SYSMAC CJ1W-NC271/NC471/NCF71/ NCF71-MA CS1W-NC271/NC471/NCF71

Position Control Units

OPERATION MANUAL



CJ1W-NC271/NC471/NCF71/NCF71-MA CS1W-NC271/NC471/NCF71 Position Control Units

Operation Manual

Revised December 2009

Notice:

OMRON products are manufactured for use according to proper procedures by a qualified operator and only for the purposes described in this manual.

The following conventions are used to indicate and classify precautions in this manual. Always heed the information provided with them. Failure to heed precautions can result in injury to people or damage to property.

- **<u>Indicates</u>** Additionally, there may be severe property damage.
- **WARNING** Indicates a potentially hazardous situation which, if not avoided, could result in death or serious injury. Additionally, there may be severe property damage.
- **Caution** Indicates a potentially hazardous situation which, if not avoided, may result in minor or moderate injury, or property damage.

OMRON Product References

All OMRON products are capitalized in this manual. The word "Unit" is also capitalized when it refers to an OMRON product, regardless of whether or not it appears in the proper name of the product.

The abbreviation "Ch," which appears in some displays and on some OMRON products, often means "word" and is abbreviated "Wd" in documentation in this sense.

The abbreviation "PLC" means Programmable Controller. "PC" is used, however, in some Programming Device displays to mean Programmable Controller.

Visual Aids

The following headings appear in the left column of the manual to help you locate different types of information.

- **Note** Indicates information of particular interest for efficient and convenient operation of the product.
- *1,2,3...* 1. Indicates lists of one sort or another, such as procedures, checklists, etc.

© OMRON, 2004

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted, in any form, or by any means, mechanical, electronic, photocopying, recording, or otherwise, without the prior written permission of OMRON.

No patent liability is assumed with respect to the use of the information contained herein. Moreover, because OMRON is constantly striving to improve its high-quality products, the information contained in this manual is subject to change without notice. Every precaution has been taken in the preparation of this manual. Nevertheless, OMRON assumes no responsibility for errors or omissions. Neither is any liability assumed for damages resulting from the use of the information contained in this publication.

1 Intended Audience. xxiv 2 General Precautions. xxiv 3 Safety Precautions. xxiv 4 Operating Environment Precautions. xxv 5 Application Precautions. xxviii 6 Conformance to EC Directives xxviii 7 SECTION 1 1 1-1 Features and System Configuration 1 1-2 System Configuration 3 1-3 Basic Operations 4 1-4 List of Functions and Specifications 6 1-5 List of Functions by Purpose. 8 1-6 Comparison with Existing Models 9 9 SECTION 2 11 2-1 Basic Flow of Operations 12 2-2 Starting Operation 16 SECTION 2 11 11 2-1 Basic Flow of Operations 12 2-2 Starting Operation 16 SECTION 3 11 31 3-1 Nomenclature and Functions 32 3-2 Installation and Wiring 3	PRF	ECAUTIONS	xxiii
2 General Precautions	1	Intended Audience	xxiv
3 Safety Precautions xxiv 4 Operating Environment Precautions xxv 5 Application Precautions xxvi 6 Conformance to EC Directives xxviii 7 SECTION 1 Features and System Configuration 1 1-1 Features and System Configuration 1 1 1-1 Features 2 2 12 System Configuration 3 3 1-3 Basic Operations and Specifications 6 1-5 List of Functions and Specifications 6 1-6 Comparison with Existing Models 9 SECTION 2 Basic Procedures 11 2-1 Basic Flow of Operations 12 2-2 Starting Operation 16 SECTION 3 Installation and Wiring 31 3-1 Nomenclature and Functions 32 3-2 Installing the Position Control Unit. 35 3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4 Data Areas 76 4-1	2	General Precautions	xxiv
4 Operating Environment Precautions xxv 5 Application Precautions xxvi 6 Conformance to EC Directives xxviii SECTION 1 Features and System Configuration 1 1-1 Features 2 1-2 System Configuration 3 1-3 Basic Operations 4 1-4 List of Functions and Specifications 6 1-5 List of Functions by Purpose 8 1-6 Comparison with Existing Models 9 SECTION 2 Basic Procedures 11 2-1 Basic Flow of Operations 12 2-2 Starting Operation 16 SECTION 3 Installation and Wiring 31 3-1 Nomenclature and Functions 32 3-2 Installation Control Unit 35 3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4 Data Areas 75 4-1 Overall Struc	3	Safety Precautions	xxiv
5 Application Precautions	4	Operating Environment Precautions	XXV
6 Conformance to EC Directives xxviii SECTION 1 1-1 Features and System Configuration 1 1-1 Features 2 1-2 System Configuration 3 1-3 Basic Operations 4 1-4 List of Functions and Specifications 6 1-5 List of Functions by Purpose 8 1-6 Comparison with Existing Models 9 SECTION 2 Basic Procedures 11 2-1 Basic Flow of Operations 12 2-2 Starting Operation 16 SECTION 2 Basic Procedures 11 2-1 Basic Flow of Operations 12 2-2 Starting Operation 16 SECTION 3 Installation and Wiring 31 3-1 Nomenclature and Functions 32 3-2 Installing the Position Control Unit. 35 3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4	5	Application Precautions	xxvi
SECTION 1 Features and System Configuration 1 1-1 Features 2 1-2 System Configuration 3 1-3 Basic Operations 4 1-4 List of Functions and Specifications 6 1-5 List of Functions by Purpose. 8 1-6 Comparison with Existing Models 9 SECTION 2 Basic Procedures 11 2-1 Basic Flow of Operations 12 2-2 Starting Operation 16 SECTION 3 Installation and Wiring 31 3-1 Nomenclature and Functions. 32 3-2 Installing the Position Control Unit. 35 3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4 Data Areas 75 4-1 Overall Structure 76 4-2 Data Areas 79 4-3 Common Parameter Area 101 4-5 Servo Parameter Area 101 4-5 Servo Parameter Area 104 4-6 Common Operating Memo	6	Conformance to EC Directives	xxviii
Features and System Configuration 1 1-1 Features 2 1-2 System Configuration 3 3 Basic Operations 4 1-4 List of Functions and Specifications 6 1-5 List of Functions by Purpose 8 1-6 Comparison with Existing Models 9 SECTION 2 Basic Procedures 11 2-1 Basic Flow of Operations 12 2-2 Starting Operation 16 SECTION 3 11 2-1 2-2 Starting Operation 16 SECTION 3 11 3-1 3-1 Nomenclature and Functions. 32 3-2 Installation and Wiring 31 3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4 10 10 Data Areas 75 4-1 Overall Structure 76 4-2 Data Areas 79 4-3 Common Parameter Area 101 4-5 Servo Parameter Area 104 <td>SEC</td> <td>CTION 1</td> <td></td>	SEC	CTION 1	
1-1 Features 2 1-2 System Configuration 3 1-3 Basic Operations 4 1-4 List of Functions and Specifications 6 1-5 List of Functions by Purpose. 8 1-6 Comparison with Existing Models 9 SECTION 2 Basic Procedures 11 2-1 Basic Flow of Operations 12 2-2 Starting Operation 16 SECTION 3 Installation and Wiring 31 3-1 Nomenclature and Functions. 32 3-2 Installing the Position Control Unit. 35 3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4 Data Areas 75 4-1 Overall Structure 76 4-2 Data Areas 79 4-3 Common Parameter Area 96 4-4 Axis Parameter Area 101 4-5 Servo Parameter Area 104 4-6 Common Operating Memory Areas	Feat	tures and System Configuration	1
1-2 System Configuration 3 1-3 Basic Operations 4 1-4 List of Functions and Specifications 6 1-5 List of Functions by Purpose. 8 1-6 Comparison with Existing Models 9 SECTION 2 Basic Procedures 11 2-1 Basic Flow of Operations 12 2-2 Starting Operation 16 SECTION 3 Installation and Wiring 31 3-1 Nomenclature and Functions. 32 3-2 Installing the Position Control Unit. 35 3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4 Data Areas 79 4-3 Common Parameter Area 96 4-4 Axis Parameter Area 101 4-5 Servo Parameter Area 104 4-6 Common Operating Memory Areas 179 4-7 Axis Operating Input Memory Areas 184	1-1	Features	2
1-3 Basic Operations 4 1-4 List of Functions and Specifications 6 1-5 List of Functions by Purpose 8 1-6 Comparison with Existing Models 9 SECTION 2 Basic Procedures 11 2-1 Basic Flow of Operations 12 2-2 Starting Operation 16 SECTION 3 Installation and Wiring 31 3-1 Nomenclature and Functions 32 3-2 Installing the Position Control Unit. 35 3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4 75 4-1 Overall Structure 76 4-2 Data Areas 79 4-3 Common Parameter Area 101 4-5 Servo Parameter Area 101 4-5 Servo Parameter Area 104 4-6 Common Operating Memory Areas 179 4-7 Axis Operating Output Memory Areas 185 4-8 Axig Operatine Input Memory Areas 185	1-2	System Configuration	3
1-4 List of Functions and Specifications 6 1-5 List of Functions by Purpose 8 1-6 Comparison with Existing Models 9 SECTION 2 Basic Procedures 11 2-1 Basic Flow of Operations 12 2-2 Starting Operation 16 SECTION 3 Installation and Wiring 31 3-1 Nomenclature and Functions 32 3-2 Installing the Position Control Unit. 35 3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4 Data Areas 75 4-1 Overall Structure 76 4-2 Data Areas 79 4-3 Common Parameter Area 101 4-5 Servo Parameter Area 101 4-6 Common Operating Memory Areas 104 4-6 Common Operating Memory Areas 104 4-6 Common Operating Memory Areas 104 4-7 Axis Operating In	1-3	Basic Operations	4
1-5 List of Functions by Purpose 8 1-6 Comparison with Existing Models 9 SECTION 2 Basic Procedures 11 2-1 Basic Flow of Operations 12 2-2 Starting Operation 16 SECTION 3 16 SECTION 3 11 3-1 Nomenclature and Functions 32 3-2 Installing the Position Control Unit 35 3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4 75 4-1 Overall Structure 0ata Areas 79 4-3 Common Parameter Area 96 4-4 Axis Parameter Area 101 4-5 Servo Parameter Area 101 4-5 Servo Parameter Area 104 4-6 Common Operating Memory Areas 179 4-7 Axis Operating Output Memory Areas 185 185 185	1-4	List of Functions and Specifications	6
1-6 Comparison with Existing Models 9 SECTION 2 Basic Procedures 11 2-1 Basic Flow of Operations 12 2-2 Starting Operation 16 SECTION 3 16 SECTION 3 31 3-1 Nomenclature and Functions 32 3-2 Installing the Position Control Unit 35 3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4 75 4-1 Overall Structure 76 4-2 Data Areas 79 4-3 Common Parameter Area 96 4-4 Axis Parameter Area 101 4-5 Servo Parameter Area 104 4-6 Common Operating Memory Areas 179 4-7 Axis Operating Output Memory Areas 185 4-8 Axis Operating Input Memory Areas 194	1-5	List of Functions by Purpose	8
SECTION 2 Basic Procedures. 11 2-1 Basic Flow of Operations 12 2-2 Starting Operation 16 SECTION 3 1 Installation and Wiring 31 3-1 Nomenclature and Functions 32 3-2 Installing the Position Control Unit 35 3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4 75 4-1 Overall Structure 76 4-2 Data Areas 79 4-3 Common Parameter Area 96 4-4 Axis Parameter Area 101 4-5 Servo Parameter Area 104 4-6 Common Operating Memory Areas 185 4-8 Axis Operating Output Memory Areas 185	1-6	Comparison with Existing Models	9
Basic Procedures 11 2-1 Basic Flow of Operations 12 2-2 Starting Operation 16 SECTION 3 16 Jastallation and Wiring 31 3-1 Nomenclature and Functions 32 3-2 Installing the Position Control Unit. 35 3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4 75 4-1 Overall Structure 76 4-2 Data Areas 79 4-3 Common Parameter Area 101 4-5 Servo Parameter Area 101 4-5 Servo Parameter Area 104 4-6 Common Operating Memory Areas 185 4-8 Axis Operating Output Memory Areas 185	SEC	CTION 2	
2-1 Basic Flow of Operations 12 2-2 Starting Operation 16 SECTION 3 Installation and Wiring 31 3-1 Nomenclature and Functions 32 3-2 Installing the Position Control Unit. 35 3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4 Data Areas 75 4-1 Overall Structure 76 4-2 Data Areas 79 4-3 Common Parameter Area 96 4-4 Axis Parameter Area 101 4-5 Servo Parameter Area 104 4-6 Common Operating Memory Area 179 4-7 Axis Operating Output Memory Areas 185 4-8 Axis Operating Input Memory Areas 194	Basi	c Procedures	11
2-2 Starting Operation 16 SECTION 3 Installation and Wiring 31 3-1 Nomenclature and Functions 32 3-2 Installing the Position Control Unit. 35 3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4 75 4-1 Overall Structure 76 4-2 Data Areas 79 4-3 Common Parameter Area 96 4-4 Axis Parameter Area 101 4-5 Servo Parameter Area 104 4-6 Common Operating Memory Areas 185 4-8 Axis Operating Output Memory Areas 185	2-1	Basic Flow of Operations	12
SECTION 3313-1Nomenclature and Functions323-2Installing the Position Control Unit.353-3External I/O Circuits393-4Wiring56SECTION 4754-1Overall Structure764-2Data Areas794-3Common Parameter Area964-4Axis Parameter Area1014-5Servo Parameter Area1044-6Common Operating Memory Areas1854-8Axis Operating Output Memory Areas194	2-2	Starting Operation	16
Installation and Wiring 31 3-1 Nomenclature and Functions 32 3-2 Installing the Position Control Unit 35 3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4 56 Data Areas 75 4-1 Overall Structure 76 4-2 Data Areas 79 4-3 Common Parameter Area 96 4-4 Axis Parameter Area 101 4-5 Servo Parameter Area 104 4-6 Common Operating Memory Areas 185 4-8 Axis Operating Output Memory Areas 185	SEC	CTION 3	
3-1 Nomenclature and Functions. 32 3-2 Installing the Position Control Unit. 35 3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4 Data Areas 75 4-1 Overall Structure 76 4-2 Data Areas 79 74 4-3 Common Parameter Area 96 4-4 Axis Parameter Area 101 4-5 Servo Parameter Area 104 4-6 Common Operating Memory Area 179 4-7 Axis Operating Output Memory Areas 185 4-8 Axis Operating Input Memory Areas 194	Inst	allation and Wiring	31
3-1 Installing the Position Control Unit. 35 3-2 Installing the Position Control Unit. 35 3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4 Data Areas 75 4-1 Overall Structure 76 4-2 Data Areas 79 74 4-3 Common Parameter Area 96 4-4 Axis Parameter Area 101 4-5 Servo Parameter Area 104 4-6 Common Operating Memory Areas 179 4-7 Axis Operating Output Memory Areas 185 4-8 Axis Operating Input Memory Areas 194	3-1	Nomenclature and Functions	32
3-3 External I/O Circuits 39 3-4 Wiring 56 SECTION 4 75 4-1 Overall Structure 76 4-2 Data Areas 79 4-3 Common Parameter Area 96 4-4 Axis Parameter Area 101 4-5 Servo Parameter Area 104 4-6 Common Operating Memory Area 179 4-7 Axis Operating Output Memory Areas 185 4-8 Axis Operating Input Memory Areas 194	3-2	Installing the Position Control Unit	35
3-4 Wiring 56 SECTION 4 75 4-1 Overall Structure 76 4-2 Data Areas 79 4-3 Common Parameter Area 96 4-4 Axis Parameter Area 101 4-5 Servo Parameter Area 104 4-6 Common Operating Memory Area 179 4-7 Axis Operating Output Memory Areas 185 4-8 Axis Operating Input Memory Areas 194	3-3	External I/O Circuits	39
SECTION 4Data Areas754-1 Overall Structure764-2 Data Areas794-3 Common Parameter Area964-4 Axis Parameter Area1014-5 Servo Parameter Area1044-6 Common Operating Memory Area1794-7 Axis Operating Output Memory Areas1854-8 Axis Operating Input Memory Areas194	3-4	Wiring	56
Data Areas754-1Overall Structure764-2Data Areas794-3Common Parameter Area964-4Axis Parameter Area1014-5Servo Parameter Area1044-6Common Operating Memory Area1794-7Axis Operating Output Memory Areas1854-8Axis Operating Input Memory Areas194	SEC	TION 4	
4-1Overall Structure764-2Data Areas794-3Common Parameter Area964-4Axis Parameter Area1014-5Servo Parameter Area1044-6Common Operating Memory Area1794-7Axis Operating Output Memory Areas1854-8Axis Operating Input Memory Areas194	Date	Areas	75
4-1Overall Structure764-2Data Areas794-3Common Parameter Area964-4Axis Parameter Area1014-5Servo Parameter Area1044-6Common Operating Memory Area1794-7Axis Operating Output Memory Areas1854-8Axis Operating Input Memory Areas194	<i>A</i> -1		76
4-2Data Areas754-3Common Parameter Area964-4Axis Parameter Area1014-5Servo Parameter Area1044-6Common Operating Memory Area1794-7Axis Operating Output Memory Areas1854-8Axis Operating Input Memory Areas194	4_2		70
4-3Common rataneter Area904-4Axis Parameter Area1014-5Servo Parameter Area1044-6Common Operating Memory Area1794-7Axis Operating Output Memory Areas1854-8Axis Operating Input Memory Areas194		Common Parameter Area	06
4-5Servo Parameter Area1044-6Common Operating Memory Area1794-7Axis Operating Output Memory Areas1854-8Axis Operating Input Memory Areas194	4-J	Axis Parameter Area	101
4-6 Common Operating Memory Area 104 4-6 Common Operating Memory Area 179 4-7 Axis Operating Output Memory Areas 185 4-8 Axis Operating Input Memory Areas 194	4-4 4_5	Servo Parameter Area	101
4-7 Axis Operating Output Memory Areas 185 4-8 Axis Operating Input Memory Areas 194	-+-J 1 6	Common Operating Memory Area	104
4-8 Axis Operating Input Memory Areas	4-0 4-7	Axis Operating Output Memory Areas	185
	4-8	Axis Operating Input Memory Areas	194

SEC	CTION 5	
Tra	nsferring and Saving Data	217
5-1	Transferring Data	218
5-2	Transferring PCU Parameters	220
5-3	Transferring Servo Parameters	226
SEC	CTION 6	
ME	CHATROLINK	237
6-1	MECHATROLINK Overview	238
6-2	MECHATROLINK Settings	239
6-3	MECHATROLINK Communications Control	248
6-4	Standard Settings for Servo Drives Using MECHATROLINK	263
SEC	CTION 7	
Posi	tion Control Structure	271
7-1	PCU Control System	272
7-2	Control Units	273
7-3	Coordinate System and Present Position	276
7-4	Acceleration and Deceleration Operations	277
7-5	Limit Input Operations	286
SEC	CTION 8	
Defi	ning the Origin	291
8-1	Overview	292
8-2	Origin Search Operation	293
8-3	Present Position Preset.	317
8-4	Origin Return	319
8-5	Phase Z Margin	324
8-6	Absolute Encoder Origin	326
SEC	CTION 9	
Posi	tioning	337
9-1	Direct Operation Overview	338
9-2	Direct Operation Procedure	339
9-3	PCU Data Settings for Direct Operation	339
9-4	Using Direct Operation	343
9-5	Interrupt Feeding	351
9-6	Torque Limit Function	354
9-7	Linear Interpolation	355

SECTION 10	
Other Operations	. 369
10-1 Servo Lock/Unlock	370
10-2 Jogging	371
10-3 Override	376
10-4 Torque Limits	377
10-5 Speed Control	385
10-6 Torque Control	398
10-7 Backlash Compensation	405
10-8 Software Limits	408
10-9 Stop Functions	412
10-10 DEVIATION COUNTER RESET	418

SECTION 11 Sample Program

Sample Programs	421
11-1 Overview	422
11-2 Basic Program Examples	423
11-3 Application Examples	448

SECTION 12 Troubleshoot

rou	Ibleshooting	479
12-1	Overview of PCU Errors	480
12-2	Troubleshooting Procedure	484
12-3	LED Error Indicators	485
12-4	Error Codes	488
12-5	Troubleshooting	501
12-6	Error Reset	506
12-7	CPU Unit Error Display.	508

SECTION 13

Maintenance and Inspection		
13-1	Inspection	510
13-2	Inspection Points	510
13-3	Handling Precautions	511
13-4	Procedure for Replacing a PCU	511

Appendices

Revi	sion History	657
Inde	×	647
F	Additional Functions for the CJ1W-NCF71-MA	643
Е	Changing to CS1W/CJ1W-NC271/471/F71 from CS1W/CJ1W-NC113/133/213/233/413/433	621
D	List of Error Codes	609
С	Operation Area I/O Allocations	597
В	List of Parameters	519
А	Performance Characteristics	515

About this Manual:

This manual describes the installation and operation of the CJ1W-NC271/NC471/NCF71/NCF71-MA and CS1W-NC271/NC471/NCF71 Position Control Units and includes the sections described below.

Please read this manual carefully and be sure you understand the information provided before attempting to install or operate the Position Control Unit. Be sure to read the precautions provided in the following section.

Precautions provide general precautions for using the Position Control Unit, Programmable Controller, and related devices.

Section 1 introduces the features of the Position Control Unit, explains the system configuration in which it is used, and also provides information on basic operations, functions and specifications.

Section 2 provides an overview of the procedures required to use the Position Control Unit.

Section 3 provides information on nomenclature and functions, and describes the procedures required for wiring and installation. Information on the MECHATROLINK-II I/F Unit is also provided.

Section 4 provides an overview of the parameter and data settings used in Position Control Unit operation and provides information on memory allocations.

Section 5 explains how to transfer and save parameters and data using the data transfer bits.

Section 6 provides an overview of MECHATROLINK communications, and includes information on settings and procedures required to use MECHATROLINK with the Position Control Unit.

Section 7 provides an overview of the control system used by the Position Control Unit, including information on the control units, coordinate system, acceleration/deceleration operations, and limit input operations.

Section 8 provides information on the various operations used to determine the origin, including origin searches, origin returns, presetting the present position, calculating phase Z margins, and using the absolute encoder.

Section 9 provides an overview of direct operation and describes the parameter settings, data settings, and procedures required to perform direct operation. Information on interrupt feeding and torque limits is also provided here.

Section 10 describes the servo lock/unlock, jogging, override, torque limits, speed control, torque control, backlash compensation, software limits, and stop functions.

Section 11 provides basic program examples and application examples for using the Position Control Unit.

Section 12 provides information on troubleshooting errors that may occur, including details on the meaning of indicator displays and error codes, and the procedures required to reset errors in the Unit or axes.

Section 13 describes methods for inspecting and maintaining the Position Control Unit and the procedure required to replace a Position Control Unit.

The *Appendices* provide information on the performance characteristics, lists of parameters, I/O allocations in the operation areas, lists of error codes, alarm/warning displays, and information required when changing to the CJ1W-NC271/NC471/NCF71/NCF71-MA or CS1W-NC271/NC471/NCF71 from a CJ1W/CS1W-NC113/133/213/233/413/433 Position Control Unit.

WARNING Failure to read and understand the information provided in this manual may result in personal injury or death, damage to the product, or product failure. Please read each section in its entirety and be sure you understand the information provided in the section and related sections before attempting any of the procedures or operations given.

Unit Versions of Position Control Units

Unit Versions

A "unit version" has been introduced to manage Position Control Units according to differences in functionality accompanying Unit upgrades.

Notation of Unit Versions on Products The unit version is given to the right of the lot number on the nameplate of the products for which unit versions are being managed, as shown below.



The unit version of Position Control Units starts with unit version 1.0.

Confirming Unit Versions with Support Software

CX-Programmer version 4.0 can be used to confirm the unit version using the *Unit Manufacturing Information*.

In the *IO Table* Window, right-click the Position Control Unit and select *Unit Manufacturing information.*

The following Unit Manufacturing information Dialog Box will be displayed.

Unit Manufacturing Information		? ×	
File Help			
Manufacturing Details Revision PCB Revision Software Revision Lot Number Manufacturing ID Serial Number Unit Ver.	A A C- 10 1.0 4 1.0		Unit version
There is no Memory Laid installed	СЭІН-Н-СРИ66	Program	

The unit version is displayed as *1.0* in the *Unit Version Number* field of the above example. Use the above display to confirm the unit version of the Unit connected online.

Using Unit Version Label A unit version label is provided with the Position Control Unit. This label can be attached to the front of the Position Control Unit to differentiate between Position Control Units with different unit versions.

Model			CJ1W-NC 71	CS1W-NC	l	
	Unit Ver. 1.0	Unit Ver. 1.1	Unit Ver. 1.2	Unit Ver. 1.3	Unit Ver. 2.0	Unit Ver. 2.1
Linear interpolation		Supported.	Supported.	Supported.	Supported.	Supported.
Absolute encoder setup function			Supported.	Supported.	Supported.	Supported.
Deviation counter reset				Supported.	Supported.	Supported.
Establishing connections even when there are unconnected axes or axes with alarms that cannot be cleared				Supported.	Supported.	Supported.
Transferring servo parameters even when there is an axis error				Supported.	Supported.	Supported.
Creating servo locks during soft- ware limit detection when an absolute encoder is used				Supported.	Supported.	Supported.
Driver main circuit OFF error detection only when the servo is locked				Supported.	Supported.	Supported.
Using Holding Area address H512 and onwards for function block address allocations				Supported.	Supported.	Supported.
Addition of supported models: SMARTSTEP Junior Servo Drives (R7D-ZN□-ML2)					Supported.	Supported.
Addition of rejoin function					Supported.	Supported.
Eliminating connection restric- tion when Servo Drive alarms occur (enabling connection when alarm A.C90 occurs)					Supported.	Supported.

Addition of origin search opera-

Addition of origin search preset

Faster setting for transfer cycle and communications cycle when setting the absolute encoder PG zero point position offset with an origin search

tion modes

function

Supported.

Supported.

Supported.

Supported.

Supported.

Upgrades Made According to Unit Versions of the Position Control Unit

Functional upgrade	Unit version 1.0	Unit version 1.1
Addition of linear interpolation function	Linear interpolation cannot be used.	Linear interpolation can be performed for positioning operations combining one or more axes. Linear interpolation can performed for up to four axes each of axes 1 to 4 and axes 5 to 8 for Servo Drive axes connected to the Posi- tion Control Unit. (Refer to <i>9-7 Linear Inter-</i> <i>polation</i>)
		polation.)

Unit Version 1.0 to Unit Version 1.1

Unit Version 1.1 to Unit Version 1.2

Functional upgrade	Unit version 1.1	Unit version 1.2
Addition of setup function for	An absolute encoder must be set up the	With Position Control Units with unit version
absolute encoders	first time it is used, when the rotation data	1.2 or later, the following operation can be
	is initialized to 0, or when the absolute	used to set up the absolute encoder.
	encoder is left for a long period of time without the battery connected.	 Special software (personal computer moni- toring software) can be connected to the
	With Position Control Units with unit ver-	Servo Drive to perform the setup operation.
	sion 1.1 or earlier, the following operation is used to set up the absolute encoder.	• When the Position Control Unit is used with a CPU Unit with unit version 3.0 or later, the
	• Special software (personal computer monitoring software) must be connected to the Servo Drive to perform the setup	absolute encoder can be set up from the program by using a function block from the OMRON FB Library.
	operation.	• The absolute encoder can be set up from the CX-Motion-NCF. (Refer to <i>8-6-4 Absolute Encoder Setup.</i>)

Unit Version 1.2 to Unit Version 1.3

Unit version 1.2	Unit version 1.3
The deviation counter in the Servo Drive cannot be reset from the Position Control Unit during position control operations.	The deviation counter in the Servo Drive can be reset from the Position Control Unit during position control operations.
	To deviation reset function in the Position Control Unit works by sending a movement command in the opposite direction and of the same size as the current position deviation so that the current command position equals the current feedback position. (Refer to <i>10-10 DEVIATION COUNTER</i>
U	the deviation counter in the Servo Drive annot be reset from the Position Control Init during position control operations.

Functional upgrade	Unit version 1.2	Unit version 1.3
Establishing connections	If any of the axes registered in the scan list	Axis operations using MECHATROLINK com-
when there are unconnected	are not connected, have the control power	munications are possible for any axes regis-
axes or axes with alarms that	supply interrupted, or have an alarm that	tered in the scan list and for which
cannot be cleared	can be reset only by cycling the power	MECHATROLINK communications have
	supply, an MLK initialization error (Unit	been started (see note) regardless of
	error code 0020 (hex) will occur after the	whether there are Servo Drive alarms.
	connections are established and opera-	If there are any axes with alarms, they will be
	tions using MECHATROLINK communica-	indicated by the Error Flags and error code in
	tions will not be possible any axes,	the Axis Operating Input Memory Areas.
	including those without errors.	If there are alarms in the Servo Drive that can
	To start MECHATROLINK communications	be cleared only by recycling the power, they
	normally, all errors must be cleared for all	will be detected as Unit errors (MLK initializa-
	axes registered in the scan list before con-	tion errors) for Units with unit version 1.1 or
	nections can be established.	earlier, but they will be detected in the individ-
		ual axis areas.
		Note If R88D-WN□-ML2 W-series Servo
		Drives (Models with Built-in MECHA-
		TROLINK-II Communications) are con-
		nected, an encoder communications
		error (A.C9 \Box) will occur in the Servo
		Drive and it will not be possible to start
		MECHATROLINK communications for
		Units with unit version 1.3 or earlier.
		(Refer to 6-3-2 MECHATROLINK Communi-
		cations Status.)
Transferring parameters when	Servo parameters cannot be transferred	Servo parameters can be transferred (i.e.,
there are axis errors	(i.e., written, read, or saved) for axes with	written, read, or saved) for axes with errors. If
	errors. The errors must first be reset to	the axis error already exists, it will not be
	clear the axis error status before Servo	overwritten even if an error occurs during
	parameters can be transferred.	parameter transfer.
		If Servo parameters are written when there is
		an axis error, be sure to confirm that the
		parameters were transferred correctly.
		(Refer to 5-3 Transferring Servo Parameters.)
Locking the servo when a	If an attempt is made to lock the Servo	The Servo can be locked at any position,
software limit is being	when an absolute encoder is used, the	regardless of the type of encoder and the
detected for a Motor with an	software limits are enabled, and the	software limit settings.
absolute encoder	present position is within the software limit	(Refer to 10-8-4 Software Limit Operation.)
	area, a software limit error will occur and	
	the Servo lock operation will be canceled.	
	To lock the Servo in the above situation,	
	the software limit must first be disabled.	

Functional upgrade	Unit version 1.2	Unit version 1.3
Detecting driver main circuit OFF errors only when the Servo is locked	Servo Drive main circuit OFF errors are detected regardless of whether the Servo is locked for the axis. Once a Servo Drive main circuit OFF error is detected, it will continue to be detected even if the error is reset until the main circuit power supply is restored.	Servo Drive main circuit OFF errors are detected only when the Servo is locked for the axis. The Position Control Unit will automatically unlock the Servo when a Servo Drive main circuit OFF error is detected, allowing the error to be cleared even while the main circuit power supply is interrupted. If an attempt is made to lock the Servo while the main circuit power supply is interrupted, a Servo Drive main circuit OFF error will be detected again.
Allocating holding addresses H512 and higher as function block addresses	The function blocks in the OMRON FB Library for the Position Control Unit cannot be used if H512 (default setting) or higher are allocated for non-holding areas of func- tion block addresses. If H512 or higher are allocated, a function block error will occur when the function block is executed. The CX-Programmer must be used to change the setting to other unused words (e.g., in the DM or EM Area).	(Refer to 12-4-2 List of Error Codes.) The function blocks in the OMRON FB Library for the Position Control Unit can be used if H512 (default setting) or higher are allocated for non-holding areas of function block addresses.

Unit Version 1.3 to Unit Version 2.0

Functional upgrade	Unit version 1.3	Unit version 2.0
Addition of applicable models	Applicable Models	Applicable Models
	 R88D-WT□W-series Servo Drives (with FNY-NS115 MECHATROLINK-II I/F Unit mounted) 	 R88D-WT W-series Servo Drives (with FNY-NS115 MECHATROLINK-II I/F Unit mounted)
	• R88D-WN□-ML2 W-series Servo Drives (Models with Built-in MECHATROLINK-II Communications)	 R88D-WN -ML2 W-series Servo Drives (Models with Built-in MECHATROLINK-II Communications) R7D-ZN -ML2 SMARTSTEP Junior Servo Drive (Models with Built-in MECHA- TROLINK-II Communications)

Functional upgrade	Unit version 1.3	Unit version 2.0
Addition of rejoin function	MECHATROLINK communications are started and stopped at the same time for all axes registered in the scan list.	The following functions are supported in addition to starting and stopping MECHA-TROLINK communications for all axes at the same time.
		 Rejoin Function An axis for which communications have been stopped, e.g., due to a communica- tions error, can be restarted without stop- ping communications for the other axes. (Refer to 6-3-4 Rejoining the Connection.) Setting the Axes to Be Connected Axes registered in the scan list can be set temporarily so that they are not registered. The axes can be set so that they are tempo- rarily not used without resetting the scan list. Operations can be performed without errors occurring for these axes. (Refer to 6-3-5 Specifying the Axes to Con- nect.) The Axis Communications Status Flags have also been changed for the above functions.
		Refer to the note following this table for details.
Eliminating connection restric- tion when Servo Drive alarms occur (enabling connection when alarm A.C90 occurs)	If an encoder communications error (A.C90) occurs for a R88D-WN□-ML2 W- series Servo Drive (Model with Built-in MECHATROLINK-II Communications), MECHATROLINK communications cannot	MECHATROLINK communications can be started under the conditions given at the left, and operations, such as transferring Servo Parameters, can be performed.
Addition of origin search oper- ation modes	 Three origin search operation pattern are possible by combining the following settings: Origin search operations: 3 settings (Reversal modes 1 and 2, and Single-direction mode) 	Eleven origin search operation pattern are possible by combining the following settings: • Origin search operations: 4 settings (Reversal modes 1, 2, and 3, and Single- direction mode)
	 Origin detection method: 1 setting (With origin proximity input signal rever- sal) 	origin proximity input signal reversal, With- out origin proximity input signal reversal, Not use origin proximity input signal)
		Origin detection method Reversal mode 1 Single- mode 2 Reversal mode 3
		With origin proximity input signal reversal (See note.)
		Without origin proximity O O O O input signal (See note.)
		Not use origin proximity input signal O O •: Combinations supported by unit version 1.3 or earlier O: Combinations supported by unit version 2.0 or later
		Note: Origin search operation patterns supported by absolute encoders. (Refer to <i>8-2-4 Origin Search Operation.</i>)

Functional upgrade	Unit version 1.3	Unit version 2.0
Addition of origin search pre-	The preset function cannot be used during	The preset function can be used during ori-
set function	origin searches.	gin searches.
		For any of the origin search operations the present position can be automatically set to any specified value at the end of the origin search. When using reversal mode 1 and an absolute encoder, an offset can also be set for the absolute origin. (Refer to 8-2-6 Origin Search Preset and 8-6-2 Absolute Encoder Operating Procedure.)

Note

Changes in Axis Communications Status Flags

The conditions for setting and resetting the Axis Communications Status Flags in word n+22 of the Common Operating Memory Area have been changed accompanying the addition of the rejoin function. New conditions are underlined in the following table.

Functional upgrade	Unit version 1.3 or earlier	Unit version 2.0 or later
Setting conditions	• The flags will turn ON when connections are made for the axes registered in the scan list and MECHATROLINK communi- cations start.	• The flags will turn ON when connections are made for the axes registered in the scan list and MECHATROLINK communica- tions start.
		• The flag will turn ON when the rejoin func- tion is used to start MECHATROLINK com- munications for an axis registered in the scan list.
Resetting conditions	• The flags will remain OFF when MECHA- TROLINK communications cannot be started when connections are made for the axes registered in the scan list.	• The flags will remain OFF when MECHA- TROLINK communications cannot be started when connections are made for the axes registered in the scan list.
	• The flags will turn OFF if MECHA- TROLINK communications stop because the axis is disconnected.	• The flags will turn OFF if MECHATROLINK communications stop because the axis is disconnected.
	 The flags will turn OFF if a Unit error occurs that requires disconnection. 	• The flags will turn OFF if a Unit error occurs that requires disconnection.
		• The flags will turn OFF whenever a commu- nications error occurs after MECHA- TROLINK communications have been started for the axis.

With unit version 1.3 or earlier, once MECHATROLINK communications have been started by establishing connections, the Axis Communications Status Flags will not change unless communications are disconnected (including Unit errors that required disconnection).

With unit version 2.0 or later, the Axis Communications Status Flags will turn OFF after connections have been established whenever axis operation becomes impossible due to a communications error (synchronous communications alarm or communications alarm).

Unit Version 2.0 to Unit Version 2.1

Functional upgrade	Unit version 2.0	Unit version 2.1
Faster setting of transfer cycle	A longer communications cycle must be	The same communications cycle can be
and communications cycle when	set using the settings given in a separate	set regardless of whether the absolute
setting the absolute encoder PG	table when the absolute encoder PG zero	encoder PG zero point position offset is
zero point position offset with an	point position offset is set with an origin	set with an origin search.
origin search	search.	

Read and Understand this Manual

Please read and understand this manual before using the product. Please consult your OMRON representative if you have any questions or comments.

Warranty and Limitations of Liability

WARRANTY

OMRON's exclusive warranty is that the products are free from defects in materials and workmanship for a period of one year (or other period if specified) from date of sale by OMRON.

OMRON MAKES NO WARRANTY OR REPRESENTATION, EXPRESS OR IMPLIED, REGARDING NON-INFRINGEMENT, MERCHANTABILITY, OR FITNESS FOR PARTICULAR PURPOSE OF THE PRODUCTS. ANY BUYER OR USER ACKNOWLEDGES THAT THE BUYER OR USER ALONE HAS DETERMINED THAT THE PRODUCTS WILL SUITABLY MEET THE REQUIREMENTS OF THEIR INTENDED USE. OMRON DISCLAIMS ALL OTHER WARRANTIES, EXPRESS OR IMPLIED.

LIMITATIONS OF LIABILITY

OMRON SHALL NOT BE RESPONSIBLE FOR SPECIAL, INDIRECT, OR CONSEQUENTIAL DAMAGES, LOSS OF PROFITS OR COMMERCIAL LOSS IN ANY WAY CONNECTED WITH THE PRODUCTS, WHETHER SUCH CLAIM IS BASED ON CONTRACT, WARRANTY, NEGLIGENCE, OR STRICT LIABILITY.

In no event shall the responsibility of OMRON for any act exceed the individual price of the product on which liability is asserted.

IN NO EVENT SHALL OMRON BE RESPONSIBLE FOR WARRANTY, REPAIR, OR OTHER CLAIMS REGARDING THE PRODUCTS UNLESS OMRON'S ANALYSIS CONFIRMS THAT THE PRODUCTS WERE PROPERLY HANDLED, STORED, INSTALLED, AND MAINTAINED AND NOT SUBJECT TO CONTAMINATION, ABUSE, MISUSE, OR INAPPROPRIATE MODIFICATION OR REPAIR.

Application Considerations

SUITABILITY FOR USE

OMRON shall not be responsible for conformity with any standards, codes, or regulations that apply to the combination of products in the customer's application or use of the products.

At the customer's request, OMRON will provide applicable third party certification documents identifying ratings and limitations of use that apply to the products. This information by itself is not sufficient for a complete determination of the suitability of the products in combination with the end product, machine, system, or other application or use.

The following are some examples of applications for which particular attention must be given. This is not intended to be an exhaustive list of all possible uses of the products, nor is it intended to imply that the uses listed may be suitable for the products:

- Outdoor use, uses involving potential chemical contamination or electrical interference, or conditions or uses not described in this manual.
- Nuclear energy control systems, combustion systems, railroad systems, aviation systems, medical equipment, amusement machines, vehicles, safety equipment, and installations subject to separate industry or government regulations.
- Systems, machines, and equipment that could present a risk to life or property.

Please know and observe all prohibitions of use applicable to the products.

NEVER USE THE PRODUCTS FOR AN APPLICATION INVOLVING SERIOUS RISK TO LIFE OR PROPERTY WITHOUT ENSURING THAT THE SYSTEM AS A WHOLE HAS BEEN DESIGNED TO ADDRESS THE RISKS, AND THAT THE OMRON PRODUCTS ARE PROPERLY RATED AND INSTALLED FOR THE INTENDED USE WITHIN THE OVERALL EQUIPMENT OR SYSTEM.

PROGRAMMABLE PRODUCTS

OMRON shall not be responsible for the user's programming of a programmable product, or any consequence thereof.

Disclaimers

CHANGE IN SPECIFICATIONS

Product specifications and accessories may be changed at any time based on improvements and other reasons.

It is our practice to change model numbers when published ratings or features are changed, or when significant construction changes are made. However, some specifications of the products may be changed without any notice. When in doubt, special model numbers may be assigned to fix or establish key specifications for your application on your request. Please consult with your OMRON representative at any time to confirm actual specifications of purchased products.

DIMENSIONS AND WEIGHTS

Dimensions and weights are nominal and are not to be used for manufacturing purposes, even when tolerances are shown.

PERFORMANCE DATA

Performance data given in this manual is provided as a guide for the user in determining suitability and does not constitute a warranty. It may represent the result of OMRON's test conditions, and the users must correlate it to actual application requirements. Actual performance is subject to the OMRON Warranty and Limitations of Liability.

ERRORS AND OMISSIONS

The information in this manual has been carefully checked and is believed to be accurate; however, no responsibility is assumed for clerical, typographical, or proofreading errors, or omissions.

PRECAUTIONS

This section provides general precautions for using the Position Control Unit and related devices.

The information contained in this section is important for the safe and reliable application of Position Control Units. You must read this section and understand the information contained before attempting to set up or operate a Position Control Unit.

1	Intende	ed Audience	xxiv
2	Genera	1 Precautions	xxiv
3	Safety	Precautions	xxiv
4	Operati	ing Environment Precautions	XXV
5	Application Precautions		
6	Confor	mance to EC Directives	xxviii
	6-1	Applicable Directives	xxviii
	6-2	Concepts	xxviii
	6-3	Conformance to EC Directives	xxviii
	6-4	Installation within Control Panels	xxviii

1 Intended Audience

This manual is intended for the following personnel, who must also have knowledge of electrical systems (an electrical engineer or the equivalent).

- Personnel in charge of installing FA systems.
- Personnel in charge of designing FA systems.
- Personnel in charge of managing FA systems and facilities.

2 General Precautions

The user must operate the product according to the performance specifications described in the operation manuals.

Before using the product under conditions which are not described in the manual or applying the product to nuclear control systems, railroad systems, aviation systems, vehicles, combustion systems, medical equipment, amusement machines, safety equipment, and other systems, machines, and equipment that may have a serious influence on lives and property if used improperly, consult your OMRON representative.

Make sure that the ratings and performance characteristics of the product are sufficient for the systems, machines, and equipment, and be sure to provide the systems, machines, and equipment with double safety mechanisms.

This manual provides information for programming and operating the Unit. Be sure to read this manual before attempting to use the Unit and keep this manual close at hand for reference during operation.

WARNING It is extremely important that a Position Control Units and related devices be used for the specified purpose and under the specified conditions, especially in applications that can directly or indirectly affect human life. You must consult with your OMRON representative before applying Position Control Units and related devices to the above-mentioned applications.

3 Safety Precautions

WARNING Do not attempt to take any Unit apart while the power is being supplied. Doing so may result in electric shock.

- WARNING Do not attempt to disassemble, repair, or modify any Units. Any attempt to do so may result in malfunction, fire, or electric shock.
- WARNING Never touch any of the terminals while power is being supplied. Doing so may result in serious electric shock.
- WARNING Provide safety measures in external circuits (i.e., not in the Programmable Controller or Position Control Unit) to ensure safety in the system if an abnormality occurs due to malfunction of the PLC, malfunction of the PCU (Position Control Unit), or external factors affecting the operation of the PLC or PCU. Not providing sufficient safety measures may result in serious accidents.

- Emergency stop circuits, interlock circuits, limit circuits, and similar safety measures must be provided in external control circuits.
- The PLC will turn OFF all outputs when its self-diagnosis function detects any error or when a severe failure alarm (FALS) instruction is executed. As a countermeasure for such errors, external safety measures must be provided to ensure safety in the system.
- The PLC or PCU outputs may remain ON or OFF due to deposits on or burning of the output relays, or destruction of the output transistors. As a countermeasure for such problems, external safety measures must be provided to ensure safety in the system.
- When the 24-V DC output (service power supply to the PLC) is overloaded or short-circuited, the voltage may drop and result in the outputs being turned OFF. As a countermeasure for such problems, external safety measures must be provided to ensure safety in the system.
- External safety measures must also be taken to ensure safety in the event of unexpected operation when connecting or disconnecting the PCU's connectors.
- Caution Execute online editing only after confirming that no adverse effects will be caused by extending the cycle time. Otherwise, the input signals may not be readable.
- Caution Confirm safety at the destination node before transferring a program to another node or changing contents of the I/O memory area. Doing either of these without confirming safety may result in injury.

4 **Operating Environment Precautions**

Caution Do not operate the control system in the following locations:

- Locations subject to direct sunlight.
- Locations subject to temperatures or humidity outside the range specified in the specifications.
- Locations subject to condensation as the result of severe changes in temperature.
- Locations subject to corrosive or flammable gases.
- Locations subject to dust (especially iron dust) or salts.
- Locations subject to exposure to water, oil, or chemicals.
- Locations subject to shock or vibration.
- Caution Take appropriate and sufficient countermeasures when installing systems in the following locations:
 - Locations subject to static electricity or other forms of noise.
 - Locations subject to strong electromagnetic fields.
 - · Locations subject to possible exposure to radioactivity.
 - Locations close to power supplies.

Caution The operating environment of the PLC System can have a large effect on the longevity and reliability of the system. Improper operating environments can lead to malfunction, failure, and other unforeseeable problems with the PLC System. Make sure that the operating environment is within the specified conditions at installation and remains within the specified conditions during the life of the system.

5 Application Precautions

Observe the following precautions when using the PLC System.

- **WARNING** Always heed these precautions. Failure to abide by the following precautions could lead to serious or possibly fatal injury.
 - Always connect to a ground of 100 Ω or less when installing the Units. Not connecting to a ground of 100 Ω or less may result in electric shock.
 - Always turn OFF the power supply to the PLC before attempting any of the following. Not turning OFF the power supply may result in malfunction or electric shock.
 - Mounting or dismounting Power Supply Units, I/O Units, CPU Units, Inner Boards, or any other Units.
 - Assembling the Units.
 - Setting DIP switches or rotary switches.
 - Connecting cables or wiring the system.
 - Connecting or disconnecting the connectors.
 - **Caution** Failure to abide by the following precautions could lead to faulty operation of the PLC, the PCU, or the system, or could damage the PLC or PCU. Always heed these precautions.
 - Fail-safe measures must be taken by the customer to ensure safety in the event of incorrect, missing, or abnormal signals caused by broken signal lines, momentary power interruptions, or other causes. Not doing so may cause malfunction resulting in serious injury.
 - Interlock circuits, limit circuits, and similar safety measures in external circuits (i.e., not in the Programmable Controller) must be provided by the customer.
 - Install external breakers and take other safety measures against short-circuiting in external wiring. Insufficient safety measures against short-circuiting may result in burning.
 - For CS-series PLCs, always tighten the mounting screw at the bottom of the PCU to a torque of 0.4 N·m.
 - For CJ-series PLCs, lock the sliders securely until they click into place when connecting the Power Supply Unit, CPU Unit, I/O Units, Special I/O Units, or CPU Bus Units. Functions may not work correctly if the sliders are not locked properly.
 - Always attach the End Cover provided with the CPU Unit to the Unit on the right end of the PLC. The CJ-series PLC will not operate properly if the End Cover is not attached.

- Take appropriate measures to ensure that the specified power with the rated voltage and frequency is supplied in places where the power supply is unstable. An incorrect power supply may result in malfunction.
- Remove the label after the completion of wiring to ensure proper heat dissipation. Leaving the label attached may result in malfunction.
- Disconnect the LG (line ground) terminal and GR (ground) terminal before performing withstand voltage and insulation resistance tests.
- Confirm that set parameters and data operate properly.
- Perform wiring according to specified procedures.
- Double-check all wiring and switch settings before turning ON the power supply. Incorrect wiring may result in burning.
- Check the user program for proper execution before actually running it on the Unit. Not checking the program may result in unexpected operation.
- Confirm that no adverse effect will occur in the system before attempting any of the following. Not doing so may result in an unexpected operation.
 - Changing the operating mode of the PLC (including setting the Startup Mode).
 - Force-setting/force-resetting any bit in memory.
 - Changing the present value of any word or any set value in memory.
- After replacing Units, resume operation only after transferring to the new CPU Unit, Special I/O Units, CPU Bus Units, and externally connected devices the contents of the DM Area, Holding Area, and other data required for resuming operation. Not doing so may result in an unexpected operation.
- Do not pull on the cables or bend the cables beyond their natural limit. Doing either of these may break the cables.
- Do not place objects on top of the cables or other wiring lines. Doing so may break the cables.
- Before touching a Unit, be sure to first touch a grounded metallic object in order to discharge any static build-up. Not doing so may result in malfunction or damage.
- Never turn OFF the power to the Unit while transferring data.

6 Conformance to EC Directives

6-1 Applicable Directives

• EMC Directives

6-2 Concepts

EMC Directives

OMRON devices that comply with EC Directives also conform to the related EMC standards so that they can be more easily built into other devices or the overall machine. The actual products have been checked for conformity to EMC standards (see the following note). Whether the products conform to the standards in the system used by the customer, however, must be checked by the customer.

EMC-related performance of the OMRON devices that comply with EC Directives will vary depending on the configuration, wiring, and other conditions of the equipment or control panel on which the OMRON devices are installed. The customer must, therefore, perform the final check to confirm that devices and the overall machine conform to EMC standards.

Note Applicable EMC (Electromagnetic Compatibility) standards are as follows:

EMS (Electromagnetic Susceptibility): EN61000-6-2 EMI (Electromagnetic Interference): EN61000-6-4 (Radiated emission: 10-m regulations)

6-3 Conformance to EC Directives

The PCUs comply with EC Directives. To ensure that the machine or device in which a PCU is used complies with EC Directives, the PCU must be installed as follows:

- *1,2,3...* 1. The PCU is defined as a in-panel device and must be installed within a control panel.
 - 2. Reinforced insulation or double insulation must be used for the DC power supplies used for I/O.
 - 3. PCUs complying with EC directives also meet the common emission standard (EN61000-6-4). The measures required to ensure that the standard is met will vary with the overall configuration of the control panel, the other devices connected to the control panel, wiring, and other conditions. The customer must therefore confirm that EC directives are met for the overall machine or device, particularly for the radiated emission requirement (10 m).

6-4 Installation within Control Panels

Unnecessary clearance in cable inlet or outlet ports, operation panel mounting holes, or in the control panel door may cause electromagnetic wave leakage or interference. In this case, the product may fail to meet EC Directives. In order to prevent such interference, fill clearances in the control panel with conductive packing. (In places where conductive packing comes in contact with the control panel, ensure electrical conductivity by removing the paint coating or masking these parts when painting.)

SECTION 1 Features and System Configuration

This section introduces the features of the Position Control Unit, explains the system configuration in which it is used, and also provides information on basic operations, functions and specifications.

1-1	Feature	°S	2
1-2	System	Configuration	3
1-3	-3 Basic Operations		4
	1-3-1	Position Control (Direct Operation)	4
	1-3-2	Speed Control and Torque Control	5
	1-3-3	Other Operations	5
1-4	List of	Functions and Specifications	6
	1-4-1	General Specifications	6
	1-4-2	List of Functions and Specifications	6
1-5	List of	Functions by Purpose	8
1-6	Comparison with Existing Models		9

1-1 Features

	CJ1W-NC271/471/F71	CS1W-NC271/471/F71
	The Position Control Unit is a CS/CJ-series trol Unit (PCU) receives commands from Area and outputs positioning commands to MECHATROLINK is a registered tradema Association.	CPU Bus Unit. The Position Con- the CPU Unit's internal Auxiliary MECHATROLINK-II Servo Drives. ark of MECHATROLINK Members
Compatible with the MECHATROLINK-II High- speed Field Network	A MECHATROLINK-II high-speed (10 Mi used to control Servo Drives for up to 16 ax Shielded twisted-pair cables in daisy-chain enable multi-axis systems that require less	ops) communications interface is es with a single CS/CJ-series Unit. formation make wiring simple and wiring and are smaller in size.
High-speed, High- precision Control Using Data Communications	Optimal motor performance can be achieve munications between the Programmable C without having to set an upper limit for the o high-precision position control using high-re	ed by transmitting data using com- controller (PLC) and Servo Drives, designated speed. High-speed and esolution motors are possible.
Position Control (Direct Operation)	Positioning can be performed simply by direct target speed from the CPU Unit. Positioning tions is also possible. Interrupt feeding is a ing, positioning is continued for a specifie signal is received, and then the axis is stop	ectly setting the target position and g to either absolute or relative posi- lso supported. With interrupt feed- ed amount after an interrupt input ped.
Speed Control and Torque Control	The Servo Drive's speed and torque can be the target speed and torque from the CPU	be controlled by directly specifying Unit.
Compatible with Servomotors with Absolute Encoders	The PCU is compatible with Servomotors the such Servomotors eliminates the need to re-	hat have absolute encoders. Using epeatedly perform origin searches.
Transfer Data between Host PLC and Servo Drive	The Servo Drive's parameters and monitors the data for the multi-axis system can be PLC. This removes the difficulty in starting replacing a Unit.	s can be set from the CPU Unit. All centrally controlled from the host g up devices or setting data when

Position Control Unit

1-2 System Configuration

The PCU receives commands from the CPU Unit's ladder program and control signal status (forward/reverse rotation limit, origin, origin proximity, and interrupt input signals) from devices connected externally to the Servo Drive, and uses them to control Servo Drive positioning.

System Configuration Example



1-3 Basic Operations

The PCU's operations are as follows:



1-3-1 Position Control (Direct Operation)

Positioning can be executed either to an absolute position (i.e., to an absolute position from the origin) or to an incremental position (i.e., to a position relative to the present position). Interrupt feeding is also possible, whereby an axis is moved a specified amount when an interrupt input signal is received and then stopped.

Absolute Movements and Relative Movements

With absolute and relative movements, position and speed data are set directly from the ladder program in the CPU Unit. Positioning is executed according to operating commands sent to the PCU from the CPU Unit. It is also possible to change the speed or to send commands to move axes to different positions while positioning is being performed.



Interrupt Feeding

When an interrupt input signal is received, positioning is continued for the specified amount of movement and then stopped.



Linear Interpolation

Linear interpolation can be performed for a combination of axes (Unit Ver. 1.1 or later).

1-3-2 Speed Control and Torque Control

Speed command data and torque command data are set from the CPU Unit. Speed control and torque control of the Servomotor are executed by sending operating commands to the PCU from the CPU Unit.

1-3-3 Other Operations

Origin Searches The origin search operations find the origin for a designated axis.

Jogging Jogging moves a specified axis at a designated speed and then stops it.

Overrides When an override is enabled during positioning, the target speed is changed to the override speed.



Present Position Preset (Changing the Present Position)

Stop Functions

The PRESENT POSITION PRESET command changes the present position to a specified position.

The DECELERATION STOP command decelerates positioning to a stop.

The EMERGENCY STOP command cancels operating commands immediately and stops the axis after moving it for the number of pulses remaining in the Servo Drive's deviation counter.

1-4 List of Functions and Specifications

1-4-1 General Specifications

Item	Specification	
Model	CJ1W-NC271/471/F71 CS1W-NC271/471/F71	
Internal current consumption	360 mA max. at 5 V DC	
Dimensions	$31 \times 90 \times 65 \text{ mm} (W \times H \times D)$ $130 \times 35 \times 101 \text{ mm} (W \times H \times D)$	
Weight	95 g max. 188 g max.	
Ambient operat- ing temperature	0 to 55°C	
Approved stan- dards	CE, cULus, and C-tick	

Specifications not listed above conform to general CS/CJ Series specifications.

1-4-2 List of Functions and Specifications

Item		Specification	
Unit classification		CPU Bus Unit	
Applicable PLCs		CS/CJ Series	
Possible unit nu	mber settings	0 to F	
I/O allocations	Common Operating Memory Area	Words allocated in CPU Bus Unit Area: 25 words (15 output words, 10 input words)	
	Axis Operating Memory Area	Allocated in one of the following areas (user-specified): CIO, Work, Auxiliary, Holding, DM, or EM Area.	
		Number of words allocated: 50 words (25 output words, 25 input words) \times Highest axis No. used	
Compatible devices		 OMRON G5-series Servo Drives (Built-in MECHATROLINK-II communications) OMRON G-series Servo Drives (Built-in MECHATROLINK-II communications) OMRON W-series Servo Drives (equipped with MECHATROLINK-II I/F Unit or built-in MECHA- TROLINK-II communications) OMRON SMARTSTEP Junior Servo Drives (Built-in MECHATROLINK-II communications) Note SMARTSTEP Junior Servo Drive are supported by Position Control Units with unit version 2.0 or later 	
Control method		Control commands executed using MECHATROLINK-II synchro- nous communications.	
Maximum numb	per of controlled axes	CS1W/CJ1W-NC271: 2 axes, CS1W/CJ1W-NC471: 4 axes, CS1W/CJ1W-NCF71: 16 axes	
Control units	Position command unit	Command unit: Depends on the Electronic Gear Setting in the Servo Parameters. Default setting: Pulses	
	Speed command unit for position control	Command units/s	
	Acceleration/deceleration speeds for position control	10,000 command units/s ²	
	Speed command unit for speed control	0.001% of the motor's momentary maximum rotation speed	
	Torque command unit for torque control	0.001% of the motor's momentary maximum torque	

List of Functions and Specifications

Item		Specification		
Control com-	Position command range	-2,147,483,648 to 2,147,483,647 (command units)		
mand range	Speed command range for position control	0 to 2,147,483,647 (command units/s)		
	Acceleration/deceleration speeds for position control	1 to 65,535 (10,000 command units/s ²)		
	Speed command range for speed control	-199.999% to 199.999% The upper limit of the speed command range depends on the specifications of the Servo Drive.		
	Torque command range for torque control	-199.999% to 199.999% The upper limit of the torque command range depends on the specifications of the Servo Drive.		
Control func- tions	Servo lock/unlock	Creates (Servo lock) or releases (Servo unlock) the position loop on the PCU.		
	Position control	Positions to an absolute position or relative position according to the target position and target speed specified from the ladder program.		
	Origin determination	 Origin search: Establishes the origin using the specified search method. 		
		• Present position preset: Changes the present position to a speci- fied position to establish the origin.		
		 Origin return: Returns the axis from any position to the estab- lished origin. 		
		 Absolute encoder origin: Establishes the origin using a Servomo- tor that has an absolute encoder, without having to use an origin search. 		
	Jogging	Outputs pulses at a fixed speed in the forward rotation or reverse rotation direction.		
	Interrupt feeding	Performs positioning by moving the axis a fixed amount when an external interrupt input is received while the axis is moving.		
	Speed control	Performs speed control by sending a command to the Servo Drive speed loop.		
	Torque control	Performs torque control by sending a command to the Servo Drive current loop.		
	Stop functions	 Deceleration stop: Decelerates the moving axis to a stop. Emergency stop: Positions the moving axis for the number of pulses remaining in the deviation counter and then stops the axis. 		
Auxiliary func- tions	Acceleration/deceleration curves	Sets one of the following: a trapezoidal (linear) curve, an exponen- tial curve, or an S-curve (moving average).		
	Torque limit	Restricts the output torque during axis operation.		
	Override	Multiplies the axis command speed by a specified ratio. Override: 0.01% to 327.67%		
	Servo parameter transfer	Reads and writes the Servo Drive parameters from the ladder pro- gram in the CPU Unit.		
	Monitoring function	Monitors the control status of the Servo Drive, such as the com- mand coordinate positions, feedback position, current speed, and torque.		
	Software limits	Limits software operation within the positioning range during position control.		
	Backlash compensation	Compensates for the amount of play in the mechanical system according to a set value.		
	Deviation counter reset	The position deviation in the Servo Drive's deviation counter can be reset to 0 (unit version 1.3 or later).		
External I/O	Position Control Unit	One MECHATROLINK-II interface port		
	Servo Drive I/O	Forward/reverse rotation limit inputs, origin proximity inputs, exter- nal interrupt inputs 1 to 3 (can be used as external origin inputs)		

Item	Specification
Self-diagnostic functions	Watchdog, flash memory check, memory corruption check
Error detection functions	Overtravel, Servo Drive alarm detection, CPU error, MECHA- TROLINK communications error, Unit setting error

1-5 List of Functions by Purpose

Purpose	Category	Name	Basic function	Details
Establishing the mechanical origin of	Origin deter- mination	Origin search	The motor is operated to estab- lished the origin.	8-2 Origin Search Operation
the machine		Present position preset	The position where the motor is stopped is set to a specified position to establish the origin.	8-3 Present Position Preset
		Origin return	The axis is returned to the estab- lished origin.	8-4 Origin Return
		Absolute encoder origin	The origin is established using a Servomotor with an absolute encoder, so origin searches are not required at machine startup.	8-6 Absolute Encoder Origin
Point-to-point (PTP) positioning	Position con- trol	Direct operation (abso- lute movement or relative movement)	The position and speed are speci- fied to perform positioning using an absolute or relative movement.	9-4 Using Direct Operation
Changing the target position and speed as required during positioning		Direct operation: Changing target position or changing target speed	The target position or target speed is changed during position- ing with direct operation.	9-4-3 Changing Tar- get Position 9-4-4 Changing Tar- get Speed
Performing position- ing for a specified distance from an external input point during positioning		Interrupt feeding	When an interrupt input signal turns ON during positioning with direct operation, operation switches to positioning for a fixed amount.	9-5 Interrupt Feeding
Performing manual feeding for adjust- ment or other pur- pose		Jogging	The axis is moved at a fixed speed in the forward rotation or reverse rotation direction.	10-2 Jogging
Reducing shock while device is oper- ating	Auxiliary functions	Acceleration/decelera- tion curves	Acceleration/deceleration is per- formed according to the basic trapezoidal curve (linear accelera- tion/deceleration), an exponential curve, or an S-curve, which greatly helps to reduce mechani- cal vibration.	7-4 Acceleration and Deceleration Opera- tions
Temporarily multiply- ing the machine's operating speed by a constant ratio to per- form startup adjust- ments		Overrides	The axis command speed is mul- tiplied by a constant ratio.	10-3 Override
Restricting output torque during con- trol operations such as pushing control		Torque limit	A constant limit is applied to the output torque of the Servomotor during positioning.	10-4 Torque Limits
Stopping the device during operation	Stop func- tion	Deceleration stop or emergency stop	The moving axis is decelerated to a stop or the axis is moved for the number of pulses remaining in the deviation counter and then stopped.	10-9 Stop Functions
Changing the Servo Drive settings from the PLC	Data transfer function	Reading/writing Servo parameters	Servo Drive parameters are read or written from the CPU Unit.	5-3 Transferring Servo Parameters
Purpose	Category	Name	Basic function	Details
---	---------------------	----------------	---	---------------------
Performing speed feeding in rotary con- trol such as sheet feeding.	Speed con- trol	Speed control	The speed command value is directly specified to control the Servomotor rotation.	10-5 Speed Control
Changing the output torque sequentially during control opera- tions such as tight- ening.	Torque con- trol	Torque control	The torque command value is directly specified to control the Servomotor's output torque.	10-6 Torque Control

1-6 Comparison with Existing Models

Functions and performance	CJ1W-NC□71 CS1W-NC□71	CJ1W-NC□13/□33 CS1W-NC□13/□33
Unit type	CPU Bus Unit	Special I/O Unit
Unit number alloca-	Unit numbers can be set from 0 to F	Unit numbers can be set from 0 to 95.
tion	(CPU Bus Units).	 One-axis and two-axis PCUs: One unit number used.
		• Four-axis PCUs: Two unit numbers used.
Control method	Commands are executed using MECHA- TROLINK-II synchronous communications.	Open-loop control is performed using a pulse train output.
Format of data exchanged between PLC and PCU	Binary (hexadecimal) Example: Present position is output to the PLC in 32-bit signed binary format.	Same as CJ1W-NC□71/CS1W-NC□71.
Position command range	-2,147,483,648 to 2,147,483,647 (Unit depends on Servo Parameters)	-1,073,741,823 to 1,073,741,823 pulses
Present position range	-2,147,483,648~2,147,483,647 (Unit depends on Servo parameters)	-2,147,483,647 to 2,147,483,647 pulses
Zone range	No zone functions	-1,073,741,823 to 1,073,741,823 pulses
Speed command range	Position control: 0 to 2,147,483,647 (command units/s) (Upper limit speed depends on Servo Drive and Servomotor.)	1 to 500,000 (unit: 1 pps)
	Speed control: -199.999% to 199.999% (percentage of Servomotor's momentary maxi- mum rotation speed) The upper limit of the speed command range depends on the specifications of the Servo Drive.	
Torque command range	-199.999% to 199.999% (percentage of Servomotor's momentary maxi- mum torque) The upper limit of the torque command range depends on the specifications of the Servo Drive.	None
Overrides	0.01% to 327.67% in increments of 0.01%	1% to 999% in increments of 1%
Memory operation function	None	Absolute/relative movement, linear interpolation, interrupt feeding, speed control, forced interrupt, and teaching

Functions and performance	CJ1W-NC□71 CS1W-NC□71	CJ1W-NC□13/□33 CS1W-NC□13/□33
Origin search	Origin search method:	Origin search methods:
	 The origin input signal is detected after the ori- gin proximity input signal turns OFF. 	 The origin input signal is detected after the ori- gin proximity input signal turns ON.
	 The origin input signal is detected after the ori- gin proximity input signal turns ON. (Unit ver- sion 2.0 or later) 	 The origin input signal is detected after the origin proximity input signal turns OFF. The origin input signal is detected without
	• The origin input signal is detected without using the origin proximity input signal. (Unit version 2.0 or later)	using the origin proximity input signal.
	Origin compensation: After detecting the origin input signal, positioning is performed for the ori- gin return final travel distance (specified in Servo Parameters).	Origin compensation: The axis is moved for the amount specified by the origin compensation data (specified from the Unit) at the proximity speed.
Acceleration/decel- eration curves	Trapezoidal curve, exponential curve, or S-curve S-curve acceleration/deceleration uses a mov- ing average.	Trapezoidal curve or S-curve S-curve acceleration/deceleration uses a ter- tiary function.
Setting acceleration/ deceleration speeds	Accelerations and decelerations are specified in units of 10,000 command units/s ² . Servo parameters are set individually for each axis.	The times in milliseconds required to reach the maximum speed from the initial speed and to reach the initial speed from the maximum speed are specified
		Direct operation: Acceleration/deceleration speeds are specified as operation data from the PLC.
		Memory operation: Up to 9 acceleration/deceler- ation speeds per axis are recorded in the Unit.
Deviation counter reset	Supported (unit version 1.3 or later).	Supported.
Emergency stop	A hardware input contact is not provided on the Position Control Unit.	The PCU's hardware input contact is used.
	Stopping is possible after moving the number of pulses remaining in the deviation counter by using an allocated operation bit.	
Data transfer method	Writes/reads using the Data Transfer Bit.	 Data can be read or written using the Data Transfer Bit.
		 Data can be read or written using the IOWR/ IORD instruction.
Saving data	Parameters can be saved to the flash memory in the PCU.	Axis Parameters and Zone Data are saved in the flash memory in the PCU.
	Servo Parameters are saved in the Servo Drive.	
CPU Unit cycle time extension for END refresh	1 ms max. per 16 axes (using the CS1/CJ1-H CPU Unit)	0.5 ms max. per PCU
Response time	4 ms max. (time from when the start commands for the ladder program are sent until the Servo Drive receives the control command when four axes are connected) (See note.)	4 ms max. (time from when the start commands for the ladder program are sent until the Position Control Unit performs pulse output when all axes of a four-axis Unit are being operated simulta- neously)

Note The response time depends on the cycle time of the PLC and the MECHA-TROLINK communications settings. The time shown in the table is the maximum value obtained when calculated according to specified measurement conditions. For details, refer to *Appendix A Performance Characteristics*.

SECTION 2 Basic Procedures

This section provides an overview of the procedures required to use the Position Control Unit.

2-1	Basic F	low of Operations	12
2-2	Starting	Operation	16
	2-2-1	Overview of Operation	16
	2-2-2	System Configuration and Wiring	16
	2-2-3	Setting the PCU	18
	2-2-4	Starting MECHATROLINK Communications	21
	2-2-5	Setting Servo Parameters	22
	2-2-6	Operating the Servomotor from the PCU	25

2-1 Basic Flow of Operations

The basic flow of Position Control Unit (PCU) operation is described in this section. The steps from installation through setting the MECHATROLINK devices are required only when installing the devices for the first time. When PCU and MECHATROLINK device settings have been completed, start operation from starting MECHATROLINK communications in the flow of operation.



Note

- Perform wiring according to instructions given in the Servomotor and Servo Drive's operation manuals.
 - (2) Refer to the CJ Series PLC Operation Manual.



	Flow of operation	Details
Servo lock	(Continued from previous page.) Turn ON the SERVO LOCK Bit in the PCU's Axis Operating Memory Area. Check that the SVON Flag indicating the Servo Drive status in the PCU's Axis Operating Memory Area is ON. The Servomotor axis operations can now be controlled from the PCU.	 4-7 Axis Operating Output Memory Areas 4-8 Axis Operating Input Memory Areas 10-1 Servo Lock/Unlock
Jogging	Set the jog speed in the speed command value of the PCU's Axis Operating Memory Area. Set the acceleration and deceleration times in the Servo parameters. Set the feed direction in the Direction Designation Bit of the PCU's Axis Operating Memory Area. Turn ON the JOG Bit in the PCU's Axis Operating Memory Area. Stop the jogging operation by turning OFF the JOG Bit in the PCU's Axis Operating Memory Area. Check whether jogging has stopped by monitoring whether the Busy Flag is OFF in the PCU's Axis Operating Memory Area.	 4-7 Axis Operating Output Memory Areas 4-8 Axis Operating Input Memory Areas 7-4 Acceleration and Deceleration Operations 10-2 Jogging
Origin determination	Set the Origin Search Speed in the speed command value of the PCU's Axis Operating Memory Area. Set the Origin Search Approach Speeds 1 and 2 in the Servo Parameters. Turn ON the ORIGIN SEARCH Bit in the PCU's Axis Operating Memory Area. Check whether the ORIGIN SEARCH operation has completed by monitoring the PCU Positioning Completed Flag and No Origin Flag in the PCU's Axis Operating Memory Area.	4-7 Axis Operating Output Memory Areas 4-8 Axis Operating Input Memory Areas 7-4 Acceleration and Deceleration Operations SECTION 8 Defining the Origin

	Flow of operation	Details
	(Continued from previous page.)	
ssitioning	Set the target position in the position command value of the PCU's Axis Operating Memory Area. Set the target speed in the speed command val- ue of the PCU's Axis Operating Memory Area. Set the acceleration and deceleration times for positioning in the Servo	 4-7 Axis Operating Output Memory Areas 4-8 Axis Operating Input Memory Areas 7-4 Acceleration and Deceleration Operations
Pc		9-1 Direct Operation Overview
Stopping operation	Start axis operation using jogging, an origin search, or direct operation. Turn ON the Deceleration Stop Bit or Emergency Stop Bit in the PCU's Axis Operating Memory Area. Set the deceleration time for a deceleration stop in the Servo Parameters. Check whether the positioning operation has stopped by monitoring whether the Stop Execution Flag is ON in the PCU's Axis Operating Memory Area.	 4-7 Axis Operating Output Memory Areas 4-8 Axis Operating Input Memory Areas 7-4 Acceleration and Deceleration Operations 10-9 Stop Functions
Servo Unlock	Turn ON the SERVO UNLOCK Bit in the PCU's Axis Operating Memory Area. Check that the SVON Flag indicating the Servo Drive status in the PCU's Axis Operating Memory Area is OFF.	4-7 Axis Operating Output Memory Areas 4-8 Axis Operating Input Memory Areas 10-1 Servo Lock/Unlock
Finishing operation	Stop Servo communications by turning OFF the CONNECT Bit (releases connection) in the PCU's Common Operating Memory Area. Turn OFF the power to the Servo Drives and external input devices. Turn OFF the power to the CPU Unit. END	4-6 Common Operating Memory Area 6-3-1 Establishing Connections 6-3-2 MECHATROLINK Communications Status

2-2 Starting Operation

Examples of operating the Servomotor using RELATIVE MOVEMENT commands for direct operation are provided in this section for first-time users of a PCU.

2-2-1 Overview of Operation

The following example is for operating the Servomotor using direct operation under the following operation conditions. An OMRON W-series Servomotor and a Servo Drive with a FNY-NS115 MECHATROLINK-II I/F Unit installed is used.

Only the minimum parameters required to operate the Servomotor are set in this example. Default settings are used for the PCU and Servo Drive parameters that are not set.



A Programming Console is used in this operation example without using a ladder program and with the PLC in PROGRAM mode to manually perform the settings and execute the start commands that are required to operate the Servomotor. The Programming Console is used to set the required data for direct operation and then turn ON the RELATIVE MOVEMENT Bit to operate the motor.

2-2-2 System Configuration and Wiring

The following system configuration is used. In this example, only the motor is operated, without using a mechanical system. The unit number of the PCU is 0, and the station address of the MECHATROLINK-II I/F Unit is 1.



The devices used in this configuration diagram example are as follows:

Device	Model
CPU Unit	CJ1H-CPU67H
Power Supply Unit	CJ1W-PA202
Position Control Unit	CJ1W-NCF71
Programming Console	C200H-PRO27
Programming Console Connecting Cable	CS1W-CN224 (2 m)
Servo Drive	R88D-WT01HL
Servomotor	R88M-W10030L
Servomotor Encoder Cable	R88A-CRWA003C (3 m)
Servomotor Power Cable	R88A-CAWA003S (3 m)
MECHATROLINK-II I/F Unit	FNY-NS115
MECHATROLINK-II Connection Cable	JEPMC-W6003-01 (Yaskawa) (1 m)
MECHATROLINK-II Terminator	JEPMC-W6022 (Yaskawa)

Each of the above devices is in its factory-shipped condition.

Setting the PCU Unit Number

Set the unit number using the rotary switch on the front of the PCU. Unit number: 0

Setting the Station Address of the MECHATROLINK-II I/F Unit

Set the station address of the MECHATROLINK-II I/F Unit using the rotary switch (SW1) on the Unit. Use the default settings for the DIP switch (SW2).

SW1: 1

SW2: Default settings (pin 1: ON; pin 2: ON; pin 3: OFF; pin 4: OFF)



Wire the Units as shown in the following diagram.



The Servo Drive's CN1 input signals depend on the input signal allocations, which are set in this operation example. Of these allocated input signals, the forward drive prohibit input (forward rotation limit input) and reverse drive prohibit input (reverse rotation limit input) are used as N.C. contacts. Therefore, connect them so that they are normally ON. The origin proximity signal and external latch inputs 1 to 3 are not used in this operation example and therefore do not need to be wired.

The above diagram shows the wiring for the PCU, Servo Drive, MECHA-TROLINK-II I/F Unit, and external control input signals at the Servo Drive. Refer to each of the CPU Unit and Servo Drive operation manuals for details on wiring the CPU Unit and Servo Drive power supply and connecting the Servo Drive and Servomotor.

2-2-3 Setting the PCU

<u>Creating I/O Tables</u> Turn ON the power to the PLC and create the I/O tables. Refer to the *CJ Series PLC Operation Manual* for details on creating I/O tables.

Setting Common Parameters Set the Common Parameters of the PCU. The minimum required Common Parameters that must be set are as follows:

- Axis Operating Output Memory Area designation
- Axis Operating Input Memory Area designation
- Scan list setting (information registered for axes connected to MECHA-TROLINK communications)

Common Parameters are transferred to the PCU using the WRITE DATA Bit in the Common Operating Memory Area. D01000 to D01011 are used for data to be transferred.

Common Parameters Set in the PCU

The following settings are made in D01000 to D01011 as the Common Parameters to be set in the PCU. Refer to the following table for setting details.

DM word	Set value	Common Parameter	Details	
D01000	00B0 hex	Axis Operating Output Memory Areas	Set the beginning word of the Axis Operating Output	
D01001	0064 hex	Beginning word of Axis Operating Output Memory Areas	Memory Areas to CIO 100. Axis 1 output words: CIO 100 to CIO 124	
D01002	00B0 hex	Axis Operating Input Mem- ory Areas	Set the beginning word of the Axis Operating Input	
D01003	01F4 hex	Beginning word of Axis Operating Input Memory Areas	Memory Area to CIO 500. Axis 1 input words: CIO 500 to CIO 524	
D01004	0040 hex	Scan list setting (axes 1 and 2)	Allocate axis 1 of the PCU to the Servo Drive.	
D01005 to D01011	Set all words to 0000 hex.	Scan list setting (axes 3 to 16)		

The above settings enable the Servo Drive connected to MECHATROLINK to be controlled from the ladder program through the I/O words that are allocated as shown in the following diagram.



Writing Common Parameters to the PCU

The Common Parameter settings in D01000 to D01011 are written to the PCU. Make the settings for transferring data to the PCU's Common Operating Memory Area as shown below.

Word	Set value	Data transfer setting name	Details
CIO 1506	000C hex	Number of write words	Number of write words: 0C hex = 12 decimal
CIO 1507	0082 hex	Write source area	Beginning word of write data:
CIO 1508	03E8 hex	Write source word	D01000
CIO 1509	1838 hex	Write destination address	Write destination address in PCU: 1838 hex = Beginning word of Common Parameter Area

Data is written to the PCU by turning ON the WRITE DATA Bit in the Common Operating Memory Area. For the PCU with unit number 0, the WRITE DATA Bit is allocated in CIO 150001. Turn ON this bit using the Programming Console.



WRITE DATA Bit

Section 2-2

While data is being written to the PCU, the Data Transferring Flag in the Common Operating Memory Area turns ON. When data has finished being written, the Data Transferring Flag turns OFF. For the PCU with unit number 0, the Data Transferring Flag is allocated in CIO 151514.



Setting Axis Parameters

In this operation example, the PCU's default settings are used for each axis parameter, so they do not need to be transferred to the PCU.

Saving PCU Settings

The Common Parameters are saved in the PCU's flash memory.

Data is saved to the PCU's flash memory by turning ON the SAVE DATA Bit in the Common Operating Memory Area. For the PCU with unit number 0, the SAVE DATA Bit is allocated in CIO 150003. Turn ON this bit using the Programming Console.



While data is being saved to flash memory, the Data Transferring Flag in the Common Operating Memory Area turns ON. When data has finished being saved, the Data Transferring Flag turns OFF. For the PCU with unit number 0, the Data Transferring Flag is allocated in CIO 151514.



Restarting the PCU After the PCU settings have been saved, restart the PCU to enable the settings. Either cycle the power to the CPU Unit, or restart the PCU. For the PCU with unit number 0, the Restart Bit is allocated in A50100.

Note Do not turn OFF the power to the PLC or restart the PCU while data is being saved to the PCU's flash memory. Doing so may corrupt the PCU's memory. Always make sure that the Data Transferring Flag is OFF before turning OFF the power to the CPU Unit or restarting the PCU.

2-2-4 Starting MECHATROLINK Communications

Communications are started with the Servo Drive connected to MECHA-TROLINK based on the Common Parameter settings in the PCU.

MECHATROLINK communications are started by turning ON the CONNECT Bit in the Common Operating Memory Area. For the PCU with unit number 0, the CONNECT Bit is allocated in CIO 150100. Turn ON this bit using the Programming Console.



When connections are established, the PCU starts communications with the MECHATROLINK devices (Servo Drives) registered in the scan list set in the Common Parameters. When communications with the registered device are normal, the corresponding bits for the axes in the Axis Communications Status of the Common Operating Memory Area are turned ON. For the PCU with unit number 0, the Axis Communications Status is allocated in the bits of CIO 1522.



When communications with the registered devices are not normal, the corresponding bits for the axes in the Axis Communications Status bits of the Common Operating Memory Area are not turned ON and an MLK initialization error (Unit error code 0020 hex) occurs in the PCU. Any axis errors that occur can be checked using the Axis Error Flags and error codes in the Axis Operating Memory Areas.

The Connection Status Flag in the Common Operating Memory Area will turn ON at the start of communications when the CONNECT Bit turns ON, regardless of whether communications with all registered devices are normal.

In this operation example, if MECHATROLINK communications are started normally, the status of each flag is as follows:

Connection status (CIO 151615): 1 (Connection established)

Axis communications status (CIO 1522): 0001 (bit 00 = Communications established with axis 1)

Unit Error Flag (CIO 151512): 0 (No error)

Unit error code (CIO 1521): 0000 (No error)

Axis Error Flag for axis 1 (CIO 50012): 0 (No error) (See note.)

Axis error code for axis 1 (CIO 504): 0000 (No error) (See note.)

Note The Axis Operating Memory Area for axis 1 depends on the Axis Operating Input Memory Area settings in the Common Parameters.

2-2-5 Setting Servo Parameters

The Servo Parameters are set in the connected Servo Drive. The following table shows an example of settings for the Servo Drive's external input signal allocations (input signal selection).

Input terminal (CN1)	Input signal name	Setting	Servo Drive default setting
40		Not used	RUN (RUN command input)
41	DEC	Origin return deceleration limit switch (origin proximity input sig- nal)	MING (gain reduction input)
42	POT	Forward drive prohibit input (forward rotation limit input signal)	Not allocated.
43	NOT	Reverse drive prohibit input (reverse rotation limit input signal)	Not allocated.
44	EXT1	External latch signal 1 (external interrupt input sig- nal 1)	RESET (alarm reset input)
45	EXT2	External latch signal 2 (external interrupt input sig- nal 2)	PCL (forward rotation cur- rent limit input)
46	EXT3	External latch signal 3 (external interrupt input sig- nal 3)	NCL (reverse rotation cur- rent limit input)

These settings are standard for input signals when an R88D-WT Servo Drive and FNY-NS115 MECHATROLINK-II I/F Unit are used together.

Transferring ServoServoParametersSAVArea

Servo Parameters are transferred to the Servo Drive via the PCU using the SAVE SERVO PARAMETER Bits in the PCU's Axis Operating Output Memory Areas. The Servo Parameters are transferred one at a time. The following three parameters must be set as shown below to allocate the above input signals.

Parameter No.	Parameter name	Set value
Pn50A	Input signal selection 1	2881
Pn50B	Input signal selection 2	8883
Pn511	Input signal selection 5	6541

The settings for the parameter number, parameter size, and write data are set in the Axis Operating Output Memory Area as Servo Parameters to be set in the Servo Drive. In this operation example, the Common Parameters are set so that the beginning word of the Axis Operating Output Memory Area for axis 1 is allocated in CIO 100. Therefore, the setting words for the Servo Parameters for axis 1 are allocated as follows:

Word Details		
CIO 117	Servo Parameter No.	
CIO 118	Parameter size (Unit: bytes)	
CIO 119	Write data (rightmost word)	
CIO 120	Write data (leftmost word)	

Writing Servo Parameters to the Servo Drive

The Servo Parameter settings in CIO 117 to CIO 120 are written to the Servo Drive. In this example, to transfer three Servo Parameters, execute the operation to write to the Servo Drive three times.

Writing the Pn50A Set Value

To write the Pn50A settings, first make the settings for transferring Servo Parameters to the PCU's Axis Operating Output Memory Areas as shown below.

Word	Set value	Parameter transfer setting	Details
CIO 117	050A hex	Servo Parameter No.	Write Servo Parameter No.: Pn50A
CIO 118	0002	Parameter size	Write parameter size: 2 (bytes)
CIO 119	2881 hex	Write data (rightmost word)	Write Servo Parameter set value: 2881
CIO 120		Write data (leftmost word)	Parameter size is 2 bytes (1 word, so this is not used.) The set value is ignored.

In this example, to keep the Servo Parameters even if the Servo Drive power is turned OFF, the Servo Parameters are saved in the non-volatile memory (flash memory) in the Servo Drive. Servo Parameters are written from the PCU to the non-volatile memory (flash memory) of the Servo Drive by turning ON the SAVE SERVO PARAMETER Bit in the Axis Operating Output Memory Area.

In this operation example, the Common Parameters are set so that the beginning word of the Axis Operating Output Memory Area for axis 1 is allocated in CIO 100. Therefore, the SAVE SERVO PARAMETER Bit for axis 1 is allocated in CIO 10114. Turn ON this bit using the Programming Console.



While Servo Parameters are being saved to the Servo Drive, the Servo Parameter Transferring Flag in the Axis Operating Input Memory Area turns ON. When the Servo Parameters have finished being saved, the Servo Parameter Transferring Flag turns OFF.

In this operation example, the Common Parameters are set so that the beginning word of the Axis Operating Input Memory Area for axis 1 is allocated in CIO 500. Therefore, the Servo Parameter Transferring Flag for axis 1 is allocated in CIO 50014.

Section 2-2



Servo Parameter Transferring Flag

The set values for Pn50B and Pn511 are written in the same way, i.e., by changing the details of the Servo Parameters to be transferred and turning ON the WRITE DATA Bit.

Writing the Pn50B Set Value

Word	Set value	Parameter transfer setting	Details
CIO 117	050B hex	Servo Parameter No.	Write Servo Parameter No.: Pn50B
CIO 118	0002 hex	Parameter size	Write parameter size: 2 (bytes)
CIO 119	8883 hex	Write data (rightmost word)	Write Servo Parameter setting: 8883
CIO 120		Write data (leftmost word)	Parameter size is two bytes (one word, so this is not used.) The set- ting is ignored.

Writing the Pn511 Setting

Word	Set value	Parameter transfer setting	Details
CIO 117	0511 hex	Servo Parameter No.	Write Servo Parameter No.: Pn511
CIO 118	0002 hex	Parameter size	Write parameter size: 2 (bytes)
CIO 119	6541 hex	Write data (rightmost word)	Write Servo Parameter setting: 6541
CIO 120		Write data (leftmost word)	Parameter size is two bytes (i.e., one word, so this is not used.) The setting is ignored.

The following diagram shows the operation used to transfer parameters to the Servo Drive.

Axis Operating ()utput Memo	CPU Unit		PCU		Servo	Drive (Axis 1)
(Axis 1) CIO 101			Servo Parameters	5		Neg	
			saveu.		MECHATROLINK	Non-v	
CIO 117	050A hex	Servo parameter N	lo. (Pn50A)			<u> </u>	
CIO 118	0002 hex	Parameter length (2 bytes)			Pn50A	2881
CIO 119	2881 hex	Set value (2881)					
CIO 120		-					
Axis Operating In (Axis 1) CIO 500 Servo Pa	nput Memory 14	/ Area) while transferring)			

Ending MECHATROLINK Communications

The Servo Parameters written to the Servo Drive consist of online and offline parameters. Online parameters are enabled as soon as they are written, whereas offline parameters are not.

The input signal selection parameters set here are offline parameters that are enabled by cycling the power to the Servo Drive or executing the device setup operation. In this example, the Servo Drive power is cycled. First, stop MECHATROLINK communications before turning OFF the power supply to the Servo Drive.

MECHATROLINK communications are stopped by turning OFF the CON-NECT Bit in the Common Operating Memory Area. For the PCU with unit number 0, the CONNECT Bit is allocated in CIO 150100. Use the Programming Console to turn OFF this bit, which was turned ON at the start of MECHATROLINK communications.



When MECHATROLINK communications stop, the Connection Status Flag in the Common Operating Memory Area turns OFF. For the PCU with unit number 0, the Connection Status Flag is allocated in CIO 151615.



Turning the Servo Drive Power OFF and ON Again

After writing Servo Parameters, and when MECHATROLINK communications have stopped, enable the Servo Parameters by cycling the power to the Servo Drive. This procedure completes settings for the PCU and Servo Drive. Unless changes in installation, wiring, or parameter settings are required, the above operation does not need to be performed a second time.

2-2-6 Operating the Servomotor from the PCU

Starting MECHATROLINK Communications

Perform the operations described in *2-2-4 Starting MECHATROLINK Communications* and then start communications with the Servo Drive connected to MECHATROLINK.

Servo Lock The servo lock of the Servomotor connected to MECHATROLINK can be set by turning ON the SERVO LOCK Bit in the Axis Operating Output Memory Area.

In this operation example, the Common Parameters are set so that the beginning word of the Axis Operating Output Memory Area for axis 1 is allocated in CIO 100. Therefore, the SERVO LOCK Bit for axis 1 is allocated in CIO 10100. Turn ON this bit using the Programming Console.



- SERVO LOCK Bit

When the servo lock operation is performed, the Servomotor is placed in servo lock status. The SVON (Servo ON) Flag indicating servo status in the Axis Operating Input Memory Area turns ON when an R88D-WT Servo Drive and FNY-NS115 MECHATROLINK-II I/F Unit are used together.

In this operation example, the Common Parameters are set so that the beginning word of the Axis Operating Input Memory Area for axis 1 is allocated in CIO 500. Therefore, the SVON Flag for axis 1 is allocated in CIO 50103.



When the servo lock is no longer required, turn OFF the corresponding SERVO LOCK Bit in the Axis Operating Output Memory Area.

Positioning Using the Direct Operation RELATIVE MOVEMENT Command

Positioning can be performed for axis 1 using the RELATIVE MOVEMENT command for direct operation. The RELATIVE MOVEMENT command for direct operation sends information on the target position and target speed to the Axis Operating Output Memory Area when the RELATIVE MOVEMENT Bit turns ON.

In this operation example, the Common Parameters are set so that the beginning word of the Axis Operating Output Memory Area for axis 1 is allocated in CIO 100. Therefore, the target position and target speed are set as follows:

Word	Set value	Parameter transfer setting	Details
CIO 102	A120 hex	Position command value (rightmost word)	Target position: 0007 A120 hex = 500,000 pulses
CIO 103	0007 hex	Position command value (leftmost word)	
CIO 104	86A0 hex	Speed command value (rightmost word)	Target speed: 0001 86A0 hex = 100,000 pulses/s
CIO 105	0001 hex	Speed command value (leftmost word)	

The RELATIVE MOVEMENT Bit for axis 1 is allocated in CIO 10004. Therefore, turn this bit ON using the Programming Console.



The Servomotor starts rotating up to the target speed of 100,000 pulses/s and stops at 500,000 pulses. The acceleration/deceleration speed depends on the Servo Parameters set for the Servo Drive. When a W-series Servo Drive and the FNY-NS115 are used together, the acceleration and deceleration speeds are set in the Servo Parameters as follows:

Parameter No.	Parameter name	Default	Unit	Setting in this example
Pn80A	First-step linear acceleration con- stant	100	10,000 command units/s ²	Not used.
Pn80B	Second-step linear acceleration con- stant	100	10,000 command units/s ²	1,000,000 pulses/s ²
Pn80C	Acceleration con- stant switching speed	0	100 command units/s	Default settings used.
Pn80D	First-step linear deceleration con- stant	100	10,000 command units/s ²	Not used.
Pn80E	Second-step linear deceleration con- stant	100	10,000 command units/s ²	1,000,000 pulses/s ²
Pn80F	Deceleration con- stant switching speed	0	100 command units/s	Default settings used.

In this operation example, the default settings are used for the acceleration/ deceleration constants. With the default settings, the acceleration/deceleration movement uses a linear acceleration/deceleration waveform, and the slope of the acceleration and deceleration is 1,000,000 pulses/s². The target speed is 100,000 pulses/s, so after starting, the motor accelerates up to the target speed in 0.1 s, and decelerates to a stop from the target speed in 0.1 s.



In this operation example, the acceleration/deceleration constant switching speed is 0. Therefore, from startup, the motor accelerates according to the Second-step Linear Acceleration Constant and decelerates using the Second-step Linear Deceleration Constant until stopping. The First-step Acceleration/Deceleration Constants are not used.

The present position of each axis can be monitored in the Axis Operating Input Memory Areas.

In this operation example, the Common Parameters are set so that the beginning word of the Axis Operating Input Memory Area for axis 1 is allocated in CIO 500. Therefore, the present position (feedback present position) for axis 1 is allocated in CIO 506 and CIO 507.

CIO 506	Feedback present position (rightmost word)	
CIO 507	Feedback present position (leftmost word)	

Starting Operation



The following diagram shows the operation for the RELATIVE MOVEMENT command.

Servo Unlock

The Servomotor connected to MECHATROLINK can be set to servo unlock status by turning ON the SERVO UNLOCK Bit in the Axis Operating Output Memory Area.

In this operation example, the Common Parameters are set so that the beginning word of the Axis Operating Output Memory Area for axis 1 is allocated in CIO 100. Therefore, the SERVO UNLOCK Bit for axis 1 is allocated in CIO 10101. Turn ON this bit using the Programming Console.



When the servo unlock operation is performed, the Servomotor is placed in servo unlock status. The SVON (Servo ON) Flag indicating servo status in the Axis Operating Memory Areas turns OFF when using an R88D-WT Servo Drive with a FNY-NS115 MECHATROLINK-II I/F Unit.

In this operation example, the Common Parameters are set so that the beginning word of the Axis Operating Input Memory Area for axis 1 is allocated in CIO 500. Therefore, the SVON Flag for axis 1 is allocated in CIO 50103.



When the servo unlock status is no longer required, turn OFF the corresponding SERVO UNLOCK Bit in the Axis Operating Output Memory Area.

Stopping MECHATROLINK Communications

Always stop MECHATROLINK communications before turning OFF the power supply to the Servo Drive. MECHATROLINK communications are stopped by turning OFF the CONNECT Bit in the Common Operating Memory Area.

For the PCU with unit number 0, the CONNECT Bit is allocated in CIO 150100. Use the Programming Console to turn OFF this bit, which was turned ON at the start of MECHATROLINK communications.



When MECHATROLINK communications stop, the Connection Status Flag in the Common Operating Memory Area turns OFF. For the PCU with unit number 0, the Connection Status Flag is allocated in CIO 151615.



This completes the operations example for operating the Servomotor using the RELATIVE MOVEMENT command for direct operation. In this operation example, the commands are sent manually from the Programming Console, but the basic operation flow is the same when sequences are programmed into the ladder program. Other functions are also used in the same way by changing the parameter settings and manipulating bits.

SECTION 3 Installation and Wiring

This section provides information on nomenclature and functions, and describes the procedures required for wiring and installation. Information on the MECHATROLINK-II I/F Unit is also provided.

3-1	Nomen	clature and Functions	32
	3-1-1	Nomenclature	32
	3-1-2	MECHATROLINK-II I/F Unit for W-series Servo Drives	33
3-2	Installir	g the Position Control Unit	35
	3-2-1	System Configuration Precautions	35
	3-2-2	Unit Installation	35
	3-2-3	Installation Precautions	37
	3-2-4	Dimensions	38
3-3	Externa	1 I/O Circuits	39
	3-3-1	PCU I/O Signals	40
	3-3-2	G5-series Servo Drive I/O Signals (R88D-KN ⁻ ML2 with Built-in MECHATROLINK-II Communications)	40
	3-3-3	G-series Servo Drive I/O Signals (R88D-GN ⁻ ML2 with Built-in MECHATROLINK-II Communications)	43
	3-3-4	W-series Servo Drive I/O Signals (R88D-WT with FNY-NS115)	47
	3-3-5	W-series Servo Drive I/O Signals (R88D-WND-ML2 with MECHATROLINK-II Built-in Communications)	50
	3-3-6	SMARTSTEP Junior Servo Drive I/O Signals (R7D-ZN -ML2 with Built-in MECHATROLINK-II Communications)	53
3-4	Wiring		56
	3-4-1	MECHATROLINK-II Communications Wiring	56
	3-4-2	Wiring the Servo Drive I/O Signals	58

3-1 Nomenclature and Functions

3-1-1 Nomenclature



LED Indicators

LED	Name	Color	Status	Details
RUN	Run	Green	Lit	The PCU is operating normally.
			Not lit	Other condition
ERC	C Unit Error		Lit	A fatal error has occurred in the PCU and operation cannot con- tinue.
			Flashing	A non-fatal error has occurred in the PCU and operation can con- tinue.
			Not lit	Other condition
ERH	CPU Unit Error	Red	Lit	An error has occurred in the PLC.
			Not lit	Other condition
ERM	MECHA- TROLINK	IECHA- Red ROLINK	Lit	An error has occurred in MECHA- TROLINK communications.
	Device Error		Flashing	An error has occurred in a con- nected MECHATROLINK device.
			Not lit	Other condition
MLK	MECHA- TROLINK Com-	Yellow	Lit	MECHATROLINK communica- tions in progress
	munications Status		Not lit	MECHATROLINK communica- tions stopped

For details on errors, refer to SECTION 12 Troubleshooting.

Unit Number Setting Switch (UNIT No.) Set the unit number using the rotary switch on the front of the PCU. The PCU's unit number is a CPU Bus Unit unit number.

CJ1W-NC□71 CS1W-NC□71

The factory default setting is 0.



No.

UNIT

Setting range: 0 to F (Unit numbers 0 to 15)

Note

The unit number setting determines which words are allocated to the PCU in the CPU Bus Unit Area within the CPU Unit's CIO Area.

The PCU uses this allocated area as the "Common Operating Memory Area." For details, refer to *4-6 Common Operating Memory Area*.

Unit number	Allocated words	Unit number	Allocated words
0	1500 to 1524	8	1700 to 1724
1	1525 to 1549	9	1725 to 1749
2	1550 to 1574	10	1750 to 1774
3	1575 to 1599	11	1775 to 1799
4	1600 to 1624	12	1800 to 1824
5	1625 to 1649	13	1825 to 1849
6	1650 to 1674	14	1850 to 1874
7	1675 to 1699	15	1875 to 1899

Note Always turn OFF the power supply before changing the Unit Number Setting Switch's setting.

MECHATROLINK-II Communications Connector This connector connects the PCU with MECHATROLINK devices through the special MECHATROLINK-II Connection Cable.

For details on MECHATROLINK-II Connection Cable models and configuration, refer to *3-4-1 MECHATROLINK-II Communications Wiring*.

3-1-2 MECHATROLINK-II I/F Unit for W-series Servo Drives

The following MECHATROLINK-II I/F Unit must be mounted to a an R88D-WT W-series Servo Drive without built-in MECHATROLINK-II communications to enable connection to the PCU through MECHATROLINK-II.

Name	Model number
MECHATROLINK-II I/F Unit	FNY-NS115

When a MECHATROLINK-II I/F Unit must be mounted to a W-series Servo Drive, use the following device versions.

Device	Compatible versions	
W-series Servo Drive	Ver. 39 or later	
MECHATROLINK-II I/F Unit	VER. $\Box\Box$ 03 or later	

The versions of both the W-series Servo Drive and MECHATROLINK-II I/F Unit can be found on the nameplate on the side of each device. If an earlier version of the device is used, it will not function properly. Always use products with versions listed in the table above (or later versions).

Nomenclature This section provides a basic description of the FNY-NS115 MECHA-TROLINK-II I/F Unit's LED Indicators and Setting Switches.

> For details, refer to the Yaskawa Electric Corporation JUSP-NS115 MECHA-TROLINK-II I/F Unit User's Manual.

Note Refer to the user's manual for the Servo Drive for the nomenclature and functions of Servo Drives with Built-in MECHATROLINK-II Communications.

The LED Indicators show the operating status of the FNY-NS115.

- G5-series Servo Drives with Built-in MECHATROLINK-II Communications (R88D-KN
 –ML2)
- G-series Servo Drives with Built-in MECHATROLINK-II Communications (R88D-GN□-ML2)
- W-series Servo Drives with Built-in MECHATROLINK-II Communications (R88D-WN□-ML2)
- SMARTSTEP Junior Servo Drives with Built-in MECHATROLINK-II Communications (R7D-ZN□-ML2)

LED Indicators

\bigcirc	A
0	R

LED	Name	Color	Status	Details		
А	Alarm Status	Red	Lit	An alarm occurred in the Servo Drive. (See note.)		
			Not lit	Other condition		
R	MECHA- TROLINK-II	Green	Lit	MECHATROLINK communica- tions in progress		
	Communica- tions Status		Not lit	MECHATROLINK communica- tions stopped		

Note

Station Address Setting Switch (SW1) The Alarm Status LED will also be lit when MECHATROLINK communications are not established with the PCU.

Set the Servo Drive's station address in conjunction with the MECHA-TROLINK-II communications setting on pin 3 of the Communications Setting DIP Switch (SW2).



Setting Range: 0 to F (See the following table.)

Note

e The factory default setting is 1.

Pin 3 of SW2	SW1	Station address	Note
OFF	0		Cannot be used. Do not set.
	1 to F	1 to 15	
ON	0	16	
	1 to F		Cannot be used when connecting to the PCU. Do not set.

Communications Setting DIP Switch (SW2)

Sets the MECHATROLINK-II communications settings.



Pin	Function	Setting	Contents	Default setting	Note	
1	Baud rate	OFF	4 Mbps	ON	Turn ON this pin (10	
		ON	10 Mbps		Mbps) when connect- ing to the PCU.	
2	Transmission	OFF	17 bytes	ON	Turn ON this pin (32	
	bytes	ON 32 bytes (See note.)			bytes) when connect- ing to the PCU.	
3	Station address	OFF	1 to 15	OFF	See the explanation of	
		ON	16 to 30		SW1.	
4	Reserved by the system.	OFF		OFF	Leave this pin OFF.	

- * In some devices, the number of transmission bytes is expressed as "30 bytes," but the meaning is the same as this 32-byte setting.
- **Note** The MECHATROLINK-II I/F Unit can be ordered from OMRON with the following model number.

Name	Yaskawa Electric Co. model number	OMRON model number		
MECHATROLINK-II I/F Unit	JUSP-NS115	FNY-NS115		

3-2 Installing the Position Control Unit

3-2-1 System Configuration Precautions

- The I/O words allocated to the PCU as a CPU Bus Unit are not determined by the Unit's mounting order, but by the unit number set on the Unit Number Setting Switch on the front of the Unit.
- The PCU can be mounted in either the CPU Rack or an Expansion Rack (only up to 10 Units per Rack for a CJ-series PLC) and up to 16 Units can be controlled by one CPU Unit.
- The CS1W-NC 71 can be mounted to a CS1W-BC 3 CPU Backplane or a CS1W-BI 3 Expansion Backplane.

3-2-2 Unit Installation

Use the following procedure to install the PCU.

CJ1W-NC 71

1,2,3... 1. Align the connectors correctly and mount the PCU.



2. Secure the PCU by sliding the yellow latches on the top and bottom until they click and lock.



Note If the latches are not completely locked, the PCU may not function properly. To remove the PCU, slide the latches in the "release" direction and remove the PCU.

CS1W-NC 71

1,2,3... 1. Catch the hook on the top back of the PCU on the Backplane to mount the unit.





2. Insert the connector on the back of the CPU properly into the connector on the Backplane.



- 3. Tighten the screw on the bottom of the PCU using a Phillip's screwdriver. When mounting the PCU inside equipment, provide the minimum space indicated in the diagram to enable mounting/dismounting the PCU and to ensure proper ventilation.
- Note Always tighten the mounting screw on the bottom of the PCU to a torque of $0.4 \text{ N}\cdot\text{m}$.

To remove the PCU, loosen the screw at the bottom of the PCU using a Phillip's screwdriver and then lift up on the bottom of the PCU.



3-2-3 Installation Precautions

- Always turn OFF the CPU Unit's power supply before connecting or disconnecting cables or the Unit itself.
- To minimize the effects of noise, place I/O wiring in a separate duct from high-voltage lines and power lines.
- Wire strands may be scattered around during wiring, so leave the protective label on top of the PCU to prevent any wire strands from getting inside the PCU. Once the wiring has been completed, be sure to remove the label to allow ventilation.



3-2-4 Dimensions

CJ1W-NC 71



External I/O Circuits

CS1W-NC071



3-3 External I/O Circuits

This section describes the external I/O when a Position Control Unit is used with any of the following Servo Drives.

- G-series Servo Drives (R88D-GN -ML2 with built-in MECHATROLINK-II communications)
- W-series Servo Drives (equipped with R88D-WT and FNY-NS115)
- W-series Servo Drives (R88D-WN□-ML2 with built-in MECHATROLINK-II communications)
- SMARTSTEP Junior Servo Drives (R7D-ZN□-ML2 with built-in MECHA-TROLINK-II communications)

3-3-1 PCU I/O Signals

MECHATROLINK Connector (MLK)

Connector specifications	Explanation								
Name	ML	K		MEC	HATROLII	NK-II c	connector		
Connector used	USB connector			DUSB-ARA41-T11 (made by DDK) or equivalent					
Applicable connector	USB connector			DUS inclu	DUSB-APA41-B1-C50 (made by DDK), including shell				
Pin arrangement									
	1			Pin	Name	I/O	Description		
	'			1	(NC)				
		▋▙▌▕▕▕▖▎▌		2	SRD-	I/O	Send/receive data -		
		▋┟┛╴╡╶║५╶╽		3	SRD+	1/0	Send/receive data +		
				4	(NC)				
	4			Shell	Shield		Shield ground		

3-3-2 G5-series Servo Drive I/O Signals (R88D-KND-ML2 with Built-in MECHATROLINK-II Communications)

This section describes the standard I/O signals used with a Position Control Unit when using a G5-series Servo Drive with built-in MECHATROLINK-II communications.

Use the OMNUC G5 Series AC Servomotors/Servo Drives with Built-in MECHATROLINK-II Communication User's Manual (Cat No. I566) together with this manual for information on I/O signals.

Terminal Arrangement of the Control I/O Connector (CN1)

When using G5-series Servo Drives with built-in MECHATROLINK-II communications, the default control I/O signal allocations are the standard Servo Drive settings for using MECHATROLINK.

The following diagram shows the terminal arrangement of the Servo Drive's Control I/O Connector (CN1) when MECHATROLINK is being used with the Servo Drive's default settings.

This diagram shows only the I/O signals used when connecting to the PCU.

For details of the Servo Drive's standard settings, refer to 6-4 Standard Settings for Servo Drives Using MECHATROLINK.

External I/O Circuits

			1	+24VIN	12 to 24-VDC]			19	POT	Forward Drive
2	STOP	Emergency			Input	20	NOT	Reverse Drive			Prohibit Input
		Stop Input	3	EXT3	External Latch				21	DEC	Origin Proximity
4	EXT2	External Latch			Signal 3	22	IN0	External General-			Input
		Signal 2	5	EXT1	External Latch			purpose Input 0	23	IN2	External Gener-
6	IN1	External General-purpose			Signal	24		*			put 2
		Input 1	7	PCL	Forward				25		*
8	NCL	Reverse Torque			Input	26		*			
		Limit Input	9		*				27		*
10		*				28		*			
			11		*				29	OUTM2	General-pur-
12		*				30	OUTM2	General-			
			13		*		COM	Output 2	31	OUTM3	General-pur-
14		*				32	OUTM3	General-			
			15	/ALM	Alarm Output		COM	Output 3	33	BATCOM	Backup Battery
16	ALMCOM	Alarm Output				34	BAT	Backup Battony Input			Input
			17		*				35	OUTM1COM	General-pur-
18		*				36	OUTM1	General-			
								Output 1			

Note

(1) Do not connect anything to unused pins (*).

(2) Inputs for pins 1 to 8 are determined by parameter settings. The diagram shows the default configuration.

CN1 Connector (26 Pin)

Name	Model	Manufacturer		
Plug	10126-3000PE	Sumitama 2M		
Case	10326-52A0-008			

Control I/O Signals

The following table shows the names and functions of the Servo Drive's control I/O signals.

Section 3-3

CN1 Control Input Signals

Pin No.	Symbol	Signal name		Function/Interface			
		Name	Default setting				
6	+24VIN	12 to 24-VDC Po	wer Supply Input	Power supply +input terminal (12 to 24 VDC) for sequence inputs.			
5	IN1	General-pur-	Immediate stop	This is a general-purpose input.			
		pose input 1		Functions can be selected by parameters.			
7	IN2	General-pur- pose input 2	Forward drive prohibition	Take note that, however, external latch inputs 1 to 3 can only be allocated to IN5 to IN7 (pin			
8	IN3	General-pur- pose input 3	Reverse drive prohibition	numbers 10 to 12)			
9	IN4	General-pur- pose input 4	Origin proximity				
10	IN5	General-pur- pose input 5	External latch 3				
11	IN6	General-pur- pose input 6	External latch 2				
12	IN7	General-pur- pose input 7	External latch 1				
13	IN8	General-pur- pose input 8	Monitor input 0				
14	BAT	Backup battery input		Backup battery connection terminals when the			
15	BATGND	ABS		absolute encoder power is interrupted.			
				(Connection to this terminal is not necessary if you use the absolute encoder battery cable for backup.)			

CN1 Control Input Signal Connection Diagram



Note

(1) If the limit input signal inputs (Servo Drive's forward drive prohibited signal and reverse drive prohibited signal) are not allocated, the Servo Drive will not stop the Servomotor when the signal is input, and the Position Control Unit will also not detect limit inputs as errors. When using a Position Control Unit, always allocate the Servo Drive's forward drive prohibited signal and reverse drive prohibited signal to enable use of the limit input signals. (Refer to 6-4 Standard Settings for Servo Drives Using MECHATROLINK.)

(2) The signal width of the limit input signals (forward drive prohibited signal and reverse drive prohibited signal) and origin proximity input signal (origin return deceleration limit switch) must be longer than the MECHA-

TROLINK communications cycle. If the input signal width is shorter than the communications cycle, the Position Control Unit will be unable to detect the input signal.

(3) When selecting a sensor for the origin proximity input signal (origin return deceleration LS), use a sensor such as a Photoelectric Sensor, which does not have chattering, because the origin signal is detected after the input goes from ON to OFF during the origin search. If a switch with contacts is used, the origin position may shift due to the switch contact's chattering.

CN1 Control Output Signals

Pin No.	Symbol	Signa	l name	Function/Interface
		Name	Default setting	
3	/ALM	Alarm Output		The output is OFF when an
4	ALMCOM			alarm is generated for the Servo Drive.
1	OUTM1	General-pur-	Brake interlock	This is a general-purpose out-
2	OUTM1COM	pose output 1		put.
25	OUTM2	General-pur- Servo ready		Functions can be selected by
26	OUTM2COM	pose output 2	completed out- put	parameters.
16	GND	Signal ground		This is a signal ground.

Control Output Circuits



3-3-3 G-series Servo Drive I/O Signals (R88D-GN_-ML2 with Built-in MECHATROLINK-II Communications)

This section describes the standard I/O signals used with a Position Control Unit when using a G-series Servo Drive with Built-in MECHATROLINK-II Communications.

Use the OMNUC G Series AC Servomotors/Servo Drives with Built-in MECHATROLINK-II Communications User's Manual (Cat. No. 1566) together with this manual for information on I/O signals.

Terminal Arrangement of the Control I/O Connector (CN1)

When using G-series Servo Drives with built-in MECHATROLINK-II communications, the default control I/O signal allocations are the standard Servo Drive settings for using MECHATROLINK.

The following diagram shows the terminal arrangement of the Servo Drive's Control I/O Connector (CN1) when MECHATROLINK is being used with the Servo Drive's default settings.

This diagram shows only the I/O signals used when connecting to the PCU.

External I/O Circuits

Section 3-3

For details on the Servo Drive's standard settings, refer to 6-4 Standard Settings for Servo Drives Using MECHATROLINK.

	1	1	1	+24VIN	12 to 24-VDC Power Supply				19	POT	Forward Drive
2	STOP	Emergency			Input	20	NOT	Reverse Drive			
		Stop Input	3	EXT3	External Latch			Prohibit Input	21	DEC	Origin Proximity
4	EXT2	External Latch			Signal 3	22	IN0	External General-			Input
		Signal 2	5	EXT1	External Latch			purpose Input 0	23	IN2	External Gener-
6	IN1	External			Signal 1	24		*			put 2
		Input 1	7	PCL	Forward				25		*
8	NCL	Reverse Torque			Input	26		*			
		Limit Input	9		*				27		*
10		*				28		*			
			11		*				29	OUTM2	General-pur-
12		*				30	OUTM2	General-			pose Output 2
			13		*		СОМ	Output 2	31	OUTM3	General-pur-
14		*				32	OUTM3	General-			pose Output 3
			15	/ALM	Alarm Output		СОМ	Output 3	33	BATCOM	Backup Battery
16	ALMCOM	Alarm Output				34	BAT	Backup			Input
			17		*			Battery Input	35	OUTM1COM	General-pur-
18		*				36	OUTM1	General-			pose Output 1
								Output 1			

Note

- (1) Do not connect anything to unused pins (*).
 - (2) Inputs for pins 19 and 20 are determined by parameter settings. The diagram shows the default configuration.

CN1 Connector (36 Pin)

Name	Model	Manufacturer
Servo Drive Connector	52986-3679	Molex Japan
Cable Connector	10136-3000PE	Sumitomo 3M
Cable Case (Shell Kit)	10336-52A0-008	Sumitomo 3M
Control I/O Signals

The following table shows the names and functions of the Servo Drive's control I/O signals.

CN1 Control Input Signals

Pin No.	Symbol	Name	Function/Interface				
1	+24VIN	12 to 24-VDC Power Supply Input	Power supply input terminal (12 to 24 VDC) for sequence inputs.				
2	STOP	Emergency Stop Input	Input for emergency stop. When this signal is enabled and pin 1 is not connected to pin 2, an Emergency Stop Input error (alarm code 87) occurs. Set this signal to be enabled or disabled in the Emergency Stop Input Setting (Pn041). (Factory default: Enable)				
3	EXT3	External Latch Signal 3	This external signal input latches the current value feedback pulse counter.				
4	EXT2	External Latch Signal 2	The position data is obtained the moment the input is turned ON. Minimal signal width must be 1 ms or more.				
5	EXT1	External Latch Signal 1					
6	IN1	External Gen- eral-purpose Input 1	This input is used as external general-purpose input 1.				
7	PCL	Forward Torque Limit Input	When the Torque Limit Selection (Pn003) is set to 3 or 5, this signal input selects the torque limit.				
8	NCL	Reverse Torque Limit Input					
19 to 20	POT	Forward Drive Prohibit Input	Forward, reverse drive rotation overtravel Input.				
	NOT	Reverse Drive Prohibit Input	Pn044 sets the function assignment for pins 19 and 20. Pn066 selects the operation.				
21	DEC	Origin Proximity Input	Connect the origin proximity input signal in the origin search opera- tion. Pn042 changes the logic of the sensor.				
22	INO	External Gen- eral-purpose Input 0	This input is used as external general-purpose input 0.				
23	IN2	External Gen- eral-purpose Input 2	This input is used as external general-purpose input 2.				
11		Not used	Do not connect anything.				
12		Not used	Do not connect anything.				
13		Not used	Do not connect anything.				
14		Not used	Do not connect anything.				
9		Not used	Do not connect anything.				
10		Not used	Do not connect anything.				
27		Not used	Do not connect anything.				
28		Not used	Do not connect anything.				
34 33	BAT BATCOM	Backup battery input	Connect a battery to these terminals as a backup when the abso- lute encoder is stopped. A cable with a battery is not required if a battery is connected to these terminals (Backup voltage: 3.6 V)				
17		Not used	Do not connect anything.				
24		Not used	Do not connect anything.				
25		Not used	Do not connect anything.				
26		Not used	Do not connect anything.				
18		Not used	Do not connect anything.				

CN1 Control Input Signal Connection Diagram



- Note (1) If the limit input signal inputs (Servo Drive's forward drive prohibited signal and reverse drive prohibited signal) are not allocated, the Servo Drive will not stop the Servomotor when the signal is input, and the Position Control Unit will also not detect limit inputs as errors. When using a Position Control Unit, always allocate the Servo Drive's forward drive prohibited signal and reverse drive prohibited signal to enable use of the limit input signals. (Refer to 6-4 Standard Settings for Servo Drives Using MECHATROLINK.)
 - (2) The signal width of the limit input signals (forward drive prohibited signal and reverse drive prohibited signal) and origin proximity input signal (origin return deceleration limit switch) must be longer than the MECHA-TROLINK communications cycle. If the input signal width is shorter than the communications cycle, the Position Control Unit will be unable to detect the input signal.
 - (3) When selecting a sensor for the origin proximity input signal (origin return deceleration LS), use a sensor such as a Photoelectric Sensor, which does not have chattering, because the origin signal is detected after the input goes from ON to OFF during the origin search. If a switch with contacts is used, the origin position may shift due to the switch contact's chattering.

Pin No.	Symbol	Name	Function/Interface			
15	/ALM	Alarm Output	The output is OFF when an			
16	ALMCOM		alarm is generated in the Servo Drive.			
29	OUTM2	General-purpose Out-	This is a general-purpose out-			
30	OUTM2COM	put 2 (READY)	put. The function for this output			
31	OUTM3	General-purpose Out-	parameter.			
32	OUTM3COM	put 3 (CLIM)	Refer to Output Signal Assign-			
36	OUTM1	General-purpose Out-	ment Details on the next page.			
35	OUTM1COM	put 1 (BKIR)				

CN1 Control Output Signals

Output Signal Assignment Details

Pn112 (General-purpose Output 1 Function Selection) Pn113 (General-purpose Output 2 Function Selection) Pn114 (General-purpose Output 3 Function Selection)	OUTM1 ((OUTM2 ((OUTM3 ((General-purpose Output 1) General-purpose Output 2) General-purpose Output 3)
0	Not assigned	No output. Always OFF.
1	INP1	Positioning Completed 1 output assignment.
2	VCMP	Speed Conformity Signal out- put assignment.
3	TGON	Servomotor Rotation Speed Detection output assignment.
4	READY	Servo Ready output assign- ment.
5	CLIM	Current Limit Detection output assignment.
6	VLIM	Speed Limit Detection output assignment.
7	BKIR	Brake Interlock output assign- ment.
8	WARN	Warning Signal output assign- ment.
9	INP2	Positioning Completed 2 output assignment.

Control Output Circuits



External power supply Maximum operating voltage: 30 VDC Maximum output current: 50 mA

(Use high-speed diodes.)

W-series Servo Drive I/O Signals (R88D-WT with FNY-NS115) 3-3-4

This section explains the I/O signals used between the PCU and a W-series Servo Drive equipped with a FNY-NS115 MECHATROLINK-II I/F Unit.

Use the OMNUC W Series User's Manual together with this manual for information on I/O signals.

Terminal Arrangement of the Control I/O Connector (CN1)

The following diagram shows the terminal arrangement of the W-series Servo Drive's Control I/O Connector (CN1) when MECHATROLINK is being used with the Servo Drive's standard settings.

This diagram shows only the I/O signals used when connecting to the PCU.

For details on the Servo Drive's standard settings, refer to 6-4 Standard Settings for Servo Drives Using MECHATROLINK.

	1		1				DICID	Dud a tata	26	INP1COM	Positioning completed out-
2						27	BKIR	Brake Inter-			put 1 common
			3					юск оцриг	28	BKIRCOM	Brake interlock
4						29	READY	Servo ready			output common
			5					output	30	READYCOM	Servo ready
6						31	ALM	Alarm output			output common
			7						32	ALMCOM	Alarm output
8						33]		common
			9			1			34		
10						35			1		
			11						36		
12						37					
· -			13						38		
14			10			39					
1			15						10		Linused input
16			15			11	DEC	Origin return	+0		ondoed input
10			17			41	DEC	deceleration LS	40	DOT	Forward drive
4.0			17			40	NOT	input	42	POT	prohibit input
18						43	NOT	prohibit input			=
			19					p	44	EXT1	External latch 1
20						45	EXT2	External			input
			21	BAT	Backup battery			laten 2 input	46	EXT3	External latch 3
22	BATGND	Backup battery			+ input	47	+24VIN	+24 VDC con-			input
		– input	23					ply input	48		
24						49					
			25	INP1	Positioning]			50		
<u> </u>	•	•			output 1		•	•	1		

Note

(1) Do not connect wiring to unused pins.

(2) Connect the control I/O signal cable's shield wire to the connector shell. The connector on the Servo Drive side is connected to the FG (frame ground).

CN1 Connector (50 Pin)

Name	Model	Manufacturer
Receptacle on Servo Drive Side	10250-52A2JL	Sumitomo 3M
Soldered Plug on Cable Side	10150-3000VE	Sumitomo 3M
Case on Cable Side	10350-52A0-008	Sumitomo 3M

Control I/O Signals

The following tables show the names and functions of the Servo Drive's control I/O signals.

CN1 Control Input Signals

Pin no.	Signal	Name	Function/Interface	Control mode
40		Not used.	This control input signal is not used with the standard set- tings.	
41	DEC	Origin return deceleration LS	Used as the origin proximity input signal during the origin search operation.	Position
			With the standard settings, the signal is enabled when ON.	
42	POT	Forward drive prohibit	Used as the forward limit input.	All modes
		input (Positive overtravel)	With the standard settings, the input is normally closed and operates as follows: OFF: Drive prohibited ON: Drive allowed	
43	NOT	Reverse drive prohibit	Used as the reverse limit input.	All modes
		input (Negative over- travel)	With the standard settings, the input is normally closed and operates as follows	
11	EXT1	External latch 1 input	Input signal used for external interrunts	Position
45	EXT2	External latch 2 input	lised as an external interrupt input signal during interrupt	1 0311011
45			feeding or an external origin input signal during merupi	
46	EX13	External latch 3 input	search.	
			With the standard settings, the signal is enabled when ON.	
47	+24VIN	+24 VDC control power supply	This is the input terminal for the +24 VDC control input power supply.	All modes

Control Input Circuit



- Signal levels ON: Min. voltage (+24VIN 11 V) OFF: Max. voltage (+24VIN – 1 V)
 - Note
- (1) If the limit input signal inputs (Servo Drive's forward drive prohibited signal and reverse drive prohibited signal) are not allocated, the Servo Drive will not stop the Servomotor when the signal is input, and the Position Control Unit will also not detect limit inputs as errors. When using a Position Control Unit, always allocate the Servo Drive's forward drive prohibited signal and reverse drive prohibited signal to enable use of the limit input signals. (Refer to 6-4 Standard Settings for Servo Drives Using MECHATROLINK.)
 - (2) The signal width of the limit input signals (forward drive prohibited signal and reverse drive prohibited signal) and origin proximity input signal (origin return deceleration limit switch) must be longer than the MECHA-TROLINK communications cycle. If the input signal width is shorter than the communications cycle, the Position Control Unit will be unable to detect the input signal.

(3) When selecting a sensor for the origin proximity input signal (origin return deceleration LS), use a sensor such as a Photoelectric Sensor, which does not have chattering, because the origin signal is detected after the input goes from ON to OFF during the origin search. If a switch with contacts is used, the origin position may shift due to the switch contact's chattering.

CN1 Control Output Signals

Pin no.	Signal	Name	Function/Interface	Control mode
25	INP1	Positioning completed	The position deviation is less than positioning comple-	Position
26	INP1COM	output 1	tion range 1 (Pn500). (This signal is always OFF in con- trol modes other than position control mode.)	
27	BKIR	Brake interlock output	Prlock output This is the holding brake timing signal that is output	
28	BKIRCOM		according to the settings in parameters Pn506, Pn507, and Pn508.	
29	READY	Servo ready output	Turned ON if there are no errors after the control and	All modes
30	READYCOM		main circuit power supplies are turned ON.	
31	ALM	Alarm output	This output turns OFF when there is a Servo Drive	All modes
32	ALMCOM		alarm.	
Shell	FG	Frame ground	Use this terminal to connect the cable's shield and FG wire.	All modes

Control Output Circuit



Note The circuit is equipped with an auto-resetting circuit breaker to protect the output. Even if an overcurrent trips the breaker, the breaker will reset automatically after a certain time elapses with no current. (Ver. 37 and later Servo Drives are equipped with the auto-resetting circuit breakers.)

3-3-5 W-series Servo Drive I/O Signals (R88D-WN□-ML2 with MECHATROLINK-II Built-in Communications)

This section explains the I/O signals used between the PCU and a W-series Servo Drive equipped with built-in MECHATROLINK-II communications.

Use the *OMNUC W Series User's Manual* (Cat. No. 1544) together with this manual for information on I/O signals.

Terminal Arrangement of the Control I/O Connector (CN1)

When using W-series Servo Drives equipped with built-in MECHATROLINK-II communications, the default control I/O signal allocations are the standard Servo Drive settings for using MECHATROLINK.

The following diagram shows the terminal arrangement of the W-series Servo Drive's Control I/O Connector (CN1) when MECHATROLINK is being used with the Servo Drive's default settings.

This diagram shows only the I/O signals used when connecting to the PCU.

External I/O Circuits

For details on the Servo Drive's standard settings, refer to 6-4 Standard Settings for Servo Drives Using MECHATROLINK.

1	BKIR	Brake interlock	Ī			14	BAT	Backup battery]		
	(SO1+)	output	2	BKIRCOM (SO1–)	Brake interlock output common			+ input	15	BATGND	Backup battery – input
3	ALM	Alarm output				16					
			4	ALMCOM	Alarm output				17		
5			Ī		common	18					
			6	+24VIN	24-VDC				19		
7	POT	Forward drive			supply	20					
		prohibited	8	NOT	Reverse drive				21		
9	DEC	Origin return	Ī		prohibited	22					
		limit switch	10	EXT1	External latch]			23	SO2+	Unused output
11	EXT2	External latch	Ī		i input	24	SO2-	Unused output			
		2 input	12	EXT3	External latch	1			25	SO3+	Unused output
13		Unused input			3 input	26	SO3-	Unused output			
]					

Note

(1) Do not connect wiring to unused pins.

(2) Connect the control I/O signal cable's shield wire to the connector shell. The connector on the Servo Drive side is connected to the FG (frame ground).

CN1 Connector (26 Pins)

Name	Model	Manufacturer
Receptacle on Servo Drive Side	10226-52A2JL	Sumitomo 3M
Soldered Plug on Cable Side	10126-3000VE	Sumitomo 3M
Case on Cable Side	10326-52A0-008	Sumitomo 3M

Control I/O Signals

The following tables show the names and functions of the Servo Drive's control I/O signals.

Control Input Signals

Pin No.	Symbol	Name	Function/Interface	Control mode
6	+24VIN	+24 VDC control power supply	This is the input terminal for the +24 VDC control input power supply.	All modes
7	POT	Forward drive pro-	Used as the forward limit input.	All modes
		hibited	With the standard settings, the input is normally closed and operates as follows: OFF: Drive prohibited ON: Drive allowed	
8	NOT	Reverse drive pro- hibited	Used as the reverse limit input.	All modes
			With the standard settings, the input is normally closed and operates as follows OFF: Drive prohibited ON: Drive allowed	
9	DEC	Origin return decel- eration limit switch	Used as the origin proximity input signal during the origin search operation.	Position
			With the standard settings, the signal is enabled when ON.	

External I/O Circuits

Section 3-3

Pin No.	Symbol	Name	Function/Interface	Control mode
10	EXT1	External latch 1 input	Input signal used for external interrupts.	Position
11	EXT2	External latch 2 input	Used as an external interrupt input signal during	
12	EXT3	External latch 3 input	interrupt feeding or an external origin input signal during an origin search.	
			With the standard settings, the signal is enabled when ON.	
13		Not used.	This control input signal is not used with the standard settings.	

Control Input Circuit



Signal levels ON: Min. voltage (+24VIN – 11 V) OFF: Max. voltage (+24VIN – 1 V)

- Note
- (1) If the limit input signal inputs (Servo Drive's forward drive prohibited signal and reverse drive prohibited signal) are not allocated, the Servo Drive will not stop the Servomotor when the signal is input, and the Position Control Unit will also not detect limit inputs as errors. When using a Position Control Unit, always allocate the Servo Drive's forward drive prohibited signal and reverse drive prohibited signal to enable use of the limit input signals. (Refer to 6-4 Standard Settings for Servo Drives Using MECHATROLINK.)
 - (2) The signal width of the limit input signals (forward drive prohibited signal and reverse drive prohibited signal) and origin proximity input signal (origin return deceleration limit switch) must be longer than the MECHA-TROLINK communications cycle. If the input signal width is shorter than the communications cycle, the Position Control Unit will be unable to detect the input signal.
 - (3) When selecting a sensor for the origin proximity input signal (origin return deceleration LS), use a sensor such as a Photoelectric Sensor, which does not have chattering, because the origin signal is detected after the input goes from ON to OFF during the origin search. If a switch with contacts is used, the origin position may shift due to the switch contact's chattering.

Pin No.	Symbol	Name	Function/Interface	Control mode
1	BKIR (SO1+)	Brake interlock out- put	This is the holding brake timing signal that is output according to the settings in parameters Pn506,	All modes
2	BKIRCOM (SO1–)		Pn507, and Pn508.	
3	ALM	Alarm output	This output turns OFF when there is a Servo Drive	All modes
4	ALMCOM		alarm.	

Control Output Signals

External I/O Circuits

Section 3-3

Pin No.	Symbol	Name	Function/Interface	Control mode
23	SO2+	General-purpose	These control input signals are not used with the	All modes
24	SO2-	output	standard settings.	
25	SO3+			
26	SO3-			
Shell	FG	Frame ground	Use this terminal to connect the cable's shield and FG wire.	All modes

Control Output Circuit



Note The circuit is equipped with an auto-resetting circuit breaker to protect the output. Even if an overcurrent trips the breaker, the breaker will reset automatically after a certain time elapses with no current.

3-3-6 SMARTSTEP Junior Servo Drive I/O Signals (R7D-ZN_-ML2 with Built-in MECHATROLINK-II Communications)

This section describes the standard I/O signals used with a Position Control Unit when using a SMARTSTEP Junior Servo Drive with Built-in MECHA-TROLINK-II Communications.

Refer also to the SMARTSTEP Junior Servo Drive with MECHATROLINK-II Communications User's Manual (Cat. No. 1554).

Terminal Arrangement of Control I/O Connector (CN1)

The default control I/O signal allocations for a SMARTSTEP Junior Servo Drive with MECHATROLINK-II Communications are set to the standard Servo Drive settings for using MECHATROLINK. The default pin arrangement of the control I/O connector (CN1) on the Servo Drive are shown below.

Only the I/O signals that are connected to the Position Control Unit are shown. Refer to *6-4 Standard Settings for Servo Drives Using MECHATROLINK* for the standard Servo Drive settings for using MECHATROLINK.

			1	EXT1	External				8		
2	DEC	Origin prox- imity input			latch signal 1 input	9					
4	POT	Forward drive inhibit input	3	NOT	Reverse drive inhibit input	11			10		
6	STOP	Emergency stop input	5	+24VIN	+24 VDC control power sup- ply input	13	BKIR	Brake inter- lock output	12	ALM	Alarm output
			7	0GND	Output ground common				14		

Note

⁽¹⁾ Do not connect unused pins.

(2) Connect the shield in the control I/O signal cable to the connector hood. At the Servo Drive connector, connect it to the FG (Frame ground).

CN1 Connector (14 Pins)

Name	Model	Manufacturer
Receptacle on Servo Drive Side	10226-52A2JL	Sumitomo 3M
Soldered Plug on Cable Side	10126-3000VE	Sumitomo 3M
Case on Cable Side	10326-52A0-008	Sumitomo 3M

Control I/O Signals

The names and functions of Servo Drive control I/O signals are given in the following table.

CN1 Control Input Signals

Pin No.	Signal	Name	Function/interface	Control mode
1	EXT1	External latch signal 1 input	An external interrupt input signal.	Position
			Use as an external interrupt signal for interrupt feeding or as external origin input signal for origin searches. The signal is valid when ON.	
2	DEC	Origin proximity input	The origin proximity input signal for ori- gin searches.	All modes
			The signal is valid when ON.	
3	NOT	Reverse drive inhibit input	The limit input in the reverse direction. This input operates as follows (i.e., like a NC contact): OFF: Drive prohibited, ON: Drive enabled	All modes
4	POT	Forward drive inhibit input	The limit input in the forward direction. This input operates as follows (i.e., like a NC contact): OFF: Drive prohibited, ON: Drive enabled	All modes
5	+24VIN	+24-V power supply input for control DC	The +24 VDC input terminal for the con- trol input power supply.	All modes
6	STOP	Emergency stop input	An external input signal used to stop power supply to the motor.	All modes
			This input is used when an error occurs to unlock the Servo from a host control- ler.	
			 This input operates as follows (i.e., like a NC contact): OFF: Power to motor stopped (Servo cannot be locked). ON: Power can be supplied to motor (Servo can be locked). 	

Control Input Circuits



- Note (1) If the limit input signal inputs (Servo Drive's forward drive prohibited signal and reverse drive prohibited signal) are not allocated, the Servo Drive will not stop the Servomotor when the signal is input, and the Position Control Unit will also not detect limit inputs as errors. When using a Position Control Unit, always allocate the Servo Drive's forward drive prohibited signal and reverse drive prohibited signal to enable use of the limit input signals. (Refer to 6-4 Standard Settings for Servo Drives Using MECHATROLINK.)
 - (2) The signal width of the limit input signals (forward drive prohibited signal and reverse drive prohibited signal) and origin proximity input signal (origin return deceleration limit switch) must be longer than the MECHA-TROLINK communications cycle. If the input signal width is shorter than the communications cycle, the Position Control Unit will be unable to detect the input signal.
 - (3) When selecting a sensor for the origin proximity input signal (origin return deceleration LS), use a sensor such as a Photoelectric Sensor, which does not have chattering, because the origin signal is detected after the input goes from ON to OFF during the origin search. If a switch with contacts is used, the origin position may shift due to the switch contact's chattering.
 - (4) The SMARTSTEP Junior Servo Drive's emergency stop input stops the Servomotor using software processing. If required for system safety, add a safety circuit in an external circuit.

CN1 Control Output Signals

Pin No.	Signal	Name	Function/interface	Control mode
7	0GND	Output ground common	The ground common for control output sig- nals.	All modes
12	ALM	Alarm output	Turns OFF when an alarm occurs in the Servo Drive.	All modes
13	BKIR	Brake interlock output	Output as a timing signal for a holding brake when the Servo is locked or unlocked.	All modes
Shell	FG	Frame ground	The connection point for the cable shield and FG line.	All modes

Control Input Circuits



Note The circuit is equipped with an auto-resetting circuit breaker to protect the output. Even if an overcurrent trips the breaker, the breaker will reset automatically after a certain time elapses with no current.

3-4 Wiring

This section provides examples of the connections between the PCU and Servo Drive as well as the Servo Drive's control I/O connections.

3-4-1 MECHATROLINK-II Communications Wiring

Use the special MECHATROLINK-II Connection Cable to connect the PCU and Servo Drive (MECHATROLINK-II I/F Unit).

Connection Cable

Use the following cables (made by Yaskawa Electric) to connect MECHA-TROLINK-II devices.

Name	Model number	Cable length	Manufacturer
MECHATROLINK-II	JEPMC-W6003-A5	0.5 m	Yaskawa Electric
Connection Cable (USB connectors and ferrite cores on both ends)	JEPMC-W6003-01	1.0 m	Corporation
	JEPMC-W6003-03	3.0 m	
	JEPMC-W6003-05	5.0 m	
	JEPMC-W6003-10	10 m	
	JEPMC-W6003-20	20 m	
	JEPMC-W6003-30	30 m	

TerminatorsMake sure to connect the following Terminator at the end of the MECHA-
TROLINK-II communications line.

Name	Model number	Manufacturer	
MECHATROLINK-II Terminator	JEPMC-W6022	Yaskawa Electric Corporation	

Repeaters

The wiring distance for the MECHATROLINK-II can be extended to a maximum of 100 m by using Repeaters.

Name	Model number	Manufacturer
MECHATROLINK-II Repeater	JEPMC-REP2000	Yaskawa Electric Corporation

Note

te MECHATROLINK-II Connection Cables and Terminators can be ordered from OMRON with the following model numbers.

Name	Yaskawa Electric Co. model number	OMRON model number
MECHATROLINK-II Connection	JEPMC-W6003-A5	FNY-W6003-A5
Cable (USB connectors and ferrite	JEPMC-W6003-01	FNY-W6003-01
cores on both ends)	JEPMC-W6003-03	FNY-W6003-03
	JEPMC-W6003-05	FNY-W6003-05
	JEPMC-W6003-10	FNY-W6003-10
	JEPMC-W6003-20	FNY-W6003-20
	JEPMC-W6003-30	FNY-W6003-30
MECHATROLINK-II Terminator	JEPMC-W6022	FNY-W6022
MECHATROLINK-II Repeater	JEPMC-REP2000	FNY-REP2000

MECHATROLINK-II Communications Connections

The following example shows a PCU connected with several Servo Drives with the MECHATROLINK-II Connection Cables.



Note

When not using Repeaters, the maximum total length of the Connection Cable (L1 + L2 + ... + Ln) is 50 m when using fewer than 16 axes or 30 m when using 16 axes.

MECHATROLINK-II Connection Cable Length

The maximum total length of the Connection Cables depends on the number of MECHATROLINK devices (Servo Drives) being connected, as shown in the following table.

Repeaters	Number of MECHATROLINK devices		Minimum cable length between devices	Maximum total cable length
Without	15 or fewer		0.5 m min.	50 m max.
Repeaters	16		0.5 m min.	30 m max.
With	Between PCU and Repeater	14 or fewer	0.5 m min.	50 m max.
Repeaters		15	0.5 m min.	30 m max.
	Between Repeater and Terminator	15 or fewer	0.5 m min.	50 m max.
		16	0.5 m min.	30 m max.

Note The number of devices in the table for when Repeaters are used does not include the Repeaters. For example, if using one MECHATROLINK device between the PCU and the Repeater and 15 MECHATROLINK devices between the Repeater and the Terminator, there will be 16 MECHATROLINK devices and the maximum total cable length is 100 m.

Note Always turn OFF the power supply to the PCU and Servo Drives before connecting or disconnecting MECHATROLINK-II Connection Cables or the Terminator.

3-4-2 Wiring the Servo Drive I/O Signals

The following example shows a Servo Drive's control I/O signal connections when a Position Control Unit is connected to a G5-series Servo Drive, a G-series Servo Drive, a W-series Servo Drive, or a SMARTSTEP Junior Servo Drive.

For details on connecting the Servo Drive to the power supply or Servomotor, refer to the Servo Drive operation manual.

Control I/O Connector (CN1) Connection Example

R88D-KN -ML2 (Equipped with Built-in MECHATROLINK-II Communications)



R88D-GND-ML2 (Equipped with Built-in MECHATROLINK-II Communications)





- (1) If a backup battery is connected, a cable with a battery is not required.
- (2) Inputs for pins 19 and 20 are determined by parameter settings. The diagram shows the default configuration.

R88D-WT with FNY-NS115

The following example shows the connections when the standard I/O signal settings are being used.





R7D-ZN -ML2 (Equipped with Built-in MECHATROLINK-II Communications)



Note When a Servo Drive is controlled through MECHATROLINK-II communications, software processes are used to stop the Servomotor when the corresponding PCU control signal is received, the Servo Drive's drive prohibited input signal is received, or an error occurs. Use an external fail-safe circuit (outside of the Servo Drive), such as a circuit that disconnects the Servo Drive's main power supply, to stop the system in an emergency.

Components Provided to Wire Control I/O Signals

The following components are provided to wire the Control I/O Connector (CN1) on a G5-series, a G-series, W-series, or SMARTSTEP Junior Servo Drive.

Control I/O Connector (R88A-CNU01C) This connector connects to the Control I/O Connector (CN1) on a G-series Servo Drive (R88D- GN_-ML2).

Use this connector when making your own control cable.

Dimensions



Control I/O Connector (R88A-CNU11C) This connector connects to the W-series Servo Drive's Control I/O Connector (CN1).

Use this connector when making your own control cable.

Dimensions



Control I/O Connector (R88A-CNW01C)

This connector connects to the R88D-KN□-ML2 G5-series Servo Drive or the R88D-WN□-ML2 W-series Servo Drive's Control I/O Connector (CN1).

Use this connector when making your own control cable.

Dimensions



Connector plug: 10126-3000VE (Sumitomo 3M) Connector case: 10326-52A0-008 (Sumitomo 3M)

Control I/O Connector (R7A-CNA01R)

This connector connects to the R7D-ZN \square -ML2 SMARTSTEP Junior Servo Drive's Control I/O Connector (CN1).

Use this connector when making your own control cable.



General-purpose Control Cable (R88A-CPW S)

This cable has a connector already attached, which connects to the R88D-WT W-series Servo Drive's Control I/O Connector (CN1). There is no connector attached to the other end of the cable. Attach an appropriate connector to connect the desired I/O device in order to use the cable.

Standard Cables

Model	Length (L)	Sheath diameter	Approx. weight
R88A-CPW001S	1 m	12.8-mm dia.	0.3 kg
R88A-CPW002S	2 m		0.6 kg

Connection Configuration and Dimensions



No.	Wire/Marking colors	Signal
1	Yellow/Black ()	
2	Pink/Black ()	
3	Yellow/Red ()	
4	Pink/Red ()	
5	Orange/Red (-)	
6	Orange/Black (-)	
7	Gray/Red (-)	
8	Gray/Black (-)	
9	White/Red (-)	
10	White/Black (-)	
11	Yellow/Red (-)	
12	Yellow/Black (-)	
13	Yellow/Black ()	
14	Pink/Black (-)	
15	Pink/Red (-)	
16	Orange/Red ()	
17	Orange/Black ()	
18	Pink/Red ()	
19	Gray/Red ()	
20	Gray/Black ()	
21	Gray/Red ()	
22	Gray/Black ()	
23	White/Red ()	
24	White/Black ()	
25	Orange/Red ()	INP1
26	Orange/Black ()	INP1COM

Wiring

No.	Wire/Marking colors	Signal
27	White/Red ()	BKIR
28	White/Black ()	BKIRCOM
29	Yellow/Red ()	READY
30	Yellow/Black ()	READYCOM
31	Pink/Red ()	
32	Pink/Black ()	
33	Orange/Red ()	
34	Orange/Black ()	
35	Gray/Black ()	
36	Gray/Red ()	
37	White/Red ()	
38	White/Black ()	
39	Yellow/Red ()	
40	Pink/Red ()	(Not used.)
41	Pink/Black ()	DEC
42	Orange/Red ()	POT
43	Orange/Black ()	NOT
44	Gray/Black ()	EXT1
45	White/Red ()	EXT2
46	White/Black ()	EXT3
47	Gray/Red ()	+24VIN
48	Yellow/Red ()	
49	Yellow/Black ()	
50	Pink/Black ()	
Shell		FG

Connector Plug model: 10150-3000VE (Sumitomo 3M) Connector Case model: 10350-52A0-008 (Sumitomo 3M) Cable: 24 AWG, 25 wire, UL20276

Note

(1) Wires with the same wire color and number of marks make up a twisted pair.

For example, the Orange/Red (-) and Orange/Black (-) wires make up a twisted pair.

(2) The I/O signals listed in the table above are applicable only when connecting to the PCU. Do not wire any unused signals.

Connector-Terminal Block Cables (XW2Z-□J-B34) This cable connects to the R88D-KN□-ML2 G5 series Servo Drive's connector terminal block.

Standard Cables

Model	Length (L)	Outer diameter of cable	Approx. weight
XW2Z-100J-B34	1 m	8.0 dia.	0.1 kg
XW2Z-200J-B34	2 m		0.2 kg

Connection Configuration and Dimensions



Connector on	Conne	ector Terminal Block	Conne	ctor on Servo	Drive (CN1)
Name	No.		No.	Name	
+24V	1	1	6	+24VIN	
0V	2				
+24V	3				
0V	4	┝╼┫ │ ┊			
+24V	5				
0V	6				
STOP	7		5	STOP	
DEC	8		9	DEC	
POT	9		7	POT	
NOT	10		8	NOT	
EXT1	11		10	EXT1	
EXT2	12		11	EXT2	
EXT3	13		12	EXT3	
BATGND	14		15	BATGND	
BAT	15	<u></u>	14	BAT	
BKIRCOM	16		2	BKIRCOM	
BKIR	17		1	BKIR	
ALMCOM	18		4	ALMCOM	
ALM	19		3	ALM	
FG	20	<u>↓</u>	Shell	FG	

*Set the signal names of driver connectors shown above.

Connector on Servo Drive Connector Plug: 10126-3000PE (Sumitomo 3M) Connector Case: 10326-52A0-008 (Sumitomo 3M) Connector on Connector Terminal Block Connector Socket: XG4M-2030 (OMRON) Strain Relief: XG4T-2004 (OMRON) Cable: AWG28 x 3P + AWG28 x 7C UL2464

Connector-Terminal Block Cables (XW2Z-□J-B33)

This cable connects to the R88D-WN -ML2 G series Servo Drive's connector terminal block.

Standard Cables

Model	Length (L)	Outer diameter of cable	Approx. weight
XW2Z-100J-B33	1 m	8.0 dia.	0.1 kg
XW2Z-200J-B33	2 m		0.2 kg

Connection Configuration and Dimensions



Wiring

Terminal block		Connector			Servo Drive				
Signal	No.		No.		No.	Wire/mark color	Signal		
+24VIN	1	\vdash	1	<u> </u>	1	Blue/Red (1)	+24VIN		
0V	2	\vdash	2	┝┊──┊╢		Blue/Black (1)			
+24VIN	3	\vdash	3	╞╴┊╢╋		Pink/Red (1)			
0V	4	\vdash	4	╞╴┊┥║		Pink/Black (1)			
+24VIN	5	\vdash	5	┝┋╤╤┋┦┚		Green/Red (1)			
0V	6	\vdash	6	┝┿╾┿╹		Green/Black (1)			
STOP	7	\vdash	7		2	Orange/Red (1)	STOP		
DEC	8	\vdash	8		21	Orange/Black (1)	DEC		
POT	9	\vdash	9		19	Gray/Red (1)	POT		
NOT	10	\vdash	10		20	Gray/Black (1)	NOT		
EXT1	11	\vdash	11	\vdash	5	Blue/Red (2)	EXT1		
EXT2	12	\vdash	12		4	Blue/Black (2)	EXT2		
EXT3	13	\vdash	13		3	Pink/Red (2)	EXT3		
BATCOM	14	\vdash	14	HwH	33	Green/Red (2)	BATCOM		
BAT	15	\vdash	15	$\vdash \frown$	34	Green/Black (2)	BAT		
OUTM1COM	16		16	Hwh	35	Orange/Red (2)	OUTM1COM		
OUTM1	17	\vdash	17	$\vdash \frown$	36	Orange/Black (2)	OUTM1		
ALMCOM	18		18	Hwh	16	Gray/Red (2)	ALMCOM		
/ALM	19	\vdash	19	$\vdash \frown$	15	Gray/Black (2)	/ALM		
FG	20	\vdash	20		Shell	Not specified.	FG		

• Wires with the same wire color and the same number of marks form a twisted pair.

A pink/red (1) wire and pink/black (1) wire form a twisted pair.

Servo Drive Connector Connector plug: 10136-3000PE (Sumitomo 3M)

Connector case: 10336-52A0-008 (Sumitomo 3M)

Terminal Block Connector Connector socket: XG4M-2030 (OMRON) Strain relief: XG4T-2004 (OMRON)

Cable AWG28×10P UL2464

Connector Terminal Block Cables (XW2Z-□J-B15)

This cable connects to the R88D-WT W-series Servo Drive's connector terminal block.

Standard Cables

Model	Length (L)	Approx. weight
XW2Z-100J-B15	1 m	0.1 kg
XW2Z-200J-B15	2 m	0.2 kg

Connection Configuration and Dimensions



Connector termi	nal bloc	k end				Serv	o Drive end
Name	No.					No.	Name
+24 V	1	•	 !	 		47	+24 VIN
0 V	2		i		i -		
+24 V	3 -	├ 	!				
0 V	4	┝━♠			i		
+24 V	5		1		-		
0 V	6						
	7		1				
DEC	8		1	 		41	DEC
POT	9				1	42	POT
NOT	10		1	 		43	NOT
EXT1	11				<u>i</u>	44	EXT1
EXT2	12		1			45	EXT2
EXT3	13			 	 	46	EXT3
BATGND	14		\vdash		-	22	BATGND
BAT	15				-	21	BAT
BKIRCOM	16		\sim		1	28	BKIRCOM
BKIR	17				-	27	BKIR
ALMCOM	18		\vdash		i	32	ALMCOM
ALM	19		<u> </u>	 		31	ALM
FG	20		•	 	•	Shell	FG

Connector on Servo Drive End Connector Plug: 10150-300VE (Sumitomo 3M) Connector Case: 10350-52A0-008 (Sumitomo 3M)

Connector on Connector Terminal Block End Connector Socket: XG4M-2030 (OMRON) Strain Relief: XG4T-2004 (OMRON)

Cable: AWG28 \times 3P + AWG28 \times 7C UL2464

Note

Connector Terminal Block Cables (XW2Z-□J-B16) Signal names for the connector on the Servo Drive end are for standard I/O allocations.

This is the Connector Terminal Block Cable for W-series Servo Drives (R88D-WN \Box -ML2 with built-in MECHATROLINK-II communications).

Standard Cables

Model	Length (L)	Approx. weight
XW2Z-100J-B16	1 m	0.1 kg
XW2Z-200J-B16	2 m	0.2 kg

Connection Configuration and Dimensions



Connector termin	nal bloc	k end				Servo	Drive end
Name	No.	1				No.	Name
+24 V	1	└── ●	r	 		6	+24 VIN
0 V	2	h					
+24 V	3	} _+∳_					
0 V	4	┣━♠					
+24 V	5	++					
0 V	6						
	7		1				
DEC	8		1 1 1			9	DEC
POT	9		i I			7	POT
NOT	10		1			8	NOT
EXT1	11					10	EXT1
EXT2	12		1			11	EXT2
EXT3	13					12	EXT3
BATGND	14			\sim		15	BATGND
BAT	15					14	BAT
BKIRCOM	16	<u> </u>		\sim		2	BKIRCOM
BKIR	17					1	BKIR
ALMCOM	18		$-\sqrt{-}$	\sim		4	ALMCOM
ALM	19	<u> </u>		_^		3	ALM
FG	20		•	 	•	Shell	FG

Connector on Servo Drive End

Connector Plug: 10126-300VE (Sumitomo 3M) Connector Case: 10326-52A0-008 (Sumitomo 3M)

Connector on Connector Terminal Block End Connector Socket: XG4M-2030 (OMRON) Strain Relief: XG4T-2004 (OMRON)

Cable: AWG28 \times 3P + AWG28 \times 7C UL2464

Note

Connector Terminal Block Cable (XW2Z-□J-B19) This is the Connector Terminal Block Cable for the SMARTSTEP Junior Servo Drive Control I/O Connector (CN1).

Signal names for the connector on the Servo Drive end are for standard I/O

Standard Cables

allocations.

Model	Length (L)	Outer diameter of cable	Approx. weight
XW2Z-100J-B19	1 m	8 dia.	0.1 kg
XW2Z-200J-B19	2 m		0.2 kg

Connection Configuration and Dimensions



Terminal E	Block	С	onnect	or				Servo Drive er	nd
Signal	No.	1	No.		r	,	No.	Wire code/Mark color	Signal
+24VIN	1	\vdash	1	•	1		5	Blue/Red (-)	+24VIN
	2	\vdash	2		1				
+24VIN	3	<u> </u>	3						
	4	<u> </u>	4	-+ I					
+24VIN	5	<u> </u>	5						
	6	\vdash	6		1				
EXT1	7	<u> </u>	7				- 1	Pink/Red (-)	EXT1
DEC	8	<u> </u>	8		<u>⊢</u> ∧	+	- 2	Pink/Black (-)	DEC
NOT	9	<u> </u>	9		<u> </u>		- 3	Green/Red (-)	NOT
POT	10	<u> </u>	10		<u> </u>		- 4	Green/Black (-)	POT
	11	<u> </u>	11		+		- 8	Orange/Red (-)	
	12	<u> </u>	12		<u> </u>		- 9	Orange/Black (-)	
	13	<u> </u>	13		⊢v		10	Grey/Red (-)	
	14	<u> </u>	14		<u>⊢</u> ∧	^_÷	- 11	Grey/Black (-)	
STOP	15	<u> </u>	15				- 6	Blue/Red ()	STOP
0GND	16	<u> </u>	16		<u>⊢</u> ∧	+	- 7	Blue/Black ()	0GND
BKIR	17	<u> </u>	17		<u> </u>		13	Pink/Red ()	BKIR
	18	<u> </u>	18		<u> </u>	÷	- 14	Pink/Black ()	
/ALM	19		19		I I		12	Green/Red ()	/ALM
Shield	20		20				Shell	Shield	FG

Connector at Connector Terminal Block Connector Socket: XG4M-2030 Strain Relief: XG4T-5004

Cable: AWG28-10P UL20276

Connector at Servo Drive Connector Plug: 10114-3000PE (Sumitomo 3M) Connector Case: 10314-52A0-008 (Sumitomo 3M)

Connector-Terminal Conversion Units

The Connector-Terminal Block Conversion Unit can be used along with a Connector Terminal Block Cable (XW2Z-□J-B15/B16/B19/B33/B34) to convert the control I/O connector (CN1) of a G5-series Servo Drive, G-series Servo Drive, W-series Servo Drive, or SMARTSTEP Junior Servo Drive to a terminal block.

XW2B-20G4

The XW2B-20G4 is an M3 Screw Terminal Block.



Dimensions



Note

- (1) Use 0.30 to 1.25 mm² wire (AWG22 to AWG16).
- (2) The wire inlet is 1.8 mm (height) \times 2.5 mm (width).
- (3) Strip the insulation from the end of the wire for 6 mm as shown below.



XW2B-20G5

The XW2B-20G5 is an M3.5 Screw Terminal Block.



Dimensions



• Terminal block pitch: 8.5 mm

Note

- (1) When using crimp terminals, use crimp terminals with the following dimensions.
 - (2) When connecting wires and crimp terminals to a terminal block, tighten them with a tightening torque of 0.59 N·m.

Round Crimp Terminals	Fork Terminals
3.2-mm dia.	
6.8 mm max.	<u>3.7 mm</u> 6.8 mm max.

Applicable C	Applicable Wires		
Round Crimp	1.25-3	AWG22-16	
Terminals		(0.3 to 1.25 mm ²)	
	2-3.5	AWG16-14	
		(1.25 to 2.0 mm ²)	
Fork Terminals	1.25Y-3	AWG22-16	
		(0.3 to 1.25 mm ²)	
	2-3.5	AWG16-14	
		(1.25 to 2.0 mm ²)	

XW2D-20G6

The XW2D-20G6 is an M3 Screw Terminal Block.



Dimensions



Note (1) When using crimp terminals, use crimp terminals with the following dimensions.

(2) When connecting wires and crimp terminals to a terminal block, tighten them with a tightening torque of 0.7 N·m.

Applicable Cri	imp Termina	ls	Applicable Wires
3.2-mm di	a. $\frac{1}{5.8}$ mm max.	Ę	$\underbrace{3.2 \text{ mm}}_{3.2 \text{ mm}} 5.8 \text{ mm max}.$
Round Crimp	Terminals	Fork	Terminals

	Applicable miles	
Round Crimp	1.25-3	AWG22-16
Terminals		(0.3 to 1.25 mm ²)
Fork Terminals	1.25Y-3	AWG22-16
		(0.3 to 1.25 mm ²)

The following diagrams show typical connections between a host device and Servo Drives using a MECHATROLINK-II communications cable.

■ G5-series Servo Drives

Terminal Block Wiring Example (Same for XW2B-20G4, XW2B-20G5, and XW2D-20G6)



■ G-series Servo Drives

Terminal Block Wiring Example (Same for XW2B-20G4, XW2B-20G5, and XW2D-20G6)



Note

- (1) Absolute encoder backup battery 3.6 to 4.5 V
- (2) The XB contacts are used to turn ON/OFF the electromagnetic brake.
- (3) Assign BKIR (brake interlock) to CN1-36 pin to use.
- (4) The absolute encoder backup battery is not required when using a Servomotor with an incremental encoder.

- (5) Connect the absolute encoder backup battery to only one of either the connector terminal block or absolute encoder backup battery cable.
- (6) Use cable clips with double-sided adhesive tape to secure the absolute encoder backup battery in place.

■ W-series Servo Drives

<u>Terminal Block Wiring Example (Same for All Models; with Standard Settings for I/O Allocations)</u>



- (1) Absolute Encoder Backup Battery: 2.8 to 4.5 V When using a motor with an absolute encoder, connect a backup battery to one of the following: The Servo Drive for the R88D-WT□, the battery cable for the R88D-WN□-ML2, or the Connector Terminal Block.
 - (2) Do not connect anything to unused terminals.

Terminal Block Wiring Example (XW2B-20G4, XW2B-20G5, and XW2D-20G6)



Note

Note

(1) Use a maximum of 300 mA total for the 24-VDC inputs.(2) Do not use inputs other than sensor inputs.

Terminal Block Signal Names

Terminal Block					
Signal	No.				
+24VIN	1				
	2				
+24VIN	3				
	4				
+24VIN	5				
	6				
EXT1	7				
DEC	8				
NOT	9				
POT	10				
	11				
	12				
	13				
	14				
STOP	15				
0GND	16				
BKIR	17				
	18				
/ALM	19				
Shield	20				

Wiring Precautions

The electronic control devices may malfunction due to noise from nearby power supply lines, external loads, or other sources.

Malfunctions caused by noise can be troublesome because it can be difficult to recreate the situation and to identify the noise source.

Use the following methods to eliminate malfunctions due to noise and improve the system's reliability.

- When selecting wiring components, use wires or cables that meet or exceed the specifications listed in the Servo Drive User's Manual.
- Wire the control lines (communications lines, external I/O signal lines, etc.) separately from the power lines (AC power supply lines and motor power lines). Do not wire these lines together in the same duct or bundle them together.
- Use shielded cables for the control lines.
- Use the specified special cables to connect the PCU and Servo Drives.
- Always connect surge suppressors to nearby inductive loads (relays or solenoids).



- **Note** Connect surge-suppressing diodes or surge suppressors close to relays. Use surge-suppressing diodes with a dielectric strength of at least 5 times the circuit voltage.
 - Noise may be transferred through the power line if there is nearby equipment that generates high-frequency noise or the power supply is shared with equipment such as an electric welder or electric discharge equipment. In this case, insert a noise filter in the power supply input line.
 - Connect to a ground of 100 Ω or less and use the thickest possible wire, greater than 1.25 $\text{mm}^2.$
 - Twisted-pair cable is recommended for power lines.

SECTION 4 Data Areas

This section provides an overview of the parameter and data settings used in Position Control Unit operation and provides information on memory allocations.

4-1	Overall	Structure	76
4-2	Data A1	reas	79
4-3	Commo	on Parameter Area	96
	4-3-1	Common Parameters Overview	96
	4-3-2	Common Parameter Details	97
4-4	Axis Pa	rameter Area	101
	4-4-1	Axis Parameters Overview	101
	4-4-2	Axis Parameter Details	102
4-5	Servo P	arameter Area	104
	4-5-1	G5-series Servo Drive (R88D-KN□-ML2 with Built-in MECHATROLINK-II Communications)	104
	4-5-2	G-series Servo Drive (R88D-GND-ML2 with Built-in MECHATROLINK-II Communications)	116
	4-5-3	W-series Servo Drives (R88D-WT \square with FNY-NS115)	132
	4-5-4	W-series Servo Drive (R88D-WN□-ML2 with Built-in MECHATROLINK-II Communications)	151
	4-5-5	SMARTSTEP Junior Servo Drive (R7D-ZN□-ML2 with Built-in MECHATROLINK-II Communications)	174
4-6	Commo	on Operating Memory Area	179
	4-6-1	Common Operating Memory Area Overview	179
	4-6-2	Common Operating Memory Area Words	181
4-7	Axis O _l	perating Output Memory Areas	185
	4-7-1	Axis Operating Output Memory Area Overview	185
	4-7-2	Axis Operating Output Memory Area Allocations	186
	4-7-3	Axis Operating Output Memory Area Priority	190
4-8	Axis O _l	perating Input Memory Areas	194
	4-8-1	Axis Operating Input Memory Area Overview	194
	4-8-2	Axis Operating Input Memory Area Allocations	195
	4-8-3	Axis Control Status Flags	200
	4-8-4	Servo Status Flags	206
	4-8-5	External I/O Status Bits	211
	4-8-6	Expanded Monitoring	212

4-1 Overall Structure

The PCU is used by exchanging data with the CPU Unit as shown in the following diagram.



Data name	Contents	Setting area	Enable timing
Common Parameter Area	This area contains the parameters for basic setting of PCU operation, such as allocation of the Axis Operating Memory Areas and MECHATROLINK communica- tions. The common parameters must be set to use the PCU.	PCU's internal memory (The parameter settings can be saved in the PCU's flash memory.)	The settings saved in the PCU are enabled when read to the PCU internal memory at power ON or restart.
Axis Parameter Areas	These areas contain the parame- ters for axis control settings, such as the origin input signal selection and origin search method.	PCU's internal memory (The parameter settings can be saved in the PCU's flash memory.)	The settings saved in the PCU are read to the PCU internal memory at power ON or restart. When the settings are written, they are refreshed immediately and are enabled after they have been written.
Servo Parame- ter Area	These parameters are for setting Servo Drive operation.	Servo Drive's internal memory (The parameter settings can be saved in the Servo Drive's internal non-volatile memory.)	Online Servo Parameters are enabled as soon as they are written, and offline Servo parameters are enabled after the Servo Drive power is turned OFF and ON again, or after executing DEVICE SETUP.
Common Oper- ating Memory Area	This area is for settings for com- mon PCU operations, such as communications control and trans- ferring common parameters. The status of these operations is also input to this area.	CPU Unit's CPU Bus Unit Area	Data is updated with every I/O refresh of the CPU Unit. The set data is enabled and used with the startup of each operation.
Axis Operating Output Memory Areas	These areas are for settings and operations of axis operation, such as positioning/speed commands and operating commands for direct operation, origin search, and jog- ging.	CPU Unit's memory area set in the common parameters.	Data is updated with every I/O refresh of the CPU Unit. The set data is enabled and used with the startup of each operation.
Axis Operating Input Memory Areas	These areas are used to input sta- tus information for axis opera- tions, such as present position and axis operation status.	CPU Unit's memory area set in the common parameters.	Data is updated with every I/O refresh of the CPU Unit.

The data handled by the PCU can be classified into the following six types.

Using these data/parameter settings, the PCU executes operation (1) using the operation settings specified in the common parameters, axis parameters, and Servo parameters (2) based on operating commands received from the Axis Operating Memory Areas.

The common parameters, axis parameters, and certain Servo parameters are the basic settings for the PCU and the axes to be controlled. Therefore, these settings must be set when using the PCU. Make the settings for other data/ parameters according to the kind of operation required.

PCU Setting Procedure

	Ste	ер	Operation	Result
1		Set unit num- ber.	Set the unit number on the unit number setting switch and turn ON the power to the PCU.	The Common Operating Memory Area starts from the beginning word obtained using the following equation: $n = CIO 1500 + (unit No. \times 25)$
2		Set common parameters.	Write the common parameters to the PCU and save them in the internal flash memory using the WRITE DATA and SAVE DATA Bits in the Common Operating Memory Area.	The axes being used (scan list) and MECHA- TROLINK communications settings are deter- mined. The Axis Operating Output/Input Memory Areas for the axes to be used are allo- cated.
			After saving the common parameters, restart the PCU or cycle the power to enable the parameters.	
3		Set axis parameters.	Write the axis parameters to the PCU and save them in the internal flash memory using the WRITE DATA and SAVE DATA Bits in the Common Oper- ating Memory Area. (These parame- ters are set at the same time as the common parameters.)	The settings of the origin search operation and other parameters for the axes to be used are determined.
4		Start MECHA- TROLINK communica- tions.	Start MECHATROLINK communica- tions using the CONNECT Bit in the Common Operating Memory Area.	When MECHATROLINK communications start, the axes are operated according to the bits in the Axis Operating Output Memory Areas and the status of the axes can be monitored using the bits in the Axis Operating Input Memory Areas. Subsequent operations are performed with MECHATROLINK communications in progress.
MECHA- 5 TROLINK communi- cations in progress		Set Servo parameters.	Write the Servo parameters using the WRITE SERVO PARAMETER Bit and SAVE SERVO PARAMETER Bit in the Axis Operating Output Memory Areas. Read the Servo parameters using the READ SERVO PARAMETER Bit in the Axis Operating Input Memory Areas.	The settings for each axis or each Servo Param- eter are transferred. The transferred online parameters are enabled when writing is com- pleted, and transferred offline parameters are enabled when the Servo Drive power is cycled or DEVICE SETUP is executed.
6 Operate		Operate axes.	Start operating axes using the bits in the Axis Operating Output Memory Areas.	Commands are executed and status information is obtained when the PLC's I/O is refreshed.
			Monitor the status of axes using the Axis Operating Input Memory Areas.	

4-2 Data Areas

The following tables provide the bit/word addresses for the parameters and data handled by the PCU. For further details, refer to the relevant sections.

Only the Servo parameters for applicable models are listed here. For details on Servo parameters, refer to the operation manual for the model being used.

Common Parameter Area

PCU's	Name	Bits				
internal address		15 to 12	11 to 08	07 to 04	03 to 00	
1838 hex	Area alloca-	Axis Operatin	g Output Mem	ory Area desig	nation	
1839 hex	tions	Beginning word of Axis Operating Output Memory Area				
183A hex		Axis Operating Input Memory Area designation				
183B hex		Beginning wo	rd of Axis Ope	rating Input Me	emory Area	
183C hex	Scan list	Axis 2 allocati	ion	Axis 1 allocat	ion	
183D hex		Axis 4 allocati	ion	Axis 3 allocat	ion	
183E hex		Axis 6 allocati	ion	Axis 5 allocation		
183F hex		Axis 8 allocati	ion	Axis 7 allocation		
1840 hex		Axis 10 allocation		Axis 9 allocation		
1841 hex		Axis 12 allocation		Axis 11 alloca	ation	
1842 hex		Axis 14 allocation		Axis 13 alloca	ation	
1843 hex		Axis 16 allocation Axis 15 allocation		ation		
1844 hex to 1855 hex	Reserved by the system.	Set to 0000.				
1856 hex	MECHA-	Transfer cycle	•	Communication	ons cycle	
1857 hex	TROLINK communica- tions setting	00 (fixed)		C2 master connection	Number of communica- tions retries	
1858 hex to 185F hex	Reserved by the system.	Set to 0000.				

Axis Parameter Area

PCU's		Name		Bits			
internal address			15 to 12	11 to 08	07 to 04	03 to 00	
1860 hex	Axis 1	Input signal selection	Origin input signal	selection	Interrupt input signal selection		
1861 hex		Operation mode selection	Origin search direction	Origin detection method	Origin search operation	0 (fixed)	
			Origin search preset				
1862 hex			0 (fixed)	0 (fixed)	0 (fixed)	Encoder type	
1863 hex to 1873 hex		Reserved by the sys- tem.	Set to 0000.				
1874 hex	Axis 2	Input signal selection	Origin input signal selection		Interrupt input sigr	nal selection	
1875 hex		Operation mode selection	Origin search direction	Origin detection method	Origin search operation	0 (fixed)	
			Origin search preset				
1876 hex			0 (fixed)	0 (fixed)	0 (fixed)	Encoder type	
1877 hex to 1887 hex		Reserved by the sys- tem.	Set to 0000.				
1888 hex to 189B hex	Axis 3	Same as above	Same as above				

Data Areas

Section 4-2

PCU's		Name	Bits			
internal address			15 to 12	11 to 08	07 to 04	03 to 00
189C hex to 18AF hex	Axis 4	Same as above	Same as above			
18B0 hex to 18C3 hex	Axis 5	Same as above	Same as above			
18C4 hex to 18D7 hex	Axis 6	Same as above	Same as above			
18D8 hex to 18EB hex	Axis 7	Same as above	Same as above			
18EC hex to 18FF hex	Axis 8	Same as above	Same as above			
1900 hex to 1913 hex	Axis 9	Same as above	Same as above			
1914 hex to 1927 hex	Axis 10	Same as above	Same as above			
1928 hex to 193B hex	Axis 11	Same as above	Same as above			
193C hex to 194F hex	Axis 12	Same as above	Same as above			
1950 hex to 1963 hex	Axis 13	Same as above	Same as above			
1964 hex to 1977 hex	Axis 14	Same as above	Same as above			
1978 hex to 198B hex	Axis 15	Same as above	Same as above			
198C hex	Axis 16	Input signal selection	Origin input signal	selection	Interrupt input sig	nal selection
198D hex		Operation mode selection	Origin search direction Origin search preset	Origin detection method	Origin search operation	0 (fixed)
198E hex			0 (fixed)	0 (fixed)	0 (fixed)	Encoder type
198F hex to 199F hex		Reserved by the sys- tem.	Set to 0000.			

Servo Parameter Area

The following table provides information on the parameters for the main functions of the PCU that are described in this manual when using an R88D-WT OMRON W-series Servo Drive together with a FNY-NS115 MECHATROLINK-II I/F Unit or an R88D-WN -ML2 W-series Servo Drive with built-in MECHA-TROLINK-II communications. A list of other parameters, G5-series Servo Drive parameters, G-series Servo Drive parameters, and SMARTSTEP Junior Servo Drive parameters are provided in *4-5 Servo Parameter Area*. For further details on each of the parameters, refer to the operation manuals for OMRON G5-series Servo Drive, G-series Servo Drives, W-series Servo Drives, SMARTSTEP Junior Servo Drives, and Yaskawa JUSP-NS115 MECHATROLINK-II I/F Unit.

The default settings for parameters Pn000 to Pn601 in the parameter tables for the R88D-WT W-series Servo Drive used with the FNY-NS115 are automatically set to the default parameter settings used when the FNY-NS115 MECHATROLINK-II I/F Unit is mounted to a W-series Servo Drive. Parameters from Pn800 onwards are enabled when the FNY-NS115 is installed. The parameter names and default settings used for these parameters follow those specified in the FNY-NS115 operation manual.

For details on information provided in the Enable Setting and Details columns, refer to *4-5 Servo Parameter Area*.
Function Selection Parameters

■ <u>R88D-WT</u> and <u>R88D-WN</u>-ML2

Param-	Parame-	Param-		C	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn001	Function selection	2	0	Select stop if an alarm	0	Servomotor stopped by dynamic brake.	2			Offline	
	tion switch			Servomotor is	1	Dynamic brake OFF after Servomotor stopped.					
					2	Servomotor stopped with free run.					
			1	Select stop when prohib- ited drive is input	0	Stop according to Pn001.0 settings (release Servomo- tor after stopping)	0			Offline	
					1	Stop Servomotor using torque set in Pn406, and lock Servomotor after stopping.					
					2	Stop Servomotor using torque set in Pn406, and release Servomotor after stopping.					
			2	Select AC/DC power input			0			Offline	
			3	Select warning code output			1			Offline	
Pn002	Function selection	2	0	Torque com- mand input	0	Option command values not used.	0			Offline	
	applica- tion switch 2			ing speed con- trol)	1	Option command value 1 used as the torque limit input.					
					2	Option command value 1 used as the torque feed forward input.					
					3	Option command values 1 and 2 used as the torque limit input according to the forward/reverse rotation current limit setting.					
			1	Speed com- mand input	0	Option command value not used.	0			Offline	
				cnange (dur- ing torque con- trol)	1	Option command value 1 used as speed limit input.					
			2	Operation switch when	0	Use as absolute encoder.	0			Offline	
				encoder	1	Use as incremental encoder.					
			3	Fully-closed encoder usage method			0			Offline	

Position Control Parameters

■ <u>R88D-WT</u>

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn202	Electronic gear ratio G1 (numerator)	2	Set the pulse rate for the command pulses and Servomotor travel dis-	4		1 to 65535	Offline	
Pn203	Electronic gear ratio G2 (denominator)	2	tance. 0.01 ≤ G1/G2 ≤ 100	1		1 to 65535	Offline	

■ R88D-WN□-ML2

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn20E	Electronic gear ratio G1 (numerator)	4	Sets the pulse rate for the command pulses and Servomotor movement	4		1 to 1073741824	Offline	
Pn210	Electronic gear ratio G2 (denominator)	4	distance. $0.01 \le G1/G2 \le 1000$	1		1 to 1073741824	Offline	

Speed Control Parameters

■ <u>R88D-WT</u> and R88D-WN -ML2

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn305	Soft start acceleration time	2	Sets acceleration time during speed control soft start.	0	ms	0 to 10000	Online	
Pn306	Soft start deceleration time	2	Sets deceleration time during speed control soft start.	0	ms	0 to 10000	Online	

Torque Control (Torque Limit) Parameters

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn402	Forward torque limit	2	Forward rotation output torque limit (rated torque ratio).	350	%	0 to 800	Online	
Pn403	Reverse torque limit	2	Reverse rotation output torque limit (rated torque ratio).	350	%	0 to 800	Online	
Pn404	Forward rotation external current limit	2	Output torque limit during input of forward rotation current limit (rated torque ratio).	100	%	0 to 800	Online	
Pn405	Reverse rotation external current limit	2	Output torque limit during input of reverse rotation current limit (rated torque ratio).	100	%	0 to 800	Online	
Pn406	Emergency stop torque	2	Deceleration torque when an error occurs (rated torque ratio).	350	%	0 to 800	Online	
Pn407	Speed limit	2	Sets the speed limit in torque control mode.	3000	r/min	0 to 10000	Online	

I/O and Status Parameters

■ **R88D-WT**□

Param- eter No. Paramete	Parameter	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn500	Positioning completion range 1	2	Sets the	e range of positio	ning corr	pleted.	3	Com- mand unit	0 to 250	Online	
Pn502	Rotation speed for motor rota- tion detec- tion	2	Sets the detection	e rotation speed f n output (TGON)	ervomotor rotation	20	r/min	1 to 10000	Online		
Pn503	Speed con- formity sig- nal output width	2	Sets the s	e allowable fluctu Speed Conformity	ation ran y Flag.	ge (rotation speed)	10	r/min	0 to 100	Online	
Pn504	Positioning completion range 2	2	Sets the Flag.	e proximity range	for the P	ositioning Proximity	3	Com- mand unit	1 to 250	Online	

Section 4-2

Param- eter No. Parameter name	Param-		C	ontents		Default	Unit	Setting	Enable	Details								
eter No.	name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting								
Pn50A	Input sig- nal selec- tion 1	2	0	Not used.	1	(Do not change the setting.)	1				Fixed setting: 1							
			1	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8							
			2	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8							
			3	POT (forward drive prohib- ited input) sig-	0	Allocated to CN1, pin 40: Valid for low input	8			Offline	Stan- dard setting:							
				nal Input terminal allo- cation	1	Allocated to CN1, pin 41: Valid for low input					2							
					2	Allocated to CN1, pin 42: Valid for low input												
					3	Allocated to CN1, pin 43: Valid for low input												
					4	Allocated to CN1, pin 44: Valid for low input												
					5	Allocated to CN1, pin 45: Valid for low input												
					6	Allocated to CN1, pin 46: Valid for low input												
					7	Always valid.												
					8	Always invalid.												
					9	Allocated to CN1, pin 40: Valid for high input												
				A	Allocated to CN1, pin 41: Valid for high input													
										В	Allocated to CN1, pin 42: Valid for high input							
																С	Allocated to CN1, pin 43: Valid for high input	
						D	Allocated to CN1, pin 44: Valid for high input											
			E	E	Allocated to CN1, pin 45: Valid for high input	N1, r												
					F	Allocated to CN1, pin 46: Valid for high input												

Section 4-2

Param-	Parameter	Param-		Cc	ontents		Default	Unit	Setting	Enable	Details
eter No.	name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn50B	Input sig- nal selec- tion 2	2	0	NOT (reverse drive prohib- ited) signal input terminal allocation	0 to F	Same as Pn50A.3	8			Offline	Stan- dard setting: 3
			1	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8
			2	PCL (forward rotation current limit) signal input terminal allocation	0 to F	Same as Pn50A.3	5			Offline	Fixed setting: 8
			3	NCL (reverse rotation current limit) signal input terminal allocation	0 to F	Same as Pn50A.3	6			Offline	Fixed setting: 8
Pn50E	Output sig-	2	0	INP1 (position-	0	Not used.	1			Offline	Stan-
	nal selec- tion 1			1) signal out- put terminal	1	Allocated to CN1, pins 25, 26					dard setting: 1
				allocation	2	Allocated to CN1, pins 27, 28					
					3	Allocated to CN1, pins 29, 30					
			1	VCMP (speed conformity) signal output terminal allo- cation	0 to 3	Same as Pn50E.0	1			Offline	Stan- dard setting: 0
			2	TGON (Servo- motor rotation detection) sig- nal output ter- minal allocation	0 to 3	Same as Pn50E.0	2			Offline	Stan- dard setting: 0
			3	READY (Ser- vomotor war- mup complete) signal output terminal allo- cation	0 to 3	Same as Pn50E.0	3			Offline	Stan- dard setting: 3
Pn50F	Output sig- nal selec- tion 2	2	0	CLIMT (current limit detection) signal output terminal allo- cation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
			1	VLIMT (speed limit detection) signal output terminal allo- cation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
			2	BKIR (brake interlock) sig- nal output ter- minal allocation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 2
			3	WARN (warn- ing) signal out- put terminal allocation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0

Section 4-2

Param- eter No.	Parameter	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn510	Output sig- nal selec- tion 3	2	0	INP2 (position- ing completed 2) signal out- put terminal allocation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
			1	Not used.	0	(Do not change the setting.)	0				
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				
Pn511	Input sig- nal selec- tion 5	2	0	DEC (origin return deceler- ation limit switch) signal input terminal allocation	0 to F	Same as Pn50A.3	8			Offline	Stan- dard setting: 1
			1	EXT1 (exter- nal latch 1 input) signal input terminal allocation	0 to F	Same as Pn50A.3 (0 to 3 and 9 to C are always dis- abled.)	8			Offline	Stan- dard setting: 4
			2	EXT2 (exter- nal latch 2 input) signal input terminal allocation	0 to F	Same as Pn50A.3 (0 to 3 and 9 to C are always dis- abled.)	8			Offline	Stan- dard setting: 5
			3	EXT3 (exter- nal latch 3 input) signal input terminal allocation	0 to F	Same as Pn50A.3 (0 to 3 and 9 to C are always dis- abled.)	8			Offline	Stan- dard setting: 6

■ <u>R88D-WN□-ML2</u>

Param- Para eter No. ter n	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn502	Rotation speed for motor rota- tion detec- tion	2	Sets th rotation	ne number of rota n detection outpu	tions fo t (TGOI	r the Servomotor N).	20	r/min	1 to 10000	Online	
Pn503	Speed conformity signal out- put width	2	Sets th for the	ne allowable fluctu speed conformity	uation (r / output	number of rotations) (VCMP).	10	r/min	0 to 100	Online	

Section 4-2

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details												
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting													
Pn50A	Input sig- nal selec- tions 1	2	0	Not used.	1	(Do not change the setting.)	1				Fixed setting: 1												
			1	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8												
			2	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8												
			3	POT (forward drive prohibited input) signal	0	Allocated to CN1, pin 13: Valid for low input	1			Offline	Stan- dard setting:												
				allocation	1	Allocated to CN1, pin 7: Valid for low input					1												
					2	Allocated to CN1, pin 8: Valid for low input																	
					3	Allocated to CN1, pin 9: Valid for low input																	
				4	Allocated to CN1, pin 10: Valid for low input																		
					5	Allocated to CN1, pin 11: Valid for low input																	
					6	Allocated to CN1, pin 12: Valid for low input																	
							7	Always enabled.															
							8	Always disabled.															
					9	Allocated to CN1, pin 13: Valid for high input																	
							A	Allocated to CN1, pin 7: Valid for high input															
					В	Allocated to CN1, pin 8: Valid for high input																	
					С	Allocated to CN1, pin 9: Valid for high input	_																
																	D	Allocated to CN1, pin 10: Valid for high input	-				
														E	Allocated to CN1, pin 11: Valid for high input	-							
				F	Allocated to CN1, pin 12: Valid for high input																		
Pn50B	Input sig- nal selec- tions 2	2	0	NOT (reverse drive prohibited input) signal Input terminal allocation	0 to F	Same as Pn50A.3.	2			Offline	Stan- dard setting: 2												
			1	Not used.	8	(Do not change the setting.)	8																
			2	Not used.	8	(Do not change the setting.)	8																
			3	Not used.	8	(Do not change the setting.)	8																

Section 4-2

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn50E	Output sig-	2	0	INP1 (position-	0	Not used.	0			Offline	Stan-
	tions 1			1) signal output terminal alloca-	1	Allocated to CN1 pins 1, 2					dard setting: 0
				tion	2	Allocated to CN1 pins 23, 24					
					3	Allocated to CN1 pins 25, 26					
			1	VCMP (speed conformity) signal output terminal alloca- tion	0 to 3	Same as Pn50E.0.	0			Offline	Stan- dard setting: 0
			2	TGON (servo- motor rotation detection) sig- nal output ter- minal allocation	0 to 3	Same as Pn50E.0.	0			Offline	Stan- dard setting: 0
			3	READY (servo ready) signal output termi- nal allocation	0 to 3	Same as Pn50E.0.	0			Offline	Stan- dard setting: 0
Pn50F	Output sig- nal selec- tions 2	2	0	CLIMT (cur- rent limit detec- tion) signal output termi- nal allocation	0 to 3	Same as Pn50E.0.	0			Offline	Stan- dard setting: 0
			1	VLIMT (speed limit detection) signal output terminal alloca- tion	0 to 3	Same as Pn50E.0.	0			Offline	Stan- dard setting: 0
			2	BKIR (brake interlock) sig- nal output ter- minal allocation	0 to 3	Same as Pn50E.0.	1			Offline	Stan- dard setting: 1
			3	WARN (warn- ing) signal out- put terminal allocation	0 to 3	Same as Pn50E.0.	0			Offline	Stan- dard setting: 0
Pn510	Output sig- nal selec- tions 3	2	0	INP2 (position- ing completed 2) signal output terminal alloca- tion	0 to 3	Same as Pn50E.0.	0			Offline	
			1	Not used.	0	(Do not change the setting.)	0				
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				
Pn511	Input sig- nal selec- tions 5	2	0	DEC signal input terminal allocation	0 to F	Same as Pn50A.3.	3			Offline	Stan- dard setting: 3
			1	EXT1 signal input terminal allocation	0 to F	Same as Pn50A.3. (0 to 3 and 9 to C are always dis- abled.)	4			Offline	Stan- dard setting: 4
			2	EXT2 signal input terminal allocation	0 to F	Same as Pn50A.3. (0 to 3 and 9 to C are always dis- abled.)	5			Offline	Stan- dard setting: 5
			3	EXT3 signal input terminal allocation	0 to F	Same as Pn50A.3. (0 to 3 and 9 to C are always dis- abled.)	6			Offline	Stan- dard setting: 6

Section 4-2

Param-	Parame-	Param-		Co	ntents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn522	Position- ing com- pleted range 1	4	Setting	range for positio	ning co	mpleted range	3	Com- mand unit	0 to 1,073,741,8 24	Online	
Pn524	Position- ing com- pleted range 2	4	Setting imity F	for proximity ran lag.	ge for th	ne Positioning Prox-	3	Com- mand unit	1 to 1,073,741,8 24	Online	

Control Function Parameters

■ R88D-WT and R88D-WN -ML2)

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn800	Communi- cations control	2	0	MECHA- TROLINK-II communica- tions check mask	0	Detects both com- munications errors (A.E6) and syn- chronization errors (A.E5).	0			Online	Always set to 0.
					1	Ignores communi- cations errors (A.E6).					
					2	Ignores synchroni- zation errors (A.E5).					
					3	Ignores both com- munications errors (A.E6) and syn- chronization errors (A.E5).					
			1	Warning check mask	0	Detects parameter setting warnings (A.94), MECHA- TROLINK com- mand warnings (A.95), and com- munications errors (A.96).	4			Online	Always set to 4 or 0.
					1	Ignores parameter setting warnings (A.94).					
					2	Ignores MECHA- TROLINK-II com- mand warnings (A.95).					
					3	Ignores both parameter setting warnings (A.94) and MECHA- TROLINK-II com- mand warnings (A.95).					
					4	Ignores communi- cations errors (A.96).					
					5	Ignores both parameter setting warnings (A.94) and communica- tions errors (A.96).					
					6	Ignores both MECHATROLINK- II command warn- ings (A.95) and communications errors (A.96).					
					7	Ignores parameter setting warnings (A.94), MECHA- TROLINK-II com- mand warnings (A.95), and com- munications errors (A.96).					
			2	Communica- tions error count at single transmission			0			Online	
			3	Not used.		(Do not change the setting.)	0				

Param- P eter No. te	Parame-	Param-		C	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn801	Function selection	2	0	Software limit function	0	Software limit enabled.	(See note.)			Online	
	applica- tion (soft- ware				1	Forward software limit disabled.					
limit	limits)				2	Reverse software limit disabled.					
					3	Software limit dis- abled in both direc- tions.					
			1	Not used.		(Do not change the setting.)	0				
			2	Software limit check using references	0	No software limit check using refer- ences.	0			Online	Always set to 0.
					1	Software limit check using refer- ences.					
			3	Not used.		(Do not change the setting.)	0				

Note

R88D-WT□ with FNY-NS115: 0 R88D-WN□-ML2: 3

Param- eter No.	Parameter name	Param- eter size	Contents	Default set- ting	Unit	Setting range	Enable setting	Details
Pn803	Zero point width	2	Sets the detection range for the Origin Stop Flag.	10	Command unit	0 to 250	Online	
Pn804	Forward soft- ware limit	4	Sets the forward soft- ware limit.	819,191,808	Command unit	-1,073,741,823 to 1,073,741,823	Online	
Pn806	Reverse soft- ware limit	4	Sets the reverse soft- ware limit.	-819,191,808	Command unit	-1,073,741,823 to 1,073,741,823	Online	
Pn808	Absolute encoder zero point position offset	4	Sets the offset for the mechanical origin from the absolute encoder's absolute value data.	0	Command unit	-1,073,741,823 to 1,073,741,823	Offline	
Pn80A	First-step lin- ear accelera- tion constant	2	Sets the first-step acceleration speed for the acceleration/deceleration curve used in position control.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80B	Second-step linear accelera- tion constant	2	Sets the second-step acceleration speed for the acceleration/deceler- ation curve used in posi- tion control.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80C	Acceleration constant switching speed	2	Sets the speed for switching between first- step and second-step acceleration for the acceleration/decelera- tion curve used in posi- tion control.	0	×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).

Section 4-2

Param- eter No.	Parameter name	Param- eter size	Contents	Default set- ting	Unit	Setting range	Enable setting	Details
Pn80D	First-step lin- ear decelera- tion constant	2	Sets the first-step decel- eration speed for the acceleration/decelera- tion curve used in posi- tion control.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80E	Second-step linear decelera- tion constant	2	Sets the second-step deceleration speed for the acceleration/deceler- ation curve used in posi- tion control.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80F	Deceleration constant switching speed	2	Sets the speed for switching between first- step and second-step deceleration for the acceleration/decelera- tion curve used in posi- tion control.	0	×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn810	Exponential acceleration/ deceleration bias	2	Sets the exponential acceleration/decelera- tion bias speed for the acceleration/decelera- tion curve used in posi- tion control.	0	Command units/s	0 to 32767	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn811	Exponential acceleration/ deceleration time constant	2	Sets exponential accel- eration/deceleration time constant for the acceler- ation/deceleration curve used in position control.	0	×0.1 ms	0 to 5100	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn812	Movement average time	2	Sets the S-curve accel- eration/deceleration moving average time for the acceleration/deceler- ation curve used in posi- tion control.	0	×0.1 ms	0 to 5100	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn814	Final travel dis- tance for exter- nal positioning	4	Sets the final travel dis- tance for external posi- tioning when performing interrupt feeding using direct operation.	100	Command unit	-1,073,741,823 to 1,073,741,823	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).

Section 4-2

Param-	Parame-	Param-		Cor	ntents		Default	Unit	Setting range	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting			setting	
Pn816	Zero point	2	0	Zero point	0	Forward	0			Online	Set the
	return mode set- ting			return direction	1	Reverse					same direction as the ori- gin search direction set in the Axis Parame- ters. Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
			1 to 3	Not used.		(Do not change the setting.)	0				
Pn817	Zero point return approach speed 1	2	Sets the search ity signa search.	e origin (zero speed used a al has been d	point) in after the letected i	put signal origin proxim- in an origin	50	×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn818	Zero point return approach speed 2	2	Sets the origin (zero p travel distance positio the origin input signal an origin search.		point) re oning spe al has bee	eturn final eed used after en detected in	5	×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn819	Final travel distance to return to zero point	4	Sets the ing use been de	e amount of c d after the or etected in an	compens igin input origin se	ation position- t signal has arch.	100	Com- mand unit	-1,073,741,823 to 1,073,741,823	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).

Note The parameters for backlash compensation are different for the R88D-WT and R88D-WN \square -ML2.

■ <u>R88D-WT</u>

Param-	Parame-	Param-		(Contents	5	Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn81B	Backlash compen- sation amount	2	Sets the	e amount of b	backlash	compensation.	0	×0.1 com- mand unit	-32,768 to 32,767	Online	
Pn81D	n81D Compen- 2 0 sation		0	Backlash compensa-	0	Compensates in forward direction.	0			Offline	
	selection			tion selec- tion	1	Compensates in reverse direction.					
	1 to 3 Not used. (Do not change th setting.)		(Do not change the setting.)	0							

■ R88D-WN□-ML2

Param-	Parame-	Param-		(Contents	6	Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn207	Position control	2	0	Not used.	0	(Do not change the setting.)	0				
	settings 2		1	Not used.	1	(Do not change the setting.)	1				
			2	Backlash	0	Disabled	0			Offline	
				compensa- tion selec- tion	1	Compensates to forward rotation side.					
					2	Compensates to reverse rotation side.					
			3	INP 1 out- put timing			0			Offline	
Pn214 B C si a	Backlash compen- sation amount	2	Mechar mechar shaft be	nical system b nical gap betv eing driven)	oacklash veen the	amount (the drive shaft and the	0	Com- mand unit	-32767 to 32767	Online	
Pn215	Backlash compen- sation time constant	2	Sets the	e backlash co	ompensa	tion time constant.	0	× 0.01 ms	0 to 65535	Online	

Common Operating Memory Area

Beginning word of Common Operating Memory Area: n = CIO 1500 + (unit number \times 25)

15 14 13 12 11 10 03 06 05 04 03 02 01 00 n Not used (reserved by the system).	Word								Na	ame							
n Not we (reserved by the system). READ (nero, note) READ (note)		15	14	13	12	11	10	09	08	07	06	05	04	03	02	01	00
n+1 READ show Not used (reserved by the system). READ show READ show Reserved by the system. CON- NECT n+2 Axes to connect (See note). Image: Source of the system image: S	n	Not use	d (reserv	ed by the	system).									SAVE DATA	READ DATA	WRITE DATA	UNIT ERROR RESET
n+2 Axes to connect (See note.) n+3 Not used (reserved by the system). n+6 Number of write vords n+7 Write source area n+8 Write source vord n+8 Write source area n+9 Write destination address n+10 Number of read vords n+11 Read source address n+12 Read destination = n+13 Read destination = n+14 Not used (reserved by the system). n+15 Read destination = n+14 Not used (reserved by the system). n+15 Read destination = n+16 Not n+17 Read destination = n+18 Read destination = n+14 Not used (reserved by the system). n+15 Read destination = n+16 Not n+17 Read destination = n+18 Read destination = n+19 Not served by the system Not used (reserved by the system). reserved by the system Card system) reserved by the system. Ca	n+1	REJOI N (See note.)	Not use	d (reserve	ed by the	system).				READ BACK- UP DATA	WRITE BACK- UP DATA	Reserv	red by the	system.			CON- NECT
n+3 to n+5 Not used (reserved by the system). n+6 Number of write words Image: System of Write System of Write System of Write System of Write System of Sy	n+2	Axes to	connect	(See note	.)												
n+6 Number of write words Use words n+7 Write surce words Vertee definition n+8 Write definition Vertee definition n+9 Write definition Vertee definition n+10 Number of read words Vertee definition n+11 Read definition Vertee definition n+12 Read definition Vertee definition n+13 Read definition Vertee definition n+14 Not used (reserved by the system). Vertee definition n+15 Not used (reserved by the system). Vertee definition n+16 Con- Status Not used (reserved by the system). n+16 Softer definition Softer definition n+16 Softer definition Softer definition n+16 Softer definition Softer definition n+16 Reserved by the system. n+17 Reserved by the system. n+21 Unit error definition	n+3 to n+5	Not use	d (reserv	ed by the	system).												
n+7 Write source area n+8 Write source word n+9 Write destination destination n+10 Number of read words second and and and and and and and and and a	n+6	Number	of write	words													
n+8 Write source word n+9 Write destination address n+10 Number of read words n+11 Read source address n+12 Read destination area n+13 Read destination area n+14 Not area Not green Unit n+15 Not green Not green Unit green Unit green n+16 Con- Status Unit served Not green served Reserved by the system. n+17 to n-20 Reserved by the system. Perform Flag Reserved by the system. n+21 Unit used (re- served Not green served Reserved by the system. n+16 Reserved by the system. Reserved by the system. n+17 to n+20 Unit green served Reserved by the system. n+21 Unit error code Unit error served Not green served	n+7	Write source area Write source word															
n+19 Write destination address Image: second	n+8	Write so	Write source word Write destination address														
n+10Numer of read with sum of r	n+9	Write destination address															
n+11Read surve add s	n+10	Number of read words															
n+12Read destination realn+13Read destination realn+14Not set (reserved by the system).n+14Not set (reserved by the system).n+15Not used (real real real real real real real real	n+11	Number of read words Read source address															
n+13Read destination wordn+14Not used (reserved by the system).n+15Not used (reserved by the system).n+15Not used (reserved by the system).n+16One- tem).Not used (reserved by the system).n+16Not Unit Status FlagNot used (reserved by the system).n+17Reserved by the system.n+18Reserved by the system.n+19Not Unit Status FlagNot used (reserved by the system.n+11Not Unit Status FlagNot used (reserved by the system.n+21Unit error code	n+12	Read de	estination	area													
n+14 Not used (re-structure by the system). n+15 Not used (re-structure by the system). Not used (re-structure by the system). n+15 Not used (re-structure by the system). Fing system) n+16 Or-nec-strem). Not used (re-structure by the system). n+16 Not Usey (re-structure by the system). Not used (reserved by the system). n+16 Not Usey (re-structure by the system). Not used (reserved by the system). n+16 Status Flag Not Usey Used (re-structure by the system). n+17 Not Usey Used (re-structure by the system). Peserved by the system. n+17 Not Useg (re-structure by the system). Peserved by the system. n+21 Unit error code Vinit error code	n+13	Read de	estination	word													
n+15Not used (re- sys- tem)Data Trans- ferring FlagNot used (re- sys- tem)Not used (reserved by the system).n+16Con- nec- ton Status FlagUnit 	n+14	Not use	d (reserv	ed by the	system).												
n+16 Con- nec- status Flag Unit Busy Flag Not used (re- sys- tem). Mem- ory Card by the system. Reserved by the system. n+17 to n+20 Reserve by the system. Image: System of the system. Image: System of the system. n+17 to n+20 Unit error code Unit error code Image: System of the system.	n+15	Not used (re- served by the sys- tem).	Data Trans- ferring Flag	Not used (re- served by the sys- tem).	Unit Error Flag	Not use	d (reserv	ved by the	system).								
n+17 to n+20 Reserved by the system. n+21 Unit error code	n+16	Con- nec- tion Status Flag	Unit Busy Flag	Not used (re- served by the sys- tem).	Mem- ory Card trans- fer error	Reserve	ed by the	system.									
n+21 Unit error code	n+17 to n+20	Reserve	ed by the	system.		-											
	n+21	Unit erro	or code														

Section 4-2

Word								Na	me							
	15	14	13	12	11	10	09	08	07	06	05	04	03	02	01	00
n+22	Axis con	Axis communications status														
n+23 n+24	Not use	d (reserve	ed by the	system).												

Note The REJOIN Bit and the Axes to Connect parameter are supported for unit version 2.0 or later. The allocated bit and word are not used for earlier unit versions.

Axis Operating Output Memory Areas

Beginning Word of Axis Operating Output Memory Areas:

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) \times 25

Word								Na	me							
	15	14	13	12	11	10	09	08	07	06	05	04	03	02	01	00
a	DE- CEL- ERA- TION STOP	Over- ride En- able Bit	DEVIA- TION COUN- TER RE- SET (See note 4.)	ER- ROR RESET	Not used (re- served by the sys- tem).	Direc- tion desig- nation	JOG	PRESENT POSI- TION PRESET	ORI- GIN RE- TURN	ORIGIN SEARCH	INTER- RUPT FEED- ING	RELA- TIVE MOVE- MENT	ABSO- LUTE MOVE- MENT	Not used (re- served by the sys- tem).	LIN- EAR INTER- POLA- TION START (See note 1.)	LINEAR INTER- POLA- TION SET- TING (See note 1.)
a+1	EMER- GEN- CY STOP	SAVE SER- VO PA- RAME- TER	READ SER- VO PA- RAME- TER	WRITE SER- VO PA- RAME- TER	DE- VICE SETUP	Not use	d (reserve	ed by the	system).				TORQUE CON- TROL	SPEED CON- TROL	SER- VO UN- LOCK	SER- VO LOCK
a+2	Position	comman	d value													
a+3																
a+4	Speed c	ommand	value (for	r position	control)											
a+5																
a+6	Speed c	ommand	value (for	r speed c	ontrol)											
a+7																
a+8	Torque command value															
a+9																
a+10	Option c	command	value 1													
a+11	0															
a+12	Option c	command	value 2													
a+13	Override															
a+14	Notuco	; d (recorve	d by the	system)					Monitor	2 tupo			Monitor	1 tupo		
a+15	Ro Ro	Eor		d by the	evetom				WOINTO	z type		S ourvo	Expo	Rosonua	d by the	avetom
a+10	verse rota- tion current limit	ward rota- tion current limit	neserve		system.							desig- nation	nential curve desig- nation	neserve	d by the s	system.
a+17	Servo P	arameter	No.													
a+18	Paramet	ter size														
a+19	Write da	ta (Servo	Paramet	ers)												
a+20																
a+21	Not use	d (reserve	ed by the	system).					Interpola note 3.)	ation axis	designati	ion (See	Interpola note 2.)	ation axis	designati	ion (See
a+22	Not use	d (reserve	ed by the	system).					Interpola (See not	ation posi te 3.)	tion desig	nation	Interpola (See not	ation posi te 2.)	tion desig	nation
a+23	Interpola	ation spee	ed comma	and value	(See not	e 1.)										
a+24																

Note

- (1) Allocated in Axis Operating Output Memory Areas for axis 1 and axis 5 for Position Control Unit Ver. 1.1 or later. These bits are not used in the Axis Operating Output Memory Areas for other axes.
- (2) Allocated in Axis Operating Output Memory Area for axis 1 for Position Control Unit Ver. 1.1 or later. These bits are not used in the Axis Operating Output Memory Areas for other axes.

- (3) Allocated in Axis Operating Output Memory Area for axis 5 for Position Control Unit version 1.1 or later. These bits are not used in the Axis Operating Output Memory Areas for other axes.
- (4) The DEVIATION COUNTER RESET can be used with unit version 1.3 or later. This bit is not used for earlier unit versions.

Axis Operating Input Memory Areas

Beginning Word of Axis Operating Input Memory Areas:

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

Word								Na	ime							
	15	14	13	12	11	10	09	08	07	06	05	04	03	02	01	00
b							Ax	is Control	Status F	lags						
	Stop Execu- tion Flag	Servo Param- eters Trans- ferring Flag	Busy Flag	Error Flag	Warn- ing Flag	Not use system)	d (reserve	∋d by the	Origin Stop Flag	No Ori- gin Flag	PCU Posi- tioning Com- pleted Flag	Not use	d (reserve	ed by the	system).	Re- ceiv- ing Com- mand Flag
b+1						Servo S	Status Fla	igs (statu	s particula	ar to Serv	/o Drive)					
	Reserve system.	d by the	Re- verse Soft- ware Limit Flag	For- ward Soft- ware Limit Flag	Posi- tioning Prox- imity Flag/ Speed Limit Status Flag	Re- served by the sys- tem.	Torque Limit Status Flag	Distri- bution Com- pleted Flag/ Zero Speed Flag	Posi- tion Com- pleted Flag/ Speed Con- formi- ty Flag	Reserve system.	∍d by the	Main Power ON Flag	Servo ON Flag	Reserve	d by the ∶	system.
b+2	Reserved by the system.															
b+3							E	xternal I/C) Status E	3its						
	Reserve	ed by the s	system.			Emer- gency stop input (See note 2.)	Brake output	Exter- nal latch signal 3 input	Exter- nal latch signal 2 input	Exter- nal latch signal 1 input	Encod- er Phase Z input	En- coder Phase B input	En- coder Phase A input	Origin prox- imity input signal	Re- verse rota- tion limit in- put	For- ward rota- tion limit input
b+4	Axis err	or code												L		
b+5	Not use	d (reserve	ed by the	system).					Monitor	2 type			Monitor	1 type		
b+6	Feedbad	ck presen	t position													
b+7																
b+8	Comma	nd preser	nt positior	۱												
b+9	L															
b+10	Monitor	1														
b+11	<u> </u>															
b+12	Monitor	2														
D+13	Deed de	ata (Carrie	Daramai	to ro)												
D+14 b±15	neau ua	ila (Servo	Farame	lers)												
b+16	Not use	d (reserve	ed by the	system)												
to b+23																
b+24	Not use served t system)	d (re- y the	Linear inter- pola- tion ex- ecut- ing (See note 1.)	Not use	d (reserve	ed by the	system).									Linear inter- pola- tion setting com- pleted (See note

Note

- (1) Allocated in Axis Operating Input Memory Areas for axis 1 and axis 5 for Position Control Unit Ver. 1.1 or later. These bits are not used in the Axis Operating Output Memory Areas for other axes.
- (2) The emergency stop input status is input only for SMARTSTEP Junior Servo Drives. This status is not used (reserved by the system) by W-series Servo Drives.

4-3 Common Parameter Area

Common Parameters are used for basic settings for operating the PCU, such as allocation of the Axis Operating Memory Areas and settings for MECHA-TROLINK communications. The common parameters must be set when using the PCU.

4-3-1 Common Parameters Overview

Common Parameters are transferred to the PCU's internal memory using the data transfer command and saved in the PCU's flash memory using the SAVE DATA command. After setting and saving the Common Parameters, either cycle the power to the CPU Unit, or restart the PCU. This operation will enable the set parameters. The Common Parameter Area is used to set the following information.

PCU's	Con	tents		Setting	
address	Bits 08 to 15	Bits 00 to 07			
1838 hex	Axis Operating Outpu nation	t Memory Area desig-	Specifies the area allocate Memory Areas.	ed for the Axis Operating Output	
			0000 hex: No setting	00B0 hex: CIO Area	
			00B1 hex: Work Area	00B2 hex: Holding Area	
			00B3 hex: Auxiliary Area	0082 hex: DM Area	
			0050 to 0059, 005A, 005E	3, 005C hex: EM Area (5⊡: □ = EM Bank No.)	
1839 hex	Beginning word of Ax Memory Areas	s Operating Output	Specifies the beginning w Memory Areas.	ord of the Axis Operating Output	
			The beginning word of the Axis 1 is determined by th Area designation and the ter. Each Axis is allocated word.	e Operating Output Memory Area for he Axis Operating Output Memory word determined using this parame- 25 words in sequence from this	
183A hex	Axis Operating Input N tion	Memory Area designa-	Specifies the allocated are Operating Input Memory	ea and beginning word for the Axis Areas.	
183B hex	Beginning word of Axi Memory Areas	s Operating Input	The setting method is the same as for the Axis Operating Out- put Memory Areas.		
183C hex to 1843 hex	Scan list		Sets the axis allocations for axes 1 to 16 with 8 bytes per axis, as follows:		
			00 hex: Axis not used (de	fault setting)	
			40 hex: Allocates axis to t	he Servo Drive.	
1856 hex	Transfer cycle	Communications cycle	Transfer cycle: 00 hex: 1 ms (default se 01 hex: 1 ms 02 hex: 2 04 hex: 4 ms 05 hex: 5 07 hex: 7 ms 08 hex: 8 A5 hex: 0.5 ms	etting) 2 ms 03 hex: 3 ms 5 ms 06 hex: 6 ms 3 ms A2 hex: 0.25 ms	
			Communications cycle: Sets a multiplier used to fer cycle. Set value: 00 to 20 hex The default setting 00 is 3. Set as follows: Transfer cycle x commu	obtain integer multiples of the trans- the same as when the cycle is set to nications cycle (multiplier) \leq 32 ms	

PCU's	Contents			Setting
address	Bits 08 to 15	Bits 0	0 to 07	
1857 hex	00 (fixed)	07 to 04	03 to 00	C2 master connection:
		C2 mas- ter con- nection	Number of com- munica- tions retries	0: No C2 master (default setting) 1: C2 master connected Number of communications retries: Set value: 0 to 7, F The default setting 0 is the same as when the number of retries is set to 1.

4-3-2 Common Parameter Details

The data set in the Common Parameter Area is as follows:

Unit address	Name	Enable timing
1838 hex	Axis Operating Output Memory Area designa- tion	At powerup or restart.
1839 hex	Beginning word of Axis Operating Output Mem- ory Areas	
183A hex	Axis Operating Input Memory Area designation	
183B hex	Beginning word of Axis Operating Input Memory Areas	

Data con- figuration	15 08 07	00	Data setting range	Default
1838 hex	Axis Operating Output Memory Area dea	signation	00B0 to 00B3, 0082, 0050 to 005C hex	0000
1839 hex	Beginning word of Axis Operating Output Areas	Memory	0000 to 7FFF hex	0000
183A hex	Axis Operating Input Memory Area des	ignation	00B0 to 00B3, 0082, 0050 to 005C hex	0000
183B hex	Beginning word of Axis Operating Input N Areas	lemory	0000 to 7FFF hex	0000

Data Details

Axis Operating Output Memory Area Designation

Specifies the words allocated for the Axis Operating Output Memory Areas.

0000 hex:No setting (default setting)00B0 hex:CIO Area00B1 hex:Work Area00B2 hex:Holding Area00B3 hex:Auxiliary Area0082 hex:DM Area0050 to 0059, 005A, 005B, 005C hex:EM Area (005□ hex: □ = EM Bank No.)

Beginning Word of Axis Operating Output Memory Areas

Specifies the beginning word of the Axis Operating Output Memory Areas.

The Axis Operating Output Memory Area designation and the word determined using this parameter are used as the beginning word of the Operating Output Memory Area for Axis 1. Each axis is allocated 25 words in sequence up to the highest axis number registered in the scan list.

Axis	MECHATROLINK	
No.	station address No.	
Axis 1	No. 1	
Axis 2	No. 2	<u>1</u> `~
Axis 3	No. 3	N
Axis 4	No. 4	1 \
:	:	
Axis 14	No. 14	
Axis 15	No. 15	
Axis 16	No. 16	Ν.

	1	Area Allegati			Allegations	ig input memory Area
OLINK		Area Allocati	ons		Allocations	
ress No.		Axis 1	Word a+0		Axis 1	Word b+0
1		Operating	Word a+1		Operating	Word b+1
2		Output	:		Input Memory	:
3	$\langle N \rangle$	Memory Area	Word a+24		Area	Word b+24
4		Axis 2	Word a+25		Axis 2	Word b+25
	$\langle \rangle$	Operating	Word a+26		Operating	Word b+26
14		Output	:		Input Memory	:
15		Memory Area	Word a+49		Area	Word b+49
16	N.		-			
	I/	Axis N	Word a+(N-1)×25		Axis N	Word b+(N-1)×25
	$\langle \rangle$	Operating	Word a+(N-1)×25+1		Operating	Word b+(N-1)×25+1
		Output	:		Input Memory	:
	-1	Memory Area	Word a+(N-1)×25+24		Area	Word b+(N-1)×25+24
	-1					
	$\langle \rangle$	Axis 16	Word a+375		Axis 16	Word b+375
		Operating	Word a+376		Operating	Word b+376
	Ì	Output	:		Input Memory	:
	Y	Memory Area	Word a+399		Area	Word b+399

Axis Operating Output Memory Axis Operating Input Memory Area

a: Beginning word of Axis Operating Output Memory Areas specified in Common Parameters.

b: Beginning word of Axis Operating Input Memory Areas specified in Common Parameters.

Axis Operating Input Memory Area Designation:

Specifies the words allocated for the Axis Operating Input Memory Areas.

The setting method is the same as for the Axis Operating Output Memory Areas.

Beginning Word of Axis Operating Input Memory Areas:

Specifies the beginning word of the Axis Operating Input Memory Areas.

The setting method is the same as for the Axis Operating Output Memory Areas.

Note

(1) Set the beginning word of the Axis Operating Output Memory Areas and Axis Operating Input Memory Areas in the Common Parameters so that the words allocated to each area do not exceed the upper limit of the range for each of the CPU Unit's I/O memory areas. CPU Unit I/O Memory Areas CIO Area: CIO 0000 to CIO 6143 Work Area: W000 to W511 Holding Area: H000 to H511 Auxiliary Area: A000 to A959 DM Area: D00000 to D32767 EM Area: E_{00000} to E_{32767} ($\Box = EM Bank No.$) The maximum set value for the beginning word of the Axis Operating Output/Input Memory Area is calculated as follows: Maximum number of words in each area – Highest axis No. registered in scan list \times 25 + 1 (2) If the setting exceeds the range of the I/O memory area, an Initialization Common Parameter Check Error (Unit error code 0028) will occur when the PCU power is turned ON or the Unit is restarted. Example: Beginning word of the Axis Operating Output Memory Area: CIO 6100 Connected axes: 2 min. Axis 1: Output Area: CIO 6100 to CIO 6124 Axis 2: Output Area: CIO 6125 to CIO 6149 The highest word in the CIO Area is CIO 6143. Therefore, an error will occur.

(3) If the ranges set for the Axis Operating Output Memory Area and Axis Operating Input Memory Area overlap, an Initialization Common Parameter

Check Error (Unit error code 0028) will occur when the PCU power is turned ON or the Unit is restarted.

(4) Do not set the bank number of the EM Area that is being saved to file memory in the CPU Unit. When the EM Area for the bank saved to file memory has been specified, the information in the Operating Data Area will not be reflected, resulting in a malfunction. If multiple PCUs are mounted to a single PLC, make sure that the Operating Data Areas do not overlap. The PCU will not detect an error if the Operating Data Areas overlap, which may result in a malfunction.

Setting Example



Details

Refer to 6-2-2 Scan List and PCU Area Allocations.

Unit address		Name		En	able timing	
183C hex to 1843 hex		Scan list		At p rest	powerup or start.	
Data con- figuration	15	08 07 00	Data setting rar	nge	Default	
183C hex	Axis	s 2 allocation Axis 1 allocation	00 or 40		0000	
183D hex	Axis	s 4 allocation Axis 3 allocation	00 or 40		0000	
183E hex	Axis	s 6 allocation Axis 5 allocation	00 or 40		0000	
183F hex	Axis	s 8 allocation Axis 7 allocation	00 or 40		0000	
1840 hex	Axis	s 10 allocation Axis 9 allocation	00 or 40		0000	
1841 hex	Axis	s 12 allocation Axis 11 allocation	00 or 40		0000	
1842 hex	Axis	s 14 allocation Axis 13 allocation	00 or 40		0000	
1843 hex	Axis	s 16 allocation Axis 15 allocation	00 or 40		0000	

Data Details Set the allocations for the MECHATROLINK devices connected to the PCU. Make the following settings using eight bits per axis for axes 1 to 16. 00 hex: Axis not used (no allocation). 40 hex: Axis allocated to the Servo Drive. Up to 16 axes can be allocated for the CS1W/CJ1W-NCF71, up to 2 axes can be allocated for the CS1W/CJ1W-NC271, and up to 4 axes can be allocated for the CS1W/CJ1W-NC471. An error will occur if you allocate more axes than your Unit supports. Setting Example In this example, axes 1 to 3, axis 5, and axis 8 are allocated to the Servo Drive. (The axis number corresponds to the station number for the MECHA-TROLINK device.) 183C hex: 4040 hex 183D hex: 0040 hex 183E hex: 0040 hex 183F hex: 4000 hex 1840 to 1843 hex: 0000 hex

Details

Refer to 6-2-1 Scan List.

Unit address	Name	Enable timing
1856 hex to 1857 hex	MECHATROLINK communications setting	At Powerup or restart

Data con- figuration	15	08 07	04 03	00	Data setting range	Default
1856 hex	Transfer cycle	Comn	nunications	cycle	Transfer cycle: 00 to 08, A2, A5	0000
					Communications cycle: 00 to 20 hex	
1857 hex	00 (fixed)	C2 ma connec	ster No. of munica	com- tions	C2 master connec- tion: 0, 1	0000
					No. of communica- tions retries: 0 to 7, F	

Data Details

This parameter is used to make the settings for MECHATROLINK communications. For details on settings, refer to *6-2-3 MECHATROLINK Communications Settings*.

Transfer Cycle

This parameter is used to set the cycle for sending and receiving data with MECHATROLINK communications.

00: 1 ms (default setting)

01: 1 ms	05: 5 ms
02: 2 ms	06: 6 ms
03: 3 ms	07: 7 ms
04: 4 ms	08: 8 ms
A2 hex: 0.25 ms	A5 hex: 0.5 ms

Communications Cycle

This parameter sets the cycle for refreshing data in the PCU and MECHA-TROLINK device. The communications cycle is set in the Common Parameters by setting the factor used to obtain integer multiples of the transfer cycle.

Set value: 00 to 20 hex

The default setting 00 is the same as when the cycle is set to 3. The communications cycle, however, is set as follows:

Transfer cycle \times communications cycle (multiplier) \leq 32 ms

	Number of Communications Retries
	This parameter is used to set the maximum number of stations that will per- form communications retries when sending/receiving data between the PCU and MECHATROLINK devices.
	Set value: 0 to 7, F
	The default setting 0 is the same as when the number of retries is set to 1. When F is specified, the number of retires is 0 (no retries).
	C2 Master Connection
	This parameter is used to set whether a separate communications master is connected to the PCU for MECHATROLINK system support. This parameter will be used for connecting future system support devices. Do not change the default setting of 0 (no C2 master).
	0: No C2 master (default setting) 1: C2 master connected
Setting Example	Default Settings
	1856 hex: 0000 hex 1857 hex: 0000 hex
	The MECHATROLINK communications settings are as follows:
	Transfer cycle: 1.0 ms Communications cycle: \times 3 (1.0 ms \times 3 = 3.0 ms) Number of communications retries: 1 There is no C2 master.
Details	Refer to 6-2-3 MECHATROLINK Communications Settings.

4-4 Axis Parameter Area

The Axis Parameter Area is used to make settings for axis control, such as origin input signal selection and origin search method. The axis parameters for the axes to be used must be set.

4-4-1 Axis Parameters Overview

Axis Parameters are transferred to the PCU's internal memory using the data transfer command and saved in the PCU's flash memory using the SAVE DATA command. Turn ON the power to the CPU Unit again, or restart the PCU to read the Axis Parameters saved in flash memory to the PCU's internal memory.

The Axis Parameters can be changed at any time by sending the WRITE DATA command to the PCU. The changed parameters are enabled as soon as the data has been written without any errors. The addresses in the PCU's internal memory allocated for Axis Parameters are determined by the axis number of each axis using the following equation.

Beginning word of Axis Parameter Area for Axis N: d = 1860 hex + (N–1) \times 14 hex (N = 1 to 16)

Axis No.	Beginning word d						
Axis 1	1860 hex	Axis 5	18B0 hex	Axis 9	1900 hex	Axis 13	1950 hex
Axis 2	1874 hex	Axis 6	18C4 hex	Axis 10	1914 hex	Axis 14	1964 hex
Axis 3	1888 hex	Axis 7	18D8 hex	Axis 11	1928 hex	Axis 15	1978 hex
Axis 4	189C hex	Axis 8	18EC hex	Axis 12	193C hex	Axis 16	198C hex

The following table lists the beginning word of each Axis Parameter Area.

PCU's			Co	ontents			Setting	
address	15 to 14	13	12	11 to 08	07 to 04	03 to 00		
d	Origin inț	out signal	selection		Interrupt input sig- nal selection		Select the origin input signal and interrupt input signal. 00: Phase Z (default setting) 01: External latch signal 1 input 02: External latch signal 2 input 03: External latch signal 3 input	
d+1	0 (fixed)	Origin search preset	Origin search direction	Origin detection method	Origin search operation	0 (fixed)	Set the origin search operation. Origin search direction 0: Reversal mode 1 (default setting) 1: Reverse mode 2 2: Single-direction mode 3: Reversal mode 3 (See note.) Origin detection method (See note.) 0: With origin proximity input signal reversal (default setting) 1: Without origin proximity input signal reversal 2: Not use origin proximity input signal Origin search direction 0: Forward (default setting) 1: Reverse Origin search preset (See note.) 0: Not Set (default setting) 1: Set	
d+2	0 (fixed)			0 (fixed)	0 (fixed)	Encoder type	0: Incremental encoder (default setting) 1: Absolute encoder	

Note The reversal mode 3 setting for the origin search operation, the origin detection method, and the origin search preset can be used only with Position Control Units with unit version 2.0 or later. They cannot be used with Position Control Units with unit version 1.3 or earlier.

For details on transferring data, refer to SECTION 5 Transferring and Saving Data.

4-4-2 Axis Parameter Details

The data set in the Axis Parameter Area is as follows: d = $1860 \text{ hex} + (\text{Axis No.} -1) \times 14 \text{ hex}$

Unit address	Name	Enable timing
d	Input signal selection	After comple- tion of data writ- ing

Data con- figuration	15	08 07	00	Data setting range	Default
d	Origin input signal selection	Interrupt selection	input signal 1	Origin input signal selection: 00 to 03	0000
				Interrupt input signal selection: 00 to 03	

Data Details

Origin Input Signal Selection

Use this parameter to select the origin input signal to be used with origin search.

00: Phase Z

01: External latch signal 1 input

02: External latch signal 2 input

03: External latch signal 3 input

		Interrupt Input Signal Selection Select the interrupt input signal used for interrupt feeding. 00: Phase Z 01: External latch signal 1 input 02: External latch signal 2 input 03: External latch signal 3 input							
	Note	(1) When (input s signal t location <i>MECH</i>	01, (igna to be ns. <i>ATR</i>	02, or 03 Il select e used n (Refer <i>OLINK</i> .)	8 (extern ion/inter nust be a to <i>6-4</i>)	al latch signa rupt input sig allocated in th <i>Standard S</i>	als 1 to 3) is sele gnal selection, th ne Servo Drive's <i>ettings for Serv</i>	cted ne ex exte ro D	in the origin xternal latch rnal input al- Drives Using
		(2) When 01 (ext tion an nals.	usin erna d int	g a SM, al latch s errupt ir	ARTSTE signal 1) nput sign	P Junior Se can be used al selection.	rvo Drive, only 0 d for the origin in Do not select an	0 (p put s y oth	hase Z) and signal selec- ner input sig-
Setting Example		In this exam Z as the orig for interrupt	ple, gin i feed	the orig nput, ar ling.	in searc nd exterr	h is performe nal latch sigr	ed using the Ser nal 2 as the inter	vom rupt	otor's phase input signal
		Beginnir d = 1860 189C he terrupt ir	ng w) he: ex: 0 nput	ord of A x + (4 – 002 hex signal s	xis Para 1) \times 14 x = Originselection	meter Area f hex = 189C l n input signa : External la	for axis 4: nex I selection: Phas tch signal 2 (02 ł	e Z nex)	(00 hex); In-
Details		Refer to $8-2$	Orig	gin Sear	rch Oper	<i>ation</i> , and 9-	5 Interrupt Feed	ing.	
			x + (. – I) × I	Namo		En	able timing
		d+1, d+2	33	Operation	on mode	selection		Afte tion ing	er comple- of data writ-
		Data con- figuration	15	12 11	08 07	04 03 00	Data setting rar	nge	Default
		d+1	0 (fixed)	Bit 1	1 search Origin opera 2: Origin searc 3: Origin searc	search 0 (fixed)	Origin search ope tion: 0 to 3 Origin detection method: 0 to 2 Origin search dire tion: 0, 1 Origin search pre	era- ec- eset	0000
		d+2		00 (fixed)	0 (fi	xed) Encoder type	Encoder type: 0,	1	0000
Data Details		Use there p	arar	neters	to make	the operation	on mode setting	s (o	rigin search

Use there parameters to make the operation mode settings (origin search operation mode) for each axis of the PCU. The bits specified as "0/00 (fixed)" are reserved by the system in the PCU and must always be set to 0. For details on settings, refer to *8-2-3 Data Settings Required for Origin Search*, and *8-6-3 PCU Data Settings for Using Absolute Encoders*.

Origin Search Operation

Select the origin search operation pattern. For details on operation patterns, refer to *8-2-4 Origin Search Operation*.

0:	Reversal	mode 1

- 1: Reversal mode 2
- 2: Single-direction mode
- Reversal mode 3 (unit version 2.0 or later)

Origin Detection Method (Unit Version 2.0 or Later)

Select the origin detection method. For details on operation patterns, refer to *8-2-4 Origin Search Operation*.

- 0: With origin proximity input signal reversal
- 1: Without origin proximity input signal reversal
- 2: Not use origin proximity input signal

Origin Search Direction

This parameter sets the origin search direction to match the Zero Point Return Direction setting (Pn816) in the Servo Parameters. This parameter and Servo Parameter Pn816 must be set to the same value. For details on origin search operations, refer to *8-2-4 Origin Search Operation*.

- 0: Forward direction
- 1: Reverse direction

Origin Search Preset (Unit Version 2.0 or Later)

The present position can be automatically set to a preset value when the origin search has been completed normally. Refer to *8-2-6 Origin Search Preset* for details on the origin search preset.

- 0: Present position not set to preset value at completion of origin search
- 1: Present position set to preset value at completion of origin search

Encoder Type

Use this parameter to select incremental encoder or absolute encoder as the encoder for the Servomotor. Set the encoder type according to the functions of the Servomotor and Servo Drive being used, as follows.

In this example, an incremental encoder is used with axis 5, and origin search

- 0: Incremental encoder
- 1: Absolute encoder

Setting Example

18B1 hex: 1020 hex 18B2 hex: 0000 hex

Details

• 8-2-3 Data Settings Required for Origin Search

is set to reverse direction in single-direction mode.

- 8-2-4 Origin Search Operation
- 8-2-6 Origin Search Preset.
- 8-6-3 PCU Data Settings for Using Absolute Encoders

4-5 Servo Parameter Area

4-5-1 G5-series Servo Drive (R88D-KN -ML2 with Built-in MECHATROLINK-II Communications)

This shows a list of Servo Parameters that can be used by the Position Control Unit of a G5-series Servo Drive (R88D-KN□-ML2) with built-in MECHA-TROLINK-II Communications.

For details on each parameter, refer to the user's manual of the respective G5-series Servo Drives.

	 Some parameters are enabled by turning the power supply OFF and then ON again. (Those parameters are indicated in the table.) After changing these parameters, turn OFF the power supply, confirm that the power supply indicator has gone OFF, and then turn ON the power supply again. Do not change the parameters marked "Reserved for manufacturer use" and "Reserved". Do not change the set values marked "Not used" and "Reserved".
	 For details about the data attributes, refer to the information below. A: Enabled at all time. B: Changes are prohibited during motor operation and issuing of commands.
	The timing will fluctuate if a change is made during motor operation and issuing of commands.
	C: Enabled after a power reset or executing the CONFIG command in MECHATROLINK-II communication. B: Power reset.
	Note that this is not enabled by the CONFIG command in MECHA- TROLINK-II communication.
Relay area settings	Set the servo parameter numbers in parentheses in the list for the Servo Parameter number Relay area (a+17) when transferring Servo Parameters. For example, when transferring the Error Counter Overflow Level, set "00E"
	for the Servo Parameter number Relay area (a+17).

Parameter Tables

Basic Parameters

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
000 (000)	Rotation Direction Switching	2	 Set the relation between the command direction and the motor rotation direction. 0: With a forward rotation command, the motor rotates CW as viewed from the axis end. 1: With a forward rotation command, the motor rotates CCW as viewed from the axis end. 	1	-	0 to 1	С
001 (001)	Control Mode Selec- tion	2	Select the Servo Drive CONTROL mode. 0 to 5: Switch function 6: Full closing control	0	-	0 to 6	R
002 (002)	Realtime Autotuning Mode Selection	2	 Set the OPERATION mode for realtime autotuning. Disabled Emphasizes stability Emphasizes positioning If there is an unbalanced load on the vertical axis or the like. When friction is large. If there is an unbalanced load on the vertical axis or the like and friction is too large. When the realtime autotuning is customized. 	1	-	0 to 6	В
003 (003)	Realtime Autotuning Machine Rigidity Set- ting	2	Set the machine rigidity for executing realtime autotun- ing.	11/13	-	0 to 31	В
004 (004)	Inertia Ratio	2	Set the load inertia as a percentage of the motor rotor inertia.	250	%	0 to 10000	В
009 (009)	Electronic Gear Ratio Numerator	4	Set the electronic gear ratio. If Pn009 = 0, the encoder resolution is set as the	1	-	0 to 2 ³⁰	С
010 (00A)	Electronic Gear Ratio Denominator	4	numerator. Electronic Gear Ratio Numerator (Pn009) Electronic Gear Ratio Denominator (Pn010)	1	-	1 to 2 ³⁰	С
013 (00D)	No. 1 Torque Limit	2	Set the No. 1 limit value for the output torque of the motor.	500	%	0 to 500	В
014 (00E)	Error Counter Over- flow Level	4	Set the range of the error counter overflow level. Detec- tion of error counter overflow level error is disabled if the set value is 0.	100000	Com- mand unit	0 to 2 ²⁷	A
015 (00F)	Operation Switch when Using Absolute Encoder	2	 Select the absolute encoder usage method. 0: Used as absolute encoder. 1: Used as an incremental encoder. 2: Used as absolute encoder. (Multi-rotation counter overflows are ignored.) 	1	-	0 to 2	С
016 (010)	Regeneration Resis- tor Selection	2	 Select the Regeneration Resistor used. 0: Use the Built-in Resistor. Triggering of regeneration overload protection (Alarm No.18) depends on the Built-in Resistor (with approx. 1% duty). 1: Use an External Resistor. The regeneration processing circuit operates and regeneration overload protection (Alarm No.18) is triggered when the operating rate of the Regenera- tion Resistor exceeds 10%. 2: Use an External Resistor. Regeneration overload protection (Alarm No.18) does not operate. 3: No Regeneration Resistor All regeneration power is processed with built-in capacitors. Select the type of load ratio calculation for the External 	0/3	-	0 to 3	c
(017)	External Regenera- tion Resistor Setting	2	 Select the type of load ratio calculation for the External Regeneration Resistor. 0: Regeneration load ratio is 100% when operating rate of the External Regeneration Resistor is 10%. 1 to 4: Reserved 	U	-	U TO 4	U

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
100 (100)	Position Loop Gain 1	2	Set the position loop gain 1.	320/ 480	0.1/s	0 to 30000	В
101 (101)	Speed Loop Gain 1	2	Set the speed loop gain 1.	180/ 270	0.1Hz	1 to 32767	В
102 (102)	Speed Loop Integral Time Constant 1	2	Set the speed loop integration time constant 1.	210/ 310	0.1ms	1 to 10000	В
103 (103)	Speed Feedback Fil- ter Time Constant 1	2	The speed feedback filter 1 can be set to one of 6 values.	0	-	0 to 5	В
104 (104)	Torque Command Fil- ter Time Constant 1	2	Set the time constant for the torque filter 1.	84/126	0.01m s	0 to 2500	В
105 (105)	Position Loop Gain 2	2	Set the position loop gain 2.	380/ 570	0.1/s	0 to 30000	В
106 (106)	Speed Loop Gain 2	2	Set the speed loop gain 2.	180/ 270	0.1Hz	1 to 32767	В
107 (107)	Speed Loop Integra- tion Time Constant 2	2	Set the speed loop integration time constant 2.	10000	0.1ms	1 to 10000	В
108 (108)	Speed Feedback Fil- ter Time Constant 2	2	The speed feedback filter 2 can be set to one of 6 values.	0	-	0 to 5	В
109 (109)	Torque Command Filter Time Constant 2	2	Set the time constant for the torque filter 2.	84/126	0.01m s	0 to 2500	В
110 (10A)	Speed Feed-forward Amount	2	Set the speed feed-forward amount.	300	0.1%	0 to 1000	В
111 (10B)	Speed Feed-forward Command Filter	2	Set the speed feed-forward filter time constant.	50	0.01m s	0 to 6400	В
112 (10C)	Torque Feed-forward Amount	2	Set the torque feed-forward amount.	0	0.1%	0 to 1000	В
113 (10D)	Torque Feed-forward Command Filter	2	Set the torque feed-forward filter.	0	0.01m s	0 to 6400	В
114 (10E)	Gain Switching Input Operating Mode Selection	2	Execute optimum tuning using the gain switching function.0: Gain 1 (PI/P switching enabled)1: Gain 1 and gain 2 switching available	1	-	0 to 1	В
115 (10F)	Switching Mode in Position Control	2	Select the gain switching condition for position control. It is necessary that Pn114 be set to 1. 0: Always gain 1 1: Always gain 2 2: Gain switching command input via MECHA- TROLINK-II communications 3: Torque command change amount 4: Always gain 1 5: Command speed 6: Amount of position error 7: When the position command is received. 8: Positioning completion signal (INP) OFF 9: Actual motor speed 10:Combination of position command input and rota- tion speed	0	-	0 to 10	В
116 (110)	Gain Switching Delay Time in Position Con- trol	2	Set the delay time for switching from gain 2 to gain 1.	50	0.1ms	0 to 10000	В
117 (111)	Gain Switching Level in Position Control	2	Set the gain switching level.	50	-	0 to 20000	В
118 (112)	Gain Switching Hys- teresis in Position Control	2	Set the hysteresis for gain switching.	33	-	0 to 20000	В
119 (113)	Position Gain Switch- ing Time	2	Set the position gain switching time for gain switching.	33	0.1ms	0 to 10000	В

Gain Parameter

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
120 (114)	Switching Mode in Speed Control	2	Select the gain switching condition for speed control. It is necessary that Pn114 be set to 1. 0: Always gain 1 1: Always gain 2 2: Gain switching command input via MECHA- TROLINK-II communications 3: Torque command change amount 4: Speed command change amount 5: Command speed	0	-	0 to 5	В
121 (115)	Gain Switching Delay Time in Speed Con- trol	2	Set the delay time for switching from gain 2 to gain 1.	0	0.1ms	0 to 10000	В
122 (116)	Gain Switching Level in Speed Control	2	Set the gain switching level.	0	-	0 to 20000	В
123 (117)	Gain Switching Hys- teresis in Speed Control	2	Set the hysteresis for gain switching.	0	-	0 to 20000	В
124 (118)	Switching Mode in Torque Control	2	Select the gain switching condition for torque control. It is necessary that Pn114 be set to 1. 0: Always gain 1 1: Always gain 2 2: Gain switching command input via MECHA- TROLINK-II communications 3: Torque command change amount	0	-	0 to 3	В
125 (119)	Gain Switching Delay Time in Torque Con- trol	2	Set the delay time for switching from gain 2 to gain 1.	0	0.1ms	0 to 10000	В
126 (11A)	Gain Switching Level in Torque Control	2	Set the gain switching level.	0	-	0 to 20000	В
127 (11B)	Gain Switching Hys- teresis in Torque Control	2	Set the hysteresis for gain switching.	0	-	0 to 20000	В

Damping Suppression Parameters

Param- eter	Parameter name	Param- eter	Explanation	Default setting	Unit	Setting range	Data attribute
NO. 200	Adaptive Filter Selec-	2 SIZE	Set the operation of the adaptive filter.	0	-	0 to 4	В
(200)	tion		 Disabled One enabled.Frequency limited after adaptation. Two enabled.Frequency limited after adaptation. One enabled.Adaptation performed at all times. Two enabled.Adaptation performed with 1 filter at all times. 				
201 (201)	Notch 1 Frequency Setting	2	Set the notch frequency of resonance suppression notch filter 1.	5000	Hz	50 to 5000	В
202 (202)	Notch 1 Width Set- ting	2	Set the notch width of the resonance suppression notch filter 1.	2	-	0 to 20	В
203 (203)	Notch 1 Depth Set- ting	2	Set the notch depth of resonance suppression notch fil- ter 1.	0	-	0 to 99	В
204 (204)	Notch 2 Frequency Setting	2	Set the notch frequency of resonance suppression notch filter 2.	5000	Hz	50 to 5000	В
205 (205)	Notch 2 Width Set- ting	2	Set the notch width of the resonance suppression notch filter 2.	2	-	0 to 20	В
206 (206)	Notch 2 Depth Set- ting	2	Set the notch depth of resonance suppression notch fil- ter 2.	0	-	0 to 99	В
207 (207)	Notch 3 Frequency Setting	2	Set the notch frequency of resonance suppression notch filter 3. This is set automatically when an adaptive notch is enabled.	5000	Hz	50 to 5000	В
208 (208)	Notch 3 Width Set- ting	2	Set the notch width of the resonance suppression notch filter 3. This is set automatically when an adaptive notch is enabled.	2	-	0 to 20	В
209 (209)	Notch 3 Depth Set- ting	2	Set the notch depth of resonance suppression notch fil- ter 3. This is set automatically when an adaptive notch is enabled.	0	-	0 to 99	В

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
210 (20A)	Notch 4 Frequency Setting	2	Set the notch frequency of resonance suppression notch filter 4. This is set automatically when an adaptive notch is pachled.	5000	Hz	50 to 5000	В
211 (20B)	Notch 4 Width Set- ting	2	Set the notch width of the resonance suppression notch filter 4. This is set automatically when an adaptive notch is enabled.	2	-	0 to 20	В
212 (20C)	Notch 4 Depth Set- ting	2	Set the notch depth of resonance suppression notch fil- ter 4. This is set automatically when an adaptive notch is enabled.	0	-	0 to 99	В
213 (20D)	Damping Filter Selection	2	 Select the damping filter switching method. Damping filter 1 or 2 enabled Reserved for manufacturer use Reserved for manufacturer use Switch by the position command direction Forward direction: Damping filter 1 or 3 is enabled. Reverse direction: Damping filter 2 or 4 is enabled. 	0	-	0 to 3	В
214 (20E)	Damping Frequency 1	2	Set the damping frequency 1. The function is enabled if the set value is 10 (= 1 Hz) or greater.	0	0.1Hz	0 to 2000	В
215 (20F)	Damping Filter 1 Set- ting	2	Finely adjust damping control function 1. If torque saturation occurs, lower this setting; to increase responsiveness, raise this setting.	0	0.1Hz	0 to 1000	В
216 (210)	Damping Frequency 2	2	Set the damping frequency 2. The function is enabled if the set value is 10 (= 1 Hz) or greater.	0	0.1Hz	0 to 2000	В
217 (211)	Damping Filter 2 Set- ting	2	Finely adjust damping control function 2. If torque saturation occurs, lower this setting; to increase responsiveness, raise this setting.	0	0.1Hz	0 to 1000	В
218 (212)	Damping Frequency 3	2	Set the damping frequency 3. The function is enabled if the set value is 10 (= 1 Hz) or greater.	0	0.1Hz	0 to 2000	В
219 (213)	Damping Filter 3 Set- ting	2	Finely adjust damping control function 3. If torque saturation occurs, lower this setting; to increase responsiveness, raise this setting.	0	0.1Hz	0 to 1000	В
220 (214)	Damping Frequency 4	2	Set the damping frequency 4. The function is enabled if the set value is 10 (= 1 Hz) or greater.	0	0.1Hz	0 to 2000	В
221 (215)	Damping Filter 4 Set- ting	2	Finely adjust damping control function 4. If torque saturation occurs, lower this setting; to increase responsiveness, raise this setting.	0	0.1Hz	0 to 1000	В
222 (216)	Position Command Filter Time Constant	2	Set the time constant of the first-order lag filter for the position command.	0	0.1Hz	0 to 10000	В

Analog Control Parameters

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
312 (30C)	Soft Start Accelera- tion Time	2	Set the acceleration processing acceleration time for speed commands.	0	ms/ motor Maxi- mum speed	0 to 10000	В
313 (30D)	Soft Start Decelera- tion Time	2	Set the deceleration processing acceleration time for speed commands.	0	ms/ motor Maxi- mum speed	0 to 10000	В
314 (30E)	S-curve Accelera- tion/Deceleration Time Setting	2	Set the acceleration/deceleration processing S-curve time for speed commands.	0	ms	0 to 1000	В
317 (311)	Speed Limit Selec- tion	2	 Select the torque command and speed limit value. 0: Limit the speed by the limit set on the Speed Limit Value Setting (Pn321). 1: Limit the speed by the speed limit value (VLIM) via MECHATROLINK-II communications or by the value set by the Speed Limit Value Setting (Pn321). 	0	-	0 to 1	В
321 (315)	Speed Limit Value Setting	2	Set the speed limit value.	50	r/min	0 to 20000	A

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
323 (317)	External Feedback Pulse Type Selection	2	 Select the external feedback pulse type. 0: 90° phase difference output type 1: Serial communications type (incremental encoder specifications) 2: Serial communications type (absolute encoder specifications) 	0	-	0 to 2	R
324 (318)	External Feedback Pulse Dividing Numerator	4	Set the external feedback pulse dividing numerator.	0	-	0 to 2 ²⁰	R
325 (319)	External Feedback Pulse Dividing Denominator	4	Set the external feedback pulse dividing denominator.	10000	-	1 to 2 ²⁰	R
326 (31A)	External Feedback Pulse Direction Switching	2	Reverse the direction to count the external encoder feed back. 0: Count direction not reversed 1: Count direction reversed	0	-	0 to 1	R
327 (31B)	External Feedback Pulse Phase-Z Set- ting	2	Set to enable or disable the Phase-Z disconnection detection when an external encoder of 90° phase differ- ence output type is used. 0: Phase-Z disconnection detection enabled 1: Phase-Z disconnection detection disabled	0	-	0 to 1	R
328 (31C)	Internal/External Feedback Pulse Error Counter Over- flow Level	4	Set the threshold for feedback pulse deviation errors.	16000	Com- mand unit	1 to 2 ²⁷	С
329 (31D)	Internal/External Feedback Pulse Error Counter Reset	2	Clear to 0 the feedback pulse error value for each set rotation speed.	0	Rota- tion	0 to 100	С

Interface Monitor Setting Parameters

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
400 (400)	Input Signal Selec- tion 1	4	Set the function and logic for the general-purpose input 1 (IN1).	00949494h	-	0 to 00FFFFFFh	С
401 (401)	Input Signal Selec- tion 2	4	Set the function and logic for the general-purpose input 2 (IN2).	00818181h	-	0 to 00FFFFFFh	С
402 (402)	Input Signal Selec- tion 3	4	Set the function and logic for the general-purpose input 3 (IN3).	00828282h	-	0 to 00FFFFFFh	С
403 (403)	Input Signal Selec- tion 4	4	Set the function and logic for the general-purpose input 4 (IN4).	00222222h	-	0 to 00FFFFFFh	С
404 (404)	Input Signal Selec- tion 5	4	Set the function and logic for the general-purpose input 5 (IN5).	002B2B2Bh	-	0 to 00FFFFFFh	С
405 (405)	Input Signal Selec- tion 6	4	Set the function and logic for the general-purpose input 6 (IN6).	00212121h	-	0 to 00FFFFFFh	С
406 (406)	Input Signal Selec- tion 7	4	Set the function and logic for the general-purpose input 7 (IN7).	00202020h	-	0 to 00FFFFFFh	С
407 (407)	Input Signal Selec- tion 8	4	Set the function and logic for the general-purpose input 8 (IN8).	002E2E2Eh	-	0 to 00FFFFFFh	С
410 (40A)	Output Signal Selec- tion 1	4	Set the function assignment for the general-purpose output 1 (OUTM1)	00030303h	-	0 to 00FFFFFFh	С
411 (40B)	Output Signal Selec- tion 2	4	Set the function assignment for the general-purpose output 2 (OUTM2)	00020202h	-	0 to 00FFFFFFh	С

Section 4-5

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
416 (410)	Analog Monitor 1 Selection	2	Select the type for analog monitor 1. 0: Motor speed 1: Position command speed 2: Internal position command speed 3: Speed control command 4: Torque command 5: Command position error 6: Encoder position error 7: Full close error 8: Hybrid error 9: P-N voltage 10:Regeneration load ratio 11: Motor load ratio 12:Forward direction torque limit 13:Reverse direction torque limit 14:Speed limit value 15:Inertia ratio 16 to 18: Reserved 19:Encoder temperature 20:Servo Drive temperature 21:Encoder 1-rotation data	0	-	0 to 21	A
417 (411)	Analog Monitor 1 Scale Setting	4	Set the output gain for analog monitor 1.	0	-	0 to 214748364	A
418 (412)	Analog Monitor 2 Selection	2	Select the type for analog monitor 2. The set values for this parameter are the same as Analog Monitor 1 Type (Pn416).	4	-	0 to 21	A
419 (413)	Analog Monitor 2 Scale Setting	4	Select the output gain for analog monitor 2.	0	-	0 to 214748364	A
421 (415)	Analog Monitor Out- put Selection	2	Select the analog monitor output voltage method. 0: Output range is -10 to 10 V 1: Output range is 0 to 10 V 2: Output range is 0 to 10 V (5 V as the center)	0	-	0 to 2	A
431 (41F)	Positioning Comple- tion Range 1	4	Set the allowed number of pulses for the positioning completion range.	300	Com- mand unit	0 to 262144	A
432 (420)	Positioning Comple- tion Condition Selec- tion	2	 Set the judgment conditions for positioning completion output. 0: The Positioning completion output 1 becomes on when the positional error is lower than the value set on the Pn431. 1: The Positioning completion output 1 becomes on when there is no position command, and the positional error is lower than the value set on the Pn431. 2: The Positioning completion output 1 becomes on when there is no position command, the zerospeed detection signal is on, and the positional error is lower than the value set on the Pn431. 3. The Positioning completion output 1 becomes on when there is no position command, the zerospeed detection signal is on, and the positional error is lower than the value set on the Pn431. 3. The Positioning completion output 1 becomes on when there is no position command, and the positional error is lower than the value set on the Pn431. 3. The Positioning completion output 1 becomes on when there is no position command, and the positional error is lower than the value set on the pn431. The ON-state is retained until the Positioning Completion Hold Time (Pn433) elapses. After that, it is turned off or kept to be on, depending on the positional error then. 	0	-	0 to 3	A
433 (421)	Positioning Comple- tion Hold Time	2	Set the positioning completion hold time.	0	1ms	0 to 30000	A
434 (422)	Zero Speed Detec- tion	2	Set the output timing of the Zero speed detection out- put (ZSP) in rotation speed [r/min].	50	r/min	10 to 20000	A
435 (423)	Speed Conformity Detection Range	2	Set the detection range for the speed conformity out- put (VCMP). Set the difference between the speed command and the actual speed.	50	r/min	10 to 20000	A
436 (424)	Rotation Speed for Motor Rotation Detection	2	Set the number of motor rotation for the Motor rotation detection output signal (TGON).	1000	r/min	10 to 20000	A
437 (425)	Brake Timing when Stopped	2	Set the operation time for the mechanical brake at stop.	0	1ms	0 to 10000	В
438 (426)	Brake Timing during Operation	2	Set the operation time for the mechanical brake during operation.	0	1ms	0 to 10000	В
439 (427)	Brake Release Speed Setting	2	Set the number of motor rotation to determine a mechanical brake output during rotation.	30	r/min	30 to 3000	В

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
440 (428)	Warning Output Selection 1	2	Select the warning type for warning output 1. 0: Output by all types of warnings. 1: Overload warning 2: Excessive regeneration warning 3: Battery warning 4: Fan warning 5: Encoder communications warning 6: Encoder overheating warning 7: Vibration warning 8: Life expectancy warning 9: External encoder error warning 10:External encoder communications error warning 11:Data setting warning 12:Command warning 13:MECHATROLINK-II communications warning	0	-	0 to 13	A
441 (429)	Warning Output Selection 2	2	Select the warning type for warning output 2. The relationships among the set values for this param- eter are the same as for Warning Output Selection 1 (Pn440).	0	-	0 to 13	A
442 (42A)	Positioning Comple- tion Range 2	4	Set the allowable number of pulses for the second positioning completion range.	10	Com- mand unit	0 to 262144	A

Extended Parameters

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
504 (504)	Drive Prohibition Input Selection	2	 Set the operation to be performed upon forward/reverse direction drive prohibition input. 0: Enable the Forward and Reverse drive prohibition inputs. 1: Disable the Forward and Reverse drive prohibition inputs. 2: Enable the Forward and Reverse drive prohibition inputs. 	1	-	0 to 2	С
505 (505)	Stop Selection for Drive Prohibition Input	2	 Set the drive conditions during deceleration and after stopping, when the Forward or Reverse Drive Prohibition Inputs are enabled. 0: The torque in the drive prohibit direction is disabled, and the dynamic brake is activated. 1: The torque in the drive prohibit direction is disabled, and free-run deceleration is performed. 2: The torque in the drive prohibit direction is disabled, and an emergency stop is performed. 	0	-	0 to 2	С
506 (506)	Stop Selection with Servo OFF	2	 Set the stop operation when the servo is turned OFF. 0, 4:During deceleration: Dynamic brake After stopping: Dynamic brake Error counter: Clear 1, 5:During deceleration: Free-run After stopping: Dynamic brake Error counter: Clear 2, 6:During deceleration: Dynamic brake After stopping: Servo free Error counter: Clear 3, 7:During deceleration: Free-run After stopping: Servo free Error counter: Clear 8: During deceleration: Emergency stop After stopping: Dynamic brake Error counter: Clear 9: During deceleration: Emergency stop After stopping: Servo free Error counter: Clear 9: During deceleration: Emergency stop After stopping: Servo free Error counter: Clear 	0	-	0 to 9	В
507 (507)	Stop Selection with Main Power Supply OFF	2	Set the stop operation when the main power supply is turned OFF. The settable values are the same as those on the Pn506.	0	-	0 to 9	В
508 (508)	Undervoltage Alarm Selection	2	 Select either to let the servo off or to stop the alarm when a main power alarm occurs. Bring the Servo-OFF state in accordance with the Pn507 setting. Return to Servo on state by turning on the main power. Main power supply undervoltage (Alarm No.13.1) occurs. Stop the alarm. 	1	-	0 to 1	В

Section 4-5

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
509 (509)	Momentary Hold Time	2	Set the main power supply alarm detection time.	70	1ms	70 to 2000	С
510 (50A)	Stop Selection for Alarm Detection	2	 Select the stopping method at an alarm. 0: During deceleration: Dynamic brake After stopping: Dynamic brake 1: During deceleration: Free-run After stopping: Dynamic brake 2: During deceleration: Dynamic brake 2: During deceleration: Free-run After stopping: Servo free 3: During deceleration due to emergency stop alarm: Emergency stop During deceleration: Dynamic brake 4. During deceleration: Dynamic brake 5: During deceleration: Dynamic brake 5: During Emergency stop alarm deceleration: Emer- gency stop During deceleration: Free-run After stopping: Dynamic brake 6: During Emergency stop alarm deceleration: Emer- gency stop During deceleration: Dynamic brake 6: During Emergency stop alarm deceleration: Emer- gency stop During deceleration: Dynamic brake 7: During Emergency stop alarm deceleration: Emer- gency stop During deceleration: Dynamic brake 7: During Emergency stop alarm deceleration: Emer- gency stop During deceleration: Dynamic brake 7: During Emergency stop alarm deceleration: Emer- gency stop During deceleration: Servo free 7: During Emergency stop alarm deceleration: Emer- gency stop During deceleration: Free-run After stopping: Servo free 	0	-	0 to 7	В
511 (50B)	Emergency Stop Torque	2	Set the torque limit for emergency stops.	0	%	0 to 500	В
512 (50C)	Overload Detection Level Setting	2	Set the overload detection level.	0	%	0 to 500	А
513 (50D)	Overspeed Detec- tion Level Setting	2	Set the overspeed error detection level.	0	r/min	0 to 20000	А
514 (50E)	Overrun Limit Setting	2	Set the motor over-travel distance for position com- mands.	10	0.1 rota- tion	0 to 1000	A
515 (50F)	Control Input Signal Read Setting	2	Select the cycle to read the control input signals from the four levels. 0: 0.166 ms 1: 0.333 ms 2: 1 ms 3: 1.666 ms	0	-	0 to 3	С
520 (514)	Position Setting Unit Selection	2	Select the setting units of Positioning Completion Range 1 and 2 (Pn431 and Pn442) and of the Error Counter Overflow Level (Pn014). 0: Command unit 1: Encoder unit	0	-	0 to 1	С
521 (515)	Torque Limit Selec- tion	2	Select the method to set the forward and reverse torque limits, and the torque feed forward function during speed control.	1	-	0 to 6	В
522 (516)	No. 2 Torque Limit	2	Set the No. 2 limit value for the motor output torque.	0	%	0 to 500	В
525 (519)	Forward External Torque Limit	2	Set the forward external toque limit when the torque limit switch input is given.	0	%	0 to 500	В
526 (51A)	Reverse External Torque Limit	2	Set the reverse external toque limit when the torque limit switch input is given.	0	%	0 to 500	В
531 (51F)	Axis Number	2	Set the axis number for communication.	1	-	0 to 127	С

Special Parameters 1

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
605 (605)	Gain 3 Effective Time	2	Set effective time of gain 3 of three-step gain switch- ing.	0	0.1ms	0 to 10000	В
606 (606)	Gain 3 Ratio Setting	2	Set gain 3 as a multiple of gain 1.	100	%	50 to 1000	В
607 (607)	Torque Command Value Offset	2	Set offset torque to add to torque command.	0	%	-100 to 100	В
608 (608)	Forward Direction Torque Offset	2	Set the value to add to a torque command for forward direction operation.	0	%	-100 to 100	В

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
609 (609)	Reverse Direction Torque Offset	2	Set the value to add to a torque command for reverse direction operation.	0	%	-100 to 100	В
610 (60A)	Function Expansion Setting	2	Set the function expansion. The setting contents vary depending on the function.	0	-	0 to 63	-
611 (60B)	Electric Current Response Setting	2	Make fine adjustment on electric current response with default setting as 100%.	100	%	50 to 100	В
614 (60E)	Alarm Detection Allowable Time Set- ting	2	Set the allowable time until stopping when an emer- gency stop is actuated upon alarm detection.	200	ms	0 to 1000	В
615 (60F)	Overspeed Detec- tion Level Setting at Emergency Stop	2	During an emergency stop upon alarm detection, if the motor speed excess this set value, this is an over-speed 2 error.	0	r/min	0 to 20000	A
618 (612)	Power Supply ON Initialization Time	2	Set initialization time after power supply ON to the standard 1.5 s plus some.	0	0.1s	0 to 100	R
623 (617)	Disturbance Torque Compensation Gain	2	Set the compensation gain for disturbance torque.	0	%	-100 to 100	В
624 (618)	Disturbance Observer Filter Set- ting	2	Set the filter time constant for disturbance torque com- pensation.	53	0.01ms	10 to 2500	В
631 (61F)	Realtime Autotuning Estimated Speed Selection	2	 Set the load characteristics estimated speed when realtime autotuning is enabled. 0: Fixes estimated results at the time load estimation becomes stable. 1: Estimates in every minute from the load characteristic changes. 2: Estimates in every second from the load characteristic changes. 3: Estimates the optimum from the load characteristic changes. 	0	-	0 to 3	В
632 (620)	Realtime Autotuing Customization Mode Setting	2	Set the CUSTOMIZATION mode detail for realtime autotuning.	0	-	-32768 to 32767	В
634 (622)	Hybrid Vibration Suppression Gain	2	Set the hybrid vibration suppression gain during full closing control.	0	0.1s	0 to 30000	В
635 (623)	Hybrid Vibration Suppression Filter	2	Set the hybrid vibration suppression filter time con- stant during full closing control.	10	0.01ms	0 to 6400	В
637 (625)	Vibration Detection Threshold	2	Set the vibration detection threshold. If torque vibration that exceeds this setting is detected, the vibration detection warning occurs.	0	0.1%	0 to 1000	В
638 (626)	Warning Mask Set- ting	2	Set the warning detection mask setting.If you set the corresponding bit to 1, the corresponding warning detection is disabled.	4	-	-32768 to 32767	С

Special Parameters 2

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
700 (700)	Default Display	2	 Select a data type to display on the 7-segment LED indicator on the front panel. O: Normal state 1: Mechanical angle 2: Electric angle 3: Cumulative count of MECHATROLINK-II communications errors 4 Rotary switch setting value 5: Cumulative count of encoder communications errors 6: Cumulative count of external encoder communications errors 7: Z-phase counter 8 or over: Unused 	0	-	0 to 32767	A
701 (701)	Power-ON Address Display Duration Set- ting	2	Set the time to indicate the node address when the control power is turned on.	0	100ms	0 to 1000	R
703 (703)	Torque Limit Flag Output Setting	2	 Set the condition for torque limit output during torque control. On by the torque limit value including the torque command value. On by the torque limit value excluding the torque command value. 	0	-	0 to 1	A

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
704 (704)	Backlash Compen- sation Selection	2	 Select to enable or disable the backlash compensation during position control. Set the compensation direction when the compensation is enabled. 0: Disable the backlash compensation. 1: Compensate the backlash at the first forward operation after a Servo ON. 2: Compensate the backlash at the first reverse operation after a Servo ON. 	0	-	0 to 2	С
705 (705)	Backlash Compen- sation Amount	2	Set the backlash compensation amount during posi- tion control.	0	Com- mand unit	-32768 to 32767	В
706 (706)	Backlash Compen- sation Time Constant	2	Set the time constant to apply a backlash compensa- tion during position control.	0	0.01ms	0 to 6400	В
710 (70A)	MECHATROLINK-II Communication I/O Monitor Setting	2	Select whether to reflect the inputs to the I/O monitor of MECHATROLINK-II communications, when either the forward or reverse drive prohibition input is assigned to the input signal and the Drive Prohibition Input Selection (Pn504) is set to 1 (i.e. Disable). 0: Disable the one on the I/O monitor of MECHA- TROLINK-II communications as well. 1: Enable the one on the I/O monitor of MECHA- TROLINK-II communications.	0	-	0 to 1	A

Special Parameters 3

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
800 (800)	Communications Control	2	Controls the alarms and warnings over the MECHA- TROLINK-II communications.	0	-	-32768 to 32767	С
801 (801)	Soft Limit	2	 Select whether to enable or disable the Soft Limit Function. 0: Enable the soft limits on both directions. 1: Disable the forward soft limit, but enable the reverse soft limit. 2: Enable the forward soft limit, but disable the reverse soft limit. 3: Disable the soft limits on both directions. 	0	-	0 to 3	A
803 (803)	Origin Range	2	Set the threshold for detecting the origin in absolute values.	0	-	0 to 250	A
804 (804)	Forward Software Limit Value	4	Set the forward soft limit.	500000	Com- mand unit	-1073741823 to 1073741823	A
806 (806)	Reverse Software Limit	4	Set the reverse soft limit.	-500000	Com- mand unit	-1073741823 to 1073741823	A
808 (808)	Absolute Encoder Origin Offset	4	Set the offset volume between the encoder or external encoder position and the mechanical coordinate posi- tion, when an absolute encoder or an absolute exter- nal encoder is used.	0	Com- mand unit	-1073741823 to 1073741823	С
811 (80B)	Linear Acceleration Constant	2	Set the acceleration for positioning.	100	*1	-32768 to 32767	В
814 (80E)	Linear Deceleration Constant	2	Set the deceleration for positioning.	100	*1	-32768 to 32767	В
818 (812)	Position Command FIR Filter Time Con- stant	2	Set the time constant of FIR filter for the position command.	0	0.1ms	0 to 10000	В
820 (814)	Final Distance for External Input Posi- tioning	4	Sets the distance to travel after the latch signal input position is detected during the external input position- ing.	100	0.1ms	-1073741823 to 1073741823	В
822 (816)	Origin Return mode Setting	2	Set the direction for origin return. 0: Positive direction 1: Negative direction	0	-	0 to 1	В
823 (817)	Origin Return Approach Speed 1	2	Set the operating speed for origin returns, from when the origin proximity signal turns on to when it turns off and the latch signal is detected.	50	*2	1 to 32767	В
824 (818)	Origin Return Approach Speed 2	2	Set the operating speed for origin returns, from when the latch signal is detected to when the motor stops at the position after travelling the distance set by Final Distance for Origin Return (Pn825).	5	*2	1 to 32767	В
825 (819)	Final Distance for Origin Return	4	Set the distance from the position where the latch sig- nal is entered to the origin during origin returns.	100	Com- mand unit	-1073741823 to 1073741823	В

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
836 (824)	Option Monitor Selection 1	2	The Monitor Selection Field of MECHATROLINK-II communications displays the monitoring data that is set on this parameter.	0	-	-32768 to 32767	A
837 (825)	Option Monitor Selection 2	2	The Monitor Selection Field of MECHATROLINK-II communications displays the monitoring data that is set on this parameter.	0	-	-32768 to 32767	A

*1. 10,000 command unit/s

*2. 100 command unit/s

4-5-2 G-series Servo Drive (R88D-GN□-ML2 with Built-in MECHATROLINK-II Communications)

The Servo Parameters listed here can be used when the Position Control Unit is used with a G-series Servo Drive (R88D-GN□-ML2) with built-in MECHA-TROLINK-II communications.

For further details on each of the parameters, refer to the user's manual for G-series Servo Drives.

The timing for Servo Parameters to be enabled are classified into the following two types.

Online (online parameters): Changed settings are enabled immediately after Servo Parameters have been written.

Offline (offline parameters): Changed settings are enabled when the Servo Drive power is cycled or DEVICE SETUP is executed.

The *Details* column indicates whether the following conditions apply to the corresponding parameter.

- Online parameters that can be changed when the axis is stopped (Busy Flag = 0) only.
- Parameters with standard set values when using MECHATROLINK are indicated in the *Details* column as "standard" with the set value. The standard settings are basic settings required when using the PCU. This manual describes operations assuming standard settings are being used.

For details on fixed and standard settings, refer to 6-4 Standard Settings for Servo Drives Using MECHATROLINK.

- **Note** Parameters marked with "(RT)" are automatically set during realtime autotuning. To set these parameters manually, disable realtime autotuning by setting the Realtime Autotuning Mode Selection (Pn021) to 0 before changing the parameter.
- **Note** Make sure that the equipment will not be adversely affected before changing the Servo Parameters (WRITE SERVO PARAMETER, SAVE SERVO PARAMETER). Refer to the *Servo Drive's operation manual* and always check the effect of changing the settings before changing the Servo Parameters.
Parameter Tables

Param- eter No.	Parameter name	Param- eter size	Setting	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn000	Reserved		Do not ch	ange.	1				
Pn001	Default Dis- play	2	Selects th the front p	e data to be displayed on the 7-segment LED display on panel.	0		0 to 4	Online	
			0	Normal status ("" Servo OFF, "00" Servo ON)					
			1	Indicates the machine angle from 0 to FF hex.					
				increases when the Servomotor turns forward.					
				The count continues from "0" after exceeding "FF".					
				"nF" (not Fixed) until detecting the zero position on the encoder after the control power is turned ON.					
			2	Indicates the electrical angle from 0 to FF hex.					
				phase reaches the position peak. The angle increases					
				The count continues from "0" after exceeding "FF".					
			3	Indicates the number (total) of MECHATROLINKII com-					
				The communications error count (total) saturates at the					
				maximum of FFFFh. "h" appears only for the lowest byte. The count continues from "00" after exceeding "FF".					
				Note The communications error count (total) is cleared by turning OFF the control power.					
			4	Indicates the setting on the rotary switch (node address value) loaded at startup, in decimal.					
				This value does not change even if the rotary switch is					
			5 to	turned after startup. Beserved (Do not set)					
			32767						
Pn002	Reserved		Do not ch	ange.	0				A
Pn003	Torque Limit Selec-	2	Selects the tion during	e torque limit function, or the torque feed-forward func- g speed control.	1			Online	Settings can be
	tion		 Torque I For torque speed c 	Limit Selection ue control, always select Pn05E. For position control and ontrol, select the torque limit as follows.	1		1 to 5		changed when the axes are stopped
			1	Use Pn05E as the limit value for forward and reverse operations.					(Busy Flag for
			2	Forward: Use Pn05E. Reverse: Use Pn05F.					= 0).
			3	Switch limits by torque limit values and input signals					
				Limit in forward direction:					
				PCL is OFF = Pn05E, PCL is ON = Pn05F					
				NCL is OFF = Pn05E, NCL is ON = Pn05F					
			4	Forward: Use Pn05E as limit.					
				Only in speed control, torque limits can be switched by					
				torque limit values from the network as follows:					
				Use Pn05E command or option command value 1,					
				Limit in reverse direction:					
				Use Pn05F command or option command value 2, whichever is smaller.					
			5	Forward: Use Pn05E as limit. Reverse: Use Pn05F as limit.					
		Only in speed control, torque limits can be switched b torque limit values and input signals from the network as follows:							
	Limit in forward direction:								
				PCL is OFF = Pn05E, PCL is ON = Pn05E command or option command value 1, whichever is smaller.					
				Limit in reverse direction:					
				NGL IS OFF = Pn05F, NCL IS ON = Pn05F command or option command value 2, whichever is smaller.					

Param- eter No.	Parameter name	Param- eter size	Setting		Explanation		Defa sett	ault ing	Unit	Setting range	Enable setting	Details
Pn003	Torque Limit Selec- tion	2	Note P P • Torque P 1 to 3	CL ON: When ei pin 7) or Option F CL OFF: When be pin 7) ar tions Op Feed-forward Func Enabled only dur	ther Forward Torc MECHATROLINI ield (P-CL) is ON oth Forward Torqu d MECHATROLII tion Field (P-CL) tion Selection ng speed control	ue Limit (CN1 PCL KII Communication: L. Je Limit (CN1 PCL: NK-II Communica- are OFF. . Disabled if not	.: 1 s			1 to 5	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
			4 to 5	Always disabled	rol.							
Pn004	Drive Pro- hibit Input	2	Sets the f Inputs (CI	unction for the For N1 POT: pin 19, N0	ward and Reverse DT: pin 20)	e Drive Prohibit	0			0 to 2	Offline	
	Selection		0	Decelerates and in the Stop Select when both POT a When both POT a Prohibit Input Err	stops according to ion for Drive Proh nd NOT inputs ar and NOT inputs a or (alarm code 38	o the sequence set ibition Input (Pn066 re enabled. re OPEN, the Drive 3) will occur.	5) ;					
			1	Both POT and NO When either POT	OT inputs disabled or NOT input bee	d. comes OPEN. the						
Pn005	Communi- cations Control	2	Controls e tions. Note U at	Drive Prohibit Inp errors and warning Ise with thi]s param tely if using a value	ut Error (alarm co s for MECHATRO eter set to 0. Prog other than 0.	ode 38) will occur. LINK-II communica gram to stop immed	a- 0 di-			0 to 3955	Offline	Always set to 0.
			Set the Cd COM_ER 83) will oc every ME tively for t Detection purposes.	onsecutive Commu R (bit 8 to 11). The cur when a comm CHATROLINK-II cc he number of the 0 Count. The error a	inications Error D communications unications error, v ommunications cy Consecutive Comi and warning can b	etection Count in a error (alarm code which is assessed a role, occurs consect munications Error be masked for debug	ut u- g					
			bits 15-	12 bits 11-8	bits 7-4	bits 3-0						
				COM_ERR	MSK COM WARNG	MSK COM ALM						
			• [bits 8-1 Error De	1] COM_ERR (Construction Count) Setting range: 0 to Consecutive Comm Count = COM_ER	nsecutive Commu 15. nunications Error R + 2							
			Note T no • [bits 0-3 (MSK C	hese bits are debu ot debugging.) MECHATROLINK OM ALM) (bit0)0: Communic: 1: Communic: (bit1)0: Watchdog of 1: Watchdog of 7) MECHATROLINI	l K							
			(MSK C	OM WARNG) [bit4]0: Data setting 1: Data setting [bit5]0: Command 1: Command [bit6]0: ML-II comm (warning cc 1: ML-II comm (warning cc	y warning (warning warning (warning warning (warning warning (warning nunications warning de 96h) enabled nunications warnin de 96h) disabled	d d I						
Pn006	Power ON Address Display Duration Setting	2	Sets the c power is t Note T al	duration to display urned ON. he node address d larms or warnings	when the control	30		ms	0 to 1000	Offline		
	5		0 to 6 7 to	600 ms set value \times 100 m	IS		_					
			1000									

Param- eter No.	Parameter name	Param- eter size	Setting	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn007	Speed monitor (SP) Selec- tion	2	Selects th panel). Note T a to a	he output to the Analog Speed Monitor (SP on the front his monitor output has a delay due to filtering. The Oper- ting Direction Setting (Pn043) does not affect this moni- or output. Thus, forward rotation is always positive (+), nd reverse rotation is always negative (-).	3		0 to 11	Online	
			0	Actual Servomotor speed: 47 r/min/6 V					
			1	Actual Servomotor speed: 188 r/min/6 V					
			2	Actual Servomotor speed: 750 r/min/6 V					
			3	Actual Servomotor speed: 3000 r/min/6 V					
			4	Actual Servomotor speed: 12000 r/min/6 V					
			5	Command speed: 47 r/min/6 V					
			6	Command speed: 188 r/min/6 V					
			7	Command speed: 750 r/min/6 V					
			8	Command speed: 3000 r/min/6 V					
			9	Command speed: 12000 r/min/6 V					
			10	Outputs the Issuance Completion Status (DEN).					
			-	0 V: Issuing 5 V: Issuance complete					
			11	Outputs the Gain Selection Status. 0 V: Gain 2 5 V: Gain 1					
Pn008	Torque	2	Selects th	e output to the Analog Torque Monitor (IM on the	0		0 to 14	Online	
	Monitor (IM) Selection		front pane Note T a to a	his monitor output has a delay due to filtering. The Oper- ting Direction Setting (Pn043) does not affect this moni- or output. Thus, forward rotation is always positive (+), nd reverse rotation is always negative (–).					
			0	Torque command: 100%/3 V					
			1	Position deviation: 31 pulses/3 V					
			2	Position deviation: 125 pulses/3 V					
			3	Position deviation: 500 pulses/3 V					
			4	Position deviation: 2000 pulses/3 V					
			5	Position deviation: 8000 pulses/3 V					
			6 to 10	Reserved					
			11	Torque command: 200%/3 V					
			12	Torque command: 400%/3 V					
			13	Outputs the Issuance Completion Status (DEN). 0 V: Issuing 5 V: Issuance complete					
			14	Outputs the Gain Selection Status. 0 V: Gain 2 5 V: Gain 1					
Pn009	Reserved		Do not ch	ange.	0				
Pn00A	Prohibit	2	Allows/pro	phibits parameter changes via the network.	0		0, 1	Online	
	Parameter Changes via Network		0	Allows parameter changes from the host controller via the network.					
			1	Prohibits parameter changes from the host controller via the network. Attempting to change a parameter via the network when prohibited triggers the Command Warning (warn- ing code 95h).					
Pn00B	Operation Switch When	2	Selects he This para encoder.	ow the an absolute encoder is used. meter is disabled when using an incremental	0		0 to 2	Online	
	Absolute		0	Use as an absolute encoder.					
	Encoder		1	Use an absolute encoder as an incremental encoder.					
			2	Use as an absolute encoder but ignore absolute multi- turn counter overflow alarm (alarm code 41).					

Param- eter No.	Parameter name	Param- eter size	Setting	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn00C	RS-232	2	Sets the b	baud rate for RS-232 communications.	2		0 to 5	Online	
	Baud Rate Setting		0	2,400 bps					
	g		1	4,800 bps					
			2	9,600 bps					
			3	19,200 bps					
			4	38,400 bps					
			5	57,600 bps					
Pn00D	Reserved		Do not ch	ange.	0				
Pn00E	Reserved		Do not ch	ange.	0				
Pn00F	Reserved		Do not ch	ange.	0				
Pn010	Position Loop Gain (RT)	2	Sets the p Increasing shortens Oscillation mum resp	oosition loop responsiveness. g the gain increases position control responsiveness and stabilization time. n or overshoot will occur if set too high. Adjust for opti- ponsiveness.	400	×0.1 [1/s]	0 to 30000	Online	Settings can be changed when the axes are stopped
Pn011	Speed Loop Gain (RT)	2	Sets the s If the Iner the Servo Increasing but too m Small gain Adjust for	speed loop responsiveness. tia Ratio (Pn020) is set correctly, this parameter is set to motor response frequency. g the gain increases the speed control responsiveness, uch gain may cause oscillating. n may cause overshoot in the speed response. optimum responsiveness.	500	×0.1 Hz	1 to 30000	Online	(Busy Flag for each axis = 0).
Pn012	Speed Loop Integration Time Con- stant (RT)	2	Adjusts th Set a larg Decrease Set 9999 tion value	e speed loop integration time constant. e value for large load inertia. the setting for fast response with small inertia. to stop integration operation while retaining the integra- . A setting of 10000 disables integration.	200	×0.1 ms	1 to 10000	Online	
Pn013	Speed Feedback Filter Time Constant (RT)	2	Sets the t Normally, Increasing reduces if This para Setting (F	ype of speed detection filter time constant. use a setting of 0. g the value reduces the noise of the Servomotor but also is responsiveness. meter is disabled if the Instantaneous Speed Observer 2027) is enabled.	0		0 to 5	Online	
Pn014	Torque Command Filter Time Constant (RT)	2	Adjusts th mand sec The torqu	e first-order lag filter time constant for the torque com- tion. e filter setting may reduce machine vibration.	80	×0.01 ms	0 to 2500	Online	
Pn015	Speed	2	Sets the s	speed feed-forward amount.	300	×0.1 %	0 to 1000	Online	
	Feedfor- ward Amount (RT)		This para required.	meter is particularly useful when fast response is					
Pn016	Feed-for- ward Filter Time Con- stant (RT)	2	Sets the t ter.	ime constant for the speed feed-forward first-order lag fil-	100	×0.01 ms	0 to 6400	Online	
Pn017	Reserved		Do not ch	ange.	0				

Param- eter No.	Parameter name	Param- eter size	Setting	E	xplanation		Default setting	Unit	Setting range	Enable setting	Details
Pn018	0 to 6400	2	Sets the p Same fun	oosition loop gain when ction as Pn010.	using gain 2 switching.		200	×0.1 [1/s]	0 to 30000	Online	Settings can be
Pn019	Speed Loop Gain 2 (RT)	2	Sets the s Same fun	speed loop gain when us oction as Pn011.	sing gain 2 switching.		800	×0.1 Hz	1 to 30000	Online	when the axes are
Pn01A	Speed Loop Integration Time Con- stant 2 (RT)	2	Sets the s switching Same fun Set 9999 tion value	speed loop integration til . to stop integration opera . Setting 10000 disables	me constant when using ga ation while retaining the int s integration.	ain 2 egra-	500	×0.1 ms	1 to 10000	Online	(Busy Flag for each axis = 0).
Pn01B	Speed Feedback Filter Time Constant 2 (RT)	2	Sets the s Same fun When Ins this parar	speed detection filter wh iction as Pn013. Normal tantaneous Speed Obse neter will be disabled.	en using gain 2 switching. ly, use a setting of 0. erver Setting (Pn027) is ena	0		0 to 5	Online		
Pn01C	Torque Command Filter Time Constant 2 (RT)	2	Sets the f section w Same fun	irst-order lag filter time c hen using gain 2 switchi ction as Pn014.	onstant for the torque com ng.	100	×0.01 ms	0 to 2500	Online		
Pn01D	Notch Filter 1 Fre- quency	2	Sets the r sion. This filter load.	notch frequency of notch must be matched with th	filter 1 for resonance supp he resonance frequency of	1500	Hz	100 to 1500	Online		
			100 to 1499	Filter enabled							
Pn01E	Notch Filter 1 Width	2	Selects th	he notch width of notch fi use a setting of 2.	Iter 1 for resonance suppre	2		2	Online		
Pn01F	Reserved		Do not ch	ange.			0				
Pn020	Inertia Ratio (RT)	2	Sets the I tia. Setting [% The inerti the EEPF	oad inertia as a percent 6] = (Load inertia / Rotor a ratio estimated during ROM every 30 minutes.	age of the Servomotor roto inertia) \times 100 realtime autotuning is stor	or iner- red in	300	%	0 to 10000	Online	Settings can be changed when the axes are stopped
Pn021	Realtime Autotuning Mode Selection	2	Sets the of A setting of on the op Normally, Set to 4 to Gain switt Use a set are a prol	operating mode for realti of 3 or 6 will provide fasto eration. Operation, howe erating pattern. use a setting of 1 or 4. o 6 when the Servomoto ching is enabled at set v ting of 7 if operation cha olem.	me autotuning. er response to changes in i ever, may be unstable depe r is used as a vertical axis alues 1 to 6. Inges caused by gain switc	nertia nding ching	0		0 to 7	Online	(Busy Flag for each axis = 0).
				Realtime Autotuning	Degree of change in load inertia						
			0 [Disabled							
			1 +	lorizontal axis mode	Almost no change						
			2		Gradual changes						
			3		Sudden changes						
			4 \	/ertical axis mode	Almost no change						
			5	Gradual changes							
			6	Sudden changes							
			7 (r	7 Gain switching disable Almost no change mode							
Pn022	Realtime Autotuning Machine Rigidity Selection	2	Sets the r Increasing If the valu change ra Always st increase Cannot b	machine rigidity for realti g this value increases th le is changed suddenly l apidly, subjecting the ma art by making small cha the value while monitorir e set to 0 when using the	n will dually	2		0 to F	Online		

Section 4-5

Param- eter No.	Parameter name	Param- eter size	Setting		Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn023	Adaptive Filter Selec- tion	2	Enables of The Adap 0 when d Note V lo fo 0 1 2	or disables the adaptive prive Filter Table Numbrisabled. When the Vibration Filte ow-pass filter type (PnC procibly set to disabled (Adaptive filter disabled Adaptive filter enable Adaptive operation p Adaptive filter enable performed (i.e. retain	e filter. er Display (Pn02F) will be reset to er Selection (Pn024) is set to a 024 = 3 to 5), the adaptive filter is Pn023 = 0). ed. d. eformed. d. Adaptive operation will not be perly	0		0 to 2	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn024	Vibration Filter Selec- tion	2	Selects th • Filter ty • Norm. Vibrat • Low-p Vibrat • Switchin • No sw • Switcl Selec (Pn02 Selec	ne vibration filter type a pe selection al type: ion frequency setting r vass type: ion frequency setting r ng mode selection ritching: Both 1 and 2 a ning with command dir is Vibration Frequency B, Pn02C) ts Vibration Frequency D, Pn02E)	ange 10.0 to 200.0 Hz ange 1.0 to 200.0 Hz are enabled ection: 1 in forward direction 2 in reverse direction	0		0 to 5	Offline	
			0 M 1 2 3 L 4 5	Filter type Normal type .ow-pass type	Switching mode No switching Switching with command direction No switching Switching with command direction					
Pn025	Normal Mode Auto- tuning Operation Setting	2	Sets the o	operating pattern for no Number of rotations Repeat cycles of 2 otations Repeat cycles of single otation	Rotation direction Forward and Reverse (Alternating) Reverse and Forward (Alternating) Forward only Reverse only Forward and Reverse (Alternating) Reverse only Forward and Reverse (Alternating) Reverse only Reverse and Forward (Alternating) Forward only Reverse only	0		0 to 7	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn026	Overrun Limit Set- ting	2	Sets the s command Set to 0 to	Servomotor's allowable I input range. o disable the overrun p	e operating range for the position protective function.	10	×0.1 rota- tion	0 to 1000	Online	
Pn027	Instanta- neous Speed Observer Setting (RT)	2	The Insta accuracy, tion wher When the Feedback ter Time 0 This featu 0 1	ntaneous Speed Obse thereby improving res stopping. i instantaneous speed Filter Time Constant (Constant 2 (Pn01B) ar ire cannot be used with Disabled Enabled	rver improves speed detection ponsiveness and reducing vibra- observer is enabled, both Speed (Pn013) and Speed Feedback Fil- e disabled. h realtime autotuning.	0		0, 1	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn028	Notch Filter 2 Fre- quency	2	Sets the r sion. This para the load. 100 to 1499 1500	notch frequency of noto meter must be matche Filter enabled Filter disabled	ch filter 2 for resonance suppres- d with the resonance frequency of	1500	Hz	100 to 1500	Online	

Param- eter No.	Parameter name	Param- eter size	Setting	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn029	Notch Filter 2 Width	2	Selects th	e notch width of notch filter 2 for resonance suppression. g the value increases the notch width.	2		0 to 4	Online	Settings can be
Pn02A	Notch Filter 2 Depth	2	Selects th sion. Increasing the phase	e notch depth of notch filter 2 for resonance suppres- g this value decreases the notch depth, thereby reducing lag.	0		0 to 99	Online	when the axes are stopped (Busy
Pn02B	Vibration Frequency 1	2	Sets the vibration a Measure The frequithe Vibrat • Normal Setting f 0 to 99) • Low-pas Setting f to 9)	ribration frequency 1 for damping control to suppress at the end of the load. and set the frequency of the vibration. ency setting range depends on the filter type selected in ion Filter Selection (Pn024). type requency range: 10.0 to 200.0 Hz (Disabled when set to ss type requency range: 1.0 to 200.0 Hz (Disabled when set to 0	0	×0.1 Hz	0 to 2000	Online	each axis = 0).
Pn02C	Vibration Filter 1 Set- ting	2	When set torque sa faster. Normally, The settin tion Filter the range Note T a • Normal Setting • Low-pas Setting	ting Vibration Frequency 1 (Pn02B), reduce this setting if turation occurs, or increase it to make the movement use a setting of 0. Ig range depends on the filter type selected in the Vibra- Selection (Pn024), and if Vibration Filter 1 is enabled, s are as follows: his parameter is disabled when Vibration Filter 1 is dis- bled. type range: 100 \leq Pn02B + Pn02C \leq Pn02B \times 2 or 2000 is type range: 10 \leq Pn02B + Pn02C \leq Pn02B \times 6	0	×0.1 Hz	-200 to 2000	Online	
Pn02D	Vibration Frequency 2	2	Same fun	ction as Pn02B.	0	×0.1 Hz	0 to 2000	Online	
Pn02E	Vibration Filter 2 Set- ting	2	Same fun	ction as Pn02C.	0	×0.1 Hz	-200 to 2000	Online	
Pn02F	Adaptive Filter Table Number Display	2	Displays t the adapt This para enabled (a value of When the EEPROM enabled ti operation value. To clear ti the adapt to 0, and 0 to 4 5 to 48 49 to 64	he table entry number corresponding to the frequency of ive filter. meter is set automatically when the adaptive filter is .e., when the Adaptive Filter Selection (Pn023) is set to her than 0), and cannot be changed. adaptive filter is enabled, this parameter will be saved in approximately every 30 min. If the adaptive filter is ne next time the power supply is turned ON, adaptive will start with the data saved in EEPROM as the default his parameter and reset the adaptive operation, disable we filter by setting the Adaptive Filter Selection (Pn023) then enable it again. Filter disabled Filter enabled Enable or disable the filter with Pn022	0		0 to 64	Online	This is a read-only parame- ter. You cannot change the set- ting.
Pn030	Gain Switching Operating Mode Selection (RT)	2	Enables o 0 1	or disables gain switching. Disabled. Uses Gain 1 (Pn010 to Pn014). PI/P operation is switched from MECHATROLINK-II. The gain is switched between Gain 1 (Pn010 to Pn014) and Gain 2 (Pn018 to Pn01C).	1		0, 1	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).

Section 4-5

Param- eter No.	Parameter name	Param- eter size	Setting	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn031	Gain Switch	2	Sets the t	rigger for gain switching.	2		0 to 10	Online	Settings
	Setting (RT)		The detai	s depend on the control mode.					can be changed
			0	Always Gain 1					when the
			1	Always Gain 2					stopped
			2	Switching from the network					(Busy Flag for
			3	Amount of change in torque command					each axis
			4	Always Gain 1					= 0).
			5	Speed command					
			6	Amount of position deviation					
			7	Position command pulses received					
			8	Positioning Completed Signal (INP) OFF					
			9	Actual Servomotor speed					
			10	Combination of position command pulses received and speed					
Pn032	Gain Switch Time (RT)	2	Enabled v 10. Sets t switching	vhen the Gain Switch Setting (Pn031) is set to 3, or 5 to he lag time from the trigger detection to actual gain when switching from gain 2 to gain 1.	30	×166 μs	0 to 10000	Online	
Pn033	Gain Switch Level Set- ting (RT)	2	Sets the j when the The unit f Switch Se	udgment level to switch between Gain 1 and Gain 2 Gain Switch Setting (Pn031) is set to 3, 5, 6, 9, or 10. or the setting depends on the condition set in the Gain witting (Pn031).	600		0 to 20000	Online	
Pn034	Gain Switch Hysteresis Setting (RT)	2	Sets the h Setting (P 5, 6, 9, or in the Gai	systeresis of the judgment level for the Gain Switch Level n033) when the Gain Switch Setting (Pn031) is set to 3, 10. The unit for the setting depends on the condition set n Switch Setting (Pn031).	50		0 to 20000	Online	
Pn035	Position Loop Gain Switching Time (RT)	2	This para suddenly fer by a la When the value + 1)	meter can prevent the position loop gain from increasing when the position loop gain and position loop gain 2 dif- rge amount. position loop gain increases, it takes the duration of (set $\times 166 \mu$ s.	20	×166 μs	0 to 10000	Online	
Pn036	Reserved		Do not ch	ange.	0				
Pn037	Reserved		Do not ch	ange.	0				
Pn038	Reserved		Do not ch	ange.	0				
Pn039	Reserved		Do not ch	ange.	0				
Pn03A	Reserved		Do not ch	ange.	0				
Pn03B	Reserved		Do not ch	ange.	0				
Pn03C	Reserved		Do not ch	ange.	0				
Pn03D	Jog Speed	2	Sets the j	og operation speed with the Parameter Unit or CX-Drive. og operation is only available when the network is not stablished. Do not try to establish the network while sing jog operation. Otherwise, command alarm (alarm ode 27) will occur.	200	r/min	0 to 500	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn03E	Reserved		Do not ch	ange.	0				
Pn03F	Reserved		Do not ch	ange.	0				
Pn040	Reserved		Do not ch	ange.	0				
Pn041	Emergency Stop Input Setting	2	Enables t Note If b	this function is disabled, the response status will always 0 (disabled).	1		0, 1	Offline	
			0	Disabled.					
			1	Enabled (alarm code 87 issued on OPEN)					
Pn042	Origin Prox-	2	Sets the I	ogic for the Origin Proximity Input (DEC).	1		0, 1	Offline	
	Logic Set-		0	N.C contact (origin proximity detected on OPEN)					
	ting		1	N.O contact (origin proximity detected on CLOSE)					
Pn043	Operating Direction Setting	2	Sets the r the netwo	elationship between polarity of operation data sent over rk and the direction of Servomotor rotation.	1		0, 1	Offline	
	- 3		note ir ((SP, IM) on the front panel, forward direction is always ositive (+), and reverse rotation is always negative (–).					
			0	Disabled.					
			1	Enabled (alarm code 87 issued on OPEN)					

Param- eter No.	Parameter name	Param- eter size	Setting	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn044	Input Sig-	2	Sets the t	erminal assignment for Drive Prohibit Input.	1		0, 1	Offline	
	nal Selec-		0	Sets CN1 pin 19 to POT, CN1 pin 20 to NOT.					
	lion		1	Sets CN1 pin 19 to NOT, CN1 pin 20 to POT.					
Pn045	Reserved		Do not ch	ange.	0				
Pn046	Reserved		Do not ch	ange.	0				
Pn047	Reserved		Do not ch	ange.	0				
Pn048	Reserved		Do not ch	ange.	0				
Pn049	Reserved		Do not ch	ange.	0				
Pn04A	Reserved		Do not ch	ange.	0				
Pn04B	Reserved		Do not ch	ange.	0				
Pn04C	Reserved		Do not ch	ange.	0				
Pn04D	Reserved		Do not ch	ange.	0				
Pn04E	Reserved		Do not ch	ange.	0				
Pn04F	Reserved		Do not ch	ange.	0				
Pn050	Reserved		Do not ch	ange.	0				
Pn051	Reserved		Do not ch	ange.	0				
Pn052	Reserved		Do not ch	ange.	0				
Pn053	Speed Limit	2	Sets the s absolute This para ting (Pn07	speed limit for torque control mode. (The value is an value) meter is limited by the Overspeed Detection Level Set- 73).	50	r/min	-20000 to 20000	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn054	Reserved		Do not ch	ange.	0				
Pn055	Reserved		Do not ch	ange.	0				
Pn056	Reserved		Do not ch	ange.	0				
Pn057	Reserved		Do not ch	ange.	0				
Pn058	Soft Start Accelera- tion Time	2	Sets the a Accelerat value $\times 2$	acceleration time for speed control mode. ion time [s] from 0 r/min to maximum speed [r/min] = Set ms	0	×2 ms	0 to 5000	Online	Settings can be changed when the
Pn059	Soft Start Decelera- tion Time	2	Sets the of Decelerat value $\times 2$	deceleration time for speed control mode. ion time [s] from maximum speed [r/min] to 0 r/min = Set ms	0	×2 ms	0 to 5000	Online	axes are stopped (Busy Flag for each axis = 0).
Pn05A	Reserved		Do not ch	ange.	0				
Pn05B		2	Selects th	e speed limit for torque control mode.	0		0, 1	Online	Settings
Dr050	Poorred		0	Use the Speed Limit (Pn053) Use the speed limit value via MECHATROLINK-II or the Speed Limit (Pn053), whichever is smaller.					changed when the axes are stopped (Busy Flag for each axis = 0).
Ph05C	Reserved		Do not ch	ange.	0				
Ph05E	No 1	2	Sote the	anyo.	300	0/_	0 to 500	Online	Settings
FIIUSE	Torque Limit	2	Refer to in select the The maxi ble Servo	nformation on the Torque Limit Selection (Pn003) to torque limit. mum value of the setting range depends on the applica- motor.	300	70	010 500	Online	can be changed when the axes are stonned
Pn05F	No. 2 Torque Limit	2	Sets the I Refer to in select the The maxing ble Servo	No. 2 torque limit for the Servomotor output torque. nformation on the Torque Limit Selection (Pn003) to torque limit. mum value of the setting range depends on the applica- motor.	100	%	0 to 500	Online	(Busy Flag for each axis = 0).
Pn060	Positioning Completion Range 1	2	Sets the p tion 1 (INI Positionin exhausted verted intervented	positioning completion range when Positioning Comple- P1) Output is selected. g is complete when all positioning command pulses are d, and the absolute value of the position deviation con- o command units is less than this setting.	25	Com- mand units	0 to 10000	Online	

Param- eter No.	Parameter name	Param- eter size	Setting	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn061	Speed Con- formity Sig- nal Output Width	2	Sets the c (VCMP) s Speed co ference bo and decel less than Note T	detection width for the speed conformity detection ignal. nformity is achieved when the absolute value of the dif- etween the internal speed command (before acceleration leration limits are applied) and the Servomotor speed is the set speed. his setting has a hysteresis of 10 r/min.	20	r/min	10 to 20000	Online	
Pn062	Rotation Speed for Motor Rota- tion Detec- tion	2	Sets the t Speed rea vomotor s Note S	hreshold level for the speed reached (TGON) signal. ached is determined when the absolute value of the Ser- peed is greater than the setting speed. peed reached detection has a hysteresis of 10 r/min.	50	r/min	10 to 20000	Online	
Pn063	Positioning Completion Range 2	2	Sets the p tion 2 (INI Positionin deviation regardles cessed.	positioning completion range when Positioning Comple- P2) is selected. g is complete when the absolute value of the position converted into command units is less than this setting, s of whether position command pulses are still being pro-	100	Com- mand units	0 to 10000	Online	
Pn064	Motor Phase Cur- rent Offset Re-adjust- ment Set- ting	2	Enables c the Motor inputs. Th ON. Note T w d c c 0	r disables the offset component readjustment function of Phase Current Detector (CT) for Servo ON command le readjustment is made when control power is turned his adjustment is inaccurate if the offset is measured hile the Servomotor is rotating. To enable this function, o not rotate the Servomotor when inputting the Servo ON ommand. Disabled (only when turning ON control power) Enabled (when turning ON control power, or at Servo ON)	0		0, 1	Online	
Pn065	Undervolt- age Alarm Selection	2	Selects w function (a rupted for ing Servo 0	hether to activate the main power supply undervoltage alarm code 13) when the main power supply is inter- the duration of the Momentary Hold Time (Pn06D) dur- ON. Turns the Servo OFF according to the setting for the Stop Selection with Main Power OFF (Pn067), interrupting the positioning command generation process (positioning operation) within the Servo Drive. When the main power supply is turned back ON, Servo ON will resume. Restart the positioning operation after performing the positioning operation and recovering from Servo OFF. Causes an error due to main power supply undervoltage (alarm code 13). This parameter is disabled if Pn06D = 1,000. If Pn06D is set too long and the voltage between P and N in the main power supply converter drops below the specified value before a main power supply interruption is detected, a main power supply undervoltage (alarm code 13) will occur.	1		0, 1	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).

Param- eter No.	Parameter name	Param- eter size	Set	ting		Explar	nation		Default setting	Unit	Setting range	Enable setting	Details
Pn066	Stop Selec- tion for Drive Prohi-	2	Sets ward (NO	s the d d Drive T) is e	leceleratio e Prohibit I enabled.	n stop operation to Input (POT) or Rev	be performed after the f erse Drive Prohibit Input	For- t	0		0 to 2	Offline	
	Input			D dec	Ouring eleration	After stopping (30 r/min or less)	Deviation counter						
			0	Dyna brak	amic æ	Disables torque command in drive prohibited direction	Cleared while decel- erating with dynamic brake. Retained after stopping.						
			1	Disa torqu	ubles ue	Disables torque command in drive prohibited direction	Cleared while decel- erating. Retained after stopping.						
			2	Eme Stop (Pn0	ergency o Torque 06E)	Servo locked	Retained while decel- erating, cleared upon completion of decel- eration, and retained after stopping.						
P067	Ston Selec-	2	Note 1 Note 2 Note 3		he position peration) we acceleration orque contr control oper- bon, the inte- ill be retain oning oper- boning oper- boning oper- then the S stopped), the ven if the d function dev function dev bosition dev function dev funco	he positioning command generation process (positioning peration) within the Servo Drive will be forcibly stopped ince it enters the deceleration mode. Also, when the occleration mode is activated during speed control or rigue control, it will switch to position control. If a posi- ioning operation command is received during decelera- an, the internal positioning command generation process II be retained, and after deceleration is complete, posi- ioning operation will be activated. Then the Servomotor rotation speed is 30 r/min or less topped), the deceleration mode will not be activated ren if the drive prohibit input is enabled. Then the parameter is set to 2 and an operation com- and in the drive prohibited direction is received after opping, a command warning (warning code 95h) will be sued. When the parameter is set to 0 or 1, the operation mmand in the prohibited direction after stopping will be istion deviation will accumulate because the torque ommand is 0. Take measures such as issuing a com- and in the reverse direction from the host controller. Then the parameter is set to 2, MECHATROLINK-II com- unications are interrupted, and either Forward or everse Drive Prohibit Input (POT or NOT) is turned ON, ceiving an operation command (jog operation or normal ode autotuning) via RS232 will cause a Drive Prohibit put Error (alarm code 38). A Drive Prohibit Input Error larm code 38) will also occur if either POT or NOT is red ON while operation or normand					0 to 7	Online	Settings
	tion with Main Power OFF	-	stop ervo will t	ping a ltage be res	Alarm Selection the market when the lase dyna	ain power supply is ection (Pn065) set ne power OFF is de	turned OFF with the Ui to 0. The deviation coun tected.	nd- nter	0				can be changed when the axes are
			1 00	d E	with dyna	mic brake.	nd romain atomad with						stopped (Busy Flag for
			i an	c u	dynamic	brake.							each axis $= 0$).
			2 an	101 15	Use dyna when sto	pped.	erate, but free the motor	r					
			3 an	id 7	Use free- stopped.	run to decelerate, a	and free the motor wher	٦					
Pn068	Stop Selec- tion for Alarm Gen- eration	2	Sets issu rese	the d ed by t whe	leceleratio the protec n an alarm	n process and stop tive function. The c i is issued.	status after an alarm is leviation counter will be	3	0		0 to 3	Online	
	eration		0		Use dyna with dyna	mic brake to decel mic brake.	erate and remain stoppe	ed					
			1		Use free- dynamic	run to decelerate a brake.	nd remain stopped with	1					
			2		Use dyna when sto	mic brake to decel	erate, but free the motor	r					
			3		Use free- stopped.	run to decelerate, a	and free the motor wher	n					
Pn069	Stop Selec- tion with	2	Sets after	s the o	perational ping when	conditions to apply the Servo is turned	/ during deceleration an d OFF.	ıd	0		0 to 7	Online	
	Servo OFF		The cour Sele	relation ter prection	onship bet rocessing f with Main	ween set values, o for this parameter is Power OFF (Pn067	peration, and deviation s the same as for the St 7).	top					

Section 4-5

Param- eter No.	Parameter name	Param- eter size	Setting	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn06A	Brake Tim- ing When Stopped	2	Sets the c turns OFF command Note T fr fr	Juration from when the Brake Interlock (BKIR) signal to when the Servomotor is de-energized when the RUN I is turned OFF with the Servomotor stopped. he brake interlock signal is the logical OR of the brake elease request from the network and the release request om the Servo controller. Note, the brake release request om the network is OFF (operation request is ON) at ower ON.	10	2 ms	0 to 1000	Online	Settings can be changed when the axes are stopped (Busy Flag for acch axis
Pn06B	Brake Tim- ing during Operation	2	When the motor rota tion speed after the t BKIR turn before the Note T fr fr	run command (RUN) is turned OFF during the Servo- ation, the Servomotor will decelerate reducing the rota- d and the Brake Interlock Signal (BKIR) will turn OFF ime set by this parameter has elapsed. Is OFF if the Servomotor speed drops below 30 r/min e set time. The brake interlock signal is the logical OR of the brake please request from the network and the release request om the Servo controller. Note, the brake release request om the network is OFF (operation request is ON) at ower ON.	50	2 ms	0 to 1000	Online	= 0).
Pn06C	Regenera- tion Resis- tor Selection	2	Sets the r overload (Set this pa If using ar main pow	egeneration resistor operation and the regeneration (alarm code 18) operation. arameter to 0 if using the built-in regeneration resistor. n external regeneration resistor, be sure to turn OFF the er when the built-in thermal switch is activated.	0		0 to 3	Offline	
			0	Sets the regeneration overload to match the built-in regeneration resistor. (regeneration load ratio below 1%) The regeneration overload (alarm code 18) occurs					
			2	when the load ratio of the external regeneration resistor exceeds 10%. The regeneration processing circuit by the external regeneration resistor is activated, but the regeneration					
			3	The regeneration processing circuit is not activated. All regenerative energy is absorbed by the built-in capacitor					
Pn06D	Momentary Hold Time	2	Sets the a power sup The main set to 100	amount of time required to detect shutoff when the main oply continues to shut off. power OFF detection will be disabled if this parameter is 0.	35	2 ms	35 to 1000	Offline	
Pn06E	Emergency Stop Torque	2	Sets the t hibition In (Pn066) is When this The maxin motor.	orque limit during deceleration because of the Drive Pro- put when the Stop Selection for Drive Prohibition Input s set to 2. s parameter is set to 0, the normal torque limit will be set. mum value of the setting range depends on the Servo-	0	%	0 to 300	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn06F	Reserved		Do not ch	ange.	0				
Pn070	Reserved		Do not ch	ange.	0				
Pn071	Reserved		Do not ch	ange.	0				
Pn072	Overload Detection Level Set- ting	2	Sets the c be set at 1 of 0, and 1 level.	overload detection level. The overload detection level will 115% if this parameter is set to 0. Normally, use a setting set the level only when reducing the overload detection	0	%	0 to 500	Online	
Pn073	Overspeed Detection Level Set- ting	2	Sets the c is 1.2 time paramete only wher Note T a	pverspeed detection level. The overspeed detection level es the maximum Servomotor rotation speed when the r is set to 0. Normally, use a setting of 0, and set the level n reducing the overspeed detection level. he detection margin of error for the setting is ± 3 r/min for 7-core absolute encoder and ± 36 r/min for a 5-core acremental encoder.	0	r/min	0 to 20000	Online	
Pn074	Reserved		Do not ch	ange.	0				
Pn075	Reserved		Do not ch	ange.	0				
Pn076	Reserved		Do not ch	ange.	0				
Pn077	Reserved		Do not ch	ange.	0				
Pn078	Reserved		Do not ch	ange.	0				
Pn079	Reserved		Do not ch	ange.	0				
Pn07A	Reserved		Do not ch	ange.	0				
Pn07B	Reserved		Do not ch	ange.	0				
Pn07C	Reserved		Do not ch	ange.	0				
Pn07D	Reserved		Do not ch	ange.	0				

Param- eter No.	Parameter name	Param- eter size	Setting	Exp	blanation	Default setting	Unit	Setting range	Enable setting	Details
Pn07E	Reserved		Do not ch	ange.		0				
Pn07F	Reserved		Do not ch	ange.		0				
Pn100	Backlash Compensa-	2	Enables of trol, and s	or disables the backlash co sets the compensation dire	perpensation for position con-	• 0		0 to 2	Offline	
	tion Selec- tion		0	Disabled						
			1	Compensates in the initia Servo ON.	al positive direction after the					
			2	Compensates in the initia Servo ON.	al negative direction after the	!				
Pn101	Backlash Compensa- tion	2	Sets the t	backlash compensation an	nount for position control.	0	Com- mand units	-32768 to 32767	Online	Settings can be changed
Pn102	Backlash Compensa- tion	2	Sets the to control.	backlash compensation tin	ne constant for position	0	0.01 ms	0 to 6400	Online	axes are stopped
	Time Con- stant		Value of Pn100	Pn101 = Positive number	Pn101 = Negative number					Flag for each axis = 0).
			1	Compensates in posi- tive direction during rotation in positive direction	Compensates in nega- tive direction during rotation in positive direction					
			2	Compensates in posi- tive direction during rotation in negative direction	mpensates in posi- direction during ation in negative Compensates in nega- tive direction during rotation in negative direction e. compensates in nega- tive direction during rotation in negative direction e. compensates in nega- tive direction during rotation in negative direction e. compensates in nega- tive direction e. compensates in nega- tive direction e. compensates in nega- tive direction e. compensates in nega- tive direction					
Pn103	Reserved		Do not ch	ange.		0				
Pn104	Soft Limit	2	Enables of When ena Limit (Pn2 Note T tin a re	or disables the soft limit. abled, the soft limit values 201) and Reverse Software he response value for limit ng will be set to 0. The res Iso set to 0 when the Serv eturn to origin.	are set in Forward Software e Limit (Pn202). t signals disabled by this set- ponse value for limit signals omotor does not complete its	0 is s		0 to 3	Online	
			0	Enable both the Forward (Pn201 and Pn202)	/ Reverse Software Limits					
			1	Disable the Forward Soft Reverse Software Limit (ware Limit (Pn201), enable th Pn202)	ie				
			2	Enable the Forward Softw Reverse Software Limit (vare Limit (Pn201), disable th Pn202)	ie				
			3	Disable both the Forward (Pn201 and Pn202)	d / Reverse Software Limits					
Pn105	Origin Range	2	Sets the t values. ZPOINT = tem setup ting range	hreshold for detecting the = 1 when the return to orig b is complete) and the feed of this parameter.	origin (ZPOINT) in absolute in completes (coordinate sys lback position is within the se	10 ;- ət-	Com- mand units	0 to 250	Online	
Pn106	Reserved		Do not ch	ange.		0				
Pn107	Linear Accelera- tion Constant	2	Sets the a A setting The settin data (0 to Example:	acceleration for positioning of "0" is regarded as "1". ng will be handled after cor 65535). $-32768 \rightarrow 8000h = 3270$ $-1 \rightarrow FFFFh = 65535$	ge. eleration for positioning operations. "0" is regarded as "1". will be handled after conversion to an unsigned 16-bit 535). -32768 \rightarrow 8000h = 32768 -1 \rightarrow FFFFh = 65535			-32768 to 32767	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn108	Reserved		Do not ch	ange.		0				
Pn109	Reserved		Do not ch	ange.	nge.					
Pn10A	Linear Decelera- tion Con- stant	2	Sets the of A setting The settin data (0 to Example:	deceleration for positioning of "0" is regarded as "1". ing will be handled after cor 65535). $-32768 \rightarrow 8000h = 3270$ $-1 \rightarrow FFFFh = 65535$	celeration for positioning operations. "0" is regarded as "1". will be handled after conversion to an unsigned 16-bit 5535). $-32768 \rightarrow 8000h = 32768$ $-1 \rightarrow FFFFh = 65535$			-32768 to 32767	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn10B	Reserved		Do not ch	ange.	ge.					
Pn10C	Reserved		Do not ch	ange.		0				
Pn10D	Reserved		Do not ch	ange.		0				

Param- eter No.	Parameter name	Param- eter size	Setting	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn10E	Moving Average Time	2	Sets the i Note lf a a v	noving average time for position commands. the Moving Average Time is set, commands may not be xecuted seamlessly when switching the control mode, nd when switching between interpolation feed motions nd positioning motions (motions wherein the command vaveforms are generated inside the Servo Drive).	0	×0.1 ms	0 to 5100	Online	Settings can be changed when the axes are stopped (Busy
Pn10F	Origin	2	Sets the	lirection for origin return.	0		0, 1	Online	Flag for
	Node Set-		0	Positive direction					= 0).
	tings		1	Negative direction					
Pn110	Origin Return Approach Speed 1	2	Sets the oproximity latch sign This para nally the s	operating speed for origin return from when the origin signal is turned ON, to when it is turned OFF and the al is detected. meter can be set to a maximum value of 32767, but inter- speed is limited to the Servomotor's maximum speed.	50	100 [com- mand units/ s]	1 to 32767	Online	
Pn111	Origin Return Approach Speed 2	2	Sets the d after the l Distance This para	operating speed for origin return, from when the point atch signal is detected to when the Origin Return Final (Pn204) is reached. meter can be set to a maximum value of 32767, but inter-	5	100 [com- mand units/	1 to 32767	Online	
			nally the	speed is limited to the Servomotor's maximum speed.		sj			
Pn112	General- purpose	2	Selects th	e tunction for general-purpose output 1 (OUTM1).	7		0 to 9	Offline	Standard setting
	Output 1		0	Always OFF					coung
	Selection		1	INP1 output. Turn ON when position deviation is equal to or less than Pn060 for position control. Undefined when not using position control.					
			2	VCMP output. Turn ON when the deviation between the Servomotor speed and commanded speed is within the range set by Pn061 for speed control. Undefined when not using speed control.					
			3	TGON output. Turn ON when the absolute value of the Servomotor speed exceeds Pn062 setting in all control modes.					
			4	READY output. Turn ON when the main power is supplied, there is no alarm, and Servo SYNC with a host controller is estab- lished in all control modes.					
			5	CLIM output. Turn ON when torque limit is activated in all control modes.					
			6	VLIM output. Turn ON when the Servomotor speed reaches the speed limit for torque control. Undefined when not using torque control.					
			7	BKIR output. Turn ON with the release timing of the brake release signal in all control modes.					
			8	WARN output. Turn ON when a warning is issued in all control modes.					
			9	INP2 output. Turn ON when the position deviation is equal to or less than the Positioning Completion Range 2 (Pn063) for position control. Undefined when not using position control.					
Pn113	General- purpose Output 2 Function Selection	2	Selects th The set v pose outp	e function for general-purpose output 2 (OUTM2). alues and the functions are the same as for general-pur- ut 1 (OUTM1).	0		0 to 9	Offline	
Pn114	General- purpose Output 3 Function Selection	2	Selects th The set v pose outp	e function for general-purpose output 3 (OUTM3). alues and the functions are the same as for general-pur- ut 1 (OUTM1).	0		0 to 9	Offline	
Pn115 to Pn13F	Reserved		Do not ch	ange.	0				
Pn200	Absolute Origin Off- set	4	Sets the cal coord	offset amount for the encoder position and the mechani- nate system position when using an absolute encoder.	0	Com- mand units	-1073741823 to 1073741823	Offline	

Section 4-5

Param- eter No.	Parameter name	Param- eter size	Setting	Explar	nation		Default setting	Unit	Setting range	Enable setting	Details
Pn201	Forward Software Limit	4	Sets the soft lir If the Servomo (PSOT) will tur Note Be sur Revers Note PSOT	nit in the forward direction tor exceeds the limit, the n ON (=1). The to set the limits so that se Software Limit. The software don when o	on. e network response statu : Forward Software Limit prigin return is incomplet	us t > te.	500000	Com- mand units	-1073741823 to 1073741823	Online	
Pn202	Reverse Software Limit	4	Sets the soft lin If the Servomo (NSOT) will tur Note Be sur Revers Note NSOT	nit for the reverse directi tor exceeds the limit, the n ON (=1). e to set the limits so tha se Software Limit. is not turned ON when o	on. e network response statu t Forward Software Limit prigin return is incomplet	us t > te.	-500000	Com- mand units	-1073741823 to 1073741823	Online	
Pn203	Final Dis- tance for External Input Posi- tioning	4	Sets the distan position when The operation determined by parameter as for	ce to travel after detectin performing external inpu after detecting the latch the external input position ollows.	ng the latch signal input t positioning. signal input position will oning direction and this	be	100	Com- mand units	-1073741823 to 1073741823	Online	Settings can be changed when the axes are stopped
			External	Si	gn						(Busy Flag for
			positioning	Positive	Negative						each axis = 0).
			Positive direction	Moves in the positive direction and stops*1	Decelerates to a stop, reverses, then moves in the nega- tive direction and stops						
			Negative direction	Decelerates to a stop, reverses, then moves in the positive direction and stops Moves in the negative direction and stops tive direction and stops*1 es after decelerating to a stop if the final distance for di input positioning is short in comparison to the deceleratione stance The final distance for direction and stops							
			*1. Reverses a external inp ation distan	fter decelerating to a sto ut positioning is short in ce.	r eler-						
Pn204	Origin Return Final Dis- tance	4	Sets the distan when performin The operation determined by lows.	ce from the latch signal ng origin return. after detecting the latch the origin return directio	gin be fol-	100	Com- mand units	-1073741823 to 1073741823	Online		
			Origin	Si	gn						
			direction	Positive	Negative						
			Positive direction	Moves in the positive direction and stops*1	Decelerates to a stop, reverses, then moves in the nega- tive direction and stops						
			Negative direction	Moves in the nega- tive direction and stops ^{*1}	Decelerates to a stop, reverses, then moves in the positive direction and stops						
			*1. Reverses a for origin re tance.	fter decelerating to a sto turn is short in comparis	p if the final travel distar on to the deceleration d	nce lis-					
Pn205	Electronic Gear Ratio 1 (Numera- tor)	4	Sets the nume Setting this partion as the num 10000 for a 2.5	rator for the electronic g rameter to 0 automatical herator. (131072 for a 17 500-p/r incremental enco	ear ratio. ly sets the encoder reso -bit absolute encoder, o der).	olu- or	1		0 to 131072	Offline	
			Note Set the 100 tin occur i) for a 2,500-p/r incremental encoder). Set the electronic gear ratio within the range of 1/100 to 100 times. A parameter setting alarm (alarm code 93) will occur if the ratio is set outside of this range.							
Pn206	Electronic Gear Ratio 2 (Denomi- nator)	4	Sets the denor Note Set the 100 tin occur i	denominator for the electronic gear ratio. Set the electronic gear ratio within the range of 1/100 to 100 times. A parameter setting alarm (Alarm code 93) will occur if the ratio is set outside of this range.			1		1 to 65535	Offline	
Pn207	Reserved		Do not change	of change.			0				
Pn208	Reserved		Do not change	ot change.							
Pn209	Deviation Counter Overflow Level	4	Sets the deviat The value will b multiplying with Setting this par	deviation counter overflow level. e will become saturated at 134217728 (= 2^{27}) pulses a ng with the electronic gear ratio. his parameter to 0 will disable deviation counter overflo				Com- mand units	0 to 2147483647	Online	
Pn20A to Pn21F	Reserved		Do not change				0				

4-5-3 W-series Servo Drives (R88D-WT with FNY-NS115)

The Servo Parameters listed here can be used when the Position Control Unit is used with an OMRON W-series Servo Drive (R88D-WT) with a MECHA-TROLINK-II I/F Unit (FNY-NS115).

For further details on each of the parameters, refer to the user's manuals for W-series Servo Drives and MECHATROLINK-II I/F Unit (FNY-NS115).

The default settings for parameters Pn000 to Pn601 in the parameter tables for the R88D-WT□ W-series Servo Drive used with the FNY-NS115 are automatically set to the default parameter settings for when the FNY-NS115 MECHATROLINK-II I/F Unit is mounted to a W-series Servo Drive. Parameters from Pn800 onwards are enabled when the FNY-NS115 is mounted. The parameter names and default settings for these parameters follow those specified by the FNY-NS115.

The timing for Servo Parameters to be enabled are classified into the following two types.

Online (online parameter): Changed settings are enabled immediately after Servo Parameters have been written.

Offline (offline parameters): Changed settings are enabled when the Servo Drive power is cycled or DEVICE SETUP is executed.

The *Details* column indicates whether the following conditions apply to the corresponding parameter.

- Online parameters that can be changed when the axis is stopped (Busy Flag = 0) only.
- Parameters that are automatically set when a FNY-NS115 MECHA-TROLINK-II I/F Unit is installed and Parameters that are used with fixed set values are indicated in the *Details* column as "fixed" with the set value.
- Parameters with standard set values when using MECHATROLINK are indicated in the *Details* column as "standard" with the set value. The standard settings are basic settings required when using the Position Control Unit. This manual describes operations assuming standard settings are being used.

For details on fixed and standard settings, refer to 6-4 Standard Settings for Servo Drives Using MECHATROLINK.

Note Make sure that the equipment will not be adversely affected before changing the Servo Parameters (WRITE SERVO PARAMETER, SAVE SERVO PARAMETER). Refer to the Servo Drive's user's manual and always check the effect of changing the settings before changing the Servo Parameters.

Function Selection Parameters

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details	
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting		
Pn000	Function selection basic	2	0	Reverse rota- tion	0	CCW direction is taken for positive command.	0			Offline		
	switch				1	CW direction is taken for positive command.						
			1	Not used.	1	The setting is dis- abled. (Do not change the set- ting.)	1					
			2	Unit No. setting	0 to F	Servo Drive com- munications unit number setting when using per- sonal computer monitoring soft- ware	0			Offline		
			3	Not used.	0	(Do not change the setting.)	0					
Pn001	Function selection applica-	2	0	Select stop if an alarm occurs when	0	Servomotor stopped by dynamic brake.	2			Offline		
	1			OFF	1	Servomotor stopped by dynamic brake. Dynamic brake released after stop- ping.						
			•			2	Servomotor stopped with free run.					
			1	Select stop when prohib- ited drive is input	0	Stop according to Pn001.0 settings (release Servomo- tor after stopping).	0			Offline		
					1	Stop Servomotor using torque set in Pn406, and lock Servomotor after stopping.						
					2	Stop Servomotor using torque set in Pn406, and release Servomotor after stopping.						
			2	Select AC/DC power input	0	AC power supply: AC power supplied from L1, L2, (L3) terminals	0			Offline		
					1	DC power supply: DC power supplied from +1, - termi- nals						
			3	Select warning code output	0	Alarm code only output from ALO1, ALO2, ALO3	1			Offline		
					1	Alarm code and warning code both output from ALO1, ALO2, ALO3						

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details							
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting								
Pn002	Function selection	2	0	Torque com- mand input	0	Option command value not used.	0			Offline								
	applica- tion switch 2			ing speed con- trol)	1	Option command value 1 used as torque limit input.												
					2	Option command value 1 used as the torque feed for- ward input.												
				3	Option command values 1 and 2 used as torque limit inputs according to the forward/reverse rotation current limit setting.													
			1	Speed com- mand input change (dur-	0	Option command value not used.	0			Offline								
			2	change (dur- ing torque con- trol)	1	Option command value 1 used as speed limit input.												
			2	Operation switch when using absolute	0	Use as absolute encoder.	0			Offline								
			3	using absolute encoder	1	Use as incremen- tal encoder.												
				3	Fully-closed encoder usage method	0	Fully-closed encoder is not used.	0			Offline							
						n 						1	Fully-closed encoder is used without phase Z.					
								2	Fully-closed encoder is used with phase Z.									
	3 Fully-closed encoder is used in Reverse Rotation Mode without phase Z.																	
					4	Fully-closed encoder is used in Reverse Rotation Mode with phase Z.												

Section 4-5

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn003	Function selection applica-	2	0	Analog moni- tor 1 (AM) allo- cation	0	Servomotor rota- tion speed: 1 V/1,000 r/min	2			Online	
	tion switch 3				1	Speed command: 1 V/1,000 r/min					
					2	Torque command: 1 V/100%					
					3	Position deviation: 0.05 V/1 command unit					
					4	Position deviation: 0.05 V/100 com- mand units					
					5	Command pulse frequency: 1 V/1,000 r/min					
					6	Servomotor rota- tion speed: 1 V/250 r/min					
					7	Servomotor rota- tion speed: 1 V/125 r/min					
					8 to F	Not used.					
			1	Analog moni- tor 2 (NM) allo- cation	0 to F	Same as Pn003.0	0			Online	
			2 to 3	Not used.	0	(Do not change the setting.)	0				
Pn004	Not used.					(Do not change the setting.)	0200				Fixed setting: 0200
Pn005	Not used.					(Do not change the setting.)	0000				

Servo Gain Parameters

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn100	Speed loop gain	2	Adjusts	speed loop respo	onsivene	SS.	80	Hz	1 to 2000	Online	
Pn101	Speed loop inte- gration constant	2	Speed I	oop integral time	constant	t	2000	×0.01 ms	15 to 51200	Online	
Pn102	Position loop gain	2	Adjusts	position loop res	ponsiven	iess.	40	1/s	1 to 2000	Online	
Pn103	Inertia ratio	2	Sets us inertia a	ing the ratio betw Ind the Servomot	een the i or rotor i	machine system nertia.	300	%	0 to 20000	Online	
Pn104	Speed loop gain 2	2	Adjusts matic ga	speed loop respo ain switching inpu	onsivene ut).	ss (enabled by auto-	80	Hz	1 to 2000	Online	
Pn105	Speed loop inte- gration constant 2	2	Speed I matic ga	oop integral time ain switching inpu	constant it)	t (enabled by auto-	2000	×0.01 ms	15 to 51200	Online	
Pn106	Position loop gain 2	2	Adjusts automa	position loop res tic gain switching	ponsiven input).	less (enabled by	40	1/s	1 to 2000	Online	
Pn107	Bias rota- tional speed	2	Sets po	sition control bias	3.		0	r/min	0 to 450	Online	
Pn108	Bias addi- tion band	2	Sets the deviatio	position control n counter pulse v	bias ope vidth.	ration start using	7	Com- mand unit	0 to 250	Online	
Pn109	Feed-for- ward amount	2	Position	control feed-form	vard com	pensation value	0	%	0 to 100	Online	

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn10A	Feed-for- ward com- mand filter	2	Sets po	osition control fee	d-forward	d command filter.	0	×0.01 ms	0 to 6400	Online	
Pn10B	Speed control setting	2	0	P control switching con- ditions	0	Sets internal torque command value conditions (Pn10C).	4			Online	
					1	Sets speed com- mand value condi- tions (Pn10D).					
					2	Sets acceleration command value conditions (Pn10E).					
					3	Sets deviation pulse value condi- tions (Pn10F).					
					4	No P control switching function					
			1	Speed control	0	PI control	0			Offline	
				loop switching	1	IP control					
			2	Automatic gain switching	0	Automatic gain switching disabled	0			Offline	
					1	Gain switching using position com- mands					
					2	Gain switching using position devi- ation					
					3	Gain switching using position com- mands and position deviation					
			3	Not used.	0	(Do not change the setting.)	0				
Pn10C	P control switching (torque command)	2	Sets lev trol to F	vel of torque com ^o control.	mand to	switch from PI con-	200	%	0 to 800	Online	
Pn10D	P control switching (speed command)	2	Sets lev to P co	vel of speed comr ntrol.	nand to s	witch from PI control	0	r/min	0 to 10000	Online	
Pn10E	P control switching (accelera- tion com- mand)	2	Sets lev control	vel of acceleratior to P control.	n comma	nd to switch from PI	0	10 r/ min/s	0 to 3000	Online	
Pn10F	P control switching (deviation pulse)	2	Sets lev to P co	vel of deviation pu ntrol.	ulses to s	witch from PI control	10	Com- mand unit	0 to 10000	Online	

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn110	Online autotun- ing setting	2	0	Select online autotuning	0	Autotunes initial operations only after power is turned ON.	2			Offline	
					1	Always autotunes.					
					2	No autotuning					
			1	Select speed	0	ON	1			Online	
				feedback com- pensation function	1	OFF					
			2	Select adhe- sive friction	0	Friction compensa- tion: OFF	0			Online	
				function	1	Friction compensa- tion: Rated torque ratio small					
					2	Friction compensa- tion: Rated torque ratio large					
			3	Not used.	0	(Do not change the setting.)	0				
Pn111	Speed feedback compen- sation gain	2	Adjusts	speed loop feed	back gaiı	n.	100	%	1 to 500	Online	
Pn112 to Pn123	Not used.		(Do not	change the setti	ng.)						
Pn124	Automatic gain switching timer	2	Sets the met, wh used. (Pn10B	e switching delay then the automatic 3.2 = 1 to 3).	after cor gain sw	nditions have been itching function is	100	ms	1 to 10000	Online	
Pn125	Automatic gain switching width (amount of position deviation)	2	Sets the dition w position	e position deviatio /hen the automat n deviation (Pn10	on used a ic gain sv B.2 = 2, 3	as the switching con- witching function by 3) is used.	7	Com- mand unit	1 to 250	Online	

Position Control Parameters

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn200	Position control	2	0	Not used.	0	(Do not change the setting.)	0				
	setting 1		1	Not used.	0	(Do not change the setting.)	0				
			2	Not used.	1	(Do not change the setting.)	1				Fixed setting: 1
			3	Not used.	0	(Do not change the setting.)	0				
Pn201	Encoder divider rate	2	Sets the Drive.	e number of outp	ut pulses	from the Servo	1000	Pulses /rota- tion	16 to 16384	Offline	
Pn202	Electronic gear ratio G1 (numera- tor)	2	Sets the vomoto 0.01 ≤ 0	e pulse rate for th r travel distance. G1/G2 ≤ 100	ie comma	and pulses and Ser-	4		1 to 65535	Offline	
Pn203	Electronic gear ratio G2 (denomi- nator)	2					1		1 to 65535	Offline	
Pn204	Not used.		(Do not	change the setting	ng.)		0				

Section 4-5

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn205	Absolute encoder multi-turn limit set- ting	2	Sets the vomoto	e limit to the rotat r with an absolute	ion spee e encode	d when using a Ser- r.	65535	Rota- tions	0 to 65535	Offline	
Pn206	Number of fully- closed encoder pulses	2	Sets the each m	e number of fully- otor rotation.	closed e	ncoder pulses for	16384	Pulses /rota- tion	513 to 32768	Offline	
Pn207	Position control	2	0	Not used.	0	(Do not change the setting.)	0				
	setting 2		1	Not used.	1	(Do not change the setting.)	1				Fixed setting: 1
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				
Pn208	Not used.		(Do not	change the setti	ng.)		0				
Pn212	Not used.		(Do not	change the setti	ng.)		2048				
Pn217	Not used.		(Do not	change the setti	ng.)		1				
Pn218	Position control	2	0	Not used.	0	(Do not change the setting.)	0				
	setting 3		1	Not used.	1	(Do not change the setting.)	0				
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				

Speed Control Parameters

Param- eter No.	Parameter name	Param- eter size	Contents	Default setting	Unit	Setting range	Enable setting	Details
Pn300	Not used.		(Do not change the setting.)	1000				
Pn301	Not used.		(Do not change the setting.)	100				
Pn302	Not used.		(Do not change the setting.)	200				
Pn303	Not used.		(Do not change the setting.)	300				
Pn304	Jog speed	2	Sets the rotation speed used during jog operation (executed from the Servo Drive's Parameter Unit).	500	r/min	0 to 10000	Online	
Pn305	Soft start acceleration time	2	Sets acceleration time during speed control soft start.	0	ms	0 to 10000	Online	
Pn306	Soft start deceleration time	2	Sets deceleration time during speed control soft start.	0	ms	0 to 10000	Online	
Pn307	Not used.		(Do not change the setting.)	40				
Pn308	Speed feedback filter time constant	2	Sets filter time constant for speed feed- back.	0	×0.01 ms	0 to 65535	Online	
Pn309	Not used.		(Do not change the setting.)	60				

Torque Control (Torque Limit) Parameters

Param- Parame eter No. ter nam		Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn400	Not used.		(Do not	not change the setting.)			30				
Pn401	Torque command filter time constant	2	Sets the comma	ts the filter time constant for the internal torque nmand.			40	×0.01 ms	0 to 65535	Online	
Pn402	Forward torque limit	2	Forward ratio).	prward rotation output torque limit (rated torque tio).			350	%	0 to 800	Online	

Section 4-5

Param-	Parame-	Param-	Digit	Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn403	Reverse torque limit	2	Reverse ratio).	e rotation output	torque lin	nit (rated torque	350	%	0 to 800	Online	
Pn404	Forward rotation external current limit	2	Output rent lim	torque limit durin it (rated torque ra	g input of atio).	f forward rotation cur-	100	%	0 to 800	Online	
Pn405	Reverse rotation external current limit	2	Output rent lim	torque limit durin it (rated torque ra	g input ol atio).	f reverse rotation cur-	100	%	0 to 800	Online	
Pn406	Emer- gency stop torque	2	Deceler torque	ration torque whe ratio).	en an erro	or occurs (rated	350	%	0 to 800	Online	
Pn407	Speed limit	2	Sets the	s the speed limit in torque control mode. Selects notch 0 Notch filter 1 not				r/min	0 to 10000	Online	
Pn408	Torque command	2	0	Selects notch filter 1 function	0	Notch filter 1 not used.	0			Online	
	setting				1	Notch filter 1 used for torque com- mands.					
			1	Not used.	0	(Do not change the setting.)	0				
			2	Selects notch filter 2 function	0	Notch filter 2 not used.	0			Online	
					1	Notch filter 2 used for torque com- mands.					
			3	Not used.	0	(Do not change the setting.)	0				
Pn409	Notch fil- ter 1 fre- quency	2	Sets no	otch filter 1 freque	ency for to	orque command.	2000	Hz	50 to 2000	Online	
Pn40A	Notch fil- ter 1 Q value	2	Sets Q	Sets Q value of notch filter 1.			70	×0.01	50 to 400	Online	
Pn40B	Notch fil- ter 2 fre- quency	2	Sets no	Sets notch filter 2 frequency for torque command.			2000	Hz	50 to 2000	Online	
Pn40C	Notch fil- ter 2 Q value	2	Sets Q	ets Q value of notch filter 2.			70	×0.01	50 to 400	Online	

I/O and Status Parameters

Param- Parame- Para eter No. ter name ete siz		Param-	Contents				Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn500	Position- ing com- pletion range 1	2	Sets the	e width of the pos	sitioning o	completed range.	3	Com- mand unit	0 to 250	Online	
Pn501	Not used.		(Do not	change the setting	ng.)		10				
Pn502	Rotation speed for motor rota- tion detec- tion	2	Sets the Speed	e number of rotat Flag.	letecting the Zero	20	r/min	1 to 10000	Online		
Pn503	Speed conformity signal out- put width	2	Sets the s	ets the allowable fluctuation range (rotation speed) r the Speed Conformity Flag.			10	r/min	0 to 100	Online	
Pn504	Position- ing com- pletion range 2	2	Sets the Flag.	e proximity range	for the P	ositioning Proximity	3	Com- mand unit	1 to 250	Online	

Param-	Parame-	Param-	Contents Digit Name Set- Explanation			Default	Unit	Setting	Enable	Details	
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn505	Deviation counter overflow level	2	Sets the alarm.	e detection level f	or the de	viation counter over	1024	×256 pulses	1 to 32767	Online	
Pn506	Brake tim- ing 1	2	Sets the motor to	ts the delay from the brake command to the Servitor turning OFF.				×10 ms	0 to 50	Online	
Pn507	Brake command speed	2	Sets the mand.	ets the rotation speed for outputting the brake com- and.				r/min	0 to 10000	Online	
Pn508	Brake tim- ing 2	2	Sets the the brak	ets the delay from the Servomotor turning OFF to brake command output.			50	×10 ms	10 to 100	Online	
Pn509	Momen- tary hold time	2	Sets the when a	Sets the time during which alarm detection is disable when a power failure occurs.			20	ms	20 to 1000	Online	

Param-	Parame-	Param- eter		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn50A	Input sig- nal selec- tion 1	2	0	Not used.	1	(Do not change the setting.)	1				Fixed setting: 1
			1	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8
			2	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8
			3	POT (forward drive prohibited input) signal	0	Allocated to CN1, pin 40: Valid for low input	8			Offline	Stan- dard setting:
				allocation	1	Allocated to CN1, pin 41: Valid for low input					2
					2	Allocated to CN1, pin 42: Valid for low input					
					3	Allocated to CN1, pin 43: Valid for low input					
					4	Allocated to CN1, pin 44: Valid for low input					
					5	Allocated to CN1, pin 45: Valid for low input					
					6	Allocated to CN1, pin 46: Valid for low input					
					7	Always valid.					
					8	Always invalid.					
					9	Allocated to CN1, pin 40: Valid for high input					
					A	Allocated to CN1, pin 41: Valid for high input					
					В	Allocated to CN1, pin 42: Valid for high input					
					С	Allocated to CN1, pin 43: Valid for high input					
					D	Allocated to CN1, pin 44: Valid for high input					
					E	E	Allocated to CN1, pin 45: Valid for high input	N1, r			
					F	Allocated to CN1, pin 46: Valid for high input					

Section 4-5

Param- Parame- eter No. ter name	Param-	- Contents				Default	Unit	Setting	Enable	Details		
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting		
Pn50B	Input sig- nal selec- tion 2	2	0	NOT (reverse drive prohibited input) signal input terminal allocation	0 to F	Same as Pn50A.3	8			Offline	Stan- dard setting: 3	
			1	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8	
			2	PCL (forward rotation cur- rent limit) sig- nal input terminal alloca- tion	0 to F	Same as Pn50A.3	5			Offline	Stan- dard setting: 8	
			3	NCL (reverse rotation cur- rent limit) sig- nal input terminal alloca- tion	0 to F	Same as Pn50A.3	6			Offline	Stan- dard setting: 8	
Pn50C	Input sig- nal selec- tion 3	2	0	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8	
			1	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8	
		2	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8		
			3	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8	
Pn50D	Input sig- nal selec- tion 4	2	0	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8	
			1	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8	
			2	Not used.	8	(Do not change the setting.)	8					
			3	Not used.	8	(Do not change the setting.)	8					
Pn50E	Output sig-	2	0	INP1 (position-	0	Not used.	1			Offline	Stan-	
	tion 1			1) signal output terminal alloca-	1	Allocated to CN1, pins 25, 26	_				setting: 1	
				tion	2	Allocated to CN1, pins 27, 28	_					
					3	Allocated to CN1, pins 29, 30						
				1	VCMP (speed conformity) signal output terminal alloca- tion	0 to 3	Same as Pn50E.0	1			Offline	Stan- dard setting: 0
			2	TGON (Servo- motor rotation detection) sig- nal output ter- minal allocation	0 to 3	Same as Pn50E.0	2			Offline	Stan- dard setting: 0	
			3	READY (Ser- vomotor war- mup complete) signal output terminal alloca- tion	0 to 3	Same as Pn50E.0	3			Offline	Stan- dard setting: 3	

Param-	Parame-	Param- eter		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn50F	Output sig- nal selec- tion 2	2	0	CLIMT (cur- rent limit detec- tion) signal output termi- nal allocation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
			1	VLIMT (speed limit detection) signal output terminal alloca- tion	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
			2	BKIR (brake interlock) sig- nal output ter- minal allocation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 2
			3	WARN (warn- ing) signal out- put terminal allocation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
Pn510	Output sig- nal selec- tion 3	2	0	INP2 (position- ing completed 2) signal output terminal alloca- tion	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
			1	Not used.	0	(Do not change the setting.)	0				
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				

Param-	Parame-	Param-	m- Contents				Default Uni setting	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn511	Input sig- nal selec- tion 5	2	0	DEC signal input terminal allocation	0	Allocated to CN1, pin 40: Valid for low input	8			Offline	Stan- dard setting:
					1	Allocated to CN1, pin 41: Valid for low input					1
					2	Allocated to CN1, pin 42: Valid for low input					
					3	Allocated to CN1, pin 43: Valid for low input					
				4 Allocated to CN1 pin 44: Valid for lo input	Allocated to CN1, pin 44: Valid for low input	N					
					5	Allocated to CN1, pin 45: Valid for low input					
			6 7 8		6	Allocated to CN1, pin 46: Valid for low input					
				7	Always enabled.						
				8	Always disabled.						
					9	Allocated to CN1, pin 40: Valid for high input					
					A	Allocated to CN1, pin 41: Valid for high input					
					В	Allocated to CN1, pin 42: Valid for high input					
		C Allocated to CN1, pin 43: Valid for high input D Allocated to CN1, pin 44: Valid for high input E Allocated to CN1, pin 45: Valid for high input									
					F	Allocated to CN1, pin 46: Valid for high input					

Param-	Parame-	Param-	- Contents			Default setting	Unit	Setting	Enable	Details	
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn511	Input sig-	2	1	EXT1 signal	0 to 3	Always disabled.	8			Offline	Stan-
	tion 5			allocation	4	Allocated to CN1, pin 44: Valid for low input					dard setting: 4
					5	Allocated to CN1, pin 45: Valid for low input					
					6	Allocated to CN1, pin 46: Valid for low input					
					7	Always enabled.					
					8	Always disabled.					
					9 to C	Always disabled.					
					D	Allocated to CN1, pin 44: Valid for high input					
					E	Allocated to CN1, pin 45: Valid for high input					
					F	Allocated to CN1, pin 46: Valid for high input					
			2	EXT2 signal input terminal allocation	0 to F	Same as Pn511.1	8			Offline	Stan- dard setting: 5
			3	EXT3 signal input terminal allocation	0 to F	Same as Pn511.1	8			Offline	Stan- dard setting: 6
Pn512	Output sig-	2	0	Output signal	0	Not reversed.	0			Offline	
	nal reverse			reverse for CN1 pins 25, 26	1	Reversed.					
			1	Output signal reverse for CN1 pins 27, 28	0, 1	Same as above	0			Offline	
			2	Output signal reverse for CN1 pins 29, 30	0, 1	Same as above	0			Offline	
			3	Not used.	0	(Do not change the setting.)	0				
Pn513	Input sig- nal selec-	2	0	Not used.	8	(Do not change the setting.)	8				
	tion 6		1	Not used.	8	(Do not change the setting.)	8				
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				
Pn51A	Motor-load deviation over level	2	Sets the fully-clo	e allowable range sed encoders an	e for the r d semi-c	number of pulses for losed encoders.	0	Pulse	0 to 32767	Online	
Pn51B	Not used.		(Do not	change the setti	ng.)		100				
Pn51C	Not used.		(Do not	change the setti	ng.)		450				
Pn51E	Deviation counter overflow warning level	2	Sets the flow wa	e detection level t rning.	for the de	eviation counter over-	0	%	0 to 100	Online	

Other Parameters

Param- eter No.	Parameter name	Param- eter size	Contents	Default setting	Unit	Setting range	Enable setting	Details
Pn600	Regeneration resistor capacity	2	Setting for the regeneration resistance load ratio monitoring calculations.	0	×10 W	From 0 (Varies by Unit.)	Online	
Pn601	Not used.		(Do not change the setting.)	0				

Control Function Parameters

Param-	Parame-	Param- Contents Default Unit Set		Setting	Enable	Details						
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting		
Pn800	Communi- cations control	2	0	MECHA- TROLINK-II communica- tions check mask	0	Detects both com- munications errors (A.E6) and syn- chronization errors (A.E5).	0			Online	Always set to 0.	
					1	Ignores communi- cations errors (A.E6).						
					2	Ignores synchroni- zation errors (A.E5).						
					3	Ignores both com- munications errors (A.E6) and syn- chronization errors (A.E5).						
			1	Warning check mask	0	Detects parameter setting warnings (A.94), MECHA- TROLINK com- mand warnings (A.95), and com- munications errors (A.96).	4			Online	Always set to 4 or 0.	
					1	Ignores parame- ters setting warn- ings (A.94).						
					2	Ignores MECHA- TROLINK-II com- mand warnings (A.95).						
					3	Ignores both parameter setting warnings (A.94) and MECHA- TROLINK-II com- mand warnings (A.95).						
					4	Ignores communi- cations errors (A.96).						
					5	Ignores both parameter setting warnings (A.94) and communica- tions errors (A.96).						
		2 Communitions error count at s transmissi				6	Ignores both MECHATROLINK- II command warn- ings (A.95) and communications errors (A.96).					
				7	7	Ignores parameter setting warnings (A.94), MECHA- TROLINK-II com- mand warnings (A.95), and com- munications errors (A.96).	er - rs					
	2		Communica- tions error count at single transmission	0 to F	Detects a commu- nications error (A.E6) when the number of errors specified by the set value + 2 have occurred continu- ously.	u- 0 s set J-		Online				
			3	Not used.	0	(Do not change the setting.)	0					

Param- eter No.	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn801	Function selection	2	0	Software limit function	0	Software limit enabled.	0			Online	
	applica- tion (soft- ware				1	Forward software limit disabled.					
	limits)	imits)		3	2	Reverse software limit disabled.					
					3	Software limit dis- abled in both direc- tions.					
			1	Not used.	0	(Do not change the setting.)	0				
			2	Software limit check using references	0	No software limit check using references.	0			Online	Always set to 0.
					1	Software limit check using refer- ences.					
			3	Not used.	0	(Do not change the setting.)	0				

Param- eter No.	Parameter name	Param- eter size	Contents	Default set- ting	Unit	Setting range	Enable setting	Details
Pn802	Not used.		(Do not change the set- ting.)	0000				
Pn803	Zero point width	2	Sets the detection range for the Origin Stop Flag.	10	Command unit	0 to 250	Online	
Pn804	Forward soft- ware limit	4	Sets the forward soft- ware limit.	819,191,808	Command unit	-1,073,741,823 to 1,073,741,823	Online	
Pn806	Reverse soft- ware limit	4	Sets the reverse soft- ware limit.	-819,191,808	Command unit	-1,073,741,823 to 1,073,741,823	Online	
Pn808	Absolute encoder zero point position offset	4	Sets the offset for the mechanical origin from the absolute encoder's absolute value data.	0	Command unit	-1,073,741,823 to 1,073,741,823	Offline	
Pn80A	First-step linear acceleration constant	2	Sets the first-step acceleration speed for the acceleration/deceleration/deceleration curve used in position control.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80B	Second-step lin- ear acceleration constant	2	Sets the second-step acceleration speed for the acceleration/deceler- ation curve used in posi- tion control.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80C	Acceleration constant switch- ing speed	2	Sets the speed for switching between first- step and second-step acceleration for the acceleration/decelera- tion curve used in posi- tion control.	0	×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).

Param- eter No.	Parameter name	Param- eter size	Contents	Default set- ting	Unit	Setting range	Enable setting	Details
Pn80D	First-step linear deceleration constant	2	Sets the first-step decel- eration speed for the acceleration/decelera- tion curve used in posi- tion control.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80E	Second-step lin- ear deceleration constant	2	Sets the second-step deceleration speed for the acceleration/deceler- ation curve used in posi- tion control.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80F	Deceleration constant switch- ing speed	2	Sets the speed for switching between first- step and second-step deceleration for the acceleration/decelera- tion curve used in posi- tion control.	0	×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn810	Exponential acceleration/ deceleration bias	2	Sets the exponential acceleration/decelera- tion bias speed for the acceleration/decelera- tion curve used in posi- tion control.	0	Command units/s	0 to 32767	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn811	Exponential acceleration/ deceleration time constant	2	Sets exponential accel- eration/deceleration time constant for the acceler- ation/deceleration curve used in position control.	0	×0.1 ms	0 to 5100	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn812	Movement aver- age time	2	Sets the average move- ment time for when S- curve acceleration/ deceleration is used, and an average movement filter is used for the posi- tion command filter.	0	×0.1 ms	0 to 5100	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn813	Not used.		(Do not change the set- ting.)	0010				
Pn814	Final travel dis- tance for exter- nal positioning	4	Sets the final travel dis- tance for external posi- tioning when executing INTERRUPT FEEDING using direct operation.	100	Command unit	-1,073,741,823 to 1,073,741,823	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).

Param-	Parame-	Param-	Contents				Default	Unit	Setting range	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting			setting	
Pn816	Zero point return mode set- ting	2	0	Zero point return direc- tion	0 1	Forward Reverse	0			Online	Set the same direc- tion as the origin search direction set in the Axis Parameters. Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
			1 to 3	Not used.	0	(Do not change the setting.)	0				
Pn817	Zero point return approach speed 1	2	Sets the search signal h	e origin (zero p speed used af las been detec	point) inp ter the or ted in an	ut signal igin proximity origin search.	50	×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn818	Zero point return approach speed 2	2	Sets the distance gin inpu gin sea	e origin (zero p e positioning s ut signal has be rch.	ooint) retu peed use een deteo	urn final travel ad after the ori- cted in an ori-	5	×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn819	Final travel dis- tance to return to zero point	4	Sets the used af detecte	e amount of coi ter the origin ir d in an origin s	mpensati nput sign search.	on positioning al has been	100	Com- mand unit	-1,073,741,823 to 1,073,741,823	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn81B	Backlash compen- sation amount	2	Sets the	e amount of ba	icklash c	ompensation.	0	× 0.1 com- mand unit	-32,768 to 32,767	Online	
Pn81C	Not used.		(Do not	change the se	etting.)		0000				
Pn81D	Compen- sation function selection	2	0	Backlash compensa- tion selec- tion	0	Compen- sates in for- ward direction. Compen- sates in reverse direction.	0			Offline	
			1 to 3	NOT USED.	0	(Do not change the setting.)	U				
Pn81E to Pn823	Not used.		(Do not	change the se	etting.)		0000				

4-5-4 W-series Servo Drive (R88D-WN -ML2 with Built-in MECHATROLINK-II Communications)

The Servo Parameters listed here can be used when the Position Control Unit is used with a W-series Servo Drive with Built-in MECHATROLINK-II Communications (R88D-WN□-ML2).

For further details on each of the parameters, refer to the user's manual for W-series Servo Drives.

The timing for Servo Parameters to be enabled are classified into the following two types.

Online (online parameter): Changed settings are enabled immediately after Servo Parameters have been written.

Offline (offline parameters): Changed settings are enabled when the Servo Drive power is cycled or DEVICE SETUP is executed.

The *Details* column indicates whether the following conditions apply to the corresponding parameter.

- Online parameters that can be changed when the axis is stopped (Busy Flag = 0) only.
- Parameters with standard set values when using MECHATROLINK are indicated in the *Details* column as "standard" with the set value. The standard settings are basic settings required when using the Position Control Unit. This manual describes operations assuming standard settings are being used.

For details on fixed and standard settings, refer to 6-4 Standard Settings for Servo Drives Using MECHATROLINK.

Note Make sure that the equipment will not be adversely affected before changing the Servo Parameters (WRITE SERVO PARAMETER, SAVE SERVO PARAMETER). Refer to the Servo Drive's user's manual and always check the effect of changing the settings before changing the Servo Parameters.

Param-	Parame- ter name	Param- eter size	Contents				Default	Unit	Setting	Enable	Details
eter No.			Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn000	Function selection basic	2	0	Reverse rota- tion	0	CCW direction is taken for positive command	0			Offline	
	switches	95			1	CW direction is taken for positive command					
					2 to 3	Not used.					
			1	Not used.	0	(Do not change the setting.)	0				
			2	Unit No. setting	0 to F	Servo Drive com- munications unit number setting (necessary for mul- tiple Servo Drive connections when using personal computer monitor- ing software)	0			Offline	
			3	Not used.	0	(Do not change the setting.)	0				

Function Selection Parameters

Param-	Parame-	Param-	Contents				Default I	Unit	Jnit Setting	Enable	Details		
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	Details		
Pn001	Function selection applica- tion switches 1	2	0	Stop selection if an alarm occurs when Servomotor is OFF	0	Servomotor stopped by dynamic brake.	2			Offline			
					1	Dynamic brake OFF after Servo- motor stopped							
					2	Servomotor stopped with free run							
			1	Stop selection when drive prohibited is input	0	Stop according to Pn001.0 setting (release Servomo- tor after stopping)	0			Offline			
					1 S	Stop Servomotor using torque set in Pn406, and lock Servomotor after stopping							
					2	Stop Servomotor using torque set in Pn406, and release Servomotor after stopping							
			2	AC/DC power input selection	0	AC power supply: AC power supplied from L1, L2, (L3) terminals	0 -			Offline			
					1	DC power supply: DC power from +, - (2) terminals							
			3	Not used.	0	(Do not change the setting.)	0						
Pn002	Function selection applica- tion switches 2	tion 2 tion ca- hes 2	0	Torque com- mand input	0	Do not use option command value.	0			Offline			
				change (dur- ing speed con- trol)	_ 1	Use option com- mand value 1 as the torque limit value.							
					2	Use option com- mand value 1 as the torque feed for- ward command value.							
					3	Use option com- mand value 1 or 2 as the torque limit value, according to the forward and reverse torque lim- its that are speci- fied.							
			1	Speed com- mand input	0	Do not use option command value.	0			Offline			
				change (dur- ing torque con- trol)	1	Use option com- mand value 1 as the speed limit value.							
			2	Operation switch when	0	Use as absolute encoder.	0			Offline			
				encoder	1	Use as incremen- tal encoder.							
			3	Not used.	0	(Do not change the setting.)	0						
Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Setting Enable I range setting	Details		
----------	-----------------------------------	--------------	--------------	--	--------------	--	---------	--	---------	-----------------------------------	---------	--	--
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting			
Pn004	Function selection	2	0	Not used.	0	(Do not change the setting.)	0						
	tion switches 4		1	Not used.	1	(Do not change the setting.)	1						
			2	Not used.	1	(Do not change the setting.)	1						
			3	Not used.	0	(Do not change the setting.)	0						
Pn006	Function selection applica-	2	0 to 1	Analog moni- tor 1 (AM) sig- nal selection	00	Servomotor rota- tion speed: 1 V/ 1000 r/min	02			Online			
	tion switches 6				01	Speed command: 1 V/1000 r/min							
					02	Torque command: gravity compensa- tion torque (Pn422) (1 V per 100%)							
					03	Position deviation: 0.05 V/1 command unit							
					04	Position amplifier deviation (after electronic gear) (0.05 V per encoder pulse unit)							
					05	Position command speed (1 V/1,000 r/min)							
					06	Not used.							
					07	Not used.							
							08	Positioning com- pleted command (Positioning com- pleted: 5 V; posi- tioning not completed: 0 V					
					09	Speed feed forward (1 V/1,000 r/min)							
					0A	Torque feed for- ward (1 V per 100%)							
					0B to 1F	Not used.							
			2	Analog moni-	0	1x	0			Online			
				tor 1 signal multiplier	1	10x							
				selection	2	100x							
					3	1/10x							
					4	1/100x							
			3	Not used.	0	(Do not change the setting.)	0						

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn007	Function selection applica-	2	0 to 1	Analog moni- tor 2 (NM) sig- nal selection	00	Servomotor rota- tion speed: 1V/ 1000 r/min	00			Online	
	switches 7				01	Speed command: 1 V/1000 r/min					
					02	Torque command: gravity compensa- tion torque (Pn422) (1 V per 100%)					
					03	Position deviation: 0.05 V/1 command unit					
					04	Position amplifier deviation (after electronic gear) (0.05 V per encoder pulse unit)					
					05	Position command speed (1 V/1,000 r/min)					
					06	Not used.					
					07	Not used.					
					08	Positioning com- pleted command (Positioning com- pleted: 5 V; posi- tioning not completed: 0 V					
					09	Speed feed forward (1 V/1,000 r/min)					
					0A	Torque feed for- ward (1 V per 100%)					
					0B to 1F	Not used.					
			2	Analog moni-	0	1x	0			Online	
				tor 2 signal multiplier	1	10x					
				selection	2	100x					
					3	1/10x					
					4	1/100x					
			3	Not used.	0	(Do not change the setting.)	0				
Pn008	Pn008 Function selection applica-	2	0	Lowered bat- tery voltage alarm/warning	0	Regard battery voltage drop as alarm (A.830).	0			Offline	
applica tion switche	tion switches 8			selection	1	Regard battery voltage drop as warning (A.930).					
			1	Not used.	0	(Do not change the setting.)	e the 0				
			2	Warning detec-	0	Warnings detected.	0			Offline	
				tion selection	1	Warnings not detected.					
		3	Not used.	4	(Do not change the setting.)	4					

Servo Gain Parameters

Param-	Parameter	Param-		C	ontents		Default	Unit	Setting	Enable	Details
eter No.	name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn100	Speed loop gain	2	Adjusts	speed loop resp	onse.		800	× 0.1 Hz	10 to 20000	Online	
Pn101	Speed loop integration constant	2	Speed	loop integral time	e constan	t	2000	×0.01 ms	15 to 51200	Online	
Pn102	Position loop gain	2	Adjusts	position loop res	sponse.		400	× 0.1/s	10 to 20000	Online	
Pn103	Inertia ratio	2	Set usi tia and	ng the ratio betweet the Servomotor r	een the r otor iner	nachine system iner- tia.	300	%	0 to 20000	Online	
Pn104	Speed loop gain 2	2	Adjusts gain sw	speed loop resp ritching input).	onse (er	abled by automatic	800	× 0.1 Hz	10 to 20000	Online	
Pn105	Speed loop integration constant 2	2	Speed matic g	loop integral time ain switching inp	e constan ut).	t (enabled by auto-	2000	×0.01 ms	15 to 51200	Online	
Pn106	Position loop gain 2	2	Adjusts gain sw	position loop res itching input).	sponse (e	enabled by automatic	400	× 0.1/s	10 to 20000	Online	
Pn107	Bias rota- tional speed	2	Sets po	sition control bia	S.		0	r/min	0 to 450	Online	
Pn108	Bias addi- tion band	2	Sets the deviation	e position control on counter pulse	bias ope width.	eration start using	7	Com- mand unit	0 to 250	Online	
Pn109	Feed-for- ward amount	2	Position	n control feed-form	ward con	npensation value	0	%	0 to 100	Online	
Pn10A	Feed-for- ward com- mand filter	2	Sets po	sition control fee	d-forware	d command filter.	0	×0.01 ms	0 to 6400	Online	
Pn10B	0B Speed con- trol setting 2	2 0	0	P control switching con- ditions	0	Sets internal torque command value conditions (Pn10C).	4			Online	
					1	Sets speed com- mand value condi- tions (Pn10d).					
					2	Sets acceleration command value conditions (Pn10E).					
					3	Sets deviation pulse value condi- tions (Pn10F).					
					4	No P control switching function					
			1	Speed control	0	PI control	0			Offline	
				loop switching	1	IP control					
					2 to 3	Not used.					
			2	Position loop control method	0	Standard position control	0			Offline	
					1	Less deviation con- trol					
					2 to 3	Not used.					
			3	Not used.	0	(Do not change the setting.)	0				
Pn10C	P control switching (torque command)	2	Sets lev trol to F	vel of torque com ^o control.	200	%	0 to 800	Online			
Pn10D	P control switching (speed command)	2	Sets lev to P co	vel of speed comr ntrol.	switch from PI control	0	r/min	0 to 10000	Online		

Param-	Parameter	Param-		Co	ontents		Default	Unit	Setting	Enable D setting	Details
eter No.	name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn10E	P control switching (accelera- tion com- mand)	2	Sets lev control	vel of acceleration to P control.	ו comma	ind to switch from PI	0	r/min/s	0 to 30000	Online	
Pn10F	P control switching (deviation pulse)	2	Sets lev to P co	<i>i</i> el of deviation puntrol.	ilses to s	witch from PI control	10	Com- mand unit	0 to 10000	Online	
Pn110	Normal autotuning	2	0	Not used.	2	(Do not change the setting.)	2			Offline	
	switches		1	Speed feed-	0	ON	1			Online	
				back compen-	1	OFF	1				
				selection	2 to 3	Not used.	1				
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				
Pn111	Speed feedback compensat- ing gain	2	Adjusts	speed loop feed	back gaiı	n.	100	%	1 to 500	Online	
Pn119	Not used.		(Do not	change the setting	ng.)		500			[T
Pn11A	Not used.		(Do not	change the setting	ng.)		1000			<u> </u>	
Pn11E	Not used.		(Do not	change the setti	ng.)		1000				
Pn11F	Position integral time con- stant	2	Positior	ו loop integral tim	ie consta	ınt	0	× 0.1 ms	0 to 50000	Online	
Pn12B	Not used.		(Do not	change the setting	ng.)		400				
Pn12C	Not used.		(Do not	change the setting	ng.)		2000				
Pn12D	Not used.		(Do not	change the setting	ng.)		400				
Pn12E	Not used.		(Do not	change the setting	ng.)		400				
Pn12F	Not used.		(Do not	change the setting	ng.)		2000				
Pn130	Not used.		(Do not	change the setting	ng.)		400	<u> </u>		Γ	Γ
Pn131	Gain switching time 1	2	Switchi	ng time from No.	1 gain to) No. 2 gain	0	ms	0 to 65535	Online	
Pn132	Gain switching time 2	2	Switchi	ng time from No.	2 gain to	No. 1 gain	0	ms	0 to 65535	Online	
Pn135	Gain switching waiting time 1	2	The tim isfied u gain be	e from when gair ntil switching from gins.	ו switchir n the No.	ng condition A is sat- 1 gain to the No. 2	0	ms	0 to 65535	Online	
Pn136	Gain switching waiting time 2	2	The tim isfied u gain be	e from when gair ntil switching fron gins.	n switchir n the No.	ng condition B is sat- 2 gain to the No. 1	0	ms	0 to 65535	Online	

Param-	Parameter	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn139	Automatic gain changeover related	2	0	Gain switching selection switch	0	Manual gain switching (auto- matic gain switch- ing not used)	0			Offline	
	SWITCHES I				1	Automatic switch- ing pattern 1 Automatic switch- ing from No. 1 gain to No. 2 gain when gain switching con- dition A is satisfied. Automatic switch- ing from No. 2 gain to No. 1 gain when gain switching con- dition B is satisfied.					
					2 to 4	Not used.					
			1	Gain switching condition A	0	Positioning com- pleted output 1 (INP1) ON	0			Offline	
					1	Positioning com- pleted output 1 (INP1) OFF					
					2	Positioning com- pleted output 2 (INP2) ON					
					3	Positioning com- pleted output 2 (INP2) OFF					
					4	The position com- mand filter output is 0, and also the position command input is 0.					
					5	The position com- mand input is not 0.					
			2	Gain switching condition B	0 to 5	Same as Pn139.1	0			Offline	
			3	Not used.	0	(Do not change the setting.)	0				
Pn144	Not used.		(Do not	change the setti	ng.)		1000				
Pn150	Predictive control selection	2	0	Predictive con- trol selection	0	Predictive control not used.	0			Offline	
	switches				1	Predictive control used.	_				
					2	Not used.					
			1	Predictive con- trol type	0	Predictive control for tracking	1			Offline	
					1	Predictive control for positioning					
			2	Not used.	2	(Do not change the setting.)	2				
			3	Not used.	0	(Do not change the setting.)	0				
Pn151	Predictive control accelera- tion/decel- eration gain	2	Adjusts predicti	acceleration and ve control.	l decelera	ation response for	100	%	0 to 300	Online	
Pn152	Predictive control weighting ratio	2	Adjusts	position deviatio	n for pree	dictive control.	100	%	0 to 300	Online	
Pn1A0	Servo rigidity	2	Adjusts	the Servo rigidity	/ for the l	No. 1 gain.	60	%	1 to 500	Online	
Pn1A1	Servo rigidity 2	2	Adjusts	the Servo rigidity	/ for the I	No. 2 gain.	60	%	1 to 500	Online	

Section 4-5

Param-	Parameter	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn1A2	Speed feedback filter time constant	2	Sets the back.	e filter time const	ant for N	o. 1 gain speed feed-	72	× 0.01 ms	30 to 3200	Online	
Pn1A3	Speed feedback filter time constant 2	2	Sets the back.	e filter time const	ant for N	o. 2 gain speed feed-	72	× 0.01 ms	30 to 3200	Online	
Pn1A4	Torque command filter time constant 2	2	Sets the	e filter time const	ant for th	e torque command.	36	× 0.01 ms	0 to 2500	Online	
Pn1A7	Utility con- trol switches	2	0	Integral com- pensation pro- cessing	0	Integral compensa- tion processing not executed.	1			Online	
					1	Integral compensa- tion processing executed.					
					2	Integral compensa- tion is executed for No. 1 gain and not for No. 2 gain for less-deviation gain switching.					
					3	Integral compensa- tion is executed for No. 2 gain and not for No. 1 gain for less-deviation gain switching.					
			1	Not used.	2	(Do not change the setting.)	2				
			2	Not used.	1	(Do not change the setting.)	1				
			3	Not used.	1	(Do not change the setting.)	1				
Pn1A9	Utility inte- gral gain	2	Adjusts	the auxiliary inte	egral resp	onsive.	37	Hz	0 to 500	Online	
Pn1AA	Position propor- tional gain	2	Adjusts	ists the position proportional responsive.				Hz	0 to 500	Online	
Pn1AB	Speed inte- gral gain	2	Adjusts	djusts the speed integral responsive.				Hz	0 to 500	Online	
Pn1AC	Speed pro- portional gain	2	Adjusts	djusts the speed proportional responsive.				Hz	0 to 2000	Online	
Pn1B5	Not used.		(Do not	change the setti	ng.)		150				

Position Control Parameters

Param- eter No. t	Parame-	Param- eter		c	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn200	Not used.	2	0	Not used.	0	(Do not change the setting.)	0				
			1	Not used.	0	(Do not change the setting.)	0				
			2	Not used.	1	(Do not change the setting.)	1				
			3	Not used.	0	(Do not change the setting.)	0				
Pn205	Absolute encoder multi-turn limit set- ting	2	Sets the an abso	e multi-turn lim blute encoder is	it for whe s used.	n a Servomotor with	65535	Rota- tion	0 to 65535	Offline	

WWW.BSNEW.IR

Param-	Parame-	Param-		С	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn207	Position control	2	0	Not used.	0	(Do not change the setting.)	0				
	settings 2		1	Not used.	1	(Do not change the setting.)	1				
			2	Backlash	0	Disabled	0			Offline	
				tion selection	1	Compensates to forward rotation side.					
					2	Compensates to reverse rotation side.					
			3	INP 1 output timing	0	When the position deviation is below the INP1 range (Pn522).	0			Offline	
					1	When the position deviation is below the INP1 range (Pn522) and also the command after the position com- mand filter is 0.					
					2	When the absolute value for the posi- tion deviation is below the INP1 range (Pn522) and also the position command input is 0.					
Pn209	Not used.		(Do not	change the se	tting.)		0				
Pn20A	Not used.		(Do not	change the se	tting.)		32768				
Pn20E	Electronic gear ratio G1 (numera- tor)	4	Sets the Servorr 0.001 ≤	e pulse rate for notor movemen ⊆G1/G2 ≤ 1000	the com t distanc	mand pulses and e.	4		1 to 1073741824	Offline	
Pn210	Electronic gear ratio G2 (denomi- nator)	4					1		1 to 1073741824	Offline	
Pn212	Encoder divider rate	4	Sets the rotation	e number of ou	tput puls	es per Servomotor	1000	Pulses /rota- tion	16 to 1073741824	Offline	
Pn214	Backlash compen- sation amount	2	Sets me mechar shaft be	echanical syste nical gap betwe eing driven)	m backla en the d	ash amount (the rive shaft and the	0	Com- mand unit	-32767 to 32767	Online	
Pn215	Backlash compen- sation time constant	2	Sets the	e backlash com	pensatio	on time constant.	0	× 0.01 ms	0 to 65535	Online	
Pn216	Not used.		(Do not	change the se	tting.)		0				
Pn217	Not used.		(Do not	change the se	tting.)		0				
Pn281	Not used.		(Do not	change the se	tting.)		20				

Speed Control Parameters

Param-	Parame-	Param-		Contents		Default setting	Unit	Setting	Enable	Details	
eter No.	ter name	eter size	Digit No.	git Name o.		Explanation	setting		range	setting	
Pn300	Not used.		(Do not	change the settir	ng.)		600				
Pn301	Not used.		(Do not	change the settir	ng.)		100				
Pn302	Not used.		(Do not	change the settir		200					
Pn303	Not used.		(Do not	not change the setting.)			300				

Param-	Parame-	Param-		Co	ontents	DefaultUnitSettingEnableExplanationsettingrangesetting	Details				
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn304	Jog speed	2	Sets rot Drive's	tation speed durir personal compute	ng jog op er monito	eration (using Servo pring software)	500	r/min	0 to 10000	Online	
Pn305	Soft start accelera- tion time	2	Sets ac	celeration time d	uring spe	eed control soft start.	0	ms	0 to 10000	Online	
Pn306	Soft start decelera- tion time	2	Sets de	celeration time d	uring spe	eed control soft start.	0	ms	0 to 10000	Online	
Pn307	Not used.		(Do not	change the setting	ոց.)		40				
Pn308	Speed feedback filter time constant	2	Sets co	nstant during filte	er of spee	ed feedback.	0	× 0.01 ms	0 to 65535	Online	
Pn310 Vit de sw	Vibration detection	ion 2 ion ies	0	Vibration detection	0	Vibration detection not used.	0			Online	
	switches			selection	1	Gives warning (A.911) when vibra- tion is detected.				Online	
					2	Gives warning (A.520) when vibra- tion is detected.					
			1	Not used.		(Do not change the setting.)	0				
			2	Not used.		(Do not change the setting.)	0				
			3	Not used.		(Do not change the setting.)	0				
Pn311	Vibration detection sensitivity	2	Sets the	e vibration detect	ion sensi	itivity.	100	%	50 to 500	Online	
Pn312	Vibration detection level	2	Sets the	e vibration detect	ion level		50	r/min	0 to 500	Online	

Torque Control (Torque Limit) Parameters

Param-	Parame-	Param-		Co	ontents		Default settingUnit rangeSetting setting	Details			
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn400	Not used.		(Do not	change the settir	ng.)		30				
Pn401	1st step 1st torque command filter time constant	2	Sets the mands.	e filter time consta	ant for int	ternal torque com-	40	×0.01 ms	0 to 65535	Online	
Pn402	Forward torque limit	2	Forward ratio).	I rotation output t	orque lin	nit (rated torque	350	%	0 to 800	Online	
Pn403	Reverse torque limit	2	Reverse ratio).	e rotation output t	orque lin	nit (rated torque	350	%	0 to 800	Online	
Pn404	Forward rotation external current limit	2	Output t rent limi	orque limit durin t (rated torque ra	g input of tio)	forward rotation cur-	100	%	0 to 800	Online	
Pn405	Reverse rotation external current limit	2	Output f rent limi	orque limit during t (rated torque ra	g input of tio)	reverse rotation cur-	100	%	0 to 800	Online	
Pn406	Emer- gency stop torque	2	Deceler torque r	ation torque whe atio)	n an errc	or occurs (rated	350	%	0 to 800	Online	
Pn407	Speed limit	2	Sets the	e speed limit in to	rque con	trol mode.	3000	r/min	0 to 10000	Online	

WWW.BSNEW.IR

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn408	Torque command	2	0	Selects notch filter 1 function.	0	Notch filter 1 not used.	0			Online	
	setting				1	Notch filter 1 used for torque com- mands.					
			1	Not used.	0	(Do not change the setting.)	0				
			2	Selects notch filter 2 function.	0	Notch filter 2 not used.	0			Online	
					1	Notch filter 2 used for torque com- mands.					
			3	Not used.	0	(Do not change the setting.)	0				
Pn409	Notch fil- ter 1 fre- quency	2	Sets no	tch filter 1 freque	ency for t	orque command.	2000	Hz	50 to 2000	Online	
Pn40A	Notch fil- ter 1 Q value	2	Sets Q	value of notch filt	ter 1.		70	×0.01	50 to 1000	Online	
Pn40C	Notch fil- ter 2 fre- quency	2	Sets the mands.	e notch filter 2 fre	for torque com-	2000	Hz	50 to 2000	Online		
Pn40D	Notch fil- ter 2 Q value	2	Sets Q	ts Q value of notch filter 2. ts the filter frequency for internal torque com-				×0.01	50 to 1000	Online	
Pn40F	2nd step 2nd torque command filter fre- quency	2	Sets the mands.	e filter frequency	for intern	nal torque com-	2000	Hz	100 to 2000	Online	
Pn410	2nd step 2nd torque command filter Q value	2	Sets the	ets the torque command filter Q value.			70	×0.01	50 to 1000	Online	
Pn411	3rd step torque command filter time constant	2	Sets the mands.	ə filter time const	ant for ir	ternal torque com-	0	μs	0 to 65535	Online	
Pn412	1st step 2nd torque command filter time constant	2	Sets the torque	e filter time const commands.	ant for N	o. 2 gain internal	100	× 0.01 ms	0 to 65535	Online	
Pn413	Not used.		(Do not	change the setti	ng.)		100				
Pn414	Not used.		(Do not	change the setting	ng.)		100				
Pn420	Damping for vibra- tion sup- pression on stop- ping	2	Sets the	to not change the setting.) ets the vibration suppression value while stopped.			100	%	10 to 100	Online	
Pn421	Vibration suppres- sion start- ing time	2	Sets the become stoppin	ets the time from when the position command comes 0 until damping for vibration suppression of opping begins.		1000	ms	0 to 65535	Online		
Pn422	Gravity compen- sation torque	2	Sets the	the time from when the position command nes 0 until damping for vibration suppressio ing begins. the gravity compensation torque.			0	×0.01 %	-20000 to 20000	Online	
Pn456	Sweep torque command amplitude	2	Sets the	e sweep torque c	ommanc	l amplitude.	15	%	1 to 800	Online	

I/O and Status Parameters

Param-	Parame-	Param-		C	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn501	Not used.		(Do not	change the se	tting.)		10				
Pn502	Rotation speed for motor rota- tion detec- tion	2	Sets the rotation	number of rot detection outp	ations for out (TGON	r the Servomotor N).	20	r/min	1 to 10000	Online	
Pn503	Speed conformity signal out- put width	2	Sets the s	allowable fluc speed conformi	tuation (r	number of rotations) (VCMP).	10	r/min	0 to 100	Online	
Pn506	Brake tim- ing 1	2	Sets the vomoto	e delay from the r turning OFF.	e brake c	command to the Ser-	0	×10 ms	0 to 50	Online	
Pn507	Brake command speed	2	Sets the comma	e number of rotand.	ations for	outputting the brake	100	r/min	0 to 10000	Online	
Pn508	Brake tim- ing 2	2	Sets the OFF to	e delay time fro the brake com	m the Se mand out	ervomotor turning	50	×10 ms	10 to 100	Online	
Pn509	Momen- tary hold time	2	Sets the abled w	e time during w hen a power fa	hich alar alure occ	m detection is dis- urs.	20	ms	20 to 1000	Online	

Param-	Parame-	Param-		C	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn50A	Input sig- nal selec-	2	0	Not used.	1	(Do not change the setting.)	1				
	tions 1		1	Not used.	8	(Do not change the setting.)	8				
			2	Not used.	8	(Do not change the setting.)	8				
			3	POT (forward drive prohib- ited input)	0	Allocated to CN1, pin 13: Valid for low input	1			Offline	Stan- dard setting:
				terminal allo- cation	1	Allocated to CN1, pin 7: Valid for low input					1
					2	Allocated to CN1, pin 8: Valid for low input					
					3	Allocated to CN1, pin 9: Valid for low input	-				
					4	Allocated to CN1, pin 10: Valid for low input					
					5	Allocated to CN1, pin 11: Valid for low input					
					6	Allocated to CN1, pin 12: Valid for low input					
					7	Always enabled.					
					8	Always disabled.					
					9	Allocated to CN1, pin 13: Valid for high input					
					A	Allocated to CN1, pin 7: Valid for high input					
					В	Allocated to CN1, pin 8: Valid for high input	_				
					С	Allocated to CN1, pin 9: Valid for high input					
					D	Allocated to CN1, pin 10: Valid for high input					
					E	Allocated to CN1, pin 11: Valid for high input					
					F	Allocated to CN1, pin 12: Valid for high input					
Pn50B	Input sig- nal selec- tions 2	2	0	NOT (reverse drive prohib- ited input) signal Input terminal allo- cation	0 to F	Same as Pn50A.3	2			Offline	Stan- dard setting: 2
			1	Not used.	8	(Do not change the setting.)	8				
			2	Not used.	8	(Do not change the setting.)	8				
			3	Not used.	8	(Do not change the setting.)	8				

Param-	Parame-	Param-		C	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn50C	Input sig- nal selec-	2	0	Not used.	8	(Do not change the setting.)	8				
	tions 3		1	Not used.	8	(Do not change the setting.)	8				
			2	Not used.	8	(Do not change the setting.)	8				
			3	Not used.	8	(Do not change the setting.)	8				
Pn50D	Input sig- nal selec-	2	0	Not used.	8	(Do not change the setting.)	8				
	tions 4		1	Not used.	8	(Do not change the setting.)	8				
			2	Not used.	8	(Do not change the setting.)	8				
			3	Not used.	8	(Do not change the setting.)	8				
Pn50E	Output sig-	2	0	INP1 (posi-	0	Not used.	0			Offline	Stan-
	nal selec- tions 1			pleted 1)	1	Allocated to CN1 pins 1, 2					dard setting: 0
				terminal allo- cation	2	Allocated to CN1 pins 23, 24	-				•
					3	Allocated to CN1 pins 25, 26					
			1	VCMP (speed con- formity) sig- nal output terminal allo- cation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
			2	TGON (ser- vomotor rota- tion detection) signal output terminal allo- cation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
			3	READY (servo ready) signal output terminal allo- cation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
Pn50F	Output sig- nal selec- tions 2	2	0	CLIMT (cur- rent limit detection) signal output terminal allo- cation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
			1	VLIMT (speed limit detection) signal output terminal allo- cation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
			2	BKIR (brake interlock) signal output terminal allo- cation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 1
			3	WARN (warning) signal output terminal allo- cation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0

Param-	Parame-	Param-		C	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn510	Output sig- nal selec- tions 3	2	0	INP2 (posi- tioning com- pleted 2) signal output terminal allo- cation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
			1	Not used.	0	(Do not change the setting.)	0				
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				
Pn511	Input sig- nal selec- tions 5	2	0	DEC signal input termi- nal allocation	0	Allocated to CN1, pin 13: Valid for low input	3			Offline	Stan- dard setting:
					1	Allocated to CN1, pin 7: Valid for low input					3
					2	Allocated to CN1, pin 8: Valid for low input					
					3	Allocated to CN1, pin 9: Valid for low input					
					4	Allocated to CN1, pin 10: Valid for low input	-				
					5	Allocated to CN1, pin 11: Valid for low input					
					6	Allocated to CN1, pin 12: Valid for low input	-				
					7	Always enabled.					
					8	Always disabled.					
					9	Allocated to CN1, pin 13: Valid for high input					
					A	Allocated to CN1, pin 7: Valid for high input					
					В	Allocated to CN1, pin 8: Valid for high input					
					С	Allocated to CN1, pin 9: Valid for high input					
					D	Allocated to CN1, pin 10: Valid for high input					
					E	Allocated to CN1, pin 11: Valid for high input					
					F	Allocated to CN1, pin 12: Valid for high input					

Param-	Parame-	Param-		C	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn511	Input sig-	2	1	EXT1 signal	0 to 3	Always disabled.	4			Offline	Stan-
	tions 5			nal allocation	4	Allocated to CN1, pin 10: Valid for low input					setting: 4
					5	Allocated to CN1, pin 11: Valid for low input					
					6	Allocated to CN1, pin 12: Valid for low input					
					7	Always enabled.					
					8	Always disabled.					
					9 to C	Always disabled.	_				
					D	Allocated to CN1, pin 10: Valid for high input					
					E	Allocated to CN1, pin 11: Valid for high input					
					F	Allocated to CN1, pin 12: Valid for high input					
			2	EXT2 signal input termi- nal allocation	0 to F	Same as Pn511.1	5			Offline	Stan- dard setting: 5
			3	EXT3 signal input termi- nal allocation	0 to F	Same as Pn511.1	6			Offline	Stan- dard setting: 6
Pn512	Output sig-	2	0	Output signal	0	Not reversed.	0			Offline	
	nal reverse			CN1 pins 1, 2	1	Reversed.					
			1	Output signal reverse for CN1 pins 23, 24	0, 1	Same as above	0			Offline	
			2	Output signal reverse for CN1 pins 25, 26	0, 1	Same as above	0			Offline	
			3	Not used.	0	(Do not change the setting.)	0				
Pn513	Not used.		(Do not	change the set	tting.)		0321				
Pn515	Not used.		(Do not	change the set	tting.)		8888				
Pn51B	Not used.		(Do not	change the set	tting.)		1000				
Pn51E	Deviation counter overflow warning level	2	Sets the overflow	e detection leve v warning.	I for the	deviation counter	100	%	10 to 100	Online	
Pn520	Deviation counter overflow level	4	Sets the level.	e deviation cou	nter over	flow alarm detection	262144	Com- mand unit	1 to 1073741823	Online	
Pn522	Position- ing com- pleted range 1	4	Setting (INP1)	range for positi	oning co	mpleted range 1	3	Com- mand unit	0 to 1073741824	Online	
Pn524	Position- ing com- pleted range 2	4	Setting (INP2)	range for positi	oning co	mpleted range 2	3	Com- mand unit	1 to 1073741824	Online	

Param-	Parame-	Param-		С	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn526	Deviation counter overflow level at Servo-ON	4	Sets the level for	e deviation cou Servo ON.	nter over	flow alarm detection	262144	Com- mand unit	1 to 1073741823	Online	
Pn528	Deviation counter overflow warning level at Servo-ON	2	Sets the tion leve	e deviation cou el for Servo ON	nter over	flow warning detec-	100	%	10 to 100	Online	
Pn529	Speed limit level at Servo- ON	2	Sets the with pos	e speed limit for sition deviation	r when th accumul	e Servo turns ON ated.	10000	r/min	0 to 10000	Online	
Pn52A	Not used.		(Do not	change the set	tting.)		20				
Pn52F	Not used.		(Do not	change the se	tting.)		FFF				

Param-	Parame-	Param-	n- Contents Digit Name Set- Explanatio				Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn530	Program JOG oper- ation related switches	2	0	Program JOG operat- ing pattern	0	$\begin{array}{l} (\text{Waiting time} \\ \text{Pn535} \rightarrow \text{Forward} \\ \text{movement Pn531} \\ \times \text{Number of move-} \\ \text{ment operations} \\ \text{Pn536} \end{array}$	0			Online	
					1	$\begin{array}{l} (Waiting time \\ Pn535 \rightarrow Reverse \\ movement Pn531) \\ \times Number of movement operations \\ Pn536 \end{array}$					
					2	$\begin{array}{l} \mbox{Waiting time Pn535}\\ \rightarrow \mbox{Forward move-}\\ ment Pn531) \times \\ \mbox{Number of move-}\\ ment operations \\ \mbox{Pn536}\\ \mbox{(Waiting time}\\ \mbox{Pn535} \rightarrow \mbox{Reverse}\\ \mbox{movement Pn531}) \\ \times \mbox{Number of move-}\\ ment operations \\ \mbox{Pn536}\\ \end{array}$					
					3	$\begin{array}{l} (\text{Waiting time} \\ \text{Pn535} \rightarrow \text{Reverse} \\ \text{movement Pn531}) \\ \times \text{Number of move-} \\ \text{ment operations} \\ \text{Pn536} \\ (\text{Waiting time} \\ \text{Pn535} \rightarrow \text{Forward} \\ \text{movement Pn531}) \\ \times \text{Number of move-} \\ \text{ment operations} \\ \text{Pn536} \end{array}$					
					4	$\begin{array}{l} (\text{Waiting time} \\ \text{Pn535} \rightarrow \text{Forward} \\ \text{movement Pn531} \\ \rightarrow \text{Waiting time} \\ \text{Pn535} \rightarrow \text{Reverse} \\ \text{movement Pn531}) \\ \times \text{Number of movement operations} \\ \text{Pn536} \end{array}$	-				
					5	$\begin{array}{l} (\text{Waiting time} \\ \text{Pn535} \rightarrow \text{Reverse} \\ \text{movement Pn531} \\ \rightarrow \text{Waiting time} \\ \text{Pn535} \rightarrow \text{Forward} \\ \text{movement Pn531}) \\ \times \text{Number of movement operations} \\ \text{Pn536} \end{array}$					
			1	Not used.	0	(Do not change the setting.)	0				
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				
Pn531	Program JOG movement distance	4	Sets the	e program JOG	moveme	ent distance.	32768	Com- mand unit	1 to 1073741824	Online	
Pn533	Program JOG movement speed	2	Sets the	e program JOG	operatio	on movement speed.	500	r/min	1 to 10000	Online	

Param-	Parame-	Param-		c	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn534	Program JOG accelera- tion/decel- eration time	2	Sets the JOG op	acceleration/c eration.	decelerat	ion time for program	100	ms	2 to 10000	Online	
Pn535	Program JOG wait- ing time	2	Sets the tion star	e delay time fro delay time fro delay time fro	m the pro eration st	ogram JOG opera- arts.	100	ms	0 to 10000	Online	
Pn536	Number of program JOG move- ments	2	Sets the operation	number of rep ns.	oetitions o	of the program JOG	1	Times	1 to 1000	Online	
Pn540	Gain limit	2	Sets the	e gain limit.			2000	× 0.1 Hz	10 to 2000	Online	
Pn550	Analog monitor 1 offset volt- age	2	Sets the	analog monito	or 1 offse	t voltage.	0	× 0.1 V	-10000 to 10000	Online	
Pn551	Analog monitor 2 offset volt- age	2	Sets the	analog monito	or 2 offse	t voltage.	0	× 0.1 V	-10000 to 10000	Online	

Other Parameters

Param- eter No.	Parameter name	Param- eter size	Contents	Default setting	Unit	Setting range	Enable setting	Details
Pn600	Regeneration resistor capacity	2	Setting for regeneration resistance load ratio monitoring calculations	0	×10 W	0 to (varies by model)	Online	

Control Function Parameters

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn800	Communi- cations control	2	0	MECHA- TROLINK com- munications check mask	0	Detects both com- munications errors (A.E6 ⁻) and syn- chronization errors (A.E5 ⁻).	0			Online	Always set to 0.
					1	Ignores communi- cations errors (A.E6□).					
					2	Ignores synchroni- zation errors (A.E5□).					
					3	Ignores communi- cations errors (A.E6 ⁻) and syn- chronization errors (A.E5 ⁻).					
			1	Warning check mask	0	Detects all parame- ter setting warnings (A.94), MECHA- TROLINK com- mand warnings (A.95), and com- munications errors (A.96).	4			Online	Always set to 4 or 0.
					1	Ignores parameter setting warning (A. 94).					
					2	Ignores command warning (A.95).					
					3	Ignores A.94 \Box and A.95 \Box .					
					4	Ignores communi- cations warning (A.96□).					
					5	Ignore A.94□ and A.96□.					
					6	Ignores A.95 \Box and A.96 \Box .					
					7	Ignores A.94□, A.95□ and A.96□.					
			2	Communica- tions error count at single transmission	0 to F	Detects communi- cations errors (A.E60) if they occur consecutively for the set value plus two times.	0			Online	
			3	Not used.	0	(Do not change the setting.)	0				

Param- eter No. ter name	Param-		C	ontents		Default	Unit	Setting	Enable	Details	
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn801	Function selection	2	0	Software limit function	0	Software limit enabled.	3			Online	
tion 6 (software LS)				1	Forward software limit disabled.						
LS)					2	Reverse software limit disabled.					
				3	Forward/reverse software limits disabled.						
			1	Not used.	0	(Do not change the setting.)	0				
			2	Software limit check using reference	0	No software limit check using refer- ence	0			Online	Always set to 0.
					1	Software limit check using refer- ence					
			3	Not used.	0	(Do not change the setting.)	0				

Param- eter No.	Parameter name	Param- eter size	Contents	Default set- ting	Unit	Setting range	Enable setting	Details
Pn802	Not used.		(Do not change the set- ting.)	0000				
Pn803	Zero point width	2	Sets the origin position detection range.	10	Command unit	0 to 250	Online	
Pn804	Forward soft- ware limit	4	Sets the software limit for the positive direction.	819,191,808	Command unit	-1,073,741,823 to 1,073,741,823	Online	
Pn806	Reverse soft- ware limit	4	Sets the software limit for the negative direction.	-819,191,808	Command unit	-1,073,741,823 to 1,073,741,823	Online	
Pn808	Absolute encoder zero point position offset	4	Sets the encoder posi- tion and machine coordi- nate system offsets for when an absolute encoder is used.	0	Command unit	-1,073,741,823 to 1,073,741,823	Offline	
Pn80A	First step linear acceleration constant	2	Sets the step 1 accelera- tion for when two-step acceleration is used.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80B	Second step lin- ear acceleration constant	2	Sets the step 2 accelera- tion for when two-step acceleration is exe- cuted, or the one-step acceleration constant for when one-step accelera- tion is executed.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80C	Acceleration constant switch- ing speed	2	Sets the switching speed for the step 1 and step 2 acceleration when two- step acceleration is exe- cuted.	0	×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).

Param- eter No.	Parameter name	Param- eter size	Contents	Default set- ting	Unit	Setting range	Enable setting	Details
Pn80D	First step linear deceleration constant	2	Sets the step 1 decelera- tion for when two-step deceleration is used.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80E	Second step lin- ear deceleration constant	2	Sets the step 2 decelera- tion for when two-step deceleration is exe- cuted, or the one-step deceleration constant for when one-step decelera- tion is executed.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80F	Deceleration constant switch- ing speed	2	Sets the switching speed for the step 1 and step 2 deceleration when two- step deceleration is exe- cuted.	0	×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn810	Exponential acceleration/ deceleration bias	2	Sets the bias for when an exponential filter is used for the position command filter.	0	Command units/s	0 to 32767	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn811	Exponential acceleration/ deceleration time constant	2	Sets the time constant for when an exponential filter is used for the posi- tion command filter.	0	×0.1 ms	0 to 5100	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn812	Moving average time	2	Sets the average move- ment time for when S- curve acceleration/ deceleration is used, and an average movement filter is used for the posi- tion command filter.	0	×0.1 ms	0 to 5100	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn813	Not used.		(Do not change the set- ting.)	0				
Pn814	Final travel dis- tance for exter- nal positioning	4	Sets the distance from the external signal input position when external positioning is executed.	100	Command unit	-1,073,741,823 to 1,073,741,823	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).

Param-	Parame-	Param-		Contents			Default	Unit	Setting range	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting			setting	
Pn816	Zero point	2	0	Zero point return direc-	0	Forward direction	0			Online	Set the same direc-
mode set- tings					1	Reverse direction					origin search direction set in the Axis Parameters. Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
			1 to 3	Not used.	0	(Do not change the setting.)	0				
Pn817	Zero point return approach speed 1	2	Sets the eration	e origin search limit switch sig	fter the decel- oN.	50	×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).	
Pn818	Zero point return approach speed 2	2	Sets the eration	e origin search limit switch sig	speed a nal turns	fter the decel- ON.	5	×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn819	Final travel dis- tance to return to zero point	4	Sets the position is exect	e distance fron to the origin, t ited.	n the latc for when	h signal input origin search	100	Com- mand unit	-1,073,741,823 to 1,073,741,823	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn81B to Pn825	Not used.		(Do not	change the se	etting.)		0000				
Pn900 to Pn910	Not used.		(Do not	change the se	etting.)						
Pn920 to Pn95F	Not used.		(Do not	change the se	etting.)						

4-5-5 SMARTSTEP Junior Servo Drive (R7D-ZND-ML2 with Built-in MECHATROLINK-II Communications)

The Servo Parameters listed here can be used when the Position Control Unit is used with a SMARTSTEP Junior Servo Drive with Built-in MECHA-TROLINK-II Communications (R7D-ZN□-ML2).

For further details on each of the parameters, refer to the user's manual for SMARTSTEP Junior Servo Drives.

The timing for Servo Parameters to be enabled are classified into the following two types.

Online (online parameter): Changed settings are enabled immediately after Servo Parameters have been written.

Offline (offline parameters): Changed settings are enabled when the Servo Drive power is cycled or DEVICE SETUP is executed.

The *Details* column indicates whether the following conditions apply to the corresponding parameter.

- Online parameters that can be changed when the axis is stopped (Busy Flag = 0) only.
- Parameters with standard set values when using MECHATROLINK are indicated in the *Details* column as "standard" with the set value. The standard settings are basic settings required when using the Position Control Unit. This manual describes operations assuming standard settings are being used.

For details on fixed and standard settings, refer to 6-4 Standard Settings for Servo Drives Using MECHATROLINK.

Note Make sure that the equipment will not be adversely affected before changing the Servo Parameters (WRITE SERVO PARAMETER, SAVE SERVO PARAMETER). Refer to the Servo Drive's user's manual and always check the effect of changing the settings before changing the Servo Parameters.

	Function	Selection	Parameters
--	----------	-----------	------------

Param- Parame		Param- eter		Con	tents		Default	Unit Setting ran	Setting range	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting			setting	
Pn000	Function selection basic switch	2	0	Reverse rotation	0	CCW direc- tion is taken for positive command.	0			Offline	
					1	CW direc- tion is taken for positive command.					
					2 to 3	Not used.					
			1	Not used.	1	(Do not change the setting.)	1				
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				
Pn00A	Com- mand fil- ter setting	2	Set the When u SW2 or Drive.	command filte sing this parar the SMARTS	r constar neter, tur TEP Jun	nt. m ON bit 4 on ior Servo	0000		0000 to 000F	Online	

Position Control Parameters

Param- Parame-		Param-		Con	tents		Default	Unit	Setting range	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting			setting	
Pn20E	Elec- tronic gear ratio G1 (numera- tor)	4	Set the and Sei 0.01≤G	pulse rate for t rvomotor travel 1/G2≤100	he comn distance	nand pulses ə.	1		1 to 1073741824	Offline	
Pn210	Elec- tronic gear ratio G2 (denomi- nator)	4					1		1 to 1073741824	Offline	

■ I/O and Status Parameters

Param- Parame- eter No. ter name	Param- eter		Con	tents		Default	Default Unit	Setting range	Enable	Details	
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting			setting	
Pn50A	Input sig- nal selec- tion 1	2	0	Not used.	1	(Do not change the setting.)	1			Offline	
			1	Not used.	8	(Do not change the setting.)	8				
			2	Not used.	8	(Do not change the setting.)	8				
			3	POT (for- ward drive prohibited input) signal	2	Allocated to CN1, pin 4: Valid for low input	2			Offline	Standard setting: 2
				nal alloca- tion	8	Always invalid.					
Pn50B	Input sig- nal selec- tion 2	2	0	NOT (reverse drive prohib- ited input)	3	Allocated to CN1, pin 3: Valid for low input	3			Offline	Standard setting: 3
				signal input terminal allocation	8	Always invalid.					
			1	Not used.	8	(Do not change the setting.)	8				
			2	Not used.	8	(Do not change the setting.)	8				
			3	Not used.	8	(Do not change the setting.)	8				
Pn515	Input sig- nal selec- tion 7	2	0	Not used.	8	(Do not change the setting.)	8				
			1	Not used.	8	(Do not change the setting.)	8				
			2	STOP (emergency stop input)	4	Allocated to CN1, pin 6: Valid for low input	4			Offline	Standard setting: 4
						Always invalid.					
			3	Not used.	8	(Do not change the setting.)	8				

Param-	Parame-	Param-		Con	tents		Default Ur	Unit	Setting range	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting			setting	
Pn522	Position- ing com- pletion width 1	4	Setting range	Setting range for positioning completed ange				Com- mand unit	0 to 1,073,741,824	Online	
Pn524	Position- ing com- pletion width 2	4	Setting Proximi	for proximity ra ty Flag.	inge for t	he Positioning	100	Com- mand unit	1 to 1,073,741,824	Online	

Control Function Parameters

Param-	Parame-	Param-		Con	tents		Default	Unit	Setting range	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting			setting	
Pn800	Commu- nications control	2	0	Not used.	0	(Do not change the setting.)	1				
			1	Warning check mask	0	Detects parameter setting warn- ings (A.94), command warnings (A.95), and communica- tions errors (A.96).	4			Online	Always set to 4 or 0.
					1	Ignores parameters setting warn- ings (A.94).					
					2	Ignores com- mand warn- ings (A.95).					
					3	Ignores both parameter setting warn- ings (A.94) and com- mand warn- ings (A.95).					
					4	Ignores com- munications errors (A.96).					
					5	Ignores both parameter setting warn- ings (A.94) and commu- nications errors (A.96).					
					6	Ignores both command warnings (A.95) and communica- tions errors (A.96).					
					7	Ignores parameter setting warn- ings (A.94), command warnings (A.95), and communica- tions errors (A.96).					
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				

Param-	Parame-	Param-		Contents [Default	It Unit Setting rang		Enable	Details		
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting			setting	
Pn801	Function selection	2	0	Software limit function	0	Software limit enabled.	3			Online	
	applica- tion 6 (software				1	Forward software limit disabled.					
	limits)				2	Reverse software limit disabled.					
					3	Software limit disabled in both directions.					
			1	Not used.	0	(Do not change the setting.)	0				
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				
PN803	Zero point width	2	Sets the Flag.	e detection ran	ge for th	e Origin Stop	10	Com- mand unit	0 to 250	Online	
Pn804	Forward software limit	4	Sets the	e forward softw	/are limit		107374 1823	Com- mand unit	-1073741823 to 1073741823	Online	
Pn806	Reverse software limit	4	Sets the	e reverse softw	/are limit		-10737 41823	Com- mand unit	-1073741823 to 1073741823	Online	
Pn80B	Linear accelera- tion con- stant	2	Sets the tion/dec trol.	e acceleration celeration curv	speed fo e used ir	r the accelera- position con-	100	×10,000 com- mand units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80E	Linear decelera- tion con- stant	2	Sets the tion/dec trol.	e deceleration celeration curv	speed fo e used ir	r the accelera- position con-	100	×10,000 com- mand units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn814	Final travel dis- tance for external position- ing	4	Sets the positior FEEDIN	e final travel di ning when exec NG using direc	stance fo cuting IN t operatio	or external TERRUPT on.	100	Com- mand unit	-1073741823 to 1073741823	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).

Param-	Parame-	Param-		Con	tents		Default	Unit	Setting range	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting			setting	
Pn816	Zero	2	0	Zero point	0	Forward	0			Online	Set the
	point return mode set- ting	1 to 3 Not used.		1	Reverse					same direc- tion as the origin search direction set in the Axis Parameters. Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).	
			1 to 3	Not used.	0	(Do not change the setting.)	0				
Pn817	Zero point return approach speed 1	2	Sets the search signal h	e origin (zero p speed used af las been detec	point) inp ter the or ted in an	ut signal rigin proximity 1 origin search.	50	×100 com- mand units/ s ²	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn818	Zero point return approach speed 2	2	Sets the distanc gin inpu gin sea	e origin (zero p e positioning s ut signal has bo rch.	point) retu peed use een dete	urn final travel ad after the ori- cted in an ori-	5	×100 com- mand units/ s ²	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn819	Final travel dis- tance to return to zero point	4	Sets the used af detecte	e amount of co ter the origin i d in an origin s	mpensati nput sign search.	ion positioning al has been	100	Com- mand unit	-1073741823 to 1073741823	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn820 to Pn823	Not used.		(Do not	change the se	etting.)		0000				

4-6 Common Operating Memory Area

The Common Operating Memory Area is allocated to outputs used for common PCU operations and settings, such as communications control and transferring common parameters, and inputs used for monitoring the status of these operations.

4-6-1 Common Operating Memory Area Overview

The area allocated as the Common Operating Memory Area is contained in the CPU Bus Unit Area within the CPU Unit's CIO Area. The beginning word of the Common Operating Memory Area is determined by the unit number set for the PCU using the following equation.

Beginning word of Common Operating Memory Area: n = CIO 1500 + (unit number \times 25)

Unit number	Allocated words	Unit number	Allocated words
0	CIO 1500 to CIO 1524	8	CIO 1700 to CIO 1724
1	CIO 1525 to CIO 1549	9	CIO 1725 to CIO 1749

Unit number	Allocated words	Unit number	Allocated words
2	CIO 1550 to CIO 1574	10	CIO 1750 to CIO 1774
3	CIO 1575 to CIO 1599	11	CIO 1775 to CIO 1799
4	CIO 1600 to CIO 1624	12	CIO 1800 to CIO 1824
5	CIO 1625 to CIO 1649	13	CIO 1825 to CIO 1849
6	CIO 1650 to CIO 1674	14	CIO 1850 to CIO 1874
7	CIO 1675 to CIO 1699	15	CIO 1875 to CIO 1899

The Common Operating Memory Area is divided into outputs and inputs. Commands for common PCU operations and settings, such as transferring PCU data (reading, writing, and saving Common Parameters and Axis Parameters) and establishing connections with MECHATROLINK communications are allocated to the output memory area.

Commands are sent to the PCU when the respective bit in the output memory area turns ON or while the respective bit is ON. The common PCU status and MECHATROLINK communications status are input from the PCU to the input memory area.

WWW.BSNEW.IR

4-6-2 Common Operating Memory Area Words

The memory allocation of the Common Operating Memory Area is shown in the following table. For details on functions and operations of each word, such as operation timing, refer to the section given in the *Details* column.

Common Operating Output Memory Area

 $n = CIO 1500 + (unit number \times 25)$

I/O	Word	Bits	Category	Name	Operation	Details
Output (CPU	n	00		UNIT ERROR RESET	Resets the Unit common error when this bit turns ON.	12-6 Error Reset
Unit to PCU)		01	Data trans- fer com- mands	WRITE DATA	Writes data to the PCU from the CPU Unit when this bit turns ON.	5-2-1 Writing PCU Parameters
		02		READ DATA	Reads data from the PCU to the CPU Unit when this bit turns ON.	5-2-2 Reading PCU Parameters
		03		SAVE DATA	Saves the contents of the PCU's internal memory to its flash memory when this bit turns ON.	5-2-3 Saving PCU Parameters
		04 to 15		Not used (reserved by the system).	These bits are reserved by the system. Do not use.	
	n+1	00	Communi- cations control commands 	CONNECT	Starts/stops MECHA- TROLINK communications.	6-3-1 Establishing Connections
		01 to 05		Reserved by the sys- tem.	These bits are reserved by the system. Do not use.	
		06		WRITE BACKUP DATA	Writes data saved in the PCU's internal flash memory to the Memory Card installed in the CPU Unit when this bit turns ON.	13-4-2 Memory Card Backup
		07		READ BACKUP DATA	Reads data from the Memory Card installed in the CPU Unit to the PCU's internal flash memory when this bit turns ON.	
		08 to 14		Not used (reserved by the system).	These bits are reserved by the system. Do not use.	
		15		REJOIN (See note.)	Restarts MECHATROLINK communications.	6-3-4 Rejoining the Connection
	n+2			Axes to connect (See note.)	Specifies the axes for which to start MECHATROLINK com- munications.	6-3-5 Specifying the Axes to Connect
	n+3 to n+5			Not used (reserved by the system).	These bits are reserved by the system. Do not use.	

I/O	Word	Bits	Category	Name	Operation	Details
Output (CPU Unit to PCU)	n+6		Operating data for data trans- fer	Number of write words	Specifies the number of words to be written from the CPU Unit to the PCU.	5-2-1 Writing PCU Parameters
	n+7			Write source area	Specifies the area containing the data to be written from the CPU Unit to the PCU.	
	n+8			Write source word	Specifies the beginning word of the data to be written from the CPU Unit to the PCU.	
	n+9			Write destination address	Specifies the address in the PCU to which the data is to be written.	
	n+10			Number of read words	Specifies the number of words to be read from the PCU to the CPU Unit.	5-2-2 Reading PCU Parameters
	n+11			Read source address	Specifies the address in the PCU from which the data is to be read.	
	n+12			Read destination area	Specifies the area used to store the data read from the PCU.	
	n+13			Read destination word	Specifies the word used to store the data read from the PCU.	
	n+14			Not used (reserved by the system).	These bits are reserved by the system. Do not use.	

Note

The REJOIN Bit and the Axes to Connect parameter are supported for unit version 2.0 or later. The allocated bit and word are not used for earlier unit versions.

$\begin{array}{ll} \mbox{Common Operating Input} & n = CIO \ 1500 + (unit \ number \times 25) \\ \mbox{Memory Area} \end{array}$

I/O	Word	Bits	Category	Name	Operation	Details
Input (PCU	n+15	00 to 11	PCU com- mon status words	Not used (reserved by the system).	These bits are reserved by the system. Do not use.	
to CPU Unit)		12		Unit Error Flag	ON when a Unit common error has occurred.	12-1 Overview of PCU Errors
		13	Ī	Not used (reserved by the system).	This bit is reserved by the sys- tem. Do not use.	
		14		Data Transferring Flag	ON when data is being trans- ferred between the CPU Unit and PCU, or when data is being saved in the PCU's internal flash memory.	5-2 Transferring PCU Parameters
		15		Not used (reserved by the system).	This bit is reserved by the system. Do not use.	
	n+16	00 to 11		Reserved by the sys- tem.	These bits are reserved by the system. Do not use.	
		12		Memory Card Trans- fer Error	Indicates that an error occurred when writing/reading backup data to/from the Mem- ory Card.	13-4-2 Memory Card Backup
		13		Reserved by the sys- tem.	This bit is reserved by the sys- tem. Do not use.	
		14		Unit Busy Flag	ON when PCU is processing command sent to the Com- mon Operating Memory Area.	12-6 Error Reset
					This flag also turns ON during initial processing when PCU power is turned ON or the Unit is restarted.	
		15		Connection Status Flag	ON when PCU starts MECHATROLINK communi- cations.	6-3-2 MECHA- TROLINK Communi- cations Status
	n+17to n+20			Reserved by the system.	These bits are reserved by the system. Do not use.	
	n+21			Unit error code	Returns the error code when a Unit Common error occurs.	12-4 Error Codes

I/O	Word	Bits	Category	Name	Operation	Details
Input (PCU	n+22	00 // r	Axis com- munica- tions status bits	Axis 1 communica- tions status	Bits 00 to 15 correspond to the communications status for axes 1 to 16. The bits will turn ON if the cor- responding axes registered in the scan list are communicat- ing normally.	6-3-2 MECHA- TROLINK Communi- cations Status
to CPU Unit)		01		Axis 2 communica- tions status		
		02		Axis 3 communica- tions status		
		03		Axis 4 communica- tions status		
		04		Axis 5 communica- tions status		
		05		Axis 6 communica- tions status		
		06		Axis 7 communica- tions status		
		07		Axis 8 communica- tions status		
		08		Axis 9 communica- tions status		
		09		Axis 10 communica- tions status		
		10		Axis 11 communica- tions status		
		11		Axis 12 communica- tions status		
		12		Axis 13 communica- tions status		
		13		Axis 14 communica- tions status		
		14		Axis 15 communica- tions status		
		15		Axis 16 communica- tions status		
	n+23to n+24			Not used (reserved by the system).	These bits are reserved by the system. Do not use.	

4-7 Axis Operating Output Memory Areas

The Axis Operating Output Memory Areas contain outputs used to set operation for the Servo Drive and Servomotor axes that are connected using MECHATROLINK communications. These outputs include operating commands, such as direct operation, ORIGIN SEARCH, and JOG, and related position and speed command values.

4-7-1 Axis Operating Output Memory Area Overview

The Axis Operating Output Memory Areas are allocated 25 words per axis in the CPU Unit's memory area set in the PCU's Common Parameters (Axis Operating Output Memory Area designation, beginning word of Axis Operating Output Memory Area).

The designated beginning word corresponds to the beginning word of the area for axis 1, and the other areas are allocated words in sequence up to the highest axis number registered in the scan list. For details on area allocations, refer to *6-2-2 Scan List and PCU Area Allocations*.

The beginning words of the Axis Operating Output Memory Areas are determined by the axis number of each axis using the following equation.

Beginning Word of Axis Operating Output Memory Area for Axis N:

a= Beginning word of Axis Operating Output Memory Area specified in Common Parameters + $(N-1) \times 25$ (N = 1 to 16)



a = Beginning word of Axis Operating Output Areas specified in Common Parameters

4-7-2 Axis Operating Output Memory Area Allocations

The memory allocation of the Axis Operating Output Memory Areas is shown in the following table. For details on functions and operations of each word, such as operation timing, refer to the section given in the *Details* column.

a = Beginning word of Axis Operating Output Memory Areas specified in Common Parameters + (Axis No. -1) \times 25

I/O	Word	Bits	Category	Name	Operation	Details
Output (CPU Unit to	a	00	Direct oper- ation com- mands	LINEAR INTERPO- LATION SETTING (See note 1.)	Setting the linear interpola- tion operation is started when this bit turns ON.	9-7 Linear Interpola- tion
PCU)		01		LINEAR INTERPO- LATION START (See note 1.)	The linear interpolation opera- tion that has been set is started when this bit turns ON.	
		02		Not used (reserved by the system).	These bits are reserved by the system. Do not use.	
		03		ABSOLUTE MOVE- MENT	Starts positioning operation with the designated position treated as an absolute posi- tion when this bit turns ON.	9-4 Using Direct Operation
		04		RELATIVE MOVE- MENT	Starts positioning operation with the designated position treated as a relative position when this bit turns ON.	
		05		INTERRUPT FEED- ING	Interrupt feeding begins for ABSOLUTE MOVEMENT and RELATIVE MOVEMENT com- mands.	9-5 Interrupt Feeding
		06	Origin posi- tioning commands	ORIGIN SEARCH	Executes an origin search when this bit turns ON.	8-2 Origin Search Operation
		07		ORIGIN RETURN	Returns position to the origin when this bit turns ON.	8-4 Origin Return
		08		PRESENT POSI- TION PRESET	Forcibly changes the present position to the designated position, and thus establishes an origin relative to the posi- tion when this bit turns ON.	8-3 Present Position Preset
		09	Com- mands for special functions	JOG	Jogging is executed while this bit is ON.	10-2 Jogging
		10		Direction designation	This bit designates the direc- tion for when jogging is exe- cuted.	
		11		Not used (reserved by the system).	This bit is reserved by the system. Do not use.	
		12		ERROR RESET	Resets (turns OFF) the error status for each axis when this bit turns ON.	12-6 Error Reset
		13		DEVIATION COUNTER RESET (See note 4.)	Resets the deviation counter to 0 when this bit turns ON.	10-10 DEVIATION COUNTER RESET
		14		Override Enable Bit	This bit enables or disables the override.	10-3 Override
		15		DECELERATION STOP	Starts decelerating position- ing to a stop when this bit turns ON.	10-9 Stop Functions

I/O	Word	Bits	Category	Name	Operation	Details
Output (CPU	a+1	a+1 00	Com- mands for	SERVO LOCK	Executes SERVO LOCK when this bit turns ON.	10-1 Servo Lock/ Unlock
Unit to PCU)		01	special functions	SERVO UNLOCK	Executes SERVO UNLOCK when this bit turns ON.	
		02		SPEED CONTROL	Starts speed control using the speed command value for speed control as the target speed when this bit turns ON.	10-5 Speed Control
		03		TORQUE CON- TROL	Starts torque control using the torque command value when this bit turns ON.	10-6 Torque Control
		04 to 10		Not used (reserved by the system).	These bits are reserved by the system. Do not use.	
		11 Servo Parameter	DEVICE SETUP	Starts Servo Drive setup when this bit turns ON.	5-3-4 Device Setup	
		12	transfer commands	WRITE SERVO PARAMETER	Writes data to the Servo Drive from the CPU Unit when this bit turns ON.	5-3-1 Writing Servo Parameters
		13		READ SERVO PARAMETER	Reads data from the Servo Drive to the CPU Unit when this bit turns ON.	5-3-2 Reading Servo Parameters
		14		SAVE SERVO PARAMETER	Writes data to the Servo Drive's non-volatile memory at the same time as writing the Servo Parameters when this bit turns ON.	5-3-3 Saving Servo Parameters
		15	Special command	EMERGENCY STOP	Executes emergency stop when this bit turns ON.	10-9 Stop Functions
	a+2	Operating data for	Position command value (rightmost	Specifies the position for per- forming present position pre-	9-4 Using Direct Operation	
	a+3		control	Word)	set and direct operation.	9-7 Linear Interpola-
	uro			value (leftmost word)		8-3 Present Position Preset
	a+4			Speed command value (rightmost	Specifies the target speed for performing direct operation,	8-2 Origin Search Operation
			word)	jogging, origin searches, and origin returns	8-4 Origin Return	
	a+5			value (leftmost word)		9-4 Using Direct Operation
						9-7 Linear Interpola- tion
						10-2 Jogging

I/O	Word	Bits	Category	Name	Operation	Details
Output (CPU Unit to PCU)	a+6		Speed con- trol data	Speed command value for speed con- trol (rightmost word)	Specifies the target speed for performing speed control.	10-5 Speed Control
	a+7			Speed command value for speed con- trol (leftmost word)		
	a+8		Torque con- trol data	Torque command value (rightmost word)	Specifies the torque for per- forming torque control.	10-6 Torque Control
	a+9			Torque command value (leftmost word)		
	a+10		Position/ speed/ torque con- trol data	Option command value 1 (rightmost word)	Set the command value of the auxiliary functions for position/ speed/torque control, such as the acceleration/deceleration for linear interpolation, torque feed forward command, the torque limit for speed control, and the speed limit for torque control.	9-7 Linear Interpola- tion 10-5 Speed Control 10-6 Torque Control
	a+11			Option command value 1 (leftmost word)		
	a+12			Option command value 2 (rightmost word)		
	a+13			Option command value 2 (leftmost word)		
	a+14		Operating data for special function	Override	This bit specifies the override rate.	10-3 Override
	a+15	00 to 03	Operating data for expanded	Monitor 1 type	Specifies the monitor type for monitor 1 in the Axis Operat- ing Input Memory Areas.	4-8-6 Expanded Monitoring
	04 to 07 08 to 15	04 to 07 monitoring	Monitor 2 type	Specifies the monitor type for monitor 2 in the Axis Operat- ing Input Memory Area.		
		08 to 15		Not used (reserved by the system).	These bits are reserved by the system. Do not use.	
I/O	Word	Bits	Category	Name	Operation	Details
-----------------	------	----------	--	---	---	--
Output (CPU	a+16	00 to 02		Reserved by the sys- tem.	These bits are reserved by the system. Do not use.	
Unit to PCU)		03	Accelera- tion/decel- eration curves	Exponential curve designation	Specifies an exponential curve as the acceleration/ deceleration curve for execut- ing direct operation, jogging, origin searches, and origin returns.	 8-2 Origin Search Operation 8-4 Origin Return 9-4 Using Direct Operation
	04		*	S-curve designation	Specifies an S-curve (move- ment average) as the acceler- ation/deceleration curve for executing direct operation, jogging, origin searches, and origin returns.	10-2 Jogging
		05 to 13		Reserved by the sys- tem.	These bits are reserved by the system. Do not use.	
		14	Torque limit	Forward rotation cur- rent limit designation	Specifies the torque limit in the forward direction for axis operation.	10-4 Torque Limits
		15		Reverse rotation cur- rent limit designation	Specifies the torque limit in the reverse direction for axis operation.	
	a+17		Operating data for transferring Servo	Servo Parameter No.	Specifies the parameter num- ber for the Servo Parameters to be written from or read to the CPU Unit.	5-3 Transferring Servo Parameters
	a+18		Parameters	Parameter size	Specifies the parameter size for the Servo Parameters to be written from or read to the CPU Unit.	
	a+19			Write data (rightmost word)	Specifies the data to be writ- ten to the Servo Drive from	5-3-1 Writing Servo Parameters
	a+20			Write data (leftmost word)	the CPU Unit.	5-3-3 Saving Servo Parameters
	a+21	00 to 03	Linear inter- polation data	Interpolation axis designation for axes 1 to 4 (See note 2.)	Specifies the axes for linear interpolation for combinations of axes 1 to 4.	9-7 Linear Interpola- tion
		04 to 07		Interpolation axis designation for axes 5 to 8 (See note 3.)	Specifies the axes for linear interpolation for combinations of axes 5 to 8.	
		08 to 15		Not used (reserved by the system).	These bits are reserved by the system. Do not use.	
	a+22	00 to 03		Interpolation posi- tion designation for axes 1 to 4 (See note 2.)	Specify absolute or relative positioning for the axes for linear interpolation.	
		04 to 07		Interpolation posi- tion designation for axes 5 to 8 (See note 3.)		
		08 to 15		Not used (reserved by the system).	These bits are reserved by the system. Do not use.	
	a+23			Interpolation speed command value (rightmost word) (See note 1.)	Specifies the interpolation speed for linear interpolation.	
	a+24			Interpolation speed command value (leftmost word) (See note 1.)		

- Note (1) Allocated in Axis Operating Output Memory Areas for axis 1 and axis 5 for Position Control Unit Ver. 1.1 or later. These bits are not used in the Axis Operating Output Memory Areas for other axes.
 - (2) Allocated in Axis Operating Output Memory Area for axis 1 for Position Control Unit Ver. 1.1 or later. These bits are not used in the Axis Operating Output Memory Areas for other axes.
 - (3) Allocated in Axis Operating Output Memory Area for axis 5 for Position Control Unit Ver. 1.1 or later. These bits are not used in the Axis Operating Output Memory Areas for other axes.
 - (4) The DEVIATION COUNTER RESET can be used with unit version 1.3 or later. This bit is not used for earlier unit versions.

4-7-3 Axis Operating Output Memory Area Priority

The Position Control Unit performs exclusive control for the command executed for the axis operating output bits. Depending on the status of the Position Control Unit (i.e., the Servo locked/unlocked status, Busy Flag status, and ON/OFF status of axis operating output bits), the command is disabled or detected as an error.

The operations that are performed when an attempt is made to execute more than one function are described depending on the operating status.

When Position Control Unit Is Not Executing a Function (When Busy Flag = 0)

The operation that is performed for each command when the Position Control Unit is not executing a function (including when the Busy Flag is reset after completing or canceling operation) is shown below depending on the Servo locked/unlocked status and the ON/OFF status of axis operating output bits.

(Refer to *12-1 Overview of PCU Errors* for information on executing functions when an error exists.)

Servo Unlocked

		Function to be executed (See note.)													
Executing function	SERVO UNLOCK	DEVIATION COUNTER RESET	EMERGENCY STOP	DECELERATION STOP	JOG	ORIGIN SEARCH	ORIGIN RETURN	ABSOLUTE MOVEMENT	RELATIVE MOVEMENT	INTERRUPT FEEDING	SPEED CONTROL	TORQUE CONTROL	PRESENT POSITION PRESET	DEVICE SETUP	SERVO LOCK
SERVO UNLOCK Bit ON													Ο	Ο	
DEVIATION COUNTER RESET Bit ON	0												0	0	
EMERGENCY STOP Bit ON	О												Ο	Ο	
DECELERATION STOP Bit ON	0												Ο	Ο	
Other than above status	0				×	×	×	×	×	×	×		0	0	0

O: Execution possible

X: A servo unlock error (axis error code: 3040) will occur and the function will not be executed.

---: Execution not possible (ignored)

Note: The above table assumes that the conditions required for executing jogging, origin searches, origin returns, absolute/relative movements (including interrupt feeding), speed control, or torque control, such as command values and origin establishment, have been met.

Servo Locked

		Function to be executed (See note.)													
Executing function	SERVO UNLOCK	DEVIATION COUNTER RESET	EMERGENCY STOP	DECELERATION STOP	DOL	ORIGIN SEARCH	ORIGIN RETURN	ABSOLUTE MOVEMENT	RELATIVE MOVEMENT	INTERRUPT FEEDING	SPEED CONTROL	TORQUE CONTROL	PRESENT POSITION PRESET	DEVICE SETUP	SERVO LOCK
DEVIATION COUNTER RESET Bit ON	0												0	×	
EMERGENCY STOP Bit ON	0	Ο											0	×	
DECELERATION STOP Bit ON	0	0	0										0	×	
Other than above status	0	0	0	0	0	0	0	0	0	0	0	0	0	×	

O: Execution possible

- ×: A multistart error (axis error code: 3050) will occur and the function will not be executed.
- ---: Execution not possible (ignored)
- **Note:** The above table assumes that the conditions required for executing jogging, origin searches, origin returns, absolute/relative movements (including interrupt feeding), speed control, or torque control, such as command values and origin establishment, have been met.

When Position Control Unit Is Executing a Function (When Busy Flag = 1)

The operation that is performed for each command when the Position Control Unit is already executing a function is shown below depending on the Servo locked/unlocked status and the ON/OFF status of axis operating output bits.

		Function to be executed (See note.)													
Executing function	SERVO UNLOCK	DEVIATION COUNTER RESET	EMERGENCY STOP	DECELERATION STOP	JOG	ORIGIN SEARCH	ORIGIN RETURN	ABSOLUTE MOVEMENT	RELATIVE MOVEMENT	INTERRUPT FEEDING	SPEED CONTROL	TORQUE CONTROL	PRESENT POSITION PRESET	DEVICE SETUP	SERVO LOCK
SERVO UNLOCK															
DEVIATION COUNTER RESET	0														
EMERGENCY STOP	Ο	0													

		Function to be executed (See note.)													
Executing function	SERVO UNLOCK	DEVIATION COUNTER RESET	EMERGENCY STOP	DECELERATION STOP	JOG	ORIGIN SEARCH	ORIGIN RETURN	ABSOLUTE MOVEMENT	RELATIVE MOVEMENT	INTERRUPT FEEDING	SPEED CONTROL	TORQUE CONTROL	PRESENT POSITION PRESET	DEVICE SETUP	SERVO LOCK
DECELERATION STOP	0	Ο	Ο												
JOG	0	Ο	0	0	-	×	×	×	×	×	×	×	×	×	
ORIGIN SEARCH	0	Ο	Ο	Ο	×	×	×	×	×	×	×	×	×	×	
ORIGIN RETURN	0	0	0	0	×	×	×	×	×	×	×	×	×	×	
POSITION CONTROL (ABSO- LUTE MOVEMENT)	0	0	0	0	×	×	×	0	0	×	0	0	×	×	
POSITION CONTROL (RELA- TIVE MOVEMENT)	0	0	0	0	×	×	×	0	0	×	0	0	×	×	
INTERRUPT FEEDING	0	0	0	0	×	×	×	×	×	×	0	0	×	×	
SPEED CONTROL	0		0	0	×	×	×	0	Ο	0	0	0	×	×	
TORQUE CONTROL	Ο		0	Ο	×	×	×	0	О	Ο	Ο	0	×	×	
PRESENT POSITION PRESET	0	×	0	0	×	×	×	×	×	×	×	×	×	×	×
DEVICE SETUP															
SERVO LOCK	Ο				×	×	×	×	×	×	×	×	×	×	×

O: Execution possible

- ×: A multistart error (axis error code: 3050) will occur and the function will not be executed.
- ---: Execution not possible (ignored)
- **Note:** The above table assumes that the conditions required for executing jogging, origin searches, origin returns, absolute/relative movements (including interrupt feeding), speed control, or torque control, such as command values and origin establishment, have been met.

Executing Multiple Functions Simultaneously

The following table shows the PCU operations when multiple functions are executed at the same time for the same axis during a servo lock (including executing other functions while the Receiving Command Flag is ON.

Function	Operation for simultaneous execution
SERVO UNLOCK	Takes priority over any other function executed at the same time.
	All other operations are disabled while the SERVO UNLOCK Bit is ON.
DEVIATION COUNTER RESET	Takes priority over any other function executed at the same time except for SERVO UNLOCK.
	The following functions will not be started while the DEVI- ATION COUNTER RESET Bit is ON.
EMERGENCY STOP	Executed before any other function executed at the same time, except SERVO UNLOCK or DEVIATION COUNTER RESET.
	All other operation functions are disabled while the EMERGENCY STOP Bit is ON.

Function	Operation for simultaneous execution
DECELERATION STOP	Executed before any other function executed at the same time, except SERVO UNLOCK, DEVIATION COUNTER RESET, and EMERGENCY STOP.
	All other operation functions are disabled while the DECELERATION STOP Bit is ON.
ABSOLUTE/RELATIVE MOVEMENT (with INTERRUPT FEEDING)	A Multistart Error (error code: 3050) will occur if these commands are executed at the same time.
ORIGIN SEARCH	
ORIGIN RETURN	
PRESENT POSITION	
PRESET	
JOG	
SPEED CONTROL	
TORQUE CONTROL	
DEVICE SETUP	
SERVO LOCK	

4-8 Axis Operating Input Memory Areas

The Axis Operating Input Memory Areas are allocated inputs for monitoring the status of axis operations of the Servo Drive and Servomotor axes that are connected using MECHATROLINK communications. These include present position and axis operating status.

4-8-1 Axis Operating Input Memory Area Overview

The Axis Operating Input Memory Areas are allocated 25 words for each axis in the CPU Unit's memory area set in PCU's Common Parameters (Axis Operating Input Memory Area designation and beginning word of Axis Operating Input Memory Area).

The designated beginning word corresponds to the beginning word of the area for axis 1, and the other areas are allocated words in sequence up to the highest axis number registered in the scan list. For details on area allocations, refer to *6-2-2 Scan List and PCU Area Allocations*.

The beginning words of the Axis Operating Input Memory Areas are determined by the axis number of each axis using the following equation.

Beginning Word of Axis Operating Input Memory Area for Axis N:

b = Beginning word of Axis Operating Input Memory Area specified in Common Parameters + $(N-1) \times 25$ (N = 1 to 16)



b: Beginning word of the Axis Operating Memory Areas specified in the Common Parameters.

4-8-2 Axis Operating Input Memory Area Allocations

The memory allocation of the Axis Operating Input Memory Areas is shown in the following table. For details on functions and operations of each word, such as operation timing, refer to the section given in the *Details* column.

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

I/O	Word	Bits	Category	Name	Function	Details
Input (PCU to CPU Unit)	b	00	Axis Con- trol Status Flags	Receiving Com- mand Flag	ON when the PCU is in the process of receiving an oper- ating command. 1: Receiving command If another operating command with the same priority level is started while this flag is ON, a Multistart Error (axis error code: 3050) will occur. (Refer to 4-7-3 Axis Operating Out- put Memory Area Priority.) The Receiving Command Flag remains ON (1) for at least one cycle time, and turns OFF (0) when the com- mand reception processing is completed.	4-8-3 Axis Control Status Flags 9-4 Using Direct Operation 10-5 Speed Control 10-6 Torque Control
		01 to 04		Not used (reserved by the system).	Always 0.	
		05	T	PCU Positioning Completed Flag	 ON when a PCU operation executed in response to a positioning command is com- pleted and the axis position is within the positioning comple- tion range set for the target position. 1: Any of the following opera- tions are completed: Direct operation (ABSO- LUTE MOVEMENT, REL- ATIVE MOVEMENT, REL- ATIVE MOVEMENT, INTERRUPT FEEDING), ORIGIN SEARCH, ORI- GIN RETURN The flag turns OFF (0) when the power is turned ON, a connection is established, or when another axis operation is started. This flag remains OFF if EMERGENCY STOP or DECELERATION STOP are executed while the axis is operating. 	4-8-3 Axis Control Status Flags 8-2 Origin Search Operation 8-4 Origin Return 9-4 Using Direct Operation

I/O	Word	Bits	Category	Name	Function	Details
Input (PCU	b	06	Axis Con- trol Status	No Origin Flag	ON when the origin is not established.	4-8-3 Axis Control Status Flags
to CPU Unit)			Flags			8-2 Origin Search Operation
						8-3 Present Position Preset
		07		Origin Stop Flag	ON when the present position of the axis is stopped within	4-8-3 Axis Control Status Flags
					the origin range set in <i>Zero</i> <i>Point Width</i> .	8-2 Origin Search Operation
		08 to 10		Not used (reserved by the system).	Always 0.	
		11		Warning Flag	ON when an axis warning has occurred.	12-1 Overview of PCU Errors
		12	1	Error Flag	ON when an axis error has occurred.	
		13		Busy Flag	ON during axis operation or processing.	4-8-3 Axis Control Status Flags
		14		Servo Parameter Transferring Flag	ON during transfer of Servo Parameters.	5-3 Transferring Servo Parameters
		15		Stop Execution Flag	ON when axis operation is stopped in response to EMERGENCY STOP or DECELERATION STOP.	4-8-3 Axis Control Status Flags 10-9 Stop Functions

I/O	Word	Bits	Category	Name	Function	Details
Input (PCU	b+1	00 to 02	Servo Sta- tus Flags	Reserved by the sys- tem.	Used by the PCU system.	
to CPU Unit)		03		Servo ON (SVON) Flag	ON while Servo Drive is in Servo lock status.	4-8-4 Servo Status Flags 10-1 Servo Lock/
						Unlock
		04		Main Power ON Flag (PON)	Indicates the status of the main circuit power supply to the Servo Drive.	4-8-4 Servo Status Flags
		05 and 06		Reserved by the sys- tem.	Used by the PCU system.	
		07		Positioning Com- pleted (PSET) Flag/ Speed Conformity (V-CMP) Flag	The function of these flags depends on whether position control or speed control is being used.	4-8-4 Servo Status Flags 10-5-2 Starting Speed Control
		08		Distribution Com- pleted (DEN) Flag/ Zero Speed (ZSPD) Flag	For details, refer to the rele- vant sections.	
		09		Torque Limit (T_LIM) Status Flag	ON when the torque limit function is enabled and activated.	4-8-4 Servo Status Flags 10-4-3 Torque Limits Set by Operating Commands
		10		Reserved by the sys- tem.	Used by the PCU system.	
		11	*	Positioning Proxim- ity (NEAR) Flag/ Speed Limit (V_LIM) Status Flag	The function of these flags changes according to whether position control or torque con- trol is being used.	4-8-4 Servo Status Flags 10-6-2 Starting Torque Control
					For details, refer to the rele- vant sections.	
		12		Forward Software Limit Flag (P_SOT)	Indicates whether the present position is within the forward software limit range.	4-8-4 Servo Status Flags 10-8 Software Limits
		13		Reverse Software Limit Flag (N_SOT)	Indicates whether the present position is within the reverse software limit range.	
		14 to 15		Reserved by the system.	Used by the PCU system.	
	b+2			Reserved by the sys- tem.	Used by the PCU system.	

I/O	Word	Bits	Category	Name	Function	Details
Input (PCU	b+3	00	External I/O status	Forward rotation limit input	Returns the status of I/O sig- nals for each axis.	4-8-5 External I/O Status Bits
to CPU Unit)		01	Ī	Reverse rotation limit	1: Signal valid	
		02	ł	Origin provimity input		
		02	4		The external I/O status shows	
		03		input	signals used for PCU control.	
		04		Encoder Phase B input	This status is different from the electrical ON/OFF status	
		05		Encoder Phase Z input	or the 1/O signals.	
		06		External latch signal 1 input		
		07		External latch signal 2 input		
		08		External latch signal 3 input		
		09	1	Brake output		
		10		Emergency stop input		
		11 to 15		Reserved by the sys- tem.		
	b+4		Errors	Axis error code	Returns the error code for the axis error.	12-4 Error Codes
	b+5	00 to 03	Expanded monitoring	Monitor 1 type	Indicates the monitor type for the present monitor 1 (b+10, b+11).	4-8-6 Expanded Monitoring
		04 to 07		Monitor 2 type	Indicates the monitor type for the present monitor 2 (b+12, b+13).	
		08 to 15	1	Not used (reserved by the system).	Always 0.	
	b+6		Present position	Feedback present position (rightmost word)	Returns the feedback present position.	7-3 Coordinate Sys- tem and Present Position
	b+7			Feedback present position (leftmost word)		
	b+8			Command present position (rightmost word)	Returns the command present position.	
	b+9			Command present position (leftmost word)		

I/O	Word	Bits	Category	Name	Function	Details	
Input (PCU	b+10		Expanded monitoring	Monitor 1 (rightmost word)	Returns the value for the mon- itor specified as monitor 1	4-8-6 Expanded Monitoring	
to CPU Unit)	b+11			Monitor 1 (leftmost word)	type in the Axis Operating Output Memory Areas.		
	b+12			Monitor 2 (rightmost word)	Returns the value for the mon- itor specified as monitor 2		
	b+13			Monitor 2 (leftmost word)	type in the Axis Operating Output Memory Areas.		
	b+14		Servo Parameter	Read data (rightmost word)	Returns the data read to the CPU Unit from the Servo	5-3-2 Reading Servo Parameters	
b·	b+15		data	Read data (leftmost word)	Drive.		
	b+16to b+23			Not used (reserved by the system).	Always 0.		
	b+24	00	Linear inter- polation status	Linear interpolation setting completed (See note.)	Indicates when setting the Position Control Unit has been completed for linear interpola- tion.	9-7 Linear Interpola- tion	
					1: Linear interpolation setting completed.		
		01 to 12		Not used (reserved by the system).	Always 0.		
		13		Linear interpolation executing (See note.)	Indicates when linear interpo- lation is being performed.		
		14 and 15		Not used (reserved by the system).	Always 0.		

Note

Allocated in Axis Operating Input Memory Areas for axis 1 and axis 5 for Position Control Unit Ver. 1.1 or later. These bits are not used in the Axis Operating Output Memory Areas for other axes.

4-8-3 Axis Control Status Flags

The Axis Control Status Flags in word b indicate the control status for each axis connected to the PCU. Use these flags to interface with the CPU Unit when using PCU functions.

Name	Word	Bits	Contents	
Receiving Com-	b	00	0: Command reception enabled.	
mand Flag			$0 \rightarrow 1$: Command reception started.	
			1: Receiving command (command recep- tion disabled).	
PCU Positioning Completed Flag		05	$0 \rightarrow 1$: Positioning completed.	
No Origin Flag		06	0: Origin established.	
			1: Origin not established.	
Origin Stop Flag		07	0: Outside origin range.	
			1: Within origin range.	
Warning Flag		11	0: No warning.	
			1: Warning has occurred.	
Error Flag		12	0: No axis error.	
			1: Axis error has occurred.	
Busy Flag		13	1: Axis busy (axis operation executing).	
Servo Parameter Transferring Flag		14	0: Powerup, transfer completed, or transfer failed.	
			1: Servo Parameter transferring.	
Stop Execution Flag		15	$0 \rightarrow 1$: Stop operation completed.	

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

The flags that are particularly for controlling axis operation are described next. For details on timing charts for each flag operation, refer to the relevant section.

For details on the Error Flag and Warning Flag, refer to *SECTION 12 Trouble-shooting*. For details on the Servo Parameters Transferring Flag, refer to *5-3 Transferring Servo Parameters*.

Receiving Command Flag

Word	b
Bit	00
Function	This flag turns ON when the PCU is in the process of receiving any of the fol- lowing operating commands.
	ABSOLUTE MOVEMENT (including INTERRUPT FEEDING), RELATIVE MOVEMENT (including INTERRUPT FEEDING), ORIGIN SEARCH, ORI- GIN RETURN, PRESENT POSITION PRESET, JOG (JOG Bit ON), DE- CELERATION STOP, EMERGENCY STOP, SERVO LOCK, SERVO UNLOCK, SPEED CONTROL, TORQUE CONTROL, DEVICE SETUP, ERROR RESET, or DEVIATION COUNTER RESET.

The PCU turns ON the Receiving Command Flag when an operating command is received from the CPU Unit, and the flag remains ON for at least CPU Unit cycle time. The flag turns OFF when the command reception processing is completed.

If another operating command with the same priority level is started while this flag is ON, a Multistart Error (axis error code: 3050) will occur. (Refer to *4-7-3 Axis Operating Output Memory Area Priority*.)



The Receiving Command Flag remains ON for either the length of one CPU Unit cycle time or for the time required by the PCU to process command reception, whichever is longer.

The Receiving Command Flag status does not change when changing the target speed during position control/speed control, changing the torque command value during torque control, switching override enable/disable, changing the override value, or transferring Servo Parameters.

The Receiving Command Flag is used mainly to determine ON/OFF timing of the operating command bits when sending more than one operating command at the same time during axis operation (i.e., while the Busy Flag is ON), such as changing the target position while using direct operation.

PCU Positioning Completed Flag

Word	b
Bit	05

Function

This flag turns ON when a PCU operation is completed normally in response to any of the following positioning commands and the feedback present position of the axis has reached a position within the range of the Servo Parameter *Positioning Completion Range 1* (Pn431 for R88D-KN, Pn060 for R88D-GN, Pn500 for R88D-WT, and Pn522 for R88D-WN-ML2 and SMART-STEP Junior) specified for the target position (command position). (The position deviation must be within the range of *Positioning Completion Range 1*.)

ABSOLUTE MOVEMENT (including INTERRUPT FEEDING), RELATIVE MOVEMENT (including INTERRUPT FEEDING), ORIGIN SEARCH, or ORIGIN RETURN.



Example: Executing RELATIVE MOVEMENT COMMAND.

The PCU Positioning Completed Flag will not turn ON when axis operation stops under the following conditions.

- Stops due to an error during positioning.
- Stops due to DECELERATION STOP or EMERGENCY STOP sent during positioning.
- Jogging (JOG) stops.

The PCU Positioning Completed Flag will also not turn ON when axis operation stops during positioning due to DECELERATION STOP or EMERGENCY STOP, even if the position is within the *Positioning Completion Range 1* specified for the original target position.

The PCU Positioning Completed Flag will turn OFF when the power is turned ON, a connection is released, another axis operation is started successfully (including jogging, speed control, and torque control), or Servo Unlock is executed.

Note On machines for which position deviation occurs between the feedback position and target position (command position) when the axis is stopped, such as pushing a load or using a vertical axis, the PCU Positioning Completed Flag will not turn ON if the set value for the *Positioning Completion Range 1* is too small. Adjust the set value for the *Positioning Completion Range 1* to suit the operating conditions for the machine.

<u>No Origin Flag</u>		
Word		b
Bit		06
Function		This flag turns ON when the origin is not established. Except when using Servomotors with absolute encoders, the PCU's No Origin Flag turns ON immediately after a connection is established, at which time the origin is not established. (The No Origin Flag turns OFF, however, immediately after the PCU power is turned ON or the Unit is restarted, and the flag remains OFF until the first CONNECT is executed.) The origin for each axis is determined by either of the following origin positioning operations, after which the No Origin Flag turns OFF. ORIGIN SEARCH, PRESENT POSITION PRESET
		 When using Servomotors with absolute encoders, the origin is established by reading the absolute value data when establishing a connection or executing SERVO LOCK. The established origin is held even if SERVO UNLOCK is executed, and will not be set to no origin status. No origin status (No Origin Flag ON) occurs under the following conditions. When DEVICE SETUP is executed (during Servo unlock status) When a connection is released (See note.)
	Note	When a connection is released (including a disconnection due to error), the status for each axis becomes undefined (non-execution), except for the Error Flag and axis error code. The No Origin Flag turns ON while in no origin (origin not established) status.
Origin Stop Flag		
Word		b
Bit		07
Function		This flag turns ON when the origin has been established and the present position of each axis is within the <i>Origin Range</i> set in Servo Parameter Pn105 for the R88D-G□ or <i>Zero Point Width</i> set in Servo Parameter Pn803 for the R88D-KN□, the R88D-WT□, the R88D-WN□-ML2, or a SMARTSTEP Junior Servo Drive.
		This flag turns ON for the interval in which the present position is within the <i>Origin Range</i> or <i>Zero Point Width</i> when it passes the <i>Origin Range</i> or <i>Zero Point Width</i> during axis operation.
		Origin range (Pn105) for G Series Zero point width (Pn803) for G5 Series, W Series, SMARTSTEP Junior
		Forward direction
		Established origin Origin range (Pn105) for G Series Zero point width (Pn803) for G5 Series, W Series, SMARTSTEP Junior
		The Origin Stop Flag turns ON while the present position is within the range of \pm Pn105 for G-series Servo Drives or \pm Pn803 for G5-Series, W-series and SMARTSTEP Junior Servo Drives from the established

origin when traveling forward.

<u>Busy Flag</u>			
Word	b		
Bit	13		
Function	This flag turns ON to indicate that the Position Control Unit is executing one of the following operating commands. The flag will turn ON when the operation command is started and remain ON for at least one cycle time of the CPU Unit. For commands associated with axis operations, the Busy Flag remains ON during the axis operation. ABSOLUTE MOVEMENT (including INTERRUPT FEEDING), RELATIVE MOVEMENT (including INTERRUPT FEEDING), ORIGIN SEARCH, ORI- GIN RETURN, PRESENT POSITION PRESET, JOG (JOG ON, JOG OFF), DECELERATION STOP, EMERGENCY STOP, SERVO LOCK, SERVO UNLOCK, SPEED CONTROL, TORQUE CONTROL, DEVICE SETUP, ERROR RESET, or DEVIATION COUNTER RESET.		
	For operating commands associated with tus when the axis operation is completed ing command, as follows:	h axis operations, the Busy Flag sta- I (axis stops) depends on the operat-	
	 ABSOLUTE MOVEMENT, RELATIVE ORIGIN RETURN: The Busy Flag turns OFF when the Servo Parameter <i>Positioning Complet</i> Pn060 for R88D-GN, Pn500 for WN -ML2 and SMARTSTEP Junior Busy Flag turns OFF at the same tim Flag turns ON.) Jogging Stopped (JOG Bit OFF), DE STOP (Including Error Stop), or DEV The Busy Flag turns OFF regardless the command position reaches the ta ing the command to the Servo Drive 	E MOVEMENT, ORIGIN SEARCH, or e axis feedback position reaches the etion Range 1 (Pn431 for R88D-KN, R88D-WT, and Pn522 for R88D-) set for the target stop position. (The ie as the PCU Positioning Completed ECELERATION STOP, EMERGENCY (IATION COUNTER RESET: s of the axis feedback position when arget (stop) position (i.e., when send- is completed).	
	Speed	- Solid line: Command speed 	
	Busy Flag		
	The Busy Flag turns OFF at comple- tion of sending the command (i.e., when the command present position stops changing) when the axis stops due to jogging stop (JOG Bit OFF), DECELERATION STOP, or EMER- GENCY STOP (including error stop).	When positioning for ABSOLUTE MOVEMENT, RELATIVE MOVEMENT, ORIGIN SEARCH, or ORIGIN RETURN stops, the Busy Flag will not turn OFF until the feedback present position is within the <i>Positioning Completion Range</i> for the command position.	

Stop Execution Flag	
Word	b
Bit	15
Function	This flag turns ON in response to a DECELERATION STOP or EMERGENCY STOP execution regardless of the axis operation status. If DECELERATION STOP or EMERGENCY STOP is executed during axis operation, the Stop Execution Flag turns ON when the command position reaches the target (stop) position (i.e., when sending the command to the Servo Drive is com- pleted) regardless of the axis feedback position. This flag then turns OFF when the power is turned ON, a connection is released, or when another axis operation is started.
Details	Refer to 10-9 Stop Functions.

4-8-4 Servo Status Flags

The Servo Status Flags (word b+1) are flags that monitor the control status of the Servo Drive. This enables monitoring of the Servo Drive control status (position, speed, torque control loop).

Name	Word	Bits	Contents	
Servo ON Flag	b+1	03	0: Servo unlocked.	
			1: Servo locked.	
Main Power ON Flag		04	0: Main circuit power supply OFF.	
			1: Main circuit power supply ON.	
Positioning Com- pleted Flag/Speed		07	During position control: Positioning Com- pleted Flag	
Conformity Flag			0: Sending position control command to Servo Drive or the position is outside <i>Positioning Completion Range 1</i> .	
			1: Sending position control command to Servo Drives completed and the posi- tion is within <i>Positioning Completion</i> <i>Range 1</i> .	
			During speed control: Speed Conformity Flag	
			0: Speed does not match the speed com- mand value for speed control.	
			 Speed matches the speed command value for speed control. 	
Distribution Com- pleted Flag/Zero		08	During position control: Distribution Com- pleted Flag	
Speed Flag			0: Sending position control command to Servo Drive.	
			1: Sending position control command to Servo Drive completed.	
			During speed control: Zero Speed Flag	
			0: Not zero speed.	
			1: Detecting zero speed.	
Torque limit		09	0: Torque limit function disabled.	
			1: Torque limit function enabled.	
Positioning Proxim- ity Flag/Speed Limit		11	During position control: Positioning Proxim- ity Flag	
Status Flag			0: Outside Positioning <i>Completion Range</i> 2	
			1: Within Positioning Completion Range 2	
			During torque control: Speed Limit Status Flag	
			0: Speed limit function disabled.	
			1: Speed limit function enabled.	
Forward Software		12	0: Within forward software limit.	
Limit Flag			1: Forward software limit exceeded.	
Reverse Software		13	0: Within reverse software limit.	
Limit Flag			1: Reverse software limit exceeded.	

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. –1) \times 25

	The functions of the bits shown in this table are applicable when the Position Control Unit is connected to a G5-series Servo Drive, a G-series Servo Drive, a W-series Servo Drive combined with a FNY-NS115 MECHATROLINK-II I/F Unit, a W-series Servo Drive with Built-in MECHATROLINK-II communica- tions, or to a SMARTSTEP Junior Servo Drive. The functions of Servo Status Flags 07, 08, and 11 depend on whether position control, speed control, or torque control is being used. (The SMARTSTEP Junior Servo Drive supports only position control.)
Note	The Servo Status Flags are used to monitor the control status of the Servo Drive. The Servo Status Flags may not reflect the Servo status for short-term changes in Servo Drive status, depending on the PLC cycle time and MECHATROLINK communications cycle.
Servo ON Flag	
Word	b+1
Bit	03
Function	ON while Servo Drive is in Servo lock status during position control or in Servo ON status (Servomotor carrying current) during speed or torque con- trol.
	0: Servo unlocked (Servo OFF)
	1: Servo locked (Servo ON)
	This flag turns ON when the SERVO LOCK Bit (a+1, bit 00) in the Axis Oper- ating Output Memory Area turns ON, and turns OFF when the SERVO UNLOCK Bit (a+1, bit 01) turns ON or an error that sets the Servo in free run state occurs.
	Servo ON Flag
Details	Refer to 10-1 Servo Lock/Unlock.
Main Power ON Flag	
Word	b+1
Bit	04
Function	Indicates the status of the main circuit power supply to the Servo Drive. 0: Main circuit power supply OFF
	1: Main circuit power supply ON
Positioning Completed	Flag (during Position Control) (during Speed Control)
Word	b+1
Bit	07
Function	The function of these flags depends on whether position control or speed con-

trol is being used.

During Position Control: Positioning Completed Flag

This flag turns ON when the command position reaches the target (stop) position (Sending position control command to Servo Drive completed), and the feedback position is within the range set for the target (stop) position Servo Parameter *Positioning Completion Range 1* (Pn431 for R88D-KN \square , Pn060 for R88D-GN \square , Pn500 for R88D-WT \square , and Pn522 for R88D-WN \square -ML2 and SMARTSTEP Junior).

0: Sending position control command to Servo Drive or the position is outside *Positioning Completion Range 1*.

1: Sending position control command to Servo Drive completed and the position is within *Positioning Completion Range 1*.

The Positioning Completed Flag in the Servo Status Flags is different to the PCU Positioning Completed Flag in the Axis Control Status Flags (word b, bit 05). In addition to when positioning stops, this flag turns ON when the number of pulses remaining in the Servo Drive's deviation counter are within the *Positioning Completion Range 1*, even for jogging stops, deceleration stops, emergency stops, and error stops.



During Speed Control: Speed Conformity Flag (G5-series Servo Drives, G-series Servo Drives and W-series Servo Drives Only)

This flag turns ON when the feedback speed during speed control matches the target speed (speed command value for speed control), i.e., the speed is within the range of the Servo Parameter *Speed Conformity Signal Output Width* (Pn435 for R88D-KN, Pn061 for R88D-GN and Pn503 for R88D-W).

0: Speed does not match the speed command value for speed control.

1: Speed does match the speed command value for speed control.





Details

Distribution Completed Flag (during Position Control) Zero Speed Flag (during Speed Control)

Word	b+1
Bit	08
Function	The function of these flags depends on whether position control or speed con- trol is being used.
	During Position Control: Distribution Completed Flag
	This flag turns ON when the command position reaches the target (stop)

This flag turns ON when the command position reaches the target (stop) position (When sending the position control command to Servo Drive is completed).

- 0: Sending the position control command to Servo Drive.
- 1: Sending the position control command to Servo Drive completed.



During Speed Control: Zero Speed Flag (G5-series Servo Drives, G-series Servo Drives and W-series Servo Drives Only)

This flag turns ON when the feedback speed during speed control drops below the Servo Parameter *Rotation Speed for Motor Rotation Detection* (Pn436 for R88D-KN□, Pn062 for R88D-GN□ and Pn502 for R88D-W□).

- 0: Not at zero speed
- 1: Detecting zero speed



Details

Refer to 10-5-2 Starting Speed Control.

Torque Limit Status Flag	
Word	b+1
Bit	09
Function	This flag turns ON when the Servomotor's output torque is limited by the torque limit.
	For W-series Servo Drives, the torque limit will be the lowest among all the limit values that are enabled for the torque limits specified in <i>Forward Torque Limit</i> (Pn402), <i>Reverse Torque Limit</i> (Pn403), <i>Forward Rotation External Current Limit</i> (Pn404), <i>Reverse Rotation External Current Limit</i> (Pn405), or in the option command value for speed control.
	For G5-series Servo Drives, the torque limit will be the lowest among all the limit values that are enabled for the torque limits specified in <i>No. 1 Torque Limit</i> (Pn013), <i>No.2 Torque Limit</i> (Pn522), <i>Forward External Torque Limit</i> (Pn525), <i>Reverse External Torque Limit</i> (Pn526) or in the option command value for speed control.
	For G-series Servo Drives, the torque limit will be the lowest among all the limit values that are enabled for the torque limits specified in <i>No.1 Torque Limit</i> (Pn05E), <i>No.2 Torque Limit</i> (Pn05F), or in the option command value for speed control.
	0: Torque limit function disabled
	1: Torque limit function enabled
Details	Refer to 10-4 Torque Limits.
Desitioning Drevinsity Ele	
Speed Limit Status Flag	ng (during Position Control) (during Torque Control)
Speed Limit Status Flag	n g (during Position Control) (during Torque Control) b+1
Speed Limit Status Flag Word Bit	ig (during Position Control) (during Torque Control) b+1 11
Speed Limit Status Flag Word Bit Function	b+1 The function of these flags depends on whether position control or torque control is being used.
Speed Limit Status Flag Word Bit Function	Image (during Position Control) (during Torque Control) b+1 11 The function of these flags depends on whether position control or torque control is being used. During Position Control: Positioning Proximity Flag
Speed Limit Status Flag Word Bit Function	Image (during Position Control) (during Torque Control) b+1 11 The function of these flags depends on whether position control or torque control is being used. During Position Control: Positioning Proximity Flag This flag turns ON when the feedback position is within the range for Servo Parameter Positioning Completion Range 2 (Pn442 for R88D-KN□-ML2, Pn504 for R88D-WT□ and Pn524 for R88D-WN□-ML2 and SMARTSTEP Junior) for the target (stop) position.
Speed Limit Status Flag Word Bit Function	Image (during Position Control) (during Torque Control) b+1 11 The function of these flags depends on whether position control or torque control is being used. During Position Control: Positioning Proximity Flag This flag turns ON when the feedback position is within the range for Servo Parameter Positioning Completion Range 2 (Pn442 for R88D-KN□-ML2, Pn504 for R88D-WT□ and Pn524 for R88D-WN□-ML2 and SMARTSTEP Junior) for the target (stop) position. 0: Outside Positioning Completion Range 2
Speed Limit Status Flag Word Bit Function	b+1 11 The function of these flags depends on whether position control or torque control is being used. During Position Control: Positioning Proximity Flag This flag turns ON when the feedback position is within the range for Servo Parameter <i>Positioning Completion Range 2</i> (Pn442 for R88D-KN□-ML2, Pn504 for R88D-WT□ and Pn524 for R88D-WN□-ML2 and SMARTSTEP Junior) for the target (stop) position. 0: Outside <i>Positioning Completion Range 2</i> 1: Within <i>Positioning Completion Range 2</i>
Speed Limit Status Flag Word Bit Function	ing (during Position Control) (during Torque Control) b+1 11 The function of these flags depends on whether position control or torque control is being used. During Position Control: Positioning Proximity Flag This flag turns ON when the feedback position is within the range for Servo Parameter Positioning Completion Range 2 (Pn442 for R88D-KN□-ML2, Pn504 for R88D-WT□ and Pn524 for R88D-WN□-ML2 and SMARTSTEP Junior) for the target (stop) position. 0: Outside Positioning Completion Range 2 1: Within Positioning Completion Range 2 During Torque Control: Speed Limit Status Flag
Speed Limit Status Flag Word Bit Function	g (during Position Control) (during Torque Control) b+1 11 The function of these flags depends on whether position control or torque control is being used. During Position Control: Positioning Proximity Flag This flag turns ON when the feedback position is within the range for Servo Parameter Positioning Completion Range 2 (Pn442 for R88D-KN□-ML2, Pn504 for R88D-WT□ and Pn524 for R88D-WN□-ML2 and SMARTSTEP Junior) for the target (stop) position. 0: Outside Positioning Completion Range 2 1: Within Positioning Completion Range 2 During Torque Control: Speed Limit Status Flag This flag turns ON when the feedback speed during torque control is limited by the Servo Parameter Speed Limit (Pn407), or option command value.
Speed Limit Status Flag Word Bit Function	g (during Position Control) (during Torque Control) b+1 11 The function of these flags depends on whether position control or torque control is being used. During Position Control: Positioning Proximity Flag This flag turns ON when the feedback position is within the range for Servo Parameter Positioning Completion Range 2 (Pn442 for R88D-KNML2, Pn504 for R88D-WT□ and Pn524 for R88D-WN□-ML2 and SMARTSTEP Junior) for the target (stop) position. 0: Outside Positioning Completion Range 2 1: Within Positioning Completion Range 2 1: Within Positioning Completion Range 2 During Torque Control: Speed Limit Status Flag This flag turns ON when the feedback speed during torque control is limited by the Servo Parameter Speed Limit (Pn407), or option command value. 0: Speed limit function disabled
Speed Limit Status Flag Word Bit Function	g (during Position Control) (during Torque Control) b+1 11 The function of these flags depends on whether position control or torque control is being used. During Position Control: Positioning Proximity Flag This flag turns ON when the feedback position is within the range for Servo Parameter Positioning Completion Range 2 (Pn442 for R88D-KN□-ML2, Pn504 for R88D-WT□ and Pn524 for R88D-WN□-ML2 and SMARTSTEP Junior) for the target (stop) position. 0: Outside Positioning Completion Range 2 1: Within Positioning Completion Range 2 During Torque Control: Speed Limit Status Flag This flag turns ON when the feedback speed during torque control is limited by the Servo Parameter Speed Limit (Pn407), or option command value. 0: Speed limit function disabled 1: Speed limit function enabled

Forward Software Limit Flag and Reverse Software Limit Flag

Word	b+1		
Bits	12 and 13		
Function	These flags turn ON and OFF when the software limit function is enabled to indicate the present position status for the forward/reverse software limit. Forward Software Limit Flag 0: Within forward software limit range 1: Forward software limit range exceeded Reverse Software Limit Flag 0: Within reverse software limit range		
	1: Reverse software limit range exceeded		
Reverse direction Forward Software Limit Fla (b+1, bit 12) Reverse Software Limit Fla (b+1, bit 13)	Reverse Software Limit (Pn806) Forward Software Limit (Pn804) Feedback position g g		

Details

Refer to 10-8 Software Limits.

4-8-5 External I/O Status Bits

The External I/O Status Bits (word b + 3) show the I/O status of the I/O signals allocated to the Servo Drive.

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

I/O	Word	Bits	Category	Name	Function
Input	b+3	00	External I/O	Forward rotation limit input	Returns the status of I/O signals
(PCU		01	status	Reverse rotation limit input	for each axis.
Unit)		02	Ī	Origin proximity input signal	1: Signal valid
,		03	Ī	Encoder Phase A input	The external I/O status shows the
		04	Ī	Encoder Phase B input	enabled/disabled status of signals
		05	Ī	Encoder Phase Z input	used for PCU control.
		06	Ī	External latch signal 1 input	This status is different from the
		07	Ī	External latch signal 2 input	signals.
		08	Ī	External latch signal 3 input	
		09	Ī	Brake output]
		10	Ī	Emergency stop input]
		11 to 15	Ī	Reserved by the system.]

Bit 10 (Emergency stop input) indicates the status of the emergency stop input signal input when using a G5-series Servo Drive, G-series Servo Drive or SMARTSTEP Junior Servo Drive. This bit is always 0 when using a W-series Servo Drive, which does not have an emergency stop signal.

The SMARTSTEP Junior Servo Drive does not have inputs for external latch signals 2 and 3. When using a SMARTSTEP Junior Servo Drive, the input status for these signals will always be 0.

When using a G5-series Servo Drive or W-series Servo Drive, the allocation and polarity of Servo Drive input signals can be set in the Servo Parameters, but the external I/O status depends on the polarity of the Servo Drive's input signals, as follows:

External I/O status	Signal	Servo Drive I/O signal name	Signal polarity: Normal (Setting: 0 to 6)		Signal p Reverse (Se	oolarity: tting: 9 to F)
			Sensor status: Base block (not receiving power; Input open)	Sensor status: In operation (Input closed)	Sensor status: Base block (not receiving power; Input open)	Sensor status: In operation (Input closed)
Forward rota- tion limit input	POT	Forward drive pro- hibited	1	0	0	1
Reverse rota- tion limit input	NOT	Reverse drive pro- hibited	1	0	0	1
Origin proximity input	DEC	Origin return decel- eration limit switch	0	1	1	0
External latch 1 input	EXT1	External latch 1 input	0	1	1	0
External latch 2 input	EXT2	External latch 2 input	0	1	1	0
External latch 3 input	EXT3	External latch 3 input	0	1	1	0

The forward rotation limit input (forward drive prohibited signal), and reverse rotation limit input (reverse drive prohibited signal) are normally-closed contact inputs (N.C. inputs) when polarity is normal. The origin proximity input (origin return deceleration limit switch), and external latch inputs (1 to 3) are normally-open contact inputs (N.O. inputs) when polarity is normal.

When using a SMARTSTEP Junior Servo Drive, the following status is shown. (With the SMARTSTEP Junior, signal polarity is fixed.)

External I/O status	Signal	Servo Drive I/O signal name	Sensor status: Base block (not receiving power; Input open)	Sensor status: In operation (Input closed)
Forward rotation limit input	POT	Forward drive prohibited	1	0
Reverse rotation limit input	NOT	Reverse drive prohibited	1	0
Origin proximity input	DEC	Origin proximity input	0	1
External latch 1 input	EXT1	External latch 1 input	0	1
Emergency stop input	STOP	Emergency stop input	1	0

4-8-6 Expanded Monitoring

The PCU has a fixed monitoring function that monitors the following control information for each axis and returns the result to the Axis Operating Input Memory Areas.

Axis control status (word b)

Servo status (word b + 1)

External I/O status (word b + 3)

Present position (Feedback present position: Words b + 6, b + 7; Command present position: Words b + 8, b + 9)

The PCU also has an expanded monitoring function that monitors control information selected from a number of categories, and returns the result to the Axis Operating Input Memory Areas.

The expanded monitoring function provides two monitors in the Axis Operating Input Memory Areas (Monitor 1: Words b + 10, b + 11; Monitor 2: Words b + 12, b + 13). The information to be monitored by each is selected using Monitor 1 type (Word a + 15, bits 00 to 03) and Monitor 2 type (Word a + 15, bits 04 to 07) in the Axis Operating Output Memory Areas.

Axis Operating Output Memory Area (Operating Commands)

Name	Word	Bits	Contents
Monitor 1 type	a+15	00 to 03	Used to select the type of control informa- tion to be returned to monitor 1.
Monitor 2 type		04 to 07	Used to select the type of control informa- tion to be returned to monitor 2.
Reserved by the sys- tem.		08 to 15	Not used. The setting is disabled.

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) \times 25

Axis Operating Input Memory Area (Monitoring)

Name	Word	Bits	Contents
Monitor 1 type	b+15	00 to 03	Indicates the type of control information selected by Monitor 1.
Monitor 2 type		04 to 07	Indicates the type of control information selected by Monitor 2.
Not used.		08 to 15	Always 0.
Monitor 1	b+10 b+11		Monitor 1 (rightmost word) Monitor 2 (leftmost word)
Monitor 2	b+12 b+13		Monitor 1 (rightmost word) Monitor 2 (leftmost word)

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) \times 25

Monitor Type

The following control information can be selected for Monitor 1 and Monitor 2.

Monitor Type	Control information	Unit	Details
0	Command present position	Command unit	Returns the present position being sent to the Servo Drive.
			This value is the same as the com- mand present position in words $b + 8$ and $b + 9$.
1	Reserved by the system.		Do not set.
2	Position devia- tion	Command unit	Returns the number of pulses remain- ing in the deviation counter of the Servo Drive in command units.
			Position deviation = Command present position – Feedback present position
3	Feedback present position	Command unit	Returns the actual position of the machine based on feedback from the Servomotor's encoder.
			This value is the same as the feedback present position in words $b + 6$ and $b + 7$.

Monitor Type	Control information	Unit	Details
4 to 5	Reserved by the system.		Do not set.
6	Target position	Command unit	Returns the target position (position command value) during positioning.
7	Reserved by the system.		Do not set.
8	Feedback speed	See note 1.	Returns the speed of the machine based on feedback from the Servomo-tor's encoder.
9	Command speed	See note 1.	Returns the speed designated for the Servo Drive.
A	Target speed	See note 1.	Returns the designated target speed (speed command value).
В	Torque com- mand	See note 2.	Returns the torque command for the Servo Drive.
C to F	Reserved by the system.		Do not set.

Note

(1) Speed Monitoring Unit

When the speed (feedback speed, command speed, or target speed) is selected for the monitor type, the monitoring unit for position control/ torque control, and speed control is as follows:

Control mode	Speed monitoring unit	Unit change timing
Position control Torque control	Command units/s	The unit changes when the command to switch to position control or torque con- trol is received (when the Receiving Command Flag turns ON).
Speed control	0.001% (percentage of Servomotor's momentary maxi- mum rotation speed)	The unit changes when the command for SPEED CONTROL is received (when the Receiving Command Flag turns ON).

For further details on operation timing, refer to 10-5 Speed Control.

(2) Torque Monitoring Unit

When the torque is selected for the monitor type, the monitoring unit for position control, speed control, and torque control is as follows:

Control mode	Torque monitoring unit	Unit change timing
Position control Speed control	% (1% unit) (percentage of Servomotor's rated torque)	The unit changes when the command to switch to position control or speed con- trol is received (when the Receiving Command Flag turns ON).
Torque control	0.001% (percentage of Servomotor's momentary maxi- mum torque)	The unit changes when the command for TORQUE CONTROL is received (when the Receiving Command Flag turns ON).

For further details on operation timing, refer to 10-6 Torque Control.

Note

(1) Control data is used to monitor the control status of the Servo Drive. The resolution of the monitor value depends on the specifications of the connected Servo Drive. Example: If a W-series Servo Drive equipped with a FNY-NS115 MECHATROLINK-II I/F Unit is connected, the minimum resolution of the monitor values for feedback speed and command speed during position/torque control is 1,000 command units/s.

Depending on the length of the PLC cycle and MECHATROLINK communications cycle short-term changes of the Servo Drive status may not be monitored successfully.

(2) If MECHATROLINK communications are stopped (connection released) during speed or torque control and MECHATROLINK communications are restarted to execute SERVO LOCK (Servo ON), the power to the Servomotor will be recovered with the previous control mode maintained. At this point, the PCU cannot obtain correct monitor values for speed or torque monitor in its expanded monitoring.

To stop MECHATROLINK communications during SPEED CONTROL or TORQUE CONTROL execution, either execute DECELERATION STOP or EMERGENCY STOP, or execute SERVO UNLOCK, change to the position control mode, and then release the connection.

SECTION 5 Transferring and Saving Data

This section explains how to transfer and save parameters and data using the data transfer bits.

5-1	Transferring Data						
5-2	Transferring PCU Parameters						
	5-2-1	Writing PCU Parameters	220				
	5-2-2	Reading PCU Parameters	223				
	5-2-3	Saving PCU Parameters	225				
5-3	Transfe	rring Servo Parameters	226				
	5-3-1	Writing Servo Parameters	227				
	5-3-2	Reading Servo Parameters	229				
	5-3-3	Saving Servo Parameters	231				
	5-3-4	Device Setup	234				

5-1 Transferring Data

The following two types of data can be transferred by the Position Control Unit (PCU).

- 1,2,3... 1. PCU Parameters
 - Common Parameters
 - Axis Parameters
 - 2. Servo Drive Parameters
 - Servo Parameters

There are two ways to transfer these parameters.

- Using the Support Software to transfer parameters to and from a Windows computer connected to the CPU Unit
- Using the bits in the Operating Memory Areas allocated to the PCU in the CPU Unit to transfer parameters to and from I/O Memory in the CPU Unit

The CX-Motion-NCF Support Software can be used on a Windows computer to set and transfer parameters for the PCU, to save and print the parameters, and to monitor PCU status.



Refer to the *CX-Motion-NCF Operation Manual* (Cat. No. W436) for operating procedures and other information on the CX-Motion-NCF.

Bits in the Operating Memory Areas can be manipulated to transfer data between the CPU Unit and the PCU (and Servo Drives). The Common Operating Memory Area is used for parameters held in the PCU and the Axis Operating Memory Areas are used for parameters held in the Servo Drives.

Common Parameters and Axis Parameters are saved in the PCU's internal memory and also in the PCU's flash memory. These parameters are read/ written, or saved to flash memory from the CPU Unit using the WRITE DATA, READ DATA, and SAVE DATA Bits allocated in the Common Operating Memory Area.

The number of transfer words, transfer source, and transfer destination are specified in the Common Operating Memory Area. It is possible to transfer multiple parameters at once.

For PCU parameters, the parameters stored in memory are saved altogether to the PCU's internal flash memory.

Transferring Parameters Using Support Software

Transferring Parameters Using Bits in Operating Memory Areas

Transferring PCU Parameters



Transferring Servo Drive Parameters

Servo Parameters are stored in internal memory of the Servo Drive connected to the PCU and then saved in the Servo Drive's non-volatile memory. These parameters are read/written or saved to the non-volatile memory from the CPU Unit using the WRITE SERVO PARAMETER, READ SERVO PARAME-TER, and SAVE SERVO PARAMETER Bits allocated in the Axis Operating Memory Areas.

Servo Parameters for a single axis are written and read by parameter by specifying the parameter number. Parameters for different axes can be transferred at the same time.

Servo Parameters are saved by executing WRITE DATA to write them to the Servo Drive's non-volatile memory. Therefore, SAVE SERVO PARAMETER must be executed for each parameter separately from writing them to the Servo Drive's memory (WRITE SERVO PARAMETER).

Two types of Servo Parameters are transferred to the Servo Drive: online parameters with set values that are enabled immediately after they are written, and offline parameters, which are not enabled just by writing them. To enable set values for offline parameters in Servo Drive operations, save the parameters to the non-volatile memory, and then cycle the Servo Drive power. Alternatively, execute WRITE SERVO PARAMETER/SAVE SERVO PARAMETER and then execute the Servo Drive's DEVICE SETUP.

When DEVICE SETUP is executed, the Servo Drive enables all the Servo Parameters that are currently being transferred, and initializes the present position and output signal status to set the Servo Drive.



5-2 Transferring PCU Parameters

5-2-1 Writing PCU Parameters

The PCU's Common Parameters and Axis Parameters are written from the CPU Unit using the WRITE DATA Bit in the Common Operating Memory Area. The data to be transferred is set by specifying the number of transfer words, transfer source, and transfer destination in the Common Operating Memory Area. Multiple parameters can also be transferred all at once.

The PCU reads the contents of the Common Parameters and Axis Parameters saved in the internal flash memory when the PCU power is turned ON or the Unit is restarted, and enables them as operating parameters. Parameters transferred to the PCU using the WRITE DATA Bit will be lost if the PCU power is turned OFF or the Unit is restarted.

Note

- (1) The Common Parameter set values are enabled by cycling the PCU power or restarting the Unit after executing WRITE DATA. Therefore, when the WRITE DATA Bit is used to change the contents of the Common Parameters, execute SAVE DATA and save the Common Parameters to the PCU's internal flash memory.
 - (2) The Axis Parameter settings are enabled when WRITE DATA execution has completed. To change Axis Parameter settings permanently, execute WRITE DATA followed by SAVE DATA (while MECHATROLINK communications are stopped) to save the Axis Parameter settings in the PCU's internal flash memory.
 - (3) Execute WRITE DATA to write data to the PCU while the Data Transferring Flag is OFF. Writing data to the PCU is disabled while the Busy Flag

is ON (when the PCU is in initial processing or reading or writing the backup data from or to the Memory Card).

- (4) If data is written while different data is being transferred, a Multistart Error (Unit error code: 0021) will occur and data will not be written to the PCU.
- (5) Do not restart the PCU or turn OFF the power to the PLC while data is being written. The data will not be transferred correctly, and normal PCU operations may not be possible.

Data writing starts when the WRITE DATA Bit in the Common Operating Memory Area turns ON. When writing data to the PCU, make sure the WRITE DATA Bit remains ON until the Data Transferring Flag turns ON.

Up to 600 words can be transferred during a single data write operation. The maximum amount of data that can be transferred during a single CPU Unit cycle, however, is 20 words. When more than 20 words of data are transferred, the data is divided and WRITE DATA is executed over multiple cycles. When writing data to the PCU is completed, the Data Transferring Flag turns OFF.

Common Operating Memory Area (Output)

Name	Word	Bit	Contents
WRITE DATA Bit	n	01	$0 \rightarrow 1$: Starts writing data.
Number of write words	n+6		Specifies the number of words to be written to the PCU. Up to 600 words can be transferred.
Write source area	n+7		Specifies the area of the data to be written
Write source word	n+8		to the PCU.
			The type of area set as the write source area is specified by specifying the begin- ning word of the area in the write source word in hexadecimal.
			Write source area: 00B0 hex: CIO Area 00B1 hex: Work Area 00B2 hex: Holding Area 00B3 hex: Auxiliary Area 0082 hex: DM Area 005 hex: EM Area : EM bank No.= 0 to 9, A, B, C
Write destination address	n+9		Specifies the write destination address in the PCU.

 $n = CIO 1500 + (unit number \times 25)$

Common Operating Memory Area (Input)

Name	Word	Bit	Contents
Data Transferring Flag	n+15	14	0: Powerup, transfer completed, or transfer failed
			1: Transferring data
Unit Busy Flag	n+16	14	0: PCU is not busy
			1: PCU is busy

 $n = CIO 1500 + (unit number \times 25)$

The following diagram shows the operation for writing data to the PCU.

Transferring PCU Parameters

Section 5-2



Timing Chart for Writing Data to the PCU

Data can be written to the PCU during axis operation if the Unit Busy Flag is not ON (i.e., except when the PCU is initializing or reading/writing Memory Card backup data). WRITE DATA, READ DATA, and SAVE DATA cannot be executed at the same time.



When writing data to the PCU is completed, the Data Transferring Flag (word n+15, bit 14) turns OFF. When an error occurs during data transfer, the Unit Error Flag (word n+15, bit 12) turns ON, and the corresponding error code is input in word n+21 of the Common Operating Memory Area. Check the error code and perform appropriate troubleshooting.

Note If an error occurs while writing data (multistart error, write transfer error, Common Parameter check error during data transfer, or Axis Parameter check error during data transfer), the transferred set value will be destroyed, and the parameter will return to the set value before transferring data. WRITE DATA is still executed if an error occurs, but even if the data is transferred normally, the error that occurred will not be cleared. Make sure to execute ERROR RESET

before WRITE DATA to enable identifying whether the data that was written has been transferred correctly.

5-2-2 Reading PCU Parameters

The PCU's Common Parameters and Axis Parameters are read to the CPU Unit using the READ DATA Bit in the Common Operating Memory Area. The data to be transferred is set by specifying the number of transfer words, transfer source, and transfer destination in the Common Operating Memory Area. Multiple parameters can also be transferred all at once.

Note

- (1) Read data from the PCU when the Data Transferring Flag is OFF. READ DATA cannot be executed from the PCU while the Unit Busy Flag is ON (PCU initializing or reading/writing Memory Card backup data).
 - (2) If READ DATA is executed while data is being transferred, a Multistart Error (Unit error code: 0021) will occur and data will not be read from the PCU.

Data reading starts when the READ DATA Bit in the Common Operating Memory Area turns ON. When reading data from the PCU, make sure that the READ DATA Bit remains ON until the Data Transferring Flag turns ON.

Up to 600 words can be transferred during a single data reading operation. The maximum amount of data that can be transferred during a single CPU Unit cycle, however, is 20 words. When reading more than 20 words of data, the data is divided and read over multiple cycles. When reading data from the PCU is completed, the Data Transferring Flag turns OFF.

Common Operating Memory Area (Output)

Name	Word	Bit	Contents
READ DATA	n	02	$0 \rightarrow 1$: Starts reading data
Number of read words	n+10		Specifies the number of words to be read from the PCU.
Read source address	n+11		Specifies the read source address in the PCU.
Read destination area	n+12		Specifies the area used to store the data read from the PCU.
Read destination word	n+13		The type of area set as the read destination area is specified by specifying the begin- ning word of the area in the read destination word in hexadecimal.
			Read destination area: 00B0 hex: CIO Area 00B1 hex: Work Area 00B2 hex: Holding Area 00B3 hex: Auxiliary Area 0082 hex: DM Area 005 hex: EM Area

 $n = CIO 1500 + (unit number \times 25)$

Common Operating Memory Area (Input)

Name	Word	Bit	Contents
Data Transferring Flag	n+15	14	0: Powerup, transfer completed, or transfer failed
Unit Busy Flag	n+16	14	0: PCU is not busy
			1: PCU is busy



The following diagram shows the operation for reading data from the PCU.



Timing Chart for Reading Data From the PCU

Data can be read from the PCU during axis operation if the Unit Busy Flag is not ON (i.e., except when the PCU is initializing or reading/writing Memory Card backup data). READ DATA, WRITE DATA, and SAVE DATA cannot be executed at the same time.



When reading data from the PCU is completed, the Data Transferring Flag (word n+15, bit 14) turns OFF. When an error occurs during data transfer, the Unit Error Flag (word n+15, bit 12) turns ON, and the corresponding error code is input in word n+21 of the Common Operating Memory Area. Check the error code and perform appropriate troubleshooting.
Note If an error occurs when reading data (multistart error or read transfer error), the data will not be read from the PCU and the data in the read destination area will not be refreshed. READ DATA can still be executed if an error occurs, but even if the data is transferred normally, the error that occurred previously will not be cleared. Make sure to execute ERROR RESET before executing READ DATA to enable identifying whether the data that was read has been transferred correctly.

5-2-3 Saving PCU Parameters

When the power is turned ON, the PCU reads the Common Parameters and Axis Parameters saved in the PCU's flash memory to the PCU's internal memory and starts operations. The PCU's parameters are saved to the PCU's internal flash memory using the SAVE DATA Bit in the Common Operating Memory Area.

- Note (1) Save data to the PCU when MECHATROLINK communications are stopped and the Data Transferring Flag is OFF. The PCU's data cannot be saved while the Unit Busy Flag is ON (i.e., PCU initializing or reading/ writing Memory Card backup data).
 - (2) If SAVE DATA is executed during MECHATROLINK communications or while data is transferring, a Multistart Error (Unit error code: 0021) will occur and data will not be saved to the PCU.
 - (3) Do not restart the PCU or turn OFF the power to the PLC while data is being saved. Doing so may result in damage to the PCU's internal flash memory and the PCU may not operate normally.

Data saving starts when the SAVE DATA Bit in the Common Operating Memory Area turns ON, at which time the enabled Common Parameters and Axis Parameters are saved to the PCU's internal flash memory all at once. When saving data to the PCU, make sure that the SAVE DATA Bit remains ON until the Data Transferring Flag turns ON.

Common Operating Memory Area (Output)

Name	Word	Bit	Contents
SAVE DATA	n	03	$0 \rightarrow 1$: Starts saving data

 $n = CIO 1500 + (unit number \times 25)$

Common Operating Memory Area (Input)

Name	Word	Bit	Contents
Data Transferring Flag	n+15	14	0: Powerup, saving completed, or saving failed
			1: Saving data
Unit Busy Flag	n+16	14	0: PCU is not busy
			1: PCU is busy
Connection Status Flag		15	0: MECHATROLINK communications stopped
			1: MECHATROLINK communications exe- cuting

 $n = CIO 1500 + (unit number \times 25)$



The following diagram shows the operation for saving data in the PCU.

Timing Chart for Saving Data to the PCU

Data can be saved to the PCU while MECHATROLINK communications are stopped if the Unit Busy Flag is not ON (i.e., except when the PCU is initializing or reading/writing Memory Card backup data). SAVE DATA, WRITE DATA, and READ DATA cannot be executed at the same time.



When saving data to the PCU is completed, the Data Transferring Flag (word n+15, bit 14) turns OFF. When an error occurs during data transfer, the Unit Error Flag (word n+15, bit 12) turns ON, and the corresponding error code is input in word n+21 of the Common Operating Memory Area. Check the error code and perform appropriate troubleshooting.

Note If an error occurs while SAVE DATA is executing (multistart error), the data will not be saved to the PCU's internal flash memory, and the parameter set values saved in flash memory will be those before SAVE DATA was executed. SAVE DATA is still executed if an error occurs, but even if the data is saved normally, the error that occurred will not be cleared. Make sure that SAVE DATA is executed after ERROR RESET to enable identifying whether data has been saved correctly.

5-3 Transferring Servo Parameters

Note

Make sure that the equipment will not be adversely affected before changing the Servo Parameters (WRITE SERVO PARAMETER, SAVE SERVO PARAMETER). Refer to the Servo Drive's operation manual and always check the effect of changing the settings before changing the Servo Parameters.

5-3-1 Writing Servo Parameters

Servo Parameters are written from the CPU Unit using the WRITE SERVO PARAMETER Bit in the Axis Operating Output Memory Area. Prepare the set values for the parameter number, parameter size (data length), and transfer data in the Axis Operating Output Memory Area and execute WRITE SERVO PARAMETER.

One execution of WRITE SERVO PARAMETER transfers one parameter for each axis. To transfer multiple parameters, WRITE SERVO PARAMETER must be executed separately for each parameter. Parameters for different axes can be written at the same time.

Parameters transferred using WRITE SERVO PARAMETER are written to the control memory in the Servo Drive. Therefore, online parameters that are overwritten are used immediately in Servo Drive control of the Servomotor. After executing WRITE SERVO PARAMETER for offline parameters, DEVICE SETUP must be executed to enable the settings.

When WRITE SERVO PARAMETER is executed, however, the settings are not saved in the Servo Drive's non-volatile memory. When the Servo Drive power is turned OFF, the set parameter settings will be lost and the next time the power is turned ON, operation will be performed using the parameters saved in the non-volatile memory. To save parameters in the Servo Drive's non-volatile memory, SAVE SERVO PARAMETER must be executed for each parameter separately from WRITE SERVO PARAMETER.

- Note (1) Execute WRITE SERVO PARAMETER when the Servo Parameter Transferring Flag for the corresponding axis is OFF. If WRITE SERVO PARAM-ETER is executed while a Servo Parameter is being transferred, a Multistart Error (axis error code: 3050) will occur and the Servo Parameter will not be written.
 - (2) Do not restart the PCU or turn OFF the power to the PLC while writing a Servo Parameter. Doing so may result in parameters not being transferred correctly and cause unexpected operations.

Servo Parameter writing starts when the WRITE SERVO PARAMETER Bit in the Axis Operating Output Memory Area turns ON. When executing WRITE SERVO PARAMETER, make sure that the WRITE SERVO PARAMETER Bit remains ON until the Servo Parameter Transferring Flag turns ON.

Axis Operating Output Memory Area (Operating Commands)

Name	Word	Bit	Contents
WRITE SERVO PARAMETER Bit	a+1	12	$0 \rightarrow 1$: Starts writing Servo Parameters
Servo Parameter No.	a+17		Specifies the parameter No. of the Servo Drive to which the data is to be written.
Parameter size	a+18		Specifies the data length of the parameter to be written in byte units.
Write data (rightmost word)	a+19	Specifies the setting data (hexade be written to the Servo Parameter	Specifies the setting data (hexadecimal) to be written to the Servo Parameters. For a
Write data (leftmost word)	a+20		parameter size of two bytes, the rightmost word is used only, and the data in the left- most word is ignored. For a parameter size of four bytes, the four bytes of data from the rightmost to leftmost words are transferred.

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) \times 25

Axis Operating Input Memory Area (Monitoring)

Name	Word	Bit	Contents
Servo Parameter Transferring Flag	b	14	0: Powerup, transfer completed, or transfer failed1: Transferring Servo Parameter

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

The following diagram shows the operation used to write parameters to the W-series Servo Drive.



a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No.-1) × 25
 b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No.-1) × 25

Timing Chart for Writing Servo Parameters

WRITE SERVO PARAMETER can also be executed while the corresponding axis is operating (Busy Flag ON).

When using Position Control Units with unit version 2.0 or later, any Servo parameter write operations specified during an origin search will be performed after the origin search has been completed.

WRITE SERVO PARAMETER, READ SERVO PARAMETER, and SAVE SERVO PARAMETER cannot be executed on one axis at the same time.

Servo Parameter No. (Word a+17)	050A hex	
Parameter size (Word a+18)	2	
Write data (Words a+19, a+20)	00002881 hex	For a parameter size of two bytes, only the rightmost word (word a + 19) of the write data is
WRITE SERVO PARAMETER Bit (Word a+1, bit 12)		used, and the data in the leftmost word (word a + 20) is ignored.
Servo Parameter Transferring Flag (Word b, bit 14)		The Servo Parameter Transferring Flag turns ON for at least one cycle time when the command to
Busy Flag (Word b, bit 13)		transfer the parameter is received. The status of the Busy Flag does not change when Servo Parameters are being written

• Writing Servo Parameters during Origin Searches for Position Control Units with Unit Version 2.0 or Later



When writing a Servo Parameter is completed, the Servo Parameter Transferring Flag (word b, bit 14) turns OFF. If an error occurs during Servo Parameter transfer, the Error Flag (word b, bit 12) or the Warning Flag (word b, bit 11) will turn ON, and the corresponding axis error (warning) code is input in word b+4 of the Axis Operating Input Memory Area. Check the error code and perform appropriate troubleshooting.

Note With PCUs of unit version 1.2 or earlier, WRITE SERVO PARAMETER cannot be executed when an axis error has occurred. WRITE SERVO PARAMETER will be ignored if it is executed when an axis error has occurred. With PCUs of version 1.3 or later, WRITE SERVO PARAMETER can be executed even when an axis error has occurred. When the parameters are transferred, however, the Error Flag can no longer be used to determine if an error has occurred during the transfer. It is thus necessary to write Servo Parameters after resetting the error or, if parameters are written when there is an axis error, then steps must be taken to confirm that the parameters were transferred correctly.

5-3-2 Reading Servo Parameters

Servo Parameters are read to the CPU Unit using the READ SERVO PARAM-ETER Bit in the Axis Operating Output Memory Area. Specify the parameter number and parameter size (data length) in the Axis Operating Output Memory Area and execute READ SERVO PARAMETER. The read parameter will be input in the Read data word of the Axis Operating Input Memory Area.

One execution of READ SERVO PARAMETER transfers one parameter for each axis. To transfer multiple parameters, READ SERVO PARAMETER must be executed separately for each parameter. Parameters for different axes can also be read at the same time.

Note Execute READ SERVO PARAMETER when the Servo Parameter Transferring Flag for the corresponding axis is OFF. If READ SERVO PARAMETER is executed while a Servo Parameter is being transferred, a Multistart Error (axis error code: 3050) will occur and the Servo Parameter will not be read.

Servo Parameter reading starts when the READ SERVO PARAMETER Bit in the Axis Operating Output Memory Area turns ON. When executing READ SERVO PARAMETER, make sure that the READ SERVO PARAMETER Bit remains ON until the Servo Parameter Transferring Flag turns ON.

Axis Operating Output Memory Area (Operating Commands)

Name	Word	Bit	Contents
READ SERVO PARAMETER Bit	a+1	13	$0 \rightarrow 1$: Starts reading Servo Parameter

Name	Word	Bit	Contents
Servo Parameter No.	a+17		Specifies the Servo Drive parameter No. from which the data is to be read.
Parameter size	a+18		Specifies the data length of the parameter to be read in byte units.

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) \times 25

Axis Operating Input Memory Area (Monitoring)

Name	Word	Bit	Contents
Servo Parameter Transferring Flag	b	14	0: Powerup, transfer completed, or transfer failed
			1: Transferring Servo Parameter
Read data (rightmost word)	b+14		Contains the Servo Parameter (hexadeci- mal) that was read. For a parameter size of
Read data (leftmost word)	b+15		two bytes, the rightmost word is used only, and 0000 is stored in the leftmost word. For a parameter size of four bytes, four bytes of data from the rightmost to leftmost words are stored.

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

READ SERVO PARAMETER can be executed while the corresponding axis is operating (Busy Flag ON). The following diagram shows the operation used to read parameters to the Servo Drive.



a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) \times 25 b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. –1) \times 25

Timing Chart for Reading Servo Parameters

READ SERVO PARAMETER can be executed while the corresponding axis is operating (Busy Flag ON).

When using Position Control Units with unit version 2.0 or later, any Servo parameter read operations specified during an origin search will be performed after the origin search has been completed.

READ SERVO PARAMETER, WRITE SERVO PARAMETER, and SAVE SERVO PARAMETER cannot be executed on one axis at the same time.

Section 5-3



 Reading Servo Parameters during Origin Searches for Position Control Units with Unit Version 2.0 or Later



When READ SERVO PARAMETER execution is completed, the Servo Parameter Transferring Flag (word b, bit 14) turns OFF, and the contents of the parameter that has been read is stored in the read data words (words b+14, b+15).

If an error occurs while transferring Servo Parameters, the Error Flag (word b, bit 12) or the Warning Flag (word b, bit 11) will turn ON, and the corresponding axis error (warning) code will be input in word b+4 of the Axis Operating Input Memory Area. Check the error code and perform appropriate trouble-shooting.

Note With PCUs of unit version 1.2 or earlier, READ SERVO PARAMETER cannot be executed when an axis error has occurred. READ SERVO PARAMETER will be ignored if it is executed when an axis error has occurred. With PCUs of version 1.3 or later, READ SERVO PARAMETER can be executed even when an axis error has occurred. When the parameters are transferred, however, the Error Flag can no longer be used to determine if an error has occurred during the transfer. It is thus necessary to read Servo Parameters after resetting the error or, if parameters are read when there is an axis error, then steps must be taken to confirm that the parameters were transferred correctly.

5-3-3 Saving Servo Parameters

When Servo Parameters are saved in the Servo Drive's non-volatile memory, the settings are held even after the power is turned OFF and the next time the power is turned ON, the previous settings will be recovered. Saving Servo Parameter is treated as writing data to the Servo Drive's non-volatile memory. Therefore, SAVE SERVO PARAMETER must be executed separately from WRITE SERVO PARAMETER for each parameter.

Servo Parameters are saved from the CPU Unit using the SAVE SERVO PARAMETER Bit in the Axis Operating Output Memory Area. Specify the parameter number, parameter size (data length), and data to be transferred in the Axis Operating Output Memory Area and execute SAVE SERVO PARAM-ETER (to write settings to non-volatile memory).

One execution of SAVE SERVO PARAMETER transfers one parameter for each axis. The parameter is written to the Servo Drive's internal memory at the same time as saving it in the non-volatile memory. To save multiple parameters, SAVE SERVO PARAMETER must be executed separately for each parameter. Parameters for different axes can be saved at the same time.

- Note (1) There is a limit to the number of times non-volatile memory in the Servo Drive can be written (10,000 times). When Servo parameters are changed frequently during system operation, so not save the parameters, but rather just write them.
 - (2) Execute SAVE SERVO PARAMETER when the Servo Parameter Transferring Flag for the corresponding axis is OFF. If SAVE SERVO PARAM-ETER is executed while the Servo Parameter Transferring Flag is ON, a Multistart Error (axis error code: 3050) will occur and the Servo Parameter will not be saved.
 - (3) Do not restart the PCU or turn OFF the power to the PLC while a Servo Parameter is being saved. Doing so may result in the parameter not being saved correctly and cause unexpected operations.

Servo Parameter saving starts when the SAVE SERVO PARAMETER Bit in the Axis Operating Output Memory Area turns ON. When executing SAVE SERVO PARAMETER, make sure that the SAVE SERVO PARAMETER Bit remains ON until the Servo Parameter Transferring Flag turns ON.

Axis Operating Output Memory Area (Operating Commands)

Name	Word	Bit	Contents
SAVE SERVO PARAMETER Bit	a+1	14	$0 \rightarrow 1$: Starts writing Servo Parameter to non-volatile memory.
Servo Parameter No.	a+17		Specifies the parameter No. of the Servo Drive to which the data is to be written.
Parameter size	a+18		Specifies the data length of the parameter to be written in byte units.
Write data (rightmost word)	a+19		Specifies the setting data (hexadecimal) to be written to the Servo Parameter. For a
Write data (leftmost word	a+20		parameter size of two bytes, the rightmost word is used only, and the data in the left- most word is ignored. For a parameter size of four bytes, four bytes of data from the rightmost to leftmost words are transferred.

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) \times 25

Axis Operating Input Memory Area (Monitoring)

Name	Word	Bit	Contents
Servo Parameter Transferring Flag	b	14	0: Powerup, transfer completed, or transfer failed 1: Transferring Servo Parameter

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

Transferring Servo Parameters

Section 5-3

SAVE SERVO PARAMETER can be executed while the corresponding axis is operating (Busy Flag ON). The following diagram shows the operation used to save parameters to the W-series Servo Drive.



a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) × 25 b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

Timing Chart for Saving Servo Parameters

SAVE SERVO PARAMETER can be executed while the corresponding axis is operating (Busy Flag ON).

When using Position Control Units with unit version 2.0 or later, any Servo parameter save operations specified during an origin search will be performed after the origin search has been completed.

SAVE SERVO PARAMETER, WRITE SERVO PARAMETER, and READ SERVO PARAMETER cannot be executed on one axis at the same time.

Servo Parameter No. (word a+17)	050A hex	
Parameter size (word a+18)	2	
Write data (words a+19, a+20)	00002881 hex	For a parameter size of two bytes, only the rightmost word (word a+19) of the write data is
SAVE SERVO PARAMETER Bit (word a+1, bit 14)		used, and the data in the leftmost word (word a+20) is ignored.
Servo Parameter Transferring Flag (word b, bit 14)		The Servo Parameter Transferring Flag turns ON for at least one cycle time when the
Busy Flag (word b, bit 13)		The status of the Busy Flag does not change when Servo Parameters are being written.

• Saving Servo Parameters during Origin Searches for Position Control Units with Unit Version 2.0 or Later



When execution of SAVE SERVO PARAMETER has completed, the Servo Parameter Transferring Flag (word b, bit 14) turns OFF.

If an error occurs while executing SAVE SERVO PARAMETER, the Error Flag (word b, bit 12) or the Warning Flag (word b, bit 11) will turn ON, and the corresponding axis error (warning) code will be input in word b+4 of the Axis Operating Input Memory Area. Check the error code and perform appropriate troubleshooting.

Note With PCUs of version 1.2 or earlier, SAVE SERVO PARAMETER cannot be executed if an axis error has occurred. SAVE SERVO PARAMETER will be ignored if it is executed when an axis error has occurred. With PCUs of version 1.3 or later, SAVE SERVO PARAMETER can be executed even when an axis error has occurred. When the parameters are saved, however, the Error Flag can no longer be used to determine if an error has occurred during the save. It is thus necessary to save Servo Parameters after resetting the error or, if parameters are saved when there is an axis error, then steps must be taken to confirm that the parameters were saved correctly.

5-3-4 Device Setup

Two types of Servo Parameters are transferred to the Servo Drive: online parameters with set values that are enabled immediately after they are written, and offline parameters, which are not enabled just by writing them. To enable set values for offline parameters in Servo Drive operations, save the parameters to the non-volatile memory, and then cycle the Servo Drive power. Alternatively, execute WRITE SERVO PARAMETER/SAVE SERVO PARAME-TER and then execute the Servo Drive's DEVICE SETUP.

When DEVICE SETUP is executed, the Servo Drive enables all the Servo Parameters that have been transferred to set the Servo Drive, and initializes the present position and output signal status. DEVICE SETUP can be executed in Servo unlock status. If DEVICE SETUP is executed while in Servo lock status, a Multistart Error (axis error code: 3050) will occur and DEVICE SETUP will not be executed.

DEVICE SETUP is executed from the CPU Unit using the DEVICE SETUP Bit in the Axis Operating Output Memory Area. DEVICE SETUP can be executed for different axes at the same time.

Note Execute DEVICE SETUP when the Busy Flag for the corresponding axis is OFF. If DEVICE SETUP is executed while the axis's Busy Flag is ON, a Multi-start Error (axis error code: 3050) will occur and DEVICE SETUP will not be executed.

Device setup starts when the DEVICE SETUP Bit in the Axis Operating Output Memory Area turns ON. When executing the DEVICE SETUP command, make sure that the DEVICE SETUP Bit remains ON until the Busy Flag in the Axis Operating Input Memory Area turns ON.

Axis Operating Output Memory Area (Operating Commands)

Name	Word	Bit	Contents
DEVICE SETUP Bit	a+1	11	$0 \rightarrow 1$: Starts device setup for the Servo Drive.

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) \times 25

Axis Operating Input Memory Area (Monitoring)

Name	Word	Bit	Contents
Receiving Com-	b	00	0: Command reception enabled
mand Flag			$0 \rightarrow 1$: Command reception started
			1: Receiving command (command recep- tion disabled)
Busy Flag		13	1: Axis busy (axis operation executing)

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

The following diagram shows the device setup operation for the Servo Drive.



a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) × 25 b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

Timing Chart for DEVICE SETUP

DEVICE SETUP can be executed when the corresponding axis is in Servo unlocked status and the Busy Flag is not ON. If DEVICE SETUP is executed while in Servo lock status, a Multistart Error (axis error code: 3050) will occur and DEVICE SETUP will not be executed.



During DEVICE SETUP processing, the Receiving Command Flag (word b, bit 00) and Busy Flag (word b, bit 13) remain ON. When DEVICE SETUP execution is completed, both of these Flags turn OFF. When DEVICE SETUP is executed, the origin is lost (No Origin Flag turns ON), and the present position becomes undefined. After DEVICE SETUP is executed, establish the origin again.

During execution of DEVICE SETUP, the Servo status and Servo Drive output signals become undefined.

Note DEVICE SETUP cannot be executed if an axis error has occurred. DEVICE SETUP will be ignored if it is executed during an axis error.

SECTION 6 MECHATROLINK

This section provides an overview of MECHATROLINK communications, and includes information on settings and procedures required to use MECHATROLINK with the Position Control Unit.

6-1	MECHA	ATROLINK Overview	238
6-2	MECHA	ATROLINK Settings	239
	6-2-1	Scan List	239
	6-2-2	Scan List and PCU Area Allocations	241
	6-2-3	MECHATROLINK Communications Settings	243
6-3	MECHA	ATROLINK Communications Control	248
	6-3-1	Establishing Connections	248
	6-3-2	MECHATROLINK Communications Status	249
	6-3-3	MECHATROLINK Communications Errors	254
	6-3-4	Rejoining the Connection	258
	6-3-5	Specifying the Axes to Connect	261
6-4	Standard	d Settings for Servo Drives Using MECHATROLINK	263
	6-4-1	G5-series Servo Drive Settings (R88D-KN ⁻ -ML2 with Built-in MECHATROLINK-II Communications)	263
	6-4-2	G-series Servo Drive Settings (R88D-GN ⁻ ML2 with Built-in MECHATROLINK-II Communications)	264
	6-4-3	W-series Servo Drive Settings (R88D-WT Equipped with FNY-NS115)	265
	6-4-4	W-series Servo Drive Settings (R88D-WN□-ML2 with Built-in MECHATROLINK-II Communications)	268
	6-4-5	SMARTSTEP Junior Servo Drive Settings (R7D-ZND-ML2 with Built-in MECHATROLINK-II Communications)	269

6-1 MECHATROLINK Overview

MECHATROLINK The Position Control Unit (PCU) uses MECHATROLINK (see note) to connect to the Servo Drive.

Note MECHATROLINK is a registered trademark of Yaskawa Electric Corporation.

MECHATROLINK is a high-speed field network developed by Yaskawa Electric Corporation. High-speed, high-precision motion control is possible for up to 30 (see note) MECHATROLINK-compatible devices connected to a single communications line using high-speed communications of up to 10 Mbps.

Note Up to 16 Servo Drives can be connected to the CS1W/CJ1W-NCF71, up to 2 Servo Drives can be connected to the CS1W/CJ1W-NC271, and up to 4 Servo Drives can be connected to the CS1W/CJ1W-NC471. The highest axis number that can be set in the Position Control Unit is 16.

MECHATROLINK is available with two baud rates: 4 Mbps for MECHA-TROLINK-I and 10 Mbps for MECHATROLINK-II. This PCU is designed for MECHATROLINK-II. The connected devices must be compatible with MECHATROLINK-II.

In this manual, the use of MECHATROLINK refers to MECHATROLINK-II unless otherwise specified.

MECHATROLINK Specifications

Item	Specifications
Communications protocol	MECHATROLINK-II
Baud rate	10 Mbps
Maximum transmission distance	50 m (See note 1.)
Minimum distance between stations	0.5 m
Transmission media	Shielded, twisted-pair cables
Maximum No. of stations	30 slave stations max. (See note 2.)
Topology	Bus
Transfer cycle	250 μs to 8 ms
Communications method	Master-slave, totally synchronous
Encoding	Manchester encoding
Data length	17 bytes/32 bytes selectable (See note 3.)

Note

- (1) This distance is the total length of the cable connected between devices. However, the maximum length depends on the number of devices connected and whether Repeaters are used. For details, refer to 3-4-1 MECHATROLINK-II Communications Wiring.
 - (2) Up to 16 devices can be connected to the CS1W/CJ1W-NCF71, up to 2 devices can be connected to the CS1W/CJ1W-NC271, and up to 4 devices can be connected to the CS1W/CJ1W-NC471.
 - (3) The PCU data length is fixed at 32 bytes.

This PCU is designed to be used with MECHATROLINK-II. The maximum number of stations that can be connected to the PCU is 16, and the station addresses of the MECHATROLINK-compatible devices must be set in the range 1 to 16. The MECHATROLINK-II communications settings at the PCU are fixed, as follows:

Item	Specifications
Baud rate	10 Mbps
Data length	32 bytes (fixed) (See note.)

MECHATROLINKcompatible Devices Supported by PCU

Note Some devices indicate the data length as 30 bytes, but the meaning is the same.

Make sure that the baud rates and number of transmission bytes for the MECHATROLINK-compatible devices connected to the PCU are set to the same settings as shown in the above table.

The MECHATROLINK-compatible devices that can be connected to the PCU are shown in the following table.

Category	Compatible devices
Servo Drive	R88D-KN -ML2 OMRON G5-series Servo Drives (with MECHATROLINK-II communications)
	R88D-GN -ML2 OMRON G-series Servo Drives (with MECHATROLINK-II communications)
	R88D-WT OMRON W-series Servo Drives (equipped with FNY-NS115 MECHATROLINK-II I/F Unit)
	R88D-WN -ML2 OMRON W-series Servo Drives (equipped with built-in MECHATROLINK-II communications)
	R7D-ZN -ML2 OMRON SMARTSTEP Junior Servo Drives (equipped with built-in MECHATROLINK-II communications)

The version of R88D-WT W-series Servo Drives used must support FNY-NS115 MECHATROLINK-II I/F Unit. FNY-NS115 MECHATROLINK-II I/F Unit can be installed in R88D-WT W-series Servo Drives with version 39 or later. (For details, refer to *3-1-2 MECHATROLINK-II* I/F Unit.) The version of R88D-WT W-series Servo Drives is shown on the nameplate located on the side of the Servo Drive. Make sure that the Servo Drive is version 39 or later. Earlier versions will not function properly.

6-2 MECHATROLINK Settings

The settings required to use MECHATROLINK with the PCU are described here.

6-2-1 Scan List

The scan list is used to register the MECHATROLINK devices connected to the PCU. The PCU connects the axes registered in the scan list to MECHA-TROLINK communications, detects these axes, and monitors communications for them.

The PCU receives positioning commands allocated in the Work Area (Axis Operating Memory Area) of the CPU Unit and controls positioning of the Servo Drive. The Axis Operating Memory Areas are allocated based on the scan list that is saved in the PCU's internal flash memory. Therefore, to use the PCU, first the scan list must be created and saved, and the connected Servo Drives must be registered in the list.



The station address numbers set at the MECHATROLINK device slave stations correspond to the axis numbers registered in the PCU's scan list, regardless of the physical order in which the devices are connected to the MECHATROLINK communications line. The scan list is set as the Common Parameters for the PCU.

The axes are allocated eight bits per axis in the eight words of data in the PCU's internal addresses from 183C to 1843 hex, as shown in the following table. To allocate an axis to the Servo Drive, set the data to 40 hex and when an axis is not used, set the data to 00 hex. (Nothing is registered for the default setting.) Up to 16 axes can be allocated for the CS1W/CJ1W-NCF71, up to 2 axes can be allocated for the CS1W/CJ1W-NC271, and up to 4 axes can be allocated for the CS1W/CJ1W-NC271, and up to 4 axes can be allocated for the CS1W/CJ1W-NC271. An error will occur if you allocate more axes than your Unit supports. The Common Parameters settings saved in the PCU's internal flash memory are enabled when the PCU power is turned ON or the Unit is restarted.

PCU's	l's Contents		Setting
address	Bits 08 to 15	Bits 00 to 07	
183C hex	Axis 2 allocation	Axis 1 allocation	00 hex: Axis not used
183D hex	Axis 4 allocation	Axis 3 allocation	(default setting).
183E hex	Axis 6 allocation	Axis 5 allocation	40 hex: Allocates axis to
183F hex	Axis 8 allocation	Axis 7 allocation	
1840 hex	Axis 10 allocation	Axis 9 allocation	
1841 hex	Axis 12 allocation	Axis 11 allocation	
1842 hex	Axis 14 allocation	Axis 13 allocation	
1843 hex	Axis 16 allocation	Axis 15 allocation	

6-2-2 Scan List and PCU Area Allocations

Words are allocated to the PCU in the Work Area in the CPU Unit for use as Axis Operating Output Memory Areas and Axis Operating Input Memory Areas for the Servo Drives registered in the scan list. An output area of 25 words and an input area of 25 words are allocated to each axis according to the axis number registered in the scan list. The beginning word of these output areas and input areas can be set by the user in the Common Parameter Area.

The output areas and input area allocations are specified in the four words of data in the PCU's internal addresses from 1838 to 183B hex, as shown in the following table. The Common Parameters settings saved in the PCU's internal flash memory are enabled when the PCU power is turned ON or the Unit is restarted.

PCU's address	Contents	Setting
1838 hex	Axis Operating Out- put Memory Area designation	Specifies the words allocated for the Axis Oper- ating Output Memory Areas. 0000 hex: No setting (default) 00B0 hex: CIO Area 00B1 hex: Work Area 00B2 hex: Holding Area 00B3 hex: Auxiliary Area 00B3 hex: DM Area 0082 hex: DM Area 0050 to 0059, 005A,005B, 005C hex: EM Area (5 : = EM Bank No.)
1839 hex	Beginning word of Axis Operating Out- put Memory Areas	Specifies the beginning word of the Axis Operat- ing Output Memory Areas. The beginning word of the Operating Output Memory Area for axis 1 is determined by the Axis Operating Output Memory Area designation and the word deter- mined using this parameter. Each axis is allo- cated 25 words in sequence from this word.
183A hex	Axis Operating Input Memory Area desig- nation	Specifies the words allocated for the Axis Oper- ating Input Memory Areas. The setting method is the same as for the Axis Operating Output Memory Areas.
183B hex	Beginning word of Axis Operating Input Memory Areas	Specifies the beginning word of the Axis Operat- ing Input Memory Areas. The setting method is the same as for the Axis Operating Output Mem- ory Areas.

The Axis Operating Memory Output and Input Areas are allocated words for each axis from axis 1 up to the highest axis number registered in the scan list. Axis numbers between axis 1 and the highest registered axis number that are not registered as MECHATROLINK devices are also each allocated 25 output words and 25 input words.

			Axis Operatir	ig Output Memory Area	L	Axis Operating Input Memory Are		
Axis	MECHATROLINK		Allocations			Allocations		
No.	station address No.		Axis 1	Word a+0		Axis 1	Word b+0	
Axis 1	No. 1		Operating Output	Word a+1	Operating Input Memory	Operating	Word b+1	
Axis 2	No. 2			:		:		
Axis 3	No. 3		Memory Area	Word a+24		Area	Word b+24	
Axis 4	No. 4	À.	Axis 2	Word a+25		Axis 2	Word b+25	
:	:	$\sum_{i=1}^{n}$	Operating Output Memory Area	Word a+26	Operating Input Memory Area	Word b+26		
Axis 14	No. 14	. \		:		:		
Axis 15	No. 15	/		Word a+49		Word b+49		
Axis 16	No. 16	1				-		
		()	Axis N Operating Output	Word a+(N-1)×25	Axis N Operating	Word b+(N–1)×25		
	//	11		Word a+(N-1)×25+1		Word b+(N–1)×25+1		
		11		:	Input Memory Area		:	
	()	-1/	Memory Area	Word a+(N-1)×25+24			Word b+(N-1)×25+24	
	/ /	<u></u>						
	Ì	Axis 16	Word a+375		Axis 16	Word b+375		
	j.	Operating	Word a+376		Operating	Word b+376		
		1	Output	:		Input Memory	:	
		1	Memory Area	Word a+399	Area		Word b+399	

 $a = \text{Beginning word of Axis Operating Output Areas specified in Common Parameters } \\ b = \text{Beginning word of Axis Operating Input Areas specified in Common Parameters }$

Example:

Beginning word of Axis Operating Output Memory Areas: CIO 100 Beginning word of Axis Operating Input Memory Areas: CIO 500 Station addresses of connected MECHATROLINK devices (registered in scan list): No. 2, No. 5, and No. 7

In this example, the words occupied as Axis Operating Memory Areas are as follows:

Axis Operating Output Memory Area: CIO 100 to CIO 274 (Axes 1 to 7) Axis Operating Input Memory Area: CIO 500 to CIO 674 (Axes 1 to 7) Axis 2 Output Area:CIO 125 to CIO 149; Axis 2 Input Area:CIO 525 to CIO 549 Axis 5 Output Area:CIO 200 to CIO 224; Axis 5 Input Area:CIO 600 to CIO 624 Axis 7 Output Area:CIO 250 to CIO 274; Axis 7 Input Area:CIO 650 to CIO 674

Axes 1, 3, 4, and 6 are not used, but are still allocated words in the PCU. These words cannot be used as work words.

Note Set the beginning word of the Axis Operating Output Memory Areas and Axis Operating Input Memory Areas in the Common Parameters so that the words allocated to each area do not exceed the range for each of the CPU Unit's I/O memory areas. If the setting exceeds the range of the I/O memory area, an Initialization Common Parameter Check Error (Unit error code: 0028) will occur when the PCU power is turned ON or the Unit is restarted.

Example:

Beginning word of the Axis Operating Output Memory Areas: CIO 6100 Connected axes: 2 min.

Axis 1: Output Area: CIO 6100 to CIO 6124 Axis 2: Output Area: CIO 6125 to CIO 6149 Etc.

Areas for subsequent axes are allocated in sequence. The highest word in the CIO Area is CIO 6143. Therefore, an error will occur. If the ranges set for the Axis Operating Output Memory Areas and Axis Operating Input Memory Areas overlap, an Initialization Common Parameter Check Error (Unit error code: 0028) will occur when the PCU power is turned ON or the Unit is restarted.

6-2-3 MECHATROLINK Communications Settings

In addition to the scan list, the following four PCU settings are also for MECHATROLINK communications.

- Transfer cycle
- Communications cycle
- Number of communications retries
- C2 master connection

Of these settings, the transfer cycle and communications cycle must always be set when using the PCU. The settings for MECHATROLINK communications are set as the PCU's Common Parameters.

Set the two words of data in the PCU's internal addresses 1856 hex and 1857 hex as shown in the following table. The Common Parameters settings saved in the PCU's internal flash memory are enabled when the PCU power is turned ON or the Unit is restarted.

PCU's	Co	ontents		Setting			
address	Bits 08 to 15	Bits 0	0 to 07				
1856 hex	Transfer cycle	Communic cycle	cations	Transfer cycle: 00 hex: 1 ms (default setting) 01 hex: 1 ms05 hex: 5 ms 02 hex: 2 ms06 hex: 6 ms 03 hex: 3 ms07 hex: 7 ms 04 hex: 4 ms08 hex: 8 ms A2 hex: 0.25 msA5 hex: 0.5 ms			
				Communications cycle: Sets a multiplier to obtain integer multiples of the transfer cycle. Set value: 00 to 20 hex The default setting 00 is the same as when the cycle is set to 3. Set as follows: G5 series/G Series Transfer cycle × communications cycle (multiplier) ≤ 16 ms. Other Series Transfer cycle × communications cycle (multiplier) ≤ 32 ms The transfer cycle must be between 1 and 4 ms.			
1857 hex	00 (fixed)	07 to 04 C2 mas- ter con- nection	03 to 00 Number of com- munica- tions retries	C2 master connection: 0: No C2 master (default setting) 1: C2 master connected Number of communications retries: Set value: 0 to 7, F The default setting 0 is the same as when the number of retries is set to 1. When F is specified, the number of retries is 0 (no retries).			

Transfer Cycle

The transfer cycle is the cycle used for sending and receiving data with MECHATROLINK communications. Data (sent and received) is constantly exchanged along the MECHATROLINK communications path during every transfer cycle. The set value and setting range for the transfer cycle depends on the type and number of MECHATROLINK devices connected. When a W-series Servo Drive equipped with a FNY-NS115 MECHATROLINK-II I/F Unit or with built-in MECHATROLINK-II communications is connected to the PCU, the setting range for the transfer cycle is 0.5 to 4 ms. When a G5-series Servo Drive or a G-series Servo Drive is connected to the PCU, the setting range for the transfer cycle is 1 to 4 ms.

Communications Cycle

The communications cycle is the cycle used to refresh data in the PCU and MECHATROLINK devices. The PCU sends operating commands to the connected MECHATROLINK devices and refreshes present positions, status, and other monitoring information every communications cycle.

The set value for the PCU's communications cycle is an integer used as a multiplier to obtain integer multiples of the transfer cycle. The actual communications cycle (data refresh cycle) is a time cycle that is a multiple of the transfer cycle's set value. The set value for the communications cycle is determined by the number of connected MECHATROLINK devices and depends on the PCU communications processing.

The actual communications cycle (data refresh cycle) is a time cycle calculated by multiplying the transfer cycle by a set value. The lower limit of the communications cycle is determined by the number of connected MECHA-TROLINK devices and depends on the PCU communications processing. When the default setting of 0 is used, the communications cycle is three times the length of the transfer cycle. Set the communications cycle so that the maximum value does not exceed 32 ms. When connecting a G5-series Servo Drive or a G-series Servo Drive, set the communications cycle so that the maximum value does not exceed 16 ms.



The following diagram illustrates data exchange with the PCU.

Number of Communications Retries

The number of communications retries is the number of stations that will perform retries if data sending/receiving fails when transferring data that is exchanged between the PCU and MECHATROLINK devices using the transfer cycle. The set value is set as the maximum number of stations used to retry data transmission with the slave station for which data sending/receiving failed.

The number of communications retries can be set between 0 and 7 (communications retries for up to 7 stations). The set value for the number of retries, however, is added to the number of connected stations that determine the transfer cycle (e.g., for 3 retries, the transfer cycle must be considered as that for the number of stations connected plus 3). For normal use, use the default setting of 0 (1 communications retry).

<u>C2 Master Connection</u> The C2 master is a separate communications master connected to the PCU for MECHATROLINK system support. This setting will be used when a future system support device is connected. Use the default setting 0 (No C2 master).

Setting the Transfer Cycle and Communications Cycle

The transfer cycle and communications cycle are set as Common Parameters in the PCU. Set the PCU's transfer cycle and communications cycle according to the number of connected MECHATROLINK devices so that the set values are not lower than the values shown in the following table.

Note Make sure that all of the axis number settings (station numbers of the MECHATROLINK devices) are set to values equal to or less than the number of connected devices for the transfer cycle. Here, the number of connected MECHATROLINK devices is not the actual number of MECHATROLINK devices connected, but the highest registered axis number. For example, when using MECHATROLINK devices for axis number 9 or higher, the transfer cycle must be at least 2 ms even if less than nine devices are connected. If the set value for the transfer cycle is smaller than the number of connected devices or the axis number settings, a Transfer Cycle Setting Error (Unit error code: 0027) will occur when CONNECT is executed.

For the communications cycle, set the multiplier of the transfer cycle set value so that the value is not lower than the values shown in the following table.

For example, if up to 3 Servo Drives (axis number 1 to 3) equipped with FNY-NS115 MECHATROLINK-II I/F Unit are connected to the PCU, either set the transfer cycle to 0.5 ms min. and the communications cycle set value to 2 (multiplier) or higher, or set the transfer cycle to 1.0 ms min. and the communications cycle set value to 1 (multiplier) or higher.

Number of devices connected	R88D-KN□-ML2 G5- series Servo Drive equipped with built-in MECHATROLINK-II communications R88D-GN□-ML2 G-series Servo Drive equipped with built-in MECHATROLINK-II communications		R88D-WT W-series Servo Drive equipped with FNY-NS115 MECHATROLINK-II I/F Unit		R88D-WN -ML2 W- series Servo Drive equipped with built-in MECHATROLINK-II communications		R7D-ZN□-ML2 SMARTSTEP Junior Servo Drive equipped with built-in MECHATROLINK-II communications	
	Minimum transfer cycle set value	Minimum communi- cations cycle value	Minimum transfer cycle set value	Minimum communi- cations cycle value	Minimum transfer cycle set value	Minimum communi- cations cycle value	Minimum transfer cycle set value	Minimum communi- cations cycle value
1	1.0 ms	1.0 ms (1)	0.5 ms	1.0 ms (2)	0.5 ms	0.5 ms (1)	1.0 ms	1.0 ms (1)
2	1.0 ms	1.0 ms (1)	0.5 ms	1.0 ms (2)	0.5 ms	1.0 ms (2)	1.0 ms	1.0 ms (1)
3	1.0 ms	1.0 ms (1)	0.5 ms	1.0 ms (2)	0.5 ms	1.0 ms (2)	1.0 ms	1.0 ms (1)
4	1.0 ms	1.0 ms (1)	1.0 ms	1.0 ms (1)	1.0 ms	1.0 ms (1)	1.0 ms	1.0 ms (1)
5	1.0 ms	2.0 ms (2)	1.0 ms	2.0 ms (2)	1.0 ms	2.0 ms (2)	1.0 ms	2.0 ms (2)
6	1.0 ms	2.0 ms (2)	1.0 ms	2.0 ms (2)	1.0 ms	2.0 ms (2)	1.0 ms	2.0 ms (2)
7	1.0 ms	2.0 ms (2)	1.0 ms	2.0 ms (2)	1.0 ms	2.0 ms (2)	1.0 ms	2.0 ms (2)
8	1.0 ms	2.0 ms (2)	1.0 ms	2.0 ms (2)	1.0 ms	2.0 ms (2)	1.0 ms	2.0 ms (2)
9	2.0 ms	2.0 ms (1)	2.0 ms	2.0 ms (1)	2.0 ms	2.0 ms (1)	2.0 ms	2.0 ms (1)
10	2.0 ms	2.0 ms (1)	2.0 ms	2.0 ms (1)	2.0 ms	2.0 ms (1)	2.0 ms	2.0 ms (1)
11	2.0 ms	4.0 ms (2)	2.0 ms	4.0 ms (2)	2.0 ms	4.0 ms (2)	2.0 ms	4.0 ms (2)
12	2.0 ms	4.0 ms (2)	2.0 ms	4.0 ms (2)	2.0 ms	4.0 ms (2)	2.0 ms	4.0 ms (2)
13	2.0 ms	4.0 ms (2)	2.0 ms	4.0 ms (2)	2.0 ms	4.0 ms (2)	2.0 ms	4.0 ms (2)
14	2.0 ms	4.0 ms (2)	2.0 ms	4.0 ms (2)	2.0 ms	4.0 ms (2)	2.0 ms	4.0 ms (2)
15	2.0 ms	4.0 ms (2)	2.0 ms	4.0 ms (2)	2.0 ms	4.0 ms (2)	2.0 ms	4.0 ms (2)
16	2.0 ms	4.0 ms (2)	2.0 ms	4.0 ms (2)	2.0 ms	4.0 ms (2)	2.0 ms	4.0 ms (2)

Note

- (1) The set values in the above table apply when the default settings are used for the number of communications retries (once) and C2 master connection (No C2 master).
 - (2) The values in parentheses in the *Minimum communications cycle value* column are the PCU's communications cycle set value (multiplier to obtain integer multiples of the transfer cycle) used to set the minimum communications cycle for the minimum transfer cycle set value.
 - (3) When using the R88D-WT W-series Servo Drive and FNY-NS115 together, the SMARTSTEP Junior, or the R88D-GN, set the communications cycle so that it is an integer multiple of 1.0 ms. When using the R88D-WN --ML2 W-series Servo Drive, set the communications cycle so that it is an integer multiple of 0.5 ms.
 - (4) When connecting a G5-series Servo Drive, a G-series Servo Drive, W-series Servo Drive, or the SMARTSTEP Junior Servo Drive to the PCU, set the transfer cycle to 4 ms max.

The PCU default settings are 1.0 ms for the transfer cycle and 3 for the communications cycle (multiplier 3: 1.0 ms \times 3 = 3.0 ms). Without changing the default settings, the G5-series Servo Drive, the G-series Servo Drive, Wseries Servo Drive (equipped with FNY-NS115 MECHATROLINK-II I/F Unit or built-in MECHATROLINK-II communications), or the SMARTSTEP Junior Servo Drive can be used for up to eight axes (axis numbers 1 to 8).

- Note (1) When using linear interpolation, set the communications cycle to a value 1 ms higher than the minimum value given in the above table. If the communications cycle setting is too small, the command response time of the Position Control Unit function may be greatly extended.
 - (2) The command response time for axes not used for linear interpolation will be increased by up to four communications cycles per linear interpolation combination being executed. This applies from when setting the linear interpolation operation is started until linear interpolation is completed (i.e., while either the SETTING LINEAR INTERPOLATION Bit or the Linear Interpolation Executing Flag is ON).

Transfer Cycle and Communications Cycle When Setting the Absolute Encoder Zero Point Position Offset Using an Origin Search Operation with Position Control Units with Unit Ver. 2.0

Make the settings according to the following table when using a Position Control Unit with unit version 2.0. For a Position Control Unit with unit version 2.1 or later, the same settings can be used for the communications cycle as when not setting the absolute encoder zero point position offset using an origin search operation.

If an absolute encoder zero point position offset is set using an origin search operation for a motor with an absolute encoder, set the transfer cycle and communications cycle according to the following table. The settings will depend on the number of axes used and other functions that are executed simultaneously. Refer to *8-6-5 Absolute Encoder's Origin (Zero Point) Position Offset Setting* for details on the position offset setting.

MECHATROLINK Settings

If the settings of the transfer cycle and communications cycle are smaller than the values given in the following table when the absolute encoder zero point position offset is set using an origin search operation, the offset may not be calculated correctly and the position of the origin may not be correct.

Number of	Min. set va	Min. set values when setting the absolute encoder zero point position offset using an origin search								
connected	Linear interpolation not performed at the same time		Linear interpolation performed at the same time for one set of axes		Linear interpolation per- formed at the same time for two sets of axes		During online monitor- ing with the CX-Motion- NCF			
	Minimum transfer cycle set value	Minimum communi- cations cycle value	Minimum transfer cycle set value	Minimum communi- cations cycle value	Minimum transfer cycle set value	Minimum communica- tions cycle value				
1	0.5 ms	1.0 ms (2)	0.5 ms	3.0 ms (6)			Communications cycle			
2	0.5 ms	1.0 ms (2)	0.5 ms	4.0 ms (8)			time on the left + 1.0 ms			
3	0.5 ms	2.0 ms (4)	0.5 ms	5.0 ms (10)						
4	1.0 ms	2.0 ms (2)	1.0 ms	5.0 ms (5)						
5	1.0 ms	2.0 ms (2)	1.0 ms	6.0 ms (6)	1.0 ms	7.0 ms (7)				
6	1.0 ms	3.0 ms (3)	1.0 ms	6.0 ms (6)	1.0 ms	8.0 ms (8)				
7	1.0 ms	3.0 ms (3)	1.0 ms	6.0 ms (6)	1.0 ms	9.0 ms (9)				
8	1.0 ms	3.0 ms (3)	1.0 ms	7.0 ms (7)	1.0 ms	10.0 ms (10)				
9	2.0 ms	4.0 ms (2)	2.0 ms	8.0 ms (4)	2.0 ms	12.0 ms (6)	Same as on the left			
10	2.0 ms	4.0 ms (2)	2.0 ms	8.0 ms (4)	2.0 ms	12.0 ms (6)	Communications cycle			
11	2.0 ms	4.0 ms (2)	2.0 ms	8.0 ms (4)	2.0 ms	12.0 ms (6)	time on the left + 2.0 ms			
12	2.0 ms	4.0 ms (2)	2.0 ms	8.0 ms (4)	2.0 ms	12.0 ms (6)				
13	2.0 ms	6.0 ms (3)	2.0 ms	8.0 ms (4)	2.0 ms	12.0 ms (6)				
14	2.0 ms	6.0 ms (3)	2.0 ms	10.0 ms (5)	2.0 ms	12.0 ms (6)				
15	2.0 ms	6.0 ms (3)	2.0 ms	10.0 ms (5)	2.0 ms	12.0 ms (6)				
16	2.0 ms	6.0 ms (3)	2.0 ms	10.0 ms (5)	2.0 ms	14.0 ms (7)	Same as on the left			

Note

- (1) The values in parentheses in the *Minimum communications cycle value* column are the PCU's communications cycle set value (multiplier to obtain integer multiples of the transfer cycle) used to set the minimum communications cycle for the minimum transfer cycle set value.
 - (2) The above table gives values for when an absolute encoder zero point position offset is set using an origin search operation when a Position Control Unit of unit version 2.0 is used. If this function is not being used, set the transfer cycle and communications cycle based on the table on page 245.
 - (3) The communications cycle values shown in the above table are the set values for W-series Servo Drives. The minimum communications cycle values for G5-series Servo Drives and G-series Servo Drives are the same as those indicated in the table above, but refer to the table on page 245 for the minimum transfer cycles.
 - (4) The settings of the transfer cycle and communications cycle have a large effect on the PCU's command response time. (Refer to *Command Response Time* in *Appendix A Performance Characteristics.*) As shown in the above table, the communications cycle (time) must be increased, e.g., particularly when linear interpolation is used at the same time as an origin search is used to set the absolute encoder zero point position offset (for different axes). The affect on system performance must thus be considered when making the settings. If it is not possible to increase the communications cycle, provide an interlock circuit between the operation of different axes so that linear interpolation is not performed for one axis

while an origin search is being performed for a motor with an absolute encoder on another axis.

6-3 MECHATROLINK Communications Control

MECHATROLINK communications control used for the PCU is described here.

6-3-1 Establishing Connections

Establishing a connection refers to starting communications between the PCU and the MECHATROLINK devices registered in the scan list. Connections are established by turning ON the CONNECT Bit in the Common Operating Output Memory Area. When CONNECT is executed, the PCU executes ERROR RESET (UNIT ERROR RESET or AXIS ERROR RESET) for the PCU itself and all the axes registered in the scan list.

For any error that occurs in an axis, ERROR RESET is executed when the connection with the corresponding axis is established. Therefore, if the cause of an error is removed, unless an error occurs due to connection processing, the PCU will start MECHATROLINK communications with all errors reset when CONNECT is executed.

The PCU status is initialized when a connection is established, as follows:

- PCU Positioning Completed Flag OFF
- Origin not established (No Origin Flag ON)

When using Servomotors with absolute encoders for which the encoder type is set to absolute encoder in the axis parameters, however, the origin is established (No Origin Flag OFF) by reading the absolute value data when the connection is established.

Stop Execution Flag OFF

For details on errors when establishing connections, refer to 6-3-3 MECHA-TROLINK Communications Errors. The CONNECT Bit is allocated in CIO 1501 + (unit number \times 25), bit 00.

Common Operating Memory Area (Output)

Name	Word	Bits	Contents
CONNECT Bit	n+1	00	$0 \rightarrow 1$: Establishes connection.
			$1 \rightarrow 0$: Releases connection.

 $n = CIO 1500 + (unit number \times 25)$

		00
Word n+1		MECHATROLINK communications start when this bit is turned ON.
n = CIO 1500 +	(unit number × 25)	

If the CONNECT Bit is turned OFF while a connection is established, the PCU will stop MECHATROLINK communications (connection released).

When a connection is released, the active axis is put in the Servo free run state.

For all axes registered in the scan list, the status that is input in the Axis Operating Input Memory Areas for each axis becomes the initial status (non-executing status), except for the Error Flag and axis error code. For example, the No Origin Flag turns ON because the origin is not established.

Note When a connection is released, the PCU immediately stops controlling the active axes. If a connection is released while another command is being processed, the Servo Drive may not respond to the command properly. Make sure that command processing has been completed (Busy Flag = 0 and Servo Parameter Transferring Flag = 0) before releasing a connection. When a connection is reestablished after recovering from an error that caused the PCU to release a connection, confirm that the settings and operating status of the Servo Drive are correct before restarting operations. (If necessary, transfer the Servo parameters again and execute DEVICE SETUP.)

6-3-2 MECHATROLINK Communications Status

The MECHATROLINK communications status is given using the following two status flags allocated in the Common Operating Input Memory Area of the PCU.

- Connection Status Flag
- Axis Communications Status Flags

Connection Status Flag This status flag turns ON when the PCU has established a connection (MECHATROLINK communications have started).

The PCU starts MECHATROLINK communications for the axes registered in the scan list. This status flag will turn ON, however, when CONNECT is executed, even if nothing is registered in the scan list.

The Connection Status Flag turns OFF when the connection is released. The Connection Status Flag is allocated in CIO 1516 + (unit number \times 25), bit 15.

Common Operating Memory Area (Input)

Name	Word	Bits	Contents
Connection Status Flag	n+16	15	0: MECHATROLINK communications stopped.
			1: MECHATROLINK communications exe- cuting.

 $n = CIO 1500 + (unit number \times 25)$

15 Word n+16	This flag turns ON when MECHATROLINK communications start and turns OFF when MECHATROLINK communications stop.					
Connection Status Flag						
n = CIO 1500 + (unit number \times 25)						

Axis Communications Status Flags

These status flags turn ON when a connection has been established between the PCU and the MECHATROLINK devices connected to the PCU. The Axis Communications Status Flags are allocated in one word with 16 bits, with one bit assigned to each of the 16 axes showing the corresponding communications status.

The PCU connects the axes registered in the scan list to MECHATROLINK communications, recognizing those axes and monitoring their communications. Therefore, the Axis Communications Status Flags for axes not registered in the scan list are always OFF.

Unit version	Setting conditions	Resetting conditions
Unit version 1.3 or earlier	• The flags will turn ON when connections are made for the axes registered in the scan list and MECHA- TROLINK communications start.	 The flags will remain OFF when MECHATROLINK communications cannot be started when connections are made for the axes regis- tered in the scan list. The flags will turn OFF if MECHATROLINK communi- cations stop because the axis is disconnected. The flags will turn OFF if a Unit error occurs that requires disconnection.
Unit version 2.0 or later	 The flags will turn ON when connections are made for the axes registered in the scan list and MECHA-TROLINK communications start. The flags will turn ON when the rejoin function is used to start MECHATROLINK communications for an axis registered in the scan list. 	 The flags will remain OFF when MECHATROLINK communications cannot be started when connections are made for the axes regis- tered in the scan list. The flags will turn OFF if MECHATROLINK communi- cations stop because the axis is disconnected. The flags will turn OFF if a Unit error occurs that requires disconnection. The flags will turn OFF whenever a communica- tions error occurs after MECHATROLINK communi- cations have been started for the axis.

The operation of the Axis Communications Status Flag depends on the unit version of the PCU as shown in the following table

With unit version 1.3 or earlier, once MECHATROLINK communications have been started by establishing connections, the Axis Communications Status Flags will not change unless communications are disconnected (including Unit errors that required disconnection).

With unit version 2.0 or later, the Axis Communications Status Flags will turn OFF after connections have been established whenever axis operation becomes impossible due to a communications error (synchronous communications alarm or communications alarm).

Refer to *6-3-3 MECHATROLINK Communications Errors* for information on errors that occur when establishing connections.

The Axis Communications Status Flags are allocated in CIO 1522 + (unit number \times 25).

Common Operating Memory Area (Input)

Name	Word	Bits	Contents
Axis Communica- tions Status Flags	n+22	00	0: Axis 1 communications stopped, or axis not registered in scan list.
			1: Axis 1 communications executing.
		01	Same as above for axis 2.
		02	Same as above for axis 3.
		03	Same as above for axis 4.
		04	Same as above for axis 5.
		05	Same as above for axis 6.
		06	Same as above for axis 7.
		07	Same as above for axis 8.
		08	Same as above for axis 9.
		09	Same as above for axis 10.
		10	Same as above for axis 11.
		11	Same as above for axis 12.
		12	Same as above for axis 13.
		13	Same as above for axis 14.
		14	Same as above for axis 15.
		15	Same as above for axis 16.

$n = CIO 1500 + (unit number \times 25)$



Timing Chart

The following diagram shows the timing chart for establishing and releasing a connection.



There is no change in status of the Unit Busy Flag, or the Receiving Command Flags or Busy Flags for the connected axes when a connection is established or released.

 $n = CIO 1500 + (unit number \times 25)$

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) \times 25



In the above example, Servo Drives are registered to axes 1, 2, 4, and 5 in the scan list (set to 40 hex). Therefore, MECHATROLINK communications start with the Servo Drives with station addresses No. 1, No. 2, No. 4, and No. 5 when CONNECT is executed.

Restrictions in Establishing a Connection According to Servo Drive Status

Functionality for axis operations after establishing a connection depends on the unit version of the PCU as described below if the following conditions exist when a connection is established for an axis (Servo Drive) registered in the scan list.

- Disconnection
- Control power supply interruption
- An alarm that cannot be reset (i.e., when the power must be cycled)

■ PCUs with Unit Version 1.2 or Earlier

An MLK initialization error (unit error code 0020 hex) will occur in these situations, and operation for MECHATROLINK communications will stop for all axes. To start MECHATROLINK communications normally, any errors must be cleared for all axes registered in the scan list before establishing a connection.

PCUs with Unit Version 1.3 or Later

Axis operation is possible for axes that are registered in the scan list and for which MECHATROLINK communications was started (i.e., axes for which Axis Communications Status Flags are ON), regardless of whether an alarm is present in the Servo Drive or whether the connected axes match the scan list. If MECHATROLINK communications cannot be started for all axes registered in the scan list, an MLK initialization error (unit error code 0020 hex) will occur to provide notification of the inconsistency between the registered axes and the axes that are communicating.

The following example shows the restrictions in operations according to axis status when connections are established.





Axis No.	No. 1	No. 2	No. 3	No. 4	No. 5
Axis status (example) Function	Normal	Control power sup- ply interrupted (See note 1.)	Normal	Alarm that can- not be reset (See note 2.)	Disconnection
Connection	Possible	Not possible	Possible	Possible	Not possible
Axis Communications Status Flag	ON	OFF	ON	ON	OFF
Monitoring (including error codes)	Possible	Not possible	Possible	Possible	Not possible

Axis No.	No. 1	No. 2	No. 3	No. 4	No. 5
Servo parameter trans- fer	Possible	Not possible	Possible	Possible	Not possible
Axis operating com- mands	Possible	Not possible	Possible	Not possible	Not possible

Note

 (1) It is assumed that the MECHATROLINK communications cable is connected.

(2) For example, when A.81 (backup error) occurs. MECHATROLINK communications may not be possible, depending on the individual Servo Drive specifications and on the type of alarm that cannot be reset.

For axes 1 and 3, MECHATROLINK communications can be started and all axis operations can be executed if there is no Servo Drive error. For axis 2 (with no control power supply) and axis 5 (disconnected), MECHATROLINK communications cannot be started, and operations and monitoring are not possible. If MECHATROLINK communications can be started for axis 4 (alarm that cannot be reset), depending on the handling of the Servo Drive error, the axis can be monitored and parameters can be transferred but axis operations cannot be executed. In this example, the axes for which MECHATROLINK communications can actually be started are 1, 3, and 4. This does not match the axes registered in the scan list (i.e., axes 1 to 5), so an MLK initialization error (unit error code 0020 hex) occurs.

6-3-3 MECHATROLINK Communications Errors

The PCU monitors and detects errors if they occur at the start of or during MECHATROLINK communications. Detected errors are largely categorized as PCU common errors that stop MECHATROLINK communications and operation of all axes, and axis errors that stop individual axes. Moreover, they are classified into three types of error: Errors at Communications Startup, Errors Detected by PCU during Communications, and Errors Detected in MECHATROLINK Slave Stations Connected to PCU during Communications.

Errors that Stop MECHATROLINK Communications (Unit Common Errors)

Errors at Communications	The following errors in MECHATROLINK (MLK) communications are moni-
Startup	tored by the PCU when CONNECT is executed.
	Errors in MECHATROLINK alove station deviace connect he detected by the

Errors in MECHATROLINK slave station devices cannot be detected by the PCU before a connection is established. The PCU will execute ALARM RESET for the devices communicating with MECHATROLINK when a connection is established.

MLK Device Initialization Error (Unit Error Code: 0030 Hex)

This error occurs if initialization of MECHATROLINK communications fails when CONNECT is executed, and causes connection processing to stop. When this error occurs, confirm that the PCU's common parameters and MECHATROLINK communications settings are set correctly, restart the PCU, and then execute CONNECT again. If the error occurs again, it is a result of an error in the MECHATROLINK communications section of the PCU and requires replacement of the PCU.

MLK Initialization Error (Unit Error Code: 0020 Hex)

This error occurs if the MECHATROLINK slave station device corresponding to the axis number registered in the scan list is not detected within 10 seconds after CONNECT is executed. This error may occur if a MECHATROLINK slave station communications settings are not set, or if the settings for the MECHA-TROLINK communications line connection and slave device's station address do not match the settings in the scan list. Verify that the contents registered in the scan list matches the MECHATROLINK device settings and connection configuration.

If this error occurs, the PCU will stop and the connection status will remain ON (communications still in an executing state).

For PCUs with unit version 1.2 or earlier, operation will not be possible for all axes registered in the scan list. For PCUs with unit version 1.3 or later, all or part of the axis operations will be possible assuming that MECHATROLINK communications can be established (i.e., if the Axis Communications Status Flag is ON). (Refer to *6-3-2 MECHATROLINK Communications Status* for axis operation restrictions according to Servo Drive status.)

After recovering from the error, release the connection and then execute CONNECT again. When using a Position Control Unit with unit version 2.0 or later, the REJOIN command can be used to rejoin the connection without breaking the connection.

MLK Communications Error (Unit Error Code: 0025 Hex)

This error occurs if two or more connected MECHATROLINK slave station devices have the same station number when CONNECT is executed. This error also occurs when communications errors that are not specific to a particular slave station occur during MECHATROLINK communications, and causes MECHATROLINK communications to stop (connection released). This error may occur during MECHATROLINK communications as a result of a faulty communications path such as noise, breaks, or faulty connections in the communications line. After removing the cause of the error, restart the PCU and execute CONNECT again.

MLK Device Error (Unit Error Code: 0026 Hex)

This error occurs if an error in the MECHATROLINK communications section of the PCU is detected, and causes MECHATROLINK communications to stop (connection released). This error will occur if the MECHATROLINK communications section of the PCU is faulty. Replace the PCU if this error occurs.

Transfer Cycle Setting Error (Unit Error Code: 0027 Hex)

This error may occur when establishing a connection if the transfer cycle set in the PCU's Common Parameters is unsuitable for the axis numbers or number of slave stations connected to MECHATROLINK, and causes MECHA-TROLINK communications to stop (connection released). This error indicates that the set value for the transfer cycle is too small for the axis numbers or number of connected slave stations. Change the transfer cycle setting in the Common Parameters. After removing the cause of the error, restart the PCU and execute CONNECT again.

If a Unit common error occurs, the Unit Error Flag in the Common Operating Memory Area of the PCU turns ON, and the corresponding error code is input in the Unit Error Code Bits. After the cause of the error is removed, the Unit common error is cleared by turning ON the UNIT ERROR RESET Bit in the Common Operating Memory Area.

When clearing a Unit common error, make sure that the UNIT ERROR RESET Bit will remain ON until the Unit Busy Flag turns ON.

The UNIT ERROR RESET Bit cannot be used to clear errors that require PCU replacement (MLK Device Initialization Error or MLK Device Error), or that cause MECHATROLINK communications to stop (MLK Communications Error or Transfer Cycle Setting Error). After removing the cause of the error, turn ON the power to the PLC again or restart the PCU.

Common Operating Memory Area (Output)

Name	Word	Bits	Contents
UNIT ERROR RESET Bit	n	00	$0 \rightarrow 1$: Clears the Unit common error that occurred.

 $n = CIO 1500 + (unit number \times 25)$

Common Operating Memory Area (Input)

Name	Word	Bits	Contents
Unit Error Flag	n+15	12	0: Unit common error has not occurred.
			1: Unit common error has occurred.
Unit Busy Flag	n+16	14	0: PCU is not busy.
			1: PCU is busy.
Connection Status Flag		15	0: MECHATROLINK communications stopped.
			1: MECHATROLINK communications exe- cuting.
Unit error code	n+21	00 to 15	Indicates the error code for the common Unit error that occurred.

 $n = CIO 1500 + (unit number \times 25)$

Individual Axis Errors

Errors Detected by PCU during Communications

Synchronous Communications Alarm (Axis Error Code: 3010 Hex)

This error occurs for errors in synchronous MECHATROLINK communications processing with an axis.

If this error occurs, the corresponding axis will stop operating according to the stop method that is set in the Servo Drive for when alarms occur.

This error may occur as a result of a faulty communications path with the corresponding axis, such as noise, breaks, or faulty connections in the communications line.

If this error occurs, the corresponding axis may not be able to accept operations through MECHATROLINK communications such as ERROR RESET. To recover from this error, release the connection and remove the cause of the error, and then execute CONNECT again. When using a Position Control Unit with unit version 2.0 or later, the REJOIN command can be used to rejoin the connection without breaking the connection.

Communications Alarm (Axis Error Code: 3011 Hex)

This error occurs for errors in MECHATROLINK communications processing with an axis. The communications alarm will occur if the PCU fails twice in succession to exchange data using MECHATROLINK communications, such as not receiving communications data. The corresponding axis will stop operating if this alarm occurs according to the stop method that is set in the Servo Drive for when alarms occur.

This error may occur as a result of a faulty communications path with the corresponding axis, such as noise, breaks, or faulty connections in the communications line. If this error occurs, the corresponding axis may not be able to accept operations through MECHATROLINK communications such as ERROR RESET. To recover from this error, release the connection, remove the cause of the error, and then execute CONNECT again. When using a Position Control Unit with unit version 2.0 or later, the REJOIN command can be used to rejoin the connection without breaking the connection.

Errors Detected in MECHATROLINK Slave Station Connected to PCU during Communications Errors detected in the slave stations connected to the PCU via MECHA-TROLINK after a connection is established cause axis errors or axis warnings in the PCU, as follows:

MLK Device Alarm/Warning (Axis Error Code 40 Hex)

The boxes ($\Box\Box$) indicate the alarm number for the connected MECHA-TROLINK devices. The operations of each axis when an error occurs depend on each of the MECHATROLINK slave station devices.

For errors in MECHATROLINK communications that occur in individual axes, the Error Flags for the corresponding axes turn ON in the Axis Operating Input Memory Areas of the PCU, and the corresponding error code is input in the axis error code word.

After the cause of the error is removed, the axis error is cleared by turning ON the ERROR RESET Bit in the Axis Operating Output Memory Area. When clearing the axis error, make sure that the ERROR RESET Bit will remain ON until the Busy Flag for the corresponding axis turns ON.

The ERROR RESET Bit cannot be used to clear errors such as Synchronous Communications Alarms or Communications Alarms. To recover from these errors, release the connection, remove the cause of the error, and then execute CONNECT again. When using a Position Control Unit with unit version 2.0 or later, the REJOIN command can be used to rejoin the connection without breaking the connection.

Axis Operating Output Memory Areas

Name	Word	Bits	Contents
ERROR RESET Bit	а	12	$0 \rightarrow 1$: Clears the axis error that occurred.

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

Axis Operating Input Memory Areas

Name	Word	Bits	Contents
Warning Flag	b	11	0: Axis warning has not occurred.
			1: Axis warning has occurred.
Error Flag		12	0: Axis error has not occurred.
			1: Axis error has occurred.
Busy Flag		13	0: Axis is not busy.
			1: Axis is busy.
Axis error code	b+4	00 to 15	Indicates the axis error code/warning code for each axis.

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) \times 25

For details on indicators and troubleshooting for when errors occur in the PCU, refer to *SECTION 12 Troubleshooting*. For details on error displays and troubleshooting for MECHATROLINK slave station devices, refer to the operation manuals for the corresponding device.

Note If an axis warning occurs (Warning Flag = 1), the PCU's command response time is delayed due to the PCU's warning processing. The PCU processing time required after receiving a command from the CPU Unit until the command is sent using MECHATROLINK communications is one MECHA-TROLINK communications cycle if an axis warning has not occurred, and three MECHATROLINK communications cycles if an axis warning has occurred. If the machine's operation would be affected by the PCU's command response time, clear the axis warning status using ERROR RESET or set the Servo Drive settings so that the warnings will not occur. For details on the PCU's command response time, refer to *Command Response Time* in *Appendix A Performance Characteristics*.

6-3-4 Rejoining the Connection

Overview

The rejoin function can be used to rejoin the connection for an axis for which operations are not possible using MECHATROLINK communications either when initially establishing the connection or after the connection has been established.

This function can be used only with Position Control Units with unit version 2.0 or later.

If any axis registered in the scan list is in either of the following conditions, the error cannot be cleared with the normal error reset procedure and axis operations will not be possible from the Position Control Unit.

- If an axis cannot be detected and MECHATROLINK communications cannot be started within 10 s after executing the connection command (An MLK initialization error (Unit error code: 0020 (hex)) will occur.)
- If a synchronous communications alarm (axis error code: 3010 (hex)) occurs or a communications alarm (axis error: 3011 (hex)) occurs

To restart communications for the axis for which axis operations are not possible with Position Control Units with unit version 1.3 or earlier, the connection must be broken and MECHATROLINK communications must be stopped for all axes, and then connection must be re-established.

With Position Control Units with unit version 2.0 or later, the rejoin function can be used to restart MECHATROLINK communications for specific axes without stopping MECHATROLINK communications for all axes.



Using the Rejoin Function

The rejoin function is used by turning ON the REJOIN command bit in the Common Operating Output Memory Area of the Position Control Unit.

When the REJOIN command bit is turned ON, the process to start MECHA-TROLINK communications is performed in the same way as when a connection is established. This can be used to restart communications for any axes registered in the scan list that are not part of the connection or for which axis operations are not possible due to a communications error.

When using the REJOIN command bit, the axis for which rejoin processing is to be performed can be specified in the Axes to Connect parameter. Refer to *6-3-5 Specifying the Axes to Connect* for details on the Axes to Connect parameter.

When rejoining the connection, errors that occur in the Position Control Unit or the axes will be reset.

Functionality for axis operations after rejoining a connection are the same as when establishing the connection.

The REJOIN command bit is bit 15 of CIO 1501+ (unit number \times 25). Keep the REJOIN command bit ON until the Unit Busy Flag turns ON.

Common Operating Output Memory Area

Name	Word	Bit	Contents
REJOIN	n+1	15	Rejoining the connection is started when this bit is turned ON.

 $n = 1500 + (unit number \times 25)$

Common Operating Input Memory Area

Name	Word	Bit	Contents
Unit Busy Flag	n+16	14	0: PCU is not busy.
			1: PCU is busy.
Connection Status Flag		15	0: MECHATROLINK commu- nications stopped.
			1: MECHATROLINK commu- nications executing.
Axis Communica- tions Status Flags	n+22	00 to 15	Indicate the communications status for each axis.

 $n = 1500 + (unit number \times 25)$

The REJOIN command bit can be used when the Connection Status Flag is ON (i.e., when a connection is established).

When executing processing for an axis to rejoin a connection, the Unit Busy Flag will turn ON for a minimum of one scan time and until the Axis Communications Status Flags turn ON for the axes being processed or until a MLK initialization error (Unit error code: 0020 (hex)) occurs after the REJOIN command bit is turned ON.

Note When the REJOIN command is executed, processing to restart MECHA-TROLINK communications is performed for all specified axes. If there are any axes for which MECHATROLINK communications cannot be started, communications errors will not occur for individual axes, but rather an MLK initialization error (Unit error code: 0020 (hex)) will occur for the entire Unit. If MECHATROLINK communications cannot be restarted when attempting to recover from communications errors for only specific axes, the error status will thus change from one for specific axes before rejoin processing to one for the entire Unit after rejoin processing. Monitor the status of these error flags to determine the result of rejoin processing.
Section 6-3

Timing Chart A timing chart for rejoin processing is provided below for Position Control Units with unit version 2.0 or later. In this example, axes 1 and 2 are registered in the scan list. After a connection is established, a communications error occurs for axis 2 (i.e., either a sync communications alarm or a communications alarm).



After removing the cause of the error, the REJOIN command bit is turned ON to clear the error for axis 2 and restart MECHATROLINK communications. The Axis Communications

Status Flag turns ON.

 $n = CIO 1500 + (unit number \times 25)$

b = First word specified in the Common Parameters for the Axis Operating Input Memory Area + (Axis No. -1) × 25

6-3-5 Specifying the Axes to Connect

The Axes to Connect parameter can be used to temporarily disable MECHA-TROLINK communications for one or more axes registered in the scan list. This enables starting MECHATROLINK communications without errors occurring for axes that are not connected without changing the scan list. This can be used, for example, during system design and startup when all axes are not yet connected.

This function can be used only with Position Control Units with unit version 2.0 or later.

The Axes to Connect parameter is in CIO 1502 + (unit number \times 25).

Common Operating Output Memory Area

Name	Word	Bit	Contents
Axes to connect	n+2	00	The following settings apply if axis 1 is registered in the scan list:
			0: MECHATROLINK communica- tions started for axis 1
			1: MECHATROLINK communica- tions not started for axis 1
		01	Same as above for axis 2
		02	Same as above for axis 3
		03	Same as above for axis 4
		04	Same as above for axis 5
		05	Same as above for axis 6
		06	Same as above for axis 7
		07	Same as above for axis 8
		08	Same as above for axis 9
		09	Same as above for axis 10
		10	Same as above for axis 11
		11	Same as above for axis 12
		12	Same as above for axis 13
		13	Same as above for axis 14
		14	Same as above for axis 15
		15	Same as above for axis 16

 $n = 1500 + (unit number \times 25)$

The Axes to Connect parameter is valid only when establishing a connection or when rejoining a connection.

Example

This example shows how to use a scan list in which axes 1 to 5 are registered and connect only axes 1, 2, and 5.

- Establishing a Connection for All Axes (n+2 = 0000 (Hex)) Axes 1, 2, and 5: MECHATROLINK communications started Axes 3 and 4: MECHATROLINK communications not started In this case, the connection cannot be established for axes 3 and 4, causing an MLK initialization error (Unit error code: 0020 (hex)) to occur in the Position Control Unit.
- Establishing a Connection for All Axes Except Axes 3 and 4 (n+2 = 000C (Hex))

Axes 1, 2, and 5: MECHATROLINK communications started Axes 3 and 4: MECHATROLINK communications not started In this case, axes 3 and 4 are treated as if they are not registered in the scan list and an MLK initialization error does not occur in the Position Control Unit.

Note The Axes to Connect parameter is valid only when using the CONNECT or REJOIN command bits in the Common Operating Memory Area. When establishing connections from the CX-Motion-NCF, the Axes to Connect parameter is ignored and processing to start MECHATROLINK communications is performed for all axes registered in the scan list.

If MECHATROLINK communications cannot be started when establishing a connection from the CX-Motion-NCF, an MLK initialization error (Unit error code: 0020 (hex)) regardless of the setting of the Axes to Connect parameter. Provide interlock circuits to allow for this as required by the system.

6-4 Standard Settings for Servo Drives Using MECHATROLINK

6-4-1 G5-series Servo Drive Settings (R88D-KND-ML2 with Built-in MECHATROLINK-II Communications)

When an R88D-KN -ML2 G5-series Servo Drive (with built-in MECHA-TROLINK-II communications) is used, the default settings for the Servo Drive are the standard settings when the MECHATROLINK communications is used.

Compulsory Parameter Settings

The following Servo Parameter must be set to enable PCU control. Do not change this setting.

Pn800 Communication Control

Parameter No.	Parameter name	Unit	Fixed setting		Details	Default setting
Pn800	Communica- tions control		0	0:	MECHATROLINK-II communications errors and warnings are all detected.	0

All MECHATROLINK communications errors and warnings are used when the PCU is controlling communications. Set the parameter so that they are all detected. Leave Pn800 set to the default setting.

Standard Settings for I/O Signals Using MECHATROLINK

When an R88D-KN -ML2 G5-series Servo Drive (with built-in MECHA-TROLINK-II communications) is used, the default settings for the Servo Drive are the standard settings for the I/O signal allocations when the MECHA-TROLINK communications is used.

Standard I/O Signal Allocation



Note The PCU uses the forward drive prohibited input, reverse drive prohibited input, and origin return deceleration limit switch allocated in the Servo Drive's external inputs as the forward rotation limit input signal, reverse rotation limit input signal, and origin proximity input signal, respectively. The signal widths of these input signals must be longer than the MECHATROLINK communications cycle. If the input signal width is shorter than the communications cycle, the PCU will not be able to detect the input signal, and operation will not be performed normally.

6-4-2 G-series Servo Drive Settings (R88D-GN_-ML2 with Built-in MECHATROLINK-II Communications)

When an R88D-GN -ML2 G-series Servo Drive (with built-in MECHA-TROLINK-II communications) is used, the default settings for the Servo Drive are the standard settings when the MECHATROLINK communications is used.

Compulsory Parameter
SettingsThe following Servo Parameter must be set to enable PCU control. Do not
change this setting.

Pn005	Communications	Control
-------	----------------	---------

Parameter No.	Parameter name	Unit	Fixed setting		Details	Default setting
Pn005	Communica- tions control		0	0:	MECHATROLINK-II communications errors and warnings are all detected.	0

All MECHATROLINK communications errors and warnings are used when the PCU is controlling communications. Set the parameter so that they are all detected. Leave Pn005 set to the default setting.

Standard Settings for I/O Signals Using MECHATROLINK

When an R88D-GN -ML2 G-series Servo Drive (with built-in MECHA-TROLINK-II communications) is used, the default settings for the Servo Drive are the standard settings for the I/O signal allocations when the MECHA-TROLINK communications is used.

The standard input signal settings are used for the PCU's positioning control functions, such as limit inputs, origin searches, and interrupt feeding. Use the default settings for I/O allocations of the Servo Drive.

Standard I/O Signal Allocation



Note The PCU uses the forward drive prohibited input, reverse drive prohibited input, and origin return deceleration limit switch allocated in the Servo Drive's external inputs as the forward rotation limit input signal, reverse rotation limit input signal, and origin proximity input signal, respectively. The signal widths of these input signals must be longer than the MECHATROLINK communications cycle. If the input signal width is shorter than the communications cycle,

the PCU will not be able to detect the input signal, and operation will not be performed normally.

6-4-3 W-series Servo Drive Settings (R88D-WT Equipped with FNY-NS115)

When the R88D-WT W-series Servo Drive equipped with a FNY-NS115 MECHATROLINK-II I/F Unit is used, settings for some parameters of the W-series user constants (Servo Parameters) may be limited.

Automatically Set Servo Parameters By equipping the W-series Servo Drive with the FNY-NS115, the Servo Parameters in the following table will be automatically changed to the fixed settings and saved when the power to the W-series Servo Drive is turned ON for the first time. These parameters are used by the system as fixed settings and must not be changed. (The default settings are those used for W-series Servo Drives before installing the FNY-NS115.)

Parameter No.	Parameter name	Unit	Fixed setting	Details	Default setting
Pn004	Reserved by the system.		0200		0000
Pn200	Position control setting 1		0100		1011
Pn204	Position command filter time constant 1	×0.01 ms	0	Sets soft start for command.	0
Pn207.1	Position control setting 2 Speed command input switching		1	REF used as feed-forward input	0
Pn50A.0	Input signal selection 1 Input signal allocation mode		1	User-defined sequence input signal allocation	0
Pn50A.1	Input signal selection 1 RUN signal (RUN command) input terminal allocation		8	Always invalid.	0
Pn50A.2	Input signal selection 1 MING (gain reduction) signal input terminal allocation		8	Always invalid.	1
Pn50B.1	Input signal selection 2 RESET (alarm reset) signal input terminal allocation		8	Always invalid.	4
Pn50C	Input signal selection 3 RDIR (rotation direction command signal) SPD1 (speed selection command 1 signal) SPD2 (speed selection command 2 signal) TVSEL (control mode switching signal)		8888	Always invalid.	8888
Pn50D.0	Input signal selection 4 PLOCK (position lock command signal)		8	Always invalid.	8
Pn50D.1	Input signal selection 4 IPG (pulse disable signal)		8	Always invalid.	8

Compulsory Parameter Settings

The following Servo Parameter must be set to enable PCU control. Do not change this setting.

Pn800.0 Communications Control: MECHATROLINK Communications Check Mask

Parameter No.	Parameter name	Unit	Fixed setting	Details	Default setting
Pn800.0	Communications control MECHATROLINK-II communications check mask		0	0: Communications errors (A.E6) and synchronization errors (A.E5) both detected.	0

The MECHATROLINK communications errors A.E6 (communications errors) and A.E5 (synchronization errors) are used when the PCU is controlling communications. Make sure to set this parameter so that these errors are always detected. Leave Pn800.0 set to the default setting 0 (detects communications errors (A.E6) and synchronization errors (A.E5)).

Pn800.1 Communications Control: Warning Check Mask

Parameter No.	Parameter name	Unit	Fixed setting	Details	Default setting
Pn800.1	Communications control Warning check mask		4 or 0	4: Communications warning (A.96) ignored.	4
				0: A.94, A.95, and A.96 all detected.	

The MECHATROLINK communications warnings A.94 (parameter setting warning) and A.95 (MECHATROLINK command warning) are used when the PCU is controlling communications. Set the parameter so that they are not ignored. Either leave Pn800.1 set to the default setting 4 (ignores communications errors (A.96) only) or set to 0 (detects A.94, A.95, and A.96).

Standard Settings for I/O Signals Using MECHATROLINK When MECHATROLINK communications are used by a W-series Servo Drive equipped with FNY-NS115, the standard settings for the Servo Drive's external I/O signal allocations are as follows:

The standard input signal settings are used for the PCU's positioning control functions such as limit inputs, origin searches, and interrupt feeding. When using a system configuration combining a W-series Servo Drive and FNY-NS115, use the standard settings for the I/O signal allocations. (The following default settings are those used by W-series Servo Drives before installing the FNY-NS115.)

I/O Signal Allocations (Standard Settings)



Parameter No.	Parameter name	Standard setting	Details	Default setting
Pn50A.3	Input signal selection 1 POT (forward drive prohibited input) signal input terminal allocation	2	Allocated to CN1, pin 42 Valid for low input (N.C. con- tact)	8
Pn50B.0	Input signal selection 2 NOT (reverse drive prohibited input) signal input terminal allocation	3	Allocated to CN1, pin 43 Valid for low input (N.C. con- tact)	8
Pn50B.2	Input signal selection 2 PCL (forward rotation current limit) signal input terminal allocation	8	Always invalid.	5

Standard Settings for Servo Drives Using MECHATROLINK

Parameter No.	Parameter name	Standard setting	Details	Default setting
Pn50B.3	Input signal selection 2 NCL (reverse rotation current limit) signal input terminal allocation	8	Always invalid.	6
Pn511.0	Input signal selection 5 DEC (origin return deceleration limit switch) sig- nal input terminal allocation	1	Allocated to CN1, pin 41 Valid for low input (N.O. con- tact)	8
Pn511.1	Input signal selection 5 EXT1 (external latch 1 input) signal input termi- nal allocation	4	Allocated to CN1, pin 44 Valid for low input (N.O. con- tact)	8
Pn511.2	Input signal selection 5 EXT2 (external latch 2 input) signal input termi- nal allocation	5	Allocated to CN1, pin 45 Valid for low input (N.O. con- tact)	8
Pn511.3	Input signal selection 5 EXT3 (external latch 3 input) signal input termi- nal allocation	6	Allocated to CN1, pin 46 Valid for low input (N.O. con- tact)	8
Pn50E.0	Output signal selection 1 INP1 (positioning completed 1) signal output ter- minal allocation	1	Allocated to CN1, pins 25 and 26	1
Pn50E.1	Output signal selection 1 VCMP (speed conformity) signal output terminal allocation	0	No output	1
Pn50E.2	Output signal selection 1 TGON (Servomotor rotation detection) signal output terminal allocation	0	No output	2
Pn50E.3	Output signal selection 1 READY (Servomotor warmup complete) signal output terminal allocation	3	Allocated to CN1, pins 29 and 30	3
Pn50F.0	Output signal selection 2 CLIMT (current limit detection) signal output ter- minal allocation	0	No output	0
Pn50F.1	Output signal selection 2 VLIMT (speed limit detection) signal output ter- minal allocation	0	No output	0
Pn50F.2	Output signal selection 2 BKIR (brake interlock) signal output terminal allocation	2	Allocated to CN1, pins 27 and 28	0
Pn50F.3	Output signal selection 2 WARN (warning) signal output terminal alloca- tion	0	No output	0
Pn510.0	Output signal selection 3 INP2 (positioning completed 2) signal output ter- minal allocation	0	No output	0
Pn510.1	Not used.	0		0
Pn510.2	Output signal selection 3 PSON (command pulse factor enabled) signal output terminal allocation	0	No output	0
Pn510.3	Not used.	0		0

Note The PCU uses the forward drive prohibited input, reverse drive prohibited input, and origin return deceleration limit switch allocated in the Servo Drive's external inputs as the forward rotation limit input signal, reverse rotation limit input signal, and origin proximity input signal, respectively. Make sure that these input signals maintain a signal width no shorter than the MECHA-TROLINK communications cycle. If the input signal width is shorter than the communications cycle, the PCU is unable to detect the input signals, and operations cannot be performed normally.

6-4-4 W-series Servo Drive Settings (R88D-WN□-ML2 with Built-in MECHATROLINK-II Communications)

When the R88D-WN -ML2 W-series Servo Drive equipped with built-in MECHATROLINK-II communications is used, the Servo Drive's default settings are used as the standard settings for MECHATROLINK communications.

Section 6-4

Compulsory Parameter Settings

The following Servo parameters must be set to enable PCU control. Do not change this setting.

Pn800.0 Communications Control: MECHATROLINK Communications Check Mask

Parameter No.	Parameter name	Unit	Fixed setting	Details	Default setting
Pn800.0	Communications control MECHATROLINK communications check mask		0	0: Communications errors (A.E6 ⁽⁾) and synchroniza- tion errors (A.E5 ⁽⁾) both detected.	0

The MECHATROLINK communications errors A.E6 \Box (communications errors) and A.E5 \Box (synchronization errors) are used when the PCU is controlling communications. Make sure to set this parameter so that these errors are always detected. Leave Pn800.0 set to the default setting 0 (detects communications errors (A.E6 \Box) and synchronization errors (A.E5 \Box)).

Pn800.1 Communications Control: Warning Check Mask

Parameter No.	Parameter name	Unit	Fixed setting	Details	Default setting
Pn800.1	Communications control Warning check mask		4 or 0	4: Communications warning (A.96) ignored.	4
				all detected. $A.95 \square$, and $A.96 \square$	

The MECHATROLINK communications warnings A.94 (parameter setting warning) and A.95 (MECHATROLINK command warning) are used when the PCU is controlling communications. Set the parameter so that they are not ignored. Either leave Pn800.1 set to the default setting 4 (ignores communications errors (A.96) only) or set to 0 (detects A.94 , A.95 , and A.96).

Standard Settings for I/O Signals Using MECHATROLINK When the R88D-WN□-ML2 W-series Servo Drive equipped with built-in MECHATROLINK-II communications is used, the Servo Drive's default I/O signal allocations are used as the standard settings for MECHATROLINK communications.

The standard settings for input signals use the position control functions, such as PCU limit inputs, origin search inputs, and interrupt feeding inputs. The default I/O signal allocations of the Servo Drive should thus be used.



I/O Signal Allocations (Default Settings)

Note The PCU uses the forward drive prohibited input, reverse drive prohibited input, and origin return deceleration limit switch allocated in the Servo Drive's external inputs as the forward rotation limit input signal, reverse rotation limit input signal, and origin proximity input signal, respectively. Make sure that these input signals maintain a signal width no shorter than the MECHA-TROLINK communications cycle. If the input signal width is shorter than the communications cycle, the PCU is unable to detect the input signals, and operations cannot be performed normally.

6-4-5 SMARTSTEP Junior Servo Drive Settings (R7D-ZN_-ML2 with Built-in MECHATROLINK-II Communications)

When the R7D-ZND-ML2 SMARTSTEP Junior Servo Drive Settings equipped with built-in MECHATROLINK-II communications is used, the Servo Drive's default settings are used as the standard settings for MECHATROLINK communications.

Compulsory ParameterThe following Servo parameter must be set to enable PCU control. Do not
change this setting.

Parameter No.	Parameter name	Unit	Fixed setting	Details	Default setting
Pn800.1	Communications control Warning check mask		4 or 0	4: Communications warning (A.96) ignored.	4
				0: A.94, A.95, and A.96 all detected.	

The MECHATROLINK communications warnings A.94 (parameter setting warning) and A.95 (MECHATROLINK command warning) are used when the PCU is controlling communications. Set the parameter so that they are not ignored. Either leave Pn800.1 set to the default setting 4 (ignores communications errors (A.96) only) or set to 0 (detects A.94, A.95, and A.96).

Standard Settings for I/O Signals When Using MECHATROLINK When the R7D-ZN□-ML2 SMARTSTEP Junior Servo Drive equipped with built-in MECHATROLINK-II communications is used, the Servo Drive's default settings are the standard settings for I/O signals for MECHATROLINK communications. The standard I/O signal settings are used for limit inputs, origin searches, interrupt feeding, and other position control functions of the Position Control Unit. Use the default settings for the I/O signal allocations for the Servo Drive.





Note The PCU uses the forward drive inhibit input, reverse drive inhibit input, and origin proximity input allocated in the Servo Drive's external inputs as the forward rotation limit input signal, reverse rotation limit input signal, and origin proximity input signal, respectively. Make sure that these input signals maintain a signal width no shorter than the MECHATROLINK communications cycle. If the input signal width is shorter than the communications cycle, the PCU is unable to detect the input signals, and operations cannot be performed normally.

SECTION 7 Position Control Structure

This section provides an overview of the control system used by the Position Control Unit, including information on the control units, coordinate system, acceleration/deceleration operations, and limit input operations.

7-1	PCU Co	ontrol System	272
	7-1-1	Control System Configuration and Principles	272
	7-1-2	Position Control	272
7-2	Control	Units	273
	7-2-1	Control Units for Position Control	273
	7-2-2	Control Units for Speed/Torque Control	274
	7-2-3	Units of the PCU's Main Controlled Variables	275
7-3	Coordin	nate System and Present Position	276
7-4	Acceler	ration and Deceleration Operations	277
	7-4-1	Overview of Acceleration/Deceleration Operations	277
	7-4-2	Acceleration/Deceleration Settings	278
	7-4-3	Acceleration/Deceleration Filter Settings	280
	7-4-4	Origin Search Acceleration/Deceleration Operations	284
	7-4-5	Acceleration and Deceleration Operations during Speed Control	285
7-5	Limit Iı	nput Operations	286

7-1 PCU Control System

This section describes the configuration, principles, and basic information on position control when using a control system configured with a PCU and a W-series Servo Drive.

7-1-1 Control System Configuration and Principles



The control system configured using the PCU basically controls Servomotor operations using the semi-closed loop method. The semi-closed loop method detects the number of Servomotor rotations for the command value using the rotary encoder mounted to the Servomotor, and sends this feedback as the machine's travel distance. While calculating the deviation between the command value and actual number of Servomotor rotations, the machine is controlled so that the deviation is compensated to 0.

In the PCU system configuration, a feedback system is configured in the Servo Drive, without using feedback information for the commands sent from the CPU Unit's ladder program to the PCU and Servo Drive.

7-1-2 Position Control

The control system configured using the PCU uses the Servo Drive's position loop to perform position control. The PCU achieves positioning by using the Servo Drive's position control functions together with control units and coordinate systems.



7-2 Control Units

7-2-1 Control Units for Position Control

Control Units for Position Control

The command units are used as the basic control units for the position command values and speed command values that the PCU uses in position control axis operations, such as direct operation, origin searches, and jogging. (The unit for speeds is command units/s.) These units are determined by the electronic gear ratio in the Servo Parameters. The Servo Parameter settings such as *Positioning Completion Range* and *Software Limit* and the present position monitor unit input in the CPU Unit are also based on these command units.

Model	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
R88D-KN□- ML2	Pn009	Electronic gear ratio numerator		1 to 1073741824	4	1
	Pn010	Electronic gear ratio denominator		1 to 1073741824	4	1
R88D-GN□- ML2	Pn205	Electronic gear ratio 1 (numerator)		0 to 131072	4	1
	Pn206	Electronic gear ratio 2 (denominator)		1 to 65535	4	1
R88D-WT	Pn202	Electronic gear ratio G1 (numerator)		1 to 65535	2	4
	Pn203	Electronic gear ratio G2 (denominator)		1 to 65535	2	1
R88D- WN□-ML2	Pn20E	Electronic gear ratio G1 (numerator)		1 to 1073741824	4	4
	Pn210	Electronic gear ratio G2 (denominator)		1 to 1073741824	4	1
SMART- STEP Junior	Pn20E	Electronic gear ratio G1 (numerator)		1 to 1073741824	4	1
	Pn210	Electronic gear ratio G2 (denominator)		1 to 1073741824	4	1

The *Electronic Gear Ratio G1/G2* parameters determine the ratio between the command unit and travel distance (pulse unit) of the Servomotor.

Set the electronic gear ratio such that $0.01 \le G1/G2 \le 100$ when using R88D-GN G-series Servo Drives, R88D-WT W-series Servo Drives, or SMART-STEP Junior Servo Drives and such that $0.001 \le G1/G2 \le 1000$ when using R88D-KN -ML2 G5-series Servo Drives or R88D-WN -ML2 W-series Servo Drives.

In the Servo Drive, the control unit is the pulse unit determined by the Servomotor's encoder resolution. (The Servo Drive operates internally with a pulse multiplier of four. Therefore, the Servomotor's encoder resolution multiplied by four is the number of pulses in one rotation.) The electronic gear ratio is used to convert the command value (i.e., command units) handled by the PCU into the control unit (i.e., pulses) used in the Servo Drive (the G1/G2 unit is pulses/command unit.) When the Servo Drive's default setting (G1/G2 = 4/1) is used, the command unit is the number of pulses equivalent to the Servomotor's encoder resolution.

Servomotor with 2,048 (Pulses/Rotation) Encoder When set to G1/G2 = 8192/1000, the command units for which the Servomotor makes one rotation for a position command value of 1,000 can be obtained.

Control Units

Section 7-2



- M: Deceleration gear ratio
- V: Feed speed of object being positioned (mm/s)
- P: Feed screw pitch (mm/rotation)
- R: Servomotor encoder resolution (pulses/rotation)

The equation for calculating the distance traveled per Servomotor encoder resolution pulse is as follows:

positioned

 $\frac{\text{Feed screw pitch}}{(\text{Servomotor encoder resolution} \times 4) \times (\text{deceleration ratio})} = \frac{P \text{ (mm/rotation)}}{R \text{ (pulses/rotation)} \times 4 \times M} \text{ (mm/pulse)}$

The Servo Drive performs control internally at four times the Servomotor's encoder resolution. Therefore, in the above equation, the Servomotor's encoder resolution is multiplied by four. If the command unit used is mm, the electronic gear ratio is set as the reciprocal of the above equation and G1/G2 is set as follows:

Command unit (mm): $\frac{G1}{G2} = \frac{R \text{ (pulses/rotation)} \times 4 \times M}{P \text{ (mm/rotation)}}$ (pulses/mm)

If the command unit used is 0.01 mm (= 10μ m), the unit is converted by multiplying the travel distance per encoder resolution pulse by 100, and the electronic gear ratio is set as follows:

 $\label{eq:command} \mbox{Command unit (0.01 mm):} \ \frac{G1}{G2} = \frac{R \ (\mbox{pulses/rotation}) \times 4 \times M}{P \ (\mbox{mm/rotation}) \times 100} \ (\mbox{pulses/0.01 mm})$

7-2-2 Control Units for Speed/Torque Control

The unit used for the speed command value in speed control is the percentage (unit: 0.001%) of the Servo Drive's/Servomotor's momentary maximum rotation speed. The unit used for the torque command value in torque control is the percentage (unit: 0.001%) of the Servo Drive's/Servomotor's momentary maximum torque. The Servomotor's momentary maximum rotation speed and momentary maximum torque depend on the Servomotor model.

For details on specifications, refer to the corresponding operation manuals for the Servo Drive and Servomotor.

7-2-3 Units of the PCU's Main Controlled Variables

Function Category Controlled variable		Unit	Details	
Common	Servo Parame- ter data	Torque limit value	%	(percentage of Servomo- tor's rated torque)
	Monitor value	Feedback present position	Command unit	
		Command present position	Command unit	
		Position deviation	Command unit	
Position control	Command value	Position command value	Command unit	
		Speed command value	Command units/s	
	Servo Parame-	Acceleration	10,000 command units/s ²	
	ler dala	Deceleration	10,000 command units/s ²	
		Zero point return approach speed 1	100 command units/s	
		Zero point return approach speed 2	100 command units/s	
	Monitor value	Speed monitor value	Command units/s	
		Torque monitor value	%	(percentage of Servomo- tor's rated torque)
Speed control	Command value	Speed command value (for speed control)	0.001%	(percentage of Servomo- tor's momentary maxi- mum rotation speed)
		Torque limit value (option command value)	%	(percentage of Servomo- tor's momentary maxi- mum torque)
	Monitor value	Speed monitor value	0.001%	(percentage of Servomo- tor's momentary maxi- mum rotation speed)
		Torque monitor value	%	(percentage of Servomo- tor's rated torque)
Torque control	Command value	Torque command value	0.001%	(percentage of Servomo- tor's momentary maxi- mum torque)
		Speed limit value (option command value)	0.001%	(percentage of Servomo- tor's momentary maxi- mum rotation speed)
	Monitor value	Speed monitor value	Command units/s	
		Torque monitor value	0.001%	(percentage of Servomo- tor's momentary maxi- mum torque)

The units for speed and torque depend on the control mode (position, speed, or torque control). For details, refer to *10-5 Speed Control*, and *10-6 Torque Control*.

Note The resolution of the controlled variables that can be monitored by the PCU depends on the specifications of the connected Servo Drive.

For example, when using a W-series Servo Drive equipped with a FNY-NS115 MECHATROLINK-II I/F Unit, the minimum resolution for the command speed monitor value and feedback speed in position/torque control is 1,000 command units/s.

Depending on the PLC's cycle time and MECHATROLINK communications cycle, short-term changes in status of the Servo Drive may not be monitored successfully.

7-3 Coordinate System and Present Position

The PCU uses a single coordinate system based on the origin determined for each axis using the origin defining functions. This coordinate system provides two types of present positions for each axis: a feedback present position and a command present position.

Feedback/Command Present Position The Servo Drive uses a rotary encoder to detect Servomotor rotations (providing feedback), and the Servomotor is controlled so that the deviation between the feedback and command values is compensated to 0. Therefore, while a position deviation remains in the Servo Drive, a difference equivalent to the position deviation between the position command value and the integrated value of the feedback, which indicates the actual machine (Servomotor) position, should exist.

The PCU outputs to the I/O memory area allocated for each axis in the CPU Unit the amount of feedback indicating the actual position of the machine (Servomotor) as the feedback present position and the position given in the command as the command present position.



Note

When executing speed control and torque control commands, there is no command present position because a positioning loop is not configured. A command present position calculated from the position deviation assumed from the current speed is output as monitor information.

Present Position Upper/Lower Limits The feedback present position and command present position can be displayed as values in the range -2,147,483,648 to 2,147,483,647 (command units). When an operation in a fixed direction without a target position using jogging, speed control, or torque control is continued, the present value that exceeds the above range and is not limited by limit input signals or software limits is given as follows:



For example, if the physical position is 5,368,709,120 (command units), as indicated by point P in the above diagram, which exceeds 2,147,483,647 (command units), 1,073,741,824 will be given as the present position (command units).

The position range for the ABSOLUTE MOVEMENT and RELATIVE MOVE-MENT commands is -2,147,483,648 to 2,147,483,647 (command units) for the command present position when they are not limited by limit input signals or software limits.

If the physical position is point P in the diagram, the physical position that can be moved to using ABSOLUTE MOVEMENT or RELATIVE MOVEMENT is 2,147,483,648 to 6,442,450,943 (command units). The PCU will treat the position command value and present position in the range as -2,147,483,648 to 2,147,483,647 (command units).

Note With the direct operation position range, the operation for position command values is different for ABSOLUTE MOVEMENT and RELATIVE MOVEMENT commands. Refer to *9-4-1 Starting Direct Operation* for further details.

7-4 Acceleration and Deceleration Operations

The acceleration and deceleration operations that can be used when the PCU is connected to a G5-series Servo Drive, G-series Servo Drive, W-series Servo Drive, or SMARTSTEP Junior Servo Drive are described here.

7-4-1 Overview of Acceleration/Deceleration Operations

The PCU creates the specified acceleration/deceleration curve for the operating commands performing position control, such as direct operation or origin searches. The acceleration/deceleration speed settings are set for each axis according to the Servo Parameters for acceleration/deceleration and are enabled for the following acceleration/deceleration operations.

Operating command	Acceleration operation	Deceleration operation
ABSOLUTE MOVEMENT/ RELA-	Starting	Stopping positioning
TIVE MOVEMENT	Changing speed (acceleration)	Changing speed (deceleration)
	Changing override (acceleration)	Changing override (deceleration)
JOG	Starting jogging	Stopping jogging
	Changing speed (acceleration)	Changing speed (deceleration)
	Changing override (acceleration)	Changing override (deceleration)
ORIGIN SEARCH	See 7-4-4 Origin Search Accelera- tion/Deceleration Operations	See 7-4-4 Origin Search Accelera- tion/Deceleration Operations

Operating command	Acceleration operation	Deceleration operation
DECELERATION STOP		Decelerating to a stop
Errors with DECELERATION STOP		When errors occur

Acceleration/deceleration curves cannot be set for speed control or torque control. Continuous speed changes using speed control, however, are possible by setting the Soft Start Acceleration Time and Soft Start Deceleration Time in the Servo Parameters.

7-4-2 Acceleration/Deceleration Settings

The acceleration/deceleration settings are set for each axis according to the Servo Parameters for acceleration/deceleration. Set these Servo Parameters using WRITE/SAVE SERVO PARAMETER before sending operating commands.

Do not change the acceleration/deceleration constants during axis operation (i.e., while the Busy Flag for the corresponding axis is ON due to execution of an operating command). Changing parameters while the axis is operating may result in incorrect positioning or other malfunction.

G5-series Servo Parameters

Туре	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting	G Series
Acceleration/ deceleration	Pn811	Linear acceleration constant	10,000 com- mand units/s ²	-32768 to 32767	2	100	Supported
constants	Pn814	Linear deceleration constant	10,000 com- mand units/s ²	-32768 to 32767	2	100	Supported

Acceleration/Deceleration Constants

Note

When using G5-series Servo Drives, one-step linear acceleration/deceleration curves are set using only Pn811 and Pn814.

When the parameter is set to 0, it is taken as 1.

The setting will be used after conversion to unsigned 16-bit data (0 to 65536).

Examples: -32768 is converted to 8000 hex = 32768

-1 is converted to FFFF hex = 65535

G-series Servo Parameters

Acceleration/Deceleration Constants

Туре	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting	G Series
Acceleration/ deceleration	Pn107	Linear acceleration constant	10,000 com- mand units/s ²	-32768 to 32767	2	100	Supported
constants	Pn10A	Linear deceleration constant	10,000 com- mand units/s ²	-32768 to 32767	2	100	Supported

Note

When using G-series Servo Drives, one-step linear acceleration/deceleration curves are set using only Pn107 and Pn10A.

When the parameter is set to 0, it is taken as 1.

The setting will be used after conversion to unsigned 16-bit data (0 to 65536).

Examples: -32768 is converted to 8000 hex = 32768

-1 is converted to FFFF hex = 65535

W-series and SMARTSTEP Junior Servo Parameters

Acceleration/Deceleration Constants

Туре	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting	W Series	SMARTS TEP Junior
Accelera- tion/ decelera- tion con- stants	Pn80A	First-step linear accel- eration constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Not sup- ported
	Pn80B	Second-step linear acceleration constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Sup- ported
	Pn80C	Acceleration constant switching speed	100 com- mand units/s	0 to 65535	2	0	Sup- ported	Not sup- ported
	Pn80D	First-step linear decel- eration constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Not sup- ported
	Pn80E	Second-step linear deceleration constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Sup- ported
	Pn80F	Deceleration constant switching speed	100 com- mand units/s	0 to 65535	2	0	Sup- ported	Not sup- ported

The SMARTSTEP Junior Servo Drives do not support the following parameters: Pn80A, Pn80C, Pn80D, and Pn80F. One-step linear acceleration/deceleration curves are set using only Pn80B and Pn80E.



To operate an axis using only a first-step acceleration/deceleration curve with a W-series Servo Drive, only settings for the parameters Pn80B and Pn80E are required if the default settings (0) are used for the Servo Parameters *Acceleration Constant Switching Speed* and *Deceleration Constant Switching Speed* (Pn80C and Pn80F).

Unit for Acceleration/Deceleration Constants

The unit for the acceleration/deceleration filters set in the Position Control Unit is 10000 command units/s² using the command unit determined by the electronic gear ratio in the Servo parameters.

The setting of the command unit can cause the setting unit for the acceleration/deceleration constants to become imprecise or changes in the setting can greatly affect the size of accelerations/decelerations.

To create a more precise setting unit for accelerations/decelerations, increase the resolution of the command unit (i.e., increase the number of digits).

Example

If the command unit is 0.1 mm, the target speed is 1,000 mm/s (1,000.0 mm/s = 10,000 command units/s) and the target speed must be reached in 250 ms, the acceleration constant would be set to 4, as shown below.

1000 [command units/s]/250 [ms] = 40,000 [command units/s²]

 $= 4 [10,000 \text{ command units/s}^2]$

If the acceleration constant is increased or decreased by only 1, the acceleration time required to reach the same target speed is greatly affected, as shown below:

Acceleration constant = 3:10000 [command units/s]/3 [10,000 command units/s²] = 333 [ms]

Acceleration constant = 5:10000 [command units/s]/5 [10,000 command units/s²] = 200 [ms]

If the precision of the command unit is increased by 2 digits to 0.001 mm, the acceleration constant would be set to 400 to reach the same target speed of 1,000 mm/s (1000.000 mm/s = 1,000,000 command units/s) in the same time of 250 ms.

1,000,000 [command units/s]/250 [ms] = 4,000,000 [command units/s²]

 $= 400 [10,000 \text{ command units/s}^2]$

If the acceleration constant is increased or decreased by 1, the acceleration time required to reach the same target speed is affected very little, as shown below:

Acceleration constant = 399:1000000 [command units/s]/399 [10000 command units/s²] = 250.6 [ms]

Acceleration constant = 401:1000000 [command units/s]/401 [10000 command units/s²] = 249.3 [ms]

7-4-3 Acceleration/Deceleration Filter Settings

When a W-series Servo Drive is connected, acceleration/deceleration filters can be set for each axis to avoid rapid changes in acceleration or deceleration. A acceleration/deceleration filter can be set for each axis to enable non-linear acceleration/deceleration curves. Using an acceleration/deceleration filter enables an exponential acceleration/deceleration curve and acceleration/decelerat

The G5-series Servo Drives and the G-series Servo Drives do not support exponential filters, but acceleration/deceleration S-curves using a moving average can be used.

The SMARTSTEP Junior Servo Drives do not support acceleration/deceleration filters and the filters cannot be set.

To use the acceleration/deceleration filter, set the acceleration/deceleration curve designation in the Axis Operating Output Memory Area when specifying the operating command as well as setting the acceleration/deceleration filter settings in the Servo Parameters.

Туре	Parameter name for G Series (Parameter name for W Series)	Unit	Setting range	Paramet er size	Default setting	G5 Series	G Series	W Series	SMART- STEP Junior
Accelera- tion/ decelera- tion filters	Exponential accelera- tion/deceleration bias	Com- mand units/s	0 to 32767	2	0	Not sup- ported	Not sup- ported	Pn810	Not sup- ported
	Exponential accelera- tion/deceleration time constant	0.1 ms	0 to 5100	2	0	Not sup- ported	Not sup- ported	Pn811	Not sup- ported
	Moving average time (Movement average time)	0.1 ms	0 to 5100	2	0	Pn818	Pn10E	Pn812	Not sup- ported

Servo Parameters

Acceleration/Deceleration Constants

For W-series Servo Drives, Pn810 and Pn811 must be set when using an exponential curve, and Pn812 must be set when using an S-curve for the acceleration/deceleration curve.

For G-series Servo Drives, Pn10E must be set when using an S-curve.

For G5-series Servo Drives, Pn818 must be set when using an S-curve.

Axis Operating Output Memory Areas

Name		Word	Bit	Contents	G5 Series G Series	W Series	SMARTSTE P Junior
Accelera- tion/decel- eration	Exponential curve desig- nation	a+16	03	1: Use exponential accelera- tion/deceleration curve.	Not sup- ported	Supported	Not sup- ported
curve des- ignation	S-curve des- ignation		04	1: Use S-curve accelera- tion/deceleration curve.	Supported	Supported	Not sup- ported

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) \times 25

The acceleration/deceleration curve designation setting is enabled when the bit for the movement command turns ON.

Note

- (1) Do not set both the exponential curve designation and S-curve designation to 1 (enabled) in the acceleration/deceleration curve designation. Enabling both settings may cause a malfunction.
 - (2) If an error occurs, the acceleration/deceleration curve for the deceleration stop operation will be that currently specified in the acceleration/deceleration curve designation, except for limit inputs and software limits. To use a non-linear acceleration/deceleration curve for the deceleration stop operation both when an error occurs and when operation is normal, make sure that the acceleration/deceleration curve designation is kept enabled during axis operations.
 - (3) Change the acceleration/deceleration constants using WRITE SERVO PARAMETER while axis operation is stopped (Busy Flag =0). Do not change parameters during axis operation. Changing parameters of an active axis may result in incorrect positioning or other malfunction.

Exponential Acceleration/Deceleration Curve



The acceleration/deceleration for the exponential acceleration/deceleration curve is determined by the Servo Parameters *Exponential Acceleration/Deceleration Bias* (Pn810) and *Exponential Acceleration/Deceleration Time Constant* (Pn811). The speed set in *Exponential Acceleration/Deceleration Bias* (Pn810) is used as the start speed. The time until reaching 63.2% of the difference between the speed command value and the start speed is set in the *Exponential Acceleration/Deceleration/Deceleration/Deceleration/Deceleration/Deceleration/Deceleration*.

When an exponential acceleration/deceleration curve is used, the acceleration/deceleration time is always fixed regardless of the speed command value.

S-curve (Movement Average) Acceleration/Deceleration Curve



In the S-curve acceleration/deceleration curve, a moving average is taken based on the Moving Average Time Interval (Pn812 (this is possible with Pn10E in the G series Servo Drive and with Pn818 in the G5 series Servo Drive)) for the linear acceleration/deceleration curve specified in Pn80A to Pn80F (Pn106 to Pn10B when using a G series Servo Drive, Pn811 and Pn814 when using a G5 series Servo Drive) to smooth out sudden changes in acceleration or deceleration (angles in the acceleration/deceleration curve). The acceleration/deceleration for the S-curve acceleration/deceleration curve will not exceed the maximum acceleration/deceleration of the linear acceleration/deceleration of the linear acceleration/deceleration curve prior to taking the moving average.

The time width where the average of the acceleration/deceleration curve is taken is set as the Moving Average Time Interval (Pn812 (set by Pn10E in the G series Servo Drive and by Pn818 in the G5 series Servo Drive)). For acceleration/deceleration curves having a fixed-speed region that is large enough for the target speed (speed command value), the time to reach the target speed by the S-curve acceleration/deceleration curve is slower than the linear acceleration/deceleration curve by the amount of the setting value of the Moving Average Time Interval (Pn812 (set by Pn10E in the G series Servo Drive and by Pn818 in the G5 series Servo Drive)).

If the constant speed interval for the target speed is less than the movement (running) average time or a triangular acceleration/deceleration curve occurs, the achieved speed of S-curve acceleration/deceleration will be less than the target speed.



7-4-4 Origin Search Acceleration/Deceleration Operations

The origin search operation of the PCU consists of the following three steps.

- 1,2,3... 1. Search for origin proximity signal
 - 2. Search for origin signal after detection of origin proximity signal
 - 3. Positioning for final travel distance to return to zero point (origin) after detection of origin signal



For details on the origin search, refer to 8-2 Origin Search Operation.

The acceleration when origin search starts and the speed change when switching to another step depend on the acceleration/deceleration curve (circled in above diagram) determined by the acceleration/deceleration filter setting and acceleration/deceleration set in Servo Parameters (Pn80A to Pn812 for W Series, Pn107 and Pn10A for G Series, Pn811 and Pn814 for G5 Series).

When the final travel distance to return to zero point (origin) is a negative or small value, the axis will decelerate to a stop as shown in the following diagram, and then reverse and perform the set positioning. Both the deceleration and acceleration will be determined also by the set acceleration/deceleration curve (circled in following diagram).



If the reversal mode operation using limit input is set, however, the stop operation when a limit input signal is input will depend on the Servo Drive setting. For details on stop operations when using limit input signal inputs, refer to 7-5 *Limit Input Operations* and *Reversal Mode Operation Using Limit Inputs* in 8-2-4 Origin Search Operation.

7-4-5 Acceleration and Deceleration Operations during Speed Control

When the speed control command is executed, the Servo Drive performs stepped speed control from the present feedback speed to the target speed specified in the speed command value for speed control. Apart from smoothly increasing/decreasing the speed command value (for speed control) when starting speed control, shock in acceleration/deceleration when performing speed control can also be minimized by setting the Servo Parameters *Soft Start Acceleration Time* (Pn312 for G5 Series, Pn058 for G Series and Pn305 for W Series) and *Soft Start Deceleration Time* (Pn313 for G5 Series, Pn059 for G Series and Pn306 for W Series) to create a trapezoidal speed curve.

For details on speed control, refer to 10-5 Speed Control.

Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
Pn312	Soft start accelera- tion time	×2ms	0 to 5000	2	0
Pn313	Soft start decelera- tion time	×2ms	0 to 5000	2	0

G5-series Acceleration/Deceleration Constants for Speed Control

G-series Acceleration/Deceleration Constants for Speed Control

Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
Pn058	Soft start accelera- tion time	×2ms	0 to 5000	2	0
Pn059	Soft start decelera- tion time	×2ms	0 to 5000	2	0

W-series Acceleration/Deceleratior	Constants for Speed Control
------------------------------------	-----------------------------

Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
Pn305	Soft start accelera- tion time	ms	0 to 10000	2	0
Pn306	Soft start decelera- tion time	ms	0 to 10000	2	0

The *Soft Start Acceleration Time* and *Soft Start Deceleration Time* are set respectively as the acceleration time and deceleration time between speed 0 (speed command value for speed control: 0%) and the maximum rotation speed of the Servomotor (speed command value for speed control: 100%). The acceleration time and deceleration time for the actual target speed are as follows:

Actual acceleration (deceleration) time = $\frac{\text{Speed command value for speed control (r/min)}}{\text{Maximum rotation speed (r/min)}} \times \text{Soft start acceleration (deceleration) time}$



- **Note** The maximum rotation speed depends on the Servomotor used. Refer to the maximum momentary rotation speed for the Servomotor used. The following values apply to W-series Servomotors.
 - 3,000-r/min Servomotor (cylinder type or flat type): 5,000 r/min
 - 1,000-r/min Servomotor: 2,000 r/min
 - 1,500-r/min Servomotor (450 W to 7.5 kW): 3,000 r/min
 - 1,500-r/min Servomotor (11 to 15 kW): 2,000 r/min

7-5 Limit Input Operations

The PCU uses the Servo Drive's internal forward drive prohibited signal input and reverse drive prohibited signal input as the forward rotation limit input signal and reverse rotation limit input signal, respectively. Additionally, the forward/reverse software limits can be used by setting the Servo Parameters.

The stop operation when the forward/reverse rotation limit input signal is input, and the forward/reverse software limit is reached depend on the Servo Parameter settings, as shown in the following table.

	Travel direction limit	Reverse travel direction limit (See note 1.)
Limit input sig- nal	The Servomotor operation is stopped at the Servo Drive according to the set- ting of Servo Parameter Pn001.1 (Pn504 for G5 Series, Pn004 for G Series). At the PCU, a limit input error occurs, causing an emergency stop, except when using the origin search opera- tion's reversal mode.	An emergency stop caused by a limit input error is executed at the PCU, except when exe- cuting ORIGIN SEARCH.
Software limit	The Servomotor operation is stopped at the Servo Drive according to the set- ting of Servo Parameter Pn001.1 (Pn504 for G5 Series, Pn004 for G Series). (See note 2.) At the PCU, a software limit error occurs, causing an emergency stop.	

1. This operation occurs when a limit input signal is input in the direction opposite to the travel direction due to incorrect wiring or other cause.

2. For details on software limit functions, refer to 10-8 Software Limits.

Servo Parameters

- Note (1) If the limit input signal inputs (Servo Drive's forward drive prohibited signal and reverse drive prohibited signal) are not allocated, the Servo Drive will not stop the Servomotor when the signal is input, and the PCU will also not detect limit inputs as errors. When using a PCU, always allocate the Servo Drive's forward drive prohibited signal and reverse drive prohibited signal to enable use of the limit input signals. (Refer to 6-4 Standard Settings for Servo Drives Using MECHATROLINK.)
 - (2) Make sure that the limit input signals maintain a signal width no shorter than the MECHATROLINK communications cycle. Input signals that are shorter than the communications cycle cannot be detected by the PCU.

When using a G5-series Servo Drive, a G-series Servo Drive or a W-series Servo Drive, the Servomotor's stop operation at the Servo Drive when a limit input signal is input or a software limit is reached is set in the Servo Parameters for limit input operations.

When using a SMARTSTEP Junior Servo Drive, the Servo is locked after decelerating to a stop with the maximum Servomotor torque. The following function cannot be used.

Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
Pn505	Stop selection for drive prohibition input		0 to 2	2	0
Pn511	Emergency stop torque	%	0 to 500	2	0

G5-series Limit Input Parameters

Pn505 is used to set the stop operation of the Servomotor when a forward/reverse rotation limit is input or a forward/reverse software limit is reached. The parameter size of Pn505 is the value to be specified when transferring the Servo Parameter Pn505 including the above settings.

The following table shows the Servomotor stop operations using Pn505 settings.

Parameter No.	Setting	During deceleration	After stopping	Deviation counter
Pn505	0	Dynamic brake	Disables torque command in drive prohibited direction	Cleared while decelerating with dynamic brake. Retained after stopping.
	1	Disables torque	Disables torque command in drive prohibited direction	Cleared while decelerating. Retained after stopping.
	2	Emergency stop torque (Pn511)	Servo lock	Retained while decelerating, cleared upon com- pletion of decelera- tion, and retained after stopping.

Note

(1) If an operation command in the drive prohibited direction is received when the torque command of the drive prohibited direction is disabled, the Servomotor will not operate in the drive-prohibited direction, but the position deviation will accumulate because the position command is enabled. If drive prohibition is detected, stop the operation command from the host controller and take measures such as executing a command in the reverse direction.

- (2) The positioning command generation process (positioning operation) in the Servo Drive will be forcibly stopped after it enters the deceleration mode. Also, when the deceleration mode is activated during speed control or torque control, it will switch to position control.
- (3) When the Servomotor rotation speed is 30 r/min or less (stopped), the Servomotor does not enter into deceleration mode even if the drive prohibit input is enabled.

G-series	Limit	Input	Parameters	S
----------	-------	-------	------------	---

Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
Pn066	Stop selection for drive prohibition input		0 to 2	2	0
Pn06E	Emergency stop torque	%	0 to 300	2	0

Pn066 is used to set the stop operation of the Servomotor when a forward/reverse rotation limit is input or a forward/reverse software limit is reached. The parameter size of Pn066 is the value to be specified when transferring the Servo Parameter Pn066 including the above settings.

The following table shows the Servomotor stop operations using Pn066 settings.

Parameter No.	Setting	During deceleration	After stopping	Deviation counter
Pn066	0	Dynamic brake	Disables torque command in drive prohibited direction	Cleared while decelerating with dynamic brake. Retained after stopping.
	1	Disables torque	Disables torque command in drive prohibited direction	Cleared while decelerating. Retained after stopping.
	2	Emergency stop torque (Pn06E)	Servo lock	Retained while decelerating, cleared upon com- pletion of decelera- tion, and retained after stopping.

Note

- (1) If an operation command in the drive prohibited direction is received when the torque command of the drive prohibited direction is disabled, the Servomotor will not operate in the drive-prohibited direction, but the position deviation will accumulate because the torque command is enabled. If drive prohibition is detected, stop the operation command from the host controller and take measures such as executing a command in the reverse direction.
 - (2) The positioning command generation process (positioning operation) within the Servo Drive will be forcibly stopped after it enters the deceleration mode. Also, when the deceleration mode is activated during speed control or torque control, it will switch to position control.

(3) When the Servomotor rotation speed is 30 r/min or less (stopped), the deceleration mode will not be entered even if the drive prohibit input is enabled.

Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
Pn001.1	Function selection application switch 1		0 to 2	2	0
	Select stop when prohibited drive is input				
Pn406	Emergency stop torque	%	0 to 800	2	350

W-series and SMARTSTEP Junior Limit Input Parameters

Pn001.1 is used to set the stop operation of the Servomotor when a forward/reverse rotation limit input is input or a forward/reverse software limit is reached. The parameter size of Pn001.1 in the above table is the set value to be specified when transferring the Servo Parameter Pn001 (including Pn001.1).

The following table shows the Servomotor stop operations using Pn001.1 settings.

Parameter No.	Set value	Details			
Pn001.1	0	Stops operatio stop if an alarn	n accordi n occurs	ing to the setting in Pn001.0 (selects when Servomotor is OFF).	
		Puts Servomotor in Servo unlock status after stopping op tion.			
		Parameter No.	Set value	Details	
		Pn001.0	0	Servomotor stopped by dynamic brake.	
				Dynamic brake held after stopping.	
			1	Servomotor stopped by dynamic brake.	
				Dynamic brake OFF after stopping.	
			2	Servomotor stopped with free run.	
1		Stops Servomotor using torque set in <i>Emergency Stop Torque</i> (Pn406).			
		Puts Servomotor in Servo lock status after stopping operation.			
	2	Stops Servomotor using torque set in <i>Emergency Stop Torque</i> (Pn406).			
		Puts Servomot tion.	tor in Ser	vo unlock status after stopping opera-	

Note The Pn406 default setting corresponds to that used for W-series Servo Drives.

Limit Input Operations

Section 7-5



Note The brake interlock output (BKIR output) will not function while there is a limit input signal (Servo Drive's forwarded drive prohibited signal or reverse drive prohibited signal) or while the Servo is unlocked as the result of reaching a software limit. Unlock the Servo before activating a machine holding brake.

SECTION 8 Defining the Origin

This section provides information on the various operations used to determine the origin, including origin searches, origin returns, presetting the present position, calculating phase Z margins, and using the absolute encoder.

8-1	Overvie	W	292
8-2	Origin S	Search Operation	293
	8-2-1	Origin Search Overview	293
	8-2-2	Origin Search Procedure.	294
	8-2-3	Data Settings Required for Origin Search	295
	8-2-4	Origin Search Operation	305
	8-2-5	Starting Origin Search	314
	8-2-6	Origin Search Preset	315
8-3	Present	Position Preset	317
	8-3-1	Outline of Operation	317
	8-3-2	Data Settings for Present Position Preset	317
8-4	Origin F	Return	319
	8-4-1	Origin Return Data Settings	319
8-5	Phase Z	Margin	324
	8-5-1	Description	324
	8-5-2	Calculating the Phase Z Margin (Example)	324
	8-5-3	Phase Z Margin for Specific Operation Patterns	325
8-6	Absolut	e Encoder Origin	326
	8-6-1	Absolute Encoder Overview	326
	8-6-2	Absolute Encoder Operating Procedure	327
	8-6-3	PCU Data Settings for Using Absolute Encoders.	329
	8-6-4	Absolute Encoder Setup	331
	8-6-5	Absolute Encoder's Origin (Zero Point) Position Offset Setting	332
	8-6-6	Establishing the Origin Using an Absolute Encoder	334

8-1 Overview

When performing positioning for a system in terms of absolute positions in that system, it is first necessary to define the origin. For example, in the X-Y plane shown below, before positioning to (X, Y) = (100 mm, 200 mm), the origin (0,0) of the machine must be established, i.e., it is necessary to define the origin.



With the PCU, the mechanical origin can be defined using any of the following three methods.

Name	Contents
ORIGIN SEARCH	The origin is determined by actually operating the Servomotor and using the limit input signal, origin proximity input signal, and origin input signal.
	A photoelectric sensor or the phase Z signal from an encoder can be used as the origin input signal.
	Execute this function when using an incremental encoder Ser- vomotor or Servomotor with absolute encoder used as an incremental encoder.
PRESENT POSI- TION PRESET	The origin can be determined by forcibly setting the position at which the Servomotor is stopped to a user-specified position.
ABSOLUTE ENCODER ORIGIN	By using a Servomotor with absolute encoder, the absolute value data saved in the encoder can be read and the mechan- ical origin can be established when starting to use the device.

The PCU also features a positioning function to return to the established origin.

Name	Contents			
ORIGIN RETURN	Positioning is performed from the position where the Servo- motor is stopped to the origin that has been established using ORIGIN SEARCH or PRESENT POSITION PRESET.			

The PCU can also be operated without defining the origin but the axis operation for each function is as follows:

Function	Operation			
JOG	Can be executed.			
ORIGIN RETURN	Cannot be executed.			
	If executed, a Present Position Unknown Error (axis error code: 3030 hex) will occur.			

	Function	Operation		
Position control	ABSOLUTE MOVE-	Cannot be executed.		
	MENT	If executed, a Present Position Unknown Error (axis error code: 3030 hex) will occur.		
	RELATIVE MOVE- MENT	Can be executed.		
	INTERRUPT FEED- ING	Can be executed only with RELATIVE MOVE- MENT.		
		If executed with ABSOLUTE MOVEMENT, a Present Position Unknown Error (axis error code: 3030 hex) will occur, and the function will not be executed.		
SPEED CONTROL		Can be executed.		
TORQUE CONTROL		Can be executed.		

For G5-series Servo Drives, G-series Servo Drives and W-series Servo Drives used with incremental encoders or absolute encoders used as incremental encoders, the present position is taken as 0 when the power is turned ON. When the executable functions shown in the above table are executed by the PCU without a defined origin, the present position will be calculated from the position of the axis (0) when the power was turned ON with no origin established.

8-2 Origin Search Operation

8-2-1 Origin Search Overview

Origin searches are performed by actually operating the Servomotor, searching for the mechanical origin based on the limit input signal, origin proximity input signal, and origin input signal, and to define the origin.

Note In ORIGIN SEARCH execution, the present position is cleared to 0 at the same time when an origin is defined. Unless the Absolute Encoder Zero Point Position Offset is set using an Origin Search Operation with Position Control Units with unit version 2.0 or later, the absolute value data read to the PCU when SERVO LOCK was executed will be lost when the origin is established. (The absolute data saved in the absolute encoder will not be lost and can be obtained by executing SERVO LOCK after SERVO UNLOCK.) When using a Position Control Unit with unit version 2.0 or later and not setting an Absolute Encoder Zero Point Position Offset, execute ORIGIN SEARCH using either a Servomotor with incremental encoder or a Servomotor with absolute encoder that is used as an incremental encoder.



Origin Search Operation Diagram for W-series Servo Drives

a: Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) \times 25 b: Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25 d: 1860 hex + (Axis No. -1) × 14 hex

The speed command value set in the Axis Operating Output Memory Area using the MOV instruction is output to the PCU automatically during I/O refresh.

The origin search is started when the ORIGIN SEARCH Bit allocated in the Axis Operating Output Memory Area turns ON.

Note Once the origin is established by the PCU, it will not be lost if ORIGIN SEARCH is executed again. If ORIGIN SEARCH is executed again, the search operation will be executed using the coordinates based on the previous origin until the new origin is established. If the software limits are enabled at this point, the software limits will apply to the origin search operation, causing operation to stop due to an error. (Positioning will not be reversed at the software limits when origin searches are performed in reversal mode.) To execute ORIGIN SEARCH again after the origin has been established, either disable the software limits, or execute DEVICE SETUP or another operation that will cancel the origin (No Origin Flag = 1) before executing ORIGIN SEARCH.

Origin Search Procedure 8-2-2

Use the following procedure to perform an origin search.

- 1,2,3... Set the Common Parameters and save them.
 - 2. Set the Axis Parameters and save them.

- Turn ON the PCU again or restart the Unit. The data for the Common Parameters and Axis Parameters set in steps 1 and 2 above are enabled.
- 4. Start MECHATROLINK communications.
- Set the Servo Parameters and save them. Set the Servo Parameters required to execute origin searches. To set parameters permanently, execute SAVE SERVO PARAMETER (writes to the non-volatile memory) To enable changed offline parameters, turn the power to the Servo Drive OFF and then ON again or execute DEVICE SETUP. For details, refer to *5-3 Transferring Servo Parameters*.
- 6. Execute SERVO LOCK.
- Set the data used for origin search. Set the origin search speed data in the speed command value of the Axis Operating Output Memory Area.
- 8. Execute ORIGIN SEARCH.

When using the PCU for the first time or to change the PCU parameter settings, steps 1 to 3 must be performed.

Note To specify the Servo Parameters for acceleration/deceleration every time, make sure that the axis operation is stopped (Busy Flag = 0) while changing the acceleration/deceleration constants using WRITE SERVO PARAMETER. Do not change the settings during axis operation. Changing parameters while the axis is operating may result in incorrect positioning or other malfunction.

8-2-3 Data Settings Required for Origin Search

A simple explanation of the main parameters and data used to execute ORI-GIN SEARCH is provided here. For details on the parameters required when using absolute encoders, refer to *8-6 Absolute Encoder Origin*.

To execute ORIGIN SEARCH, apart from the parameters explained here, the following parameters also need to be set as basic settings for operating the PCU.

- External I/O Signal Allocations Refer to *6-4 Standard Settings for Servo Drives Using MECHATROLINK*. The origin proximity input signal must be allocated to execute ORIGIN SEARCH.
- Command Unit Refer to 7-2 Control Units.
 The setting units for parameters and data depend on the specified command unit.

Axis Parameter Area

The following parameters are for the operation mode for origin searches.

PCU's		Contents				Setting	
address	15 to 14	13	12	11 to 08	07 to 04	03 to 00	
d+1	0 (fixed)	Origin search preset	Origin search direction	Origin detection method	Origin search operation	0 (fixed)	 Set the origin search operation. Origin search direction Reversal mode 1 (default setting) Reverse mode 2 Single-direction mode Reversal mode 3 (See note.) Origin detection method (See note.) With origin proximity input signal reversal (default setting) Without origin proximity input signal reversal Not use origin proximity input signal Porigin search direction Forward (default setting) Reverse Origin search preset (See note.) Not Set (default setting) Set
d+2	0 (fixed)	0	0	0 (fixed)	0 (fixed)	Encoder type	0: Incremental encoder (default setting) 1: Absolute encoder

 $d = 1860 \text{ hex} + (\text{Axis No.} -1) \times 14 \text{ hex}$

* The reversal mode 3 setting for the origin search operation, the origin detection method, and the origin search preset can be used only with Position Control Units with unit version 2.0 or later. They cannot be used with Position Control Units with unit version 1.3 or earlier.

Note

- (1) Set the origin search direction of the axis parameters and Origin Search Direction for G5-series Servo Drives (Pn822), Origin Return Mode Settings for G-series Servo Drives (Pn10F) or Zero Point Return Mode Setting for W-series Servo Drives (Pn816) to the same direction. Setting different direction may result in a malfunction.
 - (2) When using an absolute encoder, make sure that the settings for the encoder type in the Axis Parameters and *Operation Switch when Using Absolute Encoder* (Pn002.2 for W Series, Pn00B for G Series and Pn015 for G5 Series) in the Servo Parameters match. If the settings do not match, ORIGIN SEARCH execution will not be possible, or another malfunction may occur.
 - (3) Make sure that the external input signals for origin searches (limit input signal and origin proximity input signal) maintain a signal width no shorter than the MECHATROLINK communications cycle. If the input signal width is shorter than the communications cycle, the PCU is unable to detect the input signals, and operations cannot be performed normally.
| Axis Parameter
Area | G5-series Servo Parameter Pn015 (Operation Switch when Using Absolute Encoder) | | | | |
|---------------------------|--|--|--|--|--|
| Encoder type | G-series Servo Parameter Pn00B (Operation Switch wh
Using Absolute Encoder) | | | | |
| | 0: Use as an absolute encoder. | 1: Use as an incremental encoder. | | | |
| | 2: Use as an absolute
encoder but ignore abso-
lute multi-turn counter over-
flow alarm | | | | |
| 0: Incremental
encoder | Do not set this combination. | This setting is used when
using an absolute encoder as
an incremental encoder. | | | |
| 1: Absolute encoder | This setting is used when
using an absolute encoder as
an absolute encoder. | Do not set this combination. | | | |

Axis Parameter Area	W-series Servo Parameter Pn002.2 (Operation switch when using absolute encoder)			
Encoder type	0: Use as absolute encoder.	1: Use as incremental encoder.		
0: Incremental encoder	Do not set this combination.	This setting is used when using an absolute encoder as an incremental encoder.		
1: Absolute encoder	This setting is used when using an absolute encoder as an absolute encoder.	Do not set this combination.		

When using a G5-series or G-series incremental encoder, only the encoder type setting in the Axis Parameters is required (When using the incremental encoder, the settings for the Servo Parameter Pn015 (G5 series) and Pn00B (G series) are disabled).

When using a W-series incremental encoder, only the encoder type setting in the Axis Parameters is required (the Pn002.2 setting in Servo Parameters is disabled when an incremental encoder is used).

An absolute encoder cannot be used with a SMARTSTEP Junior Servo Drive, and Pn002.2 is not supported. Use the Servo Drive as if the axis parameters are set for an incremental encoder.

Origin Input Signal Parameters

PCU's	Con	tents	Setting
address	Bits 15 to 08	Bits 07 to 00	
d	Origin input sig- nal selection	Interrupt input signal selection	Origin input signal selection 00: Phase Z (default setting) 01: External latch signal 1 input 02: External latch signal 2 input 03: External latch signal 3 input

 $d = 1860 \text{ hex} + (\text{Axis No.} -1) \times 14 \text{ hex}$

Use this parameter to select the origin input signal to be used with origin search. When external latch signals 1 to 3 are used as the origin input signal, the external latch signal used must be allocated in the Servo Drive's external input allocations.

When using a SMARTSTEP Junior Servo Drive, only the Z phase or the external latch signal 1 input can be set for the origin input signal.

Servo Parameters

G5-series Limit Input Operation Parameters

Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
Pn505	Stop selection for drive prohibition input		0 to 2	2	0
Pn511	Emergency stop torque	%	0 to 500	2	0

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting
	Pn066	Stop selection for drive prohibition input		0 to 2	2	0
	Pn06E	Emergency stop torque	%	0 to 500	2	0
	Pn004	Drive Prohibit Input Selection		0 to 2	2	0

G-series Limit Input Operation Parameters

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting	W Series	SMART- STEP Junior
	Pn001.1	Function selection application switch 1 Select stop when prohibited drive is input		0 to 2	2	0	Sup- ported	Not sup- ported
	Pn406	Emergency stop torque	%	0 to 800	2	350	Sup- ported	Not sup- ported

* The Pn406 default setting corresponds to that used for W-series Servo Drives.

The stop operation (Pn505 for G5 Series, Pn066 for G Series and Pn001.1 for W Series) sets the stop operation of the Servomotor for forward/reverse rotation limit inputs when prohibited drive is input.

When reversal mode is used for the origin search, it is also used as the operation setting when limit inputs are input in the origin search direction. The parameter size for Pn001.1 for W-series Servo Drives is the value to be specified when transferring the Servo Parameter Pn001 (including Pn001.1).

The emergency stop torque (Pn511 for G5 Series, Pn06E for G Series and Pn406 for W Series) sets the stopping torque when the stop operation for limit inputs is set to 1 (decelerates to a stop at set torque or lower, and locks Servo in zero clamp mode), or 2 (decelerates to a stop at set torque or lower, and puts Servo in free run state).

When using a SMARTSTEP Junior Servo Drive, the Servomotor will decelerate to a stop with the maximum Servomotor torque and the Servo will be locked when the forward or reverse rotation limit input is received.

Drive Prohibit Input Selection (G series: Pn004) sets the forward and reverse drive prohibition input (CH1 POT: pin 19, NOT: pin 20) functions.

If reversal mode 1 is used in origin search, set to 0 or 1.

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting
	Pn015	Operation switch when using absolute encoder		0 to 2	2	1

Parameters for Using G5-series Absolute Encoders

Parameters for Using	G-series	Absolute	Encoders
----------------------	----------	----------	----------

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting
	Pn00B	Operation switch when using absolute encoder		0 to 2	2	0

Parameters for Using W-series Absolute Encoders

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting	W Series	SMART- STEP Junior
	Pn002.2	Function selection application switch 2 Operation switch when using abso- lute encoder		0, 1	2	0	Sup- ported	Not sup- ported

When using a Servomotor with absolute encoder, select whether the absolute encoder will be used as an absolute encoder or as an incremental encoder. These parameter settings are disabled if a Servomotor with incremental encoder is used.

Set this parameter to 1 or 2 for a G-series Servo Drive, and 1 for a W-series Servo Drive, if using a Servomotor with absolute encoder as a Servomotor with incremental encoder.

When using an absolute encoder, make sure that the settings for the encoder type in the Axis Parameters and *Operation Switch when Using Absolute Encoder* (Pn00B for G Series and Pn002.2 for W Series) in the Servo Parameters match.

A motor with an absolute encoder cannot be used with a SMARTSTEP Junior Servo Drive. When using a SMARTSTEP Junior Servo Drive, Pn002.2 is not supported. Use the Servo Drive as for an incremental encoder.

If the settings do not agree, origin searches may not be possible and other malfunctions may occur.

Model	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
R88D-KN⊡- ML2	Pn009	Electronic gear ratio numerator		1 to 1073741824	4	1
	Pn010	Electronic gear ratio denominator		1 to 1073741824	4	1
R88D-GN□- ML2	Pn205	Electronic gear ratio 1 (numerator)		1 to 1073741824	4	1
	Pn206	Electronic gear ratio 2 (denominator)		1 to 1073741824	4	1
R88D-WT	Pn202	Electronic gear ratio G1 (numerator)		1 to 65535	2	4
	Pn203	Electronic gear ratio G2 (denominator)		1 to 65535	2	1
R88D-WN⊡- ML2	Pn20E	Electronic gear ratio G1 (numerator)		1 to 1073741824	4	4
	Pn210	Electronic gear ratio G2 (denominator)		1 to 1073741824	4	1
SMART- STEP Junior	Pn20E	Electronic gear ratio G1 (numerator)		1 to 1073741824	4	1
	Pn210	Electronic gear ratio G2 (denominator)		1 to 1073741824	4	1

Command Unit Parameters

The command unit default setting is pulses.

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting
Accelera- tion/ decelera-	Pn811	Linear acceleration con- stant	10,000 command units/s ²	-32768 to 32767	2	100
stants	Pn814	Linear deceleration con- stant	10,000 command units/s ²	-32768 to 32767	2	100
Accelera- tion/ decelera- tion filters	Pn818	Position command FIR fil- ter time constant	0.1 ms	0 to 10000	2	0

G5-series Acceleration/Deceleration Constants

G-series Acceleration/Deceleration Constants

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting
Accelera- tion/ decelera-	Pn107	Linear acceleration con- stant	10,000 command units/s ²	-32768 to 32767	2	100
stants	Pn10A	Linear deceleration con- stant	10,000 command units/s ²	-32768 to 32767	2	100
Accelera- tion/ decelera- tion filters	Pn10E	Moving average time	0.1 ms	0 to 5100	2	0

W-series and SMARTSTEP Junior Acceleration/Deceleration Constants

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting	W Series	SMART- STEP Junior
Accelera- tion/ decelera-	ccelera- Pn80A First-step linear accelera- n/ tion constant ecelera-		10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Not sup- ported
stants	Pn80B	Second-step linear acceler- ation constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Sup- ported
	Pn80C	Acceleration constant switching speed	100 com- mand units/s	0 to 65535	2	0	Sup- ported	Not sup- ported
	Pn80D	First-step linear decelera- tion constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Not sup- ported
	Pn80E	Second-step linear deceler- ation constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Sup- ported
	Pn80F	Deceleration constant switching speed	100 com- mand units/s	0 to 65535	2	0	Sup- ported	Not sup- ported
Accelera- tion/	Pn810	Exponential acceleration/ deceleration bias	Command units/s	0 to 32767	2	0	Sup- ported	Not sup- ported
decelera- tion filters	Pn811	Exponential acceleration/ deceleration time constant	0.1 ms	0 to 5100	2	0	Sup- ported	Not sup- ported
	Pn812	Movement average time	0.1 ms	0 to 5100	2	0	Sup- ported	Not sup- ported

When using W-series Servo Drives, the first-step acceleration/deceleration curve operations require parameter settings for Pn80B and Pn80E only. When using an exponential curve for the acceleration/deceleration curve, Pn810 and Pn811 must be set for W-series Servo Drives. *Movement Average Time* for W-series Servo Drives (Pn812) or *Moving Average Time* for G-series Servo Drives (Pn10E) must be set when using an S-curve for the acceleration/de

The second-step acceleration/deceleration constant cannot be set using a G5-series Servo Drive or G-series Servo Drive. The exponential acceleration/ deceleration filter cannot be set because there is no exponential acceleration/ deceleration filter function.

The SMARTSTEP Junior Servo Drives do not support the following parameters: Pn80A, Pn80C, Pn80D, and Pn80F. They also do not support acceleration/deceleration filters, so the filters cannot be set. One-step linear acceleration/deceleration curves are set using only Pn80B and Pn80E.

For details on acceleration/deceleration curves, refer to 7-4 Acceleration and Deceleration Operations.

Туре	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
	Pn820 (See note.)	Final distance for external input positioning	Command unit	-1,073,741,823 to 1,073,741,823	4	100
	Pn822	Origin return mode setting		0000 hex or 0001 hex	2	0
	Pn823	Origin return approach speed 1	100 command units/s	1 to 32767	2	50
	Pn824	Origin return approach speed 2	100 command units/s	1 to 32767	2	5
	Pn825	Final distance for Origin Return	Command unit	-1,073,741,823 to 1,073,741,823	1	100

G5-series Origin Search Parameters

G-series Origin Search Parameters

Туре	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
	Pn203 (See note.)	Final distance for external input positioning	Command unit	-1,073,741,823 to 1,073,741,823	4	100
	Pn10F	Origin return mode settings		0000 hex or 0001 hex	2	0
	Pn110	Origin return approach speed 1	100 command units/s	1 to 32767	2	50
	Pn111	Origin return approach speed 2	100 command units/s	1 to 32767	2	5
	Pn204	Origin return final distance	Command unit	-1,073,741,823 to 1,073,741,823	1	100

Туре	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
	Pn814 (See note.)	Final travel distance for external positioning	Command unit	-1,073,741,823 to 1,073,741,823	4	100
	Pn816	Zero point return direction		0000 hex or 0001 hex	2	0
	Pn817	Zero point return approach speed 1	100 command units/s	0 to 65535	2	50
	Pn818	Zero point return approach speed 2	100 command units/s	0 to 65535	2	5
	Pn819	Final travel distance to return to zero point	Command unit	-1,073,741,823 to 1,073,741,823	4	100

W-series Origin Search Parameters

Note

Starting in Position Control Unit with version 2.0 and later, the *Final Distance for External Input Positioning* (G5 series: Pn820, G series: Pn203) or *Final Travel Distance for External Positioning* (W series: Pn814) is used for internal processing in the Position Control Unit when one of the following origin search operation patterns are selected.

- When the origin detection method is set to any value except for 0 (Origin proximity input signal reversal)
- When setting a zero point position offset for an origin search using an absolute encoder

Because the *Final Distance for External Input Positioning* (G5 series: Pn820, G series: Pn203) or *Final Travel Distance for External Positioning* (W series: Pn814) is automatically overwritten by the Position Control Unit when starting the origin search operation, if the interrupt feeding is used after origin search, the *Final Distance for External Input Positioning* (G5 series: Pn820, G series: Pn203) or *Final Travel Distance for External Positioning* (W series: Pn814) must be set again. Also, the Servo parameters are read out by the Position Control Unit for the following parameters when the origin search operation is started.

Origin Return Approach Speed 1, Origin Return Approach Speed 2, Origin Search Final Travel Distance

(Origin Search Approach Speed 1, Origin Search Approach Speed 2, Origin Search Final Travel Distance (Pn823, Pn824, Pn825) for G5 Series, Origin Return Approach Speed 1, Origin Return Approach Speed 2, Origin Return Final Distance (Pn110, Pn111, Pn204) for G Series or Zero Point Return Approach Speed 1, Zero Point Return Approach Speed 2, and Final Travel Distance to Return to Zero Point (Pn817, Pn818, Pn819) for W Series)

Absolute Origin Offset (Pn200) for G Series or Absolute Encoder Zero Point Position Offset (Pn808) for G5 Series/W Series) (when using an absolute encoder)

Set the origin search direction of the axis parameters and Origin Search Direction for G5-series Servo Drives (Pn822), Origin Return Mode Settings for G-series Servo Drives (Pn10F) or Zero Point Return Mode Setting for W-series Servo Drives (Pn816) to the same direction. Setting different directions may result in a malfunction.

Туре	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
R88D-KN□-	Pn431	Positioning completion range 1	Command unit	0 to 262144	4	300
ML2	Pn803	Origin range	Command unit	0 to 250	2	0
R88D-GN -	Pn060	Positioning completion range 1	Command unit	0 to 10000	2	25
ML2	Pn105	Origin range	Command unit	0 to 250	2	10
R88D-WT	Pn500	Positioning completion range 1	Command unit	0 to 250	2	3
	Pn803	Zero point width	Command unit	0 to 250	2	10
R88D-WN⊡- ML2	Pn522	Positioning completion range 1	Command unit	0 to 1073741824	4	3
Pn803		Zero point width	Command unit	0 to 250	2	10
SMART- STEP Junior	Pn522	Positioning completion range 1	Command unit	0 to 1073741824	4	10
	Pn523	Zero point width	Command unit	0 to 250	2	10

Control Status Parameters

Axis Operating Output Memory Areas (Operating Commands)

٩	lame	Word	Bits	Contents	G Series	W Series	SMARTSTE P Junior
ORIGIN SE	EARCH	а	06	$0 \rightarrow 1$: Starts origin search	Supported	Supported	Supported
Position co	mmand value	a+2 a+3		Position command value (right- most word) Position command value (left-	Supported	Not sup- ported	Supported
				most word)			
				Command range: – 2,147,483,648 to 2,147,483,647 (80000000 hex to 7FFFFFF hex)			
				This value is also used as the preset value when a preset value is being set after completing an origin search.			
Speed corr	imand value	a+4 a+5		Speed command value (right- most word)	Supported	Supported	Supported
				Speed command value (leftmost word)			
				Unit: Command units/s			
				Command range: 1 to 2,147,483,647 (00000001 hex to 7FFFFFF hex) This value is also used as the preset value when a preset value is being set after completing an origin search.			
Accelera- tion/ decelera-	Exponential curve desig- nation	a+16	03	1: Use exponential acceleration/ deceleration curve.	Not sup- ported	Supported	Not sup- ported
tion curve designa- tion	S-curve des- ignation		04	1: Use S-curve acceleration/ deceleration curve.	Supported	Supported	Not sup- ported
Forward rotation current limit			14	1: Use forward torque limit	Supported	Supported	Not sup- ported
Reverse rotation current limit			15	1: Use reverse torque limit	Supported	Supported	Not sup- ported

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

The speed command value specifies the initial origin search speed until the origin proximity signal is input. When an origin search is started with the speed command set to 0 (command units/s), a Speed Designation Error (axis error code: 3061) will occur and the origin search will not be executed.

Torque limit during operation is possible with origin searches. For further details on the torque limit function, refer to *10-4 Torque Limits*.

The speed command value, acceleration/deceleration curve designation, and forward/reverse rotation current limit designation data are enabled when the ORIGIN SEARCH Bit turns ON.

The SMARTSTEP Junior Servo Drives do not support acceleration/deceleration filters and torque limits. When using a SMARTSTEP Junior Servo Drive, do not attempt to use an acceleration/deceleration curve designation or forward/reverse torque limit designation.

Note Do not set both the exponential curve designation and S-curve designation to 1 (enabled) in the acceleration/deceleration curve designation. Enabling both settings may cause a malfunction.

Name	Word	Bits	Contents
PCU Positioning Com- pleted Flag	b	05	$0 \rightarrow 1$: Positioning completed.
No Origin Flag		06	0: Origin established.
			1: No origin established.
Origin Stop Flag		07	0: Outside origin range.
			1: Within origin range.
Error Flag		12	0: No axis error.
			1: Axis error.
Busy Flag		13	1: Axis busy (axis operation executing)
Feedback present posi- tion	b+6 b+7		Present position: Feedback position (rightmost word) Feedback position (leftmost word)
Command present posi- tion	b+8 b+9		Present position: Command position (rightmost word) Command position (leftmost word)

Axis Operating Input Memory Areas (Monitor)

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

8-2-4 Origin Search Operation

Origin Search Operation Summary Basically, the PCU's origin search operation consists of the following three steps.

- 1,2,3... 1. Search for origin proximity signal
 - 2. Search for origin signal after detection of origin proximity signal
 - 3. Positioning for final travel distance to return to zero point after detection of origin signal

Each of these steps is executed using the operation direction and speed according to the Axis Parameters, Servo Parameters, and speed command value for origin search as shown in the following diagram.

The origin proximity signal is not always used in the origin search operation pattern and an origin search can be started at the zero point return approach speed. For details on origin search operation patterns, refer to *Origin Search Operation Modes* on page 306.



Origin search direction (See note.)

(Axis Parameters)

- (1) The operation direction for origin search is determined by the setting in Servo Parameter *Zero Point Return Direction* (Pn816).
- (2) The parameters shown above are different for G-series Servo Drives.
- **Note** Make sure that the same direction is set for *Origin Search Direction* in the Axis Parameters and *Zero Point Return Direction* (Pn816) in the Servo Parameters. Setting different directions may result in a malfunction.

The override function is disabled during origin searches.

The parameter *Final Travel Distance to Return to Zero Point* (Pn819) sets the amount of compensation positioning used after the origin input signal has been detected. The value is set as the relative travel distance in the origin search direction.

When *Final Travel Distance to Return to Zero Point* (Pn819) is set to a negative or small value, the axis will decelerate to a stop, and then reverse and perform the set positioning.



Setting	Origin detection method	Origin input signal detection	Remarks
0	With origin proximity input sig- nal reversal	The origin input signal is detected after the origin prox- imity input signal turns ON and then OFF.	

Setting	Origin detection method	Origin input signal detection	Remarks
1	Without origin proximity input signal reversal	The origin input signal is detected after the origin prox- imity input signal turns ON.	This setting is possible only for Position Control Units with unit version 2.0 or later.
2	Not use origin proximity input signal	The origin proximity input sig- nal is not used.	The origin search operation is started at the zero point return approach speed 1.
			If this setting is used along with reversal mode 3, and axis parameter check error (error code: 3091) will occur when the data is transferred.
			This setting is possible only for Position Control Units with unit version 2.0 or later.

- Note (1) The Servo Drive's origin proximity input signal (origin return deceleration limit switch signal) must be allocated to execute ORIGIN SEARCH. The forward/reverse rotation limit signal (forward/reverse drive prohibited signal) must be allocated to use a reversal mode. If these input signals are not allocated, ORIGIN SEARCH cannot be executed properly. For details on allocating Servo Drive input signals, refer to *6-4 Standard Settings for Servo Drives Using MECHATROLINK*.
 - (2) Make sure that the external input signals for origin searches (limit input signal and origin proximity input signal) maintain a signal width no shorter than the MECHATROLINK communications cycle. If the input signal width is shorter than the communications cycle, the PCU is unable to detect the input signals, and operations cannot be performed normally.
 - (3) For origin searches, the origin signal is detected after the origin proximity input signal (origin return deceleration limit switch) changes from ON to OFF. Therefore, be sure to use a sensor for the origin proximity input signal, such as a photoelectric switch, that does not produce chattering. Chattering from contact switches may result in the displacement of the origin position.

Origin Search Operation



Note To use a motor with absolute encoders with the OMNUC G series and G5 series Servo Drives, use the motor as described below when setting the absolute PG origin offset using origin search.

- If the limit input in the origin search direction is ON when executing the origin search, origin search does not operate normally. Start the origin search after moving the position where the limit input in the origin search direction does not turn ON by jogging.
- If the origin search is started when the limit input is ON in the origin search direction, an error terminates when the origin search operation is completed (data setting error, error code: 3099).

Origin search operation mode	1: Reversal mode 2
Origin detection method	
0: With origin proximity input signal reversal	Origin proximity input signal 0 Origin input signal 1 (Phase Z signal) 0 Speed command value Zero point return apprbach speed 1 Final travel distance to return to zero point Origin search reverse direction Start Stop
	Origin search reverse direction Origin search reverse direction Origin search reverse direction Origin search reverse direction Start Stop Origin search direction Start Error stop Origin search direction

Origin Search Operation

Section 8-2



Origin search operation mode	2: Single-direction mode
Origin detection method	
0: With origin proximity input signal reversal	Origin proximity 1 input signal 0 Origin input signal 1 (Phase Z signal) 0
	Origin search reverse direction Start Corigin search reverse direction Start S
	Origin search reverse direction Start Error stop
A Milah and a single and in the	Origin search direction reverse direction Start Limit input signal in origin search direction
input signal reversal (unit ver- sion 2.0 or later)	Origin proximity 1 input signal 0 Origin input signal 1 D
	(Phase Z signal) 0 Zero point return approach speed 1 Speed command value Final travel distance to return to zero point
	Origin search reverse direction Start Speed command value Stop
	Origin search reverse direction Start Speed command value
	Origin search reverse direction Start Limit input signal in origin search direction
2: Not use origin proximity input signal (unit version 2.0	Origin input signal 1 (Phase Z signal) 0
Unater)	Zero point return approach speed 1 Origin search reverse direction Start Start Start Final travel distance to return to zero point - Zero point return approach speed 2 Origin search direction Error stop
	Origin search reverse direction Start Final travel distance to return to zero point Error stop
	Origin search reverse direction Start Limit input signal in origin search direction
	Origin input signal 1 (Phase Z signal) 0 Zero point return approach speed 1 Final travel distance to return to zero point - Zero point return approach speed 2
	Origin search reverse direction Start Stop



Note (1) The reversal operation for an origin search in the reverse direction is executed either when the origin proximity input signal turns OFF during the origin search in the reverse direction, or when the limit input signal turns ON during the origin search in the reverse direction while the origin proximity input signal is ON, whichever occurs first.



(2) The signal width of the origin input signal must be longer than the MECHATROLINK communications cycle when the Origin Proximity Input Signal parameter is set to 2 (Not use origin proximity input signal) during

the origin search for the Reversal Mode 1 and Reversal Mode 2 origin search operation modes. If the origin input signal length is shorter than the communications cycle, the PCU will be unable to detect the origin input signal when in reversal operation. If the established origin position is not aligned, the origin search function will not operate correctly. If using this function with this setting, use an external sensor for the origin input signal or select an origin search method that uses the origin proximity input signal because the signal width of the Servo Drive's Z-phase signal is too short.

Reversal Mode Operation Using Limit Inputs When the origin search operation mode is set to reversal mode 1 to 3, axis operation stops or changes to the opposite direction when the limit input signal turns ON.

When using a W-series Servo Drive, the axis will operate according to the *Select Stop when Prohibited Drive Is Input* (Pn001.1) setting in the Servo Parameters after the limit input signal turns ON and until the axis operation stops.

When using a SMARTSTEP Junior Servo Drive, the Servomotor will decelerate to a stop using the maximum Servomotor torque.

Reverse Pattern

For reversal mode 1, the positioning relationship between the limit input and the origin proximity signal is as shown by the two patterns in the following diagram. For reversal mode 2 or 3, only the bottom pattern is possible. When these modes are used, the axis operates according to the *Stop Selection for Drive Prohibit Input* for G-series Servo Drives (Pn066) or *Select Stop when Prohibited Drive Is Input* for W-series Servo Drives (Pn001.1) for the shaded areas in the diagram.



Note If the Servo Parameter (Pn505 for G5 Series, Pn066 for G Series and Pn001.1 for W Series) is set to stop operation with a free run when a limit input is input when using a G-series or a W-series Servo Drive, the momentum of the previous operation may cause the machine to run in the limit input direction.

When origin searches are performed using reversal mode 1 to 3 as the operation mode, either enable the Servo Drive's dynamic brake, or set the *Stop Selection for Drive Prohibition Input* for G5-series Servo Drives (Pn505), the *Stop Selection for Drive Prohibit Input* for G-series Servo Drives (Pn066) or *Select Stop when Prohibited Drive is Input* for W-series Servo Drives (Pn001.1) to 1 (decelerates to a stop using the immediate stop torque set in Pn511 for G5 Series, the emergency stop torque set in Pn06E for G Series or Pn406 for W Series or lower and locks Servo in zero clamp mode) or 2 (decelerates to a stop using the immediate stop torque set in Pn511 for G5 Series, the emergency stop torque set in Pn06E for G Series or Pn406 for W Series or lower and locks Servo in zero clamp mode) or 2 (decelerates to a stop using the immediate stop torque set in Pn511 for G5 Series, the emergency stop torque set in Pn06E for G Series or Pn406 for W Series or lower and puts Servo in free run state) so that the Servomotor stops properly when a limit input is received.

8-2-5 Starting Origin Search

Origin search is performed according to the speed command value set in the Axis Operating Output Memory Areas and started when the ORIGIN SEARCH Bit turns ON. Execute ORIGIN SEARCH when the Busy Flag for the corresponding axis is OFF. If ORIGIN SEARCH is executed while the axis's Busy Flag is ON, a Multistart Error (axis error code: 3050) will occur and ORI-GIN SEARCH will not be executed.

When starting ORIGIN SEARCH, make sure that the ORIGIN SEARCH Bit will remain ON until the Busy Flag turns ON.

If the SERVO UNLOCK, DEVIATION COUNTER RESET, EMERGENCY STOP, OR DECELERATION STOP command bit is turned ON, the ORIGIN SEARCH command will be ignored. Be sure that all of these command bits are OFF before executing an origin search.

<u>**Timing Chart</u>** The following timing charts are for when ORIGIN SEARCH is executed with the origin search operation set to reversal mode 1, the origin detection mode set to origin proximity input signal reversal, and the origin search direction set to forward.</u>

Timing Chart Operation



Origin Search Operation



Timing Chart

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) \times 25 b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. –1) \times 25

Note If the origin search preset is used, the PCU Positioning Completed Flag, No Origin Flag, and Busy Flag will change status only after the present position preset operation is completed after finishing the origin search operation.

Transferring Parameters during Origin Search Operations

The Position Control Unit can transfer Servo Parameters during axis operations (i.e., while the Busy Flag is ON). With Position Control Units with unit version 2.0 or later, Servo Parameter transfers specified during origin search operation are performed after completing the origin search.

Example: The following example illustrates reading Servo Parameters during an origin search operation.



8-2-6 Origin Search Preset

The present position can be set to any desired position after completing an origin search. This enables setting an arbitrary electrical origin point corresponding to the mechanical origin point.

This function can be used only with Position Control Units with unit version 2.0 or later.

The origin search preset operation is selected using the following Origin Search Preset parameter in the Axis Parameters.

Origin Search Operation

Section 8-2

Axis Parameters

The following parameters are used to set the origin search operation mode.

PCU's			Cont	Setting			
address	15 to 14	13	12	11 to 08	07 to 04	03 to 00	
d+1	0 (fixed)	Origin	Origin	Origin	Origin	0 (fixed)	Bit 13: Origin Search Preset
		search preset	search direction	detection method	search operation		 0: Do not set the present position to the preset position after complet- ing the origin search (default set- ting). 1: Set the present position to the pre- set position after completing the origin search.

 $d = 1860 \text{ hex} + (\text{Axis No.} -1) \times 14 \text{ hex}$

The position command value set in the Axis Operating Output Memory Areas is used as the preset value to be set after completing an origin search.

Axis Operating Output Memory Areas (Operating Commands)

Name	Word	Bits	Contents
Position command value	a+2 a+3		Position command value (rightmost word) Position command value (leftmost word)
			Unit: Command units/s
			Command range: -2,147,483,648 to
			2,147,483,647 (80000000 hex to 7FFFFFF hex)
			This value is also used as the preset value when a preset value is being set after completing an origin search.

a = Beginning word of Axis Operating Output Areas specified in Common

Parameters + (Axis No. -1) × 25

The position command value that is set when the ORIGIN SEARCH command bit turns ON is used as the origin search preset.

The origin search preset operation is executed only after the origin search operation has been completed normally, i.e., when positioning for the final travel distance to return to zero point has entered the positioning completion range after the origin signal has been detected.

If an absolute encoder is used, the origin search operation is set to reversal mode 1, and the origin search preset is enabled, a zero point position offset for the absolute encoder can be automatically set. Refer to *8-6-5 Absolute Encoder's Origin (Zero Point) Position Offset Setting* for details.

Note The origin search preset function of the Position Control Unit cannot be used with the function block for origin searches in the OMRON FB Library (_NCF05@_Home) if the Library is version is 1.13 or lower. Use the present value preset function provided in the library. To use the origin search function block and enable the origin search preset, the OMRON FB Library with a version higher than 1.13 must be used.

8-3 Present Position Preset

PRESENT POSITION PRESET changes the present position to any position and then establishes the origin.

8-3-1 Outline of Operation

When the PRESENT POSITION PRESET Bit is turned ON, the present position is changed to the value set as the position command value in the Axis Operating Output Memory Areas. during present position preset, the Busy Flag is ON (1 cycle time min.). The Busy Flag turns OFF to indicate that preset operation has been completed. After this, the origin is established. If the present position is changed to "0," then that position will become the origin.

8-3-2 Data Settings for Present Position Preset

A simple explanation of the main parameters and data used to execute PRESENT POSITION PRESET is provided here. For details on command units, refer to *7-2 Control Units*.

The setting units for parameters and data depend on the specified command unit.

Model	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
R88D-KN⊡- ML2	Pn009	Electronic gear ratio numerator		1 to 1073741824	4	1
	Pn010	Electronic gear ratio denominator		1 to 1073741824	4	1
R88D-GN⊡- ML2	Pn205	Electronic gear ratio G1 (numerator)		1 to 1073741824	4	1
	Pn206	Electronic gear ratio G2 (denominator)		1 to 1073741824	4	1
R88D-WT	Pn202	Electronic gear ratio G1 (numerator)		1 to 65535	2	4
	Pn203	Electronic gear ratio G2 (denominator)		1 to 65535	2	1
R88D-WN⊡- ML2	Pn20E	Electronic gear ratio G1 (numerator)		1 to 1073741824	4	4
	Pn210	Electronic gear ratio G2 (denominator)		1 to 1073741824	4	1
SMART- STEP Junior	Pn20E	Electronic gear ratio G1 (numerator)		1 to 1073741824	4	1
	Pn210	Electronic gear ratio G2 (denominator)		1 to 1073741824	4	1

Servo Parameter Area Command Unit Parameters

The command unit default setting is pulses.

Axis Operating Output Memory Areas (Operating Commands)

Name	Word	Bits	Contents
PRESENT POSITION PRESET	а	08	$0 \rightarrow 1$: Starts present position preset
Position command value	a+2 a+3		Position command value (rightmost word)
			Position command value (leftmost word)
			Unit: Command unit
			Command range: -2,147,483,648 to 2,147,483,647 (80000000 hex to 7FFFFFF hex)

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

Present position preset is performed according to the position command value set in the Axis Operating Output Memory Area and started when the PRESENT POSITION PRESET Bit turns ON.

Execute PRESENT POSITION PRESET when the Busy Flag for the corresponding axis is OFF. If PRESENT POSITION PRESET is executed while the axis's Busy Flag is ON, a Multistart Error (axis error code: 3050) will occur and PRESENT POSITION PRESET will not be executed. When executing PRESENT POSITION PRESET, make sure that the PRESENT POSITION PRESET Bit will remain ON until the Busy Flag turns ON.

Axis Operating Input Memory Areas (Monitor)

Name	Word	Bits	Contents
No Origin Flag	b	06	0: Origin established.
			1: No origin established.
Origin Stop Flag		07	0: Outside origin range.
			1: Within origin range.
Error Flag		12	0: No axis error.
			1: Axis error.
Busy Flag		13	1: Axis busy (axis operation executing).
Feedback present posi- tion	b+6 b+7		Present position: Feedback present position (rightmost word) Feedback present position (leftmost word)
Command present posi- tion	b+8 b+9		Present position: Command position (rightmost word) Command position (leftmost word)

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) \times 25

Timing Chart

In the following timing chart, the present position is changed to "0." When the present position changes to "0," that position becomes the origin, so the Origin Stop Flag turns ON. The status of the PCU Positioning Completed Flag (word b, bit 05) does not change.



a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) × 25

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) \times 25

8-4 Origin Return

The origin return operation is used to return the axis to the origin from any position. ORIGIN RETURN is executed when the ORIGIN RETURN Bit turns ON.

Note Execute ORIGIN RETURN when the origin has been established. If ORIGIN RETURN is executed when the origin has not been established, a Present Position Unknown Error (axis error code: 3030) will occur.

8-4-1 Origin Return Data Settings

A simple explanation of the main parameters and data used to execute ORI-GIN RETURN is provided here. To execute ORIGIN RETURN, apart from the parameters explained here, the following parameters also need to be set as basic settings for operating the PCU.

For details on command units, refer to 7-2 Control Units. The setting units for parameters and data depend on the specified command unit.

Model	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
R88D-KN⊡- ML2	Pn009	Electronic gear ratio numerator		1 to 1073741824	4	1
	Pn010	Electronic gear ratio denominator		1 to 1073741824	4	1
R88D-GN⊡- ML2	Pn205	Electronic gear ratio G1 (numerator)		1 to 1073741824	4	1
	Pn206	Electronic gear ratio G2 (denominator)		1 to 1073741824	4	1
R88D-WT	Pn202	Electronic gear ratio G1 (numerator)		1 to 65535	2	4
	Pn203	Electronic gear ratio G2 (denominator)		1 to 65535	2	1
R88D-WN⊡- ML2	Pn20E	Electronic gear ratio G1 (numerator)		1 to 1073741824	4	4
	Pn210	Electronic gear ratio G2 (denominator)		1 to 1073741824	4	1
SMART- STEP Junior	Pn20E	Electronic gear ratio G1 (numerator)		1 to 1073741824	4	1
	Pn210	Electronic gear ratio G2 (denominator)		1 to 1073741824	4	1

Servo Parameter Area

Command Unit Parameters

The command unit default setting is pulses.

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting
Accel- eration/ decel-	Pn811	Linear acceleration constant	10,000 command units/s ²	-32768 to 32767	2	100
eration con- stants	Pn814	Linear deceleration constant	10,000 command units/s ²	-32768 to 32767	2	100
Accel- eration/ decel- eration filters	Pn818	Position command FIR filter time constant	0.1 ms	0 to 10000	2	0

G5-series Acceleration/Deceleration Constants

G-series Acceleration/Deceleration Constants

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting
Accel- eration/ decel-	Pn107	Linear acceleration constant	10,000 command units/s ²	-32768 to 32767	2	100
eration con- stants	Pn10A	Linear deceleration constant	10,000 command units/s ²	-32768 to 32767	2	100
Accel- eration/ decel- eration filters	Pn10E	Moving average time	0.1 ms	0 to 5100	2	0

W-series and SMARTSTEP Junior Acceleration/Deceleration Constants

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting	W Series	SMART- STEP Junior
Accel- eration/ decel-	Pn80A	First-step linear acceleration constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Not sup- ported
con- stants	Pn80B	Second-step linear accelera- tion constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Sup- ported
	Pn80C	Acceleration constant switch- ing speed	100 com- mand units/s	0 to 65535	2	0	Sup- ported	Not sup- ported
	Pn80D	First-step linear deceleration constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Not sup- ported
	Pn80E	Second-step linear decelera- tion constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Sup- ported
	Pn80F	Deceleration constant switch- ing speed	100 com- mand units/s	0 to 65535	2	0	Sup- ported	Not sup- ported

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting	W Series	SMART- STEP Junior
Accel- eration/	Pn810	Exponential acceleration/ deceleration bias	Command units/s	0 to 32767	2	0	Sup- ported	Not sup- ported
decel- eration	Pn811	Exponential acceleration/ deceleration time constant	0.1 ms	0 to 5100	2	0	Sup- ported	Not sup- ported
mers	Pn812	Movement average time	0.1 ms	0 to 5100	2	0	Sup- ported	Not sup- ported

When using W-series Servo Drives, the first-step acceleration/deceleration curve operations require parameter settings for Pn80B and Pn80E only. When using an exponential curve for the acceleration/deceleration curve, Pn810 and Pn811 must be set for W-series Servo Drives. *Movement Average Time* for W-series Servo Drives (Pn812) or *Moving Average Time* for G-series Servo Drives (Pn10E) must be set when using an S-curve for the acceleration/deceleration/deceleration/deceleration/deceleration/deceleration/deceleration.

The second-step acceleration/deceleration constant cannot be set using a G5-series Servo Drive or G-series Servo Drive. The exponential acceleration/ deceleration filter cannot be set because there is no exponential acceleration/ deceleration filter function.

The SMARTSTEP Junior Servo Drives do not support the following parameters: Pn80A, Pn80C, Pn80D, and Pn80F. They also do not support acceleration/deceleration filters, so the filters cannot be set. One-step linear acceleration/deceleration curves are set using only Pn80B and Pn80E.

Refer to *7-4 Acceleration and Deceleration Operations* for details on acceleration/deceleration curves.

Model	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
R88D-KN□-	Pn431	Positioning completion range 1	Command unit	0 to 262144	4	300
ML2	Pn803	Origin range	Command unit	0 to 250	2	0
R88D-GN -	Pn060	Positioning completion range 1	Command unit	0 to 10000	2	25
ML2	Pn105	Origin range Command u		0 to 250	2	10
R88D-WT	Pn500	Positioning completion range 1	Command unit	0 to 250	2	3
	Pn803	Zero point width	Command unit	0 to 250	2	10
R88D-WN⊡- ML2	Pn522	Positioning completion range 1	Command unit	0 to 1073741824	4	3
	Pn803	Zero point width	Command unit	0 to 250	2	10
SMART- STEP Junior	Pn522	Positioning completion range 1	Command unit	0 to 1073741824	4	10
	Pn803	Zero point width	Command unit	0 to 250	2	10

Control Status Parameters

Na	ame	Word	Bits	Contents	G Series	W Series	SMARTSTEP Junior
ORIGIN RETURN Bit		а	07	$0 \rightarrow 1$: Starts origin return operation	Supported	Supported	Supported
Speed com	mand value	a+4 a+5		Speed command value (rightmost word)	Supported	Supported	Supported
				Speed command value (leftmost word)			
				Unit: Command units/s			
				Command range: 1 to 2,147,483,647 (00000000 hex to 7FFFFFF hex) The upper limit setting for the speed command value depends on the Servo Drive specifications.			
Accelera- tion/decel- eration nation		a+16	03	1: Use exponential acceleration/ deceleration curve.	Not sup- ported	Supported	Not sup- ported
curve des- ignation S-curve des- ignation			04	1: Use S-curve acceleration/decel- eration curve.	Supported	Supported	Not sup- ported
Forward rotation current limit			14 1: Use forward torque limit		Supported	Supported	Not sup- ported
Reverse rot limit	ation current		15	1: Use reverse torque limit	Supported	Supported	Not sup- ported

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

Origin return is performed according to the speed command value set in the Axis Operating Output Memory Areas and started when the ORIGIN RETURN Bit turns ON.

Execute ORIGIN RETURN when the Busy Flag for the corresponding axis is OFF. If ORIGIN RETURN is executed while the axis's Busy Flag is ON, a Multistart Error (axis error code: 3050) will occur and ORIGIN RETURN will not be executed. When starting ORIGIN RETURN, make sure that the ORIGIN RETURN Bit will remain ON until the Busy Flag turns ON.

If the SERVO UNLOCK, DEVIATION COUNTER RESET, EMERGENCY STOP, OR DECELERATION STOP command bit is turned ON, the ORIGIN RETURN command will be ignored. Be sure that all of these command bits are OFF before executing an origin return.

The target speed for an origin return operation can be changed while ORIGIN RETURN is being executed by setting the new speed command value in the Axis Operating Output Memory Area. (For details on changing the target speed, which is the same as positioning operations, refer to *9-4-4 Changing Target Speed*.)

Torque limit during operation is possible with origin return operations. For further details on the torque limit function, refer to *10-4 Torque Limits*.

The acceleration/deceleration curve designation, and forward/reverse rotation current limit designation data are enabled when the ORIGIN RETURN Bit turns ON.

The SMARTSTEP Junior Servo Drives do not support acceleration/deceleration filters and torque limits. When using a SMARTSTEP Junior Servo Drive, do not attempt to use an acceleration/deceleration curve designation or forward/reverse torque limit designation. **Note** Do not set both the exponential curve designation and S-curve designation to 1 (enabled) in the acceleration/deceleration curve designation. Enabling both settings may cause a malfunction.

Axis Operating Input Memory Areas (Monitor)

Name	Word	Bits	Contents
PCU Positioning Com- pleted Flag	b	05	$0 \rightarrow 1$: Positioning completed.
No Origin Flag		06	0: Origin established.
			1: No origin established.
Origin Stop Flag		07	0: Outside origin range.
			1: Within origin range.
Error Flag		12	0: No axis error.
			1: Axis error.
Busy Flag		13	1: Axis busy (axis operation executing).
Feedback present posi- tion	b+6 b+7		Present position: Feedback present position (rightmost word) Feedback present position (leftmost word)
Command present posi- tion	b+8 b+9		Present position: Command position (rightmost word) Command position (leftmost word)

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

Timing Chart

The following timing chart is for when ORIGIN RETURN is executed.



b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

8-5 Phase Z Margin

8-5-1 Description

Although there are some variations for different settings, the PCU's origin search basically works by latching (detecting) the first encoder phase Z after confirming ON/OFF status of the origin proximity signal, performing positioning for the final travel distance to return to the origin, and then setting that position as the origin.

When the time (distance) from when the origin proximity signal changes from ON to OFF until the first phase Z is detected is extremely short or is almost the same as the time taken for one rotation of the Servomotor, the position at which the phase Z is detected may be displaced by one phase due to discrepancies in the ON response of the origin proximity sensor or the Servo Drive speed. This displacement is prevented by confirming the "phase Z margin." The phase Z margin indicates the amount by which the Servomotor rotated during the time from when the origin proximity signal changed from ON to OFF until the phase Z was detected.

If this value is close to 0 or close to one motor rotation, there is a possibility of origin displacement during origin search. A simple way of reducing the probability of this is to adjust the Servomotor's mounting angle or the origin proximity sensor's mounting position so that this value is approximately half a rotation of the Servomotor.

Example: Final travel distance to return to zero point = 0



8-5-2 Calculating the Phase Z Margin (Example)

One method used to obtain the phase Z margin is to perform jogging after origin search in the opposite direction to origin search, with the final travel distance to return to origin set to 0. Then stop operation at the point where the origin proximity signal changes from OFF to ON, and read the value of the present position at that point. The value of this present position is equal to the phase Z margin.

If the speed at the point when the origin proximity signal turns ON is fast, there will be some inaccuracy in the value read for the phase Z margin. Reduce the speed of the jogging in the vicinity of the point where the origin proximity signal turns ON.

The required phase Z margin can be obtained by adjusting mounting positions and angles so that this position corresponds to approximately half the distance moved for one rotation of the motor.



8-5-3 Phase Z Margin for Specific Operation Patterns

Phase Z Margin Required for Origin Search Settings Up to a maximum of two MECHATROLINK communications cycles plus 4 ms may be required to start searching for the origin signal after detecting the origin proximity signal for the following origin search operation patterns.

- When the origin detection method is set to any setting other than 0 (Origin proximity input signal reversal)
- When an absolute encoder zero point position offset is set for the origin search

When performing an origin search with one of the above operation patterns, the following margin must be provided in the phase Z signal after the origin proximity input signal turns ON.



Provide a margin in the phase Z signal of at least two MECHATROLINK communications cycles plus 4 ms.

Phase Z Margin Required for External Input Signals

If the limit input signal in opposite direction from origin search direction and the origin proximity signal are close or overlapping, up to a maximum of two MECHATROLINK communications cycles plus 4 ms may be required to start searching for the origin signal after detecting the origin proximity signal.

In this case, the following margin must be provided in the phase Z signal after the limit input signal in opposite direction from origin search direction turns OFF.

Absolute Encoder Origin



8-6 Absolute Encoder Origin

8-6-1 Absolute Encoder Overview

The absolute encoder's battery backup enables absolute value data to be held even if the power to the CPU Unit (PCU) or Servo Drive is turned OFF. Therefore, after turning ON the power to the CPU Unit (PCU), the saved absolute value data can be read, enabling positioning at the present position that existed before the power was turned OFF.

The Servo Parameter *Absolute Origin Offset* for G-series Servo Drives (Pn200) or *Absolute Encoder Zero Point Position Offset* for G5-series/W-series Servo Drives (Pn808) for absolute value data that is read can be used to offset the mechanical origin.

When an absolute encoder is used to determine the origin, it is not necessary to perform origin searches every time the power is turned ON, as is required for devices such as incremental encoders.



The following is an example for using a W-series Servo Drive.

n: Beginning word of Common Operating Memory Area: n = CIO 1500 + (unit number \times 25) a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) \times 25 b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. –1) \times 25

Applicable Servomotors

The following Servomotors with absolute encoders support this function.

OMRON G5-series Servomotors

Model: R88M-K S/K T/K C

OMRON G-series Servomotors Model: R88M-G S/G T

OMRON W-series Servomotors Model: R88M-W S/W T

The SMARTSTEP Junior Servo Drives cannot be used with motors with absolute encoders.

Absolute Encoder Operating Procedure 8-6-2

The procedure for using an absolute encoder is as given below. The following procedure is performed just once when using the absolute encoder for the first time, when replacing the Servomotor, when the absolute encoder's battery expires, or when another reason occurs that results in loss of absolute value data.

- 1,2,3... 1. Set the Common Parameters and save them.
 - 2. Set the Axis Parameters and save them.
 - Turn the PCU OFF and then ON again or restart the Unit. The data for the Common Parameters and Axis Parameters set in steps 1 and 2 above are enabled.

- 4. Start MECHATROLINK communications.
- 5. Set the absolute encoder type to be used in *Operation Switch When Using Absolute Encoder* (Pn002.2 for W Series and Pn00B for G Series). This step is not required if using the default setting.
- Setup the absolute encoder. After setting up the absolute encoder, the power to the Servo Drive must be turned OFF and then ON again.
- 7. Start MECHATROLINK communications. When the connection is established, the absolute value data is read from the absolute encoder.
- 8. Set the mechanical origin.
- 9. Set the origin position offset for the absolute encoder.

After completing this procedure, the mechanical origin will be set whenever the power is turned ON or MECHATROLINK communications are started.

Establishing the Origin with an Absolute Encoder

gin The following differences exist in origin searches when using a motor with an absolute encoder depending on the unit version of the Position Control Unit.

Position Control Units with Unit Version 1.3 or Earlier

Position Control Units with unit version 1.3 or earlier do not support origin search operations for absolute encoders.

When the origin search operation is executed, the present position is cleared to 0 or, if the origin search preset function is enabled, it is set to the present command value after the origin has been established. Any absolute value data previously read by the Position Control Unit is lost. If the absolute value data is later read from the absolute encoder, e.g., when the Servo is unlocked and locked, the coordinate system based on the absolute value data previously saved in the absolute encoder will be used instead of the origin established by the origin search operation.

Position Control Units with Unit Version 2.0 or Later

When the origin search operation is executed, the present position is cleared to 0 or, if the origin search preset function is enabled, it is set to the present command value after the origin has been established. If the origin search operation pattern is set to reversal mode 1, the zero point position offset of the absolute encoder will be set automatically and saved in the Servo Parameters. The mechanical origin and the zero point position offset in the absolute encoder can thus be set simultaneously by using a Position Control Units with unit version 2.0 or later and reversal mode 1 with an absolute encoder.

Refer to *8-6-5 Absolute Encoder's Origin (Zero Point) Position Offset Setting* for details on setting the zero point position offset using an origin search.

Note ORIGIN SEARCH can also be executed when using Servomotors with absolute encoders. By executing ORIGIN SEARCH, however, the present position will be cleared to 0 when the origin is established. Therefore, the absolute data read to the PCU in Servo lock status will be lost unless the zero point position offset is set using the origin search operation for a Position Control Units with unit version 2.0 or later. The absolute data saved in the absolute encoder will not be lost and can be obtained by executing SERVO UNLOCK after the origin search has been completed, and then executing SERVO LOCK again. The origin position detected, however, will be displaced due to SERVO UNLOCK execution. Therefore, do not execute ORIGIN SEARCH when using a Servomotor with absolute encoder unless you set the zero point position offset using the origin search operation for a Position Control Units with unit version 2.0 or later.

8-6-3 PCU Data Settings for Using Absolute Encoders

The parameters and data required when using absolute encoders are as follows:

Axis Parameter Area

The following parameters are for the operation mode for origin searches.

PCU's		Con	Setting		
address	15 to 12	11 to 08	07 to 04	03 to 00	
d+2	0 (fixed)	0 (fixed)	0 (fixed)	Encoder type	0: Incremental encoder (default setting)
					1: Absolute encoder

 $d = 1860 \text{ hex} + (\text{Axis No.} -1) \times 14 \text{ hex}$

When using an absolute encoder, make sure that the settings for the encoder type in the Axis Parameters and *Operation Switch When Using Absolute Encoder* in the Servo Parameters match.

Axis Parameter Area	G5-series Servo Parameter Pn015 (Operation switch when using absolute encoder)			
Encoder type	G-series Servo Parameter Pn00B (Operation switch when using absolute encoder)			
	0: Use as an absolute encoder.	1: Use as an incremental encoder.		
	2: Use as an absolute encoder but ignore abso- lute multi-turn counter over- flow alarm.			
0: Incremental encoder	Do not set this combination.	This setting is used when using an absolute encoder as an incremental encoder		
1: Absolute encoder	This setting is used when using an absolute encoder as an absolute encoder. (Set this combination when using an absolute encoder.)	Do not set this combination.		

Axis Parameter Area	W-series Servo Parameter Pn002.2 (Operation switch when using absolute encoder)			
Encoder type	0: Use as absolute encoder.	1: Use as incremental encoder.		
0: Incremental encoder	Do not set this combination.	This setting is used when using an absolute encoder as an incremental encoder		
1: Absolute encoder	This setting is used when using an absolute encoder as an absolute encoder. (Set this combination when using an absolute encoder.)	Do not set this combination.		

If the settings do not match, reading absolute data will not be possible, or another malfunction may occur.

Servo Parameter Area

Parameters for Using G5-series Absolute Encoders

Туре	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
	Pn015	Operation switch when using absolute encoder		0 to 2	2	1

		5				
Туре	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
	Pn00B	Operation switch when using absolute encoder		0 to 2	2	0

Parameters for Using G-series Absolute Encoders

Parameters for Using W-series Absolute Encoders

Туре	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
	Pn002.2	Function selection application switch 2 (operation switch when using absolute encoder)		0, 1	2	0

When using a Servomotor with an absolute encoder, select whether the absolute encoder will be used as an absolute encoder or as an incremental encoder. Set this parameter to 0 if using the absolute encoder as an absolute encoder.

When using W-series Servo Drives, the parameter size is the value to be specified when transferring the Servo Parameter including Pn002.

G-series Origin Position Offset Parameters

Туре	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
	Pn200	Absolute origin offset	Command unit	-1,073,741,823 to 1,073,741,823	4	0

G5-series/W-series Origin Position Offset Parameters

Туре	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
	Pn808	Absolute encoder zero point position offset	Command unit	-1,073,741,823 to 1,073,741,823	4	0

This parameter sets the offset for the mechanical origin from the absolute encoder's absolute value data. The present position will be the absolute value data that was read added to the value in this parameter, and that position is set as the origin.

Note When using the zero point position offset setting function with Position Control Units with unit version 2.0 or later, Pn808 is automatically overwritten when an origin search is performed with an absolute encoder. (Refer to *8-6-5 Absolute Encoder's Origin (Zero Point) Position Offset Setting*) To achieve this, the Position Control Unit reads and writes Servo Parameter for Pn808 when the origin search operation is started and when it is completed.

Axis Operating Input Memory Areas (Monitor)

Name	Word	Bits	Contents
No Origin Flag	b	06	0: Origin established.
			1: No origin established.
Origin Stop Flag		07	0: Outside origin range.
			1: Within origin range.
Error Flag		12	0: No axis error.
			1: Axis error.
Busy Flag		13	1: Axis busy (axis operation executing).

Name	Word	Bits	Contents
Feedback present posi- tion	b+6 b+7		Present position: Feedback present position (rightmost word) Feedback present position (leftmost word)
Command present posi- tion	b+8 b+9		Present position: Command position (rightmost word) Command position (leftmost word)

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. –1) \times 25

8-6-4 Absolute Encoder Setup

The absolute encoder setup must be performed when using the absolute encoder for the first time, initializing the rotation amount to 0, or having left the absolute encoder for a long time without connecting a battery.

The following methods can be used to set up the absolute encoder when using a Position Control Unit with unit version 1.2 or later.

- The absolute encoder can be set up using the CX-Motion-NCF.
- The absolute encoder can be set up from the user program using a function block from the OMRON FB Library.
- The absolute encoder can be set up using the Servo Drive's Setting Tool.

Refer to the *SYSMAC CX-Motion-NCF Programmable Controller Operation Manual* (Cat. No. W436) for information on setting up the absolute encoder from the CX-Motion-NCF.

When setting up an absolute encoder using the function block from the OMRON FB Library, use a CPU Unit with unit version 3.0 or later.

Absolute Encoder Setup Function Block in OMRON FB Library: _NCF700_InitializeAbsEncoder



The absolute encoder setup can be executed by specifying the unit number and axis number for the function block, and setting the Start Bit.

For information on function block operating requirements and precautions, refer to the *OMRON FB Library Reference Manual* (Cat. No. W442).

When using a PCU with unit version 1.1 or earlier, absolute encoder setup is executed using the Servo Drive's Setting Tool. For details on absolute encoder setup operations using the Servo Drive Setting Tool, refer to the operation manual for the connected Servo Drive.

Note After executing absolute encoder setup, be sure to turn ON the Servo Drive's control power supply again. Not doing so will prevent the Servo Drive's response to PCU commands, and normal operation will not be possible.

8-6-5 Absolute Encoder's Origin (Zero Point) Position Offset Setting

When Absolute Origin Offset for G Series (Pn200) or Absolute Encoder Zero Point Position Offset for G5 Series/W Series (Pn808) = 0 (Default)

- *1,2,3...* 1. Start MECHATROLINK communications (establish a connection using CONNECT).
 - 2. Use jogging or other operation to place the machine in the mechanical origin position. ORIGIN SEARCH cannot be used when using an absolute encoder.
 - 3. Confirm the feedback present position in the Servo unlocked state and set the origin position offset based on this present position. When the Servo is locked, the present position may oscillate, or deviation will accumulate in the Servo Drive, which may prevent the correct set value from being obtained for the origin position offset, causing displacement of the origin.
 - 4. Reverse the sign of the value for the feedback present position read in step 3 and write this value in *Absolute Encoder Zero Point Position Offset* (Pn808). Use SAVE SERVO PARAMETER to write the value to the Servo Drive's non-volatile memory.
 - 5. Execute DEVICE SETUP or turn the power to the Servo Drive OFF and then ON again.

The absolute encoder origin position offset that has been set is now enabled.

When Absolute Origin Offset for G Series (Pn200) or Absolute Encoder Zero Point Position Offset for G5 Series/W Series (Pn808) = Not 0

When the *Absolute Origin Offset* for G Series (Pn200) or *Absolute Encoder Zero Point Position Offset* for G5 Series/W Series (Pn808) is set to a value other than 0, such as when replacing the Servomotor, use either of the following methods to set the origin position offset.

- Method 1: First set the origin position offset to 0 (set *Absolute Origin Offset* for G Series (Pn200) or *Absolute Encoder Zero Point Position Offset* for G5 Series/W Series (Pn808) to 0) using SAVE SERVO PARAM-ETER), and then set it according to the setting method for when Pn808 is 0.
- Method 2: Use the above setting method for when the absolute encoder zero point position offset is 0, and calculate the set value in step 4 using the following equation. New zero point position offset value = Present zero point position
- **Note** There is a limit to the number of times non-volatile memory in the Servo Drive can be written (10,000 times). Do not save *Absolute Origin Offset* (Pn200) for G Series or *Absolute Encoder Zero Point Position Offset* (Pn808) for G5 Series/W Series too frequently.

offset value - Feedback present position

Setting the Absolute Encoder Zero Point Position Offset Using an Origin Search Operation

The Absolute Origin Offset for G Series (Pn200) and Absolute Encoder Zero Point Position Offset for W Series (Pn808) can be automatically set at the end of an origin search by executing the origin search with the following operation pattern with a Position Control Unit with unit version 2.0 or later.

Operation pattern: Reversal mode 1

Origin detection method: Origin proximity input signal reversal, No origin proximity input signal reversal, or Do not use origin proximity input signal
To enable setting the absolute encoder zero point position offset with an origin search operation, the following axis parameter settings must be used with the above origin search operation pattern.

Origin search preset: Set.

Encoder type: Absolute Encoder

Axis Parameters

The following parameters are used to set the origin search operation mode.

PCU's			Con	tents			Setting
address	15 to 14	13	12	11 to 08	07 to 04	03 to 00	
d+1	0 (fixed)	Origin search preset	Origin search direction	Origin detection method	Origin search operation	0 (fixed)	Set the origin search operation. Origin search direction 0: Reversal mode 1 Origin detection method 0: With origin proximity input signal reversal (default setting) 1: Without origin proximity input sig- nal reversal 2: Not use origin proximity input sig- nal Origin search direction 0: Forward (default setting) 1: Reverse Origin search preset 1: Set
d+2	0 (fixed)	0	0	0 (fixed)	0 (fixed)	Encoder type	1: Absolute encoder

 $d = 1860 \text{ hex} + (\text{Axis No.} -1) \times 14 \text{ hex}$

When setting the absolute encoder's zero point position offset using an origin search, set the position command value (i.e., the present position preset value) to 0 and enable the origin search preset even when the preset value is not being used for the origin search.

- Each time the absolute encoder zero point position offset is set using an origin search operation, the *Absolute Origin Offset* for G Series (Pn200) or *Absolute Encoder Zero Point Position Offset* for G5 Series/W Series (Pn808) is saved to non-volatile memory in the Servo Drive. There is a limit to the number of times non-volatile memory in the Servo Drive can be written (10,000 times). Do not save the *Absolute Origin Offset* for G Series (Pn200) or *Absolute Encoder Zero Point Position Offset* for G Series (Pn200) or *Absolute Encoder Zero Point Position Offset* for G Series (Pn200) or *Absolute Encoder Zero Point Position Offset* for W Series (Pn808) too frequently.
 - The absolute encoder zero point position offset will not be set correctly using an origin search operation if an error occurs before origin search operation is completed. Repeat the operation if an error occurs.
 - If the absolute encoder PG zero point position offset is set using an origin search operation, the MECHATROLINK transfer cycle and communications cycle settings will need to be changed. Refer to *Transfer Cycle and Communications Cycle When Setting the Absolute Encoder Zero Point Position Offset Using an Origin Search Operation with Position Control Units with Unit Ver. 2.0* on page 246 and set the transfer cycle and communications cycle correctly. If the transfer cycle and communications cycle are not set correctly, the function to set the absolute encoder PG zero point position offset using an origin search operation will not operate correctly.

- To use a motor with absolute encoders with the OMNUC G series and G5 series Servo Drives, use the motor as described below when setting the absolute PG origin offset using origin search.
 - If the limit input in the origin search direction is ON when executing the origin search, origin search does not operate normally.
 Start the origin search after moving the position where the limit input in the origin search direction does not turn ON by jogging.
 - If the origin search is started when the limit input is ON in the origin search direction, an error terminates when the origin search operation is completed (data setting error, error code: 3099).

8-6-6 Establishing the Origin Using an Absolute Encoder

The Position Control Unit establishes the origin by reading the absolute value data saved in the absolute encoder using the following timing.

The zero point position offset that is set as a Servo parameter is added to the absolute value data that is read and input to the Axis Operating Input Memory Areas as the present position of the Position Control Unit.

In the following sections, the present position is given as the sum of the travel distances for the operations for each axis.

G5-series/G-series Servo
DrivesTurning ON the Servo Drive
When the control power supply is turned ON, the G5-series Servo Drive or G-

series Servo Drive reads the absolute value data of the Servo Drive or Gseries Servo Drive reads the absolute value data of the Servomotor with the absolute encoder to the Servo Drive. When MECHATROLINK communications start (i.e., when a connection is established), the Position Control Unit updates the present position using the absolute value data read from the Servo Drive.



The present position data will be read from the Servo Drive and the origin will be established when CONNECTION Bit is set and the MECHATROLINK is in communications status (the Connection Status Flag is ON). The origin will not be established (the No Origin Flag will be ON) and the present position will become invalid data when CONNECTION Bit is set and MECHATROLINK communications are stopped (the Connection Status Flag is OFF).

Note: Once the Present Position Preset has been executed, the preset data will be read when the connection is established until the absolute value data is read again.

n = Beginning word of Common Operating Memory Area: n = 1500 + (unit number × 25) b = Beginning word of Axis Operating Input Memory Areas specified in Common Parameters + (Axis No. –1) × 25

Absolute Encoder Origin

Executing DEVICE SETUP

The absolute value data for the axis for which DEVICE SETUP was executed is read.

DEVICE SETUP (word a+1, bit 11)			
SERVO LOCK (word a+1, bit 00)	<u>_</u>		
SERVO UNLOCK (word a+1, bit 01)			
Present Position Preset (word a, bit 08)		1	
Busy Flag (word b, bit 13)			<u> </u>
Positioning Completed (word b, bit 05)			
No Origin Flag (word b, bit 06)			
Servo ON (word b+1, bit 03)			
Feedback Present Position (words b+6, b+7)	Absolute value data	Preset data	Rereading absolute value data
Command Present Position (words b+8, b+9)	Absolute value data	Preset data	Rereading absolute value data

Once the connection has been established, the zero point position offset will be set, and the origin will not be established (the No Origin Flag will be OFF) if the offset is enabled using the device setup with MECHATROLINK in communications status. Then, the origin will be updated using the set present position offset when SERVO LOCK is executed.

 $\begin{array}{l} n = Beginning \mbox{ word of Common Operating Memory Area: } n = 1500 + (unit number \times 25) \\ a = Beginning \mbox{ word of Axis Operating Output Memory Areas specified in Common Parameters + (Axis No. -1) \times 25 \\ b = Beginning \mbox{ word of Axis Operating Input Memory Areas specified in Common Parameters + (Axis No. -1) \times 25 \\ \end{array}$

W-series Servo Drives

Starting MECHATROLINK Communications (When a Connection Has **Been Established)**

Absolute value data is read for all axes of the Servomotors with absolute encoders that are registered in the scan list.

CONNECT Bit (word n+1, bit 00)				
Connection Status Flag (word n+16, bit 15)				
Present Position Preset (word a, bit 08)				
Positioning Completed (word b, bit 05)				
No Origin Flag (word b, bit 06)				
Busy Flag (word b, bit 13)				
Feedback Present Position (words b+6, b+7)	Absolute value data	Preset data	Position data invalid	Rereading absolute value data
Command Present Position (words b+8, b+9)	Absolute value data	Preset data	Position data invalid	Rereading absolute value data

The absolute value data will be read and the origin will be established when the CONNECTION Bit is set and MECHATROLINK is in communications status (the Connection Status Flag is ON).

The origin will not be established (the No Origin Flag will be ON) and the present position will become invalid data when the CONNECT Bit is OFF and the MECHATROLINK communications are stopped (the Connection Status Flag is

OFF). n = Beginning word of Common Operating Memory Area: $n = 1500 + (unit number \times 25)$

b = Beginning word of Axis Operating Input Memory Areas specified in Common Parameters + (Axis No. -1) × 25

Executing SERVO LOCK

Absolute value data for the axis for which SERVO LOCK was executed is read.

If the absolute encoder zero point offset or other offline parameter is set and DEVICE SETUP is executed to enable the absolute encoder zero point offset while continuing MECHATROLINK communications, the origin for that axis will not be established. In that case, the absolute value data will be read again and the origin will be established based on the new parameter setting by executing SERVO LOCK.

CONNECT Bit (word n+1, bit 00)				
Present Position Preset (word a, bit 08)		 		
SERVO LOCK (word a+1, bit 00)				
SERVO UNLOCK (word a+1, bit 01)				
DEVICE SETUP (word a +1, bit 11)		 		
Connection Status Flag (word n+16, bit 15)		 		
Busy Flag (word b, bit 13)		 		
Positioning Completed (word b, bit 05)				
No Origin Flag (word b, bit 06)				
Servo ON (word b+1, bit 03)				
Feedback Present Position (words b+6, b+7)	Absolute value data	Rereading absolute value data	Preset data	Rereading absolute value data
Command Present Position (words b+8, b+9)	Absolute value data	Rereading absolute value data	Preset data	Rereading absolute value data

Once the connection has been established, the zero point position offset will be set, and the origin will not be established (the No Origin Flag will be OFF) if the offset is enabled using the device setup while continuing MECHATROLINK communications. Then, the absolute value data will be read again and the origin updated with the set present position offset will be established by executing SERVO LOCK.

- n = Beginning word of Common Operating Memory Area: n = $1500 + (unit number \times 25)$
- a = Beginning word of Axis Operating Output Memory Areas specified in Common Parameters + (Axis No. -1) × 25
- b = Beginning word of Axis Operating Input Memory Áreas specified in Common Parameters + (Axis No. -1) × 25

Note

- (1) Parameters that are not saved in the non-volatile memory built into the Servo Drive will be lost when DEVICE SETUP is executed. If data must be saved to the non-volatile memory, execute SAVE SERVO PARAME-TER first.
 - (2) The set data will be overwritten with the present position data of the Servo Drive when PRESENT POSITION PRESET is executed. To read the absolute value data again, turn OFF and ON again the power supply of the Servo Drive or execute DEVICE SETUP for a G5-series Servo Drive or G-series Servo Drive, and start MECHATROLINK communications or execute SERVO LOCK for a W-series Servo Drive.

SECTION 9 Positioning

This section provides an overview of direct operation and describes the parameter settings, data settings, and procedures required to perform direct operation. Information on interrupt feeding and torque limits is also provided here.

9-1	Direct Operation Overview							
9-2	Direct (Deration Procedure	339					
9-3	PCU Da	ata Settings for Direct Operation	339					
9-4	Using Direct Operation							
	9-4-1	Starting Direct Operation						
	9-4-2	Direct Operation Timing Charts	344					
	9-4-3	Changing Target Position	348					
	9-4-4	Changing Target Speed	350					
9-5	Interrup	t Feeding	351					
	9-5-1	Overview	351					
	9-5-2	Interrupt Feeding Procedure	351					
	9-5-3	Data Settings for Using Interrupt Feeding	352					
	9-5-4	Timing Chart for Interrupt Feeding	353					
	9-5-5	Present Position during Interrupt Feeding	354					
9-6	Torque	Limit Function	354					
9-7	Linear I	nterpolation	355					
	9-7-1	Overview of Linear Interpolation Function	355					
	9-7-2	Linear Interpolation Operation Procedure	357					
	9-7-3	Setting Data for Linear Interpolation Operation	358					
	9-7-4	Linear Interpolation Operation	363					

9-1 Direct Operation Overview

Direct operation of the PCU enables positioning by simply writing target position data and target speed data directly from a ladder program to a specified area in the CPU Unit.

Positioning operations using direct operation are executed according to the position command value and speed command value in the Axis Operating Memory Area set in Common Parameters and the acceleration/deceleration constants set in the Servo Parameters.

The following is an example for a W-series Servo Drive.



a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) \times 25

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

The position command value and speed command value set in the Axis Operating Output Memory Area using the MOV instruction are output to the PCU automatically during I/O refresh. Direct operation is started when the ABSO-LUTE MOVEMENT Bit or RELATIVE MOVEMENT Bit allocated in the Axis Operating Output Memory Area turns ON.

9-2 Direct Operation Procedure

The procedure for using direct operation is as follows:

- *1,2,3...* 1. Set the Common Parameters and save them.
 - Turn ON the PCU again or restart the Unit. The data for the Common Parameters set in step 1 above is enabled.
 - 3. Start MECHATROLINK communications.
 - 4. Set the Servo Parameters and save them. Set the Servo Parameters required to execute direct operation. To set parameters permanently, execute SAVE SERVO PARAMETER (writes to the non-volatile memory). To enable changed offline parameters, turn ON the power to the Servo Drive again or execute DEVICE SETUP. For details, refer to *5-3 Transferring Servo Parameters*.
 - 5. Execute SERVO LOCK.
 - Set the data used for direct operation. Set the positioning data for direct operation in the position command value of the Axis Operating Output Memory Area. Set the speed data for direct operation in the speed command value of the Axis Operating Output Memory Area.
 - 7. Start ABSOLUTE MOVEMENT or RELATIVE MOVEMENT.

When using the PCU for the first time or to change the Common Parameter data, steps 1 and 2 must be performed.

After executing SERVO LOCK, WRITE SERVO PARAMETER can be executed to set the acceleration/deceleration used each time positioning is performed.

Note When specifying the Servo Parameters for acceleration/deceleration every time, make sure that the axis operation is stopped (Busy Flag = 0) while changing the acceleration/deceleration constants using WRITE SERVO PARAMETER. Do not change parameters during axis operation.

Changing parameters while the axis is operating may result in incorrect positioning or other malfunction.

9-3 PCU Data Settings for Direct Operation

A simple explanation of the main parameters and data used to execute direct operation is provided here. To execute direct operation, apart from the parameters explained here, the following parameters also need to be set as basic settings for operating the PCU.

- External I/O Signal Allocations Refer to 6-4 Standard Settings for Servo Drives Using MECHATROLINK.
- Command Unit Refer to *7-2 Control Units*. The setting units for parameters and data depend on the specified command unit.

G5-series Acceleration/Deceleration Servo Parameters

Servo Parameter Area

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting
Accelera- tion/decelera-	Pn811	Linear acceleration constant	10,000 command units/s ²	-32768 to 32767	2	100
tion constants	Pn814	Linear deceleration constant	10,000 command units/s ²	–32768 to 32767	2	100
Accelera- tion/decelera- tion filters	Pn818	Position command FIR filter time constant	0.1 ms	0 to 10000	2	0

G-series Servo Parameters

Servo Parameter Area

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting
Accelera- tion/decelera-	Pn107	Linear acceleration constant	10,000 command units/s ²	-32768 to 32767	2	100
tion constants	Pn10A	Linear deceleration constant	10,000 command units/s ²	-32768 to 32767	2	100
Accelera- tion/decelera- tion filters	Pn10E	Moving average time	0.1 ms	0 to 5100	2	0

W-series and SMARTSTEP Junior Servo Parameters

Servo Parameter Area

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting	W Series	SMART- STEP Junior
Accel- eration/ decel-	Pn80A	First-step linear accelera- tion constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Not sup- ported
con- stants	Pn80B	Second-step linear acceler- ation constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Sup- ported
	Pn80C	Acceleration constant switching speed	100 com- mand units/s	0 to 65535	2	0	Sup- ported	Not sup- ported
	Pn80D	First-step linear decelera- tion constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Not sup- ported
	Pn80E	Second-step linear deceler- ation constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Sup- ported
	Pn80F	Deceleration constant switching speed	100 com- mand units/s	0 to 65535	2	0	Sup- ported	Not sup- ported
Accel- eration/	Pn810	Exponential accelera- tion/deceleration bias	Command units/s	0 to 32767	2	0	Sup- ported	Not sup- ported
decel- eration filters	Pn811	Exponential accelera- tion/deceleration time con- stant	0.1 ms	0 to 5100	2	0	Sup- ported	Not sup- ported
	Pn812	Movement average time	0.1 ms	0 to 5100	2	0	Sup- ported	Not sup- ported

First-step acceleration/deceleration curve operations require parameter settings for Pn80B and Pn80E only. For the acceleration/deceleration curve, Pn810 and Pn811 must be set when using an exponential curve, and Pn812 must be set when using an S-curve.

The SMARTSTEP Junior Servo Drives do not support the following parameters: Pn80A, Pn80C, Pn80D, and Pn80F. They also do not support acceleration/deceleration filters, so the filters cannot be set. One-step linear acceleration/deceleration curves are set using only Pn80B and Pn80E.

For details on acceleration/deceleration curves, refer to 7-4 Acceleration and Deceleration Operations.

Model	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
R88D-KN⊡- ML2	Pn431	Positioning completion range 1	Command unit	0 to 262144	4	300
R88D-GN⊡- ML2	Pn060	Positioning completion range 1	Command unit	0 to 10000	2	25
R88D-WT	Pn500	Positioning completion range 1	Command unit	0 to 250	2	3
R88D-WN□- ML2	Pn522	Positioning completion range 1	Command unit	0 to 1073741824	4	3
SMART- STEP Junior	Pn522	Positioning completion range 1	Command unit	0 to 1073741824	4	10

Control Status Parameters

Minimum Limit Settings

Under the following conditions, the Servo Parameters listed above, apart from Pn80B, Pn80E, and Pn500/Pn522, are used with the default settings and do not need to be changed.

- Command unit: Pulse
- Acceleration/deceleration curve: Linear acceleration/deceleration for a single step only (trapezoidal curve)

Axis Operating Output Memory Areas (Operating Commands)

Name	Word	Bits	Contents	G5 Series G Series	W Series	SMARTST EP Junior
ABSOLUTE MOVEMENT	а	03	$0 \rightarrow 1$: Starts absolute movement.	Supported	Supported	Supported
RELATIVE MOVEMENT		04	$0 \rightarrow 1$: Starts relative movement.	Supported	Supported	Supported
INTERRUPT FEEDING		05	1: Enables interrupt feeding.	Supported	Supported	Supported
Position command value	a+2 a+3		Position command value (right- most word)	Supported	Supported	Supported
			Position command value (leftmost word)			
			Unit: Command unit			
			Command range: -2,147,483,648 to 2,147,483,647 (80000000 hex to 7FFFFFF hex) The command value is limited by the allowable positioning range when using RELATIVE MOVE- MENT.			
			(For details, refer to 9-4-1 Starting Direct Operation.)			

Name		Word	Bits	Contents	G5 Series G Series	W Series	SMARTST EP Junior
Speed command value		a+4 a+5		Speed command value (rightmost word)	Supported	Supported	Supported
				Speed command value (leftmost word)			
				Unit: Command units/s			
				Command range: 0 to 2,147,483,647 (00000000 hex to 7FFFFFF hex) The upper limit setting of the speed command value depends on the Servo Drive specifications.			
Accelera- tion/decel- eration	Exponential curve desig- nation	a+16	03	1: Use exponential accelera- tion/deceleration curve.	Not sup- ported	Supported	Not sup- ported
curve designa- tion	S-curve des- ignation		04	1: Use S-curve acceleration/deceleration curve.	Supported	Supported	Not sup- ported
Forward rotation current limit			14	1: Use forward torque limit.	Supported	Supported	Not sup- ported
Reverse ro limit	tation current		15	1: Use reverse torque limit.	Supported	Supported	Not sup- ported

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

The settings of the following are enabled when the movement command start bit turns ON: Position command value, Acceleration/Deceleration Curve Designation Bit, Interrupt Feeding Designation Bit, and Forward/Reverse Rotation Current Limit Designation Bit.

The speed command value can always be changed during operation. By overwriting the speed command value, the speed for the positioning operation can be changed.

The G5-series Servo Drives and G-series Servo Drives do not support exponential curve designation filters. When using a G5-series Servo Drive or G-series Servo Drive, do not attempt to use an exponential curve designation.

The SMARTSTEP Junior Servo Drives do not support acceleration/deceleration filters and torque limits. When using a SMARTSTEP Junior Servo Drive, do not attempt to use an acceleration/deceleration curve designation or forward/reverse torque limit designation.

Note Do not set both the exponential curve designation and S-curve designation to 1 (enabled) in the acceleration/deceleration curve designation. Enabling both settings may cause a malfunction.

Axis Operating Input Memory Areas (Monitoring)

(6,			
Name	Word	Bits	Contents
Receiving Command	b	00	0: Command reception enabled.
Flag			$0 \rightarrow 1$: Command reception started.
			1: Receiving command (command reception disabled).
PCU Positioning Com- pleted Flag		05	$0 \rightarrow 1$: Positioning is completed.
No Origin Flag		06	0: Origin established.
			1: No origin established.
Error Flag		12	0: No axis error.
			1: Axis error has occurred.
Busy Flag		13	1: Axis busy (axis operation executing)
Feedback present posi- tion	b+6 b+7		Present position: Feedback present position (rightmost word) Feedback present position (leftmost word)
Command present posi- tion	b+8 b+9		Present position: Command position (rightmost word) Command position (leftmost word)

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) \times 25

The Receiving Command Flag turns ON for at least one cycle time when the command to start direct operation is received by the PCU. Use the Receiving Command Flag when starting and changing target positions to control the ON/OFF timing for ABSOLUTE MOVEMENT and RELATIVE MOVEMENT.

Using Direct Operation 9-4

Starting Direct Operation 9-4-1

Two methods for starting direct operation are available, as follows:

1,2,3	 When the ABSOLUTE MOVEMENT Bit turns ON
	2. When the RELATIVE MOVEMENT Bit turns ON
	When starting direct operation, make sure that the ABSOLUTE MOVE- MENT/RELATIVE MOVEMENT Bit remains ON until the Receiving Command Flag or the Busy Flag in the Axis Operating Input Memory Area turns ON.
	Direct operation commands will be ignored if the SERVO UNLOCK, DEVIA- TION COUNTER RESET, EMERGENCY STOP, or DECELERATION STOP Bit is ON. Be sure these bits remain OFF while performing direct operation.
Executing ABSOLUTE MOVEMENT	ABSOLUTE MOVEMENT positions the axis at the specified position using the position command value in the Axis Operating Output Memory Area as absolute data. If ABSOLUTE MOVEMENT is started without the origin established (No Origin Flag = 1), a Present Position Unknown Error (axis error code: 3030) will occur, and positioning will not be executed.
	The positioning range for ABSOLUTE MOVEMENT is any absolute position for which the travel amount from the present position is between

for which the travel amount from the present position is between -2,147,483,648 and 2,147,483,647 (command units).

If ABSOLUTE MOVEMENT is executed for an absolute position that exceeds this range, positioning will be performed not to a position based on the present origin, but to a position exceeding the upper or lower limit of the command present position.



Positioning is not possible (Position Designation error).

For details on acceleration/deceleration operations during direct operation, refer to 7-4 Acceleration and Deceleration Operations.

9-4-2 Direct Operation Timing Charts

Timing Chart for Movement Command Execution The following timing chart is for when RELATIVE MOVEMENT is executed. The timing chart for ABSOLUTE MOVEMENT is the same, except that positioning is executed with the absolute position.

Using Direct Operation

Section 9-4



remains OFF for at least one cycle time.

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) × 25 b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

Even for a zero or minute travel distance, the Receiving Command Flag and Busy Flag for the movement command will remain ON for at least one CPU Unit cycle time.

The PCU Positioning Completed Flag for the movement command will remain OFF for at least one cycle time if positioning stops within the positioning completion range for the target position.

Timing Chart for Error/Warning Flag during **Command Execution**

If an error occurs due to an illegal command value when a direct operation command is executed, the command will not be executed, and the timing chart will be as follows:



b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

If a Servo Drive warning occurs when a movement command is being executed, such as the speed command value exceeding the Servo Drive's limit value, the Error Flag in the above timing chart is replaced by the Warning Flag (word b, bit 11). After the command has been executed and the Busy Flag has been ON for at least one CPU Unit cycle time, the Error Flag and Warning Flag will turn ON at the same time as the Receiving Command Flag and Busy Flag turn OFF after either one PLC cycle time or after one MECHATROLINK communications time, whichever is longer.

Timing Chart for Errors or Warnings during Axis Operation

When an error occurs during direct operation, such as a limit input during axis operation, the axis will either decelerate to a stop, immediately stop (emergency stop), or continue operation, depending on the error.

The timing charts when an error occurs during axis operation are provided below.



When an Error that Stops (Deceleration/Emergency Stop) Axis Operation Occurred

The PCU Positioning Completed Flag turns OFF during execution of the movement command. This flag will not turn ON if the executed positioning command is not completed due to an error. If an error occurs, the Error Flag will turn ON, the axis error code stored, and the axis will stop according to the stopping method set for when errors occur. The Error Flag will turn OFF when AXIS ERROR RESET is executed. If an error occurs, after the Busy Flag turns ON, the Error Flag and axis error code will remain ON and set for either one cycle time or MECHATROLINK communications cycle, whichever is longer.

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) × 25

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) \times 25

When a Warning Allowing Axis Operation to Continue Occurred

Example: Parameter Setting Warning during Servo Parameter Transfer



a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) × 25 b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

9-4-3 Changing Target Position

during direct operation, the target position for the positioning operation can be changed by setting a new position command value in the Axis Operating Output Memory Area and executing ABSOLUTE MOVEMENT or RELATIVE MOVEMENT again.

The Busy Flag remains ON during direct operation until the positioning operation is completed. The Receiving Command Flag in the Axis Operating Input Memory Area is used to control the ON/OFF timing of the ABSOLUTE/RELA-TIVE MOVEMENT Bit to change the target position. The target position can be changed when the Receiving Command Flag is OFF. When executing the target position change, make sure that the ABSOLUTE/RELATIVE MOVE-MENT Bit remains ON until the Receiving Command Flag turns ON.



Target position changed

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) × 25

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) \times 25

Note

- (1) When executing movement commands sequentially, make sure that the movement command bit remains OFF for a minimum of either the PLC cycle time × 2 or the MECHATROLINK communications cycle × 2, whichever is longer. If the time when the movement command bit is OFF is too short, the PCU will not be able to detect the rising edge of the movement command bit, preventing reception of the command.
 - (2) When executing the movement command again during axis operation to change the target position, do not change the acceleration/deceleration curve designation from the original setting when operation was started. Changing the acceleration/deceleration curve designation when changing the target position during axis operation may result in displaced positioning or other malfunction.

Operation Pattern when Changing the Target Position



Executing Change to Target Position for Small Travel Distance If the target position is changed using RELATIVE MOVEMENT executed for a zero or minute travel distance or ABSOLUTE MOVEMENT executed to a position near the present position, positioning will be performed to the target position after decelerating to a stop and reversing direction.



9-4-4 Changing Target Speed

during direct operation, by setting a new speed command value in the Axis Operating Output Memory Area, the target speed for the positioning operation can be changed. The speed command value is enabled at the I/O refresh after it is set in the Axis Operating Memory Area. The PCU starts changing to the new target speed at the acceleration/deceleration set in the Servo Parameters when the new speed command value is enabled.



Timing Chart for Changing Target Position and Target Speed



9-5 Interrupt Feeding

9-5-1 Overview

The PCU's interrupt feeding is a positioning operation that moves the axis for a specified amount, separate from the position command value, from the position at which the external interrupt signal was input either in the same direction as the direction of travel or in the opposite direction. The specified amount of travel is performed at the speed being used when the external interrupt signal was input.

Moving for a Specified Distance in the Direction of Travel Mo

Moving for a Specified Distance in Opposite Direction



When an interrupt input signal is not input, positioning is performed to the target position specified in the position command value, as normal.

9-5-2 Interrupt Feeding Procedure

The interrupt feeding function is a supplementary function for positioning operations using direct operation (ABSOLUTE MOVEMENT, RELATIVE MOVEMENT). Interrupt feeding can be executed using the same procedure as for direct operation by simply specifying interrupt feeding at the start of direct operation.

The following procedure omits the steps up to SERVO LOCK, showing the start procedure only.

1. Transfer the Servo Parameters for interrupt feeding. To set parameters permanently, execute SAVE SERVO PARAMETER

- Set the data used for direct operation.
 Set the positioning data for direct operation in the position command value of the Axis Operating Output Memory Area.
 Set the speed data for direct operation in the speed command value of the Axis Operating Output Memory Area.
- 3. Turn ON the INTERRUPT FEEDING Bit.

(writes to the non-volatile memory).

4. Start ABSOLUTE MOVEMENT or RELATIVE MOVEMENT.

9-5-3 Data Settings for Using Interrupt Feeding

To execute interrupt feeding, the following parameters and data must be set in addition to the parameters and data required for direct operation.

Axis Parameter Area

Interrupt Input Signal Parameters

PCU's	Cont	tents	Setting
address	Bits 08 to 15	Bits 00 to 07	
d	Origin input sig- nal selection	Interrupt input signal selection	Interrupt input signal selection 00: Phase Z 01: External latch signal 1 input 02: External latch signal 2 input 03: External latch signal 3 input

 $d = 1860 \text{ hex} + (\text{Axis No.} -1) \times 14 \text{ hex}$

When using external latch signals 1 to 3, the external latch signal to be used must be allocated in the Servo Drive's external input allocations.

When using a SMARTSTEP Junior Servo Drive, only 00 (phase Z) and 01 (external latch signal 1) can be used for the origin input signal selection.

Servo Parameter Area

Interrupt Feeding Parameters

Parameter name for G Series (Parameter name for W Series)	Unit	Setting range	Paramet er size	Default setting	G5 Series	G Series	W Series and SMARTS TEP Junior Series
Final distance for exter- nal input positioning (Final travel distance for external positioning)	Command unit	-1,073,741,823 to 1,073,741,823	4	100	Pn820	Pn203	Pn814

Set the travel distance from the position at which the interrupt input signal is input. The travel direction for when the interrupt input signal is input is determined by the sign (positive or negative) in the set value. The operation performed when the interrupt input signal is input depends on the sign for this parameter and the sign of the position command value for direct operation, as follows:

Position command value for direct	Servo parameter Pn203/Pn814/Pn820 (External Positioning Final Travel Distance)				
operation	Positive value	Negative value			
Positive value	When the interrupt input sig- nal is input, interrupt feeding is executed in the same direc- tion as the travel direction when movement was started.	When the interrupt input sig- nal is input, interrupt feeding is executed in the opposite direction from the travel direc- tion when movement was started. (Reversal mode operation)			
Negative value	When the interrupt input sig- nal is input, interrupt feeding is executed in the opposite direction from the travel direc- tion when movement was started. (Reversal mode operation)	When the interrupt input sig- nal is input, interrupt feeding is executed in the same direc- tion as the travel direction when movement was started.			

Note Starting in Position Control Unit with version 2.0 and later, the *External Positioning Final Travel Distance* for G5-series Servo Drives (Pn820), the *Final Distance for External Input Positioning* for G-series Servo Drives (Pn203) or

Final Travel Distance for External Positioning for W-series Servo Drives (Pn814) is used for internal processing in the Position Control Unit when one of the following origin search operation patterns are selected.

- When the origin detection method is set to any value except for 0 (Origin proximity input signal reversal)
- When setting a zero point position offset for an origin search using an absolute encoder

Because Pn814 is automatically overwritten by the Position Control Unit when starting the origin search operation, if using the interrupt feeding after executing the origin search in the above operation pattern, be sure to set the *External Positioning Final Travel Distance* for G5-series Servo Drives (Pn820), the *Final Distance for External Input Positioning* for G-series Servo Drives (Pn203) or *Final Travel Distance for External Positioning* for W-series Servo Drives (Pn814) again.

Axis Operating Output Memory Areas (Operating Commands)

Name	Word	Bits	Contents
INTERRUPT FEEDING	а	05	1: Enables interrupt feeding

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) \times 25

9-5-4 Timing Chart for Interrupt Feeding

Interrupt feeding is executed by turning ON the INTERRUPT FEEDING Bit in the Axis Operating Output Area and starting direct operation.

Timing Chart for Interrupt Feeding Using ABSOLUTE MOVEMENT

The following timing chart is for when INTERRUPT FEEDING is executed with ABSOLUTE MOVEMENT. The timing chart for RELATIVE MOVEMENT is the same, except that positioning is executed with the relative position if no interrupt input signal is input.



In interrupt feeding as well as normal direct operations, the target position (target position for when no interrupt input signal is input), target speed, and override value can be changed by executing another movement command. Note that, however, in the interval of positioning for the final travel distance for external positioning after the input of the interrupt input signal, another INTERRUPT FEEDING cannot be executed (it will be ignored).

9-5-5 Present Position during Interrupt Feeding

The positioning range for direct operation (ABSOLUTE MOVEMENT and RELATIVE MOVEMENT) is a command present position range of -2,147,483,648 to 2,147,483,647 (command units) when it is not limited by limit input signals or software limits. (For details on the command present position, refer to 7-3 Coordinate System and Present Position. For details on RELATIVE MOVEMENT, refer to Executing RELATIVE MOVEMENT in 9-4-1 Starting Direct Operation.)

When INTERRUPT FEEDING is executed, the positioning range and position command value setting range are the same, but the positioning range is subject to the following conditions when interrupt input signals are input near the upper limit/lower limit of the positioning range.

When the interrupt feeding operation after interrupt input exceeds the upper or lower limit of the positioning range due to the *Final Travel Distance for External Positioning* setting and the input position of the interrupt input signal, the current position reference point (i.e., the origin) and the positioning range change and positioning to the original position is no longer possible.



When interrupt feeding is used, set the limit inputs, software limits, and other required settings, and make sure that the positioning range is not exceeded due to the interrupt feeding operation.

9-6 Torque Limit Function

When using a W-series Servo Drive, set Servo Parameters *Forward Rotation External Current Limit* (Pn404) and *Reverse Rotation External Current Limit* (Pn405), and turn ON the Forward Rotation External Current Limit Designation Bit and Reverse Rotation External Current Limit Designation Bit and Reverse Rotation External Current Limit Designation Bit in the Axis Operating Output Memory Area when starting a direct operation to apply a current (torque) limit during direct operation.

When using a G-series Servo Drive, set Servo Parameter *No. 2 Torque Limit* (Pn05F) and turn ON the Forward Rotation External Current Limit Designation Bit and Reverse Rotation External Current Limit Designation Bit in the Axis Operating Output Memory Area when starting a operation to apply a current or torque limit during direct operation.

When using a G5-series Servo Drive, set Servo Parameter No.1 Torque Limit (Pn013) and turn ON the Forward rotation output torque limit and Reverse rotation torque limit in the Axis Operating Output Memory Area when starting a operation to apply torque limit during direct operation.

If another direct operation command is executed at this point, the torque limit function can be switched during operation.

For further details on the torque limit function, refer to 10-4 Torque Limits.

The SMARTSTEP Junior Servo Drives do not support torque control.

9-7 Linear Interpolation

With unit version 1.1 or later of the Position Control Unit, a linear interpolation function that can perform linear interpolation using multiple axes in combination has been added.

Linear interpolation can be executed for the axes of a Servo Drive connected to the Position Control Unit, using up to four axes in any combination for axes 1 to 4 or axes 5 to 8.

Note Do not write data to the CJ1W-NC□71 unit version 1.1 or later that has been backed up from the CJ1W-NC□71 unit version 1.0 to a Memory Card in the CPU Unit (using either the CPU Unit's simple backup function or the Position Control Unit's backup function). When unit version 1.0 backup data is restored to unit version 1.1 or later, the linear interpolation function cannot be used. To transfer unit version 1.0 settings to unit version 1.1 or later, use the parameter transfer function using the READ DATA, WRITE DATA, or SAVE DATA Bit in the Common Operating Memory Area.

9-7-1 Overview of Linear Interpolation Function

Linear interpolation operates according to the following settings data for axes 1 to 4 or axes 5 to 8.

- Designation of axes to combine for linear interpolation (Interpolation axis designation)
- · Position command values for individual interpolation axes
- Designation of absolute and relative positions for position command values for individual interpolation axes (interpolation position designation)
- Interpolation speed command values and speed command values for individual interpolation axes
- Interpolation acceleration and deceleration times

Based on these commands, the respective linear interpolation operations are started by the LINEAR INTERPOLATION START Bit allocated in the Axis Operating Output Memory Area (axis 1 for interpolation operations using axes 1 to 4, and axis 5 for interpolation operations using axes 5 to 8).



The speeds for individual axes in linear interpolation are determined by the following equations. (The same equations are used for 2-axis or 3-axis linear interpolation.)

Example: Linear Interpolation for 4-axis (Axis 1 to Axis 4) Operation

Axis 1 speed = Interpolation speed \times Movement of Axis 1 / Total movement Axis 2 speed = Interpolation speed \times Movement of Axis 2 / Total movement Axis 3 speed = Interpolation speed \times Movement of Axis 3 / Total movement Axis 4 speed = Interpolation speed \times Movement of Axis 4 / Total movement

Total movement = $\sqrt{(Movement of Axis 1)^2 + (Movement of Axis 2)^2 + (Movement of Axis 3)^2 + (Movement of Axis 4)^2}$

If speeds set using speed command values for individual interpolation axes are exceeded when interpolation speed command values are resolved into individual axis speeds by the above equations, those speeds are automatically lowered so that each axis involved in linear interpolation is moved at the set speed command value.

Example: Linear Interpolation for 2-axis (Axis 1 and Axis 2) Operation



- **Note** In the linear interpolation function of the Position Control Unit, the acceleration and deceleration operations for each axis based on the interpolation acceleration time and interpolation deceleration time that were set are performed by using the *Second-step Linear Acceleration Constant* for W-series Servo Drives (Pn80B), *Linear Acceleration Constant* for G-series Servo Drives (Pn107) or *Linear Acceleration Constant* for G5-series Servo Drives (Pn80E), *Linear Deceleration Constant* for W-series Servo Drives (Pn80E), *Linear Deceleration Constant* for G-series Servo Drives (Pn814).

The setting unit for the Second-step Linear Acceleration Constant and Second-step Linear Deceleration Constant is 10,000 command units/s². In operation, the actual acceleration and deceleration times are thus affected by the precision of the command unit.

9-7-2 Linear Interpolation Operation Procedure

Just as with direct operation (absolute and relative movement commands), linear interpolation is executed by setting operation commands in the Axis Operating Output Memory Area.

Common commands for linear interpolation operations, such as designation of interpolation axis combinations, interpolation speeds, and interpolation acceleration and deceleration times, are set in the Axis Operating Output Memory Area for axis 1 or axis 5, and position and speed command values for individual linear interpolation axes are set in the Axis Operating Output Memory Area for those particular axes. Then linear interpolation operations are started by using the LINEAR INTERPOLATION SETTING and LINEAR INTERPOLATION START Bits.

The following procedure describes the steps related to starting linear interpolation operations, from servolock onwards. 1,2,3... 1. Set the data for linear interpolation operation. Set the axis combination designation, the absolute and relative positioning designation for individual interpolation axes, the interpolation acceleration and deceleration times, and the interpolation speed command value in the Axis Operating Output Memory Area for axis 1 (when using any combination of axes 1 to 4 for linear interpolation) or axis 5 (when using any combination of axes 5 to 8 for linear interpolation). T 2. Set the data for individual axis interpolation operations. In the Axis Operating Output Memory Area, for each axis specified for interpolation, set the position data in the position command value and the speed data in the speed command value. \downarrow 3. Set linear interpolation. Turn ON the LINEAR INTERPOLATION SETTING Bit in the Axis Operating Output Memory Area for axis 1 (or axis 5). \downarrow 4. Start linear interpolation operation. Turn ON the LINEAR INTERPOLATION START Bit in the Axis Operating Output Memory Area for axis 1 (or axis 5).

- Note (1) When the linear interpolation function is used, set the communications cycle (PCU address 1856 hex: bits 07 to 00) in the MECHATROLINK communications settings in the CPU's Common Parameter Area to a value at least one greater than the normal minimum value. If the communications cycle is set too low, the command response times for PCU functions may become too long.
 - (2) From the point where linear interpolation is first set until axis operation is finished (i.e., while the LINEAR INTERPOLATION SETTING Bit is ON or until the Linear Interpolation Executing Flag turns OFF), command response times for the other axes not involved in the linear interpolation are delayed by up to four communications cycles per combination of axes executing linear interpolation.

9-7-3 Setting Data for Linear Interpolation Operation

PCU linear interpolation operations can be executed simultaneously for two combinations of axes, from axes 1 to 4 and axes 5 to 8. For a combination of axes 1 to 4, set the following linear interpolation-related data in the Axis 1 Operating Output Memory Area, and for a combination of axes 5 to 8 set the data in the Axis 5 Operating Output Memory Area.

- · Interpolation acceleration and deceleration times
- Interpolation axis designation
- Interpolation position designation
- Interpolation speed command value

Also, in the Axis Operating Output Memory Area for each axis, make the following settings related to individual axes specified for linear interpolation.

- Position command value (Command value absolute and relative positions are handled according to the interpolation position designation.)
- Speed command value (Functions as the maximum speed designation for individual interpolation axes, with respect to the interpolation speed command value.)

Once the above settings have been made, then execute linear interpolation operations by using the following two bits allocated in the Axis Operating Output Memory Area for axis 1 or axis 5.

- LINEAR INTERPOLATION SETTING
- LINEAR INTERPOLATION START

Example: Linear Interpolation for 4-axis (Axis 1 to Axis 4) Operation



Note

- (1) When executing interpolation operations using a combination of axes 5 to 8, the interpolation axis designation and interpolation position designation are made for axes 5 to 8 respectively in the Axis 5 Operating Output Memory Area, in bits 04 to 07 in the corresponding words in the above diagram.
 - (2) Set the acceleration/deceleration curve designation (the S-curve designation) and the forward/reverse torque designation for each designated interpolation axis.

Common Parameter Area

MECHATROLINK Communications Settings

PCU's address	Con	tents	Setting
	Bits 08 to 15	Bits 00 to 07	
1856 hex	Transfer cycle	Communica- tions cycle	Set the communications cycle to a value one greater than the min- imum communications cycle value determined by the number of connected devices. (Refer to <i>6-2-3 MECHATROLINK Communica-</i> <i>tions Settings</i> .)

Servo Parameter Area

Acceleration/Deceleration Constants

Туре	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
Accelera- tion/deceler-	Pn80C	Acceleration constant switching speed	100 command units/s	Must be set to 0.	2	0
ation constants	Pn80F	Deceleration constant switching speed	100 command units/s	Must be set to 0.	2	0

Note

When executing linear interpolation operations, the Acceleration Constant Switching Speed (Pn80C) and the Deceleration Constant Switching Speed (Pn80F) in the Servo Parameter Area must each be set to 0. Linear interpolation operations cannot be properly executed if any number other than 0 is set. Linear interpolation operations can be executed for this PCU only with a 1-step acceleration/deceleration curve.

Axis Operating Output Memory Areas (Operating Commands)

Settings for Linear Interpolation	on Operations:	: Axis 1,	Axis 5 (Operating
Output Memory Areas				

Name	Word	Bits	Contents
LINEAR INTERPO- LATION SETTING	а	00	$0 \rightarrow 1$: Starts linear interpolation setting 1: LINEAR INTERPOLATION START valid.
LINEAR INTERPO- LATION START		01	$0 \rightarrow 1$: Starts linear interpolation movement (Valid only when LINEAR INTERPOLATION SETTING = 1)
Option Command Value 1	a+10		Linear interpolation acceleration time Unit: ms Command range: 0 to 65535 (0000 hex to FFFF hex)
	a+11		Not used (Set value disabled.)
Option Command Value 2	a+12		Linear interpolation deceleration time Unit: ms Command range: 0 to 65535 (0000 hex to FFFF hex)
	a+13		Not used. (Set value disabled.)
Interpolation Axis Designation	a+21	00 to 03 (Axis 1 to Axis 4) 04 to 07 (Axis 5 to Axis 8)	0: Not designated as interpolation axis. 1: Designated as interpolation axis. Designates axes for executing linear interpolation. Combinations of axes 1 to 4 are designated in bits 00 to 03 (Axis 1 Operat- ing Output Memory Area), and combinations of axes 5 to 8 are designated in bits 04 to 07 (Axis 5 Operating Output Memory Area). Bits 00 to 03, and bits 04 to 07, correspond respectively to axes 1 to 4 and axes 5 to 8.
Interpolation Posi- tion Designation	a+22	00 to 03 (Axis 1 to Axis 4) 04 to 07 (Axis 5 to Axis 8)	 0: Execute interpolation operations using absolute position. 1: Execute interpolation operations using relative position. Specifies, for axes designated by the interpolation axis designation, whether positioning in interpolation operations is to be executed using absolute or relative positions. Positions for axes 1 to 4 are designated in bits 00 to 03 (Axis 1 Operating Output Memory Area), and positions for axes 5 to 8 are designated in bits 04 to 07 (Axis 5 Operating Output Memory Area). These bits correspond respectively to axes 1 to 4 and axes 5 to 8.
Interpolation Speed Command Value	a+23 a+24		Speed command value (rightmost word) Speed command value (leftmost word) Unit: Command units/s Command range: 0 to 2,147,483,647 (00000000 hex to 7FFFFFFF hex) For details on speeds for interpolation operations, refer to <i>Linear Interpola-</i> <i>tion Operation Speeds</i> on page 366.

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

Linear Interpolation

Section 9-7

Na	me	Word	Bits	Contents	G Series	W Series	SMART-
							STEP Junior
Position Command Value a+2 a+3		a+2 a+3		Position command value (rightmost word) Position command value (leftmost word) Unit: Command unit Command range: -2,147,483,648 to 2,147,483,647 (8000000 hex to 7FFFFFF hex) The command value is limited by the allowable positioning range. (For details, refer to <i>9-4-1 Starting Direct</i> <i>Operation</i> .)	Supported	Supported	Supported
Speed Command Value (for Position Control)		a+4 a+5		Speed command value (rightmost word) Speed command value (leftmost word) Unit: Command units/s Command range: 0 to 2,147,483,647 (00000000 hex to 7FFFFFF hex) For details on speeds for interpolation operations, refer to <i>Linear Interpolation</i> <i>Operation Speeds</i> on page 366.	Supported	Supported	Supported
Accelera- tion/decel- eration	Exponen- tial curve designation	a+16	03	1: Use exponential acceleration/decelera- tion curve.	Not sup- ported	Supported	Not sup- ported
curve des- ignation	S-curve designation		04	1: Use S-curve acceleration/deceleration curve.	Supported	Supported	Not sup- ported
Forward rotation current limit			14	1: Use forward torque limit.	Supported	Supported	Not sup- ported
Reverse rota limit	tion current		15	1: Use reverse torque limit.	Supported	Supported	Not sup- ported

Settings for Interpolation Axis Operations: Axis Operating Output Memory Areas for Axes Designated by Interpolation Axis Designation

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

The G5-series Servo Drives and G-series Servo Drives do not support exponential curve designation filters. When using a G-series Servo Drive, do not attempt to use an exponential curve designation filter.

The SMARTSTEP Junior Servo Drives do not support acceleration/deceleration filters and torque limits. When using a SMARTSTEP Junior Servo Drive, do not attempt to use an acceleration/deceleration curve designation or forward/reverse torque limit designation.

- Note (1) For interpolation acceleration and deceleration time settings, the upper and lower limits for axis acceleration and deceleration speeds are restricted to a range of 1 to 65,535 [×10,000 command units/s²]. If the acceleration or deceleration speeds calculated from the linear interpolation operation settings for an interpolation axis are extremely low or high, the linear interpolation following may not be correct.
 - (2) The interpolation acceleration time and interpolation deceleration time are set by converting to the units [×10,000 command unit/s2] of the Second-step Linear Acceleration Constant for W-series Servo Drives (Pn80B), Linear Acceleration Constant for G-series Servo Drives (Pn107) or Linear Acceleration Constant for G5-series Servo Drives (Pn811) and Second-step Linear Deceleration Constant for W-series Servo Drives (Pn80E), Linear Deceleration Constant for G-series Servo Drives

(Pn10A), or *Linear Deceleration Constant* for G5-series Servo Drives (Pn814).

- (3) If the linear interpolation operation is performed using an S-curve, be sure to set the *Movement Average Time* for W-series Servo Drives (Pn812), *Moving Average Time* for G-series Servo Drives (Pn10E), and *Moving Average Time Interval* for G5-series Servo Drives (Pn818) to the same value before for all axes that are specified an interpolation axes, and specify the S-curve for all interpolation axes when starting the linear interpolation operation. If different movement average times are set for interpolation axes, and the S-curve designation does not match, operation will not be possible using the linear interpolation following.
- (4) With linear interpolation operations, an exponential curve cannot be designated. In the acceleration/deceleration curve designation, the designation of an exponential curve will be disabled.

Axis Operating Input Memory Areas (Monitoring)

Linear Interpolation Operation Status: Axis 1, Axis 5 Operating Input Memory Areas

Name	Word	Bits	Contents
Linear Interpolation Setting Completed Flag	b+24	00	 0: Linear interpolation setting command reception enabled. 0 to 1: Linear interpolation setting completed. 1: Linear interpolation setting completed (setting command reception disabled).
Linear Interpolation Executing Flag		13	1: Linear interpolation operation executing

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

Interpolation Axis Operation Status: Axis Operating Input Memory Areas for Axes Designated by Interpolation Axis Designation

Name	Word	Bits	Contents
Receiving Com- mand Flag	b	00	0: Command reception enabled. 0 to 1: Command reception started. 1: Receiving command (command reception disabled).
PCU Positioning Completed Flag		05	0 to 1: Positioning is completed.
No Origin Flag		06	0: Origin established. 1: No origin established.
Error Flag		12	0: No axis error. 1: Axis error has occurred.
Busy Flag		13	1: Axis busy (axis operation executing)
Feedback Present	b+6		Present position:
Position	b+7		Feedback present position (rightmost word) Feedback present position (leftmost word)
Command Present Position	b+8 b+9		Present position: Command position (rightmost word) Command position (leftmost word)

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

When a linear interpolation operation command (LINEAR INTERPOLATION SETTING or LINEAR INTERPOLATION START) is executed, the linear interpolation operation status can be checked in the Axis Operating Input Memory Area of the axis (Axis 1 or Axis 5) for which the command was executed. The operation status of axes designated by linear interpolation designation can be checked in the Axis Operating Input Memory Areas for individual interpolation axes.

9-7-4 Linear Interpolation Operation

After the data related to linear interpolation has been set, linear interpolation operation is started by means of the following two operations.

Executing LINEAR INTERPOLATION SETTING

Turn ON the LINEAR INTERPOLATION SETTING Bit in the Axis 1 (or Axis 5) Operating Output Memory Area.

When this bit is turned ON, the settings for linear interpolation operation that have been set in the Axis Operating Output Memory Area for axis 1 (or axis 5) are obtained, along with the position command values and position speed values for individual interpolation axes, and preparations will be executed for linear interpolation operation. Execute LINEAR INTERPOLATION SETTING when the Linear Interpolation Setting Completed Flag in the Axis Operating Input Memory Area for axis 1 (or axis 5), and the Busy Flag and Error Flag for the axes designated by the interpolation axis designation, are reset. If the Linear Interpolation Setting Completed Flag for axis 1 (or axis 5) is set, or if an axis designated by the interpolation axis designation is busy (i.e., Busy Flag ON), or if an error has occurred (i.e., Error Flag ON), the LINEAR INTERPOLATION SETTING command will be ignored and will not be executed.

When LINEAR INTERPOLATION SETTING has been successfully completed, the Linear Interpolation Setting Completed Flag for axis 1 (or axis 5) will be set.

Note

- e (1) LINEAR INTERPOLATION SETTING is automatically written from the PCU to the Servo Parameters. When executing LINEAR INTERPOLA-TION SETTING, do not execute any of the Servo Parameter transfer operations (WRITE, READ, or SAVE).
 - (2) When LINEAR INTERPOLATION SETTING is executed, the Second-step Linear Acceleration Constant for W-series Servo Drives (Pn80B), Linear Acceleration Constant for G-series Servo Drives (Pn107) or Linear Acceleration Constant for G5-series Servo Drives (Pn811) and Second-step Linear Deceleration Constant for W-series (Pn80E), Linear Deceleration Constant for G-series Servo Drives (Pn10A) or Linear Deceleration Constant for G5-series Servo Drives (Pn814) parameters in the Servo Parameters are changed. Be careful of changes to acceleration and deceleration speed settings if interpolation axes are operated independently (by direct operation, JOG, etc.) after LINEAR INTERPOLA-TION SETTING has been executed (regardless of whether LINEAR INTERPOLATION START has been executed). Change the Second-step Linear Acceleration Constant for W-series Servo Drives (Pn80B), Linear Acceleration Constant for G-series Servo Drives (Pn107) or Linear Acceleration Constant for G5-series Servo Drives (Pn811) and Second-step Linear Deceleration Constant for W-series (Pn80E), Linear Deceleration Constant for G-series Servo Drives (Pn10A) or Linear Deceleration Constant for G5-series Servo Drives (Pn814) as required, by using a WRITE SERVO PARAMETERS command.

Executing LINEAR INTERPOLATION START

Turn ON the LINEAR INTERPOLATION START Bit in the Axis 1 (or Axis 5) Operating Output Memory Area. Linear interpolation operation will start, based on the settings from when the LINEAR INTERPOLATION SETTING Bit was turned ON just before. If the LINEAR INTERPOLATION SETTING Bit is OFF, LINEAR INTERPOLATION START will be disabled.

	Execute LINEAR INTERPOLATION START when the Linear Interpolation Executing Flag in the Axis Operating Input Memory Area for axis 1 (or axis 5), and the Busy Flag and Error Flag for the axes designated by the interpolation axis designation, are reset. If the Linear Interpolation Setting Executing for axis 1 (or axis 5) is set, or if an axis designated by the interpolation axis desig- nation is busy (i.e., Busy Flag ON), or if an error has occurred (i.e., Error Flag ON), the LINEAR INTERPOLATION START command will be ignored and will not be executed.	
Note	(1) From the time that the LINEAR INTERPOLATION SETTING Bit turns ON until the LINEAR INTERPOLATION START Bit turns ON, do not make changes to any data related to linear interpolation operations. In addition, do not make any changes to speed command values for individual axes while linear interpolation operations are in progress (i.e., while the Busy Flag for any interpolation axis is ON).	
	(2) For an axis designated as 0 (absolute position) by the interpolation position designation, the origin must be established. If the origin is not established (i.e., if the No Origin Flag is ON), a Present Position Unknown Error (axis error code: 3030) will occur when LINEAR INTERPOLATION START is executed (i.e., when linear interpolation operation is started), and the axis will not be operated. At that time, the other interpolation axes will also be decelerated to a stop.	
Example: Linear Interpolation Operation.	The basic flow of operations for linear interpolation is shown below. In this example, "a" indicates the beginning word of the Axis Operating Output Memory Area for axis 1 when a combination of axes 1 to 4 is used for linear interpolation, and axis 5 when a combination of axes 5 to 8 is used. Likewise, "b" indicates the beginning word of the Axis Operating Input Memory Area for axis 1 when a combination of axes 1 to 4 is used, and axis 5 when a combina- tion of axes 5 to 8 is used.	
1,2,3	 With the bits in the interpolation axis designation (word a+21), designate the axes to be combined for the linear interpolation operation. With the bits in the interpolation position designation (word a+22), desig- nate either absolute or relative position for the position command values for the axes designated as interpolation axes. 	
	 With the interpolation speed command value (words a+23, a+24), designate the interpolation speed (i.e., the integrated speed). With the speed command values for the axes designated as interpolation axes, designate the maximum speed for each axis. For details on the relation between linear interpolation operation speeds and interpolation speed command values, refer to the <i>Linear Interpolation Operation Speeds</i> on page 366. 	
	 With option command value 1 (word a+10) and option command value 2 (word a+12), set the interpolation acceleration time and the interpolation deceleration time. 	
	5. Turn ON the LINEAR INTERPOLATION SETTING Bit (word a, bit 00).	
	6. When the linear interpolation operation settings have been completed, the Linear Interpolation Setting Completed Flag (word b+24, bit 00) will turn ON.	

7. After the above settings have been completed, turning ON the LINEAR IN-TERPOLATION START Bit (word a, bit 01) while the LINEAR INTERPO-LATION SETTING Bit (word a, bit 00) is still ON will start positioning for the designated axes, to the position designated in the interpolation position designation.

during the linear interpolation operation, the Linear Interpolation Executing Flag (word b+24, bit 13) will turn ON. When executing a linear interpolation operation, leave the LINEAR INTERPOLATION SETTING Bit and the LIN-EAR INTERPOLATION START Bit ON until the Linear Interpolation Executing Flag turns ON.

Timing Charts

 When LINEAR INTERPOLATION START is Executed After Linear Interpolation Settings are Completed



remain ON until the Linear Interpolation Executing Flag turns ON. The Linear Interpolation Setting Completed Flag turns ON until LINEAR INTERPOLATION SETTING turns OFF or errors occur on the axes specified in the Interpolation axis designation The Linear Interpolation Executing Flag turns ON until the interpolation operation has completed or LINEAR INTERPOLATION START turns OFF.

a = Beginning Word of Axis 1 (Axis 5) Operating Output Memory Areas b = Beginning Word of Axis 1 (Axis 5) Operating Input Memory Areas

> If the LINEAR INTERPOLATION SETTING Bit turns OFF before LINEAR INTERPOLATION START is executed. LINEAR INTERPOLATION START is disabled.

 When LINEAR INTERPOLATION SETTING and LINEAR INTERPOLA-TION START are Executed Simultaneously



a = Beginning Word of Axis 1 (Axis 5) Operating Output Memory Areas b = Beginning Word of Axis 1 (Axis 5) Operating Input Memory Areas

> If the LINEAR INTERPOLATION SETTING and LINEAR INTERPOLA-TION START Bits are turned ON simultaneously, the linear interpolation operation is started after the linear interpolation settings are completed.

Note

(1) The same LINEAR INTERPOLATION START cannot be executed again with a new command while linear interpolation operation is in progress (i.e., while the Linear Interpolation Executing Flag is ON). If it is executed again, it is disabled. It is possible, however, to start an independent linear interpolation for a combination of axes 1 to 4, or of axes 5 to 8, while an operation involving the other combination is already in progress.

(2) While a linear interpolation operation is in progress, do not make changes to the speed command value for any axis involved in that operation (i.e., for any axis designated in the interpolation axis designation). If a speed command value is changed during operation, the operation will not be executed correctly.

	If direct operation (absolute movement of an axis designated in the interpolation and lation operation is in progress, that axis of linear interpolation operation. The acceleration and cution of LINEAR INTERPOLATION SE resolved with the interpolation accelerate ear Interpolation Executing Flag will state all the originally designated interpolation pleted, or until the LINEAR INTERPOLATION A	or relative movement) is executed for ixis designation while a linear interpo- will start operation separately from the eleration and deceleration speeds at d deceleration speeds set by the exe- ETTING (determined from the speed tion and deceleration times). The Lin- y ON, however, until the operations of n axes (including this axis) are com- TION START Bit turns OFF.	
	If MECHATROLINK communications are lation Setting Completed Flag and the L ON, the flags will hold their status until M resumed, regardless of the status of the and LINEAR INTERPOLATION START B	e interrupted while the Linear Interpo- inear Interpolation Executing Flag are /IECHATROLINK communications are LINEAR INTERPOLATION SETTING Bits.	
Linear Interpolation Operation Speeds	There are two types of settings relations speeds: interpolation speed command windividual axes designated by the interpolation	ere are two types of settings related to linear interpolation operatio eeds: interpolation speed command values and speed command values for dividual axes designated by the interpolation axis designation.	
	The interpolation speed command value linear interpolation. This command value lation axes and becomes the speed com	e designates the integrated speed for e is resolved for the individual interpo- mand value for each axis.	
	The speed command values for individu tion operation, on the other hand, are individual axes. For all of the interpolatic is lowered so that the speed resolved fr value does not exceed this set value. (time is operated at 100% of the speed co other axes are lowered in order to perfor	al axes at the time of linear interpola- the maximum speed settings for the on axes, the linear interpolation speed om the interpolation speed command (The axis with the longest movement ommand value, and the speeds for the rm the linear interpolation.)	
Example	2-axis interpolation (axis 1 and axis 2), with target positions of 3,000 and 4,000 (command units) respectively, and with relative positioning designated and the following speed command values set:		
	Interpolation speed command value	:2,000 (command units/s)	
	Axis 1 speed command value:	600 (command units /s)	
	Axis 2 speed command value:	1,000 (command units /s)	
	The speeds resolved from the interpolation speed command value (in com- and units/s) are 1,200 for axis 1 and 1,600 for axis 2, but for linear interpola- on operation the speeds are reduced to the following values in order for the aximum speeds set by the speed command values for individual axes to not e exceeded.		
	Interpolation speed command value	:1,000 (command units/s)	
	Axis 1 speed command value:	600 (command units/s)	
	Axis 2 speed command value:	800 (command units/s)	
	If a speed command value for an indi- speed limit will set for that interpolation interpolation speed command value will	a speed command value for an individual axis is set to 0, no maximun sed limit will set for that interpolation axis and the speed resolved from the erpolation speed command value will be used as is.	
	If the interpolation speed command value is set to 0, there is no interpolation speed designation and the interpolation speed is determined by the speed command values for the individual interpolation axes.		

If either the interpolation speed command value or a speed command value for an individual interpolation axis is set to 0, the individual interpolation axes are operated at the same speed, at the minimum unit (1 command unit/s) for the speed command value.

If less than the minimum unit (1 command unit/s) for an individual axis speed command value results when the interpolation speed command value is resolved for individual interpolation axes, that axis will be operated at the minimum unit (1 command unit/s) for the speed command value.

Note Do not use an override during linear interpolation operation. The linear interpolation operation will not be executed correctly if an override is used.

Stopping during Linear Interpolation

If a stop command, error, or Servo unlock command occurs for one of multiple axes for which linear interpolation is being performed, the operation of the axes will be as follows and the axes will not stop at the same time.

Axis Error Occurs in One of the Interpolated Axes:

The other interpolated axes will be decelerated to a stop when the Error Flag turns ON for the axis in which the error has occurred.

Stop Command Given for One of the Interpolated Axes:

The other interpolated axes will be decelerated to a stop when the Stop Execution Flag turns ON for the axis for which the stop command was given.

Servo Unlock Command Given for One of the Interpolated Axes:

The other interpolated axes will be decelerated to a stop when the Servo ON Flag turns OFF for the axis for which the servo unlock command was given.

If a stop command is given for only one axis to stop a linear interpolation operations, the axes will not stop at the same time, as described above. To stop all of the interpolated axes at the same time, stop commands must be given for all interpolated axes at the same time.

If an alarm in the Servo Drive results in an axis error, the axis for which the error occurred will be stopped according to the stop method set in the Servo Drive, but the other interpolated axes will be decelerated to a stop. The distances required to stop will thus be different.
SECTION 10 Other Operations

This section describes the following operations: servo lock/unlock, jogging, override, torque limits, speed control, torque control, backlash compensation, software limits, and stop functions.

10-1	Servo L	ock/Unlock	70
	10-1-1	Overview	70
	10-1-2	Servo Lock/Unlock Operation	70
10-2	Jogging	3	71
	10-2-1	Overview of Operation	71
	10-2-2	Procedure for Jogging Operations	71
	10-2-3	PCU Data Settings for Jogging Operations	72
	10-2-4	Starting Jogging	75
10-3	Overrid	e	76
	10-3-1	Overview	76
	10-3-2	Override Operation	76
10-4	Torque	Limits	77
	10-4-1	Overview	77
	10-4-2	Constant Torque Limits	78
	10-4-3	Torque Limits Set by Operating Commands 3'	79
10-5	Speed C	Control	85
	10-5-1	Overview	85
	10-5-2	Starting Speed Control	85
	10-5-3	Switching Control Mode	96
10-6	Torque	Control	98
	10-6-1	Overview	98
	10-6-2	Starting Torque Control	98
10-7	Backlas	h Compensation	05
	10-7-1	Overview	05
	10-7-2	Backlash Compensation Procedure 4	05
	10-7-3	Backlash Compensation Data Settings 44	06
10-8	Softwar	e Limits	08
	10-8-1	Overview	08
	10-8-2	Procedure for Using Software Limits 4	08
	10-8-3	Software Limit Data Settings 4	09
	10-8-4	Software Limit Operation	10
10-9	Stop Fu	nctions	12
	10-9-1	Overview	12
	10-9-2	Deceleration Stop	12
	10-9-3	Emergency Stop	14
	10-9-4	Stop Function Timing Chart	15
10-10	DEVIA	TION COUNTER RESET 4	18
	10-10-1	Overview of DEVIATION COUNTER RESET 4	18
	10-10-2	Using DEVIATION COUNTER RESET 4	19

10-1 Servo Lock/Unlock

10-1-1 Overview

The Servo lock/unlock function either creates a Servo Drive position loop (SERVO LOCK) or releases the position loop (SERVO UNLOCK).

10-1-2 Servo Lock/Unlock Operation

The SERVO LOCK/UNLOCK operations start when the SERVO LOCK/UNLOCK Bit in the Axis Operating Output Memory Area turns ON.

When a Servomotor with absolute encoder is used, the absolute value data is read by executing SERVO LOCK. For details, refer to *8-6-6 Establishing the Origin Using an Absolute Encoder*.

Axis Operating Output Memory Areas (Operating Commands)

Name	Word	Bits	Contents
SERVO LOCK	a+1	00	$0 \rightarrow 1$: Starts SERVO LOCK.
SERVO UNLOCK		01	$0 \rightarrow 1$: Starts SERVO UNLOCK.

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) \times 25

Execute SERVO LOCK when the Busy Flag for the corresponding axis is OFF. If SERVO LOCK is executed while the axis's Busy Flag is ON, a Multistart Error (axis error code: 3050) will occur and SERVO LOCK will not be executed. When executing SERVO LOCK, make sure that the SERVO LOCK Bit remains ON until the Receiving Command Flag or Busy Flag turns ON.

If the SERVO UNLOCK, DEVIATION COUNTER RESET, EMERGENCY STOP, OR DECELERATION STOP command bit is turned ON, the SERVO LOCK command will be ignored. Be sure that all of these command bits are OFF before attempting to lock the Servo.

SERVO UNLOCK can be executed in any PCU status. When executing SERVO UNLOCK, make sure that the SERVO UNLOCK Bit remains ON until either the Receiving Command Flag turns ON, or the Busy Flag turns OFF after executing SERVO UNLOCK.

Other axis operation commands will be ignored while SERVO UNLOCK is being executed and while the SERVO UNLOCK Bit is ON.

Note When executing SERVO UNLOCK, the PCU executes DECELERATION STOP for the Servo Drive, followed immediately (without waiting for the axis to stop) by SERVO UNLOCK. If SERVO UNLOCK is executed during axis operation, the corresponding axis will be put in Servo free run state. Therefore, prevent the machine from moving due to momentum by using the dynamic brake or executing DECELERATION STOP before executing SERVO UNLOCK to stop axis operation. Movement of the machine due to momentum may result in damage to the machine or an accident. When executing SERVO UNLOCK during speed control or torque control, the Servo Drive will recover in the position control mode's Servo lock status the next time SERVO LOCK is executed.

Jogging

Axis Operating Input Memory Areas (Monitoring)

Name	Word	Bits	Contents
Receiving Command	b	00	0: Command reception enabled.
Flag			$0 \rightarrow 1$: Command reception started.
			1: Receiving command (command reception disabled).
Error Flag		12	0: No axis error
			1: Axis error has occurred.
Busy Flag		13	1: Axis busy (axis operation executing).
SVON (Servo ON) Flag	b+1	03	0: Servo unlocked.
			1: Servo locked.

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. – 1) $\times\,25$

The Servo lock status of each axis can be confirmed by checking the SVON Flag in the Axis Operating Input Memory Areas.

Timing Chart

The following timing chart is for when SERVO LOCK and SERVO UNLOCK are executed.



a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) × 25

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

10-2 Jogging

10-2-1 Overview of Operation

The axis travels in the specified direction at the specified speed while the JOG Bit is ON, and decelerates to a stop when the JOG Bit turns OFF. Jogging operations can be executed even if the origin has not been established.

10-2-2 Procedure for Jogging Operations

The procedure for using jogging is as follows:

- *1,2,3...* 1. Set the Common Parameters and save them.
 - Turn ON the PCU again or restart the Unit. The data for the Common Parameters set in step 1 above is enabled.
 - 3. Start MECHATROLINK communications.

- 4. Set the Servo Parameters and save them. Set the Servo Parameters required to execute direct operation. To set parameters permanently, execute SAVE SERVO PARAMETER (writes to the non-volatile memory). To enable changed offline parameters, turn ON the power to the Servo Drive again or execute DEVICE SETUP. For details, refer to *5-3 Transferring Servo Parameters*.
- 5. Execute SERVO LOCK.
- Set the data used for the jogging operation. Set the jogging operation speed data in the speed command value of the Axis Operating Output Memory Area.
- 7. Start the jogging operation.

When using the PCU for the first time or to change the Common Parameter data, steps 1 and 2 must be performed. After executing SERVO LOCK, WRITE SERVO PARAMETER can be executed to set the acceleration/deceleration used each time the jogging operation is performed.

Note When specifying the Servo Parameters for acceleration/deceleration every time, make sure that the axis operation is stopped (Busy Flag = 0) while changing the acceleration/deceleration constants using WRITE SERVO PARAMETER. Do not change parameters during axis operation. Changing parameters while the axis is operating may result in displaced positioning or other malfunction.

10-2-3 PCU Data Settings for Jogging Operations

A simple explanation of the main parameters and data used to execute the jogging operation is provided here. To execute jogging operations, apart from the parameters explained here, the following parameters also need to be set as basic settings for operating the PCU.

- External I/O Signal Allocations
 - Refer to 6-4 Standard Settings for Servo Drives Using MECHATROLINK.
- Command Unit
- Refer to 7-2 Control Units.

The setting units for parameters and data depend on the specified command unit.

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
Acceleration/ deceleration con-	Pn811	Linear acceleration constant	10,000 com- mand units/s ²	-32768 to 32767	2	100
sianis	Pn814	Linear deceleration constant	10,000 com- mand units/s ²	-32768 to 32767	2	100
Acceleration/ deceleration filters	Pn818	Position command FIR filter time constant	0.1 ms	0 to 10000	2	0

G5-series Acceleration/Deceleration Constants

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
Acceleration/ deceleration con-	Pn107	Linear acceleration constant	10,000 com- mand units/s ²	-32768 to 32767	2	100
sianis	Pn10A	Linear deceleration constant	10,000 com- mand units/s ²	-32768 to 32767	2	100
Acceleration/ deceleration filters	Pn10E	Moving average time	0.1 ms	0 to 5100	2	0

G-series Acceleration/Deceleration Constants

W-series and SMARTSTEP Junior Acceleration/Deceleration Constants

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting	W Series	SMART- STEP Junior
Accelera- tion/ decelera-	Pn80A	First-step linear acceleration constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Not sup- ported
stants	Pn80B	Second-step linear accelera- tion constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Sup- ported
Accelera- tion/	Pn810	Exponential accelera- tion/deceleration bias	Command units/s	0 to 32767	2	0	Sup- ported	Not sup- ported
decelera- tion filters	Pn811	Exponential accelera- tion/deceleration time con- stant	0.1 ms	0 to 5100	2	0	Sup- ported	Not sup- ported
	Pn812	Movement average time	0.1 ms	0 to 5100	2	0	Sup- ported	Not sup- ported

First-step acceleration/deceleration curve operations require parameter settings for Pn80B and Pn80E only. For the acceleration/deceleration curve, Pn810 and Pn811 must be set when using an exponential curve, and Pn812 must be set when using an S-curve.

The SMARTSTEP Junior Servo Drives do not support the following parameters: Pn80A, Pn80C, Pn80D, and Pn80F. They also do not support acceleration/deceleration filters, so the filters cannot be set. One-step linear acceleration/deceleration curves are set using only Pn80B and Pn80E.

For details on acceleration/deceleration curves, refer to 7-4 Acceleration and Deceleration Operations.

Axis Operating Output Memory Areas (Operating Commands)

Name	Word	Bits	Contents	G Series	W Series	SMART- STEP Junior
JOG	а	09	$0 \rightarrow 1$: Starts jogging. 1 $\rightarrow 0$: Stops jogging.	Sup- ported	Sup- ported	Sup- ported
Direction designation		10	0: Forward rotation direction 1: Reverse rotation direction	Sup- ported	Sup- ported	Sup- ported
Speed command value	a+4 a+5		Speed command value (rightmost word) Speed command value (leftmost word) Unit: Command units/s Command range: 0 to 2,147,483,647 (00000000 hex to 7FFFFFF hex) The upper limit setting of the speed com- mand value depends on the specifica- tions of the Servo Drive.	Sup- ported	Sup- ported	Sup- ported

Name		Word	Bits	Contents	G Series	W Series	SMART- STEP Junior
Accelera- tion/decelera- tion curve	Exponential curve desig- nation	a+16	03	1: Use exponential acceleration/decelera- tion curve.	Not sup- ported	Sup- ported	Not sup- ported
designation	S-curve des- ignation		04	1: Use S-curve acceleration/deceleration curve.	Sup- ported	Sup- ported	Not sup- ported
Forward rotation current limit			14	1: Use forward torque limit.	Sup- ported	Sup- ported	Not sup- ported
Reverse rotation current limit			15	1: Use reverse torque limit.	Sup- ported	Sup- ported	Not sup- ported

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) \times 25

The direction designation setting when the jogging operation starts (JOG Bit turns ON) is enabled. Even if the designation is changed during jogging, the direction will not change.

The settings for the Acceleration/Deceleration Curve Designation Bit, and Forward/Reverse Rotation Current Limit Designation Bit when jogging starts (JOG Bit turns ON) and stops (JOG Bit turns OFF) are enabled.

The speed command value can always be changed during operation. By overwriting the speed command value, the speed for the jogging operation can be changed.

The G5-series Servo Drives and G-series Servo Drives do not support exponential curve designation. When using a G-series Servo Drive, do not attempt to use an exponential curve designation.

The SMARTSTEP Junior Servo Drives do not support acceleration/deceleration filters and torque limits. When using a SMARTSTEP Junior Servo Drive, do not attempt to use an acceleration/deceleration curve designation or forward/reverse torque limit designation.

Note Do not set both the exponential curve designation and S-curve designation to 1 (enabled) in the acceleration/deceleration curve designation. Enabling both settings may cause a malfunction.

Axis Operating Input Memory Areas (Monitoring)

Name	Word	Bits	Contents
Receiving Command	b	00	0: Command reception enabled.
Flag			$0 \rightarrow 1$: Command reception started.
			1: Receiving command (command reception disabled).
Error Flag		12	0: No axis error.
			1: Axis error has occurred.
Busy Flag		13	1: Axis busy (axis operation executing).
Feedback present posi- tion	b+6 b+7		Present position Feedback position (rightmost word) Feedback position (leftmost word)
Command present posi- tion	b+8 b+9		Present position Command position (rightmost word) Command position (leftmost word)

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

10-2-4 Starting Jogging

Jogging is performed according to the direction designation and speed command value set in the Axis Operating Output Memory Areas and started when the Jog Bit turns ON. Execute JOG when the Busy Flag for the corresponding axis is OFF. If the jogging operation is started while the axis's Busy Flag is ON, a Multistart Error (axis error code: 3050) will occur and JOG will not be executed.

If the SERVO UNLOCK, DEVIATION COUNTER RESET, EMERGENCY STOP, OR DECELERATION STOP command bit is turned ON, the jog command will be ignored. Be sure that all of these command bits are OFF before attempting a jog operation.

The jogging operation will continue while the Jog Bit is ON. To enable the PCU to receive the JOG command correctly, make sure that the Jog Bit remains ON while the Receiving Command Flag is ON.

The direction designation setting when the jogging operation starts (Jog Bit turns ON) is enabled. Even if the designation is changed during jogging, the direction will not change. Jogging decelerates to a stop when the Jog Bit turns OFF.

The Busy Flag remains ON during the jogging operation. The PCU Positioning Completed Flag does not turn ON when jogging stops.

The following timing chart is for when JOG is executed in the forward rotation direction.



a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) × 25 b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

Timing Chart

10-3 Override

10-3-1 Overview

The override operation is used to change the speed of an active axis. While the Override Enable Bit is ON, the target speed is changed by applying the override value that is set in the Axis Operating Output Memory Area.

The override can be set from 0.01% to 327.67% in units of 0.01%. The speed command values (Speed command value for speed control in speed control mode) set for direct operation, origin return, jogging, and speed control operations are taken to be 100%.

The override function is disabled during origin search or torque control.

The target speed when override is enabled is as follows:

Target speed = Speed command value $\times \frac{\text{Override}}{10000}$

10-3-2 Override Operation

The override function can be used for the following operations.

- Enabling an Override for a Series of Operations
 - Set the override beforehand, and execute each operating command with the Override Enable Bit turned ON. The target speed for the operation will be the initial speed command value multiplied by the override.
- Switching between Enabling and Disabling Override during Axis Operation

When the Override Enable Bit is turned ON for an active axis performing direct operation, origin return, or jogging operation, the speed will change to the above target speed using the acceleration/deceleration set in the Servo Parameters for acceleration/deceleration (the acceleration/dece

When using speed control, the speed can be changed without using acceleration/deceleration by using the Override Enable Bit during axis operation, however, the linear acceleration/deceleration curve can be applied to the speed change by setting the *Soft Start Acceleration/Deceleration Time* in the Servo Parameters. (For details on the acceleration/deceleration curve, refer to 7-4 Acceleration and Deceleration Operations. For details on speed control, refer to 10-5 Speed Control.)

If the Override Enable Bit is turned OFF, the speed is changed in the same way as when the Override Enable Bit was ON from the present target speed (speed command value multiplied by override value) to the speed set in the speed command value.

Axis Operating Output Memory Areas (Operating Commands)

Name	Word	Bits	Contents
Override Enable Bit	а	14	0: Override disabled.
			1: Override enabled.
Override	a+14		Override ratio
			Unit: 0.01%
			Command range: 1 to 32,767 (0001 hex to 7FFF hex)

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) \times 25

The override can be set in the range 1 to 32767 (0.01% to 327.67%). The actual Servomotor speed control range and minimum control unit depend on the Servo Drive specifications.

While override is enabled, any changes to the override value are enabled immediately. While override is enabled, the override ratio can be changed during operation by setting a new override value in the Axis Operating Output Memory Area.



The following timing chart is for when the override function is executed during a jogging operation.



b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

10-4 Torque Limits

10-4-1 Overview

When using a G5-series Servo Drive, a G-series Servo Drive or a W-series Servo Drive, torque limits can be applied to Servomotor axis operation in position control functions (ABSOLUTE MOVEMENT, RELATIVE MOVEMENT), ORIGIN SEARCH, ORIGIN RETURN, JOG, SPEED CONTROL, TORQUE CONTROL, and Stop Functions (DECELERATION STOP, EMERGENCY STOP).

There are two types of torque limit, one that constantly functions for forward and reverse rotation and the other that can be disabled/enabled in operating commands. For speed control, torque limits can also be added to option command values. (For details, refer to *Option Command Value during Speed Control* in *10-5-2 Starting Speed Control*.)

When enabling multiple torque limit functions simultaneously, the Servomotor's output torque is limited by the minimum torque limit setting.

The SMARTSTEP Junior Servo Drives do not support torque control. The following settings and functions cannot be used.

10-4-2 Constant Torque Limits

G5-series Servo Drives To apply a constant torque limit to axis operation, set the Servo Parameters *No.1 Torque Limit* (Pn013) and *No.2 Torque Limit* (Pn522). By setting these parameters, the output torque of the operating Servomotor will always be limited by the set ratio of the rated torque. Also by setting the *Torque Limit Selection* (Pn521), the forward torque limit and reverse torque limit will be as shown in the following table.

Parameter No.	Set value	Forward torque limit	Reverse torque limit
Pn521	0.1	Pn013	Pn013
	2	Pn013	Pn522

Torque Limit Parameters

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn013	No.1 torque limit	%	0 to 500	2	500
	Pn522	No.2 torque limit	%	0 to 500	2	500

G-series Servo Drives

To apply a constant torque limit to axis operation, set the Servo Parameters *No.1 Torque Limit* (Pn05E) and *No.2 Torque Limit* (Pn05F). By setting these parameters, the output torque of the operating Servomotor will always be limited by the set ratio of the rated torque. Also by setting the *Torque Limit Selection* (Pn003), the forward torque limit and reverse torque limit will be as shown in the following table.

Parameter No.	Set value	Forward torque limit	Reverse torque limit
Pn003	1	Pn05E	Pn05E
	2	Pn05E	Pn05F

Torque Limit Parameters

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn05E	No.1 torque limit	%	0 to 500	2	300
	Pn05F	No.2 torque limit	%	0 to 500	2	100

W-series Servo Drives

To apply constant torque limits to axis operation, set the Servo Parameters *Forward Torque Limit* (Pn402) and *Reverse Torque Limit* (Pn403). By setting this parameter, the output torque of the operating Servomotor will always be limited by the set ratio of the rated torque.

Torque Limit Parameters

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn402	Forward torque limit	%	0 to 800	2	350
	Pn403	Reverse torque limit	%	0 to 800	2	350

10-4-3 Torque Limits Set by Operating Commands

Torque limit functions set by operating commands are executed by turning ON the Forward/Reverse Rotation Current Limit Bit in the Axis Operating Output Area and starting the operating command. The *Forward/Reverse Rotation Current Limit* setting is enabled when the Start Bit for the operating command turns ON.

When using W-series Servo Drives, the output torque is limited when the Servomotor is operating according to the set values in the Servo Parameters *Forward Rotation External Current Limit* (Pn404) and *Reverse Rotation External Current Limit* (Pn405) by turning ON the Forward/Reverse Current Limit Designation Bit when using operating commands.

When using G-series Servo Drives, the output torque is limited when the Servo Parameter *Torque Limit Selection* (Pn003) is set to 3. The Forward Rotation Current Limit Designation Bit will be enabled when the PCL is turned ON. The Reverse Rotation Current Limit Designation Bit will be enabled when the NCL is turned ON.

- PCL ON: When either the Forward Torque Limit (CN1 PCL: Pin 7) or Axis Output Operating Area is turned ON.
- NCL ON When either the Reverse Torque Limit (CN1 NCL: Pin 8) or Axis Output Operating Area is turned ON.

Forward Torque Limit

PCL OFF	PCL ON
Pn05E	Pn05F

Reverse Torque Limit

NCL OFF	NCL ON
Pn05E	Pn05F

A torque limit that has been enabled will continue to function until the Forward/Reverse Current Limit Designation Bit is turned OFF and the next operating command is sent (the specified torque limits continue to be applied even in Servo lock status after positioning stops).

Note If the axis stops due to an error during axis operation with the torque limit function enabled, the torque limit when stopped will depend on the setting of the Forward/Reverse Current Limit Designation Bit. When the Current Limit Designation Bit is turned ON only at the start of axis operation, and turned OFF during axis operation, the torque limit will be disabled if the axis stops due to an error. Make sure that the Current Limit Designation Bit remains ON for the interval that the torque limit needs to be enabled.

The Torque Limit Status Flag in the Servo Status Flags in the Axis Operating Input Memory Area turns ON for the time the torque limit is functioning.

G-series Servo Parameter Area

Torque Limit Parameters

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn05E	No.1 torque limit	%	0 to 500	2	300
	Pn05F	No.2 torque limit	%	0 to 500	2	100

W-Series Servo Parameter Torque Limit Parameters Area

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn404	Forward rotation external current limit	%	0 to 800	2	100
	Pn405	Reverse rotation external current limit	%	0 to 800	2	100

Axis Operating Output Memory Areas (Operating Commands)

Name	Word	Bits	Contents
ABSOLUTE MOVEMENT	а	03	$0 \rightarrow 1$: Starts absolute movement.
RELATIVE MOVEMENT		04	$0 \rightarrow 1$: Starts relative movement.
ORIGIN SEARCH		06	$0 \rightarrow 1$: Starts origin search.
ORIGIN RETURN		07	$0 \rightarrow 1$: Starts origin return.
JOG		09	$0 \rightarrow 1$: Starts jogging.
			$1 \rightarrow 0$: Stops jogging
Direction designation		10	0: Forward rotation direction
			1: Reverse rotation direction
DECELERATION STOP		15	$0 \rightarrow 1$: Starts deceleration stop.
SPEED CONTROL	a+1	02	$0 \rightarrow 1$: Starts speed control.
TORQUE CONTROL		03	$0 \rightarrow 1$: Starts torque control.
EMERGENCY STOP		15	$0 \rightarrow 1$: Starts emergency stop.
Forward rotation current limit designation	a+16	14	1: Use forward torque limit.
Reverse rotation current limit designation		15	1: Use reverse torque limit.

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) \times 25

Axis Operating Input Memory Areas (Monitoring)

Name	Word	Bits	Contents
Torque Limit Status Flag	b+1	09	1: Torque limit in progress

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

When using W-series Servo Drives, the set values for Servo Parameters *Forward Rotation External Current Limit* (Pn404) and *Reverse Rotation External Current Limit* (Pn405) can be overwritten at any time to change the torque limit value while the torque limit function is enabled.

When using G-series Servo Drives, the set values for Servo Parameters *No.1 Torque Limit* (Pn05E) and *No.2 Torque Limit* (Pn05F) can be overwritten at any time to change the torque limit value.

During ABSOLUTE MOVEMENT, RELATIVE MOVEMENT (except for INTER-RUPT FEEDING), SPEED CONTROL, and TORQUE CONTROL execution, the Current Limit Designation Bit setting can be changed while the axis is active by turning ON the command's start bit again.

For ORIGIN SEARCH, ORIGIN RETURN, and JOG, torque limits can be specified only at the start of operation.

Note When executing movement commands sequentially, make sure that the movement command bit remains OFF for a minimum of either the PLC cycle time \times 2 or the MECHATROLINK communications cycle \times 2, whichever is longer. If the time that the movement command bit is OFF is too short, the

PCU will not be able to detect the rising edge of the movement command bit, preventing reception of the command.

Timing Charts

Example: Using Torque Limit during Position Control Execution

The following timing chart is for when RELATIVE MOVEMENT is executed. The timing chart for ABSOLUTE MOVEMENT is the same, except that positioning is executed with the absolute position.



a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) \times 25

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. $-1) \times 25$

Example: Enabling Torque Limit during Positioning



a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) \times 25

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) \times 25

Positioning that Does Not Reach Target Position (Example: Pushing a Load)



a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) × 25

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

Note

In applications such as pushing a load, when using the torque limit function to limit the output torque of the Servomotor and stop the machine, the command present position reaches the target position while the feedback present position does not, which causes the pulses to accumulate for the position deviation in the Servo Drive's deviation counter. To change movement to the opposite direction, send the operating command with the Current Limit Designation Bit setting enabled in the pushing direction.

If the Current Limit Designation Bit setting is disabled for the direction in which the load is being pushed when operation is started in the opposite direction (i.e., an operating command is sent without specifying the torque limit), torque will be generated in the pushing direction at the start of operation. This is due to the position deviation accumulated in the pushing direction and may result in damage to the machine or an accident. When performing an operation to return from the pushing position, the Servomotor will remain stopped for the movement command while the position deviation accumulated due to pushing is distributed. When the difference between the pushing operation target position and pushing stop position is large (the accumulated position deviation is large), the Servomotor will immediately accelerate to the target speed after the accumulated position deviation is distributed.

To prevent this from occurring, set the positioning target position as close to the pushing stop position as possible to minimize the accumulated position deviation during pushing. (See below.)



Alternatively, before performing the return operation, send a movement command to move the axis to the pushing stop position (feedback present position at pushing stop) and distribute the accumulated position deviation beforehand, as shown in the following diagram.

Speed Control



Note

The DEVIATION COUNTER RESET can be used to reset the remaining position deviation to 0 for PCUs with unit version 1.3 or later. Refer to *10-10 DEVI-ATION COUNTER RESET* for details on DEVIATION COUNTER RESET.

10-5 Speed Control

10-5-1 Overview

In Speed control, the PCU controls the rotation speed of the Servomotor by directly executing speed commands in a speed loop without using the Servo Drive's position loop.

Speed control can be used when a G5-series Servo Drive, a G-series Servo Drive, or a W-series Servo Drive is connected.

With the SMARTSTEP Junior Servo Drives, speed commands cannot be given directly for a speed loop. When using a SMARTSTEP Junior Servo Drive, position control can be performed while changing the speed using jog-ging or direct operation to achieve feeding at desired speeds.

10-5-2 Starting Speed Control

Speed control is performed by setting the target speed in the speed command value for speed control in the Axis Operating Output Memory Areas and started when the SPEED CONTROL Bit turns ON. When starting speed control, make sure that the SPEED CONTROL Bit remains ON until the Receiving Command Flag or Busy Flag in the Axis Operating Input Memory Area turns ON.

If the SERVO UNLOCK, DEVIATION COUNTER RESET, EMERGENCY STOP, OR DECELERATION STOP command bit is turned ON, the SPEED CONTROL command will be ignored. Be sure that all of these command bits are OFF before executing speed control.

The speed command value for speed control is set in units of 0.001% as a ratio of the Servomotor's momentary maximum rotation speed. (The unit is different from that used for the speed command value for position control (command units/s).) The speed command value is specified in the range -199.999% to 199.999%, and the rotation direction is determined by the sign (positive or negative) of the speed command value.

The actual Servomotor speed control range and minimum control unit depend on the Servo Drive specifications.

To stop speed control, execute DECELERATION STOP or EMERGENCY STOP. (For details on stop functions, refer to *10-9 Stop Functions*.)

Servo Parameter Area G5-series Servo Drive Speed Control Parameters

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn521	Torque limit selection		0 to 6	2	1
	Pn436	Rotation speed for motor rotation detection	r/min	10 to 20000	2	1000
	Pn435	Speed conformity detection range	r/min	10 to 20000	2	50

Pn003 is used to select the function of the option command value for speed control.

The option command value for speed control is described on the following pages.

Pn436 is set as the detection threshold of the Zero Speed Flag (word b+1, bit 08) in the Axis Operating Input Memory Area.

Pn435 is set as the detection width of the Speed Conformity Flag (word b+1, bit 07) in the Axis Operating Input Memory Area.

Servo Parameter Area

G-series Servo Drive Speed Control Parameters

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn003	Torque limit selection		1 to 5	2	1
	Pn062	Rotation speed for motor rotation detection	r/min	10 to 20000	2	50
	Pn061	Speed conformity signal output width	r/min	10 to 20000	2	20

Pn003 is used to select the function of the option command value for speed control.

The option command value for speed control is described on the following pages.

Pn062 is set as the detection threshold of the Zero Speed Flag (word b+1, bit 08) in the Axis Operating Input Memory Area. Pn061 is set as the detection width of the Speed Conformity Flag (word b+1, bit 07) in the Axis Operating Input Memory Area.

W-series Servo Drive Speed Control Parameters

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn002.0	Function selection application switch 2 Torque command input change		0 to 3	2	0
	Pn502	Rotation speed for motor rotation detection	r/min	1 to 10000	2	20
	Pn503	Speed conformity signal output width	r/min	0 to 100	2	10

Pn002.0 is used to select the function of the option command value for speed control. The option command value for speed control is described in the following pages.

The data length for Pn002.0 is the set value specified to transfer Pn002 (including Pn002.0) when transferring Servo Parameters.

Pn502 is set as the detection threshold of the Zero Speed Flag (word b+1, bit 08) in the Axis Operating Input Memory Area. Pn503 is set as the detection width of the Speed Conformity Flag (word b+1, bit 07) in the Axis Operating Input Memory Area.

Name	Word	Bits	Contents
DECELERATION STOP	а	15	$0 \rightarrow 1$: Starts deceleration stop.
SPEED CONTROL	a+1	02	$0 \rightarrow 1$: Starts speed control.
EMERGENCY STOP	1	15	$0 \rightarrow 1$: Starts emergency stop.
Speed command value	a+6		Speed command value (rightmost word)
(for speed control)	a+7		Speed command value (leftmost word)
			Unit: 0.001% (percentage of Servomotor's momentary maximum rotation speed)
			Command range: -199.999% to 199.999% (FFFCF2C1 to 00030D3F hex)
			Note The rotation direction is determined by the sign.
Option command value 1	a+10		Torque limit/torque feed forward (rightmost word)
	a+11		Torque limit/torque feed forward (leftmost word)
			Unit: %
			(percentage of Servomotor's momentary maximum torque)
			Command range: 0% to 399% (See note.) (00000000 to 0000018F hex)
			When using G5-series Servo Drives, the option command value 1 can be used as the torque limit command value or torque feed forward command value by setting the Servo Parameter <i>Torque Limit Selection</i> (Pn521).
			When using G-series Servo Drives, the option command value 1 can be used as the torque limit command value or torque feed forward command value by setting the Servo Parameter <i>Torque Limit Selection</i> (Pn003).
			When using W-series Servo Drives, the option command value 1 can be used as the torque limit command value or torque feed forward command value by setting <i>Torque Command Input Change</i> in Servo Parameter <i>Func-</i> <i>tion Selection Application Switch 2</i> (Pn002.0).
Option command value 2	a+12		Torque limit (rightmost word)
	a+13		Torque limit (leftmost word)
			Unit: %
			(percentage of Servomotor's momentary maximum torque)
			Command range: 0 to 399% (See note.) (00000000 to 0000018F hex)
			When using G5-series Servo Drives, the option command value 2 can be used as the torque limit command value or torque feed forward command value by setting the Servo Parameter <i>Torque Limit Selection</i> (Pn521).
			When using G-series Servo Drives, the option command value 2 can be used as the torque limit command value or torque feed forward command value by setting the Servo Parameter <i>Torque Limit Selection</i> (Pn003).
			When using W-series Servo Drives, the option command value 2 can be used as the torque limit command value by setting <i>Torque Command Input</i> <i>Change</i> in Servo Parameter <i>Function Selection Application Switch 2</i> (Pn002.0).

Axis Operating Output Memory Areas (Operating Commands)

Name	Word	Bits	Contents	
Forward rotation current limit designation	a+16	14	1: Use forward torque limit.	
Reverse rotation current limit designation		15	1: Use reverse torque limit.	
		a = Beg Parame	ginning word of Axis Operating Output Areas specified in Common ters + (Axis No. – 1) $\times25$	
	Note	The torce ified be dependent tions are R88D-W	que limit for option command values during speed control can be spec- tween 0 and 399 for the PCU, but the effective command value s on Servo Drive specifications. The following command range restric- e for W-series Servo Drives. $VT\Box$ + FNY-NS115	
		Command range: 0% to 199% (00000000 to 000000C7 hex) Torque limits will not be set correctly for command values betwee ad 399%. Use values between 0% and 199%.		
		R88D-WN□-ML2 (built-in MECHATROLINK-II Communications) Command range: 0% to 199% (00000000 to 000000C7 hex) Torque limits will not be set correctly for command values between ad 399%. Use values between 0% and 199%.		
		A command range check (0 to 399) is performed for the option convalue for speed control when the speed control command is given option command value is out of range, an option command value error code: 3064 or 3065) will occur. The command range check will be per even if the Speed Command Input Change parameter in the Function tion Application Switches 2 is set to disable using the option command value and value to 0 when it is not being used		
I	Note	The sam torque of however sure that rent type	me option command value parameter is used for both speed and control. The allowable setting ranges for speed and torque control, r, are different. When switching between speed and torque control, be tt the option command value is set within the proper range for the cur- e of control.	
		Change always value fo speed fo	s to the speed command value for speed control during operation are effective. During speed control, by setting a new speed command r speed control in the Axis Operating Output Memory Area, the target or speed control can be changed.	
		The settings for option command values 1 and 2 (torque limit/torque fe ward command values) and Forward/Reverse Rotation Current Limit De tion Bits are enabled when the SPEED CONTROL Bit turns ON. Ch these command values while speed control is being executed and the ing ON the SPEED CONTROL Bit again enables the option command and Current Limit Designation Bits to be changed during operation.		
		It is nec from the stop. Fo	t is necessary to pay attention to the change of output torque when switc rom the speed control mode to the position control mode with decelera stop. For details, refer to <i>10-5-3 Switching Control Mode</i> .	
	Note	When e moveme time × 2 longer. PCU will preventi	executing movement commands sequentially, make sure that the ent command bit remains OFF for a minimum of either the PLC cycle 2 or the MECHATROLINK communications cycle \times 2, whichever is If the time that the movement command bit is OFF is too short, the I not be able to detect the rising edge of the movement command bit, ng reception of the command.	

Axis Operating Input Memory Areas (Monitoring)

Name	Word	Bits	Contents
Receiving Command	b	00	0: Command reception enabled.
Flag			$0 \rightarrow 1$: Command reception started.
			1: Receiving command (command reception disabled).
Error Flag		12	0: No axis error.
			1: Axis error has occurred.
Busy Flag		13	1: Axis busy (axis operation executing).
Speed Conformity Flag	b+1	07	1: Speed matches the speed command value for speed control.
Zero Speed Flag		08	1: Detecting zero speed.
Torque Limit Status Flag		09	1: Torque limit in progress.
Feedback present posi- tion	b+6 b+7		Present position Feedback position (rightmost word) Feedback position (leftmost word)
Command present posi- tion	b+8 b+9		Present position Command position (rightmost word) Command position (leftmost word)

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

The Receiving Command Flag turns ON for at least one cycle time when the command to start speed control is received by the PCU. Use the Receiving Command Flag when starting and changing option commands to control the ON/OFF timing for SPEED CONTROL.

During speed control, bits 07 and 08 in the Servo Status Flags (word b+1) of the Axis Operating Input Memory Area function as the Speed Conformity Flag and Zero Speed Flag respectively.

The command present position that is output during speed control is calculated from the feedback position and the position deviation that is inferred from the current speed.

Acceleration and Deceleration during Speed Control

When SPEED CONTROL is executed, the Servo Drive performs stepped speed control from the present feedback speed to the target speed specified in the speed command value for speed control. Apart from smoothly increasing/decreasing the speed command value (for speed control) when starting speed control, shock in acceleration/deceleration when performing speed control can also be minimized by setting the Servo Parameters *Soft Start Acceleration Time* and *Soft Start Deceleration Time* to create a trapezoidal speed curve.

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn312	Soft start acceleration time	ms	0 to 10000	2	0
	Pn313	Soft start deceleration time	ms	0 to 10000	2	0

G5-series Acceleration/Deceleration Constants for Speed Control

G-series Acceleration/Deceleration Constants for Speed Control

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn058	Soft start acceleration time	ms	0 to 5000	2	0
	Pn059	Soft start deceleration time	ms	0 to 5000	2	0

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn305	Soft start acceleration time	ms	0 to 10000	2	0
	Pn306	Soft start deceleration time	ms	0 to 10000	2	0

W-series Acceleration/Deceleration Constants for Speed Control

The Soft Start Acceleration Time and Soft Start Deceleration Time are set respectively as the acceleration time and deceleration time between speed 0 (speed command value for speed control: 0%) and the momentary maximum rotation speed of the Servomotor (speed command value for speed control: 100%).

The acceleration time and deceleration time for the actual target speed are as follows:

Speed command value for speed control (r/min) × Soft start acceleration (deceleration) time Actual acceleration (deceleration) time = Momentary maximum rotation speed (r/min)



Note

The momentary maximum rotation speed depends on the Servomotor used. Refer to the momentary maximum rotation speed for the Servomotor used. The following values apply to W-series Servomotors.

3,000-r/min Servomotor (cylinder type or flat type): 5,000 r/min

- 1,000-r/min Servomotor: 2,000 r/min
- 1,500-r/min Servomotor (450 W to 7.5 kW): 3,000 r/min
- 1,500-r/min Servomotor (11 to 15 kW): 2,000 r/min

Option Command Value during Speed Control

When speed control is used, the torque limit and torque feed forward functions can be applied during speed control by setting Servo Parameter Torque Limit Selection (Pn521) when using G5-series Servo Drives, Servo Parameter Tolque Limit Selection (Pn003) when using G-series Servo Drives or Servo Parameter Torque Command Input Change of Function Selection Application Switch 2 (Pn002.0) when using W-series Servo Drives and setting the Current Limit Designation Bits and option command values in the Axis Operating Output Memory Area.

When using G5-series Servo Drives, the torque limit/torque feed forward function depends on the Servo Parameter and output bit combinations, as follows:

Torque Limit Selection (Pn521)	PCL	NCL	R88D-GN□-ML2 (with built-in MECHATROLINK-II communications)
1			Option command value 1 (words a+10, a+11) functions as a torque feed forward command value when the torque feed forward function is enabled. Pn013 functions as a forward torque limit value/reverse torque limit value when the torque limit function is enabled.
2			Option command value 1 (words $a+10$, $a+11$) functions as a torque feed forward command value when the torque feed forward function is enabled. Pn013 functions as a forward torque limit value, and Pn522 functions as a reverse torque limit value when the torque limit function is enabled.
3			Option command value 1 (words a+10, a+11) functions as a torque feed forward command value when the torque feed forward function is enabled.
	OFF	OFF	Pn013 functions as a forward torque limit value/reverse torque limit value when the torque limit function is enabled.
	OFF	ON	Pn013 functions as a forward torque limit value, and Pn522 functions as a reverse torque limit value when the torque limit function is enabled.
	ON	OFF	Pn522 functions as a forward torque limit value, and Pn013 functions as a reverse torque limit value when the torque limit function is enabled.
	ON	ON	Pn522 functions as a forward torque limit value/reverse torque limit value when the torque limit function is enabled.
4			The torque feed forward function is disabled. Whichever is smaller of the Servo Parameter Pn013 and option command value 1 (word $a+10$, $a+11$) functions as a forward torque limit value, and whichever is smaller of the Servo Parameter Pn522 and option command 2 (word $a+12$, $a+13$) functions as a reverse torque limit value when the torque limit function is enabled.
5			The torque feed forward function is disabled.
	OFF	OFF	Pn013 functions as a forward torque limit value when the torque limit function is enabled. Pn522 functions as a reverse torque limit value when the torque limit function is enabled.
	OFF	ON	Pn013 functions as a forward torque limit value when the torque limit function is enabled. Whichever is smaller of the Servo Parameter Pn522 and option command value 2 (word a+12, a+13) functions as a reverse torque limit value when the torque limit function is enabled.
	ON	OFF	Whichever is smaller of the Servo Parameter Pn013 and option com- mand value 1 (word a+10, a+11) functions as a forward torque limit value when the torque limit function is enabled. Pn522 functions as a reverse torque limit value when the torque limit function is enabled.
	ON	ON	Whichever is smaller of the Servo Parameter Pn013 and option com- mand value 1 (word $a+10$, $a+11$) functions as a forward torque limit value when the torque limit function is enabled. Whichever is smaller of the Servo Parameter Pn522 and option com- mand value 2 (word $a+12$, $a+13$) functions as a reverse torque limit value when the torque limit function is enabled.
6			Option command value 1 (words $a+10$, $a+11$) functions as a torque feed forward command value when the torque feed forward function is enabled.
	OFF	OFF	Pn013 functions as a forward torque limit value, and Pn522 functions as a reverse torque limit value when the torque limit function is enabled.
	OFF	ON	Pn013 functions as a forward torque limit value, and Pn526 functions as a reverse torque limit value when the torque limit function is enabled.

Torque Limit Selection (Pn521)	PCL	NCL	R88D-GN⊡-ML2 (with built-in MECHATROLINK-II communications)
	ON	OFF	Pn525 functions as a forward torque limit value, and Pn522 functions as a reverse torque limit value when the torque limit function is enabled.
	ON	ON	Pn525 functions as a forward torque limit value, and Pn526 functions as a reverse torque limit value when the torque limit function is enabled.

Note PCL ON: When either the external signal Forward Torque Limit or the Forward Rotation Current Designation (a+1614) is ON.

NCL ON: When either the external signal reverse torque limit or the Forward Rotation Current Designation (a+1615) is ON.

When using G-series Servo Drives, the torque limit/torque feed forward function depends on the Servo Parameter and output bit combinations, as follows:

Torque Limit Selection (Pn003)	PCL	NCL	R88D-GN□-ML2 (with built-in MECHATROLINK-II communications)
1			Option command value 1 (words a+10, a+11) functions as a torque feed forward command value and the torque feed forward function is enabled. Pn05E functions as a forward torque limit value/reverse torque limit value and the torque limit function is enabled.
2			Option command value 1 (words a+10, a+11) functions as a torque feed forward command value and the torque feed forward function is enabled. Pn05E functions as a forward torque limit value, Pn05F functions as a reverse torque limit value and the torque limit function is enabled.
3			Option command value 1 (words a+10, a+11) functions as a torque feed forward command value and the torque feed forward function is enabled.
	OFF	OFF	Pn05E functions as a forward torque limit value/reverse torque limit value and the torque limit function is enabled.
	OFF	ON	Pn05E functions as a forward torque limit value, Pn05F functions as a reverse torque limit value and the torque limit function is enabled.
	ON	OFF	Pn05F functions as a forward torque limit value, Pn05E functions as a reverse torque limit value and the torque limit function is enabled.
	ON	ON	Pn05F functions as a forward torque limit value/reverse torque limit value and the torque limit function is enabled.
4			The torque feed forward function is disabled. Whichever is smaller of the Servo Parameter Pn05E and option command value 1 (word $a+10$, $a+11$) functions as a forward torque limit value, whichever is smaller of the Servo Parameter Pn05F and option command 2 (word $a+12$, $a+13$) functions as a reverse torque limit value and the torque limit function is enabled.

Torque Limit Selection (Pn003)	PCL	NCL	R88D-GN□-ML2 (with built-in MECHATROLINK-II communications)
5			The torque feed forward function is disabled.
	OFF	OFF	Pn05E functions as a forward torque limit value and the torque limit func- tion is enabled. Pn05F functions as a reverse torque limit value and the torque limit func- tion is enabled.
	OFF	ON	Pn05E functions as a forward torque limit value and the torque limit function is enabled. Whichever is smaller of the Servo Parameter Pn05F and option command value 2 (word $a+12$, $a+13$) functions as a reverse torque limit value and the torque limit function is enabled.
	ON	OFF	Whichever is smaller of the Servo Parameter Pn05E and option com- mand value 1 (word a+10, a+11) functions as a forward torque limit value and the torque limit function is enabled. Pn05F functions as a reverse torque limit value and the torque limit func- tion is enabled.
	ON	ON	Whichever is smaller of the Servo Parameter Pn05E and option com- mand value 1 (word $a+10$, $a+11$) functions as a forward torque limit value and the torque limit function is enabled. Whichever is smaller of the Servo Parameter Pn05F and option com- mand value 2 (word $a+12$, $a+13$) functions as a reverse torque limit value and the torque limit function is enabled.

Note PCL ON: When either the Forward Torque Limit (CN1 PCL: Pin 7) or Forward Rotation Current Limit Designation (word a+1614) is turned ON.

NCL ON: When either the Reverse Torque Limit (CN1 NCL: Pin 8) or Reverse Rotation Current Limit Designation (word a+1615) is turned ON.

When an R88D-WT W-series Servo Drive equipped with a FNY-NS115 MECHATROLINK-II I/F Unit or an R88D-WN -ML2 W-series Servo Drive with built-in MECHATROLINK-II communications is connected to the PCU, the torque limit/torque feed forward function depends on the Servo Parameter and output bit combinations, as follows:

Torque com- mand input change (Pn002.0)	Forward rota- tion current limit designation (word a+16, bit 14	Reverse rota- tion current limit designation (word a+16, bit 15)	R88D-WT⊡ + FNY-NS115	R88D-WN⊡-ML2 with built-in MECHATROLINK-II commu- nications
0			Option command values 1 and 2 are disabled. Use set value 0.	Option command values 1 and 2 are disabled. Use set value 0.
1			Option command value 1 (words a+10, a+11) functions as a torque limit value and the torque limit function is enabled in both directions.	Option command value 1 (words a+10, a+11) functions as a forward torque limit value, option command value 2 (words a+12, a+13) functions as a reverse torque limit value and the torque limit function is enabled.
2			Option command value 1 (words a+10, a+11) functions as a torque feed forward com- mand value and torque feed forward is enabled.	Option command value 1 (words a+10, a+11) functions as a torque feed forward com- mand value and torque feed forward is enabled.

Speed Control

Section 10-5

Torque com- mand input change (Pn002.0)	Forward rota- tion current limit designation (word a+16, bit 14	Reverse rota- tion current limit designation (word a+16, bit 15)	R88D-WT□ + FNY-NS115	R88D-WN□-ML2 with built-in MECHATROLINK-II commu- nications
3	0	0	Option command values 1 and 2 are disabled. Use set value 0.	Option command values 1 and 2 are disabled. Use set value 0.
	0	1	Option command value 2 (words a+12, a+13) functions as a reverse torque limit value and the torque limit function is enabled.	Option command value 2 (words a+12, a+13) functions as a reverse torque limit value and the torque limit function is enabled.
	1	0	Option command value 1 (words a+10, a+11) functions as a forward torque limit value and the torque limit function is enabled.	Option command value 1 (words a+10, a+11) functions as a forward torque limit value and the torque limit function is enabled.
	1	1	Option command value 1 (words a+10, a+11) functions as a torque limit value and the torque limit function is enabled in both directions.	Option command value 1 (words a+10, a+11) functions as a forward torque limit value, option command value 2 (words a+12, a+13) functions as a reverse torque limit value and the torque limit function is enabled.

The settings for option command values 1 and 2 (torque limit/torque feed forward command values) and Forward/Reverse Rotation Current Limit Designation Bits are enabled when the SPEED CONTROL Bit turns ON. Changing these command values while speed control is being executed and then turning ON the SPEED CONTROL Bit again enables the option command values and Current Limit Designation Bits to be changed during operation.

The torque limit for speed control set in the option command values can be used together with the torque limit function (refer to *10-4 Torque Limits*). When enabling multiple torque limit functions simultaneously, the Servomotor's output torque is limited by the minimum torque limit setting.

Speed Control

Example Timing Chart for W-series Speed Control



a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) \times 25 b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. –1) \times 25

When the speed (feedback speed, command speed, or target speed) is selected for monitor 1 or 2, the monitoring unit for position control/torque control, and speed control will change. The speed command value for speed control (unit: 0.001%) is used from when the speed control command is received (when the Receiving Command Flag turns OFF) and the speed command value for position control (command units/s) is used from when the command to switch to position control or torque control is received.

Status and Monitoring during Speed Control

Servo Status Flags in Axis Operating Input Memory Area (Word b+1) When SPEED CONTROL is executed, some of the Status Flags, monitoring functions, and display units change.

The functions of bits 07 and 08 in the Servo Status Flags of the Axis Operating Input Memory Area change depending on whether position control or speed control is used.

Word	Bits	Control mode	Flag name	Function	Function change timing
b+1	07	Position control	Positioning Com- pleted (PSET) Flag	Sending the position control com- mand to the Servo Drive com- pleted and the position is within <i>Positioning Completion Range 1</i> (in position).	The function changes when the command to switch to position control has been received (when the Receiving Command Flag turns OFF).
		Speed con- trol	Speed Conformity (V-CMP) Flag	Speed matches the speed com- mand value for speed control.	The function changes when SPEED CONTROL has been received (when the Receiving Command Flag turns OFF).
	08	Position control	Distribution Com- pleted (DEN) Flag	Sending the position control com- mand to the Servo Drive com- pleted (operating for the number of pulses remaining in the deviation counter).	The function changes when the command to switch to position control has been received (when the Receiving Command Flag turns OFF).
		Speed con- trol	Zero Speed (ZSPD) Flag	Speed control with zero speed in progress.	The function changes when SPEED CONTROL has been received (when the Receiving Command Flag turns OFF).

Monitors 1 and 2 in Axis Operating Input Memory Areas When the speed (feedback speed, command speed, or target speed) is selected for monitor 1 or 2, the monitoring unit for position control/torque control, and speed control will change, as follows:

Control mode	Speed monitoring unit	Unit change timing
Position con- trol/Torque control	Command units/s	The unit changes when the command to switch to position control/torque con- trol has been received (when the Receiving Command Flag turns OFF).
Speed control	0.001% (percentage of Ser- vomotor's momen- tary maximum rotation speed)	The unit changes when SPEED CON- TROL has been received (when the Receiving Command Flag turns OFF).

10-5-3 Switching Control Mode

The PCU's operating commands can be used to switch between position control, speed control, and torque control of active axes, as follows:

Operating command	Control mode	Operation when operating command is executed
ABSOLUTE MOVEMENT	Position control	Switches to position control from any control mode when the command starts and executes positioning with an absolute position.
RELATIVE MOVEMENT	Position control	Switches to position control from any control mode when the command starts and executes positioning with a relative position.
ORIGIN SEARCH	Position control	A multistart error occurs when the command starts from any control mode, and deceleration stop is executed in position control mode.
ORIGIN RETURN	Position control	A multistart error occurs when the command starts from any control mode, and deceleration stop is executed in position control mode.

Operating command	Control mode	Operation when operating command is executed
JOG	Position control	A multistart error occurs when the command starts from any control mode, and deceleration stop is executed in position control mode.
DECELERA- TION STOP	Position control	Switches to position control when the command starts from any control mode, and deceleration stop is executed.
SPEED CON- TROL	Speed control	Switches to speed control when the command starts from any control mode.
TORQUE CON- TROL	Torque control	Switches to torque control when the command starts from any control mode.
EMERGENCY STOP	Position control	Switches to position control when the command starts from any control mode, and emergency stop is executed.
SERVO UNLOCK	Position control	When executing SERVO UNLOCK, the PCU executes DECELERATION STOP for the Servo Drive, followed immediately (without waiting for the axis to stop) by SERVO UNLOCK. When SERVO UNLOCK is executed during speed con- trol or torque control, the Servo Drive will recover in the position control mode's Servo lock status the next time the SERVO LOCK com- mand is executed.
DEVICE SETUP	Position control	Switches to position control when the command starts, and recovers in position control mode when SERVO LOCK is executed.

If operation stops due to an error (except stopping that puts Servo in free run state), the Servomotor stops in position control mode (deceleration stop or emergency stop) and remains in position control mode (Servo lock status) after stopping. After turning ON the power to the PCU or restarting the Unit and executing CONNECT or SERVO LOCK, the PCU will be put in the position control mode and in Servo lock status.

Note When SERVO UNLOCK is executed during speed control or torque control, the Servo Drive will recover in the position control mode and in Servo lock status the next time the SERVO LOCK command is executed. If MECHATROLINK communications are stopped (connection released) or the power to the PCU is interrupted during speed control or torque control, the Servo Drive will be put in Servo free run state in the previous control mode. While in this state, if MECHATROLINK communications are started (connection established) and SERVO LOCK is executed, the Servomotor starts operating immediately in the previous command state. The speed monitor and torque monitor in the PCU's expanded monitoring functions will also not function properly. To stop MECHATROLINK communications while in speed control or torque control mode, use DECELERATION STOP or another command that will switch the Servo Drive to position control, and execute the next SERVO LOCK command in position control mode. If the Servo is in a free run state during speed control or torque control (due to an error or other cause), execute DEVICE SETUP before the next SERVO LOCK and switch the Servo Drive to position control mode.

When switching to position control from speed control or torque control, the feedback speed for executing operating commands in position control is used as the start speed. Movement to the position control target speed or stop operation is then performed using the acceleration/deceleration set in the Servo Parameters for acceleration/deceleration.

Changes in Torque when Switching Control Modes	When switching to another control mode during torque control, position con- trol with the torque limit enabled, or speed control, the torque in the control mode to be switched to must be considered. For example, if DECELERATION STOP is executed with the torque limit disabled during speed control with the torque limit enabled, position control will be used without the output torque limit during the time deceleration stop is executed. When switching from the control mode that requires a constant torque limit, such as for pushing or pulling a load, make sure that the torque limit is also enabled for the next operating command to be executed to prevent a sudden change in output torque.
Note	In speed control mode, axis stop operations due to errors are executed after switching to position control. The torque limit for these stop operations depends on the setting of the Current Limit Designation Bit at when the error occurred. When executing speed control with the torque limit enabled using the option command value, if the Current Limit Designation Bit is not set to ON, the axis stop operation due to error will be executed with the torque limit disabled. To enable the torque limit for axis stop operations due to errors, turn ON the Current Limit Designation Bit and set Servo Parameters <i>No.1 Torque Limit</i> (Pn013) or <i>No.2 Torque Limit</i> (Pn522) when using G5-series Servo Drives, Servo Parameters <i>No.1 Torque Limit</i> (Pn05F) when using G-series Servo Drives, and <i>Forward Rotation External Current Limit</i> (Pn404) or <i>Reverse Rotation External Current Limit</i> (Pn405) when using W-series Servo Drives.

10-6 Torque Control

10-6-1 Overview

In torque control, the PCU controls the output torque of the Servomotor by directly executing torque commands in a torque (current) loop without using the Servo Drive's position loop or speed loop.

Torque control can be used when a G5-series Servo Drive, a G-series Servo Drive or a W-series Servo Drive is connected. The SMARTSTEP Junior Servo Drives do not support torque control.

10-6-2 Starting Torque Control

Torque control is performed according to the target torque set in the torque command value in the Axis Operating Output Memory Areas and started when the TORQUE CONTROL Bit turns ON. When starting torque control, make sure that the TORQUE CONTROL Bit remains ON until the Receiving Command Flag or Busy Flag in the Axis Operating Input Memory Area turns ON.

If the SERVO UNLOCK, DEVIATION COUNTER RESET, EMERGENCY STOP, OR DECELERATION STOP command bit is turned ON, the TORQUE CONTROL command will be ignored. Be sure that all of these command bits are OFF before executing torque control.

The torque command value is set in units of 0.001% as a percentage of the Servomotor's momentary maximum torque. The torque command value is specified in the range –199.999% to 199.999%, and the rotation direction is determined by the sign (positive or negative) of the torque command value. The actual Servomotor torque control range and minimum control unit depend on the Servo Drive specifications.

To stop torque control, execute DECELERATION STOP or EMERGENCY STOP. (For details on stop functions, refer to *10-9 Stop Functions*.)

Servo Parameter Area

G5-series Torque Control Parameters

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn317	Speed limit selection		0, 1	2	0
	Pn321	Speed limit value setting	r/min	0 to 20000	2	50

Pn317 is used to select the function of the option command value for torque control.

The option command value for torque control is described in the following pages.

Pn321 is used to set the fixed speed limit for torque control.

G-series Torque Control Parameters

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn05B	Speed limit selection		0, 1	2	0
	Pn053	Speed limit	r/min	-20000 to 20000	2	50

Pn05B is used to select the function of the option command value for torque control.

The option command value for torque control is described in the following pages.

Pn053 is used to set the fixed speed limit for torque control.

W-series Torque Control Parameters

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn002.1	Function selection application switch 2 Speed command input switching		0, 1	2	0
	Pn407	Speed limit	r/min	0 to 10000	2	3000

Note The default settings correspond to those used for W-series Servo Drives.

Pn002.1 is used to select the function of the option command value for torque control. The option command value for torque control is described in the following pages.

The data length for Pn002.1 is the set value specified to transfer Pn002 (including Pn002.1) when transferring Servo Parameters. Pn407 is used to set the fixed speed limit for torque control.

Axis Operating Output Memory Areas (Operating Commands)

Name	Word	Bits	Contents
DECELERATION STOP	а	15	$0 \rightarrow 1$: Starts deceleration stop.
TORQUE CONTROL	a+1	03	$0 \rightarrow 1$: Starts torque control.
EMERGENCY STOP		15	$0 \rightarrow 1$: Starts emergency stop.

Name	Word	Bits	Contents
Torque command value	a+8		Torque command value (rightmost word)
	a+9		Torque command value (leftmost word)
			Unit: 0.001% (percentage of Servomotor's momentary maximum torque)
			Command range: -199.999 to 199.999% (FFFCF2C1 hex to 00030D3F hex)
			Note The rotation direction is determined by the sign.
Option command value 1	a+10		Speed limit value (rightmost word)
	a+11		Speed limit value (leftmost word)
			Unit: 0.001% (percentage of Servomotor's momentary maximum rotation speed)
			Command range: 0 to 100.000% (See note 1.) (00000000 hex to 000186A0 hex)
			The option command value 1 can be used as the speed limit value during torque control by setting the Servo Parameter <i>Speed Limit Selection</i> (Pn317) when using G5-series Servo Drives, <i>Speed Limit Selection</i> (Pn05B) when using G-series Servo Drives, and the <i>Speed Command Input Change</i> in Servo Parameter <i>Function Selection Application Switch</i> 1 (Pn002.1) when using W-series Servo Drives.
Forward rotation current limit designation	a+16	14	1: Use forward torque limit.
Reverse rotation current limit designation	Ĭ	15	1: Use reverse torque limit.

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

Note

- (1) A command range check (0 to 10,000) is performed for the option command value for torque control when the torque control command is given. If the option command value is out of range, an option command value error (error code: 3064) will occur. The command range check will be performed even if the Servo Parameter *Speed Limit Selection* (Pn317) when using G5-series Servo Drives, Servo Parameter *Speed Limit Selection* (Pn05B) when using G-series Servo Drives, or the Torque Command Input Change parameter in the Function Selection Application Switches 2 when using W-series Servo Drives is set to disable using the option command value. Always set the option command value to 0 when it is not being used.
 - (2) The same option command value parameter is used for both speed and torque control. The allowable setting ranges for speed and torque control, however, are different. When switching between speed and torque control, be sure that the option command value is set within the proper range for the current type of control.

Changes to the torque command value during operation are always effective. during torque control, by setting a new torque command value in the Axis Operating Output Memory Area, the target torque for torque control can be changed.

The data for option command value 1 (speed limit value) is enabled when the start bit for TORQUE CONTROL turns ON. Changing these set values while torque control is being executed and then turning ON the TORQUE CON-TROL Bit again enables the speed limit value to be changed during operation.

The torque limit can also be applied by turning ON the Forward/Reverse Rotation Current Limit Designation Bit when executing TORQUE CONTROL. (For details, refer to *10-4-3 Torque Limits Set by Operating Commands.*)

Axis Operating Input Memory Areas (Monitoring)

Name	Word	Bits	Contents
Receiving Command	b	00	0: Command reception enabled.
Flag			$0 \rightarrow 1$: Command reception started.
			1: Receiving command (command reception disabled).
Error Flag		12	0: No axis error.
			1: Axis error has occurred.
Busy Flag		13	1: Axis busy (axis operation executing).
Torque Limit Status Flag	b+1	09	1: Torque limit in progress.
Speed Limit Status Flag		11	1: Speed limit in progress.
Feedback present posi- tion	b+6 b+7		Present position Feedback position (rightmost word) Feedback position (leftmost word)
Command present posi- tion	b+8 b+9		Present position Command position (rightmost word) Command position (leftmost word)

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

The Receiving Command Flag turns ON for at least one cycle time when the command to start torque control is received by the PCU. Use the Receiving Command Flag when starting and changing option commands to control the ON/OFF timing for TORQUE CONTROL.

During torque control, bit 11 in the Servo Status Flags (word b+1) of the Axis Operating Input Memory Area functions as the Speed Limit Status Flag.

The command present position that is output during torque control is calculated from the feedback position and the position deviation that is inferred from the current speed.

Changes in output torque may affect operation when switching to speed control or position control with deceleration stop, while torque control is being performed. For details, refer to *10-5-3 Switching Control Mode*.

Note When executing movement commands sequentially, make sure that the movement command bit remains OFF for a minimum of either the PLC cycle time × 2 or the MECHATROLINK communications cycle × 2, whichever is longer. If the time that the movement command bit is OFF is too short, the PCU will not be able to detect the rising edge of the movement command bit, preventing reception of the command.

Option Command Value during Torque Control

A speed limit can be applied during torque control by setting Servo Parameter *Speed Limit Selection* (Pn317) when using G5-series Servo Drives, Servo Parameter *Speed Limit Selection* (Pn05B) when using G-series Servo Drives, and the *Speed Command Input Change* in Servo Parameter *Function Selection Application Switch 2* (Pn002.1) when using W-series Servo Drives and the option command value.

The speed limit functions depend on the Servo Parameter and output bit combinations, as follows:

G5-series Servo Drives

Speed command input change (Pn317)	Function
0	Option command values 1 and 2 are disabled. Use set value 0.
1	Limits speed during torque control, using option command value 1 (words a+10, a+11) as the speed limit value.
	Option command value 2 is not used. Use set value 0.

G-series Servo Drives

Speed command input change (Pn05B)	Function
0	Option command values 1 and 2 are disabled. Use set value 0.
1	Limits speed during torque control, using option command value 1 (words a+10, a+11) as the speed limit value.
	Option command value 2 is not used. Use set value 0.

W-series Servo Drives

Speed command input change (Pn002.1)	Function	
0	Option command values 1 and 2 are disabled. Use set value 0.	
1	Limits speed during torque control, using option command value 1 (words a+10, a+11) as the speed limit value.	
	Option command value 2 is not used. Use set value 0.	

The data for option command value 1 (speed limit value) is enabled when the bit for TORQUE CONTROL turns ON. Changing option command value 1 while torque control is being executed and then turning ON the TORQUE CONTROL Bit again enables the speed limit value to be changed during operation.

The speed limit for torque control set in the option command value can be used together with a Servo Parameter (Pn321 for the G5 Series, Pn053 for the G Series and Pn407 for the W Series) (speed limit). When both speed limit functions are enabled at the same time, the Servomotor's rotation speed is limited by the lowest speed limit setting.

Note The speed limit functions for torque control are used to limit the Servomotor's rotation speed in a steady state, and not the upper limit (maximum value) of the Servomotor's rotation speed. Transitional overshooting may occur in the Servomotor's rotation speed and the speed limit value may be exceeded due to conditions such as the torque command value, Servomotor load, and inertia.

Example Timing Chart for Torque Control



a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) × 25 b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

For details on switching to torque control from position control or speed control, refer to 10-5-3 Switching Control Mode.

position control or speed control is received (when the

Receiving Command Flag turns OFF).

Note

(1) When SERVO UNLOCK is executed during speed control or torque control, the Servo Drive will recover in the position control mode and in Servo lock status the next time SERVO LOCK is executed. If MECHATROLINK communications are stopped (connection released) or the power to the PCU is interrupted during speed control or torgue control, the Servo Drive will be put in Servo free run state in the previous control mode. While in this state, if MECHATROLINK communications are started (connection established) and SERVO LOCK is executed, the Servomotor starts operating immediately in the previous command state. The speed monitor and torgue monitor in the PCU's expanded monitoring functions will also not function properly. To stop MECHATROLINK communications while in speed control or torgue control mode, use DECELERATION STOP or other command that will switch the Servo Drive to position control, and execute the next SERVO LOCK in position control mode. If the Servo is in a free run state during speed control or torgue control (due to an error or other cause), execute DEVICE SETUP before the next SERVO LOCK and switch the Servo Drive to position control mode.

(2) In torque control mode, axis stop operations due to errors are executed after switching to position control. The output torque for these stop operations depends on the setting for error operation in the Current Limit Designation Bit. during torque control execution, if the Current Limit Designation Bit is not set to ON, the axis stop operation due to error will be executed with the torque limit disabled. To enable the torque limit for axis stop operations due to errors, turn ON the Current Limit Designation Bit while torque control is being executed and set Servo Parameters *No.2 Torque Limit* (Pn05F) when using G-series Servo Drives, and *Forward Rotation External Current Limit* (Pn404) or *Reverse Rotation External Current Limit* (Pn405) when using W-series Servo Drives.

Status and Monitoring during Torque Control

Servo Status Flags in Axis Operating Input Memory Area (Word b+1) When TORQUE CONTROL is executed, some of the Status Flags, monitoring functions, and display units change.

The function of bit 11 in the Servo Status Flags of the Axis Operating Input Memory Area depends on whether position control or torque control is used.

Word	Bits	Control mode	Flag name	Function	Function change timing
b+1	11	Position control	Positioning Proxim- ity (NEAR) Flag	ON when within <i>Positioning Com-</i> pletion Range 2	The function changes when the command to switch to position control has been received (when the Receiving Command Flag turns OFF).
		Torque con- trol	Speed Limit (V_LIM) Status Flag	ON when speed limit in progress using speed limit value specified in option command value 1.	The function changes when TORQUE CONTROL has been received (when the Receiving Command Flag turns OFF).

Monitors 1 and 2 in Axis Operating Input Memory Areas

When the torque is selected for monitor 1 or 2, the monitoring unit for position control/speed control, and torque control changes, as follows:

Control mode	Speed monitoring unit	Unit change timing
Position con- trol/speed control	% (1% unit) (percentage of Ser- vomotor's rated torque)	The unit changes when the command to switch to position control or speed control has been received (when the Receiving Command Flag turns OFF).
Torque control	0.001% (percentage of Ser- vomotor's momen- tary maximum torque)	The unit changes when TORQUE CONTROL has been received (when the Receiving Command Flag turns OFF).
10-7 Backlash Compensation

10-7-1 Overview

"Backlash" is the play between the driving axis and the mechanical system being driven. If there is backlash in positioning from the positive or negative direction, it will cause a discrepancy of the same extent in the positioning. When using a G5-series Servo Drive, a G-series Servo Drive or a W-series Servo Drive, backlash compensation can be used to make this discrepancy as small as possible.



In the above diagram, after positioning from the positive side, backlash is compensated for up to the amount of the output of pulses set for the backlash compensation either when positioning in the negative direction or when the positioning direction is reversed.

The SMARTSTEP Junior Servo Drives do not support backlash compensation.

10-7-2 Backlash Compensation Procedure

The PCU uses the Servo Drive's backlash compensation function. The Servo Parameters for backlash compensation must be set to enable the backlash compensation function. After setting the Servo Parameters, execute SERVO LOCK, and then perform backlash compensation with the following position control operations.

ABSOLUTE MOVEMENT, RELATIVE MOVEMENT, ORIGIN SEARCH, ORIGIN RETURN, and JOG

Backlash compensation is disabled when executing SPEED CONTROL or TORQUE CONTROL. Backlash compensation operation functions as follows:

• After SERVO LOCK (First Position Control Operating Command) Compensation is applied to the first operating command that operates in the specified backlash compensation direction. Compensation is not applied to operations prior to this that moved axes in the opposite direction to backlash. • After Backlash Compensation

Backlash compensation is applied for the first operating command that operates in the opposite direction from which backlash compensation was previously applied. Once backlash compensation has been applied, it will not be applied again as long as operation is continued in the same direction.



Specified backlash compensation direction

Note The PCU's present position compensation according to the backlash compensation function is applied to both the feedback present position and command present position.

10-7-3 Backlash Compensation Data Settings

When a G5-series Servo Drive, a G-series Servo Drive or a W-series Servo Drive is connected, the data settings for backlash compensation are as follows:

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn704	Backlash compensation selec- tion		0 to 2	2	0
	Pn705	Backlash compensation amount	Command unit	-32,768 to 32,767	2	0
	Pn706	Backlash compensation time constant	0.01ms	0 to 6400	2	0

Servo Parameter Area Backlash Compensation Parameters (R88D-KN_-ML2)

The backlash compensation amount is set in Pn705 in command units. The minimum setting unit for the backlash compensation amount, however, depends on the encoder resolution.

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn100	Backlash compensation selec- tion		0 to 2	2	0

Backlash Compensation Parameters (R88D-GN - ML2)

Backlash Compensation

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn101	Backlash compensation	Command unit	-32,768 to 32,767	2	0
	Pn102	Backlash compensation time constant	0.01ms	0 to 6400	2	0

The backlash compensation amount is set in Pn101 in command units. The minimum setting unit for the backlash compensation amount, however, depends on the encoder resolution.

Backlash Comper	nsation Parameters	(R88D-WT +	FNY-NS115)
-----------------	--------------------	------------	------------

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn81B	Backlash compensation amount	0.1 command unit	-32,768 to 32,767	2	0
	Pn81D.0	Compensation function selection		0, 1	2	0

The backlash compensation amount is set in Pn81B in 0.1 command units. The minimum setting unit for the backlash compensation amount, however, depends on the encoder resolution.

Backlash Compensation Parameters (R88D-WN - ML2)

Туре	Parameter No.	Parameter name	Unit	Setting range	Data length	Default setting
	Pn207.2	Backlash compensation selec- tion		0 to 2	2	0
	Pn214	Backlash compensation amount	Command unit	-32,768 to 32,767	2	0
	Pn215	Backlash compensation constant	0.01 ms	0 to 65535	2	0

The backlash compensation function is set in Pn214 in command units. The minimum setting unit for the backlash compensation amount, however, depends on the encoder resolution.

The direction in which backlash compensation is first applied after execution of SERVO LOCK using the backlash compensation function is set in Servo Parameters *Backlash Compensation Selection* (Pn704) when using R88D-KN□-ML2, *Backlash Compensation Selection* (Pn100) when using R88D-GN□-ML2, *Backlash Compensation Direction* (Pn81D.0) when using R88D-WT□, and *Backlash Compensation Selection* (Pn207.2) when using R88D-WN□-ML2, as follows:

R88D-KN -ML2	R88D-GN□ -ML2	R88D-WT⊡ + FNY-NS115	R88D-WN -ML2	Function
Pn704	Pn100	Pn81D.0	Pn207.2	
0	0		0	No backlash compensation
1	1	0	1	Compensates in the forward direction. The first backlash compensation direction is forward.
2	2	1	2	Compensates in the reverse direction. The first backlash compensation direction is forward.

Note When using R88D-WT \square + FNY-NS115 without backlash compensation, set the backlash compensation amount to (Pn81B) to 0.

WWW.BSNEW.IR

10-8 Software Limits

10-8-1 Overview

To prevent or reduce damage to external devices due to unexpected positioning operations when there is a malfunction, in addition to the forward rotation and reverse rotation limit input signals, the PCU also has a function that monitors positioning at a software level. This function is called the "software limit" function.

The range in which the software limit function monitors the position is determined by the values of two settings: the forward software limit and the reverse software limit. Normally, positioning is only possible within the range determined by these software limit settings. The relationship between the forward rotation and reverse rotation limit input signals, the software limits, and the mechanical stoppers used in the system is shown below.



The software limit function is enabled if an origin has been established (i.e., the No Origin Flag is OFF). The software limit function is disabled when the origin has not been established.

When the Servo Drive power is turned ON, the PCU's present position is set to 0, and the origin is not established. RELATIVE MOVEMENT can be executed under these conditions, but the positioning range (present position range possible for positioning) will be -2,147,483,648 to 2,147,483,647 (command units), and the present position will be refreshed by adding the travel distance without an established origin.

If RELATIVE MOVEMENT is executed using a position command value that exceeds the positioning range, a Position Designation Error (error code: 3060) will occur when the command is executed.

When a command such as JOG continues to be executed, the present position will be repeated between -2,147,483,648 and 2,147,483,647 (command units). (Refer to *7-3 Coordinate System and Present Position*.)

10-8-2 Procedure for Using Software Limits

The PCU uses the Servo Drive's software limit function. The Servo Parameters for software limits must be set to enable the software limit function. After setting the Servo Parameters, the software limit function is enabled by executing ORIGIN SEARCH and PRESENT POSITION PRESET, or detecting the origin using the absolute encoder.

10-8-3 Software Limit Data Settings

Servo Parameter Area

G5-series Software Limit Parameters

Туре	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
	Pn801	Soft limit		0 to 3	2	0
	Pn804	Forward software limit	Command unit	-1,073,741,823 to 1,073,741,823	4	500000
	Pn806	Reverse software limit	Command unit	-1,073,741,823 to 1,073,741,823	4	-500000

G-series Software Limit Parameters

Туре	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
	Pn104	Software limit function		0 to 3	2	0
	Pn201	Forward software limit	Command unit	-1,073,741,823 to 1,073,741,823	4	500000
	Pn202	Reverse software limit	Command unit	-1,073,741,823 to 1,073,741,823	4	-500000

W-series and SMARTSTEP Junior Software Limit Parameters

Туре	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
	Pn801.0	Software limit function		0 to 3	2	(See note 2.)
	Pn801.2	Software limit check using references		0 (Do not change the set- ting.)	(See note 1.)	0
	Pn804	Forward software limit	Command unit	-1,073,741,823 to 1,073,741,823	4	819,191,808
	Pn806	Reverse software limit	Command unit	-1,073,741,823 to 1,073,741,823	4	-819,191,808

Note

- (1) The data length for Pn801.0 and Pn801.2 is that specified when Servo Parameter Pn801 is transferred.
 - (2) The default setting for Pn801.0 is 0 when using R88D-WT□, and 3 when using the R88D-WN□-ML2 or a SMARTSTEP Junior Servo Drive.

The software limit settings are provided in the following pages. Set the software limits so that the reverse software limit is less than the forward software limit. If the forward software limit is less than or equal to the reverse software limit, the software limit will always be enabled.

When the software limit function is enabled, the status is indicated by the Forward/Reverse Software Limit Flags in the Axis Operating Input Memory Area, as follows:

Axis Operating Input Memory Areas (Monitoring)

Name	Word	Bits	Contents
Forward Software Limit Flag	b+1	12	0: Within forward software limit range 1: Forward software limit range exceeded
Reverse Software Limit Flag		13	0: Within reverse software limit range 1: Reverse software limit range exceeded

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

10-8-4 Software Limit Operation

The software limit function operates as described in the following table for the present position when the software limits are set.

Softv	ware limit function	Software limit	t by command		
		0: No software limit check using references. (Default setting)	1: Software limit check using references.		
0	Software limit enabled	When the feedback present position for the active axis reaches the forward/reverse software limit, in the same way as when forward/reverse rotation limit input signals are input, the axis will stop and a Forward/Reverse Software Limit Error (error code: 3002/3003) will occur.	Do not use this setting. Otherwise, the axis operation will not be performed properly.		
1	Forward software	The forward software limit is disabled.			
	limit disabled	When the feedback present position for the active axis reaches the reverse software limit, the axis will stop and a Reverse Software Limit Error (error code: 3003) will occur in the same way as when a reverse rotation limit input signal is input.			
2	Reverse software	The reverse software limit is disabled.			
	limit disabled	When the feedback present position for the active axis reaches the forward software limit, the axis will stop and a Forward Software Limit Error (error code: 3002) will occur in the same way as when a forward rotation limit input signal is input.			
3	Software limit dis- abled in both direc-	The software limits are disabled in both directions.			
	tions	The refresh range and positioning range for the present position is $-2,147,483,647$ to 2,147,483,647 (command units), and the possible operating range is within the for- ward/reverse rotation limit input signal range.			

When a movement command is executed using direct operation for which the target position will exceed the software limit range, the operation when the software limit is reached will be as shown in the following diagram.



If RELATIVE MOVEMENT is executed using a position command value that exceeds the positioning range (-2,147,483,647 to 2,147,483,647 command units), a Position Designation Error (error code: 3060) will occur when the command is executed.

Note The stop operations when the software limit is reached is the same as when limit input signals are input. For details, refer to *7-5 Limit Input Operations*.

When the axis movement command is executed (after alarm reset) from the position at which the software limit is exceeded (software limit has been detected), the operation will depend on the direction in which the software limit is detected, as follows:

Movement command	Opposite direction of software limit	Direction of software limit
ABSOLUTE MOVEMENT RELATIVE MOVEMENT	Operation is possible. Even if the target position is still beyond the software limit in the direction that has been detected, positioning will be performed to the target position and a Software Limit Error will not be detected even after positioning is completed.	A Software Limit Error will occur when the com- mand is executed.
	If the target position is beyond the software limit in the other direc- tion, a Software Limit Error will occur when the other software limit is reached.	
	If, however, RELATIVE MOVEMENT is executed using a position command value that exceeds the positioning range (-2,147,483,647 to 2,147,483,647), a Position Designation Error (error code: 3060) will occur when the command is executed.	
INTERRUPT FEEDING	Operation depends on ABSOLUTE MOVEMENT or RELATIVE MOVEMENT.	A Software Limit Error will occur when the com-
	If the interrupt feeding direction after an interrupt input is opposite to the starting movement direction (reversal operation), the opera- tion is as follows:	mand is executed.
	• If interrupt feeding is performed after passing the software limit in the direction that has been detected. When the interrupt feeding position does not reach the software limit, positioning will be com- pleted. If the software limit will be exceeded due to interrupt feed- ing, a Software Limit Error will occur when the software limit position is reached again.	
	• If interrupt feeding is executed without passing the software limit in the direction that has been detected, positioning will stop at the reversal position and reverse operation will not be executed due to the Driver's drive prohibited status. In this state, a Software Limit Error will not be detected.	
ORIGIN RETURN	Origin return operations are possible when performed in the oppo- site direction of the software limit that has been detected. If the ori- gin position exceeds the other software limit, however, a Software Limit Error will occur when the other software limit is reached.	A Software Limit Error will occur when the com- mand is executed.

Movement command	Opposite direction of software limit	Direction of software limit				
JOG	Operation is possible.	A Software Limit Error will				
	A Software Limit Error will not be detected even if jogging is stopped, or if DECELERATION STOP or EMERGENCY STOP is executed, at a position that is still beyond the software limit in the direction that has been detected.	occur when the com- mand is executed.				
	If operation is not stopped, a Software Limit Error will occur when the other software limit is reached.					
SPEED CONTROL	Operation is possible.	A Software Limit Error will				
TORQUE CONTROL	A Software Limit Error will not be detected even if DECELERA- TION STOP or EMERGENCY STOP is executed at a position that is still beyond the software limit in the direction that has been detected.					
	If operation is not stopped, a Software Limit Error will occur when the other software limit is reached.					
PRESENT POSITION PRESET	If PRESENT POSITION PRESET is executed for a position beyond present travel direction, the present position will change and a Softw detected. If PRESENT POSITION PRESET is executed for a position limit in the direction opposite to the travel direction, the present positive ware Limit Error will occur for the software limit in the opposite direction	the software limit in the ware Limit Error will not be on beyond the software tion will change and a Soft- ction.				
SERVO LOCK	SERVO LOCK can be executed regardless of software limit detection Incremental Encoder or when using an Absolute Encoder as an Incrusing an Absolute Encoder, operation depends on the unit version of	on status when using an remental Encoder. When of the PCU as follows:				
	PCUs with Unit Version 1.2 or Earlier					
	A software limit error will be detected when SERVO LOCK is exe LOCK operation will be cancelled. Disable the software limit to en LOCK.	cuted and the SERVO nable executing SERVO				
	PCUs with Unit Version 1.3 or Later					
	SERVO LOCK can be executed regardless of software limit deter way as when using an Incremental Encoder.	ction status in the same				

10-9 Stop Functions

10-9-1 Overview

The stop function is used to stop active axes operated by position control, speed control, or torque control. Axis operation can be stopped using the deceleration stop method, which stops the axis using a specified deceleration, or the emergency stop method, which cancels the operating command immediately and stops operation after moving for the remaining pulses in the Servo Drive's deviation counter. Axis operations are also stopped by the PCU if an error occurs.

For details on PCU operations when errors occur, refer to *SECTION 12 Troubleshooting*.

10-9-2 Deceleration Stop

Deceleration stop stops the active axis using the deceleration set in the Servo Parameters when the DECELERATION STOP Bit turns ON in the Axis Operating Output Memory Area. The deceleration stop operation is executed using position control with the specified deceleration curve.

When DECLARATION STOP is executed during speed control or torque control, the feedback speed at that point is used as the start speed, the control mode switches to position control, and operation decelerates to a stop.

WWW.BSNEW.IR

Servo Parameter Area



DECELERATION STOP can be executed for an axis that is servo-locked as long as SERVO UNLOCK, DEVIATION COUNTER RESET, or EMERGENCY STOP is not being executed and the command bits for these commands are not ON and as long as an axis error has not occurred.

When executing DECELERATION STOP, make sure that the DECELERA-TION STOP Bit remains ON until the Receiving Command Flag or Stop Execution Flag in the Axis Operating Input Memory Area turns ON.

All axis operation commands except for SERVO UNLOCK, DEVIATION COUNTER RESET, and EMERGENCY STOP will be ignored during DECEL-ERATION STOP execution or while the DECELERATION STOP Bit is ON.

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting
Acceleration/ deceleration con-	Pn811	Linear acceleration constant	10,000 command units/s ²	-32768 to 32767	2	100
sianis	Pn814	Linear deceleration constant	10,000 command units/s ²	-32768 to 32767	2	100
Acceleration/ deceleration fil- ters	Pn818	Position command FIR filter time constant	0.1 ms	0 to 10000	2	0

G5-series Acceleration/Deceleration Parameters

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting
Acceleration/ deceleration con-	Pn107	Linear acceleration constant	10,000 command units/s ²	-32768 to 32767	2	100
stants	Pn10A	Linear deceleration constant	10,000 command units/s ²	-32768 to 32767	2	100
Acceleration/ deceleration fil- ters	Pn10E	Moving average time	0.1 ms	0 to 5100	2	0

G-series Acceleration/Deceleration Parameters

W-series and SMARTSTEP Junior Acceleration/Deceleration Parameters

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting	W Series	SMART STEP Junior
Accelera- tion/ decelera-	Pn80D	First-step linear deceleration constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Not sup- ported
stants	Pn80E	Second-step linear decelera- tion constant	10,000 command units/s ²	1 to 65535	2	100	Sup- ported	Sup- ported
	Pn80F	Deceleration constant switch- ing speed	100 com- mand units/s	0 to 65535	2	0	Sup- ported	Not sup- ported

Stop Functions

Туре	Parame- ter No.	Parameter name	Unit	Setting range	Parame- ter size	Default setting	W Series	SMART STEP Junior
Accelera- tion/	Pn810	Exponential accelera- tion/deceleration bias	Command units/s	0 to 32767	2	0	Sup- ported	Not sup- ported
decelera- tion filters	Pn811	Exponential accelera- tion/deceleration time con- stant	0.1 ms	0 to 5100	2	0	Sup- ported	Not sup- ported
	Pn812	Movement average time	0.1 ms	0 to 5100	2	0	Sup- ported	Not sup- ported

First-step deceleration curve operations require parameter settings for Pn80E only. For the acceleration/deceleration curve, Pn810 and Pn811 must be set when using an exponential curve, and Pn812 must be set when using an S-curve.

The SMARTSTEP Junior Servo Drives do not support the following parameters: Pn80D and Pn80F. They also do not support acceleration/deceleration filters, so the filters cannot be set. A one-step linear deceleration curve is set using only Pn80E.

Axis Operating Output Memory Areas (Operating Commands)

Name	Word	Bits	Contents
DECELERATION STOP	а	15	$0 \rightarrow 1$: Starts deceleration stop.

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

Axis Operating Input Memory Areas (Monitoring)

Name	Word	Bits	Contents
Receiving Command	b	00	0: Command reception enabled.
Flag			$0 \rightarrow 1$: Command reception started.
			1: Receiving command (command reception disabled).
Busy Flag		13	1: Axis busy (axis operation executing).
Stop Execution Flag		15	$0 \rightarrow 1$: Stop operation completed.

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. –1) \times 25

The Stop Execution Flag turns ON when DECELERATION STOP or EMER-GENCY STOP is received, but does not check whether the remaining pulses in the Servo Drive are within the positioning completion range, unlike the PCU Positioning Completed Flag in direct operation.

The Stop Execution Flag turns OFF when direct operation, jogging, speed commands or other movement commands are executed.

10-9-3 Emergency Stop

The emergency stop operation cancels the operating command immediately and stops the active axis when the EMERGENCY STOP Bit turns ON in the Axis Operating Output Memory Area.

When EMERGENCY STOP is executed, axis operation switches to position control mode, and stops in Servo locked status. When EMERGENCY STOP is executed during position control, the axis operation distributes the pulses remaining in the Servo Drive's deviation counter and then stops.

WWW.BSNEW.IR



EMERGENCY STOP can be executed for an axis that is servo-locked as long as SERVO UNLOCK or DEVIATION COUNTER RESET is not being executed and the command bits for these commands are not ON and as long as an axis error has not occurred.

When executing EMERGENCY STOP, make sure that the EMERGENCY STOP Bit remains ON until the Receiving Command Flag or Stop Execution Flag in the Axis Operating Input Memory Area turns ON.

All axis operation commands except for SERVO UNLOCK and DEVIATION COUNTER RESET will be ignored during EMERGENCY STOP execution or while the EMERGENCY STOP Bit is ON.

Axis Operating Output Memory Areas (Operating Commands)

Name	Word	Bits	Contents
EMERGENCY STOP	a+1	15	$0 \rightarrow 1$: Starts emergency stop.

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) \times 25

Axis Operating Input Memory Areas (Monitoring)

Name	Word	Bits	Contents
Receiving Command	b	00	0: Command reception enabled.
Flag			$0 \rightarrow 1$: Command reception started.
			1: Receiving command (command reception disabled).
Busy Flag		13	1: Axis busy (axis operation executing).
Stop Execution Flag		15	$0 \rightarrow 1$: Stop operation completed.

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. –1) \times 25

The Stop Execution Flag turns ON when DECELERATION STOP or EMER-GENCY STOP is received, but does not check whether the remaining pulses in the Servo Drive are within the positioning completion range, unlike the PCU Positioning Completed Flag in direct operation.

The Stop Execution Flag turns OFF when direct operation, jogging, speed commands or other movement commands are executed.

10-9-4 Stop Function Timing Chart

The following timing chart is for when DECELERATION STOP is executed. The timing chart for EMERGENCY STOP is the same, except that operation is stopped after distributing the pulses remaining in the Servo Drive.



a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) × 25 b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

When DECELERATION STOP or EMERGENCY STOP is executed during direct operation (ABSOLUTE MOVEMENT/RELATIVE MOVEMENT) or ORI-GIN SEARCH, the PCU Positioning Completed Flag will not turn ON when operation stops. Even if the stop position for DECELERATION STOP/EMER-GENCY STOP is within the positioning completion range of the target position for the initial positioning operation, the PCU Positioning Completed Flag will not turn ON.

The status of the Receiving Command Flag, Busy Flag, and Stop Execution Flag will change if DECELERATION STOP or EMERGENCY STOP is executed when an axis operating command is not being executed.

The stop operation will also be performed in the same way as for an active axis if still in the controlled status while the axis is being stopped (Busy Flag is ON), such as during direct operation, jogging, when the speed command value is set to 0 for speed control, or when the axis is being stopped during torque control.

Stop Functions



Example: Executing DECELERATION STOP for Speed Control with a Speed Command Value of 0

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) × 25 b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

10-10 DEVIATION COUNTER RESET

10-10-1 Overview of DEVIATION COUNTER RESET

The deviation counter can be reset during position control for PCUs with unit version 1.3 or later.

The deviation counter reset function resets to 0 the position deviation accumulating in the Servo Drive's position loop deviation counter during position control. When pressing using torque control, when the machine catches and stops moving, or in other cases when the command present position and the feedback present position vary greatly, the positioning operation may not finished because the position deviation never becomes less than the completion width. DEVIATION COUNTER RESET can be used in such cases to reset the accumulated position deviation to 0 and thus complete the positioning operation.

With this function, the PCU's deviation counter reset function uses a movement command in the direction opposite to and the same size as the accumulated position deviation so that the command present position will match the feedback present position.



DEVIATION COUNTER RESET is valid only during position control after the servo has been locked and will be ignored if executed during speed control or torque control, or when the servo is unlocked.

Note Execute the PCU's DEVIATION COUNTER RESET only while the motor axis (machine) is stopped, by being pressed against something for example. This function executes a movement command in the direction opposite to and of the same size as the accumulated position deviation, so the machine may be subjected to shock if a movement command in the opposite direction is suddenly executed for a rotating axis. Also, do not execute DEVIATION COUNTER RESET for a vertical axis that is stopped by a servo lock only, without being held mechanically. While the vertical axis is servo-locked, the axis is controlled so that the vertical position is held using the accumulated position deviation. Therefore, executing a deviation counter reset may cause the motor load to fall.

10-10-2 Using DEVIATION COUNTER RESET

DEVIATION COUNTER RESET is executed by turning ON the DEVIATION COUNTER RESET Bit in the Axis Operating Output Area to execute a movement command that effectively cancels the position deviation.

DEVIATION COUNTER RESET executes position control according to the position deviation at a maximum speed of 128,000. If the accumulated position deviation is large, processing time equivalent to several MECHATROLINK communications cycles will be required to complete resetting the deviation counter.

Example

MECHATROLINK communications cycle: 2 ms

Position deviation: 800 (command units)

The movement command executed for DEVIATION COUNTER RESET will be as shown below for the above conditions. The direction will be so that the position deviation is eliminated.



In this example, the movement command to eliminate the position deviation of 800 command units exceeds the movement possible in one MECHATROLINK communications cycle at a speed of 128,000 command units/s, so the command is executed over multiple cycles (four total).

When DEVIATION COUNTER RESET is executed, the torque limit does not depend on the setting of the Torque Limit Designation Bit, but on the status in the previous axis operation. (The previous torque limit status is continued, and DEVIATION COUNTER RESET is executed.)

Axis Operating Output Area (Operation Commands)

Name	Word	Bit	Description
DEVIATION COUNTER RESET Bit	а	13	$0 \rightarrow 1$: Deviation counter reset started.

a = Beginning word of Axis Operating Output Area specified in Common Parameters + (Axis No. - 1) x 25

A deviation counter reset can be executed for any axis for which position control that includes a Servo lock is being executed, as long as there is no axis error, the Servo is locked, and the Servo Unlock Bit is not ON. The command is invalid if executed during speed control or torque control, or when the Servo is unlocked.

When executing a deviation counter reset, keep the Deviation Counter Reset Bit ON until Receiving Command Flag turns ON in the Axis Operating Output Area or until the Busy Flag is turned OFF. While the Deviation Counter Reset Bit is ON and a deviation counter reset is being executed, all operation commands other than SERVO UNLOCK are disabled.

Axis Operating Input Area (Monitoring)

Name	Word	Bits	Contents
Receiving Command	b	00	0: Command reception enabled.
Flag			$0 \rightarrow 1$: Command reception started.
			1: Receiving command (command reception disabled).
Error Flag		12	0: No axis error
			1: Axis error has occurred.
Busy Flag		13	1: Axis busy (axis operation executing).

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. $-1) \times 25$

The status of the PCU Positioning Completed Flag, the No Origin Flag, and the Stop Execution Bit are not changed by turning ON the DEVIATION COUNTER RESET Bit. When the DEVIATION COUNTER RESET is executed, the Busy Flag is held.

Timing Chart

The timing chart for DEVIATION COUNTER RESET is shown below.



 $a = Beginning word of Axis Operating Output Area specified in Common Parameters + (Axis No. -1) \times 25$ $b = Beginning word of Axis Operating Input Area specified in Common Parameters + (Axis No. -1) \times 25$

WWW.BSNEW.IR

SECTION 11 Sample Programs

This section provides basic program examples and application examples for using the Position Control Unit.

11-1	Overvie	w	422
11-2	Basic Pr	ogram Examples	423
	11-2-1	Transferring PCU Parameters	423
	11-2-2	Starting and Stopping MECHATROLINK Communications	428
	11-2-3	Transferring Servo Parameters	429
	11-2-4	Servo Lock/Unlock	436
	11-2-5	Origin Search	438
	11-2-6	Positioning (Absolute Movement or Relative Movement)	439
	11-2-7	Speed Control	442
	11-2-8	Torque Control	443
	11-2-9	Deceleration Stop or Emergency Stop	445
	11-2-10	Jogging	446
11-3	Applica	tion Examples	448
	11-3-1	Initial PCU Settings	448
	11-3-2	Servo Parameter Backup	454
	11-3-3	Switching among Position, Speed, and Torque Control	466

11-1 Overview

This section shows sample programs that execute the various PCU functions from the ladder program.

Refer to *11-2 Basic Program Examples* for descriptions of the sample ladder program sections containing the functions. Refer to *11-3 Application Examples* for sample programs that combine those various program sections.

This section explains the example system and conditions used for the sample programs.

Note The sample programs shown in this section use the PCU's functions and are configured only with ladder programs related to the interface between the PLC and PCU.

When creating the program for the actual equipment in the system, add program sections to perform tasks such as interlocking the equipment's movement, managing I/O from other equipment, and controlling operation so that the program will provide the required operating commands to the PCU.

In addition, always test the new program thoroughly to verify proper operation before switching to full-scale operation in the system.

Devices in the Sample System

Position Control Unit	CJ1W-NCF71 Unit number: 0 (Beginning word of Common Operating Memory Area: CIO 1500)
	Only one PCU is used in the PLC.
	When a different unit number is required, use the following equation to deter- mine the beginning word of Common Operating Memory Area.
	Beginning word of Common Operating Memory Area: n = 1500 + (unit number $\times25)$
MECHATROLINK Devices (Servo Drives)	W-series Servo Drive (R88D-WT□) FNY-NS115 MECHATROLINK-II I/F Unit mounted to Servo Drive
	Axis 1 (MECHATROLINK-II station address 1) is used to demonstrate axis operation in these sample ladder programs.
	When a different number of axes or station number is required, use the follow- ing equation to determine the beginning CIO word addresses for each axis.
	Beginning word of Axis Operating Output Memory Area for Axis N: Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) \times 25
	Beginning word of Axis Operating Input Memory Area for Axis N: Beginning word of Axis Operating Input Areas specified in Common Pa- rameters + (Axis No. -1) \times 25
	Refer to <i>11-3-1 Initial PCU Settings</i> for an example that shows how to set the beginning words of Axis Operating Input/Output Areas.
	The standard settings are used for the Servo Drive's I/O signals; for details on the standard settings, refer to <i>6-4 Standard Settings for Servo Drives Using MECHATROLINK</i> .
CPU Unit	Model number: CJ1H-CPU
	In <i>11-3 Application Examples</i> , the CPU Unit's EM Area is used to store the PCU's Common Parameter, Axis Parameter, and Servo Parameter data.
	If the CJ1H-CPU CPU Unit being used is not equipped with an EM Area, change the storage area to an unused section of the DM Area.

WWW.BSNEW.IR

 Installation and
 Install and wire the devices according to the directions in SECTION 3 Installation and Wiring.

 Operation
 Operation

Connect a 24-VDC control input power supply for the Servo Drive's I/O signals and connect normally closed switches/sensors for the positive and negative overtravel inputs (forward drive prohibited and reverse drive prohibited signals).

Also, when an origin search operation is to be executed, connect a normally open switch/sensor for the origin proximity input signal (origin return deceleration LS).

11-2 Basic Program Examples

This section provides sample ladder programs related to the PCU's basic functions.

11-2-1 Transferring PCU Parameters

Overview

The parameters set in the PCU (Common Parameters and Axis Parameters) can be transferred to and from the CPU Unit's EM Area using the READ DATA Bit and WRITE DATA Bit.

Also, the parameters that have been written to the PCU can be saved to the PCU's internal flash memory using the SAVE DATA Bit.

These sample programs use the following parts of the CPU Unit's I/O memory.

CPU Bus Unit Area

CIO 1500 to CIO 1524

These word addresses are contained in the CPU Bus Unit Area words allocated to unit number 0.

Data Memory Area (DM Area)

D10000 to D10039

These words contain the write data when writing or the read data when reading.

In this sample program, 40 words of data are written and read, and D10000 is specified as the starting word, so D10000 to D10039 are used.

Work Area (WR Area)

Writing data: W300.00 to W300.06 Reading data: W301.00 to W301.06 Saving data: W302.00 to W302.06

These Work Area bits are used to show the progress of function execution and the execution status.

Writing DataUse the WRITE DATA Bit in the Common Operating Memory Area to write the
write data prepared in the DM Area of the CPU Unit to the PCU.

When the execution condition for the write operation goes ON, the 40 words of write data (preset in the CPU Unit's DM Area starting at D10000) are transferred to the PCU's internal memory starting at address 1838 hex.

When the data has been written, W300.05 is turned ON for one cycle.



Reading Data

Use the READ DATA Bit in the Common Operating Memory Area to read the read data from the PCU's internal memory to the CPU Unit's DM Area.

When the execution condition for the read operation goes ON, the 40 words of read data are read from the PCU's internal memory (starting at address 1838 hex) to the CPU Unit's DM Area (starting at D10000).

When the data has been read, W301.05 is turned ON for one cycle.



Saving DataUse the SAVE DATA Bit in the Common Operating Memory Area to save the
PCU's Common Parameters and Axis Parameters to the PCU's flash memory.
The PCU's Save Data operation is executed when the Save Data execution
condition goes from OFF to ON.
When the data has been saved, W302.05 is turned ON for one cycle.
The Save Data operation cannot be executed while MECHATROLINK com-
munications are active. If the Save Data is executed during MECHATROLINK
communications, a Multi-start Error (Unit error code: 0021).
Execute this sample program only when MECHATROLINK communications
are stopped.

Sample Ladder Program

	Program name	: Save Data					
	Section name:	Save Data					
000000						W302.00	Start Unit Data Save
(000000)	Save Data execution condi	tion				()	-
000001	Save Data						1
(000002)	W302.00	1515.12	1515.14		 	1500.03	SAVE DATA Bit
	Start Unit Data Save 1500.03	Unit Error Flag	Data Transferring Flag			W302.01	Execute Unit Data Save
					 	0	-
000002	W302.02	1515.14			 	W302.03	Unit Data Save Received
(000000)	Waiting for Unit Data Save Received	Data Transferr	ing Flag			Ũ	
000003	W302.04	1515.12	1515.14			W302.05	Unit Data Save Normal End
(000011)	Waiting for Unit Data Save End	Unit Error Flag	Data Transferri	ng Flag			
000004 (000015)	W302.02	1515.12			 	W302.06	Unit Data Save Error End
	Waiting for Unit Data Save Received W302.04	Unit Error Flag					
	Waiting for Unit Data Save End						
000005	W302.01	W302.03	W302.06			W302.02	Waiting for Unit Data Save Received
(000019)	Execute Unit Data Save W302.02	Unit Data Save Received	Unit Data Save	Error End			
	Waiting for Unit I	Data					-
000006	W302.03	W302.05	W302.06		 	W302.04	Waiting for Unit Data Save End
(000024)	I I Unit Data Save Received W302.04	Unit Data Save Normal End	Unit Data Save	Error End		0	
	Waiting for Unit						-
	Data Save Ellu						

11-2-2 Starting and Stopping MECHATROLINK Communications

Overview

This program starts MECHATROLINK communications (establishes a connection) based on the MECHATROLINK communications settings and the scan list set in the PCU's Common Parameters; the program can also stop communications (release the connection).

While the PCU connection is established, MECHATROLINK communications will be stopped automatically if normal communications are not established with all of the axes registered in the scan list within the preset timeout time.

(If the PCU cannot start communications with an axis registered in the scan list within 10 seconds, an MLK Initialization Error will occur. This sample program contains a timer function that stops communications before the MLK Initialization Error is detected.)

This program uses the MECHATROLINK communications settings and the scan list that are already set in the PCU's Common Parameters.

This sample program uses the following parts of the CPU Unit's I/O memory.

CPU Bus Unit Area

CIO 1500 to CIO 1524

These word addresses are contained in the CPU Bus Unit Area words allocated to unit number 0.

Work Area (WR Area)

W303.00 to W303.04

These Work Area bits are used to show the progress of function execution and the execution status.

Timer Area

TIM0000

Used as a watchdog timer to monitor the establishment of a connection.

This program uses the CONNECT Bit in the Common Operating Memory Area to start and stop MECHATROLINK communications.

When the Start Communications condition goes ON, MECHATROLINK communications are started (a connection is established). Likewise, when the Stop Communications condition goes ON, MECHATROLINK communications are stopped (the connection is released).

In this sample program, only Axis 1 is registered in the scan list.

If MECHATROLINK communications start normally with Axis 1, Work Area bit W303.03 will be turned ON.

The connection will be stopped and Work Area bit W303.04 will be turned ON if MECHATROLINK communications do not start with Axis 1 5 seconds after execution of CONNECT.

Since the PCU will continue MECHATROLINK communications while the CONNECT Bit (CIO 1501.00) is ON, the CONNECT Bit is programmed as a self-holding bit in the ladder program. If an error occurs that stops communications, such as a communications error, the PCU will release the connection regardless of the status of the CONNECT Bit.

To reestablish the connection after eliminating the cause of the error, the CONNECT Bit must be turned OFF and then ON again. When releasing the connection after an error occurs, use W303.02 to turn OFF the self-holding circuit that holds the CONNECT Bit.



11-2-3 Transferring Servo Parameters

Overview

The Servo Parameters in Servo Drives connected through MECHATROLINK communications can be transferred to and from the CPU Unit's DM Area with the PCU's WRITE SERVO PARAMETER Bit, READ SERVO PARAMETER Bit, and SAVE SERVO PARAMETER Bit.

The application example shown in *11-3-2 Servo Parameter Backup* incorporates this sample program to transfer all the W-series Servo Drive's Servo Parameters at once between the Servo Drive (equipped with a FNY-NS115 I/F Unit) and the CPU Unit's EM Area.

Execute these sample programs only after starting MECHATROLINK communications (establishing a connection).

In these examples, the Servo Parameters are transferred to and from the Servo Drive registered as axis 1.

The Axis Operating Output/Input Memory Areas for the axis are based on the settings of the Common Parameter Area, as shown below. The application example in *11-3-1 Initial PCU Settings* shows a sample program that can be used to make the following settings.

Beginning word of Axis Operating Output Memory Area: CIO 0000 (Axis 1 Operating Output Memory Area: CIO 0000 to CIO 0024)

Beginning word of Axis Operating Input Memory Area: CIO 1000 (Axis 1 Operating Input Memory Area: CIO 1000 to CIO 1024)

This sample program uses the following parts of the CPU Unit's I/O memory in addition to the CIO Area words listed above.

Writing Servo

Parameters

Data Memory Area (DM Area)

D10500 to D10503

These words are used to store the data required for a single Servo Parameter transfer, including the Servo Parameter number, parameter size, and transfer data (2 words).

Work Area (WR Area)

Writing Servo Parameters:W400.00 to W400.06Reading Servo Parameters:W401.00 to W401.06Saving Servo Parameters:W402.00 to W402.06

These Work Area bits are used to show the progress of function execution and the execution status.

Use the WRITE SERVO PARAMETER Bit in the Axis Operating Output Memory Area to write the Servo Parameter data (preset in the CPU Unit's DM Area) to the Servo Drive.

When the Servo Parameter transfer operation's execution condition goes ON, the Servo Parameters preset in the following DM Area words will be transferred to the Servo Drive.

Word	Details
D10500	Servo Parameter No.
D10501	Parameter size (Unit: bytes)
D10502	Write data (rightmost word)
D10503	Write data (leftmost word)

When the Servo Parameter data has been written, W400.05 is turned ON for one cycle.

When the write operation was not completed normally, i.e., an error occurred during the transfer, W400.06 will be turned ON for one cycle. (There must be no axis error before execution of this sample program.)



Use the READ SERVO PARAMETER Bit in the Axis Operating Output Memory Area to read the Servo Parameters from the Servo Drive to the CPU Unit's DM Area.

When the Servo Parameter transfer operation's execution condition goes ON, the Servo Parameter (specified in the DM Area as shown below) will be read.

Word	Details
D10500	Servo Parameter No.
D10501	Parameter size (Unit: bytes)

When the Servo Parameter data has been read normally, W401.05 is turned ON for one cycle and the parameter is stored in the DM Area as shown in the following table.

Word	Details
D10502	Read data (rightmost)
D10503	Read data (leftmost)

When the read operation was not completed normally, i.e., an error occurred during the transfer, W401.06 will be turned ON for one cycle. (There must be no axis error before execution of this sample program.)



Saving Servo Parameters

Use the SAVE SERVO PARAMETER Bit in the Axis Operating Output Memory Area to write the transfer data (preset in the CPU Unit's DM Area) to the Servo Drive's non-volatile flash memory.

When the Servo Parameter transfer operation's execution condition goes ON, the Servo Parameters preset in the following DM Area words will be transferred to the Servo Drive and written to the Servo Drive's non-volatile flash memory at the same time.

Word	Details
D10500	Servo Parameter No.
D10501	Parameter size (Unit: bytes)
D10502	Write data (rightmost word)
D10503	Write data (leftmost word)

When the Servo Parameter data has been saved normally, W402.05 is turned ON for one cycle.

When the save operation was not completed normally, i.e., an error occurred during the transfer, W402.06 will be turned ON for one cycle. (There must be no axis error before execution of this sample program.)



11-2-4 Servo Lock/Unlock

|--|

This sample program executes the Servo Lock/Unlock function on a Servo Drive connected through MECHATROLINK communications.

Also, timeout processing will be performed when the Servo Lock function has been executed but the Servo Lock did not engage for the axis within a preset time.

Execute this sample program only after starting MECHATROLINK communications (establishing a connection).

In this example, the Servo Lock/Unlock function is executed on the Servo Drive registered as axis 1 using the SERVO LOCK Bit or SERVO UNLOCK Bit in the Axis Operating Output Memory Area.

The Axis Operating Output/Input Memory Areas are based on the settings of the Common Parameter Area, as shown below. The application example in *11-3-1 Initial PCU Settings* shows a sample program that can be used to make the following settings.

Beginning word of Axis Operating Output Memory Area: CIO 0000 (Axis 1 Operating Output Memory Area: CIO 0000 to CIO 0024)

Beginning word of Axis Operating Input Memory Area: CIO 1000 (Axis 1 Operating Input Memory Area: CIO 1000 to CIO 1024)

This sample program uses the following parts of the CPU Unit's I/O memory in addition to the CIO Area words listed above.

Work Area (WR Area)

W305.00 to W305.05

These Work Area bits are used to show the progress of function execution and the execution status.

Timer Area

TIM0001

Used as a watchdog timer to monitor the execution of the Servo Lock function.

This sample program alternately executes the Servo Lock and Servo Unlock functions each time that the Servo Lock Switch execution condition goes from OFF to ON.

If the Axis 1 Servo Lock is not engaged within 5 seconds after the Servo Lock function is executed, a Servo Lock Execution Timeout will occur, bit W305.05 will be turned ON for one cycle, and the Servo Unlock function will be executed.



Note With the above programming example, if the Servo is unlocked by any means other than the SERVO UNLOCK command bit (e.g., if a Driver main circuit OFF error occurs), the status of the work bit will be held. When using this programming example as reference for actual programming, add interlocks suitable for the operating conditions of the equipment.

11-2-5 Origin Search

Overview

This sample program executes an origin search on a Servo Drive connected through MECHATROLINK communications.

Execute this sample program only after starting MECHATROLINK communications (establishing a connection), executing the Servo Lock function, and verifying that the Servo Lock is engaged.

In these examples, the origin search is performed on the Servo Drive registered as axis 1.

The Axis Operating Output/Input Memory Areas are based on the settings of the Common Parameter Area, as shown below. The application example in *11-3-1 Initial PCU Settings* shows a sample program that can be used to make the following settings.

Beginning word of Axis Operating Output Memory Area: CIO 0000 (Axis 1 Operating Output Memory Area: CIO 0000 to CIO 0024)

Beginning word of Axis Operating Input Memory Area: CIO 1000 (Axis 1 Operating Input Memory Area: CIO 1000 to CIO 1024)

This sample program uses the following parts of the CPU Unit's I/O memory in addition to the CIO Area words listed above.

Work Area (WR Area)

W350.00 to W350.02

These Work Area bits are used to show the progress of function execution and the execution status.

Timer Area

TIM0002

Used as a watchdog timer to detect an origin search timeout error.

This sample program starts an origin search on axis 1 when the Origin Search execution condition goes from OFF to ON. The speed command value and origin search speed must be set in advance in the Axis 1 Operating Output Memory Area before executing this sample program.

If the Axis 1 origin is not established within 10 seconds after the origin input signal is detected and the *Final Travel Distance to Return to Zero Point* positioning is performed, the Origin Search Timeout will occur and W305.02 will be turned ON for one cycle.



11-2-6 Positioning (Absolute Movement or Relative Movement)

Overview

This sample program sends a positioning command to a Servo Drive connected through MECHATROLINK communications.

Execute this sample program only after starting MECHATROLINK communications (establishing a connection), executing the Servo Lock function, and verifying that the Servo Lock is engaged.

In addition, when executing ABSOLUTE MOVEMENT, establish the origin by performing an origin search before sending the position command.

In this example, positioning is performed on the Servo Drive registered as axis 1 using the ABSOLUTE MOVEMENT Bit or RELATIVE MOVEMENT Bit in the Axis Operating Output Memory Area.

The Axis Operating Output/Input Memory Areas are based on the settings of the Common Parameter Area, as shown below. The application example in *11-3-1 Initial PCU Settings* shows a sample program that can be used to make the following settings.

Beginning word of Axis Operating Output Memory Areas: CIO 0000 (Axis 1 Operating Output Memory Area: CIO 0000 to CIO 0024)

Beginning word of Axis Operating Input Memory Areas: CIO 1000 (Axis 1 Operating Input Memory Area: CIO 1000 to CIO 1024)

This sample program uses the following parts of the CPU Unit's I/O memory in addition to the CIO Area words listed above.

Work Area (WR Area)

W201.00 to W201.03

These Work Area bits are used to show the progress of function execution and the execution status.

Holding Area (HR Area)

H1.00

This bit is used as a switch to indicate whether positioning is performed with ABSOLUTE MOVEMENT or RELATIVE MOVEMENT.

Timer Area

TIM0003

Used as a watchdog timer to monitor the positioning operation.

This sample program starts Axis 1 positioning when the Positioning execution condition goes ON.

The position command value must be set to the target position and the speed command value must be set to the target speed in the Axis Operating Output Memory Area before executing this sample program.

At this point, the ON/OFF status of HR1.00 determines whether the positioning is executed with RELATIVE MOVEMENT (HR1.00 OFF) or ABSOLUTE MOVEMENT (HR1.00 ON).

If axis 1 positioning is not completed within 10 seconds after completion of sending the positioning command, i.e., after pulse distribution is completed, a Positioning Timeout will occur and bit W201.03 will be turned ON for one cycle.

The target position can be changed during an Absolute Movement or Relative Movement positioning operation by setting a new position command value and turning ON the ABSOLUTE or RELATIVE MOVEMENT Bit. Torque limits can also be enabled/disabled during positioning by turning ON/OFF the forward and reverse rotation current limit bits in the Axis Operating Output Memory Area and turning ON the ABSOLUTE or RELATIVE MOVEMENT Bit.

This sample program allows the target position to be changed during an operation by turning the execution condition OFF and then ON again.

A Multistart Error will occur if a Movement command is sent while the PCU's Receiving Command Flag is already ON, so the Receiving Command Flag is ANDed with the Positioning execution condition (in the W201.00 output circuit). An OR circuit containing the ABSOLUTE MOVEMENT Bit and RELA-TIVE MOVEMENT Bit has been added so that the W201.00 is not output again when the Receiving Command Flag goes from ON to OFF.

The target speed can be changed sequentially during operation by overwriting the speed command value in the Axis Operating Output Memory Area.
Sample Ladder Program



11-2-7 Speed Control

Overview

This sample program performs speed control on a Servo Drive connected through MECHATROLINK communications.

Execute this sample program only after starting MECHATROLINK communications (establishing a connection), executing the Servo Lock function, and verifying that the Servo Lock is engaged.

In this example, speed control is performed on the Servo Drive registered as axis 1 using the SPEED CONTROL Bit in the Axis 1 Operating Output Memory Area.

The Axis Operating Output/Input Memory Areas are based on the settings of the Common Parameter Area, as shown below. The application example in *11-3-1 Initial PCU Settings* shows a sample program that can be used to make the following settings.

Beginning word of Axis Operating Output Memory Area: CIO 0000 (Axis 1 Operating Output Memory Area: CIO 0000 to CIO 0024)

Beginning word of Axis Operating Input Memory Area: CIO 1000 (Axis 1 Operating Input Memory Area: CIO 1000 to CIO 1024)

This sample program uses the following parts of the CPU Unit's I/O memory in addition to the CIO Area words listed above.

Work Area (WR Area)

W202.00 to W202.02

These Work Area bits are used to show the progress of function execution and the execution status.

This sample program starts Axis 1 speed control when the Speed Control execution condition goes ON.

The speed command value in the Axis 1 Operating Output Memory Area must be set to the target speed in advance before executing this sample program.

The target speed can be changed sequentially during operation by overwriting the speed command value for speed control in the Axis Operating Output Memory Area.

Torque limits can be enabled/disabled and the torque limits can be changed during operation by turning ON/OFF the forward and reverse rotation current limit bits and changing the option command values in the Axis Operating Output Memory Area and then turning ON the SPEED CONTROL Bit.

This sample program allows the speed control command to be sent again during an operation by turning the Speed Control execution condition OFF and then ON again.

A Multistart Error will occur if a Speed Control command is sent while the PCU's Receiving Command Flag is already ON, so the condition that the Receiving Command Flag is OFF is ANDed with the Speed Control execution condition (in the W202.00 output circuit). The Axis 1 SPEED CONTROL Bit has been added as an OR condition so that the W202.00 is not output again when the Receiving Command Flag goes from ON to OFF.

Sample Ladder Program



11-2-8 Torque Control

Overview

This sample program performs torque control on a Servo Drive connected through MECHATROLINK communications.

Execute this sample program only after starting MECHATROLINK communications (establishing a connection), executing the Servo Lock function, and verifying that the Servo Lock is engaged.

In this example, torque control is performed on the Servo Drive registered as axis 1 using the TORQUE CONTROL Bit in the Axis 1 Operating Output Memory Area.

The Axis Operating Output/Input Memory Areas are based on the settings of the Common Parameter Area, as shown below. The application example in *11-3-1 Initial PCU Settings* shows a sample program that can be used to make the following settings.

Beginning word of Axis Operating Output Memory Area: CIO 0000 (Axis 1 Operating Output Memory Area: CIO 0000 to CIO 0024)

Beginning word of Axis Operating Input Memory Area: CIO 1000 (Axis 1 Operating Input Memory Area: CIO 1000 to CIO 1024)

This sample program uses the following parts of the CPU Unit's I/O memory in addition to the CIO Area words listed above.

Work Area (WR Area)

W203.00 to W203.02

These Work Area bits are used to show the progress of function execution and the execution status.

This sample program starts Axis 1 torque control when the Torque Control execution condition goes ON.

The torque command value in the Axis 1 Operating Output Memory Area must be set to the desired output torque in advance before executing this sample program.

The torque control's output torque can be changed sequentially during operation by overwriting the torque command value in the Axis Operating Output Memory Area.

Torque limits can be enabled/disabled and the speed limit value can be changed during operation by turning ON/OFF the forward and reverse rotation current limit bits and changing the option command values in the Axis Operating Output Memory Area and then turning ON the TORQUE CONTROL Bit.

This sample program allows the TORQUE CONTROL command to be sent again during an operation by turning the Torque Control execution condition OFF and then ON again.

A Multistart Error will occur if a TORQUE CONTROL command is sent while the PCU's Receiving Command Flag is already ON, so include an AND of the Receiving Command Flag OFF in the execution condition for Torque Control execution (i.e. from the W203.00 output in the following example). The TORQUE CONTROL Bit has been added as an OR condition so that the W203.00 is not output again when the Receiving Command Flag goes from ON to OFF.



Sample Ladder Program

11-2-9 Deceleration Stop or Emergency Stop

Overview

This sample program performs a deceleration stop or emergency stop on a Servo Drive connected through MECHATROLINK communications. Execute this sample program only after starting MECHATROLINK communications with the servo locked (i.e., after establishing a connection).

In this example, a deceleration stop or emergency stop is performed on the Servo Drive registered as axis 1 using the DECELERATION STOP Bit or EMERGENCY STOP Bit in the Axis 1 Operating Output Memory Area.

The Axis Operating Output/Input Memory Areas are based on the settings of the Common Parameter Area, as shown below. The application example in *11-3-1 Initial PCU Settings* shows a sample program that can be used to make the following settings.

Beginning word of Axis Operating Output Memory Area: CIO 0000 (Axis 1 Operating Output Memory Area: CIO 0000 to CIO 0024)

Beginning word of Axis Operating Input Memory Area: CIO 1000 (Axis 1 Operating Input Memory Area: CIO 1000 to CIO 1024)

This sample program uses the following parts of the CPU Unit's I/O memory in addition to the CIO Area words listed above.

Work Area (WR Area)

Deceleration Stop: W204.00 and W204.01 Emergency Stop: W205.00 and W205.01

These Work Area bits are used to show the progress of function execution and the execution status.

This sample program executes a deceleration stop or emergency stop on Axis 1 when the Deceleration Stop or Emergency Stop execution condition goes ON.

Sample Ladder Program (Deceleration Stop)



Sample Ladder Program (Emergency Stop)



Note Deceleration and emergency stops are possible only when the Servo is locked. If the Servo is unlocked, these stop commands will be ignored. In the above programming example, the Servo ON Flag is inserted as an output condition so that DECELERATION STOP, EMERGENCY STOP, and work bits do not remain ON even if a deceleration stop or emergency stop is executed when the Servo is not locked.

11-2-10 Jogging

Overview

This sample program performs jogging with the Servo Drive connected via MECHATROLINK communications. MECHATROLINK communications start (i.e., a connection is established), and after executing SERVO LOCK, jogging is executed while the Servomotor is in a servo locked state.

In this programming example, jogging is executed for the Servo Drive registered at axis 1 when the Jog Bit turns ON in the Axis Operating Output Memory Area. The Axis Operating Output/Input Memory Areas are based on the settings of the Common Parameter Area, as shown below. The application example in *11-3-1 Initial PCU Settings* shows a sample program that can be used to make the following settings.

- Beginning word of Axis Operating Output Memory Area: CIO 0000
- (Axis 1 Operating Output Memory Area: CIO 0000 to CIO 0024)
- Beginning word of Axis Operating Input Memory Area: CIO 1000
- (Axis 1 Operating Input Memory Area: CIO 1000 to CIO 1024)
- This sample program uses the following parts of the CPU Unit's I/O memory.

Work Area (WR Area)

W206.00 to W206.01

These Work Area bits are used to create the operation timing for function execution.

In this program example, jogging starts when the jogging condition turns ON, and stops when the jogging condition turns OFF. Jogging is performed in the forward direction when the jogging direction is OFF at the start of jogging. Jogging is performed in the reverse direction when the jogging direction is ON at the start of jogging.

When executing this programming example, the target speed must be set in the speed command value of the Axis 1 Operating Output Memory Area. The target speed for jogging can be changed successively during operation by writing the speed command value to the Axis Operating Output Memory Area.

In this programming example, application in combination with the PCU jogging (adjustment operation) Smart Active Parts Library provided in the OMRON NS-series Programmable Terminals has been considered and thus the program is constructed with self-holding bits for the Jog Bit and Direction Designation Bit in the Axis Operating Output Memory Area. Therefore, the command for the jogging condition is sent via the Work Area bit W206.00.

The output circuit of Work Area bit W206.00 is also provided with a self-holding circuit that operates according to the Receiving Command Flag and Busy Flag to enable the JOG command to be received correctly at the PCU.



Ladder Programming Example

This section shows sample application programs created by combining the basic sample programs introduced in the previous section.

To make the basic sample programs easier to understand, the basic sample programs reused in this section's application examples keep the same Work Area word and bit addresses that were allocated in the previous section.

11-3-1 Initial PCU Settings

Overview

This application combines the data-writing and data-saving functions from 11-2-1 Transferring PCU Parameters with the communications function from 11-2-2 Starting and Stopping MECHATROLINK Communications to make the Initial PCU Settings. The 80 words of parameter data (including unused words) must be preset in EM Area words E0_00000 to E0_00079 with the 40 words of Common Parameters and 40 words of Axis Parameters (20 words each for axes 1 and 2).

In this program, the initial PCU settings are made when H0.00 is OFF and the task switches from INI to RUN status. After the initial settings are made, the PCU is restarted. The transferred and saved parameters are enabled and H0.00 is turned ON (Set) at the same time so that the initial settings are not made again.

To make the initial settings again, reset H0.00 and execute the program again.

When H0.00 is reset, the Initial Task Execution Flag is used as the Write Data execution condition and the Unit Data Write End Flag (W300.05) is used as the Save Data execution condition. When the Save Data operation is completed (W302.05 ON), the Unit is restarted and H0.00 is turned ON.

MECHATROLINK communications will be controlled by the new Common Parameters in the PCU either after the Unit restart at the end of initial settings if H0.00 is OFF, or after starting the program (task) if H0.00 is ON.

This sample program uses the following parts of the CPU Unit's I/O memory.

CPU Bus Unit Area

CIO 1500 to CIO 1524

These word addresses are contained in the CPU Bus Unit Area words allocated to unit number 0.

EM Area bank 0

E0_00000 to E0_00079

These 80 EM words are used to transfer to the PCU the 40 words of Common Parameters and 40 words of Axis Parameters (20 words each for axes 1 and 2).

Work Area (WR Area)

Writing Data:	W300.00 to W300.06
Saving Data:	W302.00 to W302.06
Starting Communications:	W303.00 to W303.04
Restarting the Unit:	W304.00 to W304.03

These Work Area bits are used to show the progress of function execution and the execution status.

The following parts of I/O memory are used in the combined program example.

Holding Area (HR Area)

H0.00

Used as a switch to enable/disable the initial PCU settings function.

I/O Bits

Restart Communications:400.00 Stop Communications: 400.01

When the corresponding I/O bit goes ON, stopped MECHATROLINK communications are restarted or established MECHATROLINK communications are stopped.

This sample program writes the following Common Parameters and Axis Parameters (initial settings).

Common Parameters

Scan List: Registers Servo Drives to axes 1 and 2 (station numbers 1 and 2).

Beginning word of Axis Operating Output Memory Area: CIO 0000

Beginning word of Axis Operating Input Memory Area: CIO 1000

MECHATROLINK communications settings:

Transfer cycle: 1 ms Communications cycle: \times 2 (1 ms \times 2 = 2 ms) Number of communications retries: 0 (1 retry) C2 Master connection: 0 (No C2 Master)

Axis Parameters for Axes 1 and 2 (Shared by both Axes.)

Origin input signal selection: 00 (Phase Z)

Interrupt input signal selection: 01 (External latch signal 1 input)

Origin search direction: 0 (Forward)

Origin search operation mode: 1 (Reversal mode 2)

Encoder type: 0 (Incremental encoder)

The following parameter settings must be written in advance to the CPU Unit's EM Area words E0_00000 to E0_00079.

EM word	Set value	Item	Details
E0_00000	00B0 hex	Common Parameters	Sets the beginning word of the Axis Operating
		Axis Operating Output Memory Area	Output Memory Areas to CIO 0000.
		designation	Axis 1 output words: CIO 0000 to CIO 0024
E0_00001	0000 hex	Common Parameters	Axis 2 output words: CIO 0025 to CIO 0049
		Beginning word of Axis Operating Out- put Memory Areas	
E0_00002	00B0 hex	Common Parameters	Sets the beginning word of the Axis Operating
		Axis Operating Input Memory Area des-	Input Memory Areas to CIO 1000.
		ignation	Axis 1 input words: CIO 1000 to CIO 1024
E0_00003	03E8 hex	Common Parameters	Axis 2 input words: CIO 1025 to CIO 1049
		Beginning word of Axis Operating Input Memory Areas	
E0_00004	4040 hex	Common Parameters	Allocates axes 1 and 2 to the Servo Drive.
		Scan list setting (axes 1 and 2)	
E0_00005 to	0000 hex	Common Parameters	
E0_00011		Scan list setting (axes 3 to 16)	
E0_00012 to E0_00029	0000 hex	Not used.	This part of the Common Parameters is unused.
			Set all words to 0000 hex.

EM word	Set value	Item	Details
E0_00030	0102 hex	Common Parameters	Transfer cycle: 01 (1 ms)
		MECHATROLINK communications set- tings Transfer cycle and communications	Communications cycle: 02 (×2)
E0_00031	0000 bex	Common Parameters	Number of communications retries: 0 (1 retry)
20_00001	0000 nex	MECHATROLINK communications set- tings	C2 Master connection: 0 (No C2 Master)
		Number of communications retries and C2 Master connection	
E0_00032 to E0_00039	0000 hex	Not used.	This part of the Common Parameters is unused.
			Set all words to 0000 hex.
E0_00040	0001 hex	Axis 1 Parameters	Origin input signal selection: 00 (Phase Z)
		Origin input signal selection and Inter- rupt input signal selection	Interrupt input signal selection: 01 (External latch signal 1 input)
E0_00041	0010 hex	Axis 1 Parameters	Origin search direction: 0 (Forward)
		Origin search direction and Origin search operation	Origin search operation mode: 1 (Reversal mode 2)
E0_00042	0000 hex	Axis 1 Parameters Encoder type	Encoder type: 0 (Incremental encoder)
E0_00043 to E0_00059	0000 hex	Not used.	This part of the Common Parameters is unused.
			Set all words to 0000 hex.
E0_00060	0001 hex	Axis 2 Parameters	Origin input signal selection: 00 (Phase Z)
		Origin input signal selection and Inter- rupt input signal selection	Interrupt input signal selection: 01 (External latch signal 1 input)
E0_00061	0010 hex	Axis 2 Parameters	Origin search direction: 0 (Forward)
		Origin search direction and Origin search operation	Origin search operation mode: 1 (Reversal mode 2)
E0_00062	0000 hex	Axis 2 Parameters	Encoder type: 0 (Incremental encoder)
		Encoder type	
E0_00063 to E0_00079	0000 hex	Not used.	This part of the Common Parameters is unused.
			Set all words to 0000 hex.

The PCU does not use regions E0_00012 to E0_00029, E0_00032 to E0_00039, E0_00043 to E0_00059, and E0_00063 to E0_00079 in the Common Parameter Area and Axis Parameter Areas.

These unused areas are reserved because this program transfers all 80 words (40 words of Common Parameters and 40 words of Axis Parameters for axes 1 and 2) to the PCU's internal addresses at once.

Set all words in the unused addresses to 0000 hex.

When Axis Parameters are being transferred for axis 3 or higher axes, prepare the parameter settings in 20-word regions for each axis (starting from E0_00080) with the same data configuration as axis 1. This sample program can be used to make the initial settings for multi-axis operation by changing the number of write words to 40 words (Common Parameters) + 20 words × Number of axes (Axis Parameters).

Conversely, when only axis 1 is being used, set the number of write words to 60 words and prepare only EM words E0_00000 to E0_00059 from the table above. (Change the scan list in the Common Parameters so that only axis 1 is used.)

When the initial PCU settings have been made, W304.03 is turned ON for one cycle.

Also, W303.03 will be turned ON when MECHATROLINK communications have started and all axes (axes 1 and 2 in this case) are communicating normally.

Sample Ladder Program



Section 11-3





11-3-2 Servo Parameter Backup

Overview

This application incorporates the Servo Parameter transfer functions from *11-2-3 Transferring Servo Parameters* to transfer (write, read, or save) all of the Servo Drive's Servo Parameters at one time with the CPU Unit's EM Area.

This application example is normally executed after the initial PCU settings application shown in *11-3-1 Initial PCU Settings*.

The Servo Parameters to be transferred are set in the CPU Unit's EM Area as a parameter list.

When reading Servo Parameters, the parameter data read from the Servo Drive is stored in the specified EM words based on this parameter list.

When writing/saving Servo Parameters, the parameter data is stored in the EM Area in the prescribed format and written to the Servo Drives in order.

In this program example, the Servo Drives connected as axes 1 and 2 are both W-series Servo Drives and the same Servo Parameter list is transferred to both.



Written/Saved together

The Axis Operating Output/Input Memory Areas for axes 1 and 2 are based on the settings of the Common Parameter Area, as shown below. The application example in *11-3-1 Initial PCU Settings* shows a sample program that can be used to make the following settings.

Beginning word of Axis Operating Output Memory Area: CIO 0000

(Axis 1 Operating Output Memory Area: CIO 0000 to CIO 0024)

(Axis 2 Operating Output Memory Area: CIO 0025 to CIO 0049)

Beginning word of Axis Operating Input Memory Area: CIO 1000

(Axis 1 Operating Input Memory Area: CIO 1000 to CIO 1024)

(Axis 2 Operating Input Memory Area: CIO 1025 to CIO 1049)

This sample program uses the following parts of the CPU Unit's I/O memory.

CPU Bus Unit Area

CIO 1500 to CIO 1524

These word addresses are contained in the CPU Bus Unit Area words allocated to unit number 0.

Data Memory Area (DM Area)

Axis 1 Servo Parameter words: D10500 to D10503 Axis 2 Servo Parameter words: D10600 to D10603

These words are used to store the data required for a single Servo Parameter transfer, including the Servo Parameter number, parameter size, and transfer data (2 words).

EM Area bank 1

Servo Parameter list:	E1_00000 to E1_00257
Axis 1 Servo Parameter words:	E1_01000 to E1_01511
Axis 2 Servo Parameter words:	E1_02000 to E1_02511

When the Servo Parameters are being read, these EM words are used to store the parameter list read from the Servo Drive. These words are also used to store the Servo Parameters that were read and set the Servo Parameters that will be written.

Index Registers (IR)

IR6 to IR8

The Index Registers are used to reference the Servo Parameter list and reference the EM Area addresses that contain the parameter data being read/written.

Work Area (WR Area)

Write Axis 1 Servo Parameters:	W400.00 to W400.06
Read Axis 1 Servo Parameters:	W401.00 to W401.06
Save Axis 1 Servo Parameters:	W402.00 to W402.06
Write Axis 2 Servo Parameters:	W410.00 to W410.06
Read Axis 2 Servo Parameters:	W411.00 to W411.06
Save Axis 2 Servo Parameters:	W412.00 to W412.06
Batch transfer sequence stepping:	W500.00 to W500.10

These Work Area bits are used to show the progress of function execution and the execution status.

Holding Area (HR Area)

Axis 1 Transfer Function Selection: H10.00 to H10.02 Axis 2 Transfer Function Selection: H11.00 to H11.02

These bits are used to select the Servo Parameter transfer function for each axis.

The following table shows how the status of bits H10.00 to H10.02 and H11.00 to H11.02 specify the Servo Parameter transfer function in this sample program.

Transfer function	Transfer function Axis 1					
	H10.00	H10.01	H10.02	H11.00	H11.01	H11.02
Write Servo Parameter	1	0	0	1	0	0
Read Servo Parameter	0	1	0	0	1	0
Save Servo Parameter	0	0	1	0	0	1

A different transfer function can be selected for each axis and executed for 2 axes simultaneously.

Different transfer functions cannot be performed at the same time on a single axis. If bits 00 to 02 are all OFF or more than one bit is ON, the transfer function will not be executed for that axis.

Also, when executing this sample program, always transfer the Servo Parameters for 2 axes.

CIO 0510.00 is used as the Servo Parameter Batch Transfer Start Switch.

In this program, the parameter list for the Read Servo Parameters function is stored in the CPU Unit's EM Area as follows.

The beginning words of the parameter list specify the parameter number and parameter size of each parameter.

- +0: Parameter number (hex)
- +1: Parameter size

The parameter number and size are specified in order for all of the parameters being transferred.

At the end of the parameter list, the parameter size is set to 0000 hex.

WWW.BSNEW.IR

Parameter List

The following table shows a setting example for the parameter list in this program.

The Servo Parameters listed here are for an OMRON W-series Servo Drive with a FNY-NS115 MECHATROLINK-II I/F Unit mounted. All 256 parameter numbers and parameter sizes shown in the table in *4-5 Servo Parameter Area* are included even if the parameters are not being used.

When the two-word end code (0000 0000) at the end of the parameter list is included, a total of 256 words (including unused words) are used, from $E1_{00000}$ to $E1_{00257}$.

Word	+0	+1	+2	+3	+4	+5	+6	+7	+8	+9
E1_00000	0000	0002	0001	0002	0002	0002	0003	0002	0004	0002
E1_00010	0005	0002	0100	0002	0101	0002	0002	0002	0103	0002
E1_00020	0104	0002	0105	0002	0106	0002	0107	0002	0108	0002
E1_00030	0109	0002	010A	0002	010B	0002	010C	0002	010D	0002
E1_00040	010E	0002	010F	0002	0110	0002	0111	0002	0112	0002
E1_00050	0113	0002	0114	0002	0115	0002	0116	0002	0117	0002
E1_00060	0118	0002	0119	0002	0120	0002	0121	0002	0122	0002
E1_00070	0123	0002	0124	0002	0125	0002	0200	0002	0201	0002
E1_00080	0202	0002	0203	0002	0204	0002	0205	0002	0206	0002
E1_00090	0207	0002	0208	0002	0212	0002	0217	0002	0218	0002
E1_00100	0300	0002	0301	0002	0302	0002	0303	0002	0304	0002
E1_00110	0305	0002	0306	0002	0307	0002	0308	0002	0309	0002
E1_00120	0400	0002	0401	0002	0402	0002	0403	0002	0404	0002
E1_00130	0405	0002	0406	0002	0407	0002	0408	0002	0409	0002
E1_00140	040A	0002	040B	0002	040C	0002	0500	0002	0501	0002
E1_00150	0502	0002	0503	0002	0504	0002	0505	0002	0506	0002
E1_00160	0507	0002	0508	0002	0509	0002	050A	0002	050B	0002
E1_00170	050C	0002	050D	0002	050E	0002	050F	0002	0510	0002
E1_00180	0511	0002	0512	0002	0513	0002	051A	0002	051B	0002
E1_00190	051C	0002	051E	0002	0600	0002	0601	0002	0800	0002
E1_00200	0801	0002	0802	0002	0803	0002	0804	0004	0806	0004
E1_00210	0808	0004	080A	0002	080B	0002	080C	0002	080D	0002
E1_00220	080E	0002	080F	0002	0810	0002	0811	0002	0812	0002
E1_00230	0813	0002	0814	0004	0816	0002	0817	0002	0818	0002
E1_00240	0819	0004	081B	0002	081C	0002	081D	0002	081E	0002
E1_00250	081F	0002	0820	0004	0822	0004	0000	0000	0000	0000

Read/Write Parameter Format This program stores the Servo Parameters with the following format (read according to the parameter list above) in the specified part of the EM Area.

The parameter number, parameter size, and set value of each parameter are stored in four-word sets.

- +0: Parameter number (hex)
- +1: Parameter size
- +2: Servo Parameter set value (rightmost word)
- +3: Servo Parameter set value (leftmost word)

These four-word sets are stored in the order that the parameters appear in the parameter list.

When the parameter size is 2 bytes (0002 hex), the leftmost word of the set value in "+3: Servo Parameter set value" will be set to 0000 hex.

Word	+0	+1	+2	+3	+4	+5	+6	+7
E1_01000	0000	0002	0010	0000	0001	0002	1002	0000
E1_01008	0002	0002	0000	0000	0003	0002	0002	0000
E1_01016	0004	0002	0200	0000	0005	0002	0000	0000
E1_01024	0100	0002	0050	0000	0101	0002	07D0	0000
E1_01032	0102	0002	0028	0000	0103	0002	012C	0000
E1_01040	0104	0002	0050	0000	0105	0002	07D0	0000
E1_01048	0106	0002	0028	0000	0107	0002	0000	0000
E1_01056	0108	0002	0007	0000	0109	0002	0000	0000
E1_01064	010A	0002	0000	0000	010B	0002	0004	0000
:								

Example: First Part of Default Settings Read from an OMRON W-series Servo Drive

Note All of the data is expressed in hexadecimal.

For example, the data in E1_01024 to E1_01027 indicates parameter Pn100, a parameter size of 2 bytes, and a set value of 0050 hex (80 decimal).

This data format is also used in the sample program when writing or saving data from the CPU Unit's EM Area to the Servo Drive. (The parameter size at the end of the transfer data is 0000 hex, which is the end code.)

Consequently, this program can be used to read the Servo Parameters to the CPU Unit's EM Area to back up the Servo Parameters and those Servo Parameters can be transferred to a new Servo Drive if it is necessary to replace the Servo Drive in the future.

This sample program will start transferring the Servo Parameters when the Servo Parameter Batch Transfer Bit (CIO 510.00) is ON and W303.03 has been turned ON (indicating that axes 1 and 2 are both communicating normally) by the application example shown in *11-3-1 Initial PCU Settings*.

When this program is completed (Servo Parameter Transfer Complete), W500.05 is turned ON for one cycle. If an error occurs while the Servo Parameters are being transferred (Error Flag ON), the transfer will be cancelled at that point and W500.10 will be turned ON for one cycle.

Sample LadderThe Write Servo Parameters, Read Servo Parameters, and Save ServoProgramParameters functions for axis 1 (program steps 000016 to 000033) are the
same as the functions described in *11-2-3 Transferring Servo Parameters*.

The functions for axis 2 (program steps 000034 to 000051) are the same as the functions for axis 1, except for the different I/O words. To prevent duplications, be sure to change the axis 1 output words (CIO 0000 to CIO 0024) to the corresponding axis 2 output words (CIO 0025 to CIO 0049) and change the axis 1 input words (CIO 1000 to CIO 1024) to the corresponding axis 2 input words (CIO 1025 to CIO 1029).



Section 11-3



inued on next page.)



Section 11-3



000019	W400.02 1000.12	W400.06	Axis 1 Write Error End
(000102)	Waiting for Axis 1 Error Axis 1 Write Flag	0	
	W400.04 1522.00		
	Waiting for Axis 1 Communicating Axis 1 Write		
000020	End W400.01 W400.03 W400.06	W400.02	Waiting for Axis 1
(000168)	Execute Axis 1 Axis 1 Write Axis 1 Write	0	write Received
	Parameter Write Received Error End W400.02		
	Waiting for Avis 1		-
	Write Received	W/400.04	Waiting for Axis 1
000021 (000173)		O	Write End
	Axis 1 Write Axis 1 Write Axis 1 Write Received Normal End Error End		
	W400.04		
	Waiting for Axis 1 Write End		
000022	W401: Read Axis 1 Servo Parameter		I
(000178)	W401.00 1522.00 1000.12 1000.14	1.13	Axis 1 READ SERVO
	Start Axis 1 Axis 1 Error Axis 1 Storvo	(]
	1.13	W401.01	Execute Axis 1 Read
	Axis 1 READ SERVO	0	
000000	PARAMETER W401.02 1000.14	W401.03	Axis 1 Read Received
(000185)	Waiting for Axis 1 Servo Parameter Transferring		-
	Axis 1 Read Received	W/401.05	Avia 1 Bood Normal End
000024 (000188)		0	Axis i Neau Normai Enu
	Waiting for Axis Axis 1 Axis 1 Servo Parameter Transferring 1 Read End Error Flag		
000025 (000192)		W401.06	Axis 1 Read Error End
	Waiting for Axis 1 Axis 1 Read Error Flag Received Error Flag		
	W401.04 1522.00		
	Waiting for Axis Axis 1 Communicating 1 Read End		
000026	Waiting for Axis Axis 1 Communicating 1 Read End W401.01 W401.03 W401.06	W401.02	Waiting for Axis 1 Read
000026 (000198)	Waiting for Axis 1 Communicating 1 Read End W401.01 W401.03 W401.06 Execute Axis Axis 1 Read Axis 1 Read Error End	W401.02	Waiting for Axis 1 Read Received
000026 (000198)	Waiting for Axis Axis 1 Communicating 1 Read End W401.01 W401.03 W401.06 Execute Axis Axis 1 Read Error End 1 Read W401.02	W401.02	Waiting for Axis 1 Read Received
000026 (000198)	Waiting for Axis Axis 1 Communicating 1 Read End W401.01 W401.03 W401.06 Execute Axis 1 Read W401.02 Waiting for Axis 1 Waiting for Axis 1	W401.02	Waiting for Axis 1 Read Received
000026 (000198)	Waiting for Axis Axis 1 Communicating 1 Read End W401.01 W401.03 W401.06 Execute Axis 1 Read W401.02 Waiting for Axis 1 Read Received W401.03 W401.05 W401.06	W401.02	Waiting for Axis 1 Read Received Waiting for Axis 1 Read
000026 (000198) 000027 (000203)	Waiting for Axis Axis 1 Communicating 1 Read End W401.01 W401.03 W401.06 Execute Axis Axis 1 Read Axis 1 Read Error End 1 Read W401.02 Waiting for Axis 1 Read Received W401.03 W401.05 W401.06 Avis 1 Read Error End Received Error End Re	W401.02	Waiting for Axis 1 Read Received Waiting for Axis 1 Read End
000026 (000198) 000027 (000203)	Waiting for Axis Axis 1 Communicating 1 Read End W401.01 W401.03 W401.06 Execute Axis Axis 1 Read Axis 1 Read Error End Received W401.02 Waiting for Axis 1 Read Received W401.03 W401.05 W401.06 Axis 1 Read Axis 1 Read Axis 1 Read Error End Normal End W401.04	W401.02	Waiting for Axis 1 Read Received Waiting for Axis 1 Read End
000026 (000198) 000027 (000203)	Waiting for Axis 1 Communicating 1 Read End W401.01 W401.03 W401.06 Execute Axis 1 Read W401.02 W401.02 W401.02 W401.03 W401.05 W401.06 W401.03 W401.05 W401.06 Axis 1 Read Axis 1 Read Error End Axis 1 Read Received W401.04 Axis 1 Read Axis 1 Read Error End Axis 1 Read Error End Axis 1 Read Axis 1 Read Error End	W401.02	Waiting for Axis 1 Read Received Waiting for Axis 1 Read End
000026 (000198) 000027 (000203)	Waiting for Axis Axis 1 Communicating 1 Read End W401.01 W401.03 W401.06 Execute Axis 1 Read W401.02 Waiting for Axis 1 Read Received W401.03 W401.05 W401.06 Axis 1 Read Axis 1 Read Error End Axis 1 Read Error End Axis 1 Read Error End Normal End Waiting for Axis 1 Read End	W401.02	Waiting for Axis 1 Read Received Waiting for Axis 1 Read End

Section 11-3

000028	W402: Save Axis 1 Servo Parameter		
(000208)	W/02 00 1522 00 1000 12 1000 14	1 1/	Axis 1 SAVE SEBVO
		-0	PARAMETER
	Start Axis 1 Axis 1 Axis 1 Error Axis 1 Servo	-	
	1 14	V402 01	Execute Axis 1 Save
		-0	
	Axis 1 SAVE SERVO PARAMETER		
	W402 02 1000 14	V402.03	Axis 1 Save Received
000029 (000215)		0	
. ,	Waiting for Axis 1 Servo Parameter Transferring		
	Received W402.04 1000.12 1000.14	V402.05	Axis 1 Save Normal End
(000030		-0	
	Waiting for Axis 1 Xis 1 Error Axis 1 Servo Parameter Transferring		
000001	W402 02 1000 12 V	V402.06	Axis 1 Save Error End
(000222)		-0	
	Waiting for Axis 1 Error Axis 1 Save		
	W402.04 1522.00		
		-	
	1 Save End		
000000	W402.01 W402.03 W402.06 V	V402.02	Waiting for Axis 1 Save
(000032		-0	Received
	1 Save Received		
	W402.02		
		-	
	Save Received		
000033	W402.03 W402.05 W402.06 V	V402.04	Waiting for Axis 1 Save
(000233)		-0	Ellu
	Received Normal End		
	W402.04		
	Waiting for Axis	1	
	1 Save End		
000034	W410: Save Axis 2 Servo Parameter		
(000238)	W410 00 1522 01 1025 12 1025 14	26.12	Axis 2 WRITE SERVO
		-0	PARAMETER
	Start Axis 2 Axis 2 Axis 2 Error Axis 2 Servo		
	26.12 Transferring V	V410.01	Execute Axis 2 Write
		-0	
	Axis 2 SAVE SERVO PARAMETER		
000025	W410.02 1025.14 V	V410.03	Axis 2 Write Received
(000245)		-0	
	Waiting for Axis 2 Servo Parameter Transferring Axis 2 Save		
000036	W410.04 1025.12 1025.14 V	V410.05	Axis 2 Write Normal End
(000248)		-0	
	Waiting for Axis 2 Error Axis 2 Servo Parameter Transferring 2 Save End		
000037	W410.02 1025.12 V	V410.06	Axis 2 Write Error End
(000252)	Waiting for Axis 2 Error	-0	
	Axis 2 Save Received		
	W410.04 1522.01		
	Waiting for Axis Axis 2 Communicating		
	2 Save End		
	(Continued on next page.)		
		I	

000038 (000258)	W410.01 W410.03 W410.06 W410.02	Waiting for Axis 2 Write Received
()	Execute Axis 2 Servo Parameter Write Received Error End	
	W410.02	
	Waiting for Axis 2 Write Received	
000039	W410.03 W410.05 W410.06 W410.04	Waiting for Axis 2 Write End
(000200)	Axis 2 Write Axis 2 Write Axis 2 Write Received Normal End Error End	
	W410.04	_
	Waiting for Axis 2 Write End	
000040	W411: Read Axis 2 Servo Parameter]
(000200)	W411.00 1522.01 1025.12 1025.14 26.13	Axis 2 READ SERVO PARAMETER
	Start Axis 2 Axis 2 Axis 2 Error Axis 2 Servo Read Communicating Parameter	
	26.13 Transferring W411.01	Execute Axis 2 Read
	Axis 2 READ SERVO PARAMETER	
000041	W411.02 1025.14 W411.03	Axis 2 Read Received
(000273)	Waiting for Axis 2 Servo Parameter Transferring Axis 2 Read Beceived	
000042	W411.04 1025.12 1025.14 W411.05	Axis 2 Read Normal End
(000270)	Waiting for Axis Axis 2 Axis 2 Servo Parameter Transferring 2 Read End Error Flag	
000043	W411.02 1025.12 W411.06	Axis 2 Read Error End
(000202)	Waiting for Axis 2 Axis 2 Read Error Flag	
	W411.04 1522.01	
	Waiting for Axis 2 Communicating 2 Read End	
000044	W411.01 W411.03 W411.06 W411.02	Waiting for Axis 2 Read Received
(000200)	Execute Axis Axis 2 Read Error End 2 Read Received	
	W411.02	
	Waiting for Axis 2 Read Received	
000045	W411.03 W411.05 W411.06 W411.04	Waiting for Axis 2 Read End
(000200)	Axis 2 Read Axis 2 Read Axis 2 Read Error End Received Normal End	
	W411.04	_
	Waiting for Axis 2 Read End	
000046	W412: Save Axis 2 Servo Parameter	7
(000200)		Axis 2 SAVE SERVO PARAMETER
	Start Axis 2 Axis 2 Error Axis 2 Servo Save Communicating Parameter	
	26.14 W412.01	Execute Axis 2 Save
	Axis 2 SAVE SERVO PARAMETER	
000047	W412.02 1025.14 W412.03	Axis 2 Save Received
(Waiting for Axis 2 Servo Parameter Transferring Axis 2 Save Received	
	(Continued on next page.)	

Section 11-3



11-3-3 Switching among Position, Speed, and Torque Control

Overview

The basic program examples introduced in the following sections are combined to perform position, speed, and torque control.

11-2-3 Transferring Servo Parameters (Used to transfer acceleration/deceleration constants for positioning.)

11-2-4 Servo Lock/Unlock

11-2-6 Positioning (Absolute Movement or Relative Movement)

11-2-7 Speed Control

11-2-8 Torque Control

11-2-9 Deceleration Stop or Emergency Stop

This program uses I/O bits to execute the Servo Lock/Unlock function, start controlling an axis (positioning, speed control, or torque control), or switch the control status of an operating axis.

In this example, only axis 1 is controlled.

This application example is normally executed after the initial PCU settings application shown in *11-3-1 Initial PCU Settings*.

The Axis Operating Output/Input Memory Areas are based on the settings of the Common Parameter Area, as shown below. The application example in *11-3-1 Initial PCU Settings* shows a sample program that can be used to make the following settings.

Beginning word of Axis Operating Output Memory Area: CIO 0000

(Axis 1 Operating Output Memory Area: CIO 0000 to CIO 0024)

Beginning word of Axis Operating Input Memory Area: CIO 1000

(Axis 1 Operating Input Memory Area: CIO 1000 to CIO 1024)

This sample program uses the following parts of the CPU Unit's I/O memory.

CPU Bus Unit Area

CIO 1500 to CIO 1524

These word addresses are contained in the CPU Bus Unit Area words allocated to unit number 0.

Data Memory Area (DM Area)

Axis operation command settings	: D1000 to D1011
Units conversion calculation:	D1500 to D1515
Expanded Monitoring:	D1020 to D1023
Axis 1 Servo Parameter transfer:	D10500 to D10503

These words are used to store the axis operation command values, calculate the command's units, store expanded monitoring values, and store the Servo Parameter transfer information (parameter number, parameter size, and two-word transfer data).

Work Area (WR Area)

Servo Lock/Unlock:	W305.00 to W305.06
Positioning:	W201.00 to W201.03
Speed Control:	W202.00 to W202.02
Torque Control:	W203.00 to W203.02
Deceleration Stop:	W204.00 and W204.01
Emergency Stop:	W205.00 and W205.01
Expanded Monitoring Switch:	W210.00 to W210.02
Axis 1 Write Servo Parameters:	W400.00 to W400.06
Change Acceleration/Deceleration	Constant: W501.00 to W501.06

These Work Area bits are used to show the progress of function execution and the execution status.

Holding Area (HR Area)

H1.00

This bit is used as a switch to indicate whether the positioning command is ABSOLUTE MOVEMENT or RELATIVE MOVEMENT.

Timer Area

TIM0003

Used as a watchdog timer to monitor the positioning operation.

I/O Bits

CIO 500.00 to CIO 500.05 and CIO 500.14 to CIO 500.15

The following table shows the axis 1 operations that can be started by this program as well as the I/O bit operations and command values (in the DM Area), which start that axis 1 operation.

Word/bit	Function	Explanation			
CIO 050000	Axis 1 Servo Lock/ Unlock Switch	Each time this bit is turned ON, it switches axis 1 between Servo Lock and Servo Unlock.			
		When the Servo Unlock is executed during speed control or torque control, the Servo Unlock is executed after executing an Emergency Stop.			
CIO 050001	Start Axis 1 Position- ing	When this bit goes from OFF to ON, positioning starts according to the position command value in D1000 and D1001 or the speed command value in D1002 and D1003.			
		The positioning method can be switched between absolute/relative movement by turning H1.00 ON/OFF.			
		H1.00 ON: ABSOLUTE MOVEMENT H1.00 OFF: RELATIVE MOVEMENT			
		Also, the acceleration constant (deceleration constant) can be set to the value in D1010 (D1011) before starting the positioning operation if CIO 0500.14 (CIO 0500.15) is turned ON before turning this bit from OFF to ON.			
		When positioning is started with CIO 0500.14 and/or CIO 0500.15 OFF, position- ing will be performed with the acceleration constant and/or deceleration constant set in the Servo Drive.			
CIO 050002	Start Axis 1 Speed Control	When this bit goes from OFF to ON, speed control starts according to the speed command value in D1002 and D1003.			
CIO 050003	Start Axis 1 Torque Control	When this bit goes from OFF to ON, torque control starts according to the torque command value in D1004 and D1005, with the speed command value in D1002 and D1003 as the speed limit value.			
CIO 050004	Start Axis 1 Deceler- ation Stop	A Deceleration Stop is executed when this bit goes from OFF to ON.			
CIO 050005	Start Axis 1 Emer- gency Stop	An Emergency Stop is executed when this bit goes from OFF to ON.			
CIO 050014	Specify Axis 1 Acceleration Constant	If CIO 0500.01 (Axis 1 Start Positioning) is turned from OFF to ON while this bit is ON, positioning will be executed with the acceleration constant specified in D1010. (Before positioning is started, the new constant is written to the acceler- ation constant Servo Parameter.)			
CIO 050015	Specify Axis 1 Deceleration Con- stant	If CIO 0500.01 (Axis 1 Start Positioning) is turned from OFF to ON while this bit is ON, positioning will be executed with the deceleration constant specified in D1011. (Before positioning is started, the new constant is written to the deceler- ation constant Servo Parameter.)			
D1000 D1001	Axis 1 Position Com- mand Value	These words contain the position command value used in positioning. Set in the "command units."			
D1002 Axis 1 Speed Com- D1003 Mand Value These words contain the speed command v control.		These words contain the speed command value used in positioning and speed control.			
		This value is also used as the speed limit value during torque control.			
		Set in "command units/s."			
		Speed command value for speed control and speed limit value for torque control are calculated in this program from this set value and the Servomotor's momentary maximum rotation speed, encoder resolution, and electronic gear ratio.			
D1004	Axis 1 Torque Com-	This is the torque command value used in torque control.			
mand Value		Set the torque command value as a percentage of the Servomotor's momentary maximum torque, in units of 0.001%.			

Word/bit	Function	Explanation	
D1010 A	Axis 1 Acceleration Constant	This word contains the acceleration constant for positioning.	
		If CIO 0500.14 is ON when positioning is started, the contents of this word are written to the Servo Parameter to change the acceleration rate.	
		Set in units of "×10,000 command units/s ² ."	
D1011	Axis 1 Deceleration Constant	This word contains the deceleration constant for positioning.	
		If CIO 0500.15 is ON when positioning is started, the contents of this word are written to the Servo Parameter to change the deceleration rate.	
		Set in units of "×10,000 command units/s ² ."	

The setting value in D1002 and D1003 is used as the speed command value in position and speed control. In torque control, it is used as the speed limit value.

The PCU's speed command value for speed control and the speed limit value for torque control (option command value 1) are set as a percentage of the Servomotor's momentary maximum rotation speed (units: 0.001%), so they are calculated as follows in this program.

Use the following equation as a template and change the equation in this program when the Servomotor's momentary maximum rotation speed, encoder resolution, or electronic gear ratio are different.

Speed command value for speed control and speed limit value for torque control = Speed command value (command units/s) \times 60 (s) / (Servomotor momentary maximum rotation speed (r/min) \times Encoder resolution (pulses/rotation) \times 4/Electronic gear ratio)/0.001%

An R88M-W10030L W-series Servomotor is used in this sample program. In this case,

Momentary maximum rotation speed:5,000 r/min Encoder resolution: 2,048 pulses/rotation

The electronic gear ratio is contained in Servo Parameters (Pn202/Pn203) and the default setting is 4 (Pn202 = 4 and Pn203 = 1).

Consequently, the units calculation is as follows:

Speed command value for speed control and speed limit value for torque control

= Speed command value (command units/s) \times 60/ (5,000 \times 2,048 \times 4/4) \times 100,000

= Speed command value (command units/s) × 6,000,000 / 10,240,000

In this program, the fraction is reduced to 1,200/2,048 to prevent the multiplication or division from overflowing.

Note When torque control is being performed, option command value 1 is used as the speed limit value, so set parameter Pn002.1 (Speed command input change) to 1 to select option command value 1.

If the Servo Unlock function is executed during speed control or torque control, this program will output the Servo Unlock command after executing an Emergency Stop (see the output circuit for W305.06).

In addition, the feedback speed and torque command are monitored in this program with the PCU's Expanded Monitoring function.

Since the speed and torque units used for monitoring depend on the control mode (position control, speed control, or torque control), the monitoring units are converted in this program and output to D1020 to D1021 and D1022 to D1023. Refer to *4-8-6 Expanded Monitoring* for details on the monitoring units used in each control mode.

In this program, the unit conversion methods are switched at the start of axis operation. The speed monitoring units are converted ($0.001\% \rightarrow$ command units/s) during speed control and the torque monitoring units are converted ($0.001\% \rightarrow \%$) during torque control.

(The conversion of speed monitoring units is performed by reversing the equation used to convert the speed command value for speed control.)

Sample Ladder Program



Section 11-3









000038	W501.01	MOV (021)	[OP1] [OP2]
(000234)	Execute Axis 1 Acceleration Constant	#080B D10500	Axis 1 Transfer Parameter Number
		MOV (021) #0002 D10501	[OP1] [OP2] Axis 1 Transfer Parameter Size
		MOVL (498) D1010 D10502	[OP1] Axis 1 Acceleration Constant [OP2] Axis 1 Transfer Data
000039 (000238)	W501.01 W501.03 Execute Axis Execute Axis 1 1 Acceleration Deceleration	MOV (021) #080E D10500	[OP1] [OP2] Axis 1 Transfer Parameter Number
	Transfer Transfer	MOV (021) #0002 D10501	[OP1] [OP2] Axis 1 Transfer Parameter Size
		MOVL (498) D1011 D10502	[OP1] Axis 1 Deceleration Constant [OP2] Axis 1 Transfer Data
000040 (000243)	W501.01 Execute Axis 1 Acceleration Constant Transfer	MOV (021) D10500 17	[OP1] Axis 1 Transfer Parameter Number [OP2] Axis 1 Servo Parameter Number
	W501.01 W501.03 W501.05 Execute Axis Execute Axis 1 Axis 1 1 Acceleration Deceleration Acceleration Constant Constant	-	(004)
	Transfer Transfer End	MOV (021) D10501 18	[OP1] Axis <u>1 Transfer Parameter Siz</u> e [OP2] Axis 1 Parameter Size
		MOVL (498) D10502 19	[OP1] Axis 1 Transfer Data [OP2] Axis 1 Write Data
		DIFU (013) W400.00	Start Axis 1 Write
000041 (000252)	W501.01 W400.05 Execute Axis 1 Acceleration Axis 1 Write Constant Transfer Normal End	W501.05	Axis 1 Acceleration Constant Transfer End
	W501.01 W501.03 Execute Axis 1 Execute Axis 1 Execute Axis 1 Acceleration Constant Transfer	W501.06	Axis 1 Deceleration Constant Transfer End
000042 (000260)	stant Transfer Constant Transfer W400: Write Axis 1 Servo Parameter W400 00 1522 00 W400 12 1000 14	1 12	Axis 1 WRITE SERVO
	Start Axis 1 Axis 1 Axis 1 Error Axis 1 Servo Write Communicating Parameter Transferring	O	PARAMETER
	Axis 1 WRITE SERVO PARAMETER	vv400.01	LACCUC AXIS I WITH
	(Continued on next page.)		
Application Examples

000043	W400.02	1000.14				W400.03	Axis 1 Write Received
(000267	Waiting for Axis 1 Write	Axis 1 Servo Transferring	Parameter				
000044	W400.04	1000.12	1000.14			 W400.05	Axis 1 Write Normal End
(000270	Waiting for Axis 1 Write End	Axis 1 Error Flag	Axis 1 Servo F Transferring	Parameter		Ũ	
000045	W400.02	1000.12				W400.06	Axis 1 Write Error End
(000274	Waiting for Axis 1 Write	Axis 1 Error Flag				0	
	Received W400.04	1522.00					
	Waiting for Axis 1 Write	Axis 1 Comm	unicating			-	
000046	End W400.01	W400.03	W400.06			W400.02	Waiting for Axis 1 Write
(000280	Execute Axis 1 Servo Parame-	Axis 1 Write	Axis 1 Write			0	noceweu
	ter Write W400.02	neceiveu	Enorend				
	Waiting for Axis	1				-	
000047	Write Received W400.03	W400.05	W400.06			W400.04	Waiting for Axis 1 Write
(000285	Axis 1 Write	Axis 1 Write	Axis 1 Write			0	End
	Received W400.04	Normal End	Error End				
	Waiting for Axis	1				-	
000040	Write End Switch Monito	oring Units					
(000290	CE113	-					[OP1]
						(021) #00B8	[OP2] Axis 1 Monitor Type
	Always ON Fla	ıg				15	
000049	A200.15	W202.01	W203.01			W210.00	Position Control Monitor
(000292	P_First_Cycle_ Task	Axis 1 Speed Control	Axis 1 Torqu Control Com	e mand		0	
	Initial Lask Execution Flag	Command Reception	Reception C	omplete			
	Axis 1 Movement					-	
	Command Recep- tion Complete						
		-				-	
	tion Stop Recep- tion Complete						
	W205.02	-				-	
	Stop Reception Complete						
	W210.00					-	
000050	Position Contro W202.01	W201.01	W203.01	W204.02	W205.02	W210.01	Speed Control Monitor
(000300	Axis 1 Speed Con-	Axis 1	Axis 1 Torque	Axis 1	Axis 1 Emergency	 O	
	W210.01	Command Reception	Control Command Reception	Stop Reception	Stop Reception Complete		
	Speed Control M	Complete Monitor	Complete	Complete		-	
000051	W203.01	W201.01	W202.01	W204.02	W205.02	W210.02	Torque Control Monitor
,000007	Axis 1 Torque Con- trol Command Re-	Axis 1 Movement	Axis 1 Speed Control	Axis 1 Deceleration	Axis 1 Emergency Stop Reception	0	
	W210.02	Reception Complete	Reception Complete	Complete	Complete		
	Torque Control (Continued on	Monitor next page.)					

Application Examples



SECTION 12 Troubleshooting

This section provides information on troubleshooting errors that may occur, including details on the meaning of indicator displays and error codes, and the procedures required to reset errors in the Unit or axes.

12-1	Overvie	ew of PCU Errors	480
12-2	Trouble	shooting Procedure	484
12-3	LED Er	ror Indicators	485
	12-3-1	Error Indicators at Powerup	485
	12-3-2	Error Display for CPU Unit Errors	486
	12-3-3	Error Display for PCU Errors	486
	12-3-4	Error Display for MECHATROLINK Communications Errors	487
	12-3-5	Error Display for PCU Settings and Operations.	487
	12-3-6	Error Display for MECHATROLINK Slave Station Device Errors.	488
12-4	Error C	odes	488
	12-4-1	Overview	488
	12-4-2	List of Error Codes	489
12-5	Trouble	shooting	501
	12-5-1	Initial Troubleshooting	501
	12-5-2	Troubleshooting during PCU Operations	502
	12-5-3	Troubleshooting Communications Errors.	504
12-6	Error R	eset	506
	12-6-1	Resetting Unit Common Errors	506
	12-6-2	Resetting Individual Axis Errors and Warnings	507
	12-6-3	Restarting the PCU	508
12-7	CPU U	nit Error Display	508

12-1 Overview of PCU Errors

The PCU detects errors that occur in the PCU, MECHATROLINK communications, or the MECHATROLINK slave device and sends notification of the error to the CPU Unit.

The errors that are detected by the PCU can be divided broadly into problems common to the PCU and those in each of the controlled axes. The CPU Unit is notified of any errors according to the cause of the detected error and its location, as shown in the following table.

Category	Source of error	Error/ warning			Cause	of error			Notifica-	Interface
			Errors at powerup (initializa- tion errors)	CPU Unit errors	PCU internal errors	MECHA- TROLINK communi- cations errors	PCU set- tings/ opera- tion errors	MECHA- TROLINK slave device errors	tion from PCU	
Unit com- mon	PCU	Error	Causes Unit com- mon error	Causes Unit com- mon error	Causes Unit com- mon error	Causes Unit com- mon error	Causes Unit com- mon error	Does not cause Unit common error	Unit Error Flag, Unit error code	Common Operating Memory Area
Axis errors	PCU	Error	Does not cause axis error	Does not cause axis error	Does not cause axis error	Causes axis error	Causes axis error	Does not cause axis error	(Axis) Error Flag, Axis error	Axis Oper- ating Input Memory Areas
	MECHA- TROLINK slave device	Error	Does not cause axis error	Does not cause axis error	Does not cause axis error	Causes axis error	Causes axis error	Causes axis error	⁻ code	
		Warning	Does not cause axis error	Does not cause axis error	Does not cause axis error	Does not cause axis error	Does not cause axis error	Causes axis error	(Axis) Warning Flag, Axis error code	

Note

- MECHATROLINK communications errors are either related to all communications or occur in individual axes, and are notified either as Unit common errors or as axis errors.
 - (2) The PCU settings/operation errors are caused due to operations either common to the PCU or in each axis, and are notified either as Unit common errors or as axis errors.



The following diagram shows the operation error notification.

 $n = CIO 1500 + (unit number \times 25)$

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. - 1) × 25

If an error occurs, the PCU operates according to the following table depending on the cause of the detected error.

Overview of PCU Errors

The errors common to the PCU can be cleared by setting the UNIT ERROR RESET Bit in the Common Operating Memory Area, and errors that occur in axes are cleared by setting the ERROR RESET Bit in the Axis Operating Output Memory Areas.

Operation			Cause	of error	_		Status a	fter error	ERROR
after error	Errors at powerup (initializa- tion errors)	CPU Unit errors	PCU inter- nal errors	MECHA- TROLINK communi- cations errors	PCU Set- tings/oper- ation errors	MECHA- TROLINK slave device errors	MECHA- TROLINK communi- cations	Servomo- tor carry- ing current	RESET
System stopped	Causes PCU opera- tion on the left	Does not cause PCU operation on the left	Causes PCU opera- tion on the left	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Connection released.	Servomo- tor not car- rying current (See note 2.) (Servo OFF)	Not possi- ble
Connection released suddenly	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Causes PCU opera- tion on the left	Causes PCU opera- tion on the left	Does not cause PCU operation on the left	Connection released.	Servomo- tor not car- rying current (See note 2.) (Servo OFF)	Not possi- ble
Connection released after decel- eration stop	Does not cause PCU operation on the left	Causes PCU opera- tion on the left	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Connection released.	Servomo- tor not car- rying current (See note 2.) (Servo OFF)	Possible only for CPU Unit monitor errors.
Servo OFF	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Causes PCU opera- tion on the left	Connection not released.	Servomo- tor not car- rying current (Servo OFF)	Possible
Emergency stop	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Causes PCU opera- tion on the left	Connection not released.	Maintains previous status	Possible
Decelera- tion stop	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Causes PCU opera- tion on the left	Causes PCU opera- tion on the left	Causes PCU opera- tion on the left	Connection not released.	Maintains previous status	Possible (See note 4.)
Maintains	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Does not cause PCU operation on the left	Causes PCU opera- tion on the left	Causes PCU opera- tion on the left	See note 1.	Maintains previous status	Maintains previous status	Possible

Note

- (1) If a warning occurs in a MECHATROLINK slave device, the Warning Flag will turn ON and operation will continue.
- (2) When a connection is released by the PCU, the Servo Drive will be put in the Servo OFF state.
- (3) The operations and status for MECHATROLINK slave device errors above indicate the PCU operation (i.e., the command sent to the slave device when an error occurs). The operation of the MECHATROLINK slave device in which the error occurred depends on the device's specifications.
- (4) MECHATROLINK communications errors that occur in individual axes cannot be cleared by setting the AXIS ERROR RESET Bit in the Axis Operating Output Memory Areas. (For details, refer to *Errors in MECHA-TROLINK Communications* in this section.)

If two or more errors occur at the same time, the errors are processed in the order of priority starting from the most important error, as shown in the following table, and the error codes are refreshed.

If two or more errors of the same priority level occur at the same time, the error status of the error that was detected first will be maintained (the error code for the error that was detected first will be held).

Category	PCU operation if error occurs	Priority level	Details
Unit common errors	System stopped Connection released sud- denly Connection released after deceleration stop	High	
Individual axis error	MECHATROLINK communi- cations error (See note.) Servo OFF Emergency stop Deceleration stop Operation continues	Low	If a Unit common error occurs while an individ- ual axis error has occurred, the status of the Error Flag and axis error code for the axis error will be held.

Note MECHATROLINK communications errors that occur in individual axes are processed in order of highest priority among the errors that occur in each axis. (For details, refer to *Errors in MECHATROLINK Communications* in this section.)

The PCU functions are limited when an error occurs, as follows:

Error	Functions that can be performed	Functions that cannot be performed
Unit common errors (Unit Error Flag ON in the Common Operating Memory Area)	UNIT ERROR RESET WRITE DATA READ DATA SAVE DATA Releasing connection (CON- NECT Bit OFF) ERROR RESET	Establishing connection (CONNECT Bit ON) ABSOLUTE/RELATIVE
(Error Flag ON in the Axis Operating Input Memory Area)	 SERVO UNLOCK WRITE SERVO PARAME- TER (See note.) READ SERVO PARAME- TER (See note.) SAVE SERVO PARAME- TER (See note.) Switching between Monitor type 1 and 2. For details on MECHA- TROLINK communications errors, refer to <i>Errors in</i> <i>MECHATROLINK Communi- cations</i> in this section. 	MOVEMENT (with INTER- RUPT FEEDING) • ORIGIN SEARCH • ORIGIN RETURN • PRESENT POSITION PRE- SET • JOG • DEVIATION COUNTER RESET • Changing override • DECELERATION STOP • SERVO LOCK • SPEED CONTROL • TORQUE CONTROL • TORQUE CONTROL • DEVICE SETUP • WRITE SERVO PARAME- TER (See note.) • READ SERVO PARAME- TER (See note.) • SAVE SERVO PARAMETER • EMERGENCY STOP (See note.)

Note For PCUs with unit version 1.2 or earlier, parameters cannot be read, written, or saved when there is an axis error. Attempts to execute these commands

will be ignored. For PCUs with unit version 1.3 or later, parameters can be read, written, or saved even when there is an axis error.

Functions that cannot be performed during errors can be performed after executing ERROR RESET to clear the error.

Errors in MECHATROLINK Communications

The following errors are MECHATROLINK communications errors that occur in individual axes.

- Synchronous Communications Alarm (axis error code: 3010 hex)
- Communications Alarm (axis error code: 3011 hex)
- Command timeout (axis error code: 3012 hex)

These errors cannot be cleared using the AXIS ERROR RESET Bit in the Axis Operating Output Memory Areas.

These errors are related to the interface between the PCU and Servo Drive and are, therefore, given the highest priority in error processing of individual axis errors.

If these errors occur, the PCU will execute DECELERATION STOP for the corresponding axis (Servo Drive operation will depend on the stop method that is specified to be used when alarms occur in the Servo Drive) and operation of the axis will be prohibited.

To clear this error, stop MECHATROLINK communications (release connection) and then execute CONNECT again.

PCU Operation when CPU Unit Load Is Turned OFF

When the CPU Unit turns OFF the load, the PCU decelerates the active axis to a stop and releases the MECHATROLINK communications connection.

After recovering from the load OFF status, reestablish the MECHATROLINK communications connection by turning the CONNECT Bit OFF and then ON again.

<u>Warnings</u> The MECHATROLINK slave device warnings that occur indicate that an error will occur if operation is continued without correcting the problem.

The type of warning and detection conditions depend on the connected MECHATROLINK slave device (Servo Drive).

When the PCU detects a warning for the MECHATROLINK slave station device, the PCU notifies the CPU Unit, by turning ON the Warning Flag for the corresponding axis, and stores the warning code in the axis error code.

When a warning is detected, the Unit continues operation unchanged.

When a warning is detected and then the associated error occurs for the same axis, the Error Flag turns ON and the PCU performs error processing.

The axis error code will be overwritten by the error code for the error that has occurred.



Timing Chart for Axis Errors and Warnings

a = Beginning word of Axis Operating Output Areas specified in Common Parameters + (Axis No. -1) \times 25

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

Note If an axis warning occurs (Warning Flag = 1), the PCUs' command response time will be delayed due to the warning processing. The PCU processing time required from receiving a command from the CPU Unit until the command is sent using MECHATROLINK communications is one MECHATROLINK communications cycle if an axis warning has not occurred, and three MECHA-TROLINK communications cycles if an axis warning has occurred. If the machine's operation is affected by the PCU's command response time, clear the axis warning status using ERROR RESET or disable the warnings using the Servo Drive settings.

For details on the PCU's command response time, refer to *Command Response Time* in *Appendix A Performance Characteristics*.

12-2 Troubleshooting Procedure

The basic flow of troubleshooting errors that occur in the PCU is as follows:



12-3 LED Error Indicators

The status of the LED indicators on the front of the PCU can be used to diagnose the following errors.



RUN: The PCU is operating.

ERC: An error has occurred in the PCU.

ERH: An error has occurred in the CPU Unit.

ERM: An error has occurred in the MECHATROLINK device.

MLK: Indicates MECHATROLINK communications status

The indicator displays indicate the first error that is detected.

If the cause of an error has been cleared using ALARM RESET when two or more errors have occurred at the same time, the PCU will detect the next error with the highest priority and the indicator display will switch to show the new error. If multiple errors occur with the same priority level, the error for the smaller axis number will be displayed first.

12-3-1 Error Indicators at Powerup

	Inc	dicator sta	tus		Category	Probable cause of	Countermeasure
RUN	ERC	ERH	ERM	MLK		error	
Not lit	Not lit	Not lit	Not lit	Not lit	CPU Unit power interruption	Power is not being supplied correctly to the CPU Unit.	Check the power sup- ply voltage being sup- plied to the CPU Unit and make sure the cor- rect power is being supplied.
					Unit system error	The PCU system is not operating correctly.	Make sure that the PCU is installed cor- rectly, and turn the power OFF and ON again. If the error occurs again, replace the PCU.
Not lit	Not lit	Lit	Lit Not lit	Not lit	Setting error	The PCU's unit number is the same as another Unit.	After changing the unit number, cycle the power and create the I/ O tables from the CPU Unit.
						The PCU is not regis- tered in the CPU Unit's I/O tables, or the regis- tered status does not match.	Create the I/O tables for the CPU Unit again.
					Data transfer error	The transfer of data between the PCU and CPU Unit could not be executed correctly.	Make sure that the PCU is installed cor- rectly, and turn the power OFF and ON again. If the error occurs again, replace the PCU or CPU Unit.

	Inc	dicator sta	tus		Category	Probable cause of	Countermeasure
RUN	ERC	ERH	ERM	MLK		error	
Not lit	Lit	Not lit	Not lit	Not lit	Unit malfunction	The internal circuits of the PCU have malfunc- tioned.	Replace the PCU.
Not lit	Lit	Lit	Not lit	Not lit	Unit recognition error	The CPU Unit has not recognized the PCU correctly.	Replace the PCU.

12-3-2 Error Display for CPU Unit Errors

	Inc	licator sta	tus		Category	Probable cause of	Countermeasure
RUN	ERC	ERH	ERM	MLK		error	
Lit	Flashing	Lit	Not lit	Not lit	CPU Unit error	The CPU Unit system is not operating cor- rectly. A PLC bus operation error has occurred.	Make sure that the CPU Unit and PCU are installed correctly, and turn the power OFF and ON again. If the error occurs again,
							replace the CPU Unit.
						A fatal error has occurred in the CPU Unit.	Remove the cause of the CPU Unit stopping.

The indicator displays shown in the above table indicate errors that have occurred in the CPU Unit. Check the Unit's installation status and causes for the errors in the CPU Unit.

12-3-3 Error Display for PCU Errors

	Inc	dicator sta	tus		Category	Probable cause of	Countermeasure
RUN	ERC	ERH	ERM	MLK		error	
Lit	Lit	Not lit	Not lit	Not lit	Data corrupted	The data saved in the PCU is corrupted.	Transfer and save the PCU data again, and then restart the Unit or turn the power OFF and ON again.
					Unit error	An error has occurred in the internal circuits of the PCU.	Replace the PCU.

The indicator displays shown in the above table indicate that data in the PCU is corrupted or the PCU's internal circuits have malfunction.

Data may be corrupted if the PCU power is interrupted while data is being saved to the PCU's internal flash memory. Use PCU's SAVE DATA or the CPU Unit's simple backup function to restore (read) data saved in the Memory Card or perform another process to recover saved data.

If the error occurs again, the probable cause is a faulty PCU. Replace the PCU.

12-3-4 Error Display for MECHATROLINK Communications Errors

	Inc	dicator sta	tus		Category	Probable cause of	Countermeasure
RUN	ERC	ERH	ERM	MLK		error	
Lit	Flashing	Not lit	Lit	Undeter- mined	Communica- tions error	MECHATROLINK com- munications cannot be performed correctly.	Check the connection of the MECHA- TROLINK communica- tions cable.
					Scan list mis- match	The MECHATROLINK slave station device corresponding to the axis number regis- tered in the PCU scan list is not connected.	Check whether the set- tings for the MECHA- TROLINK communications line connection or slave device's station address match the set- tings in the scan list.

This display occurs when MECHATROLINK communications are started (connection established), or during communications when MECHATROLINK communications are not connected correctly.

Check for disconnected, broken, or faulty contact in the MECHATROLINK communications cables, and check the status (power interruption, etc.) of the connected MECHATROLINK slave station devices.

Also check whether the actual connection status matches the connections registered in the scan list.

Communications errors, scan list mismatches, or other error types, and the axis in which the error has occurred can be checked using the PCU Error Flags and error codes.

12-3-5 Error Display for PCU Settings and Operations

	Inc	dicator sta	tus		Category	Probable cause of	Countermeasure
RUN	ERC	ERH	ERM	MLK		error	
Lit	Flashing	Not lit	Not lit	Undeter- mined	Illegal operation	An operation command that cannot be exe- cuted has been sent to the PCU.	Check the timing of the last command and change the operation sequence.
					Illegal data	Illegal parameters have been set for the PCU, or an operation com- mand has been sent with illegal parameters.	Check the contents of the last command, and correct the contents of the data.

The indicator display shown in the above table indicates that illegal data is set in the PCU parameters or an illegal operation has been attempted, such as multistart.

This error will occur if a problem has occurred in the use of the PCU, which requires the PCU settings or user program to be corrected.

The error types, and the axis in which the error has occurred can be checked using the PCU Error Flags and error codes.

12-3-6 Error Display for MECHATROLINK Slave Station Device Errors

	Inc	dicator sta	tus		Category	Probable cause of	Countermeasure	
RUN	ERC	ERH	ERM	MLK	error			
Lit	Flashing	Not lit	Flashing	g Lit External sense input		An error has occurred due to external sensor input detection.	Check the machine status and remove the cause of the error.	
					Servo Drive error	The main circuit power of the Servo Drive has been turned OFF.	Check the power sup- ply voltage being sup- plied to the Servo Drive's main circuit power supply and make sure the correct power is being sup- plied.	

The indicator display shown in the above table indicates that the limit input has been detected, Servo Drive's main circuit power supply is OFF, or an error specific to the Servo Drive or another error in the machine or Servo Drive has occurred.

Check whether the cause of the error is a machine error, a faulty sensor, disconnection, or Servo Drive error.

The error types, and the axis in which the error has occurred can be checked using the PCU Error Flags and error codes.

12-4 Error Codes

12-4-1 Overview

When an error occurs in the PCU, an Error Flag is turned ON and an error code is input in the Common Operating Memory Area or the Axis Operating Input Memory Areas.

The area in which the Error Flag is turned ON and the input error code indicate the type of error and in which axis the error occurred.

Common Operating Memory Area

Name	Word	Bit	Contents
Unit Error Flag	n+15	12	0: Unit common error has not occurred. 1: Unit common error has occurred.
Unit error code	n+21		Stores the error code when a Unit common error occurs.

 $n = CIO 1500 + (unit number \times 25)$

Error information in this area indicates errors that occur in the PCU or that are not specific to a particular axis.

The Unit common error code is indicated as $00\Box\Box$ hex.

If the Unit Error Flag turns ON, error processing must be performed for the PCU or the whole system including the PCU.

Axis Operating Input Memory Areas

Name	Word	Bit	Contents
Error Flag	b	12	0: No error 1: Error has occurred
Axis error code	b+4		Stores the error code for the error that occurred in an individual axis.

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

Error information in this area indicates errors that have occurred in individual axes.

The Axis error code is indicated as $3\square\square$ hex or $4\square\square$ hex.

If an Axis Error Flag turns ON, error processing must be performed for the corresponding axis.

Determine the cause of the error using the indicator display and error code, isolating the location of the error as either in the PCU or in the Servo Drive/ Machine.

12-4-2 List of Error Codes

PCU Common Errors

		Indi	cator st	atus		Category	Error Erro		Error Probable	Clearing method	Operation after error
	RUN	ERC	ERH	ERM	MLK		name	code	cause		after error
at powerup	Not lit	Not lit	Not lit	Not lit	Not lit	CPU Unit power interrup- tion	Power interrup- tion		Power is not being supplied correctly to the CPU Unit.	Check the power supply voltage being supplied to the CPU Unit and make sure the correct power is being supplied.	
Errors						Unit sys- tem error	Watch- dog timer operation		The PCU sys- tem is not operating cor- rectly.	Make sure that the PCU is installed correctly, and turn the power OFF and ON again. If the error occurs again, replace the PCU.	System stopped
	Not lit	Not lit	Lit	Not lit	Not lit	Setting error	Unit num- ber error		The PCU's unit number is the same as another Unit.	After changing the unit number, cycle the power and create the I/O tables from the CPU Unit.	System stopped
							Unit rec- ognition incom- plete		The PCU is not registered in the CPU Unit's I/O tables, or the registered sta- tus does not match.	Create the I/O tables for the CPU Unit again.	System stopped
						Data transfer error	Data transfer error		The transfer of data between the PCU and CPU Unit at initialization of the PCU could not be exe- cuted cor- rectly.	Make sure that the PCU is installed correctly, and turn the power OFF and ON again. If the error occurs again, replace the PCU or CPU Unit.	System stopped
	Not lit	Lit	Not lit	Not lit	Not lit	Unit mal- function	Common memory error		The internal circuits of the PCU have malfunc- tioned.	Replace the PCU.	System stopped
	Not lit	Lit	Lit	Not lit	Not lit	Unit rec- ognition error	Unit rec- ognition error		The CPU Unit has not recog- nized the PCU correctly.	Replace the PCU.	System stopped

The above errors for which the RUN indicator is not lit occur when starting PCU operations, such as turning ON the power or restarting the Unit. The PCU system does not start correctly if any of these errors occurs, so the Error Flags and error codes cannot be checked from the CPU Unit.

		Indicator status				Category	Error	Error	Probable	Clearing method	Operation
	RUN	ERC	ERH	ERM	MLK		name	code	cause		after error
PU Unit errors	Lit	Flash- ing	Lit	Not lit	Not lit	CPU Unit error	CPU fatal error	000A	An error caus- ing the CPU Unit to stop has occurred.	Remove the cause of the CPU Unit stopping.	Connec- tion released after decel- eration stop
0							CPU Unit watchdog timer error	000B	The CPU Unit system is not operating cor- rectly.	Make sure that the CPU Unit and PCU are installed correctly, and turn the power OFF and ON again. If the error occurs again, replace the CPU Unit.	Connec- tion released after decel- eration stop
							CPU Unit monitor error	000C	The cyclic refresh from the CPU Unit to the PCU has stopped.	Check the error status of the CPU Unit and per- form appropriate error processing. After restart- ing the cyclic refresh with the CPU Unit, exe- cute PCU's ERROR RESET.	Connec- tion released after decel- eration stop
							Bus error	000D	PLC bus oper- ation error	Make sure that the CPU Unit and PCU are installed correctly, and turn the power OFF and ON again. If the error occurs again, replace the CPU Unit.	Connec- tion released after decel- eration stop
ernal errors	Lit	Lit	Not lit	Not lit	Not lit	Unit error	MLK device error	0026	An error has occurred in the internal circuits of the PCU.	Replace the PCU.	System stopped
PCU int							MLK device ini- tialization error	0030	An error has been detected in the MECHA- TROLINK communica- tions part dur- ing PCU initialization processing.	Check the MECHA- TROLINK communica- tions settings in the Common Parameters, and then restart the Unit or turn the power OFF and ON again. If the error occurs again, replace the PCU.	System stopped
						Data cor- rupted	Memory error	00F1	The data saved in the PCU is cor- rupted.	Transfer and save the PCU data again, and then restart the Unit or turn the power OFF and ON again. If the error occurs again, replace the PCU.	System stopped

		Indi	cator st	atus		Category	Error	Error	Probable	Clearing method	Operation
	RUN	ERC	ERH	ERM	MLK		name	code	cause		after error
INK communications errors	Lit	Flash- ing	Not lit	Lit	Unde- ter- mined	Scan list mismatch	MLK ini- tialization error	0020	The MECHA- TROLINK slave station device corre- sponding to the axis num- ber registered in the PCU scan list is not connected.	Check whether the set- tings for the MECHA- TROLINK communications line connection or slave device's station address match the settings in the scan list, and then exe- cute CONNECT again.	Maintains (Axes that have a connection estab- lished can- not be operated for PCUs with unit version 1.2 or earlier.)
MECHATROL						Communi- cations error	MLK com- munica- tions error	0025	MECHA- TROLINK communica- tions cannot be performed correctly, or two or more MECHA- TROLINK slave station devices are using the same station number.	Check the connection of the MECHATROLINK communications cable. After removing the noise or other the cause pre- venting communications, restart the PCU.	Connec- tion released suddenly

Section 12-4

		Indi	cator st	atus		Category	Error	Error	Probable	Clearing method	Operation
	RUN	ERC	ERH	ERM	MLK		name	code	cause		after error
erations errors	Lit	Flash- ing	Not lit	Not lit	Unde- ter- mined	Illegal operation	Multistart error	0021	An operation command that cannot be executed has been sent to the PCU.	The operation com- mand that was sent can- not be executed. Check the last command tim- ing and change the operation sequence.	Maintains
PCU settings and op						FINS command time moni- toring error	0024	Communica- tions between the computer and PCU were interrupted for more than 5 s after switch- ing the CX- Motion-NCF to an opera- tion monitor window.	Check the communica- tions path between the computer and PLC and remove any problems. The Position Control Unit is set to continue monitoring communica- tions with the computer. Restart the PCU to clear the communications monitoring status.	Connec- tion released after decel- eration stop	
						Illegal data	Write transfer error	0022	An attempt has been made for the PCU to write data to an ille- gal address, or to write data using an illegal data size.	The data transfer for the command cannot be executed. Check the contents of the last command, and correct the data transfer settings.	Maintains
							Read transfer error	0023	An attempt has been made for the PCU to read data from an illegal address, or to read data with an illegal data size.	The data transfer for the command cannot be executed. Check the contents of the last command, and correct the data transfer settings.	Maintains
							Transfer cycle set- ting error	0027	The set value for the transfer cycle set in the PCU's Common Parameters is too small for the number and type of connected MECHA- TROLINK devices or the maximum axis number.	Set and save a transfer cycle set value in the Common Parameters that is suitable for the number and type of con- nected MECHA- TROLINK devices and the maximum axis num- ber, and then restart the PCU.	Connec- tion released suddenly
							Initializa- tion com- mon parame- ter check error	0028	An illegal set value has been detected in the Com- mon Parame- ters during PCU initializa- tion. (See note.)	When this error occurs, the corresponding set- ting in the Common Parameters is set to the default value (0). After executing ERROR RESET, transfer and save the correct Com- mon Parameter setting and restart the PCU.	Maintains
							Data transfer common parame- ter check error	0029	An illegal set value in the Common Parameters was trans- ferred to the PCU using WRITE DATA. (See note.)	The transferred set value is discarded and the set value in the Com- mon Parameters before the transfer is restored. After executing ERROR RESET, transfer the cor- rect Common Parame- ters setting.	Maintains

Note

te An error will occur if the scan list contains more than the maximum number of axes for the CS1W-NC271/NC471 or CJ1W-NC271/NC471.

Individual Axis Errors

		Indi	cator st	atus		Category	Error	Error	Probable	Clearing method	Operation
	RUN	ERC	ERH	ERM	MLK		name	code	cause		after error
MECHATROLINK communications errors	Lit	Flash- ing	Not lit	Lit	Unde- ter- mined	Communi- cations error	Synchro- nous communi- cations alarm	3010	MECHA- TROLINK communica- tions cannot be performed correctly with the corre- sponding axis.	Check the connection of the MECHATROLINK communications cable. After removing the cause preventing com- munications, such as breaks or noise in the connection, execute CONNECT again.	Operation stops according to the method used when the Servo Drive alarm occurs. (The PCU executes DECELER- ATION STOP.) Operation of the cor- respond- ing axis is prohibited.
							Communi- cations alarm	3011	MECHA- TROLINK communica- tions cannot be performed correctly with the corre- sponding axis.	Check the connection of the MECHATROLINK communications cable. After removing the cause preventing com- munications, such as breaks or noise in the connection, execute CONNECT again.	Operation stops according to the method set in the Servo Drive for alarms. (The PCU executes DECELER- ATION STOP.) Operation of the cor- respond- ing axis is prohibited.
							Com- mand tim- eout	3012	No MECHA- TROLINK communica- tions response has been received from the cor- responding axis. When con- nected to an R88D-WN□- ML2 Servo Drive, this error will occur if a Servo Parameter is transferred with an illegal parameter size.	After checking that no error has occurred in the MECHATROLINK device connected to the corre- sponding axis, execute CONNECT again. If this error occurs when transferring a Servo Parameter to the R88D- WNML2 Servo Drive, check whether the Servo Parameter is being transferred with an ille- gal parameter size.	Operation stops according to the method used when the Servo Drive alarm occurs. (The PCU executes DECELER- ATION STOP.) Operation of the cor- respond- ing axis is prohibited.

		Indicator status				Category	Error	Error	Probable	Clearing method	Operation
	RUN	ERC	ERH	ERM	MLK		name	code	cause		after error
nd operation errors	Lit	Flash- ing	Not lit	Not lit	Lit	Illegal operation	Present position unknown error	3030	ABSOLUTE MOVEMENT or ORIGIN RETURN was executed before the ori- gin was estab- lished.	Execute ORIGIN SEARCH or PRESENT POSITION PRESET and after defining the origin, execute the previously unsuccessful command again.	Decelera- tion stop
PCU settings a							Servo unlock error	3040	A command to start the axis was executed while in Servo unlock status.	Execute the SERVO LOCK and then execute the previously unsuc- cessful command again.	The com- mand to start the axis is ignored and the Servo Drive remains in Servo unlock sta- tus.
							Multistart error	3050	An attempt was made to execute two or more of the following com- mands at the same time for the same axis. ABSOLUTE MOVE- MENT, REL- ATIVE MOVE- MENT, ORI- GIN SEARCH, ORIGIN RETURN, PRESET, JOG, SPEED CONTROL, TORQUE CONTROL, OR DEVICE SETUP	After editing the ladder program so that multiple command bits do not turn ON at the same time for the same axis, execute the previously unsuccessful command again.	Decelera- tion stop
									An attempt was made to execute one of the following commands for a busy axis. ORIGIN SEARCH, ORIGIN RETURN, PRESENT POSITION PRESET, JOG, or DEVICE SETUP	After editing the ladder program so that com- mand bits do not turn ON for a busy axis, exe- cute the previously unsuccessful command again.	

		Indi	cator st	atus		Category	Error Error		Probable	Clearing method	Operation after error
	RUN	ERC	ERH	ERM	MLK		name	code	cause		after error
ettings and operation errors	Lit	Flash- ing	Not lit	Not lit	Lit	Illegal data	Position designa- tion error	3060	An attempt was made to execute REL- ATIVE MOVE- MENT using a position com- mand value for the target position that is outside the positioning range.	Edit the position com- mand value to be within the positioning range and execute the com- mand again.	Decelera- tion stop
PCU s							Speed designa- tion error	3061	An attempt was made to execute one of the following commands with a nega- tive value as the speed command value. ABSOLUTE MOVE- MENT, REL- ATIVE MOVE- MENT, ORI- GIN SEARCH, ORIGIN RETURN, and JOG	Edit the speed com- mand value to be within the setting range and execute the command again.	Decelera- tion stop
									An attempt was made to execute ORI- GIN SEARCH with a speed command value of 0.	Edit the speed com- mand value to be within the setting range and execute the command again.	
							Speed control speed designa- tion error	3062	An attempt was made to execute SPEED CON- TROL using a command value that exceeds the speed com- mand range.	Edit the speed com- mand value to be within the setting range and execute the command again.	Decelera- tion stop
							Torque command value error	3063	An attempt was made to execute TORQUE CONTROL using a com- mand value that exceeds the torque command range.	Edit the torque com- mand value to be within the setting range and execute the command again.	Decelera- tion stop

		Indi	cator st	atus		Category	Error	Error	Probable	Clearing method	Operation
	RUN	ERC	ERH	ERM	MLK		name	code	cause		after error
settings and operation errors	Lit	Flash- ing	Not lit	Not lit	Lit	Illegal data	Option command value 1 error	3064	An attempt was made to execute SPEED/ TORQUE CONTROL using a com- mand value that exceeds the command range in option com- mand value 1.	Edit the option com- mand value to be within the setting range and execute the command again.	Decelera- tion stop
POL							Option command value 2 error	3065	An attempt was made to execute SPEED CON- TROL using a command value that exceeds the command range in option com- mand value 2.	Edit the option com- mand value to be within the setting range and execute the command again.	Decelera- tion stop
							Override	3070	An attempt was made to execute the override using an override value outside the setting range.	Edit the override value to be within the setting range and execute the command again.	Decelera- tion stop
							Initializa- tion axis parame- ter check error	3090	An illegal set value has been detected in the Axis Parameters during PCU initialization.	When this error occurs, the corresponding set- ting in the Axis Parame- ters is set to the default value (0). After executing ERROR RESET, transfer the correct Axis Parame- ter.	Decelera- tion stop
							Data transfer axis parame- ter check error	3091	An illegal set value in the Axis Parame- ters was transferred to the PCU using WRITE DATA.	The transferred set value is discarded and the set value in the Axis Parameters before the transfer is restored. After executing ERROR RESET, transfer the cor- rect Axis Parameter.	Decelera- tion stop
							Data setting error	3099	An attempt was made to transfer data for an illegal parameter number and outside the setting range using SERVO PARAMETER TRANSFER.	The transferred set value is discarded and the set value for the Servo Parameter before the transfer is restored. After executing ERROR RESET, transfer the cor- rect Servo Parameter.	Decelera- tion stop
									A function not supported by SMARTSTEP Junior Servo Drives (e.g., torque control commands) was executed	Check the commands for the relevant axis and correct the ladder pro- gram so that only sup- ported functions are used.	

		Indi	cator st	atus		Category	Error	Error	Probable	Clearing method	Operation
	RUN	ERC	ERH	ERM	MLK		name	code	cause		after error
JLINK Slave Station Device Errors	Lit	Flash- ing	Not lit	Flash- ing	Lit	External sensor input	Forward rotation limit input	3000	A forward rotation limit input signal was detected.	After executing ERROR RESET, perform move- ment in the reverse rota- tion direction.	Operation stops according to the Servo Drive's specified stop method for limit inputs. (The PCU executes EMER- GENCY STOP.)
MECHATRO							Reverse rotation limit input	3001	A reverse rotation limit input signal was detected.	After executing ERROR RESET, perform move- ment in the forward rota- tion direction.	Operation stops according to the Servo Drive's specified stop method for limit inputs. (The PCU executes EMER- GENCY STOP.)
							Forward software limit	3002	The forward software limit was reached or exceeded during axis movement.	After checking the posi- tion command value and executing ERROR RESET, execute a movement command to move the axis to a cor- rect position within the software limit range.	Operation stops according to the Servo Drive's specified stop method for limit inputs. (The PCU executes EMER- GENCY STOP.)
							Reverse software limit	3003	The reverse software limit was reached or exceeded during axis movement.	After checking the posi- tion command value and executing ERROR RESET, execute a movement command to move the axis to a cor- rect position within the software limit range.	Operation stops according to the Servo Drive's specified stop method for limit inputs. (The PCU executes EMER- GENCY STOP.)
							Emer- gency stop sig- nal input	3004	The emer- gency stop input signal (an external control input signal) was detected when the Servo was locked or an attempt was made to lock the Servo while the emergency stop input sig- nal was ON.	Turn OFF the emer- gency stop input signal, reset the axis error, and repeat the operation from locking the Servo. (This error will not occur if the EMERGENCY STOP command in the Axis Operating Memory Area is used.)	Servo OFF

Section 12-4

	Indicator status		Category	Error	Error	Probable	Clearing method	Operation												
	RUN	ERC	ERH	ERM	MLK		name	code	cause		after error									
e Station Device Errors	Lit	Flash- ing	Flash- Not I	Flash- Not li	Flash- ing	Flash- ing	Flash- ing	Flash- ing	Flash- ing	Flash- ing	Flash- ing	Flash- Not lit	Flash- ing	Lit	Origin search error	No origin proximity or origin input sig- nal	3020	The origin proximity input signal could not be detected within the range of both limit input sig- nals during an origin search.	Check the origin proxim- ity input signal wiring and the signal's alloca- tion setting in the Servo Parameters. Check that the dog width of the ori- gin proximity input signal is no shorter than the communications cycle.	Operation stops according to the Servo Drive's specified stop method for limit inputs.
MECHATROLINK Slav									After detecting the origin proximity input signal during an origin search opera- tion, a limit input signal was detected before detect- ing the origin input signal.	Check that the origin input signal selection in the PCU's Axis Parame- ters is correct. When the external latch signal is selected as the origin input signal, check the external latch signal wir- ing and the allocation setting in the Servo Parameters.	(The PCU executes EMER- GENCY STOP.)									
							Limit input already ON	3021	The limit input signal in the origin search direction has already been input during a single-direc- tion origin search.	Check the limit input sig- nal wiring for the corre- sponding direction and check the limit input sig- nal's allocation setting in the Servo Parameters.	Emer- gency stop (the opera- tion com- mand is not exe- cuted)									
							Limit input signal ON in both directions	3022	Origin search cannot be executed due to limit input signals being input in both directions.	Check the limit input sig- nal wiring in both direc- tions and check the limit input signal allocation settings in the Servo Parameters.	Emer- gency stop (the opera- tion com- mand is not exe- cuted)									
						Servo Drive error	Drive main cir- cuit OFF error (See note.)	3080	The main cir- cuit power of the Servo Drive has been turned OFF.	Check the power supply voltage being supplied to the Servo Drive's main circuit power supply and make sure the correct power is being supplied.	Servo OFF									
						MECHA- TROLINK device alarm		4000 + Alarm code for each device	The error proce device.	essing depends on the	Operation stops according to the specified method when the Servo Drive alarm occurs. (The PCU executes DECELER- ATION STOP.)									

Note

For PCUs with unit version 1.2 or earlier, driver main circuit OFF errors can be detected regardless of whether the axis is servo-locked. For PCUs with unit version 1.3 or later, driver main circuit OFF errors will be detected only when the axis is servo-locked and will not be detected when the axis is not servo-locked.

Individual Axis Warnings

		Indi	cator st	atus		Category	Error Error		r Probable Clea	Clearing method	Operation
	RUN	ERC	ERH	ERM	MLK		name	code	cause		after error
MECHATROLINK slave station device	Lit	Not lit	Not lit	Not lit	Lit	MECHA- TROLINK device warning		4000 + Warn- ing code for each device	The error proce device.	essing depends on the	Operation continues

MECHATROLINK Device Alarms (Alarm Code 4

When an alarm or warning occurs for one of the connected MECHATROLINK slave station devices, the PCU turns ON the Error Flag or Warning Flag and returns the following error code.

G5-series Servo Drives/G-series Servo Drives

Error code: 4

The numbers in the boxes of the error codes displayed for alarms are given as decimal numbers for the Servo Drives and as hexadecimal numbers for the Position Control Units.

The same numbers are given in the boxes of the error codes displayed for warnings for the Servo Drives and the Position Control Units.

Example: Position deviation overflow alarm (deviation counter overflow)

Servo Drive alarm display: A.24

The PCU's error code in this example is 4018.

Example: Battery warning

Servo Drive warning display: A.92

The PCU's error code in this example is 4092.

For details on alarm display and warning display numbers, and the difference between Servo Drives and Position Control Units, refer to the list in *Appendix D List of Error Codes*.

W-series and SMARTSTEP Junior Servo Drives

Error code: 40 hex

The boxes $(\Box\Box)$ indicate the alarm/display number for the Servo Drive.

When using an R88D-WN□-ML2 W-series Servo Drive (with built-in MECHA-TROLINK-II communications), however, only the upper two digits of the display number's three digits will be displayed.

Example: Position deviation overflow alarm (deviation counter overflow)

R88D-WT W-series Servo Drive:

Servo Drive alarm display: A.D0

The PCU's error code in this example is 40D0 hex.

R88D-WN -ML2 W-series Servo Drive:

Position deviation overflow alarm: A.D00

Servo ON position deviation overflow alarm: A.D01

Servo ON speed limit position deviation overflow alarm: A.D02

With the R88D-WN \Box -ML2 W-series Servo Drives, more detailed information is provided for position deviation overflow alarms, as shown above, but only the upper two digits of the alarm code (40D0 hex) are displayed as the PCU's error code for any of these alarms.

For details on alarm displays and warning displays in Servo Drives, refer to the list in *Appendix D List of Error Codes*.

For details on alarm/warning contents, refer to the operation manual for the Servo Drive being used.

12-5 Troubleshooting

This section provides troubleshooting flowcharts when Servo Drive communications cannot be established or movement is not possible even when movement commands are sent from the ladder program.

12-5-1 Initial Troubleshooting

The following troubleshooting flowchart is for errors that occur during initial checking.

This flowchart can be used to determine whether the PCU is faulty and cannot be started.



12-5-2 Troubleshooting during PCU Operations

The following troubleshooting flowchart is for problems that occur after the PCU has been started (RUN indicator is lit).

Use this flowchart to determine whether the error is a CPU Unit error, a setting or usage problem, or a MECHATROLINK communications error.





12-5-3 Troubleshooting Communications Errors

The following troubleshooting flowchart is for when MECHATROLINK communications errors occur.

Use this flowchart to determine the cause, such as whether the error is due to MECHATROLINK communications settings or installation.





12-6 Error Reset

The PCU indicates Unit common errors and individual axis errors in the Common Operating Memory Area and Axis Operating Input Memory Areas, respectively, using Error Flags and error codes.

These errors can be cleared by removing the cause of the error and executing ERROR RESET.

The methods for resetting Unit common errors and individual axis errors are different.

12-6-1 Resetting Unit Common Errors

Unit common errors are cleared by turning ON () the UNIT ERROR RESET Bit in the Common Operating Memory Area.

Reset Unit common errors when the Unit Busy Flag for the Position Control Unit is OFF. If errors are reset when the Unit Busy Flag is ON, a multistart error (Unit error code: 0021) will occur and it will not be possible to reset Unit common errors.

When clearing Unit common errors, make sure that the UNIT ERROR RESET Bit in the Common Operating Memory Area remains ON until the Unit Busy Flag turns ON.

Common Operating Memory Area (Output)

Name	Word	Bit	Contents
UNIT ERROR RESET Bit	n	00	

 $n = CIO 1500 + (unit number \times 25)$

Common Operating Memory Area (Input)

Name	Word	Bit	Contents
Unit Error Flag	n+15	12	0: Unit common error has not occurred.
			1: Unit common error has occurred.
Unit Busy Flag	n+16	14	0: PCU is not busy 1: PCU is busy
Unit error code	n+21		Stores the error code when a Unit common error occurs.

 $n = CIO 1500 + (unit number \times 25)$

Timing Chart Example for UNIT ERROR RESET



 $n = CIO 1500 + (unit number \times 25)$

12-6-2 Resetting Individual Axis Errors and Warnings

Individual axis errors/warnings are cleared by turning ON (1) the ERROR RESET Bits in the Axis Operating Output Memory Areas.

Reset individual axis errors and warning when the Axis Busy Flag for the Axis is OFF. If errors are reset when the Axis Busy Flag is ON, a multistart error (Axis error code: 3050) will occur and it will not be possible to reset errors/ warnings.

When clearing individual axis errors/warnings, make sure that the ERROR RESET Bit remains ON until the Busy Flag in the Axis Operating Input Memory Areas turns ON.

Axis Operating Output Memory Areas

Name	Word	Bit	Contents
ERROR RESET Bit	а	12	▲ : Resets axis error

a = Beginning word of Axis Operating Output Memory Areas specified in Common Parameters + (Axis No. –1) \times 25

Axis Operating Input Memory Areas

Name	Word	Bit	Contents
Warning Flag	b	11	0: No warning 1: Warning has occurred
Error Flag		12	0: No error 1: Error has occurred
Busy Flag		13	1: Axis busy (AXIS ERROR RESET execut- ing)
Axis error code	b+4		Stores the error code for the error that occurred in an individual axis.

b = Beginning word of Axis Operating Input Memory Areas specified in Common Parameters + (Axis No. –1) \times 25

Timing Chart Example for ERROR RESET



a = Beginning word of Axis Operating Output Memory Areas specified in Common Parameters + (Axis No. -1) \times 25 b = Beginning word of Axis Operating Input Memory Areas specified in Common Parameters + (Axis No. -1) \times 25

12-6-3 Restarting the PCU

For errors that cannot be cleared using ERROR RESET, remove the cause of the error and either cycle the power to the CPU Unit or restart the CPU Bus Unit to enable the PCU to recover from the error.

Name	Word	Details	Read/write
CPU Bus Unit Restart Flag	A50100 to A50115	The CPU Bus Unit is restarted when this flag is set from 0 to 1. Each bit cor- responds to the unit number. Bits 00 to 15: Unit numbers 0 to F	Read/write possible

12-7 CPU Unit Error Display

The CPU Unit monitors the PCU, which is a CPU Bus Unit, as follows:

- Monitors PCU for hardware errors
- Monitors unit number duplication
- Monitors refreshing between the CPU Unit and PCU.

If an error is detected during the above monitoring operations, the corresponding flag in the Auxiliary Area of the CPU Unit will turn ON (for details on each flag, refer to the CPU Unit's operation manual). The flags are as follows:

- CPU Bus Unit Number Duplication Flags (A41000 to A41015)
- CPU Bus Unit Error, Unit Number Flags (A41700 to A41715)
- CPU Bus Unit Setting Error, Unit Number Flags (A42700 to A42715)
- CPU Bus Unit Setting Error Flag (A40203)
- CPU Bus Unit Error Flag (A40207)

Perform error processing according to the tables provided in 12-3 LED Error Indicators and 12-4-2 List of Error Codes.

SECTION 13 Maintenance and Inspection

This section describes methods for inspecting and maintaining the Position Control Unit and the procedure required to replace a Position Control Unit.

13-1	Inspection	510
13-2	Inspection Points	510
13-3	Handling Precautions	511
13-4	Procedure for Replacing a PCU	511
	13-4-1 Procedure for Replacing a PCU	511
	13-4-2 Memory Card Backup	512

13-1 Inspection

Perform daily or periodic inspections to make sure that the PCU continues operating in its optimum condition.

13-2 Inspection Points

The main components of the PCU are semiconductors, which have a long service life. These components may deteriorate faster, however, due to factors such as the operating environment. Therefore, routine inspection is required.

The standard inspection schedule is once every six months to one year. More frequent inspections may be advisable depending on the operating environment.

Maintain the inspection schedule once it has been set.

Routine Inspection Points

No.	Inspection point	Details	Criteria	Remarks
1	I/O Power Sup- ply	Measure the voltage varia- tions at the I/O power supply terminal block. Do they meet the standards?	24 V DC: 21.6 to 26.4 V DC	Using a voltage tester, check between the terminals adjusting to make sure that the power supply is within the acceptable range.
2	Environment conditions	Is the ambient temperature within the acceptable range? (When used in a panel, the ambient temperature of the PCU inside the panel must be checked.)	0 to 55°C	Using a thermometer, check the ambi- ent temperature inside the panel and make sure that it is within the accept- able range.
		Is the ambient humidity (humidity within panel) within the acceptable range? (When used in a panel, the ambient humidity of the PCU inside the panel must be checked.)	10% to 90% (with no con- densation)	Using a hygrometer, check the ambient humidity inside the panel and make sure that it is within the acceptable range. In particular, check that there is no condensation due to sudden changes in temperature.
		Is the PCU exposed to direct sunlight?	The PCU must not be exposed to direct sun- light.	Shield the PCU from direct sunlight.
		Is there any accumulation of dust (including iron dust) or salts?	There must be none of these present.	Remove any accumulation of dust (including iron dust) or salts and protect the PCU against them.
		Is the PCU exposed to any sprays of water, oil, or chemicals?	The PCU must not be exposed to any of these.	Protect the PCU against water, oil, and chemicals.
		Is the PCU installed in a loca- tion subject to corrosive or flammable gases?	The PCU must not be exposed to any of these.	Check for odors by smelling or using a gas sensor.
		Is the PCU installed in a loca- tion subject to shock or vibra- tion?	The amount of shock or vibration must be within the acceptable ranges given in the specifica- tions.	Install a cushion or other device to reduce shock and vibration.
		Is the PCU installed in a loca- tion near any source of noise?	There must be no noise.	Remove the PCU from the noise source or take countermeasures to protect the PCU from noise.
3	Installation and wiring	Is the PCU securely mounted?	The mounting must not be loose.	Lock the sliders securely.
		Are the cable connectors properly inserted and locked?	The mounting must not be loose.	Insert and lock all cable connectors properly.
		Are any external cables dis- connected?	There must be no exter- nal abnormalities.	Do a visual check and connect or replace cables as required.

Required Inspection Tools

The following tools, materials, and equipment are required for performing inspections.

- Voltage tester or digital voltage meter
- Industrial alcohol and a clean cotton cloth

The following measuring devices may be required when performing an inspection.

- Oscilloscope
- Memory recorder
- Thermometer
- Hygrometer

13-3 Handling Precautions

- Turn OFF the power before replacing the PCU.
- If a PCU is found to be faulty and is replaced, also check the new PCU to make sure that there are no errors.
- When returning a faulty PCU for repair, make a detailed record of the PCU's malfunction and take it together with the PCU to your nearest OMRON office or sales representative.
- If a contact is faulty, clean the contact by applying some industrial alcohol to a clean cotton cloth and wiping the surface, and then install the PCU.

13-4 Procedure for Replacing a PCU

Use the following procedure when it is necessary to replace a PCU due to malfunction or other cause. The method for backing up data saved in the PCU's internal flash memory is also provided here.

13-4-1 Procedure for Replacing a PCU

Use the following procedure to replace a PCU.

- *1,2,3...* 1. Make a note of the unit number of the PCU to be replaced.
 - 2. Use the READ DATA Bit to read the parameters saved in the PCU to be replaced to the PLC's DM Area first. Then save the DM Area data to a floppy disk or hard disk using, for example, CX-Programmer. Alternatively, use the CPU Unit's simple backup function or the PCU's backup function to back up (write) the PCU parameter data to the Memory Card installed in the CPU Unit.
 - 3. Turn OFF the power to the CPU Unit.
 - 4. Disconnect the MECHATROLINK-II communications cables that are connected to the Servo Drive.
 - 5. Replace the PCU and reconnect the MECHATROLINK-II communications cables, making sure that they are securely locked.
 - 6. Set the unit number for the PCU.
 - 7. Turn ON the power supply to the CPU Unit.

- 8. Transfer the saved parameters and operating data to the CPU Unit's DM Area using, for example, CX-Programmer, and then transfer them to the PCU using the WRITE DATA Bit. After transferring, save the parameters and operating data to the flash memory. When the PCU's parameter data has been saved to the Memory Card using the CPU Unit's simple backup function or the PCU's backup function, execute the command to restore (read) the data from the Memory Card. Save the backed up parameters to the PCU's flash memory by reading them from the Memory Card.
- 9. Turn OFF the power supply to the CPU Unit, and then turn it ON again.

For details on transferring PCU data to the Memory Card installed in the CPU Unit, refer to the next section *13-4-2 Memory Card Backup*.

Note The above procedure is used when replacing the PCU, and transferring only the parameters saved in the PCU. This procedure is not used to transfer Servo Parameters saved in the Servo Drive.

When replacing the Servo Drive, use the PCU's READ SERVO PARAMETER Bit to read the parameters saved in the Servo Drive to be replaced to the CPU Unit's DM Area first. Then save the DM Area data to a floppy disk or hard disk using, for example, CX-Programmer. After replacing the Servo Drive, transfer the saved Servo Parameters to the CPU Unit's DM Area using, for example, CX-Programmer, and then write them to the Servo Drive's non-volatile memory using the PCU's SAVE SERVO PARAMETER Bit.

13-4-2 Memory Card Backup

The following data can be saved to the PCU's internal flash memory.

- Common Parameters
- Axis Parameters

The PCU can save this data to the Memory Card installed in the CPU Unit by using the CPU Unit's simple backup function or the PCU's backup function. The data saved to the Memory Card can also be read to the PCU's flash memory.

- Note Do not read out data backed up to a Memory Card installed in the CPU Unit from a Position Control Unit with unit version 1.0 to a Position Control Unit with unit version 1.1 or later (for both the simple backup function and the PCU's backup function). If Position Control Unit Ver. 1.0 backup data is restored for Position Control Unit Ver. 1.1 or later, the linear interpolation function will be disabled. To transfer setting data from Position Control Unit Ver. 1.0 to Position Control Unit Ver. 1.1 or later, use the parameter transfer function using the data transfer bits (WRITE DATA, READ DATA, and SAVE DATA) in the Common Operating Memory Area.
- **<u>CPU Unit's Simple</u>** <u>**Backup Function**</u> **CS/CJ-series CPU Units have a simple backup function that enables all the setup data in the CPU Unit for every Unit to be backed up (written) to the Memory Card that is installed in the CPU Unit or restored (read) from the Memory Card at the same time by simply operating a switch on the front of the CPU Unit.**

When replacing the PCU, the time required to reset each Unit can be shortened by reading the backup file.

The CJ1W-NC□71/CS1W-NC□71 PCUs described in this manual support the CS/CJ-series CPU Unit simple backup function.

Note The simple backup function can be used to back up the PCU's Common Parameters and Axis Parameters. The Servo Parameters for each Servo Drive connected to the PCU will not be saved to the Memory Card.
The PCU data backed up using the CPU Unit's simple backup function is stored in the Memory Card under the following filename.

File name: BACKUP

(Note: The boxes indicate the PCU's unit number + 10 hex)

For details on the CPU Unit's simple backup function, refer to *5-2-6 Simple* Backup Function in the CS/CJ-series Programmable Controllers Programming Manual (Cat. No. W394).

Note The CPU Unit's simple backup function is used to backup or restore data for the entire CPU Unit's system at the same time. Changing the system configuration may prevent restoration of the PCU's data. To backup or restore data for a single PCU, use the PCU's backup function as described next.

PCU's BackupThe PCU has a backup function for separately transferring data that has been
saved in the PCU's internal flash memory to the Memory Card installed in the
CPU Unit using the READ BACKUP DATA Bit and WRITE BACKUP DATA Bit
in the Common Operating Memory Area.

The PCU's backup function can be executed while MECHATROLINK communications are stopped (connection released) and the Unit Busy Flag is OFF. If this command is executed during MECHATROLINK communications, the command will be ignored and backup will not be performed.

Writing and reading of backup data starts at the rising edge (f) when the WRITE BACKUP DATA Bit and READ BACKUP DATA Bit turn ON, respectively, in the Common Operating Memory Area.

Common Operating Memory Area (Output)

Name	Word	Bit	Contents
WRITE BACKUP DATA	n+1	06	✓ : Starts writing backup data
READ BACKUP DATA		07	✓ : Starts reading backup data

 $n = CIO 1500 + (unit number \times 25)$

When executing the PCU's backup function, make sure that the WRITE BACKUP DATA Bit or READ BACKUP DATA Bit remains ON until the Unit Busy Flag turns ON.

The Unit Busy Flag is ON while data is being transferred between the PCU and Memory Card. When transferring data is completed, the Unit Busy Flag turns OFF.

Common Operating Memory Area (Input)

Name	Word	Bit	Contents
Memory Card Trans-	n+16	12	0: No Memory Card transfer error
fer Error Flag			1: A Memory Card transfer error has occurred
Unit Busy Flag		14	0: PCU is not busy 1: PCU is busy

 $n = CIO 1500 + (unit number \times 25)$

If an error occurs in transferring data between the PCU and Memory Card when using the PCU's backup function, the data transfer will be cancelled, and the Memory Card Transfer Error Flag in the Common Operating Memory Area will turn ON.

If a Memory Card Transfer Error occurs, execute UNIT ERROR RESET and after the error is reset, execute the backup function again.

The PCU data backed up using the PCU's backup function is stored in the Memory Card under the following filename.

File name: UNIT ... PRM (Note: The boxes indicate the PCU's unit number + 10 hex)

Note

- (1) Execute PCU's READ BACKUP DATA to restore (read) data from the Memory Card to the PCU's internal flash memory. The data restored in the internal flash memory is enabled by restarting the PCU or cycling the power to the CPU Unit.
 - (2) If a Memory Card Transfer Error occurs while using the PCU's backup function, the backup to the Memory Card or restoration of data to the PCU's internal flash memory will not be executed correctly. If such an error occurs, always execute the backup function again, and backup or restore the correct data.

Appendix A Performance Characteristics

The performance characteristics of the CJ1W-NCF71 are described here.

Refer to these characteristics when building applications.

The numerical values for each performance characteristic are obtained using the following machines and configuration.

CPU Unit: CJ1H-CPU H (without an Expansion Rack) Position Control Unit: A single PCU installed on the CPU Rack. Servo Drive: R88D-WT01HL Servo Drive equipped with a FNY-NS115 MECHATROLINK-II I/F Unit

Note The performance values shown in this Appendix will depend on conditions such as the CPU Unit, user program, and other installed Units used. Therefore, these values are for reference purposes only.

Power Up Time

When the PCU is powered up or restarted, the time required for the Unit to complete its initial processing and be ready to receive commands from the CPU Unit is 500 ms max.

The Unit Busy Flag turns ON during initial processing after turning ON the PCU power or restarting the Unit.

Restart completed (Unit reset released)

Reset		
	Unit initial processir	ng
Unit Busy Flag		

Command Response Time

The PCU's command response time is the time from when the CPU Unit (user program) sends a command to the PCU until the Servo Drive actually starts operating in response to the command, and the changes in Servo Drive, such as the present position and statuses are reflected in the CPU Unit's user program.

The command response time depends on the CPU Unit's cycle time, MECHATROLINK's transfer time and communications cycle, and the command processing time of MECHATROLINK slave station device.

Servo Drive's Response Time for Commands from the CPU Unit

The response time starts with refreshing outputs for the cycle in which the command was sent from the CPU Unit and continues until the Servo Drive starts operating in response to the command.



The maximum time from the end refresh of the user program's cycle in which the command was sent until the command is actually received by the PCU is indicated as T_{OUT_REF}:

(CPU cycle time or MECHATROLINK communications cycle, whichever is longer) + communications cycle \times 1

The PCU processing time from when the command is received by the PCU until it is sent using MECHA-TROLINK communications is indicated as T_{NC} :

MECHATROLINK communications cycle \times 1

Note If a warning occurs, PCU retry processing is required, so the value for T_{NC} is the MECHATROLINK communications cycle \times 3.

The time for a command sent from the PCU using MECHATROLINK communications to reach the Servo Drive is indicated as T_{MLK} :

MECHATROLINK transfer cycle × 1

The processing time until execution of the command received by the Servo Drive is indicated as T_{SRV OUT}:

250 μ s (using an R88D-GN \Box -ML2 to send commands relating to interpolation feed, speed control and torque control)

1,250 µs (using an R88D-GN□-ML2 to send commands other than those described above)

625 μ s (using an R88D-WT \Box and FNY-NS115 or the SMARTSTEP Junior)

425 µs (using an R88D-KN□-ML2 or an R88D-WN□-ML2)

The maximum value for the time until the Servo Drive actually starts operating in response to the command sent from the CPU Unit's user program, as indicated by T_{CMD} , is calculated as the total of these values, as follows:

 $T_{CMD(MAX)} = T_{OUT_REF} + T_{NC} + T_{MLK} + T_{SRV_OUT}$

Note T_{CMD} varies depending on the width of T_{OUT_REF}

Response Time to Reflect Servo Drive Status Changes in CPU Unit

The response time starts when status transmitted via MECHATROLINK communications is sampled and continues until that status is refreshed in the CPU Unit's input status.



The processing time for the changes in status detected in the Servo Drive to be sent using MECHATROLINK communications is indicated as $T_{SRV IN}$:

250 µs + transfer cycle (using R88D-GN□-ML2)

625 μs (using an R88D-WT and FNY-NS115 MECHATROLINK-II I/F Unit or the SMARTSTEP Junior)

450 μ s (using an R88D-WN \Box -ML2)

300 μ s (using an R88D-KN \Box -ML2)

The time for the Servo Drive status to reach the PCU via MECHATROLINK communications is indicated as T_{MLK} :

MECHATROLINK transfer cycle × 1

The maximum time from the end refresh of the user program's cycle in which the command was sent until the command is actually received by the PCU is $T_{IN REF}$:

(CPU cycle time or MECHATROLINK communications cycle, whichever is longer) + CPU Unit's cycle time \times 1

The maximum value for the time until changes in the Servo Drive status are reflected in the CPU Unit's user program, as indicated by T_{RES} , is a total of these values, as follows:

 $T_{RES(MAX)} = T_{SRV_IN} + T_{MLK} + T_{IN_REF}$

Note T_{RES} varies depending on the width of T_{IN_REF}

Example:

CPU Unit cycle time: 1 ms

PCU communications setting: Transfer cycle: 1 ms

Communications cycle = 1 (transfer cycle \times 1 = 1 ms)

Connected Servo Drive: W-series Servo Drive + FNY-NS115

The command response time for the above conditions is as follows:

(This example applies when no warning has occurred in the PCU.)

Item	Maximum response time	Calculation method
CPU Unit to PCU	4.625 ms	$T_{CMD(MAX)} = T_{OUT_REF} + T_{NC} + T_{MLK} + T_{SRV_OUT}$ = 2 ms + 1 ms + 1 ms +625 µs
PCU to CPU Unit	3.625 ms	$T_{\text{RES}(\text{MAX})} = T_{\text{SRV}_{\text{IN}}} + T_{\text{MLK}} + T_{\text{IN}_{\text{REF}}}$ $= 625 \mu\text{s} + 1 \text{ms} + 2 \text{ms}$

Note When mounting a Position Control Unit to a CS-series Long-distance Expansion Rack, refer to the graph on I/O refresh time coefficients in the *CS-series CPU Unit Operation Manual* (Cat. No. W339).

Effect on CPU Unit Cycle Time

When a single PCU is mounted, the CPU Unit's cycle time increases with the amount of Servo Drive axes connected to the PCU, as follows:

Model	Number of axes connected	Additional time for each PCU connected
CJ1W-NCF71	1	0.2 ms
	3	0.3 ms
	6	0.4 ms
	16	1.0 ms

External Input Response Time

The response time is as shown below for external input signals, such as limit inputs and interrupt inputs. The following response times indicate the time until the Servo Drive starts operating according to the function corresponding to the input signal.

The command response time is required for the status of the signal that was input to be reflected in the input area of the CPU Unit.

External input signal	Response time	Details
Limit input signal	2 ms max.	Response time until the Servo Drive starts the Servomotor's stop operation according to the setting in Servo Parameter Pn001.1 in response to input of the limit input signal.
Interrupt input signal	3 ms max.	Response time until the interrupt feeding operation starts after input of the interrupt input signal for interrupt feeding.

Note The numerical values are obtained when a W-series Servo Drive is used with a FNY-NS115 MECHA-TROLINK-II I/F Unit.



Difference in Start Times for Simultaneously Started Axes

There is no difference in start times between axes when multiple axes are started at the same time (the bit to start operation turns ON at the same I/O refresh timing).

Appendix B List of Parameters

Common Parameter Area

PCU's internal	Name	e Bit										
address		15 to 12	11 to 08	07 to 04	03 to 00							
1838 hex	Area allocations	Axis Operating Out 0000 hex: No sett 00B0 hex: CIO Ar 00B1 hex: WR Ar 00B2 hex: HR Are 00B3 hex: AR Are 0082 hex: DM Are 0050 to 0059, 005	0000 hex: No setting 00B0 hex: CIO Area 00B1 hex: WR Area 00B2 hex: HR Area 00B3 hex: AR Area 00B2 hex: DM Area 00S0 to 0059, 005A, 005B, 005C: EM Area (5 \Box : \Box = EM Bank No.)									
1839 hex	ţ ,	Beginning word of A	Axis Operating Outpu	ut Memory Area: 000	0 to 7FFF hex							
183A hex		Axis Operating Inpu 0000 hex: No sett 00B0 hex: CIO Ar 00B1 hex: WR Ard 00B2 hex: HR Are 00B3 hex: AR Are 0082 hex: DM Are 0050 to 0059, 005	ut Memory Area desig rea ea ea ea ea ea 5A, 005B, 005C: EM	gnation Area (5⊡: □ = EM B	Bank No.)							
183B hex		Beginning word of A	Axis Operating Input	Memory Area: 0000	to 7FFF hex							
183C hex	Scan list	Axis 2 allocation 00 hex: Axis not use 40 hex: Allocates as Drive.	ed (no allocation) xis to the Servo	Axis 1 allocation 00 hex: Axis not used (no allocation) 40 hex: Allocates axis to the Servo Drive.								
183D hex	† ,	Axis 4 allocation (sa	ame as above)	Axis 3 allocation (sa	ame as above)							
183E hex	† .	Axis 6 allocation (sa	ame as above)	Axis 5 allocation (sa	ame as above)							
183F hex	† I	Axis 8 allocation (sa	ame as above)	Axis 7 allocation (sa	ame as above)							
1840 hex	Į į	Axis 10 allocation (s	same as above)	Axis 9 allocation (sa	ame as above)							
1841 hex	Ĭ	Axis 12 allocation (s	same as above)	Axis 11 allocation (same as above)							
1842 hex	Ĭ	Axis 14 allocation (s	same as above)	Axis 13 allocation (s	same as above)							
1843 hex		Axis 16 allocation (s	same as above)	Axis 15 allocation (same as above)							
1844 hex to 1855 hex	Reserved by the sys- tem.	0	0	0	0							
1856 hex	MECHATROLINK communications set- ting	Transfer cycle 00: 1 ms (default : 01: 1 ms 02: 2 03: 3 ms 04: 4 05: 5 ms 06: 6 07: 7 ms 08: 8 A2 hex: 0.25 ms A5 hex: 0.5 ms	setting) ms ms ms ms	Communications cycle Sets the factor used as the integer multiple for the transfer cycle. Set value: 00 to 20 hex The default setting 00 is the same a when the cycle is set to 3.								
1857 hex		0	0	C2 Master Con- nection 0: No C2 master (default setting) 1: C2 master con- nected	Number of com- munications retries Set value: 0 to 7, F hex							
1858 hex to 185F hex	Reserved by the sys- tem.	0	0	0	0							

Axis Parameter Area

Beginning word of Axis Parameter Area for Axis N: d = 1860 hex + $(N - 1) \times 14$ hex (N = 1 to 16)

					PCU's	s intern	al add	ress (h	exade	cimal)						Data			
Axis 1	Axis 2	Axis 3	Axis 4	Axis 5	Axis 6	Axis 7	Axis 8	Axis 9	Axis 10	Axis 11	Axis 12	Axis 13	Axis 14	Axis 15	Axis 16				
1860	1874	1888	189C	18B0	18C4	18D8	18EC	1900	1914	1928	193C	1950	1964	1978	198C	Input signa	al selection		
																Bit	Function	Settings	
																00 to 07	Interrupt input sig- nal selec- tion	00: Phase Z 01: External latch signal 1 input 02: External latch signal 2 input 03: External latch signal 3 input	
																08 to 15	Origin input sig- nal selec- tion	00: Phase Z 01: External latch signal 1 input 02: External latch signal 2 input 03: External latch signal 3 input	
1861	1875	1889	189D	18B1	18C5	18D9	18ED	1901	1915	1929	193D	1951	1965	1979	198D	Operation	mode select	tion	
																Bit	Function	Settings	
																00 to 03	Reserved by the system.	0	
																04 to 07	Origin search operation	0: Reversal mode 1 1: Reversal mode 2 2: Single-direc- tion mode 3: Reversal mode 3 (See note.)	
																08 to 11	Origin detec- tion method (See note.)	0: Origin prox- imity input signal rever- sal 1: No origin proximity input signal reversal 2: Do not use origin prox- imity input signal	
																12	Origin search direction	0: Forward 1: Reverse	
																13	Origin search preset (See note.)	0: Not set. 1: Set.	
																14 to 15	Reserved by the system.	0	

Appendix B

					PCU's	s interr	al add	ress (h	exade	cimal)							Data		
Axis 1	Axis 2	Axis 3	Axis 4	Axis 5	Axis 6	Axis 7	Axis 8	Axis 9	Axis 10	Axis 11	Axis 12	Axis 13	Axis 14	Axis 15	Axis 16				
1862	1876	188A	189E	18B2	18C6	18DA	18EE	1902	1916	192A	193E	1952	1966	197A	198E	Operation	mode selec	tion	
																Bit	Function	Settings	
																00 to 03	Encoder type	0: Incremental encoder 1: Absolute encoder	
																04 to 07	Reserved by the system.	0	
																08 to 11	Reserved by the system.	0	
																12 to 15	Reserved by the system.	0	
1863 to 1873	1877 to 1887	188B to 189B	189F to 18AF	18B3 to 18C3	18C7 to 18D7	18DB to 18EB	18EF to 18FF	1903 to 1913	1917 to 1927	192B to 193B	193F to 194F	1953 to 1963	1967 to 1977	197B to 198B	198F to 199F	0000 (Reser	ved by the s	ystem.)	

Note The reversal mode 3 setting for the origin search operation, the origin detection method, and the origin search preset can be used only with Position Control Units with unit version 2.0 or later. They cannot be used with Position Control Units with unit version 1.3 or earlier.

Servo Parameter Area

The Servo Parameters listed here can be used when the PCU is used with either an R88D-KN -ML2 G5series Servo Drive, an R88D-GN -ML2 G-series Servo Drive, OMRON R88D-WT W-series Servo Drive equipped with a FNY-NS115 MECHATROLINK-II I/F Unit, an W-series Servo Drive with built-in MECHA-TROLINK-II communications (R88D-WN -ML2), or a SMARTSTEP Junior Servo Drive with Built-in MECHA-TROLINK-II Communications (R7D-ZN -ML2).

For further details on each of the parameters, refer to the operation manuals for OMRON G5-series Servo Drives, G-series Servo Drives and Yaskawa JUSP-NS115 MECHATROLINK-II I/F Unit.

G5-series Servo Drive (R88D-KN - ML2 with Built-in MECHATROLINK-II Communications)

This shows a list of Servo Parameters that can be used by the Position Control Unit of a G5-series Servo Drive (R88D-KN□-ML2) with built-in MECHATROLINK-II communications.

For details on each parameter, refer to the user's manual of the respective G5-series Servo Drives.

• Some parameters are enabled by turning the power supply OFF and then ON again. (Those parameters are indicated in the table.)

After changing these parameters, turn OFF the power supply, confirm that the power supply indicator has gone OFF, and then turn ON the power supply again.

- Do not change the parameters marked "Reserved for manufacturer use" and "Reserved". Do not change the set values marked "Not used" and "Reserved".
- For details about the data attributes, refer to the information below.

A: Enabled at all time.

B: Changes are prohibited during motor operation and issuing of commands.

The timing will fluctuate if a change is made during motor operation and issuing of commands.

C: Enabled after a power reset or executing the CONFIG command in MECHATROLINK-II communication. R: Power reset.

Note that this is not enabled by the CONFIG command in MECHATROLINK-II communication.

Relay area settings

Set the servo parameter numbers in parentheses in the list for the Servo Parameter number Relay area (a+17) when transferring Servo Parameters.

For example, when transferring the Error Counter Overflow Level, set "00E" for the Servo Parameter number Relay area (a+17).

Basic Parameters

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
000 (000)	Rotation Direction Switching	2	 Set the relation between the command direction and the motor rotation direction. 0: With a forward rotation command, the motor rotates CW as viewed from the axis end. 1: With a forward rotation command, the motor rotates CCW as viewed from the axis end. 	1	-	0 to 1	С
001 (001)	Control Mode Selec- tion	2	Select the Servo Drive CONTROL mode. 0 to 5: Switch function 6: Full closing control	0	-	0 to 6	R
002 (002)	Realtime Autotuning Mode Selection	2	 Set the OPERATION mode for realtime autotuning. Disabled Emphasizes stability Emphasizes positioning If there is an unbalanced load on the vertical axis or the like. When friction is large. If there is an unbalanced load on the vertical axis or the like and friction is too large. When the realtime autotuning is customized. 	1	-	0 to 6	В
003 (003)	Realtime Autotuning Machine Rigidity Set- ting	2	Set the machine rigidity for executing realtime autotun- ing.	11/13	-	0 to 31	В
004 (004)	Inertia Ratio	2	Set the load inertia as a percentage of the motor rotor inertia.	250	%	0 to 10000	В
009 (009)	Electronic Gear Ratio Numerator	4	Set the electronic gear ratio. If $Pn009 = 0$, the encoder resolution is set as the	1	-	0 to 2 ³⁰	С
010 (00A)	Electronic Gear Ratio Denominator	4	Electronic Gear Ratio Numerator (Pn009) Electronic Gear Ratio Denominator (Pn010)	1	-	1 to 2 ³⁰	С
013 (00D)	No. 1 Torque Limit	2	Set the No. 1 limit value for the output torque of the motor.	500	%	0 to 500	В
014 (00E)	Error Counter Over- flow Level	4	Set the range of the error counter overflow level. Detec- tion of error counter overflow level error is disabled if the set value is 0.	100000	Com- mand unit	0 to 2 ²⁷	A
015 (00F)	Operation Switch when Using Absolute Encoder	2	 Select the absolute encoder usage method. 0: Used as absolute encoder. 1: Used as an incremental encoder. 2: Used as absolute encoder. (Multi-rotation counter overflows are ignored.) 	1	-	0 to 2	С
016 (010)	Regeneration Resistor Selection	2	 Select the Regeneration Resistor used. 0: Use the Built-in Resistor. Triggering of regeneration overload protection (Alarm No.18) depends on the Built-in Resistor (with approx. 1% duty). 1: Use an External Resistor. The regeneration processing circuit operates and regeneration overload protection (Alarm No.18) is triggered when the operating rate of the Regenera- tion Resistor exceeds 10%. 2: Use an External Resistor. Regeneration overload protection (Alarm No.18) does not operate. 3: No Regeneration Resistor All regeneration power is processed with built-in capacitors. 	0/3	-	0 to 3	C
017 (011)	External Regenera- tion Resistor Setting	2	 Select the type of load ratio calculation for the External Regeneration Resistor. 0: Regeneration load ratio is 100% when operating rate of the External Regeneration Resistor is 10%. 1 to 4: Reserved 	0	-	0 to 4	С

Gain Parameter

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
100 (100)	Position Loop Gain 1	2	Set the position loop gain 1.	320/ 480	0.1/s	0 to 30000	В
101 (101)	Speed Loop Gain 1	2	Set the speed loop gain 1.	180/ 270	0.1Hz	1 to 32767	В
102 (102)	Speed Loop Integral Time Constant 1	2	Set the speed loop integration time constant 1.	210/ 310	0.1ms	1 to 10000	В
103 (103)	Speed Feedback Fil- ter Time Constant 1	2	The speed feedback filter 1 can be set to one of 6 values.	0	-	0 to 5	В
104 (104)	Torque Command Fil- ter Time Constant 1	2	Set the time constant for the torque filter 1.	84/126	0.01m s	0 to 2500	В
105 (105)	Position Loop Gain 2	2	Set the position loop gain 2.	380/ 570	0.1/s	0 to 30000	В
106 (106)	Speed Loop Gain 2	2	Set the speed loop gain 2.	180/ 270	0.1Hz	1 to 32767	В
107 (107)	Speed Loop Integra- tion Time Constant 2	2	Set the speed loop integration time constant 2.	10000	0.1ms	1 to 10000	В
108 (108)	Speed Feedback Fil- ter Time Constant 2	2	The speed feedback filter 2 can be set to one of 6 values.	0	-	0 to 5	В
109 (109)	Torque Command Fil- ter Time Constant 2	2	Set the time constant for the torque filter 2.	84/126	0.01m s	0 to 2500	В
110 (10A)	Speed Feed-forward Amount	2	Set the speed feed-forward amount.	300	0.1%	0 to 1000	В
111 (10B)	Speed Feed-forward Command Filter	2	Set the speed feed-forward filter time constant.	50	0.01m s	0 to 6400	В
112 (10C)	Torque Feed-forward Amount	2	Set the torque feed-forward amount.	0	0.1%	0 to 1000	В
113 (10D)	Torque Feed-forward Command Filter	2	Set the torque feed-forward filter.	0	0.01m s	0 to 6400	В
114 (10E)	Gain Switching Input Operating Mode Selection	2	Execute optimum tuning using the gain switching func- tion. 0: Gain 1 (PI/P switching enabled) 1: Gain 1 and gain 2 switching available	1	-	0 to 1	В
115 (10F)	Switching Mode in Position Control	2	 Select the gain switching condition for position control. It is necessary that Pn114 be set to 1. O: Always gain 1 1: Always gain 2 2: Gain switching command input via MECHA- TROLINK-II communications 3: Torque command change amount 4: Always gain 1 5: Command speed 6: Amount of position error 7: When the position command is received. 8: Positioning completion signal (INP) OFF 9: Actual motor speed 10:Combination of position command input and rota- tion speed 	0	-	0 to 10	В
116 (110)	Gain Switching Delay Time in Position Con- trol	2	Set the delay time for switching from gain 2 to gain 1.	50	0.1ms	0 to 10000	В
117 (111)	Gain Switching Level in Position Control	2	Set the gain switching level.	50	-	0 to 20000	В
118 (112)	Gain Switching Hys- teresis in Position Control	2	Set the hysteresis for gain switching.	33	-	0 to 20000	В
119 (113)	Position Gain Switch- ing Time	2	Set the position gain switching time for gain switching.	33	0.1ms	0 to 10000	В
120 (114)	Switching Mode in Speed Control	2	 Select the gain switching condition for speed control. It is necessary that Pn114 be set to 1. O: Always gain 1 1: Always gain 2 2: Gain switching command input via MECHA- TROLINK-II communications 3: Torque command change amount 4: Speed command change amount 5: Command speed 	0	-	0 to 5	В

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute			
121 (115)	Gain Switching Delay Time in Speed Con- trol	2	Set the delay time for switching from gain 2 to gain 1.	0	0.1ms	0 to 10000	В			
122 (116)	Gain Switching Level in Speed Control	2	Set the gain switching level.	t the gain switching level. 0 - 0						
123 (117)	Gain Switching Hys- teresis in Speed Control	2	Set the hysteresis for gain switching.	0	-	0 to 20000	В			
124 (118)	Switching Mode in Torque Control	2	Select the gain switching condition for torque control. It is necessary that Pn114 be set to 1. 0: Always gain 1 1: Always gain 2 2: Gain switching command input via MECHA- TROLINK-II communications 3: Torque command change amount	0	-	0 to 3	В			
125 (119)	Gain Switching Delay Time in Torque Con- trol	2	Set the delay time for switching from gain 2 to gain 1.	0	0.1ms	0 to 10000	В			
126 (11A)	Gain Switching Level in Torque Control	2	Set the gain switching level.	0	-	0 to 20000	В			
127 (11B)	Gain Switching Hys- teresis in Torque Control	2	Set the hysteresis for gain switching.	0	-	0 to 20000	В			

Damping Suppression Parameters

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
200 (200)	Adaptive Filter Selec- tion	2	Set the operation of the adaptive filter. 0: Disabled 1: One enabled.Frequency limited after adaptation. 2: Two enabled.Frequency limited after adaptation. 3: One enabled.Adaptation performed at all times. 4: Two enabled.Adaptation performed with 1 filter at all times.	0	-	0 to 4	В
201 (201)	Notch 1 Frequency Setting	2	Set the notch frequency of resonance suppression notch filter 1.	5000	Hz	50 to 5000	В
202 (202)	Notch 1 Width Set- ting	2	Set the notch width of the resonance suppression notch filter 1.	2	-	0 to 20	В
203 (203)	Notch 1 Depth Set- ting	2	Set the notch depth of resonance suppression notch fil- ter 1.	0	-	0 to 99	В
204 (204)	Notch 2 Frequency Setting	2	Set the notch frequency of resonance suppression notch filter 2.	5000	Hz	50 to 5000	В
205 (205)	Notch 2 Width Set- ting	2	Set the notch width of the resonance suppression notch filter 2.	2	-	0 to 20	В
206 (206)	Notch 2 Depth Set- ting	2	Set the notch depth of resonance suppression notch fil- ter 2.	0	-	0 to 99	В
207 (207)	Notch 3 Frequency Setting	2	Set the notch frequency of resonance suppression notch filter 3. This is set automatically when an adaptive notch is enabled.	5000	Hz	50 to 5000	В
208 (208)	Notch 3 Width Set- ting	2	Set the notch width of the resonance suppression notch filter 3. This is set automatically when an adaptive notch is enabled.	2	-	0 to 20	В
209 (209)	Notch 3 Depth Set- ting	2	Set the notch depth of resonance suppression notch fil- ter 3. This is set automatically when an adaptive notch is enabled.	0	-	0 to 99	В
210 (20A)	Notch 4 Frequency Setting	2	Set the notch frequency of resonance suppression notch filter 4. This is set automatically when an adaptive notch is enabled.	5000	Hz	50 to 5000	В
211 (20B)	Notch 4 Width Set- ting	2	Set the notch width of the resonance suppression notch filter 4. This is set automatically when an adaptive notch is enabled.	2	-	0 to 20	В
212 (20C)	Notch 4 Depth Set- ting	2	Set the notch depth of resonance suppression notch fil- ter 4. This is set automatically when an adaptive notch is enabled.	0	-	0 to 99	В
213 (20D)	Damping Filter Selection	2	Select the damping filter switching method. 0: Damping filter 1 or 2 enabled 1: Reserved for manufacturer use 2: Reserved for manufacturer use 3: Switch by the position command direction • Forward direction: Damping filter 1 or 3 is enabled. • Reverse direction: Damping filter 2 or 4 is enabled.	0	-	0 to 3	В
214 (20E)	Damping Frequency 1	2	Set the damping frequency 1. The function is enabled if the set value is 10 (= 1 Hz) or greater.	0	0.1Hz	0 to 2000	В
215 (20F)	Damping Filter 1 Set- ting	2	Finely adjust damping control function 1. If torque saturation occurs, lower this setting; to increase responsiveness, raise this setting.	0	0.1Hz	0 to 1000	В
216 (210)	Damping Frequency 2	2	Set the damping frequency 2. The function is enabled if the set value is 10 (= 1 Hz) or greater.	0	0.1Hz	0 to 2000	В
217 (211)	Damping Filter 2 Set- ting	2	Finely adjust damping control function 2. If torque saturation occurs, lower this setting; to increase responsiveness, raise this setting.	0	0.1Hz	0 to 1000	В
218 (212)	Damping Frequency 3	2	Set the damping frequency 3. The function is enabled if the set value is 10 (= 1 Hz) or greater.	0	0.1Hz	0 to 2000	В
219 (213)	Damping Filter 3 Set- ting	2	Finely adjust damping control function 3. If torque saturation occurs, lower this setting; to increase responsiveness, raise this setting.	0	0.1Hz	0 to 1000	В

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
220 (214)	Damping Frequency 4	2	Set the damping frequency 4. The function is enabled if the set value is 10 (= 1 Hz) or greater.	0	0.1Hz	0 to 2000	В
221 (215)	Damping Filter 4 Set- ting	2	Finely adjust damping control function 4. If torque saturation occurs, lower this setting; to increase responsiveness, raise this setting.	0	0.1Hz	0 to 1000	В
222 (216)	Position Command Filter Time Constant	2	Set the time constant of the first-order lag filter for the position command.	0	0.1Hz	0 to 1000	В

Analog Control Parameters

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
312 (30C)	Soft Start Accelera- tion Time	2	Set the acceleration processing acceleration time for speed commands.	0	ms/ motor Maxi- mum speed	0 to 10000	В
313 (30D)	Soft Start Decelera- tion Time	2	Set the deceleration processing acceleration time for speed commands.	0	ms/ motor Maxi- mum speed	0 to 10000	В
314 (30E)	S-curve Accelera- tion/Deceleration Time Setting	2	Set the acceleration/deceleration processing S-curve time for speed commands.	0	ms	0 to 1000	В
317 (311)	Speed Limit Selec- tion	2	 Select the torque command and speed limit value. 0: Limit the speed by the limit set on the Speed Limit Value Setting (Pn321). 1: Limit the speed by the speed limit value (VLIM) via MECHATROLINK-II communications or by the value set by the Speed Limit Value Setting (Pn321). 	0	-	0 to 1	В
321 (315)	Speed Limit Value Setting	2	Set the speed limit value.	50	r/min	0 to 20000	A
323 (317)	External Feedback Pulse Type Selection	2	 Select the external feedback pulse type. 0: 90° phase difference output type 1: Serial communications type (incremental encoder specifications) 2: Serial communications type (absolute encoder specifications) 	0	-	0 to 2	R
324 (318)	External Feedback Pulse Dividing Numerator	4	Set the external feedback pulse dividing numerator.	0	-	0 to 2 ²⁰	R
325 (319)	External Feedback Pulse Dividing Denominator	4	Set the external feedback pulse dividing denominator.	10000	-	1 to 2 ²⁰	R
326 (31A)	External Feedback Pulse Direction Switching	2	Reverse the direction to count the external encoder feed back. 0: Count direction not reversed 1: Count direction reversed	0	-	0 to 1	R
327 (31B)	External Feedback Pulse Phase-Z Set- ting	2	 Set to enable or disable the Phase-Z disconnection detection when an external encoder of 90° phase difference output type is used. O: Phase-Z disconnection detection enabled 1: Phase-Z disconnection detection disabled 	0	-	0 to 1	R
328 (31C)	Internal/External Feedback Pulse Error Counter Over- flow Level	4	Set the threshold for feedback pulse deviation errors.	16000	Com- mand unit	1 to 2 ²⁷	С
329 (31D)	Internal/External Feedback Pulse Error Counter Reset	2	Clear to 0 the feedback pulse error value for each set rotation speed.	0	Rota- tion	0 to 100	С

Interface Monitor Setting Parameters

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
400 (400)	Input Signal Selec- tion 1	4	Set the function and logic for the general-purpose input 1 (IN1).	00949494h	-	0 to 00FFFFFFh	С
401 (401)	Input Signal Selec- tion 2	4	Set the function and logic for the general-purpose input 2 (IN2).	00818181h	-	0 to 00FFFFFFh	С
402 (402)	Input Signal Selec- tion 3	4	Set the function and logic for the general-purpose input 3 (IN3).	00828282h	-	0 to 00FFFFFFh	С
403 (403)	Input Signal Selec- tion 4	4	Set the function and logic for the general-purpose input 4 (IN4).	00222222h	-	0 to 00FFFFFFh	С
404 (404)	Input Signal Selec- tion 5	4	Set the function and logic for the general-purpose input 5 (IN5).	002B2B2Bh	-	0 to 00FFFFFFh	С
405 (405)	Input Signal Selec- tion 6	4	Set the function and logic for the general-purpose input 6 (IN6).	00212121h	-	0 to 00FFFFFFh	С
406 (406)	Input Signal Selec- tion 7	4	Set the function and logic for the general-purpose input 7 (IN7).	00202020h	-	0 to 00FFFFFFh	С

Appendix **B**

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
407 (407)	Input Signal Selec- tion 8	4	Set the function and logic for the general-purpose input 8 (IN8).	002E2E2Eh	-	0 to 00FFFFFFh	С
410 (40A)	Output Signal Selec- tion 1	4	Set the function assignment for the general-purpose output 1 (OUTM1)	00030303h	-	0 to 00FFFFFFh	С
411 (40B)	Output Signal Selec- tion 2	4	Set the function assignment for the general-purpose output 2 (OUTM2)	00020202h	-	0 to 00FFFFFFh	С
416 (410)	Analog Monitor 1 Selection	2	Select the type for analog monitor 1. 0: Motor speed 1: Position command speed 2: Internal position command speed 3: Speed control command 4: Torque command 5: Command position error 6: Encoder position error 7: Full close error 8: Hybrid error 9: P-N voltage 10:Regeneration load ratio 11: Motor load ratio 12:Forward direction torque limit 13:Reverse direction torque limit 14:Speed limit value 15:Inertia ratio 16 to 18: Reserved 19:Encoder temperature 20:Servo Drive temperature 21:Encoder 1-rotation data	0	-	0 to 21	A
417 (411)	Analog Monitor 1 Scale Setting	4	Set the output gain for analog monitor 1.	0	-	0 to 214748364	A
418 (412)	Analog Monitor 2 Selection	2	Select the type for analog monitor 2. The set values for this parameter are the same as Analog Monitor 1 Type (Pn416).	4	-	0 to 21	A
419 (413)	Analog Monitor 2 Scale Setting	4	Select the output gain for analog monitor 2.	0	-	0 to 214748364	A
421 (415)	Analog Monitor Out- put Selection	2	Select the analog monitor output voltage method. 0: Output range is -10 to 10 V 1: Output range is 0 to 10 V 2: Output range is 0 to 10 V (5 V as the center)	0	-	0 to 2	A
431 (41F)	Positioning Comple- tion Range 1	4	Set the allowed number of pulses for the positioning completion range.	300	Com- mand unit	0 to 262144	A
432 (420)	Positioning Comple- tion Condition Selec- tion	2	 Set the judgment conditions for positioning completion output. O: The Positioning completion output 1 becomes on when the positional error is lower than the value set on the Pn431. 1: The Positioning completion output 1 becomes on when there is no position command, and the positional error is lower than the value set on the Pn431. 2: The Positioning completion output 1 becomes on when there is no position command, the zero-speed detection signal is on, and the positional error is lower than the value set on the Pn431. 3. The Positioning completion output 1 becomes on when there is no position command, and the positional error is lower than the value set on the Pn431. 3. The Positioning completion output 1 becomes on when there is no position command, and the positional error is lower than the value set on the Pn431. 4. The ON-state is retained until the Positioning Completion Hold Time (Pn433) elapses. After that, it is turned off or kept to be on, depending on the positional error then. 	0	-	0 to 3	A
433 (421)	Positioning Comple- tion Hold Time	2	Set the positioning completion hold time.	0	1ms	0 to 30000	A
434 (422)	Zero Speed Detec- tion	2	Set the output timing of the Zero speed detection out- put (ZSP) in rotation speed [r/min].	50	r/min	10 to 20000	A
435 (423)	Speed Conformity Detection Range	2	Set the detection range for the speed conformity out- put (VCMP). Set the difference between the speed command and the actual speed.	50	r/min	10 to 20000	A
436 (424)	Rotation Speed for Motor Rotation Detection	2	Set the number of motor rotation for the Motor rotation detection output signal (TGON).	1000	r/min	10 to 20000	A

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
437 (425)	Brake Timing when Stopped	2	Set the operation time for the mechanical brake at stop.	0	1ms	0 to 10000	В
438 (426)	Brake Timing during Operation	2	Set the operation time for the mechanical brake during operation.	0	1ms	0 to 10000	В
439 (427)	Brake Release Speed Setting	2	Set the number of motor rotation to determine a mechanical brake output during rotation.	30	r/min	30 to 3000	В
440 (428)	Warning Output Selection 1	2	 Select the warning type for warning output 1. Output by all types of warnings. Overload warning Excessive regeneration warning Battery warning Fan warning Encoder communications warning Encoder overheating warning Encoder overheating warning Life expectancy warning External encoder error warning 10:External encoder communications error warning 11:Data setting warning 12:Command warning 13:MECHATROLINK-II communications warning 	0	-	0 to 13	A
441 (429)	Warning Output Selection 2	2	Select the warning type for warning output 2. The relationships among the set values for this param- eter are the same as for Warning Output Selection 1 (Pn440).	0	-	0 to 13	A
442 (42A)	Positioning Comple- tion Range 2	4	Set the allowable number of pulses for the second positioning completion range.	10	Com- mand unit	0 to 262144	A

Extended Parameters

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
504 (504)	Drive Prohibition Input Selection	2	 Set the operation to be performed upon forward/ reverse direction drive prohibition input. 0: Enable the Forward and Reverse drive prohibition inputs. 1: Disable the Forward and Reverse drive prohibition inputs. 2: Enable the Forward and Reverse drive prohibition inputs. 	1	-	0 to 2	С
505 (505)	Stop Selection for Drive Prohibition Input	2	 Set the drive conditions during deceleration and after stopping, when the Forward or Reverse Drive Prohibition Inputs are enabled. O: The torque in the drive prohibit direction is disabled, and the dynamic brake is activated. 1: The torque in the drive prohibit direction is disabled, and free-run deceleration is performed. 2: The torque in the drive prohibit direction is disabled, and an emergency stop is performed. 	0	-	0 to 2	С
506 (506)	Stop Selection with Servo OFF	2	 Set the stop operation when the servo is turned OFF. 0, 4: During deceleration: Dynamic brake After stopping: Dynamic brake Error counter: Clear 1, 5: During deceleration: Free-run After stopping: Dynamic brake Error counter: Clear 2, 6: During deceleration: Dynamic brake After stopping: Servo free Error counter: Clear 3, 7: During deceleration: Free-run After stopping: Servo free Error counter: Clear 8: During deceleration: Emergency stop After stopping: Dynamic brake Error counter: Clear 9: During deceleration: Emergency stop After stopping: Servo free Error counter: Clear 9: During deceleration: Emergency stop After stopping: Servo free Error counter: Clear 9: During deceleration: Emergency stop After stopping: Servo free Error counter: Clear 	0	-	0 to 9	В
507 (507)	Stop Selection with Main Power Supply OFF	2	Set the stop operation when the main power supply is turned OFF. The settable values are the same as those on the Pn506.	0	-	0 to 9	В
508 (508)	Undervoltage Alarm Selection	2	 Select either to let the servo off or to stop the alarm when a main power alarm occurs. 0: Bring the Servo-OFF state in accordance with the Pn507 setting. Return to Servo on state by turning on the main power. 1: Main power supply undervoltage (Alarm No.13.1) occurs. Stop the alarm. 	1	-	0 to 1	В
509 (509)	Momentary Hold Time	2	Set the main power supply alarm detection time.	70	1ms	70 to 2000	С
510 (50A)	Stop Selection for Alarm Detection	2	 Select the stopping method at an alarm. O: During deceleration: Dynamic brake After stopping: Dynamic brake 1: During deceleration: Free-run After stopping: Dynamic brake 2: During deceleration: Dynamic brake 3: During deceleration: Free-run After stopping: Servo free 3: During deceleration due to emergency stop alarm: Emergency stop During deceleration: Dynamic brake After stopping: Dynamic brake 5: During deceleration: Dynamic brake 5: During Emergency stop alarm deceleration: Emer- gency stop During deceleration: Free-run After stopping: Dynamic brake 6: During Emergency stop alarm deceleration: Emer- gency stop During deceleration: Free-run After stopping: Dynamic brake 6: During Emergency stop alarm deceleration: Emer- gency stop During deceleration: Spramic brake 7: During Emergency stop alarm deceleration: Emer- gency stop During deceleration: Spramic brake 7: During Emergency stop alarm deceleration: Emer- gency stop During deceleration: Free-run After stopping: Servo free 7: During Emergency stop alarm deceleration: Emer- gency stop During deceleration: Free-run After stopping: Servo free 	0	-	0 to 7	В
511 (50B)	Emergency Stop Torque	2	Set the torque limit for emergency stops.	0	%	0 to 500	В

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
512 (50C)	Overload Detection Level Setting	2	Set the overload detection level.	0	%	0 to 500	A
513 (50D)	Overspeed Detec- tion Level Setting	2	Set the overspeed error detection level.	0	r/min	0 to 20000	A
514 (50E)	Overrun Limit Setting	2	Set the motor over-travel distance for position com- mands.	10	0.1 rota- tion	0 to 1000	A
515 (50F)	Control Input Signal Read Setting	2	Select the cycle to read the control input signals from the four levels. 0: 0.166 ms 1: 0.333 ms 2: 1 ms 3: 1.666 ms	0	-	0 to 3	С
520 (514)	Position Setting Unit Selection	2	Select the setting units of Positioning Completion Range 1 and 2 (Pn431 and Pn442) and of the Error Counter Overflow Level (Pn014). 0: Command unit 1: Encoder unit	0	-	0 to 1	С
521 (515)	Torque Limit Selec- tion	2	Select the method to set the forward and reverse torque limits, and the torque feed forward function during speed control.	1	-	0 to 6	В
522 (516)	No. 2 Torque Limit	2	Set the No. 2 limit value for the motor output torque.	0	%	0 to 500	В
525 (519)	Forward External Torque Limit	2	Set the forward external toque limit when the torque limit switch input is given.	0	%	0 to 500	В
526 (51A)	Reverse External Torque Limit	2	Set the reverse external toque limit when the torque limit switch input is given.	0	%	0 to 500	В
531 (51F)	Axis Number	2	Set the axis number for communication.	1	-	0 to 127	С

Special Parameters 1

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
605 (605)	Gain 3 Effective Time	2	Set effective time of gain 3 of three-step gain switch- ing.	0	0.1ms	0 to 10000	В
606 (606)	Gain 3 Ratio Setting	2	Set gain 3 as a multiple of gain 1.	100	%	50 to 1000	В
607 (607)	Torque Command Value Offset	2	Set offset torque to add to torque command.	0	%	-100 to 100	В
608 (608)	Forward Direction Torque Offset	2	Set the value to add to a torque command for forward direction operation.	0	%	-100 to 100	В
609 (609)	Reverse Direction Torque Offset	2	Set the value to add to a torque command for reverse direction operation.	0	%	-100 to 100	В
610 (60A)	Function Expansion Setting	2	Set the function expansion. The setting contents vary depending on the function.	0	-	0 to 63	В
611 (60B)	Electric Current Response Setting	2	Make fine adjustment on electric current response with default setting as 100%.	100	%	50 to 100	В
614 (60E)	Alarm Detection Allowable Time Set- ting	2	Set the allowable time until stopping when an emer- gency stop is actuated upon alarm detection.	200	ms	0 to 1000	В
615 (60F)	Overspeed Detec- tion Level Setting at Emergency Stop	2	During an emergency stop upon alarm detection, if the motor speed excess this set value, this is an over-speed 2 error.	0	r/min	0 to 20000	A
618 (612)	Power Supply ON Initialization Time	2	Set initialization time after power supply ON to the standard 1.5 s plus some.	0	0.1s	0 to 100	R
623 (617)	Disturbance Torque Compensation Gain	2	Set the compensation gain for disturbance torque.	0	%	-100 to 100	В
624 (618)	Disturbance Observer Filter Set- ting	2	Set the filter time constant for disturbance torque com- pensation.	53	0.01ms	10 to 2500	В
631 (61F)	Realtime Autotuning Estimated Speed Selection	2	 Set the load characteristics estimated speed when realtime autotuning is enabled. 0: Fixes estimated results at the time load estimation becomes stable. 1: Estimates in every minute from the load characteristic changes. 2: Estimates in every second from the load characteristic changes. 3: Estimates the optimum from the load characteristic changes. 	0	-	0 to 3	В
632 (620)	Realtime Autotuing Customization Mode Setting	2	Set the CUSTOMIZATION mode detail for realtime autotuning.	0	-	-32768 to 32767	В
634 (622)	Hybrid Vibration Suppression Gain	2	Set the hybrid vibration suppression gain during full closing control.	0	0.1s	0 to 30000	В
635 (623)	Hybrid Vibration Suppression Filter	2	Set the hybrid vibration suppression filter time con- stant during full closing control.	10	0.01ms	0 to 6400	В
637 (625)	Vibration Detection Threshold	2	Set the vibration detection threshold. If torque vibration that exceeds this setting is detected, the vibration detection warning occurs.	0	0.1%	0 to 1000	В
638 (626)	Warning Mask Set- ting	2	Set the warning detection mask setting. If you set the corresponding bit to 1, the corresponding warning detection is disabled.	4	-	-32768 to 32767	С

Special Parameters 2

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
700 (700)	Default Display	2	Select a data type to display on the 7-segment LED indicator on the front panel. 0: Normal state 1: Mechanical angle 2: Electric angle 3: Cumulative count of MECHATROLINK-II communications errors 4 Rotary switch setting value 5: Cumulative count of encoder communications errors 6: Cumulative count of external encoder communications errors 7: Z-phase counter 8 or over: Unused	0	-	0 to 32767	A
701 (701)	Power-ON Address Display Duration Set- ting	2	Set the time to indicate the node address when the control power is turned on.	0	100ms	0 to 1000	R
703 (703)	Torque Limit Flag Output Setting	2	 Set the condition for torque limit output during torque control. On by the torque limit value including the torque command value. On by the torque limit value excluding the torque command value. 	0	-	0 to 1	A
704 (704)	Backlash Compen- sation Selection	2	 Select to enable or disable the backlash compensation during position control. Set the compensation direction when the compensation is enabled. Disable the backlash compensation. 1: Compensate the backlash at the first forward operation after a Servo ON. 2: Compensate the backlash at the first reverse operation after a Servo ON. 	0	-	0 to 2	С
705 (705)	Backlash Compen- sation Amount	2	Set the backlash compensation amount during position control.	0	Com- mand unit	-32768 to 32767	В
706 (706)	Backlash Compen- sation Time Constant	2	Set the time constant to apply a backlash compensa- tion during position control.	0	0.01ms	0 to 6400	В
710 (70A)	MECHATROLINK-II Communication I/O Monitor Setting	2	 Select whether to reflect the inputs to the I/O monitor of MECHATROLINK-II communications, when either the forward or reverse drive prohibition input is assigned to the input signal and the Drive Prohibition Input Selection (Pn504) is set to 1 (i.e. Disable). Disable the one on the I/O monitor of MECHA-TROLINK-II communications as well. Enable the one on the I/O monitor of MECHA-TROLINK-II communications. 	0	-	0 to 1	A

Special Parameters 3

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
800 (800)	Communications Control	2	Controls the alarms and warnings over the MECHA- TROLINK-II communications.	0	-	-32768 to 32767	С
801 (801)	Soft Limit	2	 Select whether to enable or disable the Soft Limit Function. O: Enable the soft limits on both directions. 1: Disable the forward soft limit, but enable the reverse soft limit. 2: Enable the forward soft limit, but disable the reverse soft limit. 3: Disable the soft limits on both directions. 	0	-	0 to 3	A
803 (803)	Origin Range	2	Set the threshold for detecting the origin in absolute values.	0	-	0 to 250	A
804 (804)	Forward Software Limit	4	Set the forward soft limit.	500000	Com- mand unit	-1073741823 to 1073741823	A
806 (806)	Reverse Software Limit	4	Set the reverse soft limit.	-500000	Com- mand unit	-1073741823 to 1073741823	A

Appendix B

Param- eter No.	Parameter name	Param- eter size	Explanation	Default setting	Unit	Setting range	Data attribute
808 (808)	Absolute Encoder Origin Offset	4	Set the offset volume between the encoder or external encoder position and the mechanical coordinate posi- tion, when an absolute encoder or an absolute exter- nal encoder is used.	0	Com- mand unit	-1073741823 to 1073741823	С
811 (80B)	Linear Acceleration Constant	2	Set the acceleration for positioning.	100	*1	-32768 to 32767	В
814 (80E)	Linear Deceleration Constant	2	Set the deceleration for positioning.	100	*1	-32768 to 32767	В
818 (812)	Position Command FIR Filter Time Con- stant	2	Set the time constant of FIR filter for the position com- mand.	0	0.1ms	0 to 10000	В
820 (814)	Final Distance for External Input Posi- tioning	4	Sets the distance to travel after the latch signal input position is detected during the external input position- ing.	100	0.1ms	-1073741823 to 1073741823	В
822 (816)	Origin Return mode Setting	2	Set the direction for origin return. 0: Positive direction 1: Negative direction	0	-	0 to 1	В
823 (817)	Origin Return Approach Speed 1	2	Set the operating speed for origin returns, from when the origin proximity signal turns on to when it turns off and the latch signal is detected.	50	*2	1 to 32767	В
824 (818)	Origin Return Approach Speed 2	2	Set the operating speed for origin returns, from when the latch signal is detected to when the motor stops at the position after travelling the distance set by Final Distance for Origin Return (Pn825).	5	*2	1 to 32767	В
825 (819)	Final Distance for Origin Return	4	Set the distance from the position where the latch sig- nal is entered to the origin during origin returns.	100	Com- mand unit	-1073741823 to 1073741823	В
836 (824)	Option Monitor Selection 1	2	The Monitor Selection Field of MECHATROLINK-II communications displays the monitoring data that is set on this parameter.	0	-	-32768 to 32767	A
837 (825)	Option Monitor Selection 2	2	The Monitor Selection Field of MECHATROLINK-II communications displays the monitoring data that is set on this parameter.	0	-	-32768 to 32767	A

*1. 10,000 command unit/s

*2. 100 command unit/s

G Series Servo Drive (Built-in MECHATROLINK-II Communication Type R88D-GN_-ML2)

Parameter Tables

Param- eter No.	Parameter name	Param- eter size	Setting	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn000	Reserved		Do not ch	ange.	1				
Pn001	Default Dis- play	2	Selects the the front p	e adata to be displayed on the 7-segment LED display on panel.	0		0 to 4	Online	
			0	Normal status ("" Servo OFF, "00" Servo ON)					
			1	Indicates the machine angle from 0 to FF hex. 0 is the zero position of the encoder. The angle increases when the Servomotor turns forward. The count continues from "0" after exceeding "FF"					
				When using an incremental encoder, the display shows "nF" (not Fixed) until detecting the zero position on the encoder after the control power is turned ON.					
			2	Indicates the electrical angle from 0 to FF hex. 0 is the position where the inductive voltage on the U phase reaches the position peak. The angle increases when the Servomotor turns forward. The count continues from "0" after exceeding "FF".					
			3	Indicates the number (total) of MECHATROLINKII com- munications errors from 0 to FF hex. The communications error count (total) saturates at the maximum of FFFFh. "h" appears only for the lowest byte. The count continues from "00" after exceeding					
				"FF". Note The communications error count (total) is cleared by turning OFF the control power.					
			4	Indicates the setting on the rotary switch (node address value) loaded at startup, in decimal. This value does not change even if the rotary switch is turned after startup.					
			5 to 32767	Reserved (Do not set.)					
Pn002	Reserved		Do not ch	ange.	0				
Pn003	Torque Limit Selec-	2	Selects the tion during	e torque limit function, or the torque feed-forward func- g speed control.	1			Online	Settings can be
	tion		 Torque I For torq speed c 	imit Selection ue control, always select Pn05E. For position control and ontrol, select the torque limit as follows.	1		1 to 5		changed when the axes are stopped
			1	Use Pn05E as the limit value for forward and reverse operations.					(Busy Flag for
			2	Forward: Use Pn05E. Reverse: Use Pn05F.					= 0).
			3	Switch limits by torque limit values and input signals from the network. Limit in forward direction: PCL is OFF = Pn05E, PCL is ON = Pn05F Limit in reverse direction: NCL is OFF = Pn05E, NCL is ON = Pn05F					
			4	Forward: Use Pn05E as limit. Reverse: Use Pn05F as limit. Only in speed control, torque limits can be switched by torque limit values from the network as follows: Limit in forward direction: Use Pn05E command or option command value 1, whichever is smaller. Limit in reverse direction: Use Pn05F command or option command value 2, whichever is smaller.					
			5	Forward: Use Pn05E as limit. Reverse: Use Pn05F as limit. Only in speed control, torque limits can be switched by torque limit values and input signals from the network as follows: Limit in forward direction: PCL is OFF = Pn05E, PCL is ON = Pn05E command or option command value 1, whichever is smaller. Limit in reverse direction: NCL is OFF = Pn05F, NCL is ON = Pn05F command or option command value 2, whichever is a command					

Param- eter No.	Parameter name	Param- eter size	Setting		Explanation			Default setting	Unit	Setting range	Enable setting	Details
Pn003	Torque Limit Selec- tion	2	Note P	CL ON: When eith pin 7) or I Option Fi CL OFF: When bot pin 7) and tions Opti	ner Forward Torc MECHATROLINI eld (P-CL) is ON th Forward Torqu d MECHATROLI	que Limit (CN1 PC KII Communicatio I. Je Limit (CN1 PCI NK-II Communica	CL: ons L: a-	1		1 to 5	Online	Settings can be changed when the axes are stopped
			• Torque F	Feed-forward Functi	on Selection							(Busy Flag for
			1 to 3	Enabled only durin using speed control	ig speed control	. Disabled if not						each axis = 0).
			4 to 5	Always disabled								
Pn004	Drive Pro- hibit Input Selection	2	Sets the finners (CN	unction for the Forw N1 POT: pin 19, NO	ard and Reverse T: pin 20)	e Drive Prohibit		0		0 to 2	Offline	
	Gelection		0	Decelerates and s in the Stop Selection when both POT and When both POT and	tops according to on for Drive Proh nd NOT inputs ar nd NOT inputs a	o the sequence so ibition Input (Pn06 re enabled. re OPEN, the Driv	et 66) ve					
			1	Prohibit Input Erro	r (alarm code 38 T inputs disable)	3) will occur.						
			2	When either POT	When either POT or NOT input becomes OPEN, the Drive Prohibit Input Error (alarm code 38) will occur. rors and warnings for MECHATROLINK-II communica-							
Pn005	Communi- cations Control	2	Controls e tions. Note U at	errors and warnings se with thi]s parametely if using a value	0		0 to 3955	Offline	Always set to 0.			
			Set the Co COM_ER 83) will oc every MEC tively for t Detection purposes.	onsecutive Commun R (bit 8 to 11). The cur when a commu CHATROLINK-II cor he number of the Co Count. The error ar	nications Error D communications nications error, v nmunications cy onsecutive Com nd warning can b	Detection Count in c error (alarm code which is assessed cle, occurs conse munications Error be masked for deb	i e d at ecu- r bug					
			bits 15-	12 bits 11-8	bits 7-4	bits 3-0						
				COM_ERR	MSK COM WARNG	MSK COM ALM						
			• [bits 8-1 Error De	1] COM_ERR (Con tection Count) Setting range: 0 to Consecutive Comm Count = COM_ERF	secutive Commu 15. Junications Error	unications Detection						
			Note Ti nd • [bits 0-3] (MSK Ci	Count = COM_ERR + 2 These bits are debug functions. Set to enable (0) when not debugging. c 0-3] MECHATROLINK-II Communications Alarms Mask K COM ALM) [bit0]0: Communications error (alarm code 83) enabled 1: Communications error (alarm code 83) disabled [bit1]0: Watchdon data error (alarm code 83) enabled								
			• [bits 4-7 (MSK C	1: Watchdoğ data error (alarm code 86) disabled s 4-7] MECHATROLINK-II Communications Warnings Mask (K COM WARNG) [bit4]0: Data setting warning (warning code 94h) disabled 1: Data setting warning (warning code 94h) disabled [bit5]0: Command warning (warning code 95h) enabled 1: Command warning (warning code 95h) disabled [bit6]0: ML-II communications warning (warning code 96h) enabled 1: ML-II communications warning (warning code 96h) disabled								
Pn006	Power ON Address Display Duration Setting	2	Sets the c power is t Note T al	e the duration to display the node address when the control er is turned ON. • The node address display has priority even if there are alarms or warnings at power ON.				30	ms	0 to 1000	Offline	
	Jenny		0 to 6	600 ms								
			7 to 1000	set value × 100 ms	3							

Param- eter No.	Parameter name	Param- eter size	Setting	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn007	Speed monitor (SP) Selec- tion	2	Selects th panel). Note T ai	he output to the Analog Speed Monitor (SP on the front his monitor output has a delay due to filtering. The Oper- ting Direction Setting (Pn043) does not affect this moni- or output. Thus, forward rotation is always positive (+),	3		0 to 11	Online	
			a	nd reverse rotation is always negative (-).					
			0	Actual Servomotor speed: 47 r/min/6 V					
			1	Actual Servomotor speed: 188 r/min/6 V					
			2	Actual Servomotor speed: 750 r/min/6 V					
			3	Actual Servomotor speed: 3000 r/min/6 V					
			4	Actual Servomotor speed: 12000 r/min/6 V					
			5	Command speed: 47 r/min/6 V					
			6	Command speed: 188 r/min/6 V					
			7	Command speed: 750 r/min/6 V					
			8	Command speed: 3000 r/min/6 V					
			9	Command speed: 12000 r/min/6 V					
			10	Outputs the Issuance Completion Status (DEN). 0 V: Issuing					
			11	Outputs the Gain Selection Status					
				0 V: Gain 2 5 V: Gain 1					
Pn008	Torque Monitor (IM) Selection	2	Selects th front pane Note T at tc at 0 1 2 3 4 5	e output to the Analog Torque Monitor (IM on the el) his monitor output has a delay due to filtering. The Oper- ting Direction Setting (Pn043) does not affect this moni- or output. Thus, forward rotation is always positive (+), nd reverse rotation is always negative (-). Torque command: 100%/3 V Position deviation: 31 pulses/3 V Position deviation: 125 pulses/3 V Position deviation: 500 pulses/3 V Position deviation: 2000 pulses/3 V Position deviation: 8000 pulses/3 V	0		0 to 14	Online	
			6 to 10	Beserved					
			11	Torque command: 200%/3 V					
			12	Torque command: 400%/3 V					
			13	Outputs the Issuance Completion Status (DEN). 0 V: Issuing 5 V: Issuance complete					
			14	Outputs the Gain Selection Status. 0 V: Gain 2 5 V: Gain 1					
Pn009	Reserved		Do not ch	ange.	0				
Pn00A	Prohibit Parameter Changes	2	Allows/pro 0	bibits parameter changes via the network. Allows parameter changes from the host controller via	0		0, 1	Online	
	via Network		1	the network. Prohibits parameter changes from the host controller via the network. Attempting to change a parameter via the network when prohibited triggers the Command Warning (warn- ing code 95h).					
Pn00B	Operation Switch When Usina	2	Selects he This para encoder.	ow the an absolute encoder is used. meter is disabled when using an incremental	0		0 to 2	Online	
	Absolute		0	Use as an absolute encoder.					
	⊏ncoder	1 Use an absolute encoder as an		Use an absolute encoder as an incremental encoder.					
			2	Use as an absolute encoder but ignore absolute multi- turn counter overflow alarm (alarm code 41).					

Param- eter No.	Parameter name	Param- eter size	Setting	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn00C	RS-232	2	Sets the b	baud rate for RS-232 communications.	2		0 to 5	Online	
	Baud Rate Setting		0	2,400 bps					
	g		1	4,800 bps					
			2	9,600 bps					
			3	19,200 bps					
			4	38,400 bps					
			5	57,600 bps					
Pn00D	Reserved		Do not ch	ange.	0				
Pn00E	Reserved		Do not ch	ange.	0				
Pn00F	Reserved		Do not ch	ange.	0				
Pn010	Position Loop Gain (RT)	2	Sets the p Increasing shortens Oscillation mum resp	osition loop responsiveness. the gain increases position control responsiveness and stabilization time. n or overshoot will occur if set too high. Adjust for opti- nonsiveness.	400	×0.1 [1/s]	0 to 30000	Online	Settings can be changed when the axes are stopped
Pn011	Speed Loop Gain (RT)	2	Sets the s If the Iner the Servo Increasing but too m Small gain Adjust for	peed loop responsiveness. tia Ratio (Pn020) is set correctly, this parameter is set to motor response frequency. g the gain increases the speed control responsiveness, uch gain may cause oscillating. n may cause overshoot in the speed response. optimum responsiveness.	500	×0.1 Hz	1 to 30000	Online	(Busy Flag for each axis = 0).
Pn012	Speed Loop Integration Time Con- stant (RT)	2	Adjusts th Set a larg Decrease Set 9999 tion value	e speed loop integration time constant. e value for large load inertia. the setting for fast response with small inertia. to stop integration operation while retaining the integra- . A setting of 10000 disables integration.	200	×0.1 ms	1 to 10000	Online	
Pn013	Speed Feedback Filter Time Constant (RT)	2	Sets the t Normally, Increasing reduces it This para Setting (P	ype of speed detection filter time constant. use a setting of 0. g the value reduces the noise of the Servomotor but also s responsiveness. meter is disabled if the Instantaneous Speed Observer 'n027) is enabled.	0		0 to 5	Online	
Pn014	Torque Command Filter Time Constant (RT)	2	Adjusts th mand sec The torqu	e first-order lag filter time constant for the torque com- tion. e filter setting may reduce machine vibration.	80	×0.01 ms	0 to 2500	Online	
Pn015	Speed Feedfor- ward Amount (RT)	2	Sets the s This para required.	speed feed-forward amount. meter is particularly useful when fast response is	300	×0.1 %	0 to 1000	Online	
Pn016	Feed-for- ward Filter Time Con- stant (RT)	2	Sets the t ter.	ime constant for the speed feed-forward first-order lag fil-	100	×0.01 ms	0 to 6400	Online	
Pn017	Reserved		Do not ch	ange.	0				

Param- eter No.	Parameter name	Param- eter size	Setting	E	xplanation		Default setting	Unit	Setting range	Enable setting	Details
Pn018	0 to 6400	2	Sets the p Same fun	oosition loop gain when to ction as Pn010.	using gain 2 switching.		200	×0.1 [1/s]	0 to 30000	Online	Settings can be
Pn019	Speed Loop Gain 2 (RT)	2	Sets the s Same fun	speed loop gain when us ction as Pn011.	sing gain 2 switching.		800	×0.1 Hz	1 to 30000	Online	when the axes are
Pn01A	Speed Loop Integration Time Con- stant 2 (RT)	2	Sets the s switching. Same fun Set 9999 tion value	speed loop integration tir ction as Pn012. to stop integration opera . Setting 10000 disables	me constant when using gai ation while retaining the inte s integration.	in 2 gra-	500	×0.1 ms	1 to 10000	Online	(Busy Flag for each axis = 0).
Pn01B	Speed Feedback Filter Time Constant 2 (RT)	2	Sets the s Same fun When Ins this paran	speed detection filter wh ction as Pn013. Normal tantaneous Speed Obse neter will be disabled.	en using gain 2 switching. ly, use a setting of 0. erver Setting (Pn027) is enal	bled,	0		0 to 5	Online	
Pn01C	Torque Command Filter Time Constant 2 (RT)	2	Sets the f section w Same fun	irst-order lag filter time c hen using gain 2 switchi ction as Pn014.	onstant for the torque comn ng.	nand	100	×0.01 ms	0 to 2500	Online	
Pn01D	Notch Filter 1 Fre- quency	2	Sets the r sion. This filter load.	notch frequency of notch must be matched with th	filter 1 for resonance supplies the resonance frequency of the resonance fr	res- the	1500	Hz	100 to 1500	Online	
			100 to 1499	Filter enabled							
Pn01E	Notch Filter	2	1500 Selects th	Filter disabled e notch width of notch fi	2		2	Online	-		
	Deserved		Normally,	use a setting of 2.	0			-	-		
Phule	Reserved		Do not ch	ange.	age of the Correspondence reter	inor	0	 0/		Online	Sottingo
11020	(RT)	2	tia. Setting [% The inerti the EEPF	b] = (Load inertia / Rotor a ratio estimated during OM every 30 minutes.	inertia) \times 100 realtime autotuning is store	ed in	300	70		Online	can be changed when the axes are stopped
Pn021	Realtime Autotuning Mode Selection	2	Sets the of A setting of on the op Normally, Set to 4 to Gain swite Use a set are a prob	operating mode for realti of 3 or 6 will provide faste eration. Operation, howe erating pattern. use a setting of 1 or 4. o 6 when the Servomoto ching is enabled at set v ting of 7 if operation cha olem.	me autotuning. er response to changes in in ever, may be unstable deper or is used as a vertical axis. alues 1 to 6. unges caused by gain switch	nertia Inding	0		0 to 7	Online	(Busy Flag for each axis = 0).
				Realtime Autotuning	Degree of change in load inertia						
			0 [Disabled							
			1 F	lorizontal axis mode	Almost no change						
			2		Gradual changes						
			3		Sudden changes						
			4 V	ertical axis mode	Almost no change						
			5	Gradual changes							
			7 0 n	Sudden changes 7 Gain switching disable mode							
Pn022	Realtime Autotuning Machine Rigidity Selection	2	Sets the r Increasing If the valu change ra Always st increase t	Sets the machine rigidity for realtime autotuning. Increasing this value increases the responsiveness. If the value is changed suddenly by a large amount, the gain will change rapidly, subjecting the machine to shock. Always start by making small changes in the value, and gradually increase the value while monitoring machine operation.					0 to F	Online	
			Cannot be	e set to 0 when using the	e Parameter Unit.						

Param- eter No.	Parameter name	Param- eter size	Setting		Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn023	Adaptive Filter Selec- tion	2	Enables of The Adap 0 when d Note V lo fo 0 1	or disables the adaptive trive Filter Table Numb isabled. When the Vibration Filte ov-pass filter type (Pnf orcibly set to disabled (Adaptive filter disable Adaptive filter enable Adaptive operation p Adaptive filter enable	e filter. er Display (Pn02F) will be reset to er Selection (Pn024) is set to a 124 = 3 to 5), the adaptive filter is Pn023 = 0). ed. ed. ed. eformed. d. Adaptive operation will not be	0		0 to 2	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn024	Vibration Filter Selec- tion	2	Selects th • Filter ty • Norm Vibrat • Low-p Vibrat • Switchin • No sw • Switchin • No sw • Switchin Selec (Pn02 Selec	performed (i.e., retain ne vibration filter type a pe selection al type: ion frequency setting r pass type: ion frequency setting r ng mode selection ritching: Both 1 and 2 a hing with command dir ts Vibration Frequency B, Pn02C) ts Vibration Frequency D, Pn02E)	and switching mode. ange 10.0 to 200.0 Hz ange 1.0 to 200.0 Hz are enabled ection: 1 in forward direction 2 in reverse direction	0		0 to 5	Offline	
				Filter type	Switching mode					
			۱ 0 1	Normal type	No switching					
			2		Switching with command direction					
			3 l 4	₋ow-pass type	No switching					
			5		Switching with command direction					
Pn025	Normal Mode Auto- tuning Operation Setting	2	Sets the o	operating pattern for no Number of rotations Repeat cycles of 2 otations	Prmal mode autotuning. Rotation direction Forward and Reverse (Alternating)	0		0 to 7	Online	Settings can be changed when the axes are stopped
			1 2		Reverse and Forward (Alternating) Forward only					(Busy Flag for each axis = 0).
			3 4 F	Repeat cycles of single	Forward and Reverse					
			5		(Alternating) Reverse and Forward (Alternating)					
			6		Forward only					
			/		Reverse only					
Pn026	Overrun Limit Set- ting	2	Sets the s command Set to 0 t	Servomotor's allowable d input range. o disable the overrun p	e operating range for the position protective function.	10	×0.1 rota- tion	0 to 1000	Online	
Pn027	Instanta- neous	2	The Insta accuracy,	ntaneous Speed Obse thereby improving res	rver improves speed detection ponsiveness and reducing vibra-	0		0, 1	Online	Settings can be
	Speed Observer Setting (RT)		When the Feedback ter Time This featu	a istopping. b instantaneous speed k Filter Time Constant Constant 2 (Pn01B) ar ure cannot be used with	observer is enabled, both Speed (Pn013) and Speed Feedback Fil- e disabled. h realtime autotuning.					changed when the axes are stopped (Busy Flag for
			0	Disabled		-				each axis = 0).
Pn028	Notch Filter 2 Fre- quency	2	Sets the sion. This para	L ⊆nabled notch frequency of note meter must be matche	ch filter 2 for resonance suppres- d with the resonance frequency of	1500	Hz	100 to 1500	Online	
			the load.	Filter enabled		-				
			1499	Filter disabled		-				

Param- eter No.	Parameter name	Param- eter size	Setting	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn029	Notch Filter 2 Width	2	Selects th Increasing	e notch width of notch filter 2 for resonance suppression. g the value increases the notch width.	2		0 to 4	Online	Settings can be
Pn02A	Notch Filter 2 Depth	2	Selects th sion. Increasing the phase	e notch depth of notch filter 2 for resonance suppres- g this value decreases the notch depth, thereby reducing e lag.	0		0 to 99	Online	when the axes are stopped (Busy
Pn02B	Vibration Frequency 1	2	Sets the vibration a Measure The frequithe Vibrat • Normal Setting f 0 to 99) • Low-pas Setting f to 9)	ibration frequency 1 for damping control to suppress at the end of the load. and set the frequency of the vibration. ency setting range depends on the filter type selected in ion Filter Selection (Pn024). type requency range: 10.0 to 200.0 Hz (Disabled when set to ss type frequency range: 1.0 to 200.0 Hz (Disabled when set to 0	0	×0.1 Hz	0 to 2000	Online	each axis = 0).
Pn02C	Vibration Filter 1 Set- ting	2	When set torque sa faster. Normally, The settir tion Filter the range Note T a • Normal Setting • Low-pas Setting	ting Vibration Frequency 1 (Pn02B), reduce this setting if uration occurs, or increase it to make the movement use a setting of 0. Ig range depends on the filter type selected in the Vibra-Selection (Pn024), and if Vibration Filter 1 is enabled, s are as follows: his parameter is disabled when Vibration Filter 1 is disbled. type range: $100 \le Pn02B + Pn02C \le Pn02B \times 2$ or 2000 ss type range: $10 \le Pn02B + Pn02C \le Pn02B \times 6$	0	×0.1 Hz	-200 to 2000	Online	
Pn02D	Vibration Frequency 2	2	Same fun	ction as Pn02B.	0	×0.1 Hz	0 to 2000	Online	
Pn02E	Vibration Filter 2 Set- ting	2	Same fun	ction as Pn02C.	0	×0.1 Hz	-200 to 2000	Online	
Pn02F	Adaptive Filter Table Number Display	2	Displays t the adapt This para enabled (a value of When the EEPROM enabled ti operation value. To clear ti the adapt to 0, and 0 to 4 5 to 48 49 to 64	he table entry number corresponding to the frequency of ve filter. meter is set automatically when the adaptive filter is .e., when the Adaptive Filter Selection (Pn023) is set to her than 0), and cannot be changed. adaptive filter is enabled, this parameter will be saved in approximately every 30 min. If the adaptive filter is ne next time the power supply is turned ON, adaptive will start with the data saved in EEPROM as the default his parameter and reset the adaptive operation, disable ve filter by setting the Adaptive Filter Selection (Pn023) then enable it again. Filter disabled Filter enabled Enable or disable the filter with Pn022	0		0 to 64	Online	This is a read-only parame- ter. You cannot change the set- ting.
Pn030	Gain Switching Operating Mode Selection (RT)	2	Enables of 0	r disables gain switching. Disabled. Uses Gain 1 (Pn010 to Pn014). PI/P operation is switched from MECHATROLINK-II. The gain is switched between Gain 1 (Pn010 to Pn014) and Gain 2 (Pn018 to Pn01C).	1		0, 1	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).

Param- eter No.	Parameter name	Param- eter size	Setting	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn031	Gain Switch	2	Sets the t	rigger for gain switching.	2		0 to 10	Online	Settings
	Setting (RT)		The detai	Is depend on the control mode.					can be changed
			0	Always Gain 1					when the
			1	Always Gain 2					axes are stopped
			2	Switching from the network					(Busy
			3	Amount of change in torque command					each axis
			4	Always Gain 1					= 0).
			5	Speed command					
			6	Amount of position deviation					
			7	Position command pulses received					
			8	Positioning Completed Signal (INP) OFF					
			9	Actual Servomotor speed					
			10	Combination of position command pulses received and speed					
Pn032	Gain Switch Time (RT)	2	Enabled v 10. Sets t switching	when the Gain Switch Setting (Pn031) is set to 3, or 5 to he lag time from the trigger detection to actual gain when switching from gain 2 to gain 1.	30	×166 μs	0 to 10000	Online	
Pn033	Gain Switch Level Set- ting (RT)	2	Sets the j when the The unit f Switch Se	udgment level to switch between Gain 1 and Gain 2 Gain Switch Setting (Pn031) is set to 3, 5, 6, 9, or 10. or the setting depends on the condition set in the Gain etting (Pn031).	600		0 to 20000	Online	
Pn034	Gain Switch Hysteresis Setting (RT)	2	Sets the h Setting (P 5, 6, 9, or in the Gai	hysteresis of the judgment level for the Gain Switch Level 20033) when the Gain Switch Setting (Pn031) is set to 3, 10. The unit for the setting depends on the condition set n Switch Setting (Pn031).	50		0 to 20000	Online	
Pn035	Position Loop Gain Switching Time (RT)	2	This para suddenly fer by a la When the value + 1)	meter can prevent the position loop gain from increasing when the position loop gain and position loop gain 2 dif- irge amount. position loop gain increases, it takes the duration of (set $\times 166$ µs.	20	×166 μs	0 to 10000	Online	
Pn036	Reserved		Do not ch	ange.	0				
Pn037	Reserved		Do not ch	ange.	0				
Pn038	Reserved		Do not ch	ange.	0				
Pn039	Reserved		Do not ch	ange.	0				
Pn03A	Reserved		Do not ch	ange.	0				
Pn03B	Reserved		Do not ch	ange.	0				
Pn03C	Reserved		Do not ch	ange.	0				
Pn03D	Jog Speed	2	Sets the j	og operation speed with the Parameter Unit or CX-Drive.	200	r/min	0 to 500	Online	Settings
			Note Ju e u c	og operation is only available when the network is not stablished. Do not try to establish the network while sing jog operation. Otherwise, command alarm (alarm ode 27) will occur.					can be changed when the axes are stopped (Busy Flag for each axis = 0).
Ph03E	Reserved		Do not ch	ange.	0				
Pn03F	Reserved		Do not ch	anye.	0				
Pn040	Reserved		Do not ch	ange.	0				
Pn041	Emergency Stop Input Setting	2	Enables t Note If b	he Emergency Stop Input (STOP). this function is disabled, the response status will always o (disabled).	1		0, 1	Offline	
			0	Disabled.					
			1	Enabled (alarm code 87 issued on OPEN)					
Pn042	Origin Prox-	2	Sets the I	ogic for the Origin Proximity Input (DEC).	1		0, 1	Offline	
	Imity Input Logic Set-		0	N.C contact (origin proximity detected on OPEN)					
	ting		1	N.O contact (origin proximity detected on CLOSE)					
Pn043	Operating Direction Setting	2	Sets the r the netwo	elationship between polarity of operation data sent over rk and the direction of Servomotor rotation.	1		0, 1	Offline	
	Journa		Note Ir (S	RS-232C communications and on the analog monitor SP, IM) on the front panel, forward direction is always ositive (+), and reverse rotation is always negative (-).					
			0	Disabled.					
			1	Enabled (alarm code 87 issued on OPEN)					

Param- eter No.	Parameter name	Param- eter size	Setting	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn044	Input Sig-	2	Sets the t	erminal assignment for Drive Prohibit Input.	1		0, 1	Offline	
	nal Selec- tion	1	0	Sets CN1 pin 19 to POT, CN1 pin 20 to NOT.					
		1	1	Sets CN1 pin 19 to NOT, CN1 pin 20 to POT.					
Pn045	Reserved		Do not ch	ange.	0				
Pn046	Reserved		Do not ch	ange.	0				
Pn047	Reserved		Do not ch	ange.	0				
Pn048	Reserved		Do not ch	ange.	0				
Pn049	Reserved		Do not ch	ange.	0				
Pn04A	Reserved		Do not ch	ange.	0				
Pn04B	Reserved		Do not ch	ange.	0				
Pn04C	Reserved		Do not ch	ange.	0				
Pn04D	Reserved		Do not ch	ange.	0				
Pn04E	Reserved		Do not ch	ange.	0				
Pn04F	Reserved		Do not ch	ange.	0				
Pn050	Reserved		Do not ch	ange.	0				
Pn051	Reserved		Do not ch	ange.	0				
Pn052	Reserved		Do not ch	ange.	0				
Pn053	Speed Limit	2	Sets the s absolute This para ting (Pn07	peed limit for torque control mode. (The value is an <i>r</i> alue) meter is limited by the Overspeed Detection Level Set- 73).	50	r/min	-20000 to 20000	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn054	Reserved		Do not ch	ange.	0				
Pn055	Reserved		Do not ch	ange.	0				
Pn056	Reserved		Do not ch	ange.	0				
Pn057	Reserved		Do not ch	ange.	0				
Pn058	Soft Start Accelera- tion Time	2	Sets the a Accelerativalue $\times 2$	cceleration time for speed control mode. ion time [s] from 0 r/min to maximum speed [r/min] = Set ms	0	×2 ms	0 to 5000	Online	Settings can be changed when the
Pn059	Soft Start Decelera- tion Time	2	Sets the c Decelerat value $\times 2$	leceleration time for speed control mode. ion time [s] from maximum speed [r/min] to 0 r/min = Set ms	0	×2 ms	0 to 5000	Online	axes are stopped (Busy Flag for each axis = 0).
Pn05A	Reserved		Do not ch	ange.	0				
Pn05B		2	Selects th	e speed limit for torque control mode.	0		0, 1	Online	Settings
Page 0	Deserved		0 1	Use the Speed Limit (Pn053) Use the speed limit value via MECHATROLINK-II or the Speed Limit (Pn053), whichever is smaller.	0				changed when the axes are stopped (Busy Flag for each axis = 0).
Ph05C	Reserved		Do not ch	ange.	0				
PhOSE	No. 1		Do not ch	ange.	200	 0/		Online	Sottingo
PNUSE	Torque Limit	2	Refer to in select the The maxin ble Servo	40. 1 forque Limit for the Servomotor output forque. formation on the Torque Limit Selection (Pn003) to torque limit. mum value of the setting range depends on the applica- motor.	300	%	0 to 500	Unine	can be changed when the axes are stonned
Pn05F	No. 2 Torque Limit	2	Sets the N Refer to in select the The maxin ble Servo	Io. 2 torque limit for the Servomotor output torque. iformation on the Torque Limit Selection (Pn003) to torque limit. mum value of the setting range depends on the applica- motor.	100	%	0 to 500	Online	(Busy Flag for each axis = 0).
Pn060	Positioning Completion Range 1	2	Sets the p tion 1 (INI Positionin exhausted	ositioning completion range when Positioning Comple- ?1) Output is selected. g is complete when all positioning command pulses are d, and the absolute value of the position deviation con-	25	Com- mand units	0 to 10000	Online	

Param- eter No.	Parameter name	Param- eter size	Setting	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn061	Speed Con- formity Sig- nal Output Width	2	Sets the c (VCMP) s Speed co ference b and dece less than Note T	detection width for the speed conformity detection ignal. nformity is achieved when the absolute value of the dif- etween the internal speed command (before acceleration leration limits are applied) and the Servomotor speed is the set speed. his setting has a hysteresis of 10 r/min.	20	r/min	10 to 20000	Online	
Pn062	Rotation Speed for Motor Rota- tion Detec- tion	2	Sets the t Speed rea vomotor s Note S	hreshold level for the speed reached (TGON) signal. ached is determined when the absolute value of the Ser- peed is greater than the setting speed. peed reached detection has a hysteresis of 10 r/min.	50	r/min	10 to 20000	Online	
Pn063	Positioning Completion Range 2	2	Sets the p tion 2 (IN Positionin deviation regardles cessed.	positioning completion range when Positioning Comple- P2) is selected. g is complete when the absolute value of the position converted into command units is less than this setting, s of whether position command pulses are still being pro-	100	Com- mand units	0 to 10000	Online	
Pn064	Motor Phase Cur- rent Offset Re-adjust- ment Set- ting	2	Enables c the Motor inputs. Th ON. Note T w d c 0 1	or disables the offset component readjustment function of Phase Current Detector (CT) for Servo ON command the readjustment is made when control power is turned his adjustment is inaccurate if the offset is measured hile the Servomotor is rotating. To enable this function, o not rotate the Servomotor when inputting the Servo ON ommand. Disabled (only when turning ON control power) Enabled (when turning ON control power, or at Servo ON)	0		0, 1	Online	
Pn065	Undervolt- age Alarm Selection	2	Selects w function (rupted for ing Servo 0 1	hether to activate the main power supply undervoltage alarm code 13) when the main power supply is inter- the duration of the Momentary Hold Time (Pn06D) dur- ON. Turns the Servo OFF according to the setting for the Stop Selection with Main Power OFF (Pn067), interrupting the positioning command generation process (positioning operation) within the Servo Drive. When the main power supply is turned back ON, Servo ON will resume. Restart the positioning operation after performing the positioning operation and recovering from Servo OFF. Causes an error due to main power supply undervoltage (alarm code 13). This parameter is disabled if Pn06D = 1,000. If Pn06D is set too long and the voltage between P and N in the main power supply converter drops below the specified value before a main power supply interruption is detected, a main power supply undervoltage (alarm code 13) will occur.	1		0, 1	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).

Appendix B

Param- eter No.	Parameter name	Param- eter size	Set	ting		Explar	nation		Default setting	Unit	Setting range	Enable setting	Details
Pn066	Stop Selec- tion for Drive Prohi-	2	Sets warc (NO	the de Drive T) is er	eceleration Prohibit I nabled.	n stop operation to nput (POT) or Revo	be performed after the F erse Drive Prohibit Input	For- t	0		0 to 2	Offline	
	Input			Dı decel	uring leration	After stopping (30 r/min or less)	Deviation counter						
			0	Dynai brake	mic	Disables torque command in drive prohibited direction	Cleared while decel- erating with dynamic brake. Retained after stopping.						
			1	Disab torque	oles e	Disables torque command in drive prohibited direction	Cleared while decel- erating. Retained after stopping.						
			2	Emerg Stop (Pn06	gency Torque 6E)	Servo locked	Retained while decel- erating, cleared upon completion of decel- eration, and retained after stopping.						
			Note Note	 1. The operative of the operation of the ope	e position eration) w ce it enter celeration que contrining operan, ning operan, the inte be retair ning oper- ten the S opped), the ten if the d popped), the ten if the d popped, the ten if the d popped, the ten the pa und in the popping, a d ued. Whe mmand is aud. Whe mmand is istion dev mmand is sition dev mmand is sition dev mmand is serverse Dri- serving an ode autotu, but Error (arm code ned ON w peived via	ing command gene vithin the Servo Drivers the deceleration is the deceleration inode is activated ol, it will switch to p ation command is re- rinal positioning cor- red, and after dece ation will be activated ervomotor rotation the deceleration mo- rive prohibit input is arameter is set to 2 drive prohibited direct drive prohibited direct arameter is set to 2 drive prohibited direct the parameter is the parameter is the prohibited direct the servomotor viation will accumula 0. Take measures reverse direction fra arameter is set to 2 are interrupted, ar ve Prohibit Input (P operation commar uning) via RS232 w alarm code 38). A 38) will also occur vhile operating on a RS232.	eration process (position ve will be forcibly stoppe mode. Also, when the during speed control or position control. If a posi received during decelera mmand generation proce leration is complete, por ed. speed is 30 r/min or les de will not be activated s enabled. and an operation com- rection is received after (warning code 95h) will set to 0 or 1, the operat action after stopping will set to 0 or 1, the operat action after stopping will will not operate and the ate because the torque such as issuing a com- rom the host controller. MECHATROLINK-II con d (jog operation or norr ill cause a Drive Prohibi Drive Prohibit Input Errc if either POT or NOT is an operation command	ing ed , i- a- ess si- es be tion be DN, mal it or					
P067	Stop Selec- tion with Main Power OFF	2	Sets stop ervo will b	the op ping af Itage A be rese	beration to iter the ma larm Sele t when th	b be performed dur ain power supply is ection (Pn065) set he power OFF is de	ing deceleration and after turned OFF with the Un to 0. The deviation count tected.	er nd- nter	0		0 to 7	Online	Settings can be changed when the axes are
			0 an	04 \	with dyna	imic brake to decei imic brake.	erate and remain stoppe	ea					stopped (Busy
			1 an	d 5	Use free- dynamic l	run to decelerate a brake.	nd remain stopped with	1					each axis = 0.
			2 an	d 6	Use dyna when sto	mic brake to decelopped.	erate, but free the motor	r					,
			3 an	d 7	Use free- stopped.	run to decelerate, a	and free the motor when	٦					
Pn068	Stop Selec- tion for Alarm Gen-	2	Sets issue rese	the de ed by th t when	eceleration he protec an alarm	n process and stop tive function. The d i is issued.	status after an alarm is leviation counter will be	5	0		0 to 3	Online	
	eration		0	1	Use dyna with dyna	mic brake to decel mic brake.	erate and remain stoppe	ed					
			1	l	Use free- dynamic l	run to decelerate a brake.	nd remain stopped with	1					
			2	1	Use dyna when sto	mic brake to decel	erate, but free the motor	r					
			3	1	Use free- stopped.	run to decelerate, a	and free the motor when	١					
Pn069	Stop Selec-	2	Sets	the op	perational	conditions to apply	/ during deceleration an	ıd	0		0 to 7	Online	
	Servo OFF		The cour Sele	relation nter pro	nship bet ocessing f vith Main	ween set values, op for this parameter is Power OFF (Pn067	peration, and deviation s the same as for the St 7).	op					

Param- eter No.	Parameter name	Param- eter size	Setting	Explanation	Default setting	Unit	Setting range	Enable setting	Details
Pn06A	Brake Tim- ing When Stopped	2	Sets the c turns OFF command Note T fr fr fr	Juration from when the Brake Interlock (BKIR) signal to when the Servomotor is de-energized when the RUN d is turned OFF with the Servomotor stopped. he brake interlock signal is the logical OR of the brake elease request from the network and the release request om the Servo controller. Note, the brake release request om the network is OFF (operation request is ON) at ower ON.	10	2 ms	0 to 1000	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis
Pn06B	Brake Tim- ing during Operation	2	When the motor rota tion speed after the t BKIR turn before the Note T re fr	run command (RUN) is turned OFF during the Servo- ation, the Servomotor will decelerate reducing the rota- d and the Brake Interlock Signal (BKIR) will turn OFF ime set by this parameter has elapsed. Is OFF if the Servomotor speed drops below 30 r/min e set time. The brake interlock signal is the logical OR of the brake elease request from the network and the release request om the Servo controller. Note, the brake release request om the network is OFF (operation request is ON) at ower ON.	50	2 ms	0 to 1000	Online	= 0).
Pn06C	Regenera- tion Resis- tor Selection	2	Sets the r overload Set this p If using an main pow	regeneration resistor operation and the regeneration (alarm code 18) operation. arameter to 0 if using the built-in regeneration resistor. n external regeneration resistor, be sure to turn OFF the er when the built-in thermal switch is activated.	0		0 to 3	Offline	
			0	Sets the regeneration overload to match the built-in regeneration resistor. (regeneration load ratio below 1%) The regeneration overload (alarm code 18) occurs					
			2	when the load ratio of the external regeneration resistor exceeds 10%. The regeneration processing circuit by the external regeneration resistor is activated but the regeneration					
			3	The regeneration processing circuit is not activated. All regenerative energy is absorbed by the built-in capacitor					
Pn06D	Momentary Hold Time	2	Sets the a power sup The main set to 100	amount of time required to detect shutoff when the main oply continues to shut off. power OFF detection will be disabled if this parameter is 10.	35	2 ms	35 to 1000	Offline	
Pn06E	Emergency Stop Torque	2	Sets the t hibition In (Pn066) is When this The maxi motor.	orque limit during deceleration because of the Drive Pro- put when the Stop Selection for Drive Prohibition Input s set to 2. s parameter is set to 0, the normal torque limit will be set. mum value of the setting range depends on the Servo-	0	%	0 to 300	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn06F	Reserved		Do not ch	ange.	0				
Pn070	Reserved		Do not ch	ange.	0				
Pn071	Reserved		Do not ch	ange.	0				
Pn072	Overload Detection Level Set- ting	2	Sets the c be set at of 0, and level.	overload detection level. The overload detection level will 115% if this parameter is set to 0. Normally, use a setting set the level only when reducing the overload detection	0	%	0 to 500	Online	
Pn073	Overspeed Detection Level Set- ting	2	Sets the c is 1.2 time paramete only wher Note T a ir	overspeed detection level. The overspeed detection level as the maximum Servomotor rotation speed when the r is set to 0. Normally, use a setting of 0, and set the level n reducing the overspeed detection level. he detection margin of error for the setting is ± 3 r/min for 7-core absolute encoder and ± 36 r/min for a 5-core acremental encoder.	0	r/min	0 to 20000	Online	
Pn074	Reserved		Do not ch	ange.	0				
Pn075	Reserved		Do not ch	ange.	0				
Pn076	Reserved		Do not ch	ange.	0				
Pn077	Reserved		Do not ch	ange.	0				
Pn078	Reserved		Do not ch	ange.	0				
Pn079	Reserved		Do not ch	ange.	0				
Pn07A	Reserved		Do not ch	ange.	0				
Pn07B	Reserved		Do not ch	ange.	0				
Pn07C	Reserved		Do not ch	ange.	0				
Pn07D	Reserved		Do not ch	ange.	0				

Appendix B

Param- eter No.	Parameter name	Param- eter size	Setting	Explanation		Defa setti	ult I ng	Unit	Setting range	Enable setting	Details
Pn07E	Reserved		Do not change.								
Pn07F	Reserved		Do not change.								
Pn100	Backlash Compensa- tion Selec- tion	2	Enables or disables the backlash compensation for position con- trol, and sets the compensation direction.			. 0			0 to 2	Offline	
			0	Disabled Compensates in the initial positive direction after the Servo ON.							
			1								
D 101			2	Compensates in the initial negative direction after the Servo ON.							0
Pn101	Backlash Compensa- tion	2	Sets the backlash compensation amount for position control.				C m ur	om- nand nits	-32768 to 32767	Online	Settings can be changed
Pn102	Backlash Compensa- tion Time Con- stant	2	Sets the backlash compensation time constant for position control.			0	0. m	.01 ns	0 to 6400	Online	axes are stopped
			Value of Pn100	Pn101 = Positive number	Pn101 = Negative number						Flag for each axis = 0).
			1	Compensates in posi- tive direction during rotation in positive direction	Compensates in nega- tive direction during rotation in positive direction						
			2	Compensates in posi- tive direction during rotation in negative direction	Compensates in nega- tive direction during rotation in negative direction						
Pn103	Reserved		Do not ch	Do not change.							
Pn104	Soft Limit	2	 Enables or disables the soft limit. When enabled, the soft limit values are set in Forward Software Limit (Pn201) and Reverse Software Limit (Pn202). Note The response value for limit signals disabled by this setting will be set to 0. The response value for limit signals is also set to 0 when the Servomotor does not complete its return to origin. 			0 is			0 to 3	Online	
			0	Enable both the Forward / Reverse Software Limits (Pn201 and Pn202) Disable the Forward Software Limit (Pn201), enable the Reverse Software Limit (Pn202) Enable the Forward Software Limit (Pn201), disable the Reverse Software Limit (Pn202)							
			1								
			2								
			3	Disable both the Forward (Pn201 and Pn202)	/ Reverse Software Limits						
Pn105	Origin Range	2	Sets the threshold for detecting the origin (ZPOINT) in absolute values. ZPOINT = 1 when the return to origin completes (coordinate system setup is complete) and the feedback position is within the setting range of this parameter.				C m ur	Com- nand inits	0 to 250	Online	
Pn106	Reserved		Do not change.								
Pn107	Linear Accelera- tion Constant	2	Sets the acceleration for positioning operations. A setting of "0" is regarded as "1". The setting will be handled after conversion to an unsigned 16-bit data (0 to 65535). Example: $-32768 \rightarrow 8000h = 32768$ $-1 \rightarrow FFFFh = 65535$				×10 [c m ur s ²	0000 com- nand nits/ ²]	-32768 to 32767	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn108	Reserved		Do not change.			0		-			
Pn109	Reserved		Do not ch	Jo not change.							
Pn10A	Linear Decelera- tion Con- stant	2	Sets the deceleration for positioning operations. A setting of "0" is regarded as "1". The setting will be handled after conversion to an unsigned 16-bit data (0 to 65535). Example: $-32768 \rightarrow 8000h = 32768$ $-1 \rightarrow FFFFh = 65535$				× 10 [c m ur s ²	0000 com- nand nits/ ²]	-32768 to 32767	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn10B	Reserved		Do not change.			0					
Pn10C	Reserved		Do not change.			0					
Pn10D	Reserved		Do not change.			0		-			
Param- eter No.	Parameter name	Param- eter size	Setting	Explanation	Default setting	Unit	Setting range	Enable setting	Details		
-----------------------	--	------------------------	--	---	-----------------	--------------------------------------	---------------------------------	----------------	---		
Pn10E	Moving Average Time	2	Sets the r Note If a a w	noving average time for position commands. the Moving Average Time is set, commands may not be xecuted seamlessly when switching the control mode, nd when switching between interpolation feed motions nd positioning motions (motions wherein the command vaveforms are generated inside the Servo Drive).	0	×0.1 ms	0 to 5100	Online	Settings can be changed when the axes are stopped (Busy		
Pn10F	Origin	2	Sets the o	direction for origin return.	0		0, 1	Online	Flag for		
	Node Set-		0	Positive direction					= 0).		
	tings		1	Negative direction							
Pn110	Origin Return Approach Speed 1	2	Sets the oproximity latch sign This para	operating speed for origin return from when the origin signal is turned ON, to when it is turned OFF and the al is detected. meter can be set to a maximum value of 32767, but inter-	50	100 [com- mand units/ s]	1 to 32767	Online			
Pn111	Origin Return Approach	2	Sets the d after the l	ppeed is infinited to the Servonitor's maximum speed. pperating speed for origin return, from when the point atch signal is detected to when the Origin Return Final (Pn204) is reached	5	100 [com- mand	1 to 32767	Online			
	Speed 2		This para nally the	meter can be set to a maximum value of 32767, but inter- speed is limited to the Servomotor's maximum speed.		units/ s]					
Pn112	General-	2	Selects th	ne function for general-purpose output 1 (OUTM1).	7		0 to 9	Offline	Standard		
	Output 1		0	Always OFF					setting		
	Function Selection		1	INP1 output. Turn ON when position deviation is equal to or less than Pn060 for position control. Undefined when not using position control.							
			2	VCMP output. Turn ON when the deviation between the Servomotor speed and commanded speed is within the range set by Pn061 for speed control. Undefined when not using speed control.							
			3	TGON output. Turn ON when the absolute value of the Servomotor speed exceeds Pn062 setting in all control modes.							
			4	READY output. Turn ON when the main power is supplied, there is no alarm, and Servo SYNC with a host controller is estab- lished in all control modes.							
			5	CLIM output. Turn ON when torque limit is activated in all control modes.							
			6	VLIM output. Turn ON when the Servomotor speed reaches the speed limit for torque control. Undefined when not using torque control.							
			7	BKIR output. Turn ON with the release timing of the brake release signal in all control modes.							
			8	WARN output. Turn ON when a warning is issued in all control modes.							
			9	INP2 output. Turn ON when the position deviation is equal to or less than the Positioning Completion Range 2 (Pn063) for position control. Undefined when not using position control.							
Pn113	General- purpose Output 2 Function Selection	2	Selects th The set v pose outp	ne function for general-purpose output 2 (OUTM2). alues and the functions are the same as for general-pur- out 1 (OUTM1).	0		0 to 9	Offline			
Pn114	General- purpose Output 3 Function Selection	2	Selects th The set v pose outp	ne function for general-purpose output 3 (OUTM3). alues and the functions are the same as for general-pur- put 1 (OUTM1).	0		0 to 9	Offline			
Pn115 to Pn13F	Reserved		Do not ch	ange.	0						
Pn200	Absolute Origin Off- set	4	Sets the c cal coord	offset amount for the encoder position and the mechani- inate system position when using an absolute encoder.	0	Com- mand units	-1073741823 to 1073741823	Offline			

Appendix B

Param- eter No.	Parameter name	Param- eter size	Setting	Explar	nation		Default setting	Unit	Setting range	Enable setting	Details
Pn201	Forward Software Limit	4	Sets the soft lir If the Servomot (PSOT) will turn Note Be sum Revers	nit in the forward direction for exceeds the limit, the n ON (=1). to set the limits so that be Software Limit. is not turged ON when o	on. e network response statu : Forward Software Limit	us t >	500000	Com- mand units	-1073741823 to 1073741823	Online	
Br202	Bayaraa	4	Note PSUT	nit for the reverse direct		le.	500000	Com	1072741922	Online	
P1202	Software Limit	4	If the Servomol (NSOT) will tur Note Be sur Revers Note NSOT	tor exceeds the limit, the n ON (=1). e to set the limits so that e Software Limit. is not turned ON when o	e network response statu t Forward Software Limit prigin return is incomplet	us t > te.	-500000	mand units	to 1073741823	Online	
Pn203	Final Dis- tance for External Input Posi- tioning	4	Sets the distan position when p The operation a determined by parameter as fo	ce to travel after detection performing external input after detecting the latch the external input position pllows.	ng the latch signal input t positioning. signal input position will ning direction and this	be	100	Com- mand units	-1073741823 to 1073741823	Online	Settings can be changed when the axes are stopped
			External	Si	gn						(Busy Flag for
			positioning direction	Positive	Negative						each axis = 0).
			Positive direction	Moves in the positive direction and stops ^{*1}	Decelerates to a stop, reverses, then moves in the nega- tive direction and stops						
			Negative direction	Decelerates to a stop, reverses, then moves in the positive direction and stops	Moves in the nega- tive direction and stops ^{*1}						
			*1. Reverses at external inp ation distan	fter decelerating to a sto ut positioning is short in ce.	r eler-						
Pn204	Origin Return Final Dis- tance	4	Sets the distan when performir The operation a determined by lows.	ce from the latch signal ng origin return. after detecting the latch the origin return directio	input position to the orig signal input position will n and this parameter as	gin be fol-	100	Com- mand units	–1073741823 to 1073741823	Online	
			Origin	Si	gn						
			return direction	Positive	Negative						
			Positive direction	Moves in the positive direction and stops*1	Decelerates to a stop, reverses, then moves in the nega- tive direction and stops						
			Negative direction	Moves in the nega- tive direction and stops ^{*1}	Decelerates to a stop, reverses, then moves in the positive direction and stops						
			*1. Reverses at for origin retance.	iter decelerating to a sto turn is short in comparis	p if the final travel distan on to the deceleration d	nce lis-					
Pn205	Electronic Gear Ratio 1 (Numera- tor)	4	Sets the numer Setting this par tion as the num 10000 for a 2,5	ator for the electronic grator for the electronic grameter to 0 automatical terator. (131072 for a 17 00-p/r incremental enco	ear ratio. ly sets the encoder reso -bit absolute encoder, o der).	olu- or	1		0 to 131072	Offline	
			Note Set the 100 tim occur i	2,500-p/r incremental encoder). t the electronic gear ratio within the range of 1/100 to 0 times. A parameter setting alarm (alarm code 93) wil cur if the ratio is set outside of this range.							
Pn206	Electronic Gear Ratio 2 (Denomi-	4	Sets the denom Note Set the 100 tim	ur if the ratio is set outside of this range. nominator for the electronic gear ratio. the electronic gear ratio within the range of 1/100 to times. A parameter setting alarm (Alarm code 93) wil			1		1 to 65535	Offline	
D 077			occur i	f the ratio is set outside							
Pn207	Reserved		Do not change.	nge.							
Pn208	Heserved		Do not change.	ange.							
FIIZUY	Counter Overflow Level	4	The value will b multiplying with Setting this par	ecome saturated at 134 the electronic gear ratio ameter to 0 will disable	یں۔ 217728 (= 2 ²⁷) pulses af deviation counter overflo	fter	20000	mand units	2147483647	Uninte	
Pn20A to Pn21F	Reserved		Do not change.				0				

WWW.BSNEW.IR

R88D-WT W-series Servo Drive Equipped with JUSP-NS115

Function Selection Parameters

Param-	Parame-	Param-		Cc	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn000	Function selection basic	2	0	Reverse rota- tion	0	CCW direction is taken for positive command.	0			Offline	
	switch				1	CW direction is taken for positive command.					
			1	Not used.		The setting is dis- abled. (Do not change the set- ting.)	1				
			2	Unit No. setting	0 to F	Servo Drive com- munications unit number setting when using per- sonal computer monitoring soft- ware	0			Offline	
			3	Not used.	0	(Do not change the setting.)	0				
Pn001	Function selection applica-	2	0	Select stop if an alarm occurs when	0	Servomotor stopped by dynamic brake.	2			Offline	
	1			OFF	1	Servomotor stopped by dynamic brake. Dynamic brake released after stop- ping.					
					2	Servomotor stopped with free run.					
			1	Select stop when prohib- ited drive is input	0	Stop according to Pn001.0 settings (release Servomo- tor after stopping).	0			Offline	
					1	Stop Servomotor using torque set in Pn406, and lock Servomotor after stopping.					
					2	Stop Servomotor using torque set in Pn406, and release Servomotor after stopping.					
			2	Select AC/DC power input	0	AC power supply: AC power supplied from L1, L2, (L3) terminals	0			Offline	
					1	DC power supply: DC power supplied from +1, – termi- nals					
			3	Select warning code output	0	Alarm code only output from ALO1, ALO2, ALO3	1			Offline	
					1	Alarm code and warning code both output from ALO1, ALO2, ALO3					

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn002	Function selection	2	0	Torque com- mand input	0	Option command value not used.	0			Offline	
	applica- tion switch 2			ing speed con- trol)	1	Option command value 1 used as torque limit input.					
					2	Option command value 1 used as the torque feed for- ward input.					
					3	Option command values 1 and 2 used as torque limit inputs according to the forward/reverse rotation current limit setting.					
			1	Speed com- mand input	0	Option command value not used.	0			Offline	
	change (dur- ing torque con- trol) 1 Option command value 1 used as speed limit input.										
			2	Operation switch when	0	Use as absolute encoder.	0			Offline	
				using absolute encoder	1	Use as incremen- tal encoder.					
			3	Fully-closed encoder usage method	0	Fully-closed encoder is not used.	0			Offline	
					1	Fully-closed encoder is used without phase Z.					
					2	Fully-closed encoder is used with phase Z.					
					3	Fully-closed encoder is used in Reverse Rotation Mode without phase Z.					
					4	Fully-closed encoder is used in Reverse Rotation Mode with phase Z.					

Appendix B

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn003	Function selection applica-	2	0	Analog moni- tor 1 (AM) allo- cation	0	Servomotor rota- tion speed: 1 V/1,000 r/min	2			Online	
	tion switch 3				1	Speed command: 1 V/1,000 r/min					
					2	Torque command: 1 V/100%					
					3	Position deviation: 0.05 V/1 command unit					
					4	Position deviation: 0.05 V/100 com- mand units					
					5	Command pulse frequency: 1 V/1,000 r/min					
					6	Servomotor rota- tion speed: 1 V/250 r/min					
					7	Servomotor rota- tion speed: 1 V/125 r/min					
					8 to F	Not used.					
			1	Analog moni- tor 2 (NM) allo- cation	0 to F	Same as Pn003.0	0			Online	
			2 to 3	Not used.	0	(Do not change the setting.)	0				
Pn004	Not used.					(Do not change the setting.)	0200				Fixed setting: 0200
Pn005	Not used.					(Do not change the setting.)	0000				

Servo Gain Parameters

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn100	Speed loop gain	2	Adjusts	speed loop respo	onsivene	SS.	80	Hz	1 to 2000	Online	
Pn101	Speed loop inte- gration constant	2	Speed I	oop integral time	constant	t	2000	×0.01 ms	15 to 51200	Online	
Pn102	Position loop gain	2	Adjusts	position loop res	ponsiver	iess.	40	1/s	1 to 2000	Online	
Pn103	Inertia ratio	2	Sets us inertia a	ing the ratio betw and the Servomot	een the for rotor i	machine system nertia.	300	%	0 to 20000	Online	
Pn104	Speed loop gain 2	2	Adjusts matic ga	speed loop respo ain switching inpu	onsivene ut).	ss (enabled by auto-	80	Hz	1 to 2000	Online	
Pn105	Speed loop inte- gration constant 2	2	Speed I matic ga	oop integral time ain switching inpu	constant ut)	t (enabled by auto-	2000	×0.01 ms	15 to 51200	Online	
Pn106	Position loop gain 2	2	Adjusts automa	position loop res tic gain switching	ponsiver input).	less (enabled by	40	1/s	1 to 2000	Online	
Pn107	Bias rota- tional speed	2	Sets po	sition control bias	6.		0	r/min	0 to 450	Online	
Pn108	Bias addi- tion band	2	Sets the deviatio	e position control n counter pulse v	bias ope vidth.	ration start using	7	Com- mand unit	0 to 250	Online	
Pn109	Feed-for- ward amount	2	Position	control feed-form	vard com	pensation value	0	%	0 to 100	Online	

WWW.BSNEW.IR

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn10A	Feed-for- ward com- mand filter	2	Sets po	sition control fee	d-forward	l command filter.	0	×0.01 ms	0 to 6400	Online	
Pn10B	Speed control setting	2	0	P control switching con- ditions	0	Sets internal torque command value conditions (Pn10C).	4			Online	
					1	Sets speed com- mand value condi- tions (Pn10D).					
					2	Sets acceleration command value conditions (Pn10E).					
					3	Sets deviation pulse value condi- tions (Pn10F).					
					4	No P control switching function					
			1	Speed control	0	PI control	0			Offline	
				loop switching	1	IP control					
			2	Automatic gain switching	0	Automatic gain switching disabled	0			Offline	
					1	Gain switching using position com- mands					
					2	Gain switching using position devi- ation					
					3	Gain switching using position com- mands and position deviation					
			3	Not used.	0	(Do not change the setting.)	0				
Pn10C	P control switching (torque command)	2	Sets lev trol to F	vel of torque com control.	mand to	switch from PI con-	200	%	0 to 800	Online	
Pn10D	P control switching (speed command)	2	Sets lev to P cor	vel of speed comm ntrol.	nand to s	witch from PI control	0	r/min	0 to 10000	Online	
Pn10E	P control switching (accelera- tion com- mand)	2	Sets lev control	vel of acceleratior to P control.	n comma	nd to switch from PI	0	10 r/ min/s	0 to 3000	Online	
Pn10F	P control switching (deviation pulse)	2	Sets lev to P cor	vel of deviation pu ntrol.	ilses to s	witch from PI control	10	Com- mand unit	0 to 10000	Online	

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn110	Online autotun- ing setting	2	0	Select online autotuning	0	Autotunes initial operations only after power is turned ON.	2			Offline	
					1	Always autotunes.					
					2	No autotuning					
			1	Select speed	0	ON	1			Online	
				pensation function	1	OFF					
			2	Select adhe- sive friction	0	Friction compensa- tion: OFF	0			Online	
				function	1	Friction compensa- tion: Rated torque ratio small					
					2	Friction compensa- tion: Rated torque ratio large					
			3	Not used.	0	(Do not change the setting.)	0				
Pn111	Speed feedback compen- sation gain	2	Adjusts	speed loop feed	back gair	ı.	100	%	1 to 500	Online	
Pn112 to Pn123	Not used.		(Do not	change the setti	ng.)						
Pn124	Automatic gain switching timer	2	Sets the met, wh used. (Pn10B	e switching delay then the automatic .2 = 1 to 3).	after cor gain sw	nditions have been itching function is	100	ms	1 to 10000	Online	
Pn125	Automatic gain switching width (amount of position deviation)	2	Sets the dition w position	e position deviatio chen the automat deviation (Pn10	on used a ic gain sv B.2 = 2, \$	as the switching con- vitching function by 3) is used.	7	Com- mand unit	1 to 250	Online	

Position Control Parameters

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn200	Position control	2	0	Not used.	0	(Do not change the setting.)	0				
	setting 1		1	Not used.	0	(Do not change the setting.)	0				
			2	Not used.	1	(Do not change the setting.)	1				Fixed setting: 1
			3	Not used.	0	(Do not change the setting.)	0				
Pn201	Encoder divider rate	2	Sets the Drive.	e number of outp	ut pulses	from the Servo	1000	Pulses /rota- tion	16 to 16384	Offline	
Pn202	Electronic gear ratio G1 (numera- tor)	2	Sets the vomoto 0.01 ≤ 0	e pulse rate for th r travel distance. G1/G2 ≤ 100	ie comma	and pulses and Ser-	4		1 to 65535	Offline	
Pn203	Electronic gear ratio G2 (denomi- nator)	2					1		1 to 65535	Offline	
Pn204	Not used.		(Do not	change the setting	ng.)		0				

Appendix B

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn205	Absolute encoder multi-turn limit set- ting	2	Sets the vomoto	e limit to the rotati r with an absolute	ion spee encode	d when using a Ser- r.	65535	Rota- tions	0 to 65535	Offline	
Pn206	Number of fully- closed encoder pulses	2	Sets the each m	e number of fully- otor rotation.	closed e	ncoder pulses for	16384	Pulses /rota- tion	513 to 32768	Offline	
Pn207	Position control	2	0	Not used.	0	(Do not change the setting.)	0				
	setting 2		1	Not used.	1	(Do not change the setting.)	1				Fixed setting: 1
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				
Pn208	Not used.		(Do not	change the settir	ng.)		0				
Pn212	Not used.		(Do not	change the settir	ng.)		2048				
Pn217	Not used.		(Do not	change the settir	ng.)		1				
Pn218	Position control	2	0	Not used.	0	(Do not change the setting.)	0				
	setting 3		1	Not used.	1	(Do not change the setting.)	0				
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				

Speed Control Parameters

Param- eter No.	Parameter name	Param- eter size	Contents	Default setting	Unit	Setting range	Enable setting	Details
Pn300	Not used.		(Do not change the setting.)	1000				
Pn301	Not used.		(Do not change the setting.)	100				
Pn302	Not used.		(Do not change the setting.)	200				
Pn303	Not used.		(Do not change the setting.)	300				
Pn304	Jog speed	2	Sets the rotation speed used during jog operation (executed from the Servo Drive's Parameter Unit).	500	r/min	0 to 10000	Online	
Pn305	Soft start acceleration time	2	Sets acceleration time during speed control soft start.	0	ms	0 to 10000	Online	
Pn306	Soft start deceleration time	2	Sets deceleration time during speed control soft start.	0	ms	0 to 10000	Online	
Pn307	Not used.		(Do not change the setting.)	40				
Pn308	Speed feedback filter time constant	2	Sets filter time constant for speed feed- back.	0	×0.01 ms	0 to 65535	Online	
Pn309	Not used.		(Do not change the setting.)	60				

Torque Control (Torque Limit) Parameters

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn400	Not used.		(Do not	change the settir	ng.)		30				
Pn401	Torque command filter time constant	2	Sets the comma	e filter time consta nd.	ant for the	e internal torque	40	×0.01 ms	0 to 65535	Online	

Appendix B

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn402	Forward torque limit	2	Forward ratio).	d rotation output	torque lin	nit (rated torque	350	%	0 to 800	Online	
Pn403	Reverse torque limit	2	Reverse ratio).	e rotation output	torque lin	nit (rated torque	350	%	0 to 800	Online	
Pn404	Forward rotation external current limit	2	Output rent lim	torque limit durin it (rated torque ra	g input of itio).	forward rotation cur-	100	%	0 to 800	Online	
Pn405	Reverse rotation external current limit	2	Output rent lim	torque limit durin it (rated torque ra	g input of ttio).	f reverse rotation cur-	100	%	0 to 800	Online	
Pn406	Emer- gency stop torque	2	Deceler torque	ration torque whe ratio).	n an erro	or occurs (rated	350	%	0 to 800	Online	
Pn407	Speed limit	2	Sets the	e speed limit in to	orque cor	ntrol mode.	3000	r/min	0 to 10000	Online	
Pn408	Torque command	2	0	Selects notch filter 1 function	0	Notch filter 1 not used.	0			Online	
	setting				1	Notch filter 1 used for torque com- mands.					
			1	Not used.	0	(Do not change the setting.)	0				
			2	Selects notch filter 2 function	0	Notch filter 2 not used.	0			Online	
					1	Notch filter 2 used for torque com- mands.					
			3	Not used.	0	(Do not change the setting.)	0				
Pn409	Notch fil- ter 1 fre- quency	2	Sets no	ets notch filter 1 frequency for torque command.				Hz	50 to 2000	Online	
Pn40A	Notch fil- ter 1 Q value	2	Sets Q	Sets Q value of notch filter 1.				×0.01	50 to 400	Online	
Pn40B	Notch fil- ter 2 fre- quency	2	Sets no	Sets notch filter 2 frequency for torque command.				Hz	50 to 2000	Online	
Pn40C	Notch fil- ter 2 Q value	2	Sets Q	value of notch filt		70	×0.01	50 to 400	Online		

I/O and Status Parameters

Param-	Parame-	Param-	Digit Name Set- Explanation				Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn500	Position- ing com- pletion range 1	2	Sets the	e width of the pos	itioning o	completed range.	3	Com- mand unit	0 to 250	Online	
Pn501	Not used.		(Do not	change the setting		10					
Pn502	Rotation speed for motor rota- tion detec- tion	2	Sets the Speed I	ets the number of rotations for detecting the Zero beed Flag.				r/min	1 to 10000	Online	
Pn503	Speed conformity signal out- put width	2	Sets the for the S	e allowable fluctua Speed Conformity	ation ran / Flag.	ge (rotation speed)	10	r/min	0 to 100	Online	

Param-	Parame-	Param-		Co		Default	Unit	Setting	Enable	Details	
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn504	Position- ing com- pletion range 2	2	Sets the Flag.	proximity range	for the P	ositioning Proximity	3	Com- mand unit	1 to 250	Online	
Pn505	Deviation counter overflow level	2	Sets the alarm.	e detection level	for the de	viation counter over	1024	×256 pulses	1 to 32767	Online	
Pn506	Brake tim- ing 1	2	Sets the motor tu	e delay from the l Irning OFF.	brake con	nmand to the Servo-	0	×10 ms	0 to 50	Online	
Pn507	Brake command speed	2	Sets the mand.	e rotation speed t	for output	ting the brake com-	100	r/min	0 to 10000	Online	
Pn508	Brake tim- ing 2	2	Sets the the brak	e delay from the se command outp	Servomot out.	or turning OFF to	50	×10 ms	10 to 100	Online	
Pn509	Momen- tary hold time	2	Sets the when a	time during which be a component of the second seco	ch alarm o curs.	detection is disabled	20	ms	20 to 1000	Online	

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details	
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting		
Pn50A	Input sig- nal selec- tion 1	2	0	Not used.	1	(Do not change the setting.)	1				Fixed setting: 1	
			1	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8	
			2	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8	
			3	POT (forward drive prohibited input) signal	0	Allocated to CN1, pin 40: Valid for low input	8			Offline	Stan- dard setting:	
				input terminal allocation	1	Allocated to CN1, pin 41: Valid for low input					2	
					2	Allocated to CN1, pin 42: Valid for low input						
					3	Allocated to CN1, pin 43: Valid for low input						
					4	Allocated to CN1, pin 44: Valid for low input						
						5	Allocated to CN1, pin 45: Valid for low input					
					6	Allocated to CN1, pin 46: Valid for low input						
					7	Always valid.						
					8	Always invalid.						
					9	Allocated to CN1, pin 40: Valid for high input						
					A	Allocated to CN1, pin 41: Valid for high input						
					В	Allocated to CN1, pin 42: Valid for high input						
					С	Allocated to CN1, pin 43: Valid for high input						
				D	Allocated to CN1, pin 44: Valid for high input	,						
			E	Allocated to CN1, pin 45: Valid for high input	J1, r							
					F	Allocated to CN1, pin 46: Valid for high input						

Appendix **B**

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn50B	Input sig- nal selec- tion 2	2	0	NOT (reverse drive prohibited input) signal input terminal allocation	0 to F	Same as Pn50A.3	8			Offline	Stan- dard setting: 3
			1	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8
			2	PCL (forward rotation cur- rent limit) sig- nal input terminal alloca- tion	0 to F	Same as Pn50A.3	5			Offline	Stan- dard setting: 8
			3	NCL (reverse rotation cur- rent limit) sig- nal input terminal alloca- tion	0 to F	Same as Pn50A.3	6			Offline	Stan- dard setting: 8
Pn50C	Input sig- nal selec- tion 3	2	0	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8
			1	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8
			2	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8
			3	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8
Pn50D	Input sig- nal selec- tion 4	2	0	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8
			1	Not used.	8	(Do not change the setting.)	8				Fixed setting: 8
			2	Not used.	8	(Do not change the setting.)	8				
			3	Not used.	8	(Do not change the setting.)	8				
Pn50E	Output sig- nal selec-	2	0	INP1 (position-	0	No output	1			Offline	Stan- dard
	tion 1			1) signal output terminal alloca-	1	Allocated to CN1, pins 25, 26	_				setting: 1
				tion	2	Allocated to CN1, pins 27, 28	-				
					3	Allocated to CN1, pins 29, 30					
			1	VCMP (speed conformity) signal output terminal alloca- tion	0 to 3	Same as Pn50E.0	1			Offline	Stan- dard setting: 0
			2	TGON (Servo- motor rotation detection) sig- nal output ter- minal allocation	0 to 3	Same as Pn50E.0	2			Offline	Stan- dard setting: 0
			3	READY (Ser- vomotor war- mup complete) signal output terminal alloca- tion	0 to 3	Same as Pn50E.0	3			Offline	Stan- dard setting: 3

WWW.BSNEW.IR

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn50F	Output sig- nal selec- tion 2	2	0	CLIMT (cur- rent limit detec- tion) signal output termi- nal allocation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
			1	VLIMT (speed limit detection) signal output terminal alloca- tion	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
			2	BKIR (brake interlock) sig- nal output ter- minal allocation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 2
			3	WARN (warn- ing) signal out- put terminal allocation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
Pn510	Output sig- nal selec- tion 3	2	0	INP2 (position- ing completed 2) signal output terminal alloca- tion	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
			1	Not used.	0	(Do not change the setting.)	0				
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				

Param-	Parame-	Param-	m- Contents				Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn511	Input sig- nal selec- tion 5	2	0	DEC signal input terminal allocation	0	Allocated to CN1, pin 40: Valid for low input	8			Offline	Stan- dard setting:
					1	Allocated to CN1, pin 41: Valid for low input					1
					2	Allocated to CN1, pin 42: Valid for low input					
					3	Allocated to CN1, pin 43: Valid for low input					
					4	Allocated to CN1, pin 44: Valid for low input					
					5	Allocated to CN1, pin 45: Valid for low input					
					6	Allocated to CN1, pin 46: Valid for low input					
					7	Always enabled.					
					8	Always disabled.					
					9	Allocated to CN1, pin 40: Valid for high input					
					A	Allocated to CN1, pin 41: Valid for high input					
					В	Allocated to CN1, pin 42: Valid for high input					
					С	Allocated to CN1, pin 43: Valid for high input					
					D	Allocated to CN1, pin 44: Valid for high input					
					E	Allocated to CN1, pin 45: Valid for high input					
					F	Allocated to CN1, pin 46: Valid for high input					

Param-	Parame-	Param-	n- Contents Digit Name Set- Explanation No. ting				Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn511	Input sig-	2	1	EXT1 signal	0 to 3	Always disabled.	8			Offline	Stan-
	nal selec- tion 5			allocation	4	Allocated to CN1, pin 44: Valid for low input					dard setting: 4
					5	Allocated to CN1, pin 45: Valid for low input					
					6	Allocated to CN1, pin 46: Valid for low input					
					7	Always enabled.					
					8	Always disabled.					
					9 to C	Always disabled.					
					D	Allocated to CN1, pin 44: Valid for high input					
					E	Allocated to CN1, pin 45: Valid for high input					
					F	Allocated to CN1, pin 46: Valid for high input					
			2	EXT2 signal input terminal allocation	0 to F	Same as Pn511.1	8			Offline	Stan- dard setting: 5
			3	EXT3 signal input terminal allocation	0 to F	Same as Pn511.1	8			Offline	Stan- dard setting: 6
Pn512	Output sig-	2	0	Output signal	0	Not reversed.	0			Offline	
	nal reverse			reverse for CN1 pins 25, 26	1	Reversed.					
			1	Output signal reverse for CN1 pins 27, 28	0, 1	Same as above	0			Offline	
			2	Output signal reverse for CN1 pins 29, 30	0, 1	Same as above	0			Offline	
			3	Not used.		(Do not change the setting.)	0				
Pn513	Input sig- nal selec-	2	0	Not used.	8	(Do not change the setting.)	8				
			1	Not used.	8	(Do not change the setting.)	8				
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				
Pn51A	Motor-load deviation over level	2	Sets the fully-clo	e allowable range sed encoders an	e for the r d semi-c	number of pulses for losed encoders.	0	Pulse	0 to 32767	Online	
Pn51B	Not used.		(Do not	change the setti	ng.)		100				
Pn51C	Not used.		(Do not	change the setti	ng.)		450				
Pn51E	Deviation counter overflow warning level	2	Sets the flow wa	e detection level t rning.	for the de	eviation counter over-	0	%	0 to 100	Online	

Other Parameters

Param- eter No.	Parameter name	Param- eter size	Contents	Default setting	Unit	Setting range	Enable setting	Details
Pn600	Regeneration resistor capacity	2	Setting for the regeneration resistance load ratio monitoring calculations.	0	×10 W	From 0 (Varies by Unit.)	Online	
Pn601	Not used.	0	(Do not change the setting.)					

Control Function Parameters

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn800	Communi- cations control	2	0	MECHA- TROLINK-II communica- tions check mask	0	Detects both com- munications errors (A.E6) and syn- chronization errors (A.E5).	0			Online	Always set to 0.
					1	Ignores communi- cations errors (A.E6).					
					2	Ignores synchroni- zation errors (A.E5).					
					3	Ignores both com- munications errors (A.E6) and syn- chronization errors (A.E5).					
			1	Warning check mask	0	Detects parameter setting warnings (A.94), MECHA- TROLINK com- mand warnings (A.95), and com- munications errors (A.96).	4			Online	Always set to 4 or 0.
					1	Ignores parame- ters setting warn- ings (A.94).					
					2	Ignores MECHA- TROLINK-II com- mand warnings (A.95).					
					3	Ignores both parameter setting warnings (A.94) and MECHA- TROLINK-II com- mand warnings (A.95).					
					4	Ignores communi- cations errors (A.96).					
					5	Ignores both parameter setting warnings (A.94) and communica- tions errors (A.96).					
					6	Ignores both MECHATROLINK- II command warn- ings (A.95) and communications errors (A.96).					
					7	Ignores parameter setting warnings (A.94), MECHA- TROLINK-II com- mand warnings (A.95), and com- munications errors (A.96).					
			2	Communica- tions error count at single transmission	0 to F	Detects a commu- nications error (A.E6) when the number of errors specified by the set value + 2 have occurred continu- ously.	0			Online	
			3	Not used.	0	(Do not change the setting.)	0				

WWW.BSNEW.IR

Appendix B

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	e Details			
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting				
Pn801	Software limits	2	0	Software limit function	0	Software limit enabled.	0			Online				
					1	Forward software limit disabled.								
					2	Reverse software limit disabled.								
					3	Software limit dis- abled in both direc- tions.								
			1	Not used.	0	(Do not change the setting.)	0							
				2	2	2	Software limit check using references	0	No software limit check using refer- ences.	0			Online	Always set to 0.
					1	Software limit check using refer- ences.								
			3	Not used.	0	(Do not change the setting.)	0							

Param- eter No.	Parameter name	Param- eter size	Contents	Default set- ting	Unit	Setting range	Enable setting	Details
Pn802	Not used.		(Do not change the set- ting.)	0000				
Pn803	Zero point width	2	Sets the detection range for the Origin Stop Flag.	10	Com- mand unit	0 to 250	Online	
Pn804	Forward soft- ware limit	4	Sets the forward soft- ware limit.	819,191,808	Com- mand unit	-1,073,741,823 to 1,073,741,823	Online	
Pn806	Reverse soft- ware limit	4	Sets the reverse soft- ware limit.	-819,191,808	Com- mand unit	-1,073,741,823 to 1,073,741,823	Online	
Pn808	Absolute encoder zero point position offset	4	Sets the offset for the mechanical origin from the absolute encoder's absolute value data.	0	Com- mand unit	-1,073,741,823 to 1,073,741,823	Offline	
Pn80A	First-step linear acceleration constant	2	Sets the first-step acceleration speed for the acceleration/decelera- tion curve used in position control.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80B	Second-step lin- ear acceleration constant	2	Sets the second-step acceleration speed for the acceleration/deceler- ation curve used in posi- tion control.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80C	Acceleration constant switch- ing speed	2	Sets the speed for switching between first- step and second-step acceleration for the acceleration/decelera- tion curve used in posi- tion control.	0	×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).

WWW.BSNEW.IR

Param- eter No.	Parameter name	Param- eter size	Contents	Default set- ting	Unit	Setting range	Enable setting	Details
Pn80D	First-step linear deceleration constant	2	Sets the first-step decel- eration speed for the acceleration/decelera- tion curve used in posi- tion control.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80E	Second-step lin- ear deceleration constant	2	Sets the second-step deceleration speed for the acceleration/deceler- ation curve used in posi- tion control.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80F	Deceleration constant switch- ing speed	2	Sets the speed for switching between first- step and second-step deceleration for the acceleration/decelera- tion curve used in posi- tion control.	0	×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn810	Exponential acceleration/ deceleration bias	2	Sets the exponential acceleration/decelera- tion bias speed for the acceleration/decelera- tion curve used in posi- tion control.	0	Com- mand units/s	0 to 32767	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn811	Exponential acceleration/ deceleration time constant	2	Sets exponential accel- eration/deceleration time constant for the acceler- ation/deceleration curve used in position control.	0	×0.1 ms	0 to 5100	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn812	Movement aver- age time	2	Sets the S-curve accel- eration/deceleration moving average time for the acceleration/deceler- ation curve used in posi- tion control.	0	×0.1 ms	0 to 5100	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn813	Not used.		(Do not change the set- ting.)	0010				
Pn814	Final travel dis- tance for exter- nal positioning	4	Sets the final travel dis- tance for external posi- tioning when executing INTERRUPT FEEDING using direct operation.	100	Com- mand unit	-1,073,741,823 to 1,073,741,823	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).

Param-	Parame-	Param-		Con	tents		Default	Unit	Setting range	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting			setting	
Pn816	Zero point return mode set- ting	2	0	Zero point return direc- tion	0	Forward Reverse	0			Online	Set the same direc- tion as the origin search direction set in the Axis Parameters. Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
			1 to 3	Not used.	used. 0 (Do n chang setting in (zero point) input signa d used after the origin pro		0				
Pn817	Zero point return approach speed 1	2	Sets the search signal h	> origin (zero p speed used aff las been detec	origin (zero point) input signal beed used after the origin proximity s been detected in an origin search porigin (zero point) return final travel positioning speed used after the ori-			×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn818	Zero point return approach speed 2	2	Sets the distance gin inpu gin sea	 origin (zero p positioning s it signal has be rch. 	origin (zero point) return final travel positioning speed used after the ori- signal has been detected in an ori- h.			×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn819	Final travel dis- tance to return to zero point	4	Sets the used af detecte	amount of cor ter the origin ir d in an origin s	amount of compensation positioning r the origin input signal has been in an origin search.			Com- mand unit	-1,073,741,823 to 1,073,741,823	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn81B	Backlash compen- sation amount	2	Sets the	amount of ba	ıcklash c	ompensation.	0	×0.1 com- mand unit	-32,768 to 32,767	Online	
Pn81C	Not used.		(Do not	change the se	etting.)		0000				
Pn81D	Compen- sation function selection	2	0	Backlash compensa- tion selec- tion	A Sa- C- C- D- D- D- D- D- D- D- D- D- D- D- Sates in for- ward direction. 1 Compen- sates in for- ward direction. 1 Compen- sates in for- ward direction. 1 Compen- sates in for- ward direction. 1 Compen- sates in for- ward direction. 1 Compen- sates in for- ward direction. 1 Compen- sates in for- ward direction.		0			Offline	
			1.0.5		0	change the setting.)	С С				
Pn81E to Pn823	Not used.		(Do not	change the setting.)			0000				

R88D-WN -ML2 W-series Servo Drive with Built-in MECHATROLINK-II Communications

Function Selection Parameters

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn000	Function selection basic	2	0	Reverse rota- tion	0	CCW direction is taken for positive command	0			Offline	
	switches				1	CW direction is taken for positive command					
					2 to 3	Not used.					
			1	Not used.	0	(Do not change the setting.)	0				
			2	Unit No. setting	0 to F	Servo Drive com- munications unit number setting (necessary for mul- tiple Servo Drive connections when using personal computer monitor- ing software)	0			Offline	
			3	Not used.	0	(Do not change the setting.)	0				
Pn001	Function selection applica-	2	0	Stop selection if an alarm occurs when	0	Servomotor stopped by dynamic brake.	2			Offline	
	tion switches 1	OFF	OFF	1	Dynamic brake OFF after Servo- motor stopped						
					2	Servomotor stopped with free run					
			1	Stop selection when drive prohibited is input	0	Stop according to Pn001.0 setting (release Servomo- tor after stopping)	0			Offline	
					1	Stop Servomotor using torque set in Pn406, and lock Servomotor after stopping					
					2	Stop Servomotor using torque set in Pn406, and release Servomotor after stopping					
		2	AC/DC power input selection	0	AC power supply: AC power supplied from L1, L2, (L3) terminals	0			Offline		
			1	1	DC power supply: DC power from +, - (2) terminals	r: -, -					
			3	Not used.	0	(Do not change the setting.)	0				

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details	
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting		
Pn002	Function selection	2	0	Torque com- mand input	0	Do not use option command value.	0			Offline		
	applica- tion switches 2			change (dur- ing speed con- trol)	1	Use option com- mand value 1 as the torque limit value.						
					2	Use option com- mand value 1 as the torque feed for- ward command value.						
					3	Use option com- mand value 1 or 2 as the torque limit value, according to the forward and reverse torque lim- its that are speci- fied.						
			1	1 Speed co mand inp	Speed com- mand input	0	Do not use option command value.	0			Offline	
				change (dur- ing torque con- trol)	1	Use option com- mand value 1 as the speed limit value.						
			2	Operation switch when	0	Use as absolute encoder	0			Offline		
				using absolute encoder	1	Use as incremen- tal encoder						
			3	Not used.	0	(Do not change the setting.)	0					
Pn004	Function selection	2	0	Not used.	0	(Do not change the setting.)	0					
	applica- tion switches 4		1	Not used.	1	(Do not change the setting.)	1					
			2	Not used.	1	(Do not change the setting.)	1					
			3	Not used.	0	(Do not change the setting.)	0					

Param-	Parame-	Param-	n- Contents				Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn006	Function selection applica-	2	0 to 1	Analog moni- tor 1 (AM) sig- nal selection	00	Servomotor rota- tion speed: 1 V/ 1000 r/min	02			Online	
	tion switches 6				01	Speed command: 1 V/1000 r/min					
					02	Torque command: gravity compensa- tion torque (Pn422) (1 V per 100%)					
					03	Position deviation: 0.05 V/1 command unit					
					04	Position amplifier deviation (after electronic gear) (0.05 V per encoder pulse unit)					
					05	Position command speed (1 V/1,000 r/min)					
					06	Not used.					
						07	Not used.				
					08	Positioning com- pleted command (Positioning com- pleted: 5 V; posi- tioning not completed: 0 V					
					09	Speed feed forward (1 V/1,000 r/min)					
					0A	Torque feed for- ward (1 V per 100%)					
					0B to 1F	Not used.					
			2	Analog moni-	0	1x	0			Online	
				tor 1 signal multiplier	1	10x					
				selection	2	100x					
					3	1/10x					
					4	1/100x					
			3	Not used.	0	(Do not change the setting.)	0				

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn007	Function selection applica-	2	0 to 1	Analog moni- tor 2 (NM) sig- nal selection	00	Servomotor rota- tion speed: 1V/ 1000 r/min	00			Online	
	switches 7				01	Speed command: 1 V/1000 r/min					
					02	Torque command: gravity compensa- tion torque (Pn422) (1 V per 100%)					
					03	Position deviation: 0.05 V/1 command unit					
					04	Position amplifier deviation (after electronic gear) (0.05 V per encoder pulse unit)					
					05	Position command speed (1 V/1,000 r/min)					
					06	Not used.					
					07	Not used.					
					08	Positioning com- pleted command (Positioning com- pleted: 5 V; posi- tioning not completed: 0 V					
					09	Speed feed forward (1 V/1,000 r/min)					
					0A	Torque feed for- ward (1 V per 100%)					
					0B to 1F	Not used.					
			2	Analog moni-	0	1x	0			Online	
				tor 2 signal multiplier	1	10x					
				selection	2	100x					
					3	1/10x					
					4	1/100x					
			3	Not used.	0	(Do not change the setting.)	0				
Pn008	Function selection applica-	2	0	Lowered bat- tery voltage alarm/warning	0	Regard battery voltage drop as alarm (A.830).	0			Offline	
	tion switches 8			selection	1	Regard battery voltage drop as warning (A.930).					
			1	Not used.	0	(Do not change the setting.)	0				
			2	Warning detec-	0	Warnings detected.	0			Offline	
				tion selection	1	Warnings not detected.					
			3	Not used.	4	(Do not change the setting.)	4				

Servo Gain Parameters

Param-	Parameter	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn100	Speed loop gain	2	Adjusts	speed loop resp	onse.		800	×0.1 Hz	10 to 20000	Online	
Pn101	Speed loop integration constant	2	Speed	loop integral time	constan	t	2000	×0.01 ms	15 to 51200	Online	
Pn102	Position loop gain	2	Adjusts	position loop res	ponse.		400	×0.1/s	10 to 20000	Online	
Pn103	Inertia ratio	2	Set usin tia and	ng the ratio betwee the Servomotor r	een the motor iner	nachine system iner- tia.	300	%	0 to 20000	Online	
Pn104	Speed loop gain 2	2	Adjusts gain sw	speed loop resp itching input).	onse (en	abled by automatic	800	×0.1 Hz	10 to 20000	Online	
Pn105	Speed loop integration constant 2	2	Speed matic g	loop integral time ain switching inp	constan ut).	t (enabled by auto-	2000	×0.01 ms	15 to 51200	Online	
Pn106	Position loop gain 2	2	Adjusts gain sw	position loop res itching input).	ponse (e	nabled by automatic	400	×0.1/s	10 to 20000	Online	
Pn107	Bias rota- tional speed	2	Sets po	sition control bia	S.		0	r/min	0 to 450	Online	
Pn108	Bias addi- tion band	2	Sets the deviation	e position control on counter pulse v	bias ope width.	eration start using	7	Com- mand unit	0 to 250	Online	
Pn109	Feed-for- ward amount	2	Positior	n control feed-forv	ward com	pensation value	0	%	0 to 100	Online	
Pn10A	Feed-for- ward com- mand filter	2	Sets po	sition control fee	d-forward	d command filter.	0	×0.01 ms	0 to 6400	Online	
Pn10B	Speed con- trol setting	2	0	P control switching con- ditions	0	Sets internal torque command value conditions (Pn10C).	4			Online	
					1	Sets speed com- mand value condi- tions (Pn10d).					
					2	Sets acceleration command value conditions (Pn10E).	-				
					3	Sets deviation pulse value condi- tions (Pn10F).					
					4	No P control switching function					
			1	Speed control	0	PI control	0			Offline	
				loop switching	1	IP control					
					2 to 3	Not used.					
			2	Position loop control method	0	Standard position control	0			Offline	
					1	Less deviation con- trol					
					2 to 3	Not used.					
			3	Not used.	0	(Do not change the setting.)	0				
Pn10C	P control switching (torque command)	2	Sets lev trol to F	Sets level of torque command to switch from PI con- trol to P control.			200	%	0 to 800	Online	
Pn10D	P control switching (speed command)	2	Sets lev to P co	vel of speed comr ntrol.	nand to s	witch from PI control	0	r/min	0 to 10000	Online	

Param-	Parameter	Param-	Contents				Default	Unit	Setting	Enable	Details
eter No.	name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn10E	P control switching (accelera- tion com- mand)	2	Sets lev control	vel of acceleratior to P control.	n comma	nd to switch from PI	0	r/min/s	0 to 30000	Online	
Pn10F	P control switching (deviation pulse)	2	Sets lev to P cor	vel of deviation pu htrol.	ilses to s	witch from PI control	10	Com- mand unit	0 to 10000	Online	
Pn110	Normal autotuning	2	0	Not used.	2	(Do not change the setting.)	2			Offline	
	switches		1	Speed feed-	0	ON	1			Offline	
				back compen-	1	OFF					
				selection	2 to 3	Not used.					
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				
Pn111	Speed feedback compen- sating gain	2	Adjusts	speed loop feed	back gair	ı.	100	%	1 to 500	Online	
Pn119	Not used.		(Do not	change the setting	ng.)		500				
Pn11A	Not used.		(Do not	change the setting	ng.)		1000				
Pn11E	Not used.		(Do not	change the setting	ng.)		1000				
Pn11F	Position integral time con- stant	2	Positior	n loop integral tim	e consta	nt	0	×0.1 ms	0 to 50000	Online	
Pn12B	Not used.		(Do not	change the setting	ng.)		400				
Pn12C	Not used.		(Do not	change the setting	ng.)		2000				
Pn12D	Not used.		(Do not	change the setting	ng.)		400				
Pn12E	Not used.		(Do not	change the setting	ng.)		400				
Pn12F	Not used.		(Do not	change the setting	ng.)		2000				
Pn130	Not used.		(Do not	change the setting	ng.)		400				
Pn131	Gain switching time 1	2	Switchi	ng time from No.	1 gain to	No. 2 gain	0	ms	0 to 65535	Online	
Pn132	Gain switching time 2	2	Switchi	ng time from No.	No. 1 gain	0	ms	0 to 65535	Online		
Pn135	Gain switching waiting time 1	2	The tim isfied u gain be	e from when gair ntil switching fron gins.	n switchir n the No.	ng condition A is sat- 1 gain to the No. 2	0	ms	0 to 65535	Online	
Pn136	Gain switching waiting time 2	2	The tim isfied u gain be	e from when gair ntil switching fron gins.	n switchir n the No.	ng condition B is sat- 2 gain to the No. 1	0	ms	0 to 65535	Online	

Param-	Parameter	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn139	Automatic gain changeover related	2	0	Gain switching selection switch	0	Manual gain switching (auto- matic gain switch- ing not used)	0			Offline	
					1	Automatic switch- ing pattern 1 Automatic switch- ing from No. 1 gain to No. 2 gain when gain switching con- dition A is satisfied. Automatic switch- ing from No. 2 gain to No. 1 gain when gain switching con- dition B is satisfied.					
					2 to 4	Not used.					
			1	Gain switching condition A	0	Positioning com- pleted output 1 (INP1) ON	0			Offline	
					1	Positioning com- pleted output 1 (INP1) OFF					
					2	Positioning com- pleted output 2 (INP2) ON					
					3	Positioning com- pleted output 2 (INP2) OFF					
					4	The position com- mand filter output is 0, and also the position command input is 0.					
					5	The position com- mand input is not 0.					
			2	Gain switching condition B	0 to 5	Same as Pn139.1	0			Offline	
			3	Not used.	0	(Do not change the setting.)	0				
Pn144	Not used.		(Do not	change the setti	ng.)	T	1000				
Pn150	Predictive control selection	2	0	Predictive con- trol selection	0	Predictive control not used.	0			Offline	
	switches				1	Predictive control used.	-				
					2	Not used.				0.00	
			1	trol type	0	Predictive control for tracking	1			Offline	
			2	Notused	2	for positioning	2				
			-	Nature of	-	(Be not change the setting.)	-				
			3	Not used.	0	(Do not change the setting.)	0				
Pn151	Predictive control accelera- tion/decel- eration gain	2	Adjusts predicti	acceleration and ve control.	decelera	ation response for	100	%	0 to 300	Online	
Pn152	Predictive control weighting ratio	2	Adjusts	position deviatio	n for pree	dictive control.	100	%	0 to 300	Online	
Pn1A0	Servo rigidity	2	Adjusts	the Servo rigidity	/ for the I	No. 1 gain.	60	%	1 to 500	Online	
Pn1A1	Servo rigidity 2	2	Adjusts	the Servo rigidity	/ for the I	No. 2 gain.	60	%	1 to 500	Online	

Param-	Parameter	Param-		Contents				Unit	Setting	Enable	Details
eter No.	name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn1A2	Speed feedback filter time constant	2	Sets the back.	e filter time const	ant for N	o. 1 gain speed feed-	72	×0.01 ms	30 to 3200	Online	
Pn1A3	Speed feedback filter time constant 2	2	Sets the back.	e filter time const	ant for N	o. 2 gain speed feed-	72	×0.01 ms	30 to 3200	Online	
Pn1A4	Torque command filter time constant 2	2	Sets the	e filter time const	ant for th	e torque command.	36	×0.01 ms	0 to 2500	Online	
Pn1A7	Utility con- trol switches	2	0	Integral com- pensation pro- cessing	0	Integral compensa- tion processing not executed.	1			Online	
					1	Integral compensa- tion processing executed.					
					2	Integral compensa- tion is executed for No. 1 gain and not for No. 2 gain for less-deviation gain switching.					
					3	Integral compensa- tion is executed for No. 2 gain and not for No. 1 gain for less-deviation gain switching.					
			1	Not used.	2	(Do not change the setting.)	2				
			2	Not used.	1	(Do not change the setting.)	1				
			3	Not used.	1	(Do not change the setting.)	1				
Pn1A9	Utility inte- gral gain	2	Adjusts	djusts the auxiliary integral responsive.			37	Hz	0 to 500	Online	
Pn1AA	Position propor- tional gain	2	Adjusts	djusts the position proportional responsive.				Hz	0 to 500	Online	
Pn1AB	Speed inte- gral gain	2	Adjusts	Adjusts the speed integral responsive.				Hz	0 to 500	Online	
Pn1AC	Speed pro- portional gain	2	Adjusts	djusts the speed proportional responsive.				Hz	0 to 2000	Online	
Pn1B5	Not used.		(Do not	change the setti	ng.)		150				

Position Control Parameters

Param-	Parame-	Param-		C	ontents	i	Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn200	Not used.	2	0	Not used.	0	(Do not change the setting.)	0				
			1	Not used.	0	(Do not change the setting.)	0				
			2	Not used.	1	(Do not change the setting.)	1				
			3	Not used.	0	(Do not change the setting.)	0				
Pn205	Absolute encoder multi-turn limit set- ting	2	Sets the an abso	e multi-turn limi blute encoder is	it for whe s used.	n a Servomotor with	65535	Rota- tion	0 to 65535	Offline	
Pn207	Position control	2	0	Not used.	0	(Do not change the setting.)	0				
	settings 2		1	Not used.	1	(Do not change the setting.)	1				
			2	Backlash	0	Disabled	0			Offline	
				compensa- tion selection	1	Compensates to forward rotation side.					
					2	Compensates to reverse rotation side.					
			3	INP 1 output timing	0	When the position deviation is below the INP1 range (Pn522).	0			Offline	
					1	When the position deviation is below the INP1 range (Pn522) and also the command after the position com- mand filter is 0.					
					2	When the absolute value for the posi- tion deviation is below the INP1 range (Pn522) and also the position command input is 0.					
Pn209	Not used.		(Do not	change the se	tting.)		0				
Pn20A	Not used.		(Do not	change the se	tting.)		32768				
Pn20E	Electronic gear ratio G1 (numera- tor)	4	Sets the Servom 0.001 ≤	e pulse rate for notor movemen $\leq G1/G2 \leq 1000$	the com t distanc	mand pulses and e.	4		1 to 1073741824	Offline	
Pn210	Electronic gear ratio G2 (denomi- nator)	4					1		1 to 1073741824	Offline	
Pn212	Encoder divider rate	4	Sets the rotation	e number of ou ı.	tput puls	es per Servomotor	1000	Pulses /rota- tion	16 to 1073741824	Offline	
Pn214	Backlash compen- sation amount	2	Mechar cal gap driven)	nical system ba between the d	cklash a rive shaf	mount (the mechani- t and the shaft being	0	Com- mand unit	-32767 to 32767	Online	
Pn215	Backlash compen- sation time constant	2	Sets the	e backlash com	npensatio	on time constant.	0	×0.01 ms	0 to 65535	Online	

Appendix B

Param-	Parame-	Param-		C		Default	Unit	Setting	Enable	Details	
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn216	Not used.		(Do not	change the set	tting.)		0				
Pn217	Not used.		(Do not	o not change the setting.)			0				
Pn281	Not used.		(Do not	o not change the setting.)			20				

Speed Control Parameters

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn300	Not used.		(Do not	change the settin	ng.)		600				
Pn301	Not used.		(Do not	change the settin	ng.)		100				
Pn302	Not used.		(Do not	change the settin	ng.)		200				
Pn303	Not used.		(Do not	change the setting	ng.)		300				
Pn304	Jog speed	2	Sets rot Drive's	tation speed durir personal compute	ng jog op er monito	eration (using Servo pring software)	500	r/min	0 to 10000	Online	
Pn305	Soft start accelera- tion time	2	Sets ac	celeration time d	eed control soft start.	0	ms	0 to 10000	Online		
Pn306	Soft start decelera- tion time	2	Sets de	eceleration time d	uring spe	eed control soft start.	0	ms	0 to 10000	Online	
Pn307	Not used.		(Do not	change the setting	ng.)		40				
Pn308	Speed feedback filter time constant	2	Sets co	nstant during filte	0	×0.01 ms	0 to 65535	Online			
Pn310	Vibration detection switches	2	0	Vibration detection	0	Vibration detection not used.	0			Online	
		licnes		Selection	1	Gives warning (A.911) when vibra- tion is detected.					
					2	Gives alarm (A.520) when vibra- tion is detected.					
			1	Not used.		(Do not change the setting.)	0				
			2	Not used.		(Do not change the setting.)	0				
			3	Not used.		(Do not change the setting.)	0				
Pn311	Vibration detection sensitivity	2	Sets the	e vibration detect	ion sensi	tivity.	100	%	50 to 500	Online	
Pn312	Vibration detection level	2	Sets the	e vibration detect	ion level		50	r/min	0 to 500	Online	

Torque Control (Torque Limit) Parameters

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn400	Not used.		(Do not	change the setting	ng.)		30				
Pn401	1st step 1st torque command filter time constant	2	Sets the mands.	Filter time const	ant for in	ternal torque com-	40	×0.01 ms	0 to 65535	Online	
Pn402	Forward torque limit	2	Forward ratio).	d rotation output t	torque lir	nit (rated torque	350	%	0 to 800	Online	
Pn403	Reverse torque limit	2	Reverse ratio).	e rotation output t	nit (rated torque	350	%	0 to 800	Online		
Pn404	Forward rotation external current limit	2	Output rent lim	torque limit during it (rated torque ra	100	%	0 to 800	Online			
Pn405	Reverse rotation external current limit	2	Output rent lim	torque limit during it (rated torque ra	100	%	0 to 800	Online			
Pn406	Emer- gency stop torque	2	Deceler torque	ration torque whe ratio)	en an erro	or occurs (rated	350	%	0 to 800	Online	
Pn407	Speed limit	2	Sets the	e speed limit in to	orque coi	ntrol mode.	3000	r/min	0 to 10000	Online	
Pn408	Torque command	orque 2 0 ommand	0	Selects notch filter 1 function.	0	Notch filter 1 not used.	0			Online	
	setting				1	Notch filter 1 used for torque com- mands.					
			1	Not used.	0	(Do not change the setting.)	0				
			2	Selects notch filter 2 function.	0	Notch filter 2 not used.	0			Online	
					1	Notch filter 2 used for torque com- mands.					
			3	Not used.	0	(Do not change the setting.)	0				
Pn409	Notch fil- ter 1 fre- quency	2	Sets no	tch filter 1 freque	ency for t	orque command.	2000	Hz	50 to 2000	Online	
Pn40A	Notch fil- ter 1 Q value	2	Sets Q	value of notch filt	ter 1.		70	×0.01	50 to 1000	Online	
Pn40C	Notch fil- ter 2 fre- quency	2	Sets the mands.	e notch filter 2 fre	equency	for torque com-	2000	Hz	50 to 2000	Online	
Pn40D	Notch fil- ter 2 Q value	2	Sets Q	value of notch filt	ter 2.		70	×0.01	50 to 1000	Online	
Pn40F	2nd step 2nd torque command filter fre- quency	2	Sets the mands.	e filter frequency	nal torque com-	2000	Hz	100 to 2000	Online		
Pn410	2nd step 2nd torque command filter Q value	2	Sets the	e torque comman	nd filter C	value.	70	×0.01	50 to 1000	Online	
Pn411	3rd step torque command filter time constant	2	Sets the mands.	e filter time const	ant for in	ternal torque com-	0	μs	0 to 65535	Online	

Appendix B

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn412	1st step 2nd torque command filter time constant	2	Sets the torque of	Sets the filter time constant for No. 2 gain internal torque commands.				×0.01 ms	0 to 65535	Online	
Pn413	Not used.		(Do not	change the setti	100						
Pn414	Not used.		(Do not	o not change the setting.)							
Pn420	Damping for vibra- tion sup- pression on stop- ping	2	Sets the	jets the vibration suppression value while stopped.				%	10 to 100	Online	
Pn421	Vibration suppres- sion start- ing time	2	Sets the become stopping	Sets the time from when the position command becomes 0 until damping for vibration suppression on stopping begins.			1000	ms	0 to 65535	Online	
Pn422	Gravity compen- sation torque	2	Sets the	ets the gravity compensation torque.			0	×0.01 %	-20000 to 20000	Online	
Pn456	Sweep torque command amplitude	2	Sets the	e sweep torque c	ommand	amplitude.	15	%	1 to 800	Online	

I/O and Status Parameters

Param-	Parame-	Param- eter size		C	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name		Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn501	Not used.		(Do not	change the se	tting.)		10				
Pn502	Rotation speed for motor rota- tion detec- tion	2	Sets the rotation	e number of rot detection outp	ations for ut (TGON	r the Servomotor N).	20	r/min	1 to 10000	Online	
Pn503	Speed conformity signal out- put width	2	Sets the for the s	ets the allowable fluctuation (number of rotations) or the speed conformity output (VCMP).				r/min	0 to 100	Online	
Pn506	Brake tim- ing 1	2	Sets the vomotor	Sets the delay from the brake command to the Ser- romotor turning OFF.				×10 ms	0 to 50	Online	
Pn507	Brake command speed	2	Sets the comma	ets the number of rotations for outputting the brake ommand.			100	r/min	0 to 10000	Online	
Pn508	Brake tim- ing 2	2	Sets the OFF to	Sets the delay time from the Servomotor turning OFF to the brake command output.			50	×10 ms	10 to 100	Online	
Pn509	Momen- tary hold time	2	Sets the abled w	e time during w hen a power fa	hich alar ilure occ	m detection is dis- urs.	20	ms	20 to 1000	Online	

Param-	Parame-	Param-		C	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn50A Input s nal sel	Input sig- nal selec-	2	0	Not used.	1	(Do not change the setting.)	1				
	tions 1		1	Not used.	8	(Do not change the setting.)	8				
			2	Not used.	8	(Do not change the setting.)	8				
			3	POT (forward drive prohib- ited input)	0	Allocated to CN1, pin 13: Valid for low input	1			Offline	Stan- dard setting:
				signal Input terminal allo- cation	1	Allocated to CN1, pin 7: Valid for low input					1
					2	Allocated to CN1, pin 8: Valid for low input					
					3	Allocated to CN1, pin 9: Valid for low input					
					4	Allocated to CN1, pin 10: Valid for low input					
					5	Allocated to CN1, pin 11: Valid for low input					
					6	Allocated to CN1, pin 12: Valid for low input					
				7	Always enabled.	1					
					8	Always disabled.					
					9	Allocated to CN1, pin 13: Valid for high input					
					A	Allocated to CN1, pin 7: Valid for high input					
					В	Allocated to CN1, pin 8: Valid for high input					
					С	Allocated to CN1, pin 9: Valid for high input					
					D	Allocated to CN1, pin 10: Valid for high input					
					E	Allocated to CN1, pin 11: Valid for high input					
					F	Allocated to CN1, pin 12: Valid for high input					
Pn50B	Input sig- nal selec- tions 2	2	0	NOT (reverse drive prohib- ited input) signal Input terminal allo- cation	0 to F	Same as Pn50A.3	2			Offline	Stan- dard setting: 2
			1	Not used.	8	(Do not change the setting.)	8				
			2 N	Not used.	8	(Do not change the setting.)	8				
			3	Not used.	8	(Do not change the setting.)	8				

Appendix B

Param-	Parame-	Param-		C	Default L	Unit	Setting	Enable	Details					
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting				
Pn50C Input sig- nal selec- tions 3	Input sig- nal selec-	2	0	Not used.	8	(Do not change the setting.)	8							
	tions 3		1	Not used.	8	(Do not change the setting.)	8							
			2	Not used.	8	(Do not change the setting.)	8							
			3	Not used.	8	(Do not change the setting.)	8							
Pn50D	Input sig- nal selec-	2	0	Not used.	8	(Do not change the setting.)	8							
	tions 4		1	Not used.	8	(Do not change the setting.)	8							
			2	Not used.	8	(Do not change the setting.)	8							
			3	Not used.	8	(Do not change the setting.)	8							
Pn50E	Output sig-	2	0	INP1 (posi-	0	Not used.	0			Offline	Stan-			
	tions 1			pleted 1) signal output	1	Allocated to CN1 pins 1, 2					setting: 0			
				terminal allo- cation	2	Allocated to CN1 pins 23, 24								
					3	Allocated to CN1 pins 25, 26				0.111				
			1	VCMP (speed con- formity) sig- nal output terminal allo- cation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0			
						2	TGON (ser- vomotor rota- tion detection) signal output terminal allo- cation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
			3	READY (servo ready) signal output terminal allo- cation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0			
Pn50F	Output sig- nal selec- tions 2	2	0	CLIMT (cur- rent limit detection) signal output terminal allo- cation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0			
					1	VLIMT (speed limit detection) signal output terminal allo- cation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0	
			2	BKIR (brake interlock) signal output terminal allo- cation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 1			
			3	WARN (warning) signal output terminal allo- cation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0			

WWW.BSNEW.IR

Param-	Parame-	Param-		С	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn510	Output sig- nal selec- tions 3	2	0	INP2 (posi- tioning com- pleted 2) signal output terminal allo- cation	0 to 3	Same as Pn50E.0	0			Offline	Stan- dard setting: 0
			1	Not used.	0	(Do not change the setting.)	0				
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				
Pn511	Input sig- nal selec- tions 5	2	0	DEC signal input termi- nal allocation	0	Allocated to CN1, pin 13: Valid for low input	3			Offline	Stan- dard setting:
					1	Allocated to CN1, pin 7: Valid for low input					3
					2	Allocated to CN1, pin 8: Valid for low input					
					3	Allocated to CN1, pin 9: Valid for low input					
					4	Allocated to CN1, pin 10: Valid for low input					
					5	Allocated to CN1, pin 11: Valid for low input					
					6	Allocated to CN1, pin 12: Valid for low input					
					7	Always enabled.					
					8	Always disabled.					
					9	Allocated to CN1, pin 13: Valid for high input					
					A	Allocated to CN1, pin 7: Valid for high input					
					В	Allocated to CN1, pin 8: Valid for high input					
					С	Allocated to CN1, pin 9: Valid for high input					
					D	Allocated to CN1, pin 10: Valid for high input					
					E	Allocated to CN1, pin 11: Valid for high input					
					F	Allocated to CN1, pin 12: Valid for high input					

Param-	Parame-	Param-		C	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn511	Input sig-	2	1	EXT1 signal	0 to 3	Always disabled.	4			Offline	Stan-
	tions 5			nal allocation	4	Allocated to CN1, pin 10: Valid for low input					setting: 4
					5	Allocated to CN1, pin 11: Valid for low input					
					6	Allocated to CN1, pin 12: Valid for low input					
					7	Always enabled.					
					8	Always disabled.					
					9 to C	Always disabled.					
				D	Allocated to CN1, pin 10: Valid for high input						
					E	Allocated to CN1, pin 11: Valid for high input					
					F	Allocated to CN1, pin 12: Valid for high input					
			2	EXT2 signal input termi- nal allocation	0 to F	Same as Pn511.1	5			Offline	Stan- dard setting: 5
			3	EXT3 signal input termi- nal allocation	0 to F	Same as Pn511.1	6			Offline	Stan- dard setting: 6
Pn512	Output sig-	2	0	Output signal	0	Not reversed.	0			Offline	
	nal reverse			reverse for CN1 pins 1, 2	1	Reversed.					
			1	Output signal reverse for CN1 pins 23, 24	0, 1	Same as above	0			Offline	
			2	Output signal reverse for CN1 pins 25, 26	0, 1	Same as above	0			Offline	
			3	Not used.	0	(Do not change the setting.)	0				
Pn513	Not used.		(Do not	change the set	tting.)		0321				
Pn515	Not used.		(Do not	change the set	tting.)		8888				
Pn51B	Not used.		(Do not	change the set	tting.)		1000				
Pn51E	Deviation counter overflow warning level	2	Sets the overflow	e detection leve v warning.	el for the	deviation counter	100	%	10 to 100	Online	
Pn520	Deviation counter overflow level	4	Sets the level.	e deviation cou	nter over	flow alarm detection	262144	Com- mand unit	1 to 1073741823	Online	
Pn522	Position- ing com- pleted range 1	4	Setting (INP1)	range for positi	oning co	mpleted range 1	3	Com- mand unit	0 to 1073741823	Online	
Pn524	Position- ing com- pleted range 2	4	Setting (INP2)	range for positi	oning co	mpleted range 2	3	Com- mand unit	1 to 1073741823	Online	
Param-	Parame-	Param-		C	ontents		Default	Unit	Setting	Enable	Details
----------	---	--------------	-----------------------	--------------------------------------	----------------------	---------------------------	---------	----------------------	--------------------	---------	---------
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn526	Deviation counter overflow level at Servo-ON	4	Sets the level for	e deviation cou Servo ON.	nter over	flow alarm detection	262144	Com- mand unit	1 to 1073741823	Online	
Pn528	Deviation counter overflow warning level at Servo-ON	2	Sets the tion leve	e deviation cou el for Servo ON	nter over	flow warning detec-	100	%	10 to 100	Online	
Pn529	Speed limit level at Servo- ON	2	Sets the with pos	e speed limit fo sition deviation	r when th accumul	e Servo turns ON ated.	10000	r/min	0 to 10000	Online	
Pn52A	Not used.		(Do not	change the se	tting.)		20				
Pn52F	Not used.		(Do not	change the se	tting.)		FFF				

Param-	Parame-	Param-		С	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn530	Program JOG oper- ation related switches	2	0	Program JOG operat- ing pattern	0	(Waiting time Pn535 → Forward movement Pn531) × Number of move- ment operations Pn536	0			Online	
					1	$\begin{array}{l} \mbox{(Waiting time} \\ \mbox{Pn535} \rightarrow \mbox{Reverse} \\ \mbox{movement Pn531} \\ \mbox{\times Number of move} \\ \mbox{ment operations} \\ \mbox{Pn536} \end{array}$					
					2	$\begin{array}{l} \mbox{Waiting time Pn535} \\ \rightarrow \mbox{Forward move-} \\ ment Pn531) \times \\ \mbox{Number of move-} \\ ment operations \\ \mbox{Pn536} \\ \mbox{(Waiting time} \\ \mbox{Pn535} \rightarrow \mbox{Reverse} \\ \mbox{movement Pn531}) \\ \times \mbox{Number of move-} \\ ment operations \\ \mbox{Pn536} \end{array}$					
					3	$\begin{array}{l} (\text{Waiting time} \\ \text{Pn535} \rightarrow \text{Reverse} \\ \text{movement Pn531}) \\ \times \text{Number of move-} \\ \text{ment operations} \\ \text{Pn536} \\ (\text{Waiting time} \\ \text{Pn535} \rightarrow \text{Forward} \\ \text{movement Pn531}) \\ \times \text{Number of move-} \\ \text{ment operations} \\ \text{Pn536} \end{array}$					
					4	$\begin{array}{l} (Waiting time \\ Pn535 \rightarrow Forward \\ movement Pn531 \\ \rightarrow Waiting time \\ Pn535 \rightarrow Reverse \\ movement Pn531) \\ \times Number of movement operations \\ Pn536 \end{array}$					
					5	(Waiting time Pn535 → Reverse movement Pn531 → Waiting time Pn535 → Forward movement Pn531) × Number of move- ment operations Pn536					
			1	Not used.	0	(Do not change the setting.)	0				
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				
Pn531	Program JOG movement distance	4	Sets the	∍ program JOG	movem	ent distance.	32768	Com- mand unit	1 to 1073741824	Online	
Pn533	Program JOG movement speed	2	Sets the	∍ program JOG	operatio	on movement speed.	500	r/min	1 to 10000	Online	

Appendix B

Param-	Parame-	Param-		C	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn534	Program JOG accelera- tion/decel- eration time	2	Sets the JOG op	e acceleration/c eration.	decelerat	ion time for program	100	ms	2 to 10000	Online	
Pn535	Program JOG wait- ing time	2	Sets the tion star	e delay time fro t input until op	m the pro eration st	ogram JOG opera- tarts.	100	ms	0 to 10000	Online	
Pn536	Number of program JOG move- ments	2	Sets the operation	e number of rep ons.	oetitions o	of the program JOG	1	Times	1 to 1000	Online	
Pn540	Gain limit	2	Sets the	e gain limit.			2000	×0.1 Hz	10 to 2000	Online	
Pn550	Analog monitor 1 offset volt- age	2	Sets the	e analog monito	or 1 offse	t voltage.	0	×0.1 V	-10000 to 10000	Online	
Pn551	Analog monitor 2 offset volt- age	2	Sets the	e analog monito	or 2 offse	t voltage.	0	×0.1 V	-10000 to 10000	Online	

Other Parameters

Param- eter No.	Parameter name	Param- eter size	Contents	Default setting	Unit	Setting range	Enable setting	Details
Pn600	Regeneration resistor capacity	2	Setting for regeneration resistance load ratio monitoring calculations	0	×10 W	0 to (varies by model)	Online	

Control Function Parameters

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn800	Communi- cations control	2	0	MECHA- TROLINK com- munications check mask	0	Detects both com- munications errors (A.E6 ⁻) and syn- chronization errors (A.E5 ⁻).	0			Online	Always set to 0.
					1	Ignores communi- cations errors (A.E6□).					
					2	Ignores synchroni- zation errors (A.E5□).					
					3	Ignores communi- cations errors (A.E6 ⁻) and syn- chronization errors (A.E5 ⁻).					
			1	Warning check mask	0	Detects all parame- ter setting warnings (A.94□), MECHA- TROLINK com- mand warnings (A.95□), and com- munications errors (A.96□).	4			Online	Always set to 4 or 0.
					1	Ignores parameter setting warning (A. 94_).					
					2	Ignores command warning (A. 95					
					3	Ignores parameter setting warning (A. 94) and com- mand warning (A. 95).					
					4	Ignores communi- cations warning (A. 96□).					
					5	Ignore setting warning (A. 94) and communica- tions warning (A. 96).					
					6	Ignores command warning (A. 95) and communica- tions warning (A. 96).					
					7	Ignores setting warning (A. 94), command warning (A. 95), and com- munications warn- ing (A. 96).					
			2	Communica- tions error count at single transmission	0 to F	Detects communi- cations errors (A.E60) if they occur consecutively for the set value plus two times.	0			Online	
			3	Not used.	0	(Do not change the setting.)	0				

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details	
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting		
Pn801	Function selection	2	0	Software limit function	0	Software limit enabled.	3			Online		
	tion 6 (software LS)				1	Forward software limit disabled.						
					2	Reverse software limit disabled.						
					3	Forward/reverse software limits disabled.						
				1	Not used.	0	(Do not change the setting.)	0				
				2	Software limit check using reference	0	No software limit check using refer- ence	0			Online	Always set to 0.
					1	Software limit check using refer- ence						
			3	Not used.	0	(Do not change the setting.)	0					

Param- eter No.	Parameter name	Param- eter size	Contents	Default setting	Unit	Setting range	Enable setting	Details
Pn802	Not used.		(Do not change the set- ting.)	0000				
Pn803	Zero point width	2	Sets the origin position detection range.	10	Command unit	0 to 250	Online	
Pn804	Forward soft- ware limit	4	Sets the software limit for the positive direction.	819,191,808	Command unit	-1,073,741,823 to 1,073,741,823	Online	
Pn806	Reverse soft- ware limit	4	Sets the software limit for the negative direction.	-819,191,808	Command unit	-1,073,741,823 to 1,073,741,823	Online	
Pn808	Absolute	4	Sets the encoder posi-	0	Command	-1,073,741,823	Offline	
	point position offset		nate system offsets for when an absolute encoder is used.		unit	1,073,741,823		
Pn80A	First step linear acceleration constant	2	Sets the step 1 accelera- tion for when two-step acceleration is used.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80B	Second step lin- ear acceleration constant	2	Sets the step 2 accelera- tion for when two-step acceleration is exe- cuted, or the one-step acceleration constant for when one-step accelera- tion is executed.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80C	Acceleration constant switch- ing speed	2	Sets the switching speed for the step 1 and step 2 acceleration when two- step acceleration is exe- cuted.	0	×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).

Param- eter No.	Parameter name	Param- eter size	Contents	Default setting	Unit	Setting range	Enable setting	Details
Pn80D	First step linear deceleration constant	2	Sets the step 1 decelera- tion for when two-step deceleration is used.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80E	Second step lin- ear deceleration constant	2	Sets the step 2 decelera- tion for when two-step deceleration is exe- cuted, or the one-step deceleration constant for when one-step decelera- tion is executed.	100	×10,000 command units/s ²	1 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn80F	Deceleration constant switch- ing speed	2	Sets the switching speed for the step 1 and step 2 deceleration when two- step deceleration is exe- cuted.	0	×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn810	Exponential acceleration/ deceleration bias	2	Sets the bias for when an exponential filter is used for the position command filter.	0	Command units/s	0 to 32767	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn811	Exponential acceleration/ deceleration time constant	2	Sets the time constant for when an exponential filter is used for the posi- tion command filter.	0	×0.1 ms	0 to 5100	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn812	Moving average time	2	Sets the average move- ment time for when S- curve acceleration/ deceleration is used, and an average movement filter is used for the posi- tion command filter.	0	×0.1 ms	0 to 5100	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn813	Reserved.		(Do not change the set- ting.)	0				
Pn814	Final travel dis- tance for exter- nal positioning	4	Sets the distance from the external signal input position when external positioning is executed.	100	Command unit	-1,073,741,823 to 1,073,741,823	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).

Param-	Parame-	Param-		Con	tents		Default	Unit	Setting range	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting			setting	
Pn816	Zero point	2	0	Zero point return direc-	0	Forward direction	0			Online	Set the same direc-
	return mode set- tings			tion	1	Reverse direction					tion as the origin search direction set in the Axis Parameters. Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
			1 to 3	Not used.	0	(Do not change the setting.)	0				
Pn817	Zero point return approach speed 1	2	Sets the eration	e origin search limit switch sig	i speed a inal turns	fter the decel- ON.	50	×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn818	Zero point return approach speed 2	2	Sets the eration	e origin search limit switch sig	speed a nal turns	fter the decels ON.	5	×100 com- mand units/s	0 to 65535	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn819	Final travel dis- tance to return to zero point	4	Sets the position is exect	e distance fron to the origin, t uted.	n the latc for when	h signal input origin search	100	Com- mand unit	-1,073,741,823 to 1,073,741,823	Online	Settings can be changed when the axes are stopped (Busy Flag for each axis = 0).
Pn81B to Pn825	Not used.		(Do not	change the se	etting.)		0000				
Pn900 to Pn910	Not used.		(Do not	change the se	etting.)						
Pn920 to Pn95F	Not used.		(Do not	change the se	etting.)						

R7D-ZN -ML2 SMARTSTEP Junior Servo Drive with Built-in MECHATROLINK-II Communications

Function Selection Parameters

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn000	Function selection basic	2	0	Reverse rota- tion	0	CCW direction is taken for positive command	0			Offline	
	switches				1	CW direction is taken for negative command					
					2 to 3	Not used.					
			1	Not used.	1	(Do not change the setting.)	0				
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				
Pn00A	Filter set- tings	2	Sets the When u the SM	e filter time consta using this parame ARTSTEP Junior	ants. ter, turn Servo D	ON bit 4 of SW2 on rive.	0000		0000 to 000F	Online	

Position Control Parameters

Param- eter No.	Parame- ter name	Param- eter size	Contents	Default setting	Unit	Setting range	Enable setting	Details
Pn20E	Electronic gear ratio G1(numer- ator)	4	Sets the pulse rate for the command pulses and Servomotor travel distance. $0.01 \le G1/G2 \le 100$	1		1 to 107374 1824	Offline	
Pn210	Electronic gear ratio G2 (denomi- nator)	4		1		1 to 107374 1824	Offline	

I/O and Status Parameters

Param-	Parame-	Param-		Co	ontents		Default	Unit	Setting	Enable	Details									
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting										
Pn50A	Input sig- nal selec-	2	0	Not used.	1	(Do not change the setting.)	1													
	tion 1		1	Not used.	8	(Do not change the setting.)	8													
			2	Not used.	8	(Do not change the setting.)	8													
			3	POT (forward drive prohibited input) signal input terminal	2	Allocated to CN1, pin 4: Valid for low input	2			Offline	Stan- dard setting: 2									
				allocation	8	Always invalid.														
Pn50B	Input sig- nal selec- tion 2	2	0	NOT (reverse drive prohibited input) signal	3	Allocated to CN1, pin 3: Valid for low input	3			Offline	Stan- dard setting:									
										-			allocation	8	Always invalid.					3
			1	Not used.	8	(Do not change the setting.)	8													
			2	Not used.	8	(Do not change the setting.)	8													
			3	Not used.	8	(Do not change the setting.)	8													

Param-	m- Parame- Param- No. ter name eter D size I			Co	ontents		Default	Unit	Setting	Enable	Details
eter No.			Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn515	Input sig- nal selec-	2	0	Not used.	8	(Do not change the setting.)	8				
	tion 7		1	Not used.	8	(Do not change the setting.)	8				
	2		2	Emergency 4 stop (STOP) signal input pin		Allocated to CN1, pin 6: Valid for low input	4			Offline	Stan- dard setting:
			allocation		8	Always invalid.	8				4
			3	Not used.	8	(Do not change the setting.)	8				
Pn522	Position- ing com- pletion range 1	4	Sets the	e width of the pos	completed range.	10	Com- mand unit	0 to 1,073,7 41,824	Online		
Pn524	Position- ing com- pletion range 2	4	Sets the s	e allowable fluctu Speed Conformity	ation ran y Flag.	ge (rotation speed)	100	Com- mand unit	0 to 1,0737, 41,824	Online	

Control Function Parameters

Param-	Parame-	Param-		Co	ontents		Default	Unit	nit Setting Enable		Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn800	Communi- cations	2	0	Not used.	0	(Do not change the setting.)	0				
	control		1	Warning check mask	0	Detects all parame- ter setting warnings (A.94), MECHA- TROLINK com- mand warnings (A.95), and com- munications errors (A.96).	4			Online	Always set to 4 or 0.
					1	Ignores parameter setting warning (A. 94).					
					2	Ignores command warning (A.95).					
					3	Ignores setting warning (A.94) and command warning (A.95).					
					4	Ignores communi- cations warning (A.96).					
					5	Ignore setting warning (A.94) and communications errors (A.96).					
				6	6	Ignores command warnings (A.95) and communica- tions errors (A.96).					
					7	Ignores setting warning (A.94, command warning (A.95), and com- munications errors (A.96).					
			2	Not used.	0	(Do not change the setting.)	0				
			3	Not used.	0	(Do not change the setting.)	0				
Pn801	Function selection	2	0	Software limit function	0	Software limit enabled.	3			Online	
	tion 6 (software				1	Forward software limit disabled.					
	LS)				2	Reverse software limit disabled.					
					3	Forward/reverse software limits disabled.					
			1	Not used.	0	(Do not change the setting.)	0				
			2	Software limit check using reference	0	No software limit check using refer- ence	0			Online	Always set to 0.
					1	Software limit check using refer- ence					
			3	Not used.	0	(Do not change the setting.)	0				
Pn803	Zero point width	2	Sets th	e origin position o	detection	range.	10	Com- mand unit	0 to 250	Online	

Param-	Parame-	Param-		Cc	ontents		Default	Unit	Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn804	Forward software limit	4	Sets the	software limit fo	r the pos	itive direction.	1,073,7 41,823	Com- mand unit	-1,073, 741,823 to 1,073,7 41,823	Online	
Pn806	Reverse software limit	4	Sets the	e software limit fo	ative direction.	-1,073, 741,823	Com- mand unit	-1,073, 741,823 to 1,073,7 41,823	Online		
Pn80B	Linear accelera- tion con- stant	2	Sets the deceler	⇒ acceleration cor ation curves for p	nstant for	r the acceleration/ g.	100	×10,0 00 com- mand units/ s ²	1 to 65535	Online	Set- tings can be change d when the axes are stopped (Busy Flag for each axis = 0).
Pn80E	Linear decelera- tion con- stant	2	Sets the deceler	deceleration con ation curves for p	nstant fo bositionin	r the acceleration/ g.	100	×10,0 00 com- mand units/ s ²	1 to 65535	Online	Set- tings can be change d when the axes are stopped (Busy Flag for each axis = 0).
Pn814	Final travel distance for exter- nal posi- tioning	4	Sets the tion who	→ distance from the second	ne extern oning is e	al signal input posi- executed.	100	Com- mand unit	-1,073, 741,823 to 1,073,7 41,823	Online	Set- tings can be change d when the axes are stopped (Busy Flag for each axis = 0).

Param-	Parame-	Param-	1	Co	ontents		Default Unit		Setting	Enable	Details
eter No.	ter name	eter size	Digit No.	Name	Set- ting	Explanation	setting		range	setting	
Pn816	Zero point return mode set- tings	2	0	Zero point return direction	0	Forward direction Reverse direction	0				Set the same direc- tion as the ori- gin search direc- tion set in the Axis Param- eters. Set- tings can be change d when the axes are stopped (Busy Flag for each axis = 0).
			1 to 3	Not used.	0	(Do not change the setting.)	0				
Pn817	Zero point return approach speed 1	2	Sets the limit sw	∍ origin search sp itch signal turns (veed afte DN.	r the deceleration	50	×100 com- mand units/s	0 to 65535	Online	Set- tings can be change d when the axes are stopped (Busy Flag for each axis = 0).
Pn818	Zero point return approach speed 2	2	Sets the limit sw	ets the origin search speed after the deceleration nit switch signal turns ON.				×100 com- mand units/s	0 to 65535	Online	Set- tings can be change d when the axes are stopped (Busy Flag for each axis = 0).
Pn819	Final travel distance to return to zero point	4	Sets the to the o	ets the distance from the latch signal input position the origin, for when origin search is executed.				Com- mand unit	-1,073, 741,823 to 1,073,7 41,823	Online	Set- tings can be change d when the axes are stopped (Busy Flag for each axis = 0).
Pn820 to Pn823	Not used.		(Do not	change the setting	ng.)		0000				

Appendix C Operation Area I/O Allocations

Common Operating Memory Area

Common Operating Output Memory Area Allocations

 $n = CIO 1500 + (unit number \times 25)$

I/O	Word	Bits	Category	Name	Function
Output (CPU	n	00		UNIT ERROR RESET	r Clears the Unit common error that occurred.
Unit to PCU)		01	Data trans-	WRITE DATA	r : Starts writing data.
,		02	mands	READ DATA	▲: Starts reading data
		03		SAVE DATA	_ Starts saving data
		04 to 15		Not used (reserved by the system).	
	n+1	00	Communi-	CONNECT	▲: Establishes connection.
	01 to 05	cations		\mathbf{V} : Releases connection.	
		01 to 05	commands	Reserved by the sys- tem.	
		06		WRITE BACKUP DATA	_ Starts writing backup data.
		07		READ BACKUP DATA	_ Starts reading backup data.
		08 to 14		Not used (reserved by the system).	
		15		REJOIN (See note.)	\mathbf{r} : Starts processing for axes to rejoin the connection.
	n+2			Axes to connect (See note.)	Sets the following for the axes registered in the scan list (bits 0 to 15 correspond to axes 1 to 16).
					0: Start MECHATROLINK communications.
					1: Do no start MECHATROLINK communications.
	n+3 to n+5			Not used (reserved by the system).	

I/O	Word	Bits	Category	Name	Function
Output (CPU	n+6		Operating data for	Number of write words	Specifies the number of words to be written to the PCU.
Unit to PCU)	Unit to PCU) n+7		data trans- fer	Write source area	Specifies the area of the data to be written to the PCU. 00B0 hex: CIO Area 00B1 hex: WR Area 00B2 hex: HR Area 00B3 hex: AR Area
					0082 nex: DM Area 005⊡ hex: EM Area □: EM bank No.= 0 to 9, A, B, C
	n+8			Write source word	Specifies in hexadecimal the beginning word of the area for the data to be written to the PCU.
	n+9			Write destination address	Specifies the write destination address in the PCU.
	n+10			Number of read words	Specifies the number of words to be read from the PCU.
	n+11			Read source address	Specifies the read source address in the PCU.
	n+12			Read destination area	Specifies the area used to store the data read from the PCU. 00B0 hex: CIO Area 00B1 hex: WR Area 00B2 hex: HR Area 00B3 hex: AR Area 0082 hex: DM Area 005 hex: EM Area
	n+13		1	Read destination word	Specifies in hexadecimal the beginning word of the area where the data read from the PCU is stored.
	n+14			Not used (reserved by the system).	

Note The REJOIN Bit and the Axes to Connect parameter are supported for unit version 2.0 or later. The allocated bit and word are not used for earlier unit versions.

Common Operating Input Memory Area Allocations

 $n = CIO 1500 + (unit number \times 25)$

I/O	Word	Bits	Category	Name	Function
Input (PCU	n+15	00 to 11	PCU com- mon status	Not used (reserved by the system).	
to CPU Unit)		12	words	Unit Error Flag	0: Unit common error has not occurred. 1: Unit common error has occurred.
		13		Not used (reserved by the system).	
		14		Data Transferring Flag	0: Powerup, transfer completed, or transfer failed. 1: Transferring data.
		15		Not used (reserved by the system).	
	n+16	00 to 11		Reserved by the sys- tem.	
		12		Memory Card Trans- fer Error	0: No Memory Card transfer error has occurred. 1: A Memory Card transfer error has occurred.
		13		Reserved by the sys- tem.	
		14		Unit Busy Flag	0: PCU is not busy. 1: PCU is busy.
		15		Connection Status Flag	0: MECHATROLINK communications stopped. 1: MECHATROLINK communications executing.
	n+17to n+20			Reserved by the sys- tem.	
	n+21			Unit error code	Shows the error code for the Unit common error that occurred.

I/O	Word	Bits	Category	Name	Function
Input (PCU	n+22	00	Axis com- munica-	Axis 1 communica- tions status	Bits 00 to 15 correspond to the communications sta- tus for axes 1 to 16.
to CPU Unit)		01	tions status bits	Axis 2 communica- tions status	The bits will turn ON if the corresponding axes regis- tered in the scan list are communicating normally.
		02		Axis 3 communica- tions status	0: Communications for the corresponding axis have stopped, or the axis not registered in scan list.
		03		Axis 4 communica- tions status	1: The corresponding axis is communicating.
		04		Axis 5 communica- tions status	
		05		Axis 6 communica- tions status	
		06		Axis 7 communica- tions status	
		07		Axis 8 communica- tions status	
		08		Axis 9 communica- tions status	
		09		Axis 10 communica- tions status	
		10		Axis 11 communica- tions status	
		11		Axis 12 communica- tions status	
		12		Axis 13 communica- tions status	
		13		Axis 14 communica- tions status	
		14		Axis 15 communica- tions status	
		15		Axis 16 communica- tions status	
	n+23to n+24			Not used (reserved by the system).	

Axis Operating Output Memory Area Allocations

a = Beginning word of Axis Operating Output Memory Areas specified in Common Parameters + (Axis No. –1) $\times\,25$

I/O	Word	Bits	Category	Name	Function
Output (CPU Unit to	a	00	Direct oper- ation com- mands	LINEAR INTERPO- LATION SETTING (See note 1.)	r Starts setting linear interpolation.
PCU)		01		LINEAR INTERPO- LATION START (See note 1.)	: Starts linear interpolation. (Valid only when LINEAR INTERPOLATION SET- TING = 1.)
		02		Not used (reserved by the system).	
		03		ABSOLUTE MOVE- MENT	r : Starts absolute movement.
		04	• •	RELATIVE MOVE- MENT	: Starts relative movement.
		05		INTERRUPT FEED- ING	1: Enables interrupt feeding.
		06	Origin posi-	ORIGIN SEARCH	r : Starts origin search.
		07	tioning commands	ORIGIN RETURN	r: Starts origin return.
		08		PRESENT POSI- TION PRESET	: Starts present position preset.
		09	Com-	JOG	_ Starts jogging.
			special		ᆛ : Stops jogging.
		10	functions	Direction designation	0: Forward rotation direction 1: Reverse rotation direction
		11	·	Not used (reserved by the system).	
		12		ERROR RESET	\mathbf{F} : Clears the axis error that occurred.
		13		DEVIATION COUNTER RESET (See note 4.)	∴ Starts DEVIATION COUNTER RESET.
		14		Override Enable Bit	0: Override disabled. 1: Override enabled.
		15		DECELERATION STOP	r Starts deceleration stop.
	a+1	00	Com-	SERVO LOCK	F : Starts SERVO LOCK.
		01	special	SERVO UNLOCK	F : Starts SERVO UNLOCK.
		02	functions	SPEED CONTROL	r : Starts speed control.
		03		TORQUE CON- TROL	r : Starts torque control.
		04 to 10		Not used (reserved by the system).	
		11	Servo	DEVICE SETUP	r Starts device setup for the Servo Drive.
		12	transfer	WRITE SERVO PARAMETER	: Starts writing Servo Parameter.
		13		READ SERVO PARAMETER	r Starts reading Servo Parameter.
		14		SAVE SERVO PARAMETER	: Starts writing Servo Parameter to non-volatile memory.
		15	Special command	EMERGENCY STOP	: Starts emergency stop.

I/O	Word	Bits	Category	Name	Function
Output (CPU Unit to	a+2		Operating data for position	Position command value (rightmost word)	Position command value (rightmost word) Position command value (leftmost word) Unit: command unit
PCU)	a+3		control	Position command value (leftmost word)	
	a+4			Speed command value (rightmost word)	Speed command value (rightmost word) Speed command value (leftmost word) Unit: Command units/s
	a+5			Speed command value (leftmost word)	
	a+6		Speed con- trol data	Speed command value for speed con- trol (rightmost word)	Speed command value (rightmost word) Speed command value (leftmost word) Unit: 0.001%
	a+7			Speed command value for speed con- trol (leftmost word)	
-	a+8		Torque con- trol data	Torque command value (rightmost word)	Torque command value (rightmost word) Torque command value (leftmost word) Unit: 0.001%
	a+9			Torque command value (leftmost word)	
	a+10		Position/ speed/ torque con-	Option command value 1 (rightmost word)	Linear interpolation: Interpolation acceleration time, Unit: ms Speed control:
	a+11		trol data	Option command value 1 (leftmost word)	Torque limit/torque feed forward, Unit: % Torque control: Speed limit, Unit: 0.001%
	a+12			Option command value 2 (rightmost word)	Linear interpolation: Interpolation deceleration time, Unit: ms Speed control:
	a+13			Option command value 2 (leftmost word)	Torque limit, Unit: %
	a+14		Operating data for special function	Override	Override ratio (unit: 0.01%)
	a+15	00 to 03	Operating	Monitor 1 type	Specifies monitor type for monitor 1.
		04 to 07	data for expanded	Monitor 2 type	Specifies monitor type for monitor 2.
		08 to 15	monitoring	Not used (reserved by the system).	

I/O	Word	Bits	Category	Name	Function
Output (CPU	a+16	00 to 02		Reserved by the sys- tem.	
Unit to PCU)		03	Accelera- tion/decel-	Exponential curve designation	1: Use exponential acceleration/deceleration curve.
		04	eration curves	S-curve designation	1: Use S-curve acceleration/deceleration curve.
		05 to 13		Reserved by the sys- tem.	
		14	Torque limit	Forward rotation cur- rent limit designation	1: Use forward torque limit.
a+		15		Reverse rotation cur- rent limit designation	1: Use reverse torque limit.
	a+17		Operating data for	Servo Parameter No.	Specifies the parameter number of the Servo Drive that will transfer the data.
	a+18transferrin Servoa+19a+20		transferring Servo	Parameter size	Specifies the data length of the parameter to be transferred in byte units.
			Farameters	Write data (rightmost word)	Specifies the data to be written to the Servo Drive.
			Write data (leftmost word)		
	a+21	00	Linear inter- polation data	Interpolation axis S designation for axis 1 (See note 2.)	Specify the axes for linear interpolation for combina- tions of axes 1 to 4. Bits 00 to 03 correspond to axes 1 to 4.
		01		Interpolation axis designation for axis 2 (See note 2.)	0: Not an interpolation axis. 1: Interpolation axis.
		02		Interpolation axis designation for axis 3 (See note 2.)	
		03		Interpolation axis designation for axis 4 (See note 2.)	
		04		Interpolation axis designation for axis 5 (See note 3.)	Specify the axes for linear interpolation for combina- tions of axes 5 to 8. Bits 04 to 07 correspond to axes 5 to 8.
		05		Interpolation axis designation for axis 6 (See note 3.)	0: Not an interpolation axis. 1: Interpolation axis.
		06		Interpolation axis designation for axis 7 (See note 3.)	
		07		Interpolation axis designation for axis 8 (See note 3.)	
		08 to 15		Not used (reserved by the system).	

I/O	Word	Bits	Category	Name	Function
Output (CPU Unit to	a+22	00	Linear inter- polation data	Interpolation posi- tion designation for axis 1 (See note 2.)	Specify absolute or relative positioning for the axes for linear interpolation. Bits 00 to 03 correspond to axes 1 to 4.
PCU)		01		Interpolation posi- tion designation for axis 2 (See note 2.)	0: Absolute positioning 1: Relative positioning
		02		Interpolation posi- tion designation for axis 3 (See note 2.)	
		03	1	Interpolation posi- tion designation for axis 4 (See note 2.)	
		04		Interpolation posi- tion designation for axis 5 (See note 3.)	Specify absolute or relative positioning for the axes for linear interpolation. Bits 04 to 07 correspond to axes 5 to 8.
		05	·	Interpolation posi- tion designation for axis 6 (See note 3.)	0: Absolute positioning 1: Relative positioning
		06		Interpolation posi- tion designation for axis 7 (See note 3.)	
		07		Interpolation posi- tion designation for axis 8 (See note 3.)	
		08 to 15		Not used (reserved by the system).	
	a+23			Interpolation speed command value (rightmost word) (See note 1.)	Interpolation speed command value (rightmost word)
	a+24			Interpolation speed command value (leftmost word) (See note 1.)	Interpolation speed command value (leftmost word) Unit: Command units/s

Note (1) Allocated in Axis Operating Output Memory Areas for axis 1 and axis 5 for Position Control Unit Ver. 1.1 or later. These bits are not used in the Axis Operating Output Memory Areas for other axes.

- (2) Allocated in Axis Operating Output Memory Area for axis 1 for Position Control Unit Ver. 1.1 or later. These bits are not used in the Axis Operating Output Memory Areas for other axes.
- (3) Allocated in Axis Operating Output Memory Area for axis 5 for Position Control Unit Ver. 1.1 or later. These bits are not used in the Axis Operating Output Memory Areas for other axes.
- (4) The DEVIATION COUNTER RESET can be used with unit version 1.3 or later. This bit is not used for earlier unit versions.

Axis Operating Input Memory Area Allocations

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) × 25

I/O	Word	Bits	Category	Name	Function
Input	b	00	Axis Con-	Axis Con- trol Status mand Flag Flags	0: Command reception enabled.
(PCU			trol Status Flags		: Command reception started.
Unit)					1: Receiving command (command reception dis- abled)
		01 to 04		Not used (reserved by the system).	Always 0.
	05		PCU Positioning Completed Flag	_ r Positioning is completed.	
		06		No Origin Flag	0: Origin established. 1: No origin established.
	07		Origin Stop Flag	0: Outside origin range. 1: Within origin range.	
		08 to 10		Not used (reserved by the system).	Always 0.
		11		Warning Flag	0: No warning has occurred. 1: Warning has occurred.
		12	-	Error Flag	0: No axis error has occurred. 1: Axis error has occurred.
		13		Busy Flag	0: Axis is not busy. 1: Axis is busy.
		14		Servo Parameter Transferring Flag	0: Powerup, transfer completed, or transfer failed.1: Servo Parameter transferring.
		15]	Stop Execution Flag	✓: Stop operation completed.

I/O	Word	Bits	Category	Name	Function	
Input (PCU	b+1	00 to 02	Servo Sta- tus Flags	Reserved by the sys- tem.	Used by the PCU system.	
to CPU Unit)		03		Servo ON Flag (SVON)	0: Servo unlocked. 1: Servo locked.	
		04		Main Power ON Flag	0: Main circuit power supply OFF	
				(PON)	1: Main circuit power supply ON	
		05 and 06			Reserved by the sys- tem.	Used by the PCU system.
		07		Position Completed (PSET) Flag/Speed Conformity (V-CMP) Flag	 During position control: Position Completed Flag 0: Servo Drive executing position control command or the position is outside <i>Positioning Completion</i> <i>Range 1.</i> 1: Servo Drive finished executing position control command and the position is within <i>Positioning</i> <i>Completion Range 1.</i> During speed control: Speed Conformity Flag 0: Speed does not match the speed designation for speed control. 1: Speed matches the speed designation for speed control. 	
		08	*	Distribution Com- pleted (DEN) Flag/ Zero Speed (ZSPD) Flag	 During position control: Distribution Completed Flag 0: Servo Drive is executing a position control command. 1: Servo Drive has completed executing a position control command. During speed control: Zero Speed Flag 0: Not at zero speed. 1: Detecting zero speed. 	
		09		Torque Limit (T_LIM) Status Flag	0: Torque limit function disabled. 1: Torque limit function enabled.	
		10		Reserved by the sys- tem.	Used by the PCU system.	
		11		Positioning Proxim- ity (NEAR) Flag/ Speed Limit (V_LIM) Status Flag	 During position control: Positioning Proximity Flag 0: Outside <i>Positioning Completion Range 2</i>. 1: Within <i>Positioning Completion Range 2</i>. During torque control: Speed Limit Status Flag 0: Speed limit function disabled. 1: Speed limit function enabled. 	
		12		Forward Software Limit Flag (P_SOT)	0: Within forward software limit range. 1: Forward software limit range exceeded.	
		13		Reverse Software Limit Flag (N_SOT)	0: Within reverse software limit range. 1: Reverse software limit range exceeded.	
		14 to 15		Reserved by the sys- tem.	Used by the PCU system.	
	b+2			Reserved by the sys- tem.	Used by the PCU system.	

I/O	Word	Bits	Category	Name	Function
Input (PCU	b+3	00	External I/O status	Forward rotation limit input	Returns the status of I/O signals for each axis. 1: Signal valid
to CPU Unit)		01		Reverse rotation limit input	0: Signal invalid The external I/O status shows the enabled/disabled
		02]	Origin proximity input	status of signals used for PCU control.
		03		Encoder Phase A input	This status is different from the electrical ON/OFF status of the I/O signals.
		04		Encoder Phase B input	
		05		Encoder Phase Z input	
		06		External latch signal 1 input	
		07		External latch signal 2 input	
		08		External latch signal 3 input	
		09		Brake output	
		10		Emergency stop input (See note 2.)	
		11 to 15		Reserved by the sys- tem.	
	b+4		Errors	Axis error code	Displays axis error code/warning code for each axis.
	b+5	00 to 03	Expanded	Monitor 1 type	Shows monitor type for monitor 1.
		04 to 07	monitoring	Monitor 2 type	Shows monitor type for monitor 2.
	08 to 15			Not used (reserved by the system).	Always 0.
	b+6		Present position	Feedback present position (rightmost word)	Present position: Feedback present position (rightmost word) Feedback present position (leftmost word)
	b+7			Feedback present position (leftmost word)	
	b+8			Command present position (rightmost word)	Present position: Command position (rightmost word) Command position (leftmost word)
	b+9			Command present position (leftmost word)	
	b+10		Expanded monitoring	Monitor 1 (rightmost word)	Monitor value specified for monitor 1 type.
	b+11			Monitor 1 (leftmost word)	
	b+12			Monitor 2 (rightmost word)	Monitor value specified for monitor 2 type.
	b+13			Monitor 2 (leftmost word)	
	b+14		Servo Parameter	Read data (rightmost word)	Contains the Servo Parameter that was read.
	b+15		data	Read data (leftmost word)	
	b+16to b+23			Not used (reserved by the system).	Always 0.

I/O	Word	Bits	Category	Name	Function
Input (PCU to CPU Unit)	b+24	00	Linear inter- polation status	Linear interpolation setting completed (See note 1.)	 0: Linear interpolation setting will be accepted. : Linear interpolation setting accepted. Linear interpolation setting completed. (New
Unit,					setting not accepted.)
		01 to 12		Not used (reserved by the system).	Always 0.
		13		Linear interpolation executing (See note 1.)	1: Linear interpolation operation in progress or linear interpolation command in progress.
		14 and 15		Not used (reserved by the system).	Always 0.

Note (1) Allocated in Axis Operating Input Memory Areas for axis 1 and axis 5 for Position Control Unit Ver. 1.1 or later. These bits are not used in the Axis Operating Output Memory Areas for other axes.

(2) The emergency stop input status is input only for SMARTSTEP Junior Servo Drives. This status is not used (reserved by the system) for W-series Servo Drives.

Appendix D List of Error Codes

PCU Common Errors

		Indi	cator st	atus		Category	Error name	Error code
	RUN	ERC	ERH	ERM	MLK			
Errors at powerup	Not lit	Not lit	Not lit	Not lit	Not lit	CPU Unit power interrup- tion	Power Interruption	
						Unit system error	Watchdog Timer Opera- tion	
	Not lit	Not lit	Lit	Not lit	Not lit	Setting error	Unit Number Error	
							Unit Recognition Incomplete	
							Data Transfer Error	
	Not lit	Lit	Not lit	Not lit	Not lit	Unit malfunction	Common Memory Error	
	Not lit	Lit	Lit	Not lit	Not lit	Unit recognition error	Unit Recognition Error	
CPU	Lit	Flash-	Lit	Not lit	Not lit	CPU Unit error	CPU Fatal Error	000A
Unit errors		ing					CPU Unit Watchdog Timer Error	000B
							CPU Unit Monitor Error	000C
							Bus Error	000D
PCU	Lit	Lit	Not lit Not lit	Not lit	Not lit	Unit error	MLK Device Error	0026
internal errors						MLK Device Initialization Error	0030	
						Data corrupted	Memory Error	00F1
MECHA-	Lit	Flash-	Not lit	Lit	Unde-	Scan list mismatch	MLK Initialization Error	0020
TROLINK commu- nications errors		ing			ter- mined	Communications error	MLK Communications Error	0025
PCU set-	Lit	Flash-	Not lit	Not lit	Unde-	Illegal operation	Multistart Error	0021
tings and opera-		ing			ter- mined		FINS Command Monitor- ing Error	0024
errors						Illegal data	Write Transfer Error	0022
							Read Transfer Error	0023
							Transfer Cycle Setting Error	0027
						Initialization Common Parameter Check Error	0028	
							Data Transfer Common Parameter Check Error	0029

Individual Axis Errors

		Indi	cator sta	atus		Category	Error name	Error code
	RUN	ERC	ERH	ERM	MLK			
MECHA- TROLINK	Lit	Flash- ing	Not lit	Lit	Unde- ter-	Communications error	Synchronous Communi- cations Alarm	3010
commu-					mined		Communications Alarm	3011
errors							Command Timeout	3012

List of Error Codes

		Indi	cator sta	atus		Category	Error name	Error code
	RUN	ERC	ERH	ERM	MLK			
PCU set- tings and	Lit	Flash- ing	Not lit	Not lit	Lit	Illegal operation	Present Position Unknown Error	3030
opera-							Servo Unlock Error	3040
errors							Multistart Error	3050
PCU set- tings and	Lit	Flash- ing	Not lit	Not lit	Lit	Illegal data	Position Designation Error	3060
opera-							Speed Designation Error	3061
errors							Speed Control Speed Designation Error	3062
							Torque Command Value Error	3063
							Option Command Value 1 Error	3064
							Option Command Value 2 Error	3065
							Override	3070
						Initialization Axis Param- eter Check Error	3090	
							Data Transfer Axis Parameter Check Error	3091
							Data Setting Error	3099
MECHA- TROLINK	Lit	Flash- ing	Not lit	Flash- ing	Lit	External sensor input	Forward Rotation Limit Input	3000
slave sta- tion							Reverse Rotation Limit	3001
errors							Forward Software Limit	3002
							Reverse Software Limit	3003
							Emergency Stop Signal Input	3004
						Origin search error	No Origin Proximity or Origin Input Signal	3020
							Limit Input Already ON	3021
							Limit Input Signal ON in Both Directions	3022
						Servo Drive error	Drive Main Circuit OFF Error	3080
						MECHATROLINK device alarm		4000 + Alarm code for each device

Alarm Displays for Servo Drives

The following table lists the alarm displays for Servo Drives.

When an alarm or warning occurs for one of the connected MECHATROLINK slave station devices, the PCU turns ON the Error Flag or Warning Flag and returns the following error code.

G5-series Servo Drives/G-series Servo Drives

Error code: 4

The numbers in the boxes of the error codes displayed for alarms are given as decimal numbers for the Servo Drives and as hexadecimal numbers for the Position Control Units.

The same numbers are given in the boxes of the error codes displayed for warnings for the Servo Drives and the Position Control Units.

Example: Position deviation overflow alarm (deviation counter overflow)

Servo Drive alarm display: A.24

The PCU's error code in this example is 4018.

Example: Battery warning

Servo Drive warning display: A.92

The PCU's error code in this example is 4092.

For details on alarm display and warning display numbers, and the difference between Servo Drives and Position Control Units, refer to the list in Appendix D List of Error Codes.

W-series and SMARTSTEP Junior Servo Drives

Error code: 40 hex

The boxes $(\Box\Box)$ indicate the alarm/warning display number for the Servo Drive.

When using an R88D-WN -ML2 W-series Servo Drive with built-in MECHATROLINK-II communications, how-ever, only the upper two digits of the display number's three digits will be displayed.

The alarms that occur in the Servo Drive correspond to error codes that are detected by the PCU when a MECHATROLINK communications connection has been established, as shown in the following table.

When using an R88D-WT Servo Drive, the two digits following "A." in the Servo Drive's display number in the table are displayed. When using an R88D-WN -ML2 W-series Servo Drive, three digits are displayed (the value indicated by the display number box is used to further classify the alarm).

For details on each alarm and countermeasures, refer to the operation manual for the Servo Drive being used.

List of G5-series Alarm Displays

The Servo Drive has built-in protective functions. When a protective function is activated, the Servo Drive turns OFF the alarm output signal (ALM) and switches to the Servo OFF status.

The alarm code will be displayed on the front panel.

Alarm type	Description
	Protective function that allows the alarm to be reset, and leaves record in the alarm history.
PR	Protective function that does not allow the alarm to be reset, and requires the control power supply to be turned OFF and turned ON again after resolving the problem.
Х	Protective function that does not leave record in the alarm history.

Servo Drive display	PCU error code	Alarm type	Error detection function	Detected error or cause of error
11	400B	х	Control power supply und- ervoltage	The DC voltage of the main circuit has dropped below the specified value.
12	400C		Overvoltage	The DC voltage of the main circuit is abnormally high.
13	400D	х	Main power supply under- voltage	The DC voltage of the main circuit is low.
14	400E	PR	Overcurrent	Overcurrent flowed to the IGBT. Servomotor power line ground fault or short circuit.
15	400F	PR	Servo Drive overheat	The temperature of the Servo Drive radiator exceeded the specified value.
16	4010		Overload	Operation was performed with torque significantly exceeding the rating for several seconds to several tens of seconds.
18	4012	PR	Regeneration overload	The regenerative energy exceeded the processing capacity of the regeneration resistor.
21	4015	PR	Encoder communications error	Communications between the encoder and the Servo Drive failed for a specified number of times, thereby activating the error detection function.
23	4017	PR	Encoder communications data error	Communications error occurred for the data from the encoder.

List of Error Codes

Servo Drive display	PCU error code	Alarm type	Error detection function	Detected error or cause of error
24	4018		Error counter overflow	The number of position deviation pulses exceeded the Error Counter Overflow Level (Pn014).
25	4019	PR	Excessive hybrid error	The difference between the position of load from external encode and the position of motor due to encoder was larger than the number of pulses set by Internal/External Feedback Pulse Error Counter Overflow Level (Pn328).
26	401A		Overspeed	The rotation speed of the Servomotor exceeded the setting of the Overspeed Detection Level Setting at Emergency Stop (Pn615) or Overspeed Detection Level Setting (Pn513).
27	401B	PR	Command error	The operation command resulted in an error.
29	401D		Internal error counter over- flow	The value of the internal deviation counter (internal control unit) exceeded 2^{27} (134217728).
30	401E	PR	Safety input error	Safety input signal turned OFF.
33	4021		Interface duplicate alloca- tion error	A duplicate setting for the interface I/O signals was detected.
34	4022		Overrun limit error	The Servomotor exceeded the allowable operating range set in the Overrun Limit Setting (Pn514) with respect to the posi- tion command input.
36	4024	PR X	Parameter error	Data in the parameter save area was corrupted when the data was read from the EEPROM at power-ON.
37	4025	PR X	Parameter destruction	The EEPROM write verification data was corrupted when the data was read from the EEPROM at power-ON.
38	4026	х	Drive prohibition input error	Forward and Reverse Drive Prohibit Inputs (NOT and POT) both became OPEN.
40	4028	PR	Absolute encoder system down error	The voltage supplied to the absolute encoder dropped below the specified value.
41	4029	PR	Absolute encoder counter overflow error	The multi-turn counter of the absolute encoder exceeded the specified value.
42	402A	PR	Absolute encoder over- speed error	The Servomotor rotation speed exceeded the specified value when power to the absolute encoder is supplied by the battery only.
43	402B	PR	Encoder initialization error	An encoder initialization error was detected.
44	402C	PR	Absolute encoder 1-rota- tion counter error	A one-turn counter error was detected.
45	402D	PR	Absolute encoder multi- rotation counter error	A multi-turn counter error or phase-AB signal error was detected.
47	402F		Absolute encoder status error	The rotation of the absolute encoder is higher than the spec- ified value.
48	4030	PR	Encoder phase Z error	A phase-Z pulse was not detected regularly.
49	4031	PR	Encoder CS signal error	A logic error was detected in the CS signal.
50	4032	PR	External encoder error	An error was detected in external encoder connection or communications data.
51	4033	PR	External encoder status error	An external encoder error code was detected.
55	4037	PR	External encoder connec- tion error	An error was detected in the external encoder phase A, B, or Z connection.
82	4052	PR	Node address setting error	The rotary switch for setting the node address of the Servo Drive was set out of range.
83	4053		Communications error	Data received during each MECHATROLINK-II communica- tions cycle repeatedly failed, exceeding the number of times set in the Communications Control (Pn800).

Servo Drive display	PCU error code	Alarm type	Error detection function	Detected error or cause of error
84	4054		Transmission cycle error	While actuating MECHATROLINK-II communications, syn- chronization frames (SYNC) were not received according to the transmission cycle.
86	4056		Watchdog data error	Synchronization data exchanged between the master and slave nodes during each MECHATROLINK-II communica- tions cycle resulted in an error.
87	4057	Х	Forced alarm input error	The Forced alarm input signal was input.
90	405A		Transmission cycle setting error	The transmission cycle setting error when the MECHA- TROLINK-II CONNECT command is received.
91	405B		SYNC command error	A SYNC-related command was issued while MECHA- TROLINK-II was in asynchronous communications mode.
92	405C		Encoder data restoration error	The initialization process of the internal position information was not performed properly.
93	405D	PR	Parameter setting error	Parameter setting exceeded the allowable range.
95	405F	PR X	Motor non-conformity	The combination of the Servomotor and Servo Drive is not appropriate.
Others	Others	PR	Other errors	The control circuit malfunctioned due to excessive noise.
				An error occurred within the Servo Drive due to the activa- tion of its self-diagnosis function.

List of G-series Warning Displays

Priority	Servo Drive display	PCU error code	Warning detection function	Warning details
	94	4094	Data setting warn-	 Command argument setting is out of the range.
			ing	Parameter write failure.
				 Command settings are wrong, and others.
	95	4095	Command warning	 Command output conditions are not satisfied.
				Received unsupported command.
				 Subcommand output conditions are not satisfied.
High	96	4096	ML-II communica- tions warning	One or more MECHATROLINK-II communications error occurred.
Ă	A.0	40A0	Overload warning	85% of the overload alarm trigger level has been exceeded.
	A.1	40A1	Regeneration over- load warning	85% of the regeneration overload alarm trigger level has been exceeded.
	A.2	40A2	Battery warning	Voltage of absolute encoder battery has dropped below 3.2 V.
	A.3	40A3	Fan lock warning	The internal cooling fan stop status continued for 1 second.
	A.4	40A4	Encoder communi- cations warning	An encoder communications error occurred for 3 consecu- tive times.
	A.5	40A5	Encoder overheat error	The encoder temperature exceeded 95°C.
Low	A.6	40A6	Vibration detection warning	Vibrations were detected.
	A.7	40A7	Limit detection warning	The limit of the capacitor or the fan fell below the specified value.
	A.8	40A8	External scale error warning	The external scale detected a warning.
	A.9	40A9	External scale com- munications warn- ing	An external scale communications error occurred 3 consecutive times.

Note (1) All warnings are retained. After resolving the problem, clear the alarms and the warnings.

(2) When multiple warnings occur, the warning codes are displayed on the front panel in the order of their priority (shown above). The value read from the network is set.

List of G-series Alarm Displays

The Servo Drive has built-in protective functions. When a protective function is activated, the Servo Drive turns OFF the alarm output signal (ALM) and switches to the Servo OFF status.

The alarm code will be displayed on the front panel.

Alarm type	Description
	Protective function that allows the alarm to be reset, and leaves record in the alarm history.
PR	Protective function that does not allow the alarm to be reset, and requires the control power supply to be turned OFF and turned ON again after resolving the problem.
Х	Protective function that does not leave record in the alarm history.

Servo Drive display	PCU error code	Alarm type	Error detection function	Detected error or cause of error
A.11	400B	х	Control power supply und- ervoltage	The DC voltage of the main circuit has dropped below the specified value.
A.12	400C		Overvoltage	The DC voltage of the main circuit is abnormally high.
A.13	400D	х	Main power supply under- voltage	The DC voltage of the main circuit is low.
A.14	400E	PR	Overcurrent	Overcurrent flowed to the IGBT. Servomotor power line ground fault or short circuit.
A.15	400F	PR	Servo Drive overheat	The temperature of the Servo Drive radiator exceeded the specified value.
A.16	4010		Overload	Operation was performed with torque significantly exceeding the rating for several seconds to several tens of seconds.
A.18	4012	PR	Regeneration overload	The regenerative energy exceeded the processing capacity of the regeneration resistor.
A.21	4015	PR	Encoder communications error	Communications between the encoder and the Servo Drive failed for a specified number of times, thereby activating the error detection function.
A.23	4017	PR	Encoder communications data error	Communications error occurred for the data from the encoder.
A.24	4018		Deviation counter overflow	The number of position deviation pulses exceeded the Deviation Counter Overflow Level (Pn209).
A.26	401A		Overspeed	The rotation speed of the Servomotor exceeded the setting of the Overspeed Detection Level Setting (Pn073).
A.27	401B	PR	Command error	The operation command resulted in an error.
A.29	401D		Internal deviation counter overflow	The value of the internal deviation counter (internal control unit) exceeded 2 ²⁷ (134217728).
A.34	4022		Overrun limit error	The Servomotor exceeded the allowable operating range set in the Overrun Limit Setting (Pn026) with respect to the posi- tion command input.
A.36	4024	PR X	Parameter error	Data in the parameter save area was corrupted when the data was read from the EEPROM at power-ON.
A.37	4025	PR X	Parameter corruption	The EEPROM write verification data was corrupted when the data was read from the EEPROM at power-ON.
A.38	4026	Х	Drive prohibit input error	Forward and Reverse Drive Prohibit Inputs (NOT and POT) both became OPEN.
A.40	4028	PR	Absolute encoder system down error	The voltage supplied to the absolute encoder dropped below the specified value.
A.41	4029	PR	Absolute encoder counter overflow error	The multi-turn counter of the absolute encoder exceeded the specified value.

Servo Drive display	PCU error	Alarm type	Error detection function	Detected error or cause of error
A.42	402A	PR	Absolute encoder over- speed error	The Servomotor rotation speed exceeded the specified value when power to the absolute encoder is supplied by the battery only.
A.44	402C	PR	Absolute encoder one-turn counter error	A one-turn counter error was detected.
A.45	402D	PR	Absolute encoder multi-turn counter error	A multi-turn counter error or phase-AB signal error was detected.
A.47	402F		Absolute encoder status error	The rotation of the absolute encoder is higher than the spec- ified value.
A.48	4030	PR	Encoder phase Z error	A phase-Z pulse was not detected regularly.
A.49	4031	PR	Encoder PS signal error	A logic error was detected in the PS signal.
A.82	4052	PR	Node address setting error	The rotary switch for setting the node address of the Servo Drive was set out of range.
A.83	4053		Communications error	Data received during each MECHATROLINK-II communica- tions cycle repeatedly failed, exceeding the number of times set in the Communications Control (Pn005).
A.84	4054		Transmission cycle error	While actuating MECHATROLINK-II communications, syn- chronization frames (SYNC) were not received according to the transmission cycle.
A.86	4056		Watchdog data error	Synchronization data exchanged between the master and slave nodes during each MECHATROLINK-II communications cycle resulted in an error.
A.87	4057	Х	Emergency stop input error	The emergency stop input became OPEN.
A.90	405A		Transmission cycle setting error	The transmission cycle setting error when the MECHA- TROLINK-II CONNECT command is received.
A.91	405B		SYNC command error	A SYNC-related command was issued while MECHA- TROLINK-II was in asynchronous communications mode.
A.93	405D	PR	Parameter setting error	Parameter setting exceeded the allowable range.
A.95	405F	PR X	Servomotor non-conformity	The combination of the Servomotor and Servo Drive is not appropriate.
Others	Others	PR	Other errors	The control circuit malfunctioned due to excessive noise.
				An error occurred within the Servo Drive due to the activa- tion of its self-diagnosis function.

List of G-series Warning Displays

Priority	Servo Drive display	PCU error code	Warning detection function	Warning details
	A.94	4094	Data setting warn-	 Command argument setting is out of the range.
High			ing	Parameter write failure.
				 Command settings are wrong, and others.
	A.95	4095	Command warning	 Command output conditions are not satisfied.
				 Received unsupported command.
				 Subcommand output conditions are not satisfied.
	A.96	4096	ML-II communica- tions warning	One or more MECHATROLINK-II communications error occurred.
	A.90	4090	Overload warning	85% of the overload alarm trigger level has been exceeded.
	A.91	4091	Regeneration over- load warning	85% of the regeneration overload alarm trigger level has been exceeded.
↓ Low	A.92	4092	Battery warning	Voltage of absolute encoder battery has dropped below 3.2 V.
	A.93	4093	Fan lock warning	The built-in cooling fan stopped, or rotated abnormally.

Note (1) All warnings are retained. After resolving the problem, clear the alarms and the warnings.

(2) When multiple warnings occur, the warning codes are displayed on the front panel in the order of their priority (shown above). The value read from the network is set.

List of W-series and SMARTSTEP Junior Alarm Display

Servo Drive display	PCU error code	Error detection function	Detected error or cause of error
A.02	4002	Parameter corrupted	Parameter checksum read from EEPROM does not match.
A.03	4003	Main circuit detection error	Error in detection data for power supply circuit
A.04	4004	Parameter setting error	The parameter setting is incorrect.
A.05	4005	Servomotor mismatch	The Servomotor and Servo Drive combination is incorrect.
A0b□ (See note 3.)	400B	Servo ON command dis- abled	The PCU sent a SERVO LOCK command when the Servo ON function was being used from personal computer monitoring software
A.10□	4010	Overcurrent	An overcurrent has occurred or the radiation shield has over- heated (1.5- to 3-kW models only).
A.28 (See note 4.)	(See note 4.)	Emergency stop	The emergency stop signal was input while the motor was oper- ating.
A.30□ (See note 5.)	4030	Regeneration error	The regenerative circuit is damaged due to large regenerative energy.
A.32□ (See note 5.)	4032	Regeneration overload	The regenerative energy has exceeded the regeneration resistor capacity.
A.33 (See note 5.)	4033	Main-circuit power sup- ply setting error	The setting of Pn001.2 (AC/DC input selection) and the AC/DC wiring method used for the main circuit power supply are not the same.
A.40	4040	Overvoltage	The main circuit DC voltage has exceeded the specified values.
A.41	4041	Undervoltage	The main circuit DC voltage is under the specified values.
A.51□	4051	Overspeed	The Servomotor rotation speed has exceeded the maximum rotation speed.
A52 (See note 3.)	4052	Fluctuation or autotuning alarm	An abnormal fluctuation was detected in the Servomotor rotation speed, or the inertia ratio calculation during autotuning was abnormal.
A.71	4071	Overload	Operating with output torque exceeding 245% of the rated torque for a W-series Servo Drive or exceeding 200% of the rated torque for a SMARTSTEP Junior Servo Drive.
A.72	4072	Overload	Operation continuing with output torque at 120% to 245% of the rated torque for a W-series Servo Drive or at 115% to 200% of the rated torque for a SMARTSTEP Junior Servo Drive.
A.73	4073	Dynamic brake overload	The rotary energy has exceeded the dynamic brake resistor capacity during dynamic brake operation.
A.74□ (See note 5.)	4074	Inrush resistance over- load	The inrush current when power was turned ON exceeded the inrush resistor capacity.
A.7A	407A	Overheat	Overheating in the radiation shield was detected or the cooling fan in the Servo Drive has stopped.
A.81□ (See note 5.)	4081	Backup error (Absolute encoders only)	The encoder's backup power supply has fallen.
A.82□ (See note 5.)	4082	Checksum error (Absolute encoders only)	An encoder memory checksum error has occurred.
A.83□ (See note 5.)	4083	Battery error (Absolute encoders only)	The encoder's battery voltage has fallen (to 2.7 V or lower).
A.84 (See note 5.)	4084	Absolute error (Absolute encoders only)	An internal encoder data error has occurred.
A.85 (See note 5.)	4085	Overspeed error (Absolute encoders only)	The Servomotor is rotating at 200 r/min. or more when the encoder power supply is turned ON.
A.86□ (See note 5.)	4086	Encoder overheating (Absolute encoders only)	Overheating in the encoder was detected.
A.b1 (See note 2.)	40B1	Speed command input reading error	The A/D completion signal from the A/D converter is not being output within the fixed interval.

List of Error Codes

Servo Drive display	PCU error code	Error detection function	Detected error or cause of error
A.b2 (See note 2.)	40B2	Torque command input reading error	The A/D completion signal from the A/D converter is not being output within the fixed interval.
A.b3□ (See note 3.)	40B3	Current detection error	The Servo Drive's current detector is faulty.
A.b6	40B6	LSI for communications corrupted	The LSI used for MECHATROLINK communications is corrupted.
A.bF	40BF	System error	A system error in the control circuit was detected.
A.C1□	40C1	Runaway detected	The Servomotor rotated in the opposite direction to the com- mand.
A.C2 (See note 4.)	40C2	Phase detection error	A Servomotor magnetic pole signal was incorrectly detected.
A.C5 (See note 4.)	40C5	Magnetic pole detection error	A Servomotor magnetic pole signal was incorrectly detected.
A.C8□ (See note 5.)	40C8	Multi-turn data error (Absolute encoders only)	The absolute encoder setup is incorrect.
A.C9	40C9	Encoder communica- tions error	Communications between the encoder and Servo Drive are not possible.
A.CA⊡ (See note 5.)	40CA	Encoder parameter error	The parameter settings in the encoder are corrupted.
A.Cb⊡ (See note 5.)	40CB	Encoder data error	Data from the encoder is corrupted.
A.CC□ (See note 5.)	40CC	Multi-turn limit discrep- ancy (Absolute encoders only)	The absolute encoder multi-turn limit for the encoder and Servo Drive do not match.
A.d0□ (See note 5.)	40D0	Deviation counter over- flow	The number of pulses in the deviation counter has exceeded the deviation counter overflow level set in Pn505 (R88D-WT) or Pn520 (R88D-WN-ML2).
A.d1□ (See note 5.)	40D1	Motor-load deviation over	The deviation between the fully-closed encoder and semi-closed encoder has reached or exceeded the command unit set in Pn51A.
A.E0	See note 1.	No option or Servo Drive error	The MECHATROLINK-II I/F Unit is not installed, or the Servo Drive has malfunctioned.
A.E1 (See note 2.)	See note 1.	Option timeout	There is no response from the MECHATROLINK-II I/F Unit.
A.E2 (See note 2.)	See note 1.	Option WDC error	An error has occurred in the MECHATROLINK-II I/F Unit. (MECHATROLINK-II I/F Unit's watchdog timer count)
A.E4 (See notes 3 and 4.)	40E4	Transfer cycle setting error	MECHATROLINK-II transfer cycle setting is incorrect.
A.E5	40E5	Synchronization error	MECHATROLINK-II synchronization error
A.E6□	40E6	Communications error	MECHATROLINK-II communications error (Continuous communications errors have occurred.)
A.EA	40EA	Servo Drive malfunction	The Servo Drive has malfunctioned.
A.EB (See note 2.)	40EB	Servo Drive initial access error	The Servo Drive initial processing cannot be executed from the MECHATROLINK-II I/F Unit.
A.EC (See note 2.)	40EC	Servo Drive error	An error has occurred in the Servo Drive. (Servo Drive's watchdog timer count)
A.Ed	40ED	Command execution incomplete	MECHATROLINK communications command aborted during execution.
A.F1□ (See note 5.)	40F1	Missing phase detected	Main circuit power supply phase is missing, or the wire is burnt out.
A.F5 (See note 2.)	40F5	Motor current error	The current to the Servomotor is too small for the torque com- mand from the Servo Drive.
A.F6 (See note 2.)	40F6	Motor conduction error	The Servo is ON, but the Servomotor is not conducting current regardless of the Servo Drive settings and external input.

- **Note** (1) Errors that occur in the MECHATROLINK-II I/F Unit cannot be detected by the PCU because the communications connection is not established. The PCU is not able to detect the corresponding axis during execution of CONNECT, so an MLK Initialization Error (Unit error code: 0020 hex) will occur.
 - (2) This alarm occurs when using an R88D-WT□ W-series Servo Drive together with the FNY-NS115 MECHATROLINK-II I/F Unit.
 - (3) This alarm occurs when using an R88D-WND-ML2 W-series Servo Drive.
 - (4) This alarm may occur when a SMARTSTEP Junior Servo Drive (R7D-ZN□-ML2) is being used. The A.28 (emergency stop) alarm is treated as an emergency stop signal input error (axis error code: 3004 (hex) in the Position Control Unit and the Servo will be unlocked for the axis.
 - (5) This alarm is not detected for SMARTSTEP Junior Servo Drives (R7D-ZND-ML2).

List of W-series and SMARTSTEP Junior Warning Displays

Servo Drive display	PCU error code	Warning detection function	Warning details
A.90 (See note 1.)	4090	Deviation counter over- flow	The number of pulses in the deviation counter has exceeded the deviation counter overflow level set in Pn505 (when using R88D-WT) or Pn520 (when using R88D-WN-ML2) multiplied by the rate (%) set in Pn51E.
A.91	4091	Overload or fluctuation	This warning occurs before the Overload Alarm (A.71, A.72) occurs. If operation is continued in this state, an alarm may occur. This warning also occurs when an abnormal fluctuation is detected in the Servomotor rotation speed.
A.92 (See note 1.)	4092	Regeneration overload	This warning occurs before the Regeneration Overload Alarm (A.32) occurs. If operation is continued in this state, an alarm may occur.
A.93 (See note 1.)	4093	Battery warning (Absolute encoders only)	This warning occurs before the Battery Error (A.83) occurs. If the power is turned OFF, an alarm may occur the next time the power is turned ON. (Replace the battery while the control circuit power supply is ON.)
A.94	See note 2.	Parameter setting warn- ing	A value outside the setting range has been set for the MECHA- TROLINK slave station device.
A.95	See note 2.	MECHATROLINK-II com- mand warning	An illegal communications command or unsupported communi- cations command has been sent to the MECHATROLINK slave station device.
A.96	4096	Communications warning	A single MECHATROLINK-II communications error has occurred. (See note 3.)

Note (1) This warning is not detected for SMARTSTEP Junior Servo Drives (R7D-ZND-ML2).

- (2) If a Parameter Setting Warning or MECHATROLINK-II Command Warning occurs in the Servo Drive, a data setting error (axis error code: 3099 hex) will occur at the PCU, and the active axis in which the error occurred will decelerate to a stop.
- (3) If a MECHATROLINK-II communications error occurs once independently, a communications warning occurs, and the PCU will perform a communications retry. If the communications warning occurs continually, a communications error will occur.
Appendix E Changing to CS1W/CJ1W-NC271/471/F71 from CS1W/CJ1W-NC113/133/213/233/413/433

The settings and methods for using CS1W/CJ1W-NC271/471/F71 Position Control Units include those that are different from the pulse-output models (CS1W/CJ1W-NC113/133/213/233/413/433).

The following information is important when replacing equipment that uses pulse-output PCUs and pulse-input Servo Drives or Stepping Motor Drives with the CS1W/CJ1W-NC271/471/F71 PCU and MECHATROLINK-II-compatible Servo Drive.

Installation

- The FNY-NS115 MECHATROLINK-II I/F Unit must be installed when using an R88D-WT OMRON Wseries Servo Drive (without built-in MECHATROLINK-II communications). The FNY-NS115 is installed on the right side of the Servo Drive, so make sure that sufficient space is provided around the Servo Drive.
- R88D-WT W-series Servo Drives must be Ver.39 or later. The Servo Drive must be changed if an earlier version is being used. The Servomotor, power cables, and encoder cables, however, can be used as is without changing.

Wiring

- MECHATROLINK-II Cables are used for the connection between the PCU and Servo Drive (MECHA-TROLINK-II I/F Unit). The cables for pulse-output PCUs and Servo Relay Units cannot be used.
- The external sensor signals, such as limit inputs and origin proximity inputs, are connected to the Servo Drive's control I/O connector (CN1). The Servo Drive's CN1 connector must be wired.
- Emergency stop input signals are not provided as external input signals for the PCU or Servo Drive. To ensure safety in the system, provide safety measures in external circuits (i.e., not in the Programmable Controller or Position Control Unit), such as using a breaker for the Servo Drive's main circuit power supply.
- To perform an origin search, the origin proximity input signal must be used. Make sure that the dog width of the limit input signal and origin proximity input signal remains at the length of the MECHATROLINK-II communications cycle or longer.

System Configuration

Pulse-output PCUs are classified as CS/CJ-series Special I/O Units, whereas CS1W/CJ1W-NC271/471/F71 PCUs are classified as CS/CJ-series CPU Bus Units. Make sure that the PCU's unit number is not already used by another CPU Bus Unit and that the area allocated to the PCU in the PLC is not occupied by another Unit.

PCU Parameters and Operating Data

The functions and classification of the PCU's parameters and operating data are shown in the following table.

CS1W/CJ1V	V-NC113/133/213/233/413/433	CS1W/CJ1W-NC271/471/F71			
Item	Setting	Item	Setting		
Common Parame- ter Area	Used to allocate the Operating Data Area and designates the Axis Param- eters. The area is set in the PLC's DM Area words allocated to Special I/O Units.	Common Parame- ter Area	Used to allocate the Axis Operating Memory Areas and MECHA- TROLINK communications settings. The settings are written to the PCU's internal memory using the data trans- fer functions.		
Axis Parameter Area	Used to allocate each Axis's operat- ing mode, maximum speed, and soft- ware limits. The area is set in the PLC's DM Area words allocated to Special I/O Units or in the PCU's internal memory.	Axis Parameter Areas Axis Operating Memory Area Servo Parameter Area	The Axis Parameter Areas are used to select operating modes (origin search) and input signals. The settings are written to the PCU's internal memory using the data trans- fer functions. Some Axis Parameters are set as command values in the Axis Operat- ing Memory Areas and are used as operating data. Other existing PCU Axis Parameters are set to the corresponding Servo Parameters.		
Operating Memory Area	Used to input and output bit data, such as operating command outputs and status (flags) inputs. The area is allocated in the Special I/O Unit Area and is determined according to the unit number set for the PCU.	Axis Operating Memory Area Servo Parameter Area	The I/O information (bit and word data) is handled in the Axis Operating Output Memory Areas and Axis Oper- ating Input Memory Areas as output data and input data, respectively. The Axis Operating Output Memory Areas and Axis Operating Input Mem- ory Areas are allocated in the PLC's memory according to the settings in the Common Parameters. Acceleration/deceleration data is set in the Servo Parameters.		
Operating Memory Data	Used to input and output parameter data, such as position command value outputs and present position inputs. The area is allocated in the PLC's DM Area according to the setting in the Common Parameter Area.				
Memory operation data	Used to set the position/speed data for memory operations. The settings are written to the PCU's internal memory using the data trans- fer functions.		Memory operation functions are not provided.		
Zone data	Used to set the zones. The settings are written to the PCU's internal memory using the data trans- fer functions.		A zone function is not provided.		
		Common Operating Memory Area	Performs operations common to the PCU such as MECHATROLINK com- munications control. The allocated area is contained in the PLC's CPU Bus Unit Memory Area according to the PCU's unit number.		

A comparison of individual parameters and operating data, and operating precautions are provided in the following pages.

Common Parameter Area

CS	1W/CJ1W-NC113/133/213/233/413/433	CS1W/CJ1W-NC271/471/F71		
	Setting		Setting	
m	Operating Data Area designation	Common Parameter Area	Used to allocate Axis Operating Memory Areas and	
m+1	Beginning word of the Operating Data Area		to perform MECHATROLINK communications set-	
m+2	Axis Parameters designation		The settings are written to the PCU's internal mem- ory using the data transfer functions.	

Axis Parameter Area

To use data in the CS1W/CJ1W-NC271/471/F71 that was set as Axis Parameters in the earlier PCUs, the data must be converted either to corresponding settings in the Axis Parameter Area and Servo Parameter Area, or assigned in the Axis Operating Output Memory Area as command values used as operating data.

The setting location in the CS1W/CJ1W-NC271/471/F71 corresponding to the Axis Parameters in earlier PCUs and the differences between the different parameters is shown in the following table.

The Axis Parameter addresses shown for CS1W/CJ1W-NC113/133/213/233/413/433 are those for the X axis. The setting locations in the CS1W/CJ1W-NC271/471/F71 indicated as word d in the Axis Parameters and word a in the Axis Operating Output Memory Area can be calculated using the following equations.

 $d = 1860 \text{ hex} + (Axis \text{ No.} -1) \times 14 \text{ hex}$

a = Beginning word of Axis Operating Output Memory Areas specified in Common Parameters + (Axis No. -1) \times 25

G-series Servo Drives

CS1	CS1W/CJ1W-NC113/133/213/233/413/433			CS1W/CJ1W-NC271/471/F71			
Word	Name	Bit	Setting	Setting lo	cation	Difference	
0004	I/O set- tings	00	Output pulse selec- tion			Output pulse selection is not pro- vided, so this parameter setting is not available.	
		01 to 03	Not used.				
		04	Limit input signal type	Servo Parame- ter Area	Pn404	Sets input allocations for the for- ward/reverse drive prohibited signals.	
						The input signal logic cannot be set.	
		05	Origin proximity input signal type	Servo Parame- ter Area	Pn042	Sets the input signal logic using the input allocations for the origin return deceleration LS signals.	
		06	Origin input signal type			The input signal logic cannot be set.	
		07	Emergency stop input signal			The input signal logic cannot be set.	
	08 Origin undefined designation				This parameter setting is not avail- able. The Servo Drive's feedback position is obtained. Therefore, the origin is not lost even when operation stops due to an external sensor.		
		09 to 15	Not used.				
0005 Opera- tion mode selection		00 to 03	Operation mode	Axis Parameter Area word d+1	Bits 00 to 03	Not used. Operation mode selection is not pro- vided, so this parameter setting is not available.	
		04 to 07	Origin search opera- tion		Bits 04 to 07	This parameter setting is the same.	
		08 to 11	Origin detection method		Bits 08 to 11	Not used. Origin searches always use the origin proximity input signal. The origin detection method is fixed as the method that receives the origin input signal after the origin proximity input signal turns from ON to OFF. (This method corresponds with the setting when the origin detection method in the earlier PCUs is set to 0.)	
		12 to 15	Origin search direc- tion		Bits 12 to 15	This parameter setting is the same. Servo Parameter Pn10F (Origin Return Mode Settings), however, must be set at the same time to the same setting as for this parameter. (For details, refer to <i>8-2-3 Data Set-</i> <i>tings Required for Origin Search.</i>)	
0006	Maximum	speed (r	ightmost word)			A maximum speed setting is not pro-	
0007	Maximum speed (rightmost word) Maximum speed (leftmost word)					vided. The maximum speed is clamped at approximately 110% when a speed command is sent for a value that is equal or higher than the Servo Drive's (Servomotor's) maximum speed.	

CS1	W/CJ1W-NC1	113/133	/213/233/413/433	CS1W/CJ1W-NC271/471/F71			
Word	Name	Bit	Setting	Setting lo	cation	Difference	
0008	Initial speed	(rightm	ost word)			An initial speed setting is not pro-	
0009	Initial speed (leftmost word)				vided. The acceleration/deceleration set- tings, however, enable two-step acceleration/deceleration curves and exponential acceleration/decelera- tion curves with bias. (For details, refer to 7-4 Acceleration and Deceler- ation Operations.)		
000A	Origin search word)	h high s	speed (rightmost	Axis Operating Output Memory	Word a+4 Word a+5	The origin search high speed is the speed command value for origin searches and is designated with the	
000B	Origin searcl	h high s	peed (leftmost word)	Aleas		operating command in the Axis Oper- ating Output Memory Areas.	
000C	Origin searcl word)	h proxir	nity speed (rightmost	Servo Parame- ter Area	Pn110 Pn111	Sets the Servo Parameters Pn110 (Origin Return Approach Speed 1) and Pn111 (Origin Return Approach	
000D	Origin searcl word)	h proxir	nity speed (lettmost			Speed 2). The setting unit is ×100 command units/s. (For details, refer to 8-2-3 Data Settings Required for Origin Search.)	
000E	Origin compo word)	ensatio	n value (rightmost	Servo Parame- ter Area	Pn204	The amount of compensation posi- tioning after detection of the origin	
000F	Origin compo word)	ensatio	n value (leftmost			Pn204 (Origin Return Final Dis- tance). (For details, refer to 8-2-3 Data Settings Required for Origin Search.)	
0010	Backlash cor	mpensa	ition	Servo Parame- ter Area	Pn100 Pn101	The backlash compensation is set in Servo Parameters Pn101 (Backlash Compensation) and Pn100 (Backlash Compensation Selection). (For details, refer to <i>10-7 Backlash Com- pensation</i> .)	
0011	Backlash cor most word)	mpensa	tion speed (right-			Backlash compensation is executed by adding it to the amount of move- ment for positioning, so these param- eter settings are not supported.	
0012	Backlash cor word)	mpensa	tion speed (leftmost				
0013	Acceleration/deceleration curves			Axis Operating Output Memory Areas	Word a+16, bits 03 to 04	The acceleration/deceleration curves are designated with the operating command in the Axis Operating Out- put Memory Areas as command val- ues for executing positioning operations.	
0014	Origin searcl most word)	h accel	eration time (right-	Servo Parame- ter Area	Pn107	The acceleration used for origin searches is the same as that used in other positioning operations and in	
0015	Origin searcl word)	h accel	eration time (leftmost			other positioning operations and is set according to the combination of settings in Servo Parameter Pn107 (Linear Acceleration Constant). (For details, refer to 7-4 Acceleration and Deceleration Operations.)	
0016	Origin searcl most word)	h decel	eration time (right-	Servo Parame- ter Area	Pn10A	The deceleration used for origin searches is the same as that used in	
0017	Origin searcl word)	h decel	eration time (leftmost			other positioning operations and is set according to the combination of settings in Servo Parameter Pn10A (Linear Deceleration Constant). (For details, refer to 7-4 Acceleration and Deceleration Operations.)	

CS1	W/CJ1W-N	C113/13	3/213/233/413/433	CS1W/CJ1W-NC271/471/F71			
Word	Name	Bit	Setting	Setting lo	cation	Difference	
0018	Positionin	g monitor	time			A positioning monitor is not provided, so this parameter setting is not avail- able. Refer to the programming examples and set a timing monitor in the user program. (Refer to <i>11-2-6 Positioning</i> <i>(Absolute Movement or Relative</i> <i>Movement)</i> .)	
0019	CCW soft	ware limit	(rightmost word)	Servo Parame- ter Area	Pn201	The forward software limit is set in Servo Parameter Pn201 (Forward Software Limit) in command units. (For details, refer to <i>10-8 Software</i> <i>Limits</i> .)	
001A	CCW soft	ware limit	(leftmost word)				
001B	CW softwa	are limit (rightmost word)	Servo Parame-	Pn202	The reverse software limit is set in	
001C	CW software limit (leftmost word)			ter Area		Servo Parameter Pn202 (Reverse Software Limit) in command units. (For details, refer to <i>10-8 Software Limits</i> .)	
001D	Reserved	by the sy	stem.				
001E	Reserved	by the sy	stem.				
001F	Initial puls	e designa	ation			Initial pulse designation is not pro- vided, so this parameter setting is not available.	

W-series Servo Drives

CS1	W/CJ1W-N	C113/13	3/213/233/413/433	CS1W/CJ1W-NC271/471/F71			
Word	Name	Bit	Setting	Setting lo	cation	Difference	
0004	I/O set- tings	00	Output pulse selec- tion			Output pulse selection is not pro- vided, so this parameter setting is not available.	
		01 to 03	Not used.				
		04	Limit input signal type	Servo Parame- ter Area	Pn50A.3 Pn50B.0	Sets the input signal logic using the input allocations for the for-ward/reverse drive prohibited signals.	
		05	Origin proximity input signal type	Servo Parame- ter Area	Pn511.0	Sets the input signal logic using the input allocations for the origin return deceleration LS signals.	
		06	Origin input signal type	Servo Parame- ter Area	Pn511.1 Pn511.2 Pn511.3	Sets the input signal logic according to the input allocations for the exter- nal latch signal when external latch signal is selected as the origin input signal. This setting is not available when the Servomotor phase Z signal is selected as the origin input signal.	
0		07	Emergency stop input signal			An emergency stop input signal is not provided, so this parameter is not available. The stop methods used for limit inputs and software limits, however, can be specified in Servo Parameter Pn001.1. (For details, refer to 7-5 <i>Limit Input Operations.</i>)	
		08	Origin undefined designation			This parameter setting is not avail- able. The Servo Drive's feedback position is obtained. Therefore, the origin is not lost even when operation stops due to an external sensor.	
		09 to 15	Not used.				
0005	Opera- tion mode selection	00 to 03	Operation mode	Axis Parameter Area word d+1	Bits 00 to 03	Not used. Operation mode selection is not pro- vided, so this parameter setting is not available.	
		04 to 07	Origin search opera- tion		Bits 04 to 07	This parameter setting is the same.	
		08 to 11	Origin detection method		Bits 08 to 11	Not used. Origin searches always use the origin proximity input signal. The origin detection method is fixed as the method that receives the origin input signal after the origin proximity input signal turns from ON to OFF. (This method corresponds with the setting when the origin detection method in the earlier PCUs is set to 0.)	
		12 to 15	Origin search direc- tion		Bits 12 to 15	This parameter setting is the same. Servo Parameter Pn816 (zero point return direction), however, must be set at the same time to the same set- ting as for this parameter. (For details, refer to <i>8-2-3 Data Settings</i> <i>Required for Origin Search.</i>)	

CS1	W/CJ1W-NC1	13/13	3/213/233/413/433	CS1W/CJ1W-NC271/471/F71			
Word	Name	Bit	Setting	Setting lo	cation	Difference	
0006 0007	Maximum sp Maximum sp	beed (ri beed (le	ghtmost word) oftmost word)			A maximum speed setting is not pro- vided. The maximum speed is clamped at approximately 110% when a speed command is sent for a value that is equal or higher than the Servo Drive's (Servomotor's) maximum speed.	
0008	Initial speed	(rightm (leftmc	nost word) ost word)			An initial speed setting is not pro- vided. The acceleration/deceleration set- tings, however, enable two-step acceleration/deceleration curves and exponential acceleration/decelera- tion curves with bias. (For details, refer to 7-4 Acceleration and Deceler- ation Operations.)	
000A 000B	Origin search word) Origin search	h high : h high :	speed (rightmost speed (leftmost word)	Axis Operating Output Memory Areas	Word a+4 Word a+5	The origin search high speed is the speed command value for origin searches and is designated with the operating command in the Axis Oper- ating Output Memory Areas.	
000C 000D	Origin search proximity speed (rightmost word) Origin search proximity speed (leftmost word)			Servo Parame- ter Area	Pn817 Pn818	Sets the Servo Parameters Pn817 (zero point return approach speed 1) and Pn818 (zero point return approach speed 2). The setting unit is ×100 command units/s. (For details, refer to <i>8-2-3 Data Settings</i> <i>Required for Origin Search.</i>)	
000E 000F	Origin compensation value (rightmost word) Origin compensation value (leftmost word)			Servo Parame- ter Area	Pn819 Pn81A	The amount of compensation posi- tioning after detection of the origin input signal is set in Servo Parameter Pn819 (final travel distance to return to zero point). (For details, refer to <i>8- 2-3 Data Settings Required for Origin</i> <i>Search.</i>)	
0010	Backlash compensation			Servo Parame- ter Area	Pn81B Pn81D.0 Pn207.2 Pn214	The backlash compensation is set in Servo Parameters Pn81B (backlash compensation amount) and Pn81D.0 (backlash compensation direction) when using R88D-WT Servo Drives and in Pn207.2 (backlash compensa- tion selection) and Pn214 (backlash compensation amount) when using R88D-WN -ML2 Servo Drives. (For details, refer to 10-7 Backlash Com- pensation.)	
0011 0012	Backlash cor most word) Backlash cor word)	mpensa mpensa	ation speed (right- ation speed (leftmost			Backlash compensation is executed by adding it to the amount of move- ment for positioning, so these param- eter settings are not supported.	
0013	Acceleration/	/decele	eration curves	Axis Operating Output Memory Areas	Word a+16, bits 03 to 04	The acceleration/deceleration curves are designated with the operating command in the Axis Operating Out- put Memory Areas as command val- ues for executing positioning operations.	

CS1	W/CJ1W-N	C113/13	3/213/233/413/433	CS1W/CJ1W-NC271/471/F71			
Word	Name	Bit	Setting	Setting lo	cation	Difference	
0014	Origin sea most word Origin sea word)	rch acce) rch acce	eration time (right- eration time (leftmost	Servo Parame- ter Area	Pn80A Pn80B Pn80C	The acceleration used for origin searches is the same as that used in other positioning operations and is set according to the combination of settings in Servo Parameters Pn80A, Pn80B, and Pn80C. (For details, refer to 7-4 Acceleration and Deceleration Operations.)	
0016	Origin sea most word	rch dece)	leration time (right-	Servo Parame- ter Area	Pn80D Pn80E	The deceleration used for origin searches is the same as that used in	
0017	Origin sea word)	rch dece	leration time (leftmost		Pn80F	other positioning operations and is set according to the combination of settings in Servo Parameters Pn80D, Pn80E, and Pn80F. (For details, refer to 7-4 Acceleration and Deceleration Operations.)	
0018	Positioning	g monitor	time			A positioning monitor is not provided, so this parameter setting is not avail- able. Refer to the programming examples and set a timing monitor in the user program. (Refer to <i>11-2-6 Positioning</i> <i>(Absolute Movement or Relative</i> <i>Movement)</i> .)	
0019	CCW softw	vare limit	(rightmost word)	Servo Parame-	Pn804	The forward software limit is set in	
001A	CCW softv	vare limit	(leftmost word)	ter Area	Pn805	Servo Parameters Pn804 and Pn805 in command units. (For details, refer to <i>10-8 Software Limits.</i>)	
001B	CW softwa	are limit (rightmost word)	Servo Parame-	Pn806	The reverse software limit is set in	
001C	CW softwa	are limit (leftmost word)	ter Area	Pn807	Servo Parameters Pn806 and Pn807 in command units. (For details, refer to <i>10-8 Software Limits</i> .)	
001D	Reserved	by the sy	stem.				
001E	Reserved	by the sy	stem.				
001F	Initial pulse	e designa	ation			Initial pulse designation is not pro- vided, so this parameter setting is not available.	

Operating Memory Area

In the CS1W/CJ1W-NC271/471/F71 PCU, the Operating Memory Area allocations and functions have changed, as shown in the following table.

A comparison of the functions of these words is provided in the following table. The differences between the functions, such as in axis operations executed by the Operating Memory Area words, are explained later in this appendix.

Operating Memory Area words for CS1W/CJ1W-NC113/133/213/233/413/433 PCUs are shown in the following table as those for the X axis of the CS1W/CJ1W-NC113 PCU, and the beginning word n is calculated using the following equation.

 $n = 2000 + (unit number \times 10)$

The Operating Memory Area words for CS1W/CJ1W-NC271/471/F71 PCUs are calculated for each axis using the following equation.

a = Beginning word of Axis Operating Output Memory Areas specified in Common Parameters + (Axis No. -1) \times 25

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) \times 25

I/O	CS1	W/CJ1W	-NC113/133	3/213/233/413/433	CS1W/CJ1W-NC271/471/F71			
	Word	Bits	Category	Name	Word	Bits	Differences	
Output n (CPU Unit to PCU)	n	00	Memory operation	Sequence Number Enable	а	00	Used as linear interpolation command bits.	
	01	com-	START		01	Memory operation functions are not		
		02	manus	INDEPENDENT START		02	mand bits are not allocated.	
		03	Direct operation com- mands	ABSOLUTE MOVE- MENT		03	The function of this bit is the same.	
		04		RELATIVE MOVE- MENT		04	The function of this bit is the same.	
		05		INTERRUPT FEED- ING		05	The method for starting interrupt feed- ing has been changed. This bit is an Interrupt Feeding Desig- nation Bit for ABSOLUTE/RELATIVE MOVEMENT, and is used together with ABSOLUTE/RELATIVE MOVEMENT bits.	
		06	Origin	ORIGIN SEARCH		06	The function of this bit is the same.	
		07	position-	ORIGIN RETURN		07	The function of this bit is the same.	
		08	mands	PRESENT POSI- TION PRESET		08	The function of this bit is the same. When the DECELERATION STOP Bit is ON, however, PRESENT POSITION PRESET cannot be received.	

I/O	CS1	W/CJ1W	/-NC113/133	3/213/233/413/433	CS1W/CJ1W-NC271/471/F71			
	Word	Bits	Category	Name	Word	Bits	Differences	
Output	n	09	Com-	JOG	а	09	The function of this bit is the same.	
(CPU Unit to PCU)		10	mands for special functions	Direction designation		10	The function of this bit is the same for JOG direction. The rotation direction cannot be designated for interrupt feeding.	
		11		TEACH		11	Not used. A teaching function is not provided, so this bit is not allocated.	
		12		ERROR RESET/RELEASE PROHIBIT		12	The function of this bit is the same for resetting errors that occur in individual axes. The function to release prohibit on pulse outputs is not provided. The function to reset Unit common errors is allocated in the Common Operating Memory Area.	
		13		Deviation counter reset output/origin- adjustment com- mand output	-	13	Same bit function as the deviation counter reset for PCUs with unit ver- sions 1.3 or later. This function, how- ever, does not turn on the deviation counter reset output but rather starts the DEVIATION COUNTER RESET command. (For PCUs with unit versions 1.2 or ear- lier, counter reset output and origin- adjustment command output functions are not provided, so this bit is not allo-	
		14	1	Override Enable		14	The function of this bit is the same	
		15		(DECELERATION) STOP	•	15	The function of this bit is the same.	
	n+1	00		Not used.	a+1	00	Used as the SERVO LOCK Bit.	
		01	-	Not used.	1	01	Used as the SERVO UNLOCK Bit.	
		02	-	Not used.	1	02	Used as the SPEED CONTROL Bit.	
		03	-	Not used.	1	03	Used as the TORQUE CONTROL Bit.	
		04 to 07		Not used.		04 to 07	Not used.	
		08		FORCED INTER- RUPT		08	Not used. A forced interrupt start function is not provided, so this bit is not allocated.	
		09 to 10	1	Not used.		09 to 10	Not used.	
		11		Not used.	1	11	Used as the DEVICE SETUP Bit.	
		12	Data transfer com- mands	WRITE DATA		12	Used as the WRITE SERVO PARAME- TER Bit. The PCU's WRITE DATA Bit is allo- cated in the Common Operating Mem- ory Area.	
		13		READ DATA		13	Used as the READ SERVO PARAME- TER Bit. The PCU's READ DATA Bit is allocated in the Common Operating Memory Area.	
		14		SAVE DATA		14	Used as the SAVE SERVO PARAME- TER Bit. The PCU's SAVE DATA Bit is allocated in the Common Operating Memory Area.	

I/O	CS1	W/CJ1W	/-NC113/13	3/213/233/413/433	CS1W/CJ1W-NC271/471/F71			
	Word	Bits	Category	Name	Word	Bits	Differences	
Output (CPU Unit to PCU)	n+1	15	Data transfer com- mands	Not used.	a+1	15	Used as the EMERGENCY STOP Bit. The Servomotor stops after moving for the amount of pulses remaining the Servo Drive's deviation counter when this bit turns ON.	
Input	n+2	00	PCU sta-	Not used.	b	00	Used as the Receiving Command Flag.	
(PCU to CPU		01 to 03	tus	Not used.		01 to 03	Not used.	
Unity		04		Waiting for Memory Operation Flag		04	Not used. Memory operation functions are not provided, so this bit is not allocated.	
		05		Positioning Com- pleted Flag		05	The function of this bit is the same. Name: PCU Positioning Completed Flag	
		06		No Origin Flag		06	The function of this bit is the same. The conditions whereby the origin is lost after being established are different from earlier PCUs.	
		07		Origin Stop Flag		07	This flag functions in the same way, showing whether the present position is within the origin positioning range (zero point width). This flag turns ON when the origin is passed, which is different from earlier PCUs.	
		08		Zone 0 Monitor Flag	1	08	Not used.	
		09		Zone 1 Monitor Flag	1	09	A zone function is not provided, so	
		10		Zone 2 Monitor Flag]	10		
		11		Teaching Completed Flag		11	A teaching function is not provided. This bit is used as a Warning Flag.	
		12		Error Flag		12	The function of this bit is the same for errors that occur in individual axes. The error flags for Unit common errors are allocated as Unit Error Flags in the Common Operating Memory Area.	
		13		Busy Flag		13	The function of this bit is the same for indicating busy status of functions for individual axes. The busy status for Unit common func- tions (including Unit initial processing) is allocated as Unit Busy Flags in the Common Operating Memory Area.	
		14		Data Transferring Flag		14	Used as the Servo Parameter Transfer- ring Flag. The PCU's Data Transferring Flag is allocated as the Data Transferring Flag in the Common Operating Memory Area.	
		15		Deceleration Stop Execution Flag		15	The function of this bit is the same. The EMERGENCY STOP Bit is turned ON by executing EMERGENCY STOP, however, and not by an external input (the CS1W/CJ1W-NC271/471/F71 does not have emergency stop inputs).	

I/O	CS1	CS1W/CJ1W-NC113/133/213/233/413/433		CS1W/CJ1W-NC271/471/F71				
	Word	Bits	Category	Name	Word	Bits	Differences	
Input (PCU to	n+3	00	External I/O status	Not used.	b+3	00	Used as the forward rotation limit input signal status.	
CPU Unit)		01		Not used.		01	Used as the reverse rotation limit input signal status.	
		02		Not used.		02	Used as the origin proximity input sig- nal status.	
	03		Not used.		03	Used as the encoder phase A input sig- nal status.		
		04		Not used.		04	Used as the encoder phase B input sig- nal status.	
		05		Not used.		05	Used as the encoder phase Z input signal status.	
		06		Not used.		06	Used as the external latch signal 1 input signal status.	
		07		Not used.		07	Used as the external latch signal 2 input signal status.	
		08		CW limit input signal		08	Used as the external latch signal 3 input signal status. The forward rotation limit input signal status is allocated to bit 00 of this word (b+3).	
		09		CCW limit input sig- nal	-	09	Used as the brake output signal status. The reverse rotation limit input signal status is allocated to bit 01 of this word (b+3).	
		10		Origin proximity input signal		10	This bit is reserved by the system. The origin proximity input signal status is allocated to bit 02 of this word (b+3).	
		11		Origin input signal			11	This bit is reserved by the system. The origin input status is provided either in bit 05 of this word (b+3) when phase Z input signal is set or in bits 06 to 08 of this word when the external latch signal is set, according to the selected signal.
		12		Interrupt input signal		12	This bit is reserved by the system. The interrupt input signal status is pro- vided in bits 06 to 08 of this word (b+3) according to the external latch signal.	
		13		Emergency stop input signal		13	An emergency stop input signal is not provided. This bit is reserved by the system.	
		14		Positioning com- pleted input signal		14	This bit is reserved by the system. The Servo Drive's positioning com- pleted signal status (indicating that the deviation counter is within the <i>Position- ing Completion Range 1</i>) is allocated as the Position Completed (PSET) Flag in bit 07 of word b+2 (Servo Status Flags) in the Axis Operating Input Memory Area.	
		15		Deviation counter reset output/origin- adjustment com- mand output		15	An deviation counter reset output/ori- gin-adjustment command output is not provided as an external output. This bit is reserved by the system.	

I/O	CS1W/CJ1W-NC113/133/213/233/413/433			CS1W/CJ1W-NC271/471/F71			
	Word	Bits	Category	Name	Word	Bits	Differences
Input (PCU to CPU Unit)	n+4		Error code	Error code	b+4		The function of this word is the same for errors that occur in individual axes. The error codes for Unit common errors are allocated as Unit error codes in the Common Operating Memory Area.

Operating Data Area

The operating data designated in the data memory of earlier PCUs is allocated in the CS1W/CJ1W-NC271/471/F71 in either the Common Operating Memory Area or Axis Operating Memory Area, as shown in the following table.

A comparison of the functions for these words is provided in the following table. The differences between these functions, such as in axis operations executed by the Operating Memory Area bits, are explained later in this appendix.

Operating Data Area words for CS1W/CJ1W-NC113/133/213/233/413/433 PCUs are shown in the following table for the X axis of the CS1W/CJ1W-NC113 PCU. The beginning word "I" is allocated in either the Special I/O Unit Area or a user-specified DM/EM Area using settings in the Common Parameters.

The Operating Memory Area words for CS1W/CJ1W-NC271/471/F71 PCUs are allocated either to the Common Operating Memory Area or Axis Operating Memory Area, which are calculated using the following equations.

 $n = CIO 1500 + (unit number \times 25)$

a = Beginning word of Axis Operating Output Memory Areas specified in Common Parameters + (Axis No. -1) \times 25

I/O	CS1W/CJ1W-NC113/133/213/233/413/433			CS1W/CJ1W-NC271/471/F71		
	Word	Category	Name	Word	Differences	
Output (CPU Unit to PCU)	Ι	Operat- ing data for data transfer	Number of write words	n+6	The function is the same as data for WRITE DATA	
	l+1		Write source area	n+7	used to transfer PCU parameters.	
	I+2		Write source word	n+8	PARAMETER function is different.	
	l+3		Write destination address	n+9		
	l+4		Number of read words	n+10	The function is the same as data for READ DATA	
	l+5		Read source address	n+11	used to transfer PCU parameters.	
	l+6		Read destination area	n+12	PARAMETER function is different.	
	l+7		Read destination word	n+13		
	l+8	Operat- ing data for direct operation	Position (rightmost word)	a+2	The function of these words is the same.	
	l+9		Position (leftmost word)	a+3	The data unit, however, is the command unit deter- mined according to the setting for the electronic gear.	
	l+10		Speed (rightmost word)	a+4	The function of these words is the same.	
	l+11		Speed (leftmost word)	a+5	The data unit, however, is the command unit/s deter- mined according to the setting for the electronic gear.	
	l+12		Acceleration time (right- most word)		The acceleration and deceleration times cannot be set in the PLC's memory area as operating data.	
	l+13		Acceleration time (left- most word)		The acceleration/deceleration constants are set by transferring them as Servo Parameters using the	
	l+14		Deceleration time (right- most word)			
	l+15		Deceleration time (left- most word)			

b = Beginning word of Axis Operating Input Areas specified in Common Parameters + (Axis No. -1) \times 25

I/O	CS1W/CJ1W-NC113/133/213/233/413/433			CS1W/CJ1W-NC271/471/F71		
	Word	Category	Name	Word	Differences	
Output (CPU Unit to PCU)	l+16	Operat- ing data for mem- ory oper- ation	Sequence number		Memory operation functions are not provided, so this data is not allocated.	
	l+17	Operat-	Override	a+14	Override ratio (unit: 0.01%)	
	l+18	ing data for special	Teaching address		Memory operation functions are not provided, so this data is not allocated.	
	l+19	TUTICUOTIS	Not used.			
Input (PCU to CPU Unit)	l+20	PCU sta- tus	Present position (right- most word)	b+6	This word functions as a feedback present position (rightmost word). The command present position (rightmost word) is allocated in word b+8.	
	l+21		Present position (left- most word)	b+7	This word functions as a feedback present position (leftmost word). The command present position (leftmost word) is allocated in word b+9.	
	l+22		Sequence number		Memory operation functions are not provided, so this data is not allocated.	
	l+23		Output code		Output code functions are not provided, so this data is not allocated.	

Functions

The main functions of the W-series Servo Drives are explained here.

ORIGIN SEARCH

Acceleration/Deceleration Settings

The acceleration/deceleration settings for origin searches are set in the parameters for each axis in earlier PCUs, but they are set in the Servo Parameters (using Servo Parameter transfer functions) of the CS1W/CJ1W-NC271/471/F71.

CS1W/CJ1W-NC113/133/213/233/413/433

The time from the initial speed until the maximum speed is reached is set in the following axis parameters.

Origin search acceleration time (ms)

Origin search deceleration time (ms)

Each parameter setting is transferred to the PCU using data transfer functions (data transfer bit or IOWR instruction), or transferred automatically from the DM Area allocated to Special I/O Units when the PCU power is turned ON or the Unit is restarted.

CS1W/CJ1W-NC271/471/F71

The rate of increase or decrease in speed within a specified time unit is set in the following Servo Parameters.

Туре	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
Acceler- ation/ deceler- ation con- stants	Pn80A	First-step linear accel- eration constant	10,000 command units/s ²	1 to 65535	2	100
	Pn80B Second-step linear acceleration constant		10,000 command units/s ²	1 to 65535	2	100
	Pn80C Acceleration constant switching speed		100 command units/s	0 to 65535	2	0
	Pn80D First-step linear decel- eration constant		10,000 command units/s ²	1 to 65535	2	100
	Pn80E	Second-step linear deceleration constant	10,000 command units/s ²	1 to 65535	2	100
	Pn80F	Deceleration constant switching speed	100 command units/s	0 to 65535	2	0

Each parameter is set in the Servo Drive via the PCU using the PCU's Servo Parameter transfer functions (WRITE/SAVE SERVO PARAMETER Bits).

When using a trapezoidal curve (linear acceleration/deceleration) with the initial speed set to 0 (pulses/s), the set values for the origin search acceleration time and origin search deceleration time used in the earlier PCUs are converted to the following settings.

Pn80A (first-step linear acceleration constant):

- The origin search uses a first-step acceleration/deceleration curve, so this parameter setting is not used.
- Pn80B (second-step linear acceleration constant):

The maximum speed, initial speed, and origin search acceleration time in the axis parameters of earlier PCUs are converted to the value determined from the following equation.

$$Pn80B = \frac{Maximum speed - Initial speed (0)}{Origin search acceleration time \times 10} (\times 10,000 \text{ pulses/s}^2)$$

Pn80C (acceleration constant switching speed):

The origin search uses a first-step acceleration/deceleration curve, so this parameter is set to 0 (default setting).

Pn80D (first-step linear deceleration constant):

The origin search uses a first-step acceleration/deceleration curve, so this parameter setting is not used.

Pn80E (second-step linear deceleration constant):

The maximum speed, initial speed, and origin search deceleration time in the axis parameters of earlier PCUs are converted to the value determined from the following equation.

$$Pn80E = \frac{Maximum speed - Initial speed (0)}{Origin search deceleration time \times 10} (\times 10,000 \text{ pulses/s}^2)$$

Pn80F (deceleration constant switching speed):

The origin search uses a first-step acceleration/deceleration curve, so this parameter is set to 0 (default setting).

CS1W/CJ1W-NC113/133/213/233/413/433



Origin Search High Speed

The origin search high speed (speed until detection of the origin proximity input signal) is set in the axis parameters in earlier PCUs. In the CS1W/CJ1W-NC271/471/F71, however, this parameter value is set as the *Speed Command Value* for origin searches in the Axis Operating Output Memory Area when the origin search is started.

Origin Search Proximity Speed

The origin search proximity speed (speed after detection of the origin proximity input signal until detection of the origin input signal) is set in the axis parameters in earlier PCUs. In the CS1W/CJ1W-NC271/471/F71, however, this parameter value is set as the *Zero Point Return Approach Speed 1* (Pn817) in the Servo Parameter Area.

The setting unit for the *Zero Point Return Approach Speed 1* parameter is 100 command units/s. Therefore, to set the command unit to pulses, the set value for earlier PCUs must be multiplied by 1/100 to obtain the set value required for CS1W/CJ1W-NC271/471/F71 PCUs.

Origin Compensation

When using origin compensation in earlier PCUs, positioning was executed for the amount of origin compensation data at the origin search proximity speed. This operation is performed after first stopping the search operation using origin input signal detection.

For CS1W/CJ1W-NC271/471/F71 PCUs, origin compensation is performed following origin input signal detection by executing positioning using the Servo Parameters *Zero Point Return Approach Speed 2* (Pn818) and *Final Travel Distance to Return to Zero Point* (Pn819).





Origin Established and No Origin Status

The No Origin Flag in both earlier PCUs and the CS1W/CJ1W-NC271/471/F71 indicates whether the origin has been established or not. The differences in the conditions whereby the No Origin Flag turns ON or OFF are as follows:

	CS1W/CJ1W-NC113/133/213/233/413/433	CS1W/CJ1W-NC271/471/F71
Initial status (when power is turned ON)	(After PCU's initial processing is completed) ON status (origin not established)	Same as left
ON condition (Origin is lost)	 At power up, or restart When executing ORIGIN SEARCH (See note.) When the emergency stop input signal or limit input signal is input (if origin undefined designation of I/O settings (for x axis, word 0004 bit 08) in the axis parameters is 	 At power up, or restart When connection is released (if CON- NECT Bit is turned OFF or the connection is released due to an error) When DEVICE SETUP is executed
OFF condition (Origin is established)	When ORIGIN SEARCH execution is completed When PRESENT POSITION PRESET is executed	 When ORIGIN SEARCH execution is completed When PRESENT POSITION PRESET is executed When using absolute encoders: When connection is established When SERVO LOCK is executed

Note In earlier PCUs, if ORIGIN SEARCH is executed again after establishing the origin, the origin will be lost (until ORIGIN SEARCH execution is completed). In the CS1W/CJ1W-NC271/471/F71, however, once the origin has been established, the origin will not be lost, even if ORIGIN SEARCH is executed again. If ORIGIN SEARCH is executed again, the search operation will be executed using the coordinates based on the previous origin until the new origin is established. When the software limits are enabled at this point, the software limits will apply to the origin search operation, causing operation to stop due to an error. (Positioning will not be reversed at the software limits when origin searches are performed in reversal mode.)

Position Control (Direct Operation)

Acceleration/Deceleration Settings

The acceleration/deceleration settings for position control (direct operation) are specified in the Operating Data Area at start up in earlier PCUs. In CS1W/CJ1W-NC271/471/F71 PCUs, however, these settings are set in the Servo Parameters (using Servo Parameter transfer functions).

CS1W/CJ1W-NC113/133/213/233/413/433:

The following operation data is designated for the time from the initial speed until the maximum speed is reached.

Acceleration time (ms) Deceleration time (ms)

For each parameter, the data that has been set in the Operating Data Area is enabled when the direct operation (ABSOLUTE MOVEMENT or RELATIVE MOVEMENT) is started.

CS1W/CJ1W-NC271/471/F71

The rate of increase or decrease in speed within a specified time unit is set in the following Servo Parameters.

Туре	Parameter No.	Parameter name	Unit	Setting range	Parameter size	Default setting
Acceler- ation/	Pn80A	First-step linear acceleration constant	10,000 command units/s ²	1 to 65535	2	100
ation con-	Pn80B	Second-step linear acceleration constant	10,000 command units/s ²	1 to 65535	2	100
stants	Pn80C	Acceleration constant switching speed	100 command units/s	0 to 65535	2	0
	Pn80D First-step linear deceleration constant		10,000 command units/s ²	1 to 65535	2	100
	Pn80E	Second-step linear deceleration constant	10,000 command units/s ²	1 to 65535	2	100
	Pn80F	Deceleration constant switching speed	100 command units/s	0 to 65535	2	0

Each parameter is set in the Servo Drive via the PCU using the PCU's Servo Parameter transfer functions (WRITE/SAVE SERVO PARAMETER Bits).

When a trapezoidal curve (linear acceleration/deceleration) is used with the initial speed set to 0 (pulses/s), the set values for the acceleration time and deceleration time in the Operating Data Area of the earlier PCUs are converted to the following settings.

Pn80A (first-step linear acceleration constant):

The operation uses a first-step acceleration/deceleration curve, so this parameter setting is not used.

Pn80B (second-step linear acceleration constant):

The maximum speed and initial speed in the axis parameters and the acceleration time in the Operating Data Area of earlier PCUs are converted to the value determined from the following equation.

Pn80B = Maximum speed – Initial speed (0) (×10,000 pulses/s²)

Acceleration time \times 10

Pn80C (acceleration constant switching speed):

The operation uses a first-step acceleration/deceleration curve, so this parameter is set to 0 (default setting).

Pn80D (first-step linear deceleration constant):

The operation uses a first-step acceleration/deceleration curve, so this parameter setting is not used. Pn80E (second-step linear deceleration constant):

The maximum speed and initial speed in the axis parameters and the deceleration time in the Operating Data Area of earlier PCUs are converted to the value determined from the following equation.

 $Pn80E = \frac{Maximum speed - Initial speed (0)}{Deceleration time \times 10} (\times 10,000 \text{ pulses/s}^2)$

Pn80F (deceleration constant switching speed):

The operation uses a first-step acceleration/deceleration curve, so this parameter is set to 0 (default setting).



Pn80C (acceleration constant switching speed) = 0 Pn80F (deceleration constant switching speed) = 0
Changing Target Position and Changing Target Speed

The target position and target speed can be changed during axis operation in the same way for both earlier PCUs and the CS1W/CJ1W-NC271/471/F71 PCU by using the following operation.

Deceleration time \times 10

Time

(×10,000 pulses/s²)

Changing target position: Set the new target position in *Position Command Value* and send an operation command again. (Strictly speaking, the new target position is to be set in Position of the Operating Data Area for earlier PCUs while it is set in Position Command Value in the Axis Operating Output Memory Area for CS1W/CJ1W-NC271/471/F71 PCUs.)

Changing target speed: Change *Speed Command Value* during axis operation to immediately change the target speed. (Strictly speaking, the new target speed is to be set in Speed of the Operating Data Area for earlier PCUs while it is set in Speed Command Value in the Axis Operating Output Memory Area for CS1W/CJ1W-NC271/471/F71 PCUs.)

When changing the target position with a CS1W/CJ1W-NC271/471/F71 PCU, however, the Receiving Command Flag must be used to control the ON/OFF timing for the operation command bits.

Interrupt Feeding

Interrupt feeding in earlier PCUs is executed by independently operating individual start bits. With CS1W/CJ1W-NC271/471/F71 PCUs, however, interrupt feeding is an additional function for the direct operation (ABSOLUTE MOVEMENT or RELATIVE MOVEMENT).

To execute interrupt feeding, turn ON the Interrupt Feeding Designation Bit in the Axis Operating Output Memory Area and execute ABSOLUTE MOVEMENT or RELATIVE MOVEMENT.

When using earlier PCUs, interrupt feeding functions by switching from speed control to position control (positioning is performed for the amount of interrupt feeding after detection of the interrupt input signal). With CS1W/CJ1W-NC271/471/F71 PCUs, however, interrupt feeding functions by switching from one position control operation (ABSOLUTE MOVEMENT or RELATIVE MOVEMENT) to another position control operation (positioning is performed for the amount of interrupt feeding after detection of the interrupt input signal), and if an interrupt input signal is not input, the original positioning operation is completed.

Memory Operation

Memory operation functions are not provided in CS1W/CJ1W-NC271/471/F71 PCUs.

The teaching function, forced interrupt operation, and output codes, which are related to memory operations, are also not provided.

Operations that use automatic patterns and continuous patterns must be created in the user program.

Jogging

Acceleration/Deceleration Settings

In earlier PCUs, the acceleration/deceleration settings for jogging are specified as operation data in the Operating Data Area when jogging is started. In CS1W/CJ1W-NC271/471/F71 PCUs, however, these settings are set in the Servo Parameters (using Servo Parameter transfer functions).

Details on acceleration/deceleration settings are the same as those provided for position control (direct operation), so refer to the section on position control (direct operation).

Changing Target Speed

The target speed can be changed during jogging operations in the same way for both earlier PCUs and the CS1W/CJ1W-NC271/471/F71 PCU by using the following operation.

Changing target speed: Change the *Speed Command Value* parameter during jogging operations to immediately change the target speed. (Strictly speaking, the new target speed is to be set in Speed of the Operating Data Area for earlier PCUs while it is set in Speed Command Value in the Axis Operating Output Memory Area for CS1W/CJ1W-NC271/471/F71 PCUs.)

Present Position

Earlier PCUs use open-loop control, and monitor the command position (PCU's output pulses) as the present position. CS1W/CJ1W-NC271/471/F71 PCUs exchange data with Servo Drives that have a semi-closed loop configuration. Therefore, both the feedback present position (total of feedback pulses from the Servomotor's encoder) and command present position (total of PCU's position command values) are monitored as the present position.

Data Transfer

In earlier PCUs, all the parameters set by the PCU are in the Unit, and data is transferred using the PCU's data transfer functions (data transfer bit or IOWR instruction).

In CS1W/CJ1W-NC271/471/F71 PCUs, however, data is classified into three types: Common Parameters, Axis Parameters, and Servo Parameters. Common Parameters and Axis Parameters are configured in the PCU, and Servo Parameters are configured in the Servo Drive. Therefore, the transfer functions used to transfer the parameters in the PCU are different than those used to transfer the parameters in the Servo Drive.

- Common Parameters and Axis Parameters These parameters are transferred to the PCU using the data transfer bits (WRITE DATA, READ DATA, and SAVE DATA) in the Common Operating Memory Area.
- Servo Parameters
 These parameters are transferred to each axis using the Servo Parameter transfer bits (WRITE SERVO PARAMETER, READ SERVO PARAMETER, and SAVE SERVO PARAMETER) in the Axis Operating Output Memory Area. Parameters are transferred between the PLC and Servo Drive by executing these Servo Parameter transfer functions.

The IOWR and IORD instructions are not supported by the CS1W/CJ1W-NC271/471/F71 PCUs.

Appendix F

Additional Functions for the CJ1W-NCF71-MA

Overview

The following functions have been added to those of the CJ1W-NCF71 Position Control Unit to produce the CJ1W-NCF71-MA Position Control Unit.

- Interpolation axis stop mode setting for linear interpolation
- Allowed interaxial deviation setting for linear interpolation
- **Note** The CS1W-NC271/471/F71 and CJ1W-NC271/471/F71 do not have the functions described in this appendix. To use these functions, use the CJ1W-NCF71-MA. If data is backed up from a CJ1W-NCF71 Position Control Unit on a Memory Card mounted in the CPU Unit (using either the CPU Unit's simple backup function or the Position Control Unit's backup function), do not read that data to a CJ1W-NCF71-MA Position Control Unit. It will not be possible to use the additional functions if data backed up from a CJ1W-NCF71 is downloaded to a CJ1W-NCF71-MA. To move CJ1W-NCF71 setting data to the CJ1W-NCF71-MA, be sure to use the parameter transfer function or CX-Motion-NCF Support Software to read, write, and save to memory the data in the Common Operating Memory Area. Also, do not download CJ1W-NCF71-MA Memory Card backup data to the CJ1W-NCF71.

Interpolation Axis Stop Mode Setting for Linear Interpolation

With the linear interpolation function of the CJ1W-NCF71, the other interpolation axes will decelerate to a stop if an error occurs on any of the interpolation axes, an axis is stopped (deceleration stop or emergency stop), or SERVO UNLOCK is executed during linear interpolation. The CJ1W-NCF71-MA enables selecting the stop mode of the other interpolation axes when this occurs with settings in the Axis Operating Output Memory Areas of axis 2 (for the combination of axes 1 to 4) or axis 6 (for the combination of axes 5 to 8).

Operating Output Memory Area for Axis 2 and Axis 6

Name	Word	Bits	Description
Interpolation Axis Stop Mode	a+21 (Axis Operating Output Mem- ory Area for Axis 2)	00 to 03	Axis 1 stop mode in linear interpolation for combinations of axes 1 to 4.
			0: Deceleration stop (default) 1: Emergency stop 2: Servo unlock
		04 to 07	Same as above for axis 2 stop mode
		08 to 11	Same as above for axis 3 stop mode
		12 to 15	Same as above for axis 4 stop mode
	a+21 (Axis Operating Output Mem- ory Area for Axis 6)	00 to 03	Axis 5 stop mode in linear interpolation for combinations of axes 5 to 8.
			0: Deceleration stop (default) 1: Emergency stop 2: Servo unlock
		04 to 07	Same as above for axis 6 stop mode
		08 to 11	Same as above for axis 7 stop mode
		12 to 15	Same as above for axis 8 stop mode

a = Beginning word of Axis Operating Output Memory Areas in Common Parameters + (Axis No. -1) \times 25

Select the stop operation for other interpolation axes if an error occurs on any of the interpolation axes, the axis is stopped (deceleration stop or emergency stop), or SERVO UNLOCK is executed during operation of linear interpolation. The setting can be made for each axis of linear interpolation combinations of axes 1 to 4 or for each axis of linear interpolation combinations of axes 5 to 8.

If a value outside the range is set, the system will treat it as if 0 was selected (i.e., deceleration stop). The interpolation axis stop mode setting is enabled when the LINEAR INTERPPOLATION START Bit turns ON. Example:

If the beginning word of the Axis Operating Output Memory Areas is set to CIO 4000, the interpolation stop mode for linear interpolation of a combination of axes 1 to 4 will be set in a+21 (CIO 4046) of the Operating Output Memory Area of axis 2.

The first word of the Operating Output Memory Area of axis 2 is calculated as follows:

a = 4000 + (2–1) × 25 = CIO 4025

In this case, the following settings will be used for the interpolation axis stop mode of each axis from 1 to 4.

Axis 1 interpolation axis stop mode: CIO 4046.00 to CIO 4046.03

Axis 2 interpolation axis stop mode: CIO 4046.04 to CIO 4046.07

Axis 3 interpolation axis stop mode: CIO 4046.08 to CIO 4046.11

Axis 4 interpolation axis stop mode: CIO 4046.12 to CIO 4046.15

In the same way, the interpolation stop mode for linear interpolation of a combination of axes 5 to 8 will be set in a+21 (CIO 4146) of the Operating Output Memory Area of axis 6.

The first word of the Operating Output Memory Area of axis 6 is calculated as follows:

 $a = 4000 + (6 - 1) \times 25 = CIO 4125$

The following settings will be used for the interpolation axis stop mode of each axis.

Axis 5 interpolation axis stop mode: CIO 4146.00 to CIO 4146.03

Axis 6 interpolation axis stop mode: CIO 4146.04 to CIO 4146.07

Axis 7 interpolation axis stop mode: CIO 4146.08 to CIO 4146.11

Axis 8 interpolation axis stop mode: CIO 4146.12 to CIO 4146.15

Allowed Interaxial Deviation Setting for Linear Interpolation

If the deviation in feedback present positions between axes 1 and 2 or axes 5 and 6 exceeds the allowed deviation that is set in the parameters for each axis, both axes 1 and 2 or both axes 5 and 6 will be decelerated to a stop. It is thus possible to automatically stop axis operation in response to deviation that occurs, for example, because of an error for one of the axes, when bilateral operation is performed with two axes.

The allowed deviation between axes is specified in the Axis Operating Output Memory Area of axis 2 for linear interpolation operation using combinations of axes 1 to 4 and in Axis Operating Output Memory Area of axis 6 for linear interpolation operation using combinations of axes 5 to 8.

Axis Operating Output Memory Areas for Axis 2 and Axis 6

Name	Word	Bits	Contents
Allowed Interaxial Deviation	a+23 a+24 (Axis Operating Output Mem- ory Area for Axis 2)		Allowed deviation between axes 1 and 2 (a+23: lower word, a+24: upper word)
			Unit: Command unit Setting range: 0 to 2,147,483,647
			The allowed deviation between axes 1 and 2 in linear interpolation for a combination of axes 1 to 4 will be set in 32-bit signed hexadecimal. The allowed deviation between axes 1 and 2 will not be detected if the setting is 0 (default) or a negative value.
	a+23 a+24 (Axis Operating Output Mem- ory Area for Axis 6)		Allowed deviation between axes 5 and 6 (a+23: lower word, a+24: upper word)
			Unit: Command unit Setting range: 0 to 2,147,483,647
			The allowed deviation between axes 5 and 6 in linear interpolation for a combination of axes 5 to 8 will be set in 32-bit signed hexadecimal. The allowed deviation between axes 5 and 6 will not be detected if the setting is 0 (default) or a negative value.

a = Beginning word of Axis Operating Output Memory Areas in Common Parameters + (Axis No. -1) \times 25

Example:

If the beginning word of the Axis Operating Output Memory Areas is set to CIO 4000, the allowed interaxial deviation for linear interpolation of a combination of axes 1 to 4 will be set in a+23 (CIO 4048) anda+24 (CIO 4049) of the Operating Output Memory Area of axis 2.

The first word of the Operating Output Memory Area of axis 2 is calculated as follows:

a = 4000 + (2–1) × 25 = CIO 4025

In the same way, the allowed interaxial deviation for linear interpolation of a combination of axes 5 to 8 will be set in a+23 (CIO 4148) anda+24 (CIO 4149) of the Operating Output Memory Area of axis 6.

The first word of the Operating Output Memory Area of axis 6 is calculated as follows:

a = 4000 + (6–1) × 25 = CIO 4125

The allowed deviation is found by comparing the difference in feedback present positions. Make settings so that operation starts from same present positions for axes 1 and 2 or axes 5 and 6 if detecting interaxial deviation is to be enabled (i.e., if the interaxial deviation is set to a positive number). Also, if either axis 1 or 2 or either axis 5 or 6 will be used for linear interpolation, set the allowed interaxial deviation of each axis to 0 (i.e., do not detect interaxial deviation). The setting of the allowed interaxial deviation becomes enabled when the setting is written to memory.

Note The CJ1W-NCF71-MA is treated as the CJ1W-NCF71 in the CX-Programmer, CX-Motion-NCF, and other Support Software. Select the CJ1W-NCF71 to make settings for the CJ1W-NCF71-MA when using Support Software.

Α

absolute encoder operating procedure 327 origin 8, 326 origin position offset 332 overview 326 parameters 298, 299 PCU data settings 329 setup 331 absolute movement 4, 8 sample program 439 ABSOLUTE MOVEMENT Bit 94, 186 acceleration during speed control 285, 389 filter settings 280 operations 277 parameters 279, 300 settings 278 speeds 10 acceleration/deceleration curves 7, 8, 10 alarm displays 610 applications precautions xxvi area allocations 241 auxiliary functions 7, 8 acceleration/deceleration curves 7 backlash compensation 7 monitoring function 7 override 7 servo parameter transfer 7 software limits 7 torque limit 7 Axis Communications Status 21 axis communications status 94 axis communications status bits 184 Axis Communications Status Flags 249 Axis Control Status Flags Busy Flag 200, 204 Error Flag 200 No Origin Flag 200, 203 Origin Stop Flag 200, 203 PCU Positioning Completed Flag 200, 201 Receiving Command Flag 200 Servo Parameter Transferring Flag 200 Stop Execution Flag 200, 205 Warning Flag 200 axis error code 198 Axis Error Flag 22

Axis Operating Input Memory Area 77, 228, 230, 232, 235 allocations 95, 195, 605 beginning word 98 designation 18, 98 monitoring 213 overview 194 Servo Parameter Transferring Flag 23 Servo Status Flags 404 Axis Operating Memory Area 6, 22 Axis Operating Output Memory Area 77, 227, 230, 232, 234, 257 allocations 94, 186, 601 beginning word 97 designation 18, 97 Forward Rotation External Current Limit Designation Bit 354 operating commands 213 overview 185 priority 190 **RELATIVE MOVEMENT Bit 26** Reverse Rotation External Current Limit Designation Bit 354 SERVO LOCK Bit 25 SERVO UNLOCK Bit 28 Axis Parameter Area 77, 520 allocations 79 overview 101 replacing PCUs 623 Axis Parameters 10, 218 setting 20

В

backlash compensation 7 data settings 406 overview 405 procedure 405 backup functions at CPU Unit 512 at PCU 513 baud rate 35 MECHATROLINK 238 BKIR 50 BKIRCOM 50 brake output 95 Busy Flag 95, 196, 204

С

C2 master connection 101, 244 CJ1W-NC113/133/213/233/413/433 9 replacing with CJ1W-NC271/471/F71/CS1W-NC271/ 471/F71 621 CN1 43, 47, 50 CN1 (Servo Drive) 53 command present position 95, 198, 276 command response time 515 command units 275 setting example 274 commands **DECELERATION STOP 5 DEVICE SETUP 234 EMERGENCY STOP 5** PRESENT POSITION PRESET 5 **RELATIVE MOVEMENT 16, 26** Common Operating Input Memory Area allocations 599 Common Operating Memory Area 18, 77, 221, 223, 225, 248, 249, 251, 256 allocated words 33 allocations 93, 181 CONNECT Bit 21, 25 Connection Status Flag 21, 25 Data Transferring Flag 20 overview 179 SAVE DATA Bit 20 Common Operating Output Memory Area allocations 597 Common Parameter Area 77, 519 allocations 79 overview 96 replacing PCUs 623 Common Parameters 18, 19, 218 Axis Operating Input Memory Areas 19 Axis Operating Output Memory Areas 19 Beginning word of Axis Operating Input Memory Areas 19 Beginning word of Axis Operating Output Memory Areas 19 scan list setting (axes 1 and 2) 19 scan list setting (axes 3 to 16) 19 writing to PCU 19 Communications Alarm 256 communications cycle 100, 244 setting 245 communications protocol 238 communications retries 101, 244

compatible devices 6 CONNECT Bit 21, 25, 93, 181 Connection Status Flag 21, 25, 93, 183, 249 connections establishing 248 Connector Terminal Block Cables 65 control command range 7 control function parameters 89, 147, 170, 588 control functions 7 interrupt feeding 7 jogging 7 origin determination 7 position control 7 servo lock/unlock 7 speed control 7 stop functions 7 torque control 7 Control I/O Connector 43, 47, 50, 61 control I/O signals 49 input +24 VDC control power supply 49 external latch inputs 49 forward drive prohibit input 49 origin return deceleration LS 49 reverse drive prohibit input 49 output brake interlock output 50, 52 frame ground 50, 53 positioning completed output 1 50 servo ready output 50 wiring 58 components 61 control method 6 control mode switching 396 control panel installation xxviii control system configuration 272 control units 6 for position control 273 for speed control 274 for torque control 274 controlled axes 6 coordinate system 276 CPU Unit 17 CPU Unit cycle time 10 current consumption 6 cycle time

D

at CPU Unit 518

data areas 79 Data Transfer Bit 10 data transfer function 8 Data Transferring Flag 20, 93, 183 deceleration during speed control 285, 389 filter settings 280 operations 277 parameters 279, 300 settings 278 speeds 10 deceleration stop 8, 412 sample program 445 **DECELERATION STOP Bit 94, 186 DEVICE SETUP 234** timing chart 235 DEVICE SETUP Bit 94, 187 dimensions 6, 38, 39 DIP switch 34 direct operation 2, 4, 8, 26 overview 338 PCU data settings 339 procedure 339 starting 343 timing charts 344 Distribution Completed Flag 197, 209 Distribution Completed Flag/Zero Speed Flag 95

Ε

EC Directives xxviii electromagnetic fields xxv EMC Directives xxviii emergency stop 8, 10 emergency stop 414 sample program 445 EMERGENCY STOP Bit 94 EMI Standard xxviii EMS Standard xxviii encoder phase inputs 95, 198 encoder type 104 error codes 488 individual axis errors 493, 609

PCU common errors 489, 609 error counter reset 10 error detection functions 8 Error Flag 95, 196 error reset 506 ERROR RESET Bit 94, 186 errors at CPU Unit 486, 508 at PCU 486 at powerup 485 indicator display 485 **MECHATROLINK** communications 483 overview 480 PCU settings and operations 487 resetting 506, 507 troubleshooting 484 expanded monitoring 212 exponential acceleration/deceleration curve 282 exponential curve designation 94, 189 external I/O 7, 39 External I/O Status Bits 95, 211 external latch signal inputs 95, 198

F

Index

FALS instruction xxv feedback present position 95, 198, 276 FG 50 forward drive prohibited signal input 286 forward rotation current limit 94 forward rotation current limit designation 189 Forward Rotation External Current Limit Designation Bit 354 forward rotation limit input 95, 198, 286 Forward Software Limit Flag 95, 197, 211 function selection parameters 81, 133, 151, 551, 569, 592

G

General-purpose Control Cable 62 ground xxvi

I

I/O allocations 6 Axis Operating Memory Area 6 Common Operating Memory Area 6

number of write words 19 parameter size 23 position command value 26 Servo Parameter No. 23 speed command value 26 write data 23 write destination address 19 write source area 19 write source word 19 I/O parameters 82, 139, 162, 557, 580, 592 I/O signals 40 control 49 MECHATROLINK standard settings 266, 268 Servo Drive 43, 47 status 211 Individual Axis Errors 256 Initial PCU Settings 448 INP1 50 **INP1COM 50** input signal selection 1 22 input signal selection 2 22 input signal selection 3 22 inspection 510 installation 35 location xxv precautions xxviii, 37 replacing PCUs 621 interrupt feeding 5, 7, 8 data settings 352 overview 351 present position 354 procedure 351 timing chart 353 **INTERRUPT FEEDING Bit 94, 186** interrupt input signal selection 103

J

JOG Bit 94, 186 jogging 5, 7, 8 operation procedure 371 PCU data settings 372 starting 375 timing chart 375

LED indicators 32

error display 486 error displays 485 errors at MECHATROLINK device 488 errors in PCU settings and operations 487 MECHATROLINK errors 487 MECHATROLINK-II Application Module 34 PCU errors 486

Μ

MECHATROLINK 12 communications 12 control 248 ending 25 errors 254, 483, 487 settings 243 starting 21, 25, 428 status 249 stopping 28, 428 communications error 8 communications settings 10 compatible devices 238 device errors 488 overview 238 scan list 239 settings 239 specifications 238 MECHATROLINK Connector 40 MECHATROLINK-compatible devices 238 MECHATROLINK-II communications connections 57 wiring 56 high-speed field network 2 interface port 7 Servo Drives 2 synchronous communications 9 MECHATROLINK-II Application Module 17, 33 compatible versions 33 LED indicators 34 part names 33 station address 17 **MECHATROLINK-II Communications Connector 33** MECHATROLINK-II Connection Cable 17, 56 MECHATROLINK-II Terminator 17, 56 memory areas Axis Operating Input Memory Areas 77 Axis Operating Output Memory Areas 77

Axis Parameter Areas 77 Common Operating Memory Area 77 Common Parameter Area 77 Memory Card backup 512 Memory Card transfer error 183 MLK Communications Error 255 MLK Device Alarm/Warning 257 MLK Device Error 255 MLK Device Initialization Error 254 MLK Initialization Error 255 monitor 1 95, 199 monitor 1 type 94, 188, 198 monitor 2 95, 199 monitor 2 type 94, 188, 198 monitor types 213 monitoring function 7

Ν

No Origin Flag 95, 196, 203 noise xxv number of read words 93, 182 number of write words 93, 182

0

online editing xxv Operating Data Area replacing PCUs 634 operating environment xxvi precautions xxv Operating Memory Area replacing PCUs 630 option command value 94, 188 during speed control 390 during torque control 401 origin determination 7, 8 origin input signal selection 102 origin proximity input 95, 198 origin return 8, 319 data settings 319 timing chart 323 ORIGIN RETURN Bit 94, 186 origin search 5, 8, 10, 284 acceleration 284 data settings 295

deceleration 284 direction 104 operation 103 operation mode 306 operation summary 305 overview 293 parameters 301, 302 procedure 294 sample program 438 starting 314 **ORIGIN SEARCH Bit 94, 186** Origin Stop Flag 95, 196, 203 outputs precautions xxv override 5, 7, 8, 9, 94, 188 operation 376 timing chart 377 Override Enable Bit 94, 186

Ρ

parameter size 94, 189 parameters Axis Parameters 20 control status 303 part names 32 MECHATROLINK-II Application Module 33 PCU Parameters 218 reading 223 replacing PCUs 622 saving 225 transferring 220 sample program 423 writing 220 PCU Positioning Completed Flag 95, 195, 201 phase Z margin calculating 324 description 324 position command range 9 position command value 94, 187 Position Completed Flag/Speed Conformity Flag 95 position control 2, 4, 7, 8, 9, 272 control units 273 parameters 81, 137, 158, 555, 577 switching 466 positioning sample program 439 Positioning Completed Flag 197, 207 Positioning Proximity Flag 95, 197, 210

power supply xxv precautions xxvi Power Supply Unit 17 power up time 515 precautions xxiii applications xxvi general xxiv handling 511 installation 37 operating environment xxv safety xxiv system configuration 35 wiring 73 present position during interrupt feeding 354 upper/lower limits 276 present position preset 5, 8, 317 data settings 317 timing chart 318 PRESENT POSITION PRESET Bit 94, 186 present position range 9 present positions 276 Programming Console 17, 25 Programming Console Connecting Cable 17

R

radiated emission xxviii radioactivity xxv READ BACKUP DATA Bit 93, 181 read data 95, 199 READ DATA Bit 93, 181 read destination area 93, 182 read destination word 182 **READ SERVO PARAMETER Bit 94, 187** read source address 93, 182 reading data from the PCU 224 **READY 50 READYCOM 50** Receiving Command Flag 95, 195, 200 relative movement 4, 8 sample program 439 RELATIVE MOVEMENT Bit 16, 26, 94, 186 replacing PCUs Axis Parameter Area 623 Common Parameter Area 623 functions 635

installation 621 operating data 622 Operating Data Area 634 Operating Memory Area 630 PCU Parameters 622 procedure 511 system configuration 621 wiring 621 replacing Units precautions xxvii resetting errors 507 resetting warnings 507 response time 10 commands from CPU Unit 516 external input 518 reflecting Servo Drive status changes in CPU Unit 517 Restart Bit 21 restarting 21 PCU 508 reversal mode 306 using limit inputs 313 reverse drive prohibited signal input 286 reverse rotation current limit 94 reverse rotation current limit designation 189 Reverse Rotation External Current Limit Designation Bit 354 reverse rotation limit input 95, 198, 286 Reverse Software Limit Flag 95, 197, 211

S

safety precautions xxiv SAVE DATA Bit 20, 93, 181 SAVE SERVO PARAMETER Bit 22, 94, 187 saving PCU settings 20 saving data 10 saving data to the PCU 226 scan list 239 allocations 241 setting 18 S-curve acceleration/deceleration curve 282 S-curve designation 94, 189 self-diagnostic functions 8 Servo 553 Servo Drive 16.17 Servo Drive Parameters 218 transferring 219

servo gain parameters 135, 155, 553, 573 SERVO LOCK Bit 25, 94, 187 servo lock/unlock 7, 25, 28 operation 370 sample program 436 timing chart 371 Servo ON (SVON) Flag 197 Servo Parameter Area 77, 104, 521 allocations 80 control function parameters 565 function selection parameters 551 I/O and status parameters 557, 592 position control parameters 555 servo gain parameters 553 speed control parameters 556 torque control parameters 556 Servo Parameter No. 94, 189 servo parameter transfer 7 Servo Parameter Transferring Flag 23, 196 Servo Parameters 10, 218 acceleration constant switching speed 27 automatic setting 265 backup sample program 454 deceleration constant switching speed 27 first-step linear acceleration constant 27 first-step linear deceleration constant 27 limit input operations 287 reading 229 timing chart 230 saving 231 timing chart 233 second-step linear acceleration constant 27 second-step linear deceleration constant 27 setting 22 transferring 22, 226 sample program 429 writing 23, 227 timing chart 228 Servo parameters reading/writing 8 Servo Parameters Transferring Flag 95 Servo Status Flags 95, 206, 404 Distribution Completed Flag 206, 209 Forward Software Limit Flag 206, 211 Positioning Completed Flag 206, 207 Positioning Proximity Flag 206, 210 Reverse Software Limit Flag 206, 211 Speed Conformity Flag 206, 207 Speed Limit Status Flag 206, 210

torque limit 206 Torque Limit Status Flag 210 Zero Speed Flag 206, 209 SERVO UNLOCK Bit 28, 94, 187 Servomotor 17 Servomotor Encoder Cable 17 Servomotor Power Cable 17 Servomotors with absolute encoders 2 short-circuits precautions xxvi signals precautions xxvi Simple 512 simple backup function 512 simultaneous execution 191 multiple functions 192 single-direction mode 306 software limits 7 data settings 409 operation 410 overview 408 procedure 408 specifications 6 speed command range 9 speed command value 94, 187, 188 Speed Conformity Flag 197, 207 speed control 2, 5, 7, 9 acceleration/deceleration 285, 389 control units 274 monitoring 396 option command value 390 overview 385 parameters 82, 85, 138, 159, 556, 578 sample program 442 starting 385 status 396 switching 466 SPEED CONTROL Bit 94, 187 Speed Limit Status Flag 95, 197, 210 standards 6 start times 518 static electricity xxv precautions xxvii station address setting switch 34 status parameters 82, 139, 162, 175, 557, 580, 592

Stop Execution Flag 95, 196, 205

stop functions 5, 7, 8 deceleration stop 412 emergency stop 414 overview 412 timing chart 415 system configuration 3 precautions 35 replacing PCUs 621

T

target position changing 348 operation pattern 349 target speed changing 350 terminal arrangement CN1 43, 47, 50 timing charts 252 changing target position/speed 350 direct operation 344 Error Warning Flag 346 errors or warnings 346 interrupt feeding 353 jogging 375 origin search 314 override 377 servo lock/unlock 371 stop functions 415 torque control 403 torque limit 382 zero/minute travel distance 345 torque command range 9 torque command value 94, 188 torque control 2, 5, 7, 9 control units 274 monitoring 404 option command value 401 overview 398 parameters 82, 138, 160, 556, 579 sample program 443 starting 398 status 404 switching 466 timing chart 403 TORQUE CONTROL Bit 94, 187 torque limit 7, 8, 354 constant 378 overview 377 setting with operating commands 379 timing chart 382 Torque Limit Status Flag 95, 197, 210 Transfer 512 transfer cycle 100, 243 setting 245 Transfer Cycle Setting Error 255 transferring data 10 transferring programs xxv troubleshooting 501 communications errors 504 during PCU operations 502 procedure 484

U

Unit Busy Flag 93, 183 Unit classification 6 unit error code 93, 183 Unit Error Flag 22, 93, 183 UNIT ERROR RESET Bit 93, 181 unit numbers setting 17 setting switch 33

W

Warning Flag 95, 196 warnings displays 619 resetting 507 weight 6 wiring 16, 18 I/O signals 58 **MECHATROLINK-II communications 56** precautions 73 replacing PCUs 621 WRITE BACKUP DATA Bit 93, 181 write data 94, 189 WRITE DATA Bit 18, 24, 93, 181 write destination address 19, 93, 182 WRITE SERVO PARAMETER Bit 94, 187 write source area 19, 93, 182 write source word 19, 93, 182 writing data to PCU 222 W-series Servo Drives 33 alarm displays 610 compatible versions 33

Control I/O Connector 43, 47, 50 warning displays 619 W-series Servomotor 16

Ζ

Zero Speed Flag 197 Zone Data 10 zone range 9
Revision History

A manual revision code appears as a suffix to the catalog number on the front cover of the manual.



The following table outlines the changes made to the manual during each revision. Page numbers refer to the previous version.

Revision code	Date	Revised content
01	July 2004	Original production
02	September 2004	 Information was expanded and revised to include new functionality, and corrections were made, as follows: Page xii: Information on unit versions added. Page 5: Information on linear interpolation added. Pages 6, 9, 122, 164, 240, 243, 245, and 300: Information on torque corrected. Pages 64, 65, 66, 97, 98, 99, 100, 107, 371, 372, 375, and 388: Tables expanded and notes added. Page 149: Notes added. Pages 165, 233, and 241: Table revised. Page 206: Information on ABSOLUTE MOVEMENT added. Page 336 and 337: Information added. Page 342: Note added.
03	April 2005	Revisions were made throughout the manual to include new functionality for the upgrade to unit version 1.2 and information on R88D-WN -ML2 W-series Servo Drives with built-in MECHATROLINK-II communications that differs from information for R88D-WT W-series Servo Drives equipped with Yaskawa JUSP-NS115 MECHATROLINK-II Application Modules.
04	February 2006	Revisions were made throughout the manual to include new functionality for the upgrade to unit version 1.3 and to include the new CS-series PCU.
05	December 2006	Revisions were made throughout the manual to include new functionality for the upgrade to unit version 2.0 and to support SMARTSTEP Junior Servo Drives (R7D-ZN□-ML2).
06	August 2007	Pages 202 and 274: Information added on setting the transfer cycle and communications cycle when setting an absolute encoder zero point position offset for an origin search.
07	February 2008	New functionality for the upgrade to unit version 2.1 were added, additional descrip- tions were added, and errors were corrected.
08	August 2008	Added information on G-series Servo Drives (R88D-GN ML2).
		Added explanations and corrected mistakes.
09	October 2008	Added the CS1W-NC271/471, CJ1W-NC271/471, and CJ1W-NCF71-MA and corrected mistakes.
10	December 2009	Added the G5 Series Servo Drive R88D-KN -ML2 and corrected mistakes.

Revision History

OMRON Corporation Industrial Automation Company Tokyo, JAPAN

Contact: www.ia.omron.com

Regional Headquarters OMRON EUROPE B.V. Wegalaan 67-69-2132 JD Hoofddorp The Netherlands Tel: (31)2356-81-300/Fax: (31)2356-81-388

OMRON ASIA PACIFIC PTE. LTD. No. 438A Alexandra Road # 05-05/08 (Lobby 2), Alexandra Technopark, Singapore 119967 Tel: (65) 6835-3011/Fax: (65) 6835-2711 OMRON ELECTRONICS LLC One Commerce Drive Schaumburg, IL 60173-5302 U.S.A. Tel: (1) 847-843-7900/Fax: (1) 847-843-7787

OMRON (CHINA) CO., LTD. Room 2211, Bank of China Tower, 200 Yin Cheng Zhong Road, PuDong New Area, Shanghai, 200120, China Tel: (86) 21-5037-2222/Fax: (86) 21-5037-2200

Authorized Distributor:

© OMRON Corporation 2009 All Rights Reserved. In the interest of product improvement, specifications are subject to change without notice.

Cat. No. W426-E1-10

Printed in Japan 1008

WWW.BSNEW.IR